LISKOW & LEWIS

ATTORNEYS AT LAW

NEW ORLEANS, LA. 70139

ONE SHELL SQUARE PIFTIETH FLOOR TELEPHONE (504) 581-7979

TWA 810 981 5252

LAFAYETTE LA 70505

321 TRAVIS ST P 0 801 52006 TE . EPHONE (318) 232-7424

New Orleans, 70139 October 1, 1984

83820.035

BRUCE J GRECK PATRICK W GRAY DEBORAH TAHN PRICE BOBERT F HOLDEN JOE B NORMAN THOMAS M MCNAMARA JAMES N MANSFIELD JAMES N MANSFIELD III
BILLY J DOMINGUE
LAMBERT M LAPEROUSE
FRANK E MASSENG'LE
PHILIP R JONES, JAME
ANNE E TATE
WILLIAM W PUGH
PAUL B DAVID
LYNN C HANTEL
JULIE F SCHWARTZ
CHARLES B GRIFFIS III
CHARLES B GRIFFIS III JULIE E SCHWARTZ
CHARLES B GRIFFIS III
RICHARD W REVELS, JR.
MARTHA O THOMAS
JOSEPH P HEBERT
MARGYERITE A. NOONAN
DONNA H. SLOCUM
BRUCE V SCHEWE
JANE C BURGER JANE C BURGER
DAVID W LEEFE
JAMES D. Mchichael
RICHARD E ANDERSON
GEORGE W PUGH JR
WM BLAKE BENNETT
MARK A. LOWE
'PROFESSIONAL CORPORATION

73

140

Mr. John L. Rankin Minerals Management Service U.S. Department of the Interior Gulf of Mexico OCS Region Imperial Office Buidling 3301 N. Causeway Blvd. P.O. Box 7944 Metairie, LA 70010-7944

and 2623

Lease Nos OCS-G 2317, 3286, 2591, 2910, 2912, 3141, 2745, 3299, 3388, 3390, 3400, 3414, 3589, 2271, 3416, 2619, 2887, 2888, 2607, 2102, 3171, 3186, 3332, 3587, 3588, 3601, 4102, 4110, 5221, 2045, 2078, 2115, 2254, 2255, 4220, 4216, 2557

Dear Mr. Rankin:

CULLEN H LISKOW (1893-1971) AUETIN W LEWIS (1910-1974)

AULTIN W LEWIS 190-1374

WILLIAM M MEYERS'
ROBERT T JORDEN'
CHARLES C GREMILLIN'
BLLLY H MINES'
JAMES L PELLETIER'
THOMAS U HATCHAN'
JOHN M RIN'
EDWARD M'
WILLIAM R PITTS'
DONALD R ABBUNZA'
JOHN M WILSON'
CHARLES M STIN'
CHARLES M STIN'
CHARLES M STIN'
CHARLES M STIN'
PREDERICR W BRADLEY
PREDERICR W BRADLEY
PREDERICR W BRADLEY

PREDEMICR W BRADLEY'
RERRY M MASSAM'
S. GENE FENDLER'
THOMAS P GITEN'
GEORGE H ROBINSON JR'
LARREY M ROCCEL'

HOON JR

Enclosed you will find counterparts of an Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production (Louisiana) dated September 27, 1984, by MTS Li: ed Partnership and Mesa Petroleum Co. and affecting interest in the captioned OCS Leases.

In order that third parties will be put on notice as to the execution and efficacy of the enclosed instrument, please file a counterpart of the enclosed instrument, together with a copy of this letter, in each of the lease files related to each of the captioned leases. Please acknowledge that the foregoing has been accomplished as requested by signing a counterpart of this letter and returning it to the undersigned.

Thomas F. Getten

ACCOMPLISHED AS REQUESTED MINERALS MANAGEMENT SERVICE GULF OF MEXICO OCS REGION

hanDATE:

DCT 3

| STATE | OF | TE | XAS |
|---------|----|------------|-------|
| COLLNIT | | 7 P | DOTTE |

35.:

)



ACT OF COLLATERAL MORTGAGE, COLLATERAL CHATTEL MORTGAGE, PLEDGE AND ASSIGNMENT OF PRODUCTION (LOUISIANA)

BE IT KNOWN, that on this 27 B day of September, 1984, before me, the undersigned authority, a Notary Public in and for the County of Harris, State of Texas. and in the presence of the two maersigned competent witnesses, personally came and appeared: MTS LIMITED PARTNERSHIP, a Texas limited partnership (herein called the "Partnership"), being herein represented by David H. Batchelder, Vice President of Mesa Petroleum Co., General Partner of the Partnership, said person being duly authorized to act by resolutions of the Board of Directors of Mesa Petroleum Co., a certified copy of which is attached here; as Exhibit A-1, and MESA PETROLEUM CO., a Delaware corporation (herein called the "Debtor"), being herein represented by David H. Batcheider, its Vice President -Finance and Treasurer, said person being duly authorized so to act by resolutions of the Board of Directors of the Debtor, a certified copy of which is attached hereto as Exhibit A-1, (the Partnership and the Debtor herein collectively called the "Mortgagor"), who, being duly sworn, did declare and say that the Debtor is justly and truly indebted unto any future holder or holders of the Mortgage Note (as hereinafter defined), in the principal sum of \$3,000,000,000. To evidence such indebtedness, the Debtor has executed one certain collateral mortgage note for the principal sum of \$3,000,000,000, of even date herewith, made payable to the order of Bearer, due on demand at the offices of Texas Commerce Bank National Association, at 712 Main Street, Houston, Texas 77002, which note stipulates to bear interest at the rate of eighteen percent (18%) per annum from the date thereof until paid, and ten percent (10%) attorney's fees (hereinafter eferred to as the "Mortgage Note"), all as set forth in the unexecuted copy of the Mortgage Note which is attached hereto as a part hereof and identified as Exhibit A-2, which Mortgage Note, after having being paraphed "Ne Varietur" by me, said Notary Public, for identification with this instrument (hereinafter referred to as the "Mortgage"), was delivered to the Debtor who hereby acknowledges receipt thereof.

The Mortgagor further declared that the Mortgage Note is given and this Mortgage is granted for the purpose of being used as collateral security by the Debtor to secure any liability, indebtedness or obligation due any future holder or holders of the Note, direct or contingent. The Mortgage Note may be issued and pledged by the Debtor as its interest and convenience may require to secure loans and advances made or to be made or to secure the debt of the Debtor or of any third party. Upon payment

of said indebtedness, the Mortgage Note may be returned to the Debtor without extinguishment of the Mortgage herein granted, and may, at any time and as many times thereafter as the interest of the Debtor may require, be again reissued or repledged by the Debtor as collateral security, and this Mortgage shall be and remain in full force and effect to secure the Mortgage Note until the Mortgage Note has been cancelled on its face and this Mortgage has been released of record. As used herein, the term "Bank" shall mean any future holder or holders of the Mortgage Note, whether one or more.

In the event that the Mortgage Note should be placed in the hands of an attorney, after its maturity, to institute legal proceedings to recover the amount thereof, or any part thereof, in principal or interest, or to protect the interests of the holder or holders thereof, or in the event that the same should be placed in the hands of an attorney for collection, compromise, or other action, the Debtor hereby binds itself to pay the fees of the attorney who may be employed for that purpose, which lees are hereby fixed at ten percent (10%) of the amount due or sued for, or claimed or sought to be protected, preserved, or enforced.

For all purposes of this instrument, unless the context otherwise requires:

- A. "oil and gas leases" shall include oil, gas and mineral leases and shall also include subleases and assignments of operating rights.
- B. "Hydrocarbons" shall mean oil, gas, casinghead gas and other liquid or gaseous hydrocarbons.
- C. "Production Sale Contracts" shall mean contracts now in effect, or hereafter entered into by 'he Mortgagor, or the Mortgagor's predecessors in interest, for the sale, purchase, exchange or processing of Hydrocarbons produced from the lands under the oil and gar leases, mineral, overriding royalty, royalty and other interests described in Exhibit B attached hereto and made a part hereof.
- D. "lands described in Exhibit B" shall include any lands either described in Exhibit B or the description of which is incorporated in Exhibit B by reference to another instrument or document or which are covered or affected by an oil and gas lease, mineral, overriding royalty, royalty and other interests, and shall also include any lands now or hereafter unitized or pooled with lands which are either described in Exhibit B or the description of which is incorporated in Exhibit B by reference.
- E. "Related Contracts" shall mean all unitization, perating and pooling agreements and other contracts and agreements that relate to the Mortgagor's interest in the lands described in Exhibit B or the oil and gas leases described on Exhibit B.
- F. "Commercial Movables" shall mean all corporeal movable property of whatever kind and character now or hereafter owned by the Mortgagor and placed upon the lands described in Exhibit B or on a unit including all or part of the lands described in Exhibit B at which location the Mortgagor operates as a commercial and

industrial facility oil and gas wells, and which corporeal movables are used by the Mortgagor in the conduct of such activity, or are dedicated to the use and exploitation of any mineral right, it being the inantion of the Mortgagor to mortgage and affect all corporeal movables now or hereal is owned by the Mortgagor that are located from time to time on the lands described in Exhibit B or on a unit including all or part of the lands described in Exhibit B for use in the conduct of the foregoing commercial and industrial activity or are dedicated to the use and exploitation of any mineral right.

- "Operating Equipment" shall mean all surface or subsurface machinery, equipment, facilities, supplies or other property of whatsoever kind or nature (excluding drilling rigs, drill pipe, mud pumps, trucks, automotive equipment or other property taken to the premises to drill a well or for other similar temporary uses) now or hereafter located on or under any of the lands described in Exhibit B or on a unit including all or part of the lands described in Exhibit B which are used or useful for the production, treatment, storage or transportation of or exploration for hydrocarbons, including, but not by way of limitation, all oil wells, gas wells, platforms, water wells, injection wells, casing, tubing, rods, pumping units and engines, christmas trees, derricks, separators, gun barrels, flow lines, tanks, gas systems (for gathering, treating, compression, disposal and injection), chemicals, solutions, water systems (for treating, disposa; and injection), pipe, pipelines, boilers, compressors, connectors, power plants, poles, lines, transformers, starters and controllers, machine shops, tools, storage yards and equipment stored therein, buildings and camps, telegraph, telephone and other communication systems, roads, loading docks, loading racks and shipping facilities, together with all improvements, betterments and additions thereto and replacements thereof.
- H. "Mortgaged Property" shall mean the properties, rights and interests hereinafter described and defined as the Mortgaged Property.
 - I. "Indebtedness" shall have the meaning stated below.

And now, in order to secure the full, due, and punctual payment of all indebtedness and/or any extension or renewal thereof, as well as ill costs, taxes, assessments, charges, insurance premiums, attorneys' and collection fees, and other costs and indebtedness incurred or paid hereunder, including the compensation of a keeper, and any sums advanced or expenses or costs incurred by the Bank (or any receiver or keeper appointed hereunder) which are made or incurred pursuant to, or permitted by, 'he terms hereof, plus interest at the rate herein specified or otherwise agreed upon, from the date of the advances or the incurring of such expenses or costs until reimbursed, and to secure the faithful performance and observance of all obligations, agreements, covenants and stipulations contained herein and in the Mortgage Note (all of the preceding being hereinafter sometimes collectively referred to as the "Indebtedness"), the Mortgagor declares that it does by these presents mortgage, affect, pledge, assign and hypothecate to the Bank, whether the Mortgage Note may be held by the Bank as an original obligation or in pledge, (i) all of the Debtor's right, title and interest, whether now owned or hereafter acquired, in all of the hereinafter described properties, rights and interests (the presently

undivided interest of the Debtor in and to said properties, rights and interests being set forth in Exhibit B opposite the term "Mesa Working Interest") and (ii) an undivided sixty-five percent (65%) of the Partnership's right, title and interest, whether now owned or hereafter acquired, in all of the hereinafter described properties, rights and interests (the presently owned undivided interest of the Partnership in and to said properties, rights and interests being set forth in Exhibit B opposite the term "MTS Working Interest"):

- (a) the lands described in Exhibit B, and the oil and gas leases, the fee, mineral, overriding royalty, royalty and other interests which are specifically described in Exhibit B,
- (b) the Related Contracts which are specifically described in Exhibit B or which relate to any of the properties and interests specifically described in Exhibit B,
- (c) the Hydrocarbons which are in, under, upon, produced or to be produced from the lands described in Exhibit B,
 - (d) the Production Sale Contracts,
 - (e) the Related Contracts.
 - (f) the Commercial Movables.
 - (g) the Operating Equipment,
- (h) all subleases, farmout agreements, assignments of interest, assignments of operating rights, contracts, operating agreements, rights of way, franchises, privileges, permits, licenses, easements, tenements, hereditaments, appurtenances and benefits now existing or in the future obtained and incident to or in the future obtained and incident and appurtenant to any of the foregoing,
- (i) all rights in and to all lease records, well records and production records which relate to any of the foregoing, and
- (j) all other rights, titles and interests of the Mortgagor in, to and under or derived from the lands, leases and properties described in Exhibit B, even though the Mortgagor's interest therein may be incorrectly or insufficiently described or referred to in Exhibit B.

(all of the aforesaid properties, rights and interests, together will any additions thereto which may be subjected to the lien of this instrument by means of supplements hereto, being hereinafter called the "Mortgaged Property")

SUBJECT, however, to (i) the restrictions, exceptions, reservations, conditions, limitations, interests and other matters, if any, set forth or specified to in the specific descriptions of such properties and interests in Exhibit B (including all presently existing royalties, overriding royalties, payments out of production and other burdens which are specified in Exhibit B and which are taken into consideration in computing any decimal interests set forth in Exhibit B), (ii) the pledge and assignment of production contained in Article i. hereof, but only insofar and so long as said assignment of production is not inoperative under the provisions of Section 2.4 hereof, and (iii) the condition that the Bank shall not be liable in any respect for the performance of any covenant or obligation of the Mortgagor in respect of the Mortgaged Property;

TO HAVE AND TO HOLD the Mortgaged Property unto the Bank forever to secure the payment of the Indebtedness and to secure the performance of the obligations of the Mortgagor herein contained.

The Mortgagor, in consideration of the premises and to induce the Bank to make the loans above described, hereby covenants and agrees with the Bank as follows:

ARTICLE I Particular Covenants and Warranties of the Mortgagor

- 1.1 Payment of the Indebtedness. The Debtor will duly and punctually pay the Indebtedness, including, without limitation, each and every obligation owing on account of the Mortgage Note.
- Warranties. The Mortgagor warrants that (a) the oil and gas leases described in Exhibit B are valid, subsisting leases, superior and paramount to all other oil and gas leases respecting the properties to which they pertain, (b) the Mortgagor, to the extent of the interest specified in Exhibit B, has and will have good and marketable title to each property right or interest constituting the Mortgaged Property and has a good and legal right to mortgage the same to the Bank, it being understood that the Mortgagor's interest in each oil and gas lease shall exceed the Mortgagor's net interest in production from such oil and gas lease to the extent of the Mortgagor's proportionate share of the burden of all royalties, overriding royalties and other such payments out of production, (c) the Mortgaged Property is free from all encumbrances or liens whatsoever, except as may be specifically set forth in Exhibit B or as permitted by the provisions of Section 1.5(e) hereof, (d) the Mortgagor is not obligated, by virtue of any prepayment under any contract providing for the sale by the Mortgagor of Hydrocarbons which contains a "take or pay" clause or under any similar arrangement including, without limitation, any "gas balancing agreements" or to deliver Hydrocarbons at some future time without then or thereafter receiving full payment therefor, (e) each of the oil and gas leases described in Exhibit B is in full force and effect, and all royalties due and payable have been timely and cor. ectly paid, and all severance and production taxes with respect thereto have been timely and

correctly paid, and (f) all producing wells located thereon have been drilled, operated, and produced in conformity with all applicable laws, and rules, regulations, and orders of all regulatory authorities having jurisdiction, and are subject to no penalties on account of past production, and that such wells are, in fact, bottomed under and are producing from, and the well bores are wholly within, lands described in Exhibit B. The Mortgagor hereby covenants that it will warrant and forever delend the Mortgaged Property unto the Bank, against every person whomsoever lawfully claiming the same or any part thereof, will maintain and preserve the Mortgaged Property free from all encumbrances or liens whatsoever, except as may be specifically set forth in Exhibit B or as permitted by the provisions of Section 1.5(e) hereof and will maintain and greserve the lien hereby created so long as any of the Indebtedness remains unpaid. The Debtor also warrants that (w) the Partnership is a limited partnership duly organized validly existing under the laws of the State of Texas and the Debtor is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware and is duly licensed or qualified and in good standing as a foreign corporation authorized to do business in each jurisdiction in which any part of the Mortgaged Property is located, (x) the execution and delivery of this instrument are within the actual and apparent power and authority of the Debtor, as General Partner of the Partnership, and have been duly authorized by the Debtor by all necessary corporate action (no shareholder action of the Debtor being required for the Debtor to execute and deliver this instrument, on its behalf or on behalf of the Partnership), (y) each of Debtor and the Partnership is now in a solvent condition and no bankruptcy or insolvency proceedings are pending or contemplated by either the Debtor or the Partnership and (z) this instrument is the legal, valid and binding obligation of the Mortgagor enforceable against the Mortgagor in accordance with its terms, except as enforceability thereof may be limited by bankruptcy, insolvency, reorganization or moratorium or other similar laws relating to creditors' rights and by general equitable principles which may limit the right to obtain equitable remedies.

- 1.3 Further Assurances. The Mortgagor will execute and deliver such other and further instruments and will do such other and further acts as in the opinion of the Bark may be necessary or desirable to carry out more effectively the purposes of this instrument, including, without limiting the generality of the foregoing, (a) prompt correction of any defect which may hereafter be discovered in the title to the Mortgaged Property or in the execution and acknowledgement of this instrument, the Mortgage Note, or any other note or document executed in connection herewith, and (b) prompt execution and delivery of all division or transfer orders which in the opinion of the Bank are needed to transfer effectually to the Bank the assigned proceeds of production from the Mortgaged Property.
- 1.4 Taxes. Subject to the Mortgagor's right to contest the same, the Mortgagor will promptly pay all taxes, assessments and governmental charges (including but not limited to severance taxes) legally imposed upon this instrument or upon the Mortgaged Property or production therefrom or upon the interest of the Bank therein, or upon the income and profits thereof.

- 1.5 Operation of the Mortgaged Property. So long as the Indebtedness or any part thereof remains unpaid, and whether or not the Mortgagor is the operator of the Mortgaged Property, the Mortgagor shall, at the Debtor's own expense:
 - (a) Do all things necessary to keep unimpaired the Mortgagor's rights in the Mortgaged Property and not, except in the ordinary course of business, abandon any well or forfeit, surrender or release any oil and gas lease or any rights in the Mortgaged Property with respect to the Mortgaged Property without the prior written consent of the Bank;
 - (b) Cause the lands described in Exhibit B to be maintained, developed, protected against drainage, and continuously operated for the production of Hydrocarbons in a good and workmanlike manner as would a prudent operator, and in accordance with generally accepted practices, applicable operating agreements, and all applicable federal, state and local laws, rules and regulations, excepting those being contested in good faith:
 - (c) Pay, or cause to be paid, promptly as and when due and payable, all rentals and royalties payable in respect of the Mortgaged Property, and all expenses incurred in or arising from the operation or development of the Mortgaged Property and furnish the Bank as and when requested with full information with respect to such payments;
 - (d) Cause the Operating Equipment to be kept in good and effective operating condition, and all repairs, renewals, replacements, additions and improvements thereof or thereto, needful to the production of Hydrocarbons from the lands described in Exhibit B, to be promptly made;
 - (e) Cause the Mortgaged Property to be kept free and clear of liens, charges and encumbrances of every character, other than (1) the lien hereof, (2) taxes constituting a lien but not due and payable, (3) defects or irregularities in title, and liens, charges or encumbrances, which are not such as to interfere materially with the development, operation or value of the Mortgaged Property and not such as to affect materially title thereto, (4) those set forth in Exhibit B, (5) those being contested by the Mortgagor in good faith in such manner as not to jeopardize the Bank's rights in and to the Mortgaged Property, and (6) those consented to in writing by the Bank; and
 - (f) Carry in standard insurance companies and in amounts satisfactory to the Bank the following insurance: (1) workmen's compensation insurance and public liability and property damage insurance in respect of all activities in which the Mortgagor might incur personal liability for the death of or injury to an employee or third person, or damage to or destruction of another's property, and (2) to the extent such insurance is generally carried by others engaged in similar undertakings in the same general areas in which the Mortgaged Property is located, insurance in respect of the Operating Equipment (with loss payable clauses in favor of the Bank), against loss or damage by fire, lightning, hail, tornado, explosion and other similar risks.

The Bank may from time to time, in its discretion, perform any agreement of the Mortgagor under this Section 1.5 or Section 1.4 hereof which the Mortgagor shall fail to perform and take any other action which the Bank deems necessary for the maintenance or preservation of any of the Mortgaged Property or its interest therein, and the Debtor agrees forthwith to reimburse the Bank for any and all expenses of the Bank in connection with such performance or action together with interest thereon at the rate of eighteen percent (18%) per annum from the date such expenses are incurred until such expenses are reimbursed by the Debtor.

- 1.3 Recording, etc. The Mortgagor will promptly, and at the Debtor's expense, record, register, deposit and file this and every other instrument in addition or supplemental hereto in such offices and places, at such times and with such frequency as may be necessary to preserve, protect and somewith lien hereof as a first lien on real and personal property, as the case may be, and the rights and remedies of the Bank, and otherwise will do and observe all things or matters necessary or expedient to be done or observed by reason of any law or regulation of any State or of the United States of America or of any other competent authority, for the purpose of effectively creating, maintaining and preserving the lien hereof on the Mortgaged Property.
- 1.7 Sale or Mortgage of the Mortgaged Property. Except as provided in Section 1.5(e) hereof and except for sales of severed Hydrocarbons in the ordinary course of the Mortgagor's business and the lien and security interest created by this instrument, the Mortgagor will not sell, convey, mortgage, pledge or otherwise dispose of or encumber the Mortgaged Property nor any Portion thereof, nor any of the Mortgagor's right, title or interest therein, without first securing the written consent of the Bank.
- Records, Statements and Reports. The Mortgagor will keep proper books of record and account in which complete and correct entries will be made of the Mortgagor's transactions in accordance with generally accepted accounting principles and will furnish or cause to be furnished to the Bank (a) upon its request, but not more than once a year, reports prepared by an independent person or firm acceptable to the Bank concerning (1) the quantity of Hydrocarbons recoverable from the Mortgaged Property, (2) the projected income and expense attributable to the Mortgaged Property, and (3) the expediency of any change in methods of treatment or operation of all or any wells productive of Hydrocarbons, any new drilling or development, any method of secondary recovery by repressuring or otherwise, or any other action with respect to the Mortgaged Property, the decision as to which may increase or reduce the quantity of Hydrocarbons ultimately recoverable, or the rate of production thereof, (b) monthly, a report showing the gross proceeds from the sale of Hydrocarbons produced from the lands tescribed in Exhibit B (including any thereof taken by the Mortgagor for the Mortgagor's own use), the quantity of such Hydrocarbons sold, the severance, gross production, occupation, or gathering taxes deducted from or paid out of such proceeds, the number of wells operated, drilled or abandoned, and such other information as the Bank may reasonably request (upon request of the Bank, such reports referred to in clauses (a) and (b) above shall set forth

such information on a lease or unit basis), and (c) such other information concerning the business and affairs and financial condition of the Mortgagor as the Bank may from time to time reasonably request.

- 1.9 No Governmental Approvals. The Mortgagor warrants that o approval or consent of any regulatory or administrative commission or authority, or of any other governmental body, is necessary to authorize the execution and delivery of this instrument or of the Mortgage Note, or to authorize the observance or performance by the Mortgagor of the covenants contained herein or in the Mortgage Note.
- 1.10 Right of Entry. The Mortgagor will permit the Bank, or its agents, to enter upon the Mortgaged Property, and all parts thereof, for the purpose of investigating and inspecting the condition and operation thereof.
- 1.11 Qualification to do Business; Maintenance of Existence. The Mortgagor (a) will continue to be duly qualified to transact business in every state which any part of the Mortgaged Property is located and in each state where the conduct of its business requires it to be qualified and (b) will not, without the prior written consent of the Bank, consolidate or merge with any other entity.
- 1.12 Legal Proceedings. The Mortgagor will notify the Bank in writing, promptly of the commencement of any legal proceedings materially and adversely affecting the Mortgaged Property or any part thereof and will take such action as may be necessary to preserve its and the Bank's rights affected thereby; and should the Mortgagor fail or refuse to take any such action, the Bank may at its election take such action in behalf and in the name of the Mortgagor and at the Debtor's expense. Moreover, the Bank at its option may take such independent action on its own behalf as holder of rights or liens on said properties as it may in its discretion deem proper, all expenses incurred in such actions, together with interest thereon at the rate of eighteen percent (18%) per annum, also being reimbursable to it by the Debtor and being payable by the Debtor to the Bank on demand.

ARTICLE II Pledge and Assignment of Production

2.1 Pledge and Assignment. As further security for the payment of the Indebtedness and to facilitate the discharge of all indebtedness for which the Mortgage Note may be pledged as security, the Mortgagor hereby transfers, pledges, assigns, warrants and conveys to the Bank, effective as of the date hereof, at 7:00 A.M., local time, all Hydrocarbons which are thereafter produced from and which accrue to the Mortgaged Property, and all proceeds therefrom. The Mortgagor will immediately cause the title to all Hydrocarbons produced from and which accrue to the Mortgaged Property, together with the proceeds therefrom, to be delivered to the credit of the Bank. All parties producing, purchasing or receiving any such Hydrocarbons, or having such, or proceeds therefrom, in their possession for which they or others are accountable to the Bank by virtue of the provisions of this Article, are authorized and

directed to treat and regard the Bank as the assignce is a transferee of the Mortgagor and entitled in the Mortgagor's place and stead to receive such Hydrocarbons and all specieds therefrom; and said parties and each of them shall be fully protected in so treating and regarding the Bank, and shall be under no obligation to see to the application by the Bank of an such proceeds or paymen's received by it. The Mortgagor will furnish the Bank of an anames of all parties purchasing or receiving any of such Hydrocarbons or proceeds therefrom, and will promptly execute and deliver any and all transfer orders, division orders, and other instruments that may be requested by the Bank for the purpose of effectuating the piedge and assignment herounder.

- 2.2 No Liggistry of the Bank in Collecting. The Bank is hereby absolved from all liability for failure to enforce collection of any proceeds so assigned and from all other responsibility in connection therewith, except the responsibility to account to the Mortgagor for funds actually received.
- 2.3 Assignment Not a Restriction on the Bar! Rights Nothing herein contained shall detract from or limit the absolute oblights on the Debtor to make payment of the Indebtedness and of all obligations for the the Mortgage Note may be pleaser as security regardless of whether the proceeds of deed and assigned by this Article are sufficient to pay the same, and the rights upder this Article shall be in death to all other security now or hereafter existing to secure the payment of the fidebtedness and of all obligations for which the Mortgage Mote may be pledged as security.
- 2.4 Status of Pledge and Assignment. Notwithstanding the other rovisions of this Article, the Bank or any receiver or keeper appointed in judicial receedings for the enforcement of this instrument so II have the right to receive all of the Hydrocarbons herein pleaged and assigned and the proceeds the refront after the Mortgage Note or any other promissory notes of the Debtor held by the Bank have been declared due and payable in accordance with the provisions of Section 3.1 hereof. Up n any sale of the Mortgaged Property or any part thereof pursuant to Article IV hereof, the Hydrocarbons thereafter and used from the property so sold, and the proceeds therefrom, shall be included in social sale and shall pass to the gurchaser free and clear of the pledge and assignment contained in the Article.
- 2.5 Indemnity. The Debtor agrees to indemnify the Bank against all claims, actions, liabilities, judgments, costs, attorneys' fees or other charges of whatsoever kind or nature (increinafter in this section 2.5 collectively called "claims") made against or incurred by the Bank as a consequence of the assertion, either be fore or after the parament in full of the Indeptedness and of all obligations for which the Mortgage Note may be pledged as security, that the Bank received Hydrocarbons herein assigned or the or ceeds thereof claimed by third persons, and the Bank shall have the right to defend against any such claims, employing attorneys therefor, and unless furnished with reasonable indemnity, the Bank shall have the right to pay or compromise and adjust all such claims. The Debtor will incommify and pay to the Bank any and all celt amounts as may be paid in respect thereof or as may be successfully adjudged against the Bank. The obligations of the Debtor as necessfully adjudged against the Bank. The obligations of the Debtor as necessfully satisfaction 2.5 small survive the release of this instrument.

ARTICLE III Eventy of Default

- "events of default" shall occur and shall not have been ran edicate
 - (a) default, and the continuance thereof for five (5) days, in the payment of principal of or interest on any indebtednes, when the by demand or otherwise;
 - (b) any warranty or representation made herein shall prove to be untrue in any material respect:
 - (c) failure by the Mortgagor within thirt; (30) days after notice thereof from the Bank, to cure a default in the are nerformance or observance of any covenant or agreement contained in this instrument and not constituting a default in the payment of principal of or in the stupen the Mortgage Note or in the payment of any other Indebtedness; or
 - (d) failure by the Mortgagor within five (5) days after demand from the Bank to pay over to the Bank any proceeds of Hydrocarbons which are paid to the Mortgagor rather than the Bank as provided in Section 2.1 hereof;
 - (e) the title of the Mortgagor to the Mortgaged Copperty of any substantial part thereof shall become the subject matter of litigation which would or might, in the Bank's opinion, upon final determination result in substantial impairment of loss of the security provided by this instrument and upon notice by the Bank to the Mortgagor such litigation is not dismissed within thirty (30) days,

then, and in any such event, the Bank, at its option, may declare the entire unpaid principal of and the interest accrued on the Mortgage Note and all other indebtedness secured hereby to be forthwith due and payable, without any notice or demand of any kind, both notice and demand being hereby expressly waived.

ARTICLE IV Enforcement of the Security

4.1 Remedies on Default. The Mortgagor for itself, its successors and assigns, does by these presents stipulate that it shall be lawful for, and the Mortgagor hereby authorizes the Bank, upon the occurrence and during the continuance of any event of default, to cause all and singular of the Mortgaged Property to be seized and sold by executory process, without appraisement, either in its entirety or in lots or parcels as the Bank may determine, to the himsest bidder for cash, or on such terms as the Bank in such proceedings may direct; and the Mortgagor for itself, its successors and assigns, hereby acknowledges the obligations secured hereby whether now existing

or to arise hereafter and confesses judgment for the full amount of the Mortgage Note, together with all costs, attorneys' and collection fees and all other charges and Indebtedness secured hereby if the same are not paid at maturity.

- 4.2 Waiver of Appraisement, Other R ghts, etc. To the extent allowed by law, the Mortgagor hereby waives: (a) the benefit of appraisement, as provided in Articles 2332, 2336, 2723 and 2724 of the buisiana Code of Civil Procedure, and all other laws conferring the same; (b) the Jemand and three (3) days delay accorded by Articles 2639 and 2721 of the Louisiana Code of Civil Procedure; (c) the notice of seizure required by Articles 2295 and 2721 of the Louisiana Code of Civil Procedure; (d) three (3) days relay provided by Articles 2331 and 2722 of the Louisiana Code of Civil Procedure; and (e) the benefit of the other provisions of Articles 2331, 1722 and 1723 of the Louisiana Code of Civil Procedure and the benefits of any other articles or laws relating to rights of appraisement, notice or delay not specifically mentioned above; and the Mortgagor expressly agrees to the immediate seizure of the Mortgaged Property in the event of suit hereon.
- 4.3 <u>Judicial Proceedings</u>. Upon the occurrence of any event of default and if such event shall be continuing, the Bank may proceed by a suit or suits in equity or at law, whether for a foreclosure hereunder, or for the sale of the Mortgaged Property, or for the specific performance of any covenant or agreement herein container, or in aid of the execution of any powers herein granted, or for the appointment of a receiver or a keeper pending any foreclosure hereunder or the gale of the Mortgaged Property, or for the enforcement of any other appropriate legal or equitable remedy.
- 4.4 Certain Aspects of a Sale. The Bank shall have the right to become the purchaser at any sale held by a court, receiver or public officer, and the Bank so purchasing shall have the right to credit the amount of the bid made therefor upon the amount payable car of the net proceeds of such sale to it. Recitals contained in any conveyance made to any purchaser at any sale made hereunder shall conclusively establish the tacts and accuracy of the matters therein stated, including, without limiting the generality of the foregoing, compayment of the unpaid principal sum of, and the interest accrued on, the Mortgage Note after the same shall have become dispand payable, and advertisement and conduct it such sale in the manner provided herein.
- 4.5 Receist a Purchaser. Upon any sale, the receipt of the Bank, or of the officer making sale under judicial proceedings, shall be sufficient discharge to the purchaser or purchaser. At any sale for his or their purchase money, and such purchaser or purchasers and his or their assigns or possonal representatives shall not, after paying such purchase money and receiving such receipt of the Bank or of such officer therefor, be obliged to see to the application of such purchase money, or be in any way arswerable for any loss, misapplication or nonapplication thereof.
- 2.6 Effect of Sale. Any sales or sites of the Mortgaged Property shall operate to direct all right, time, interest, claim and demand whatsoever a new at law

or in equity, of the Mortgagor of, in and to the premises and the property sold, and shall be a perpetual bar, both at law and in equity, against the Mortgagor, its successors and assigns, and against any and all persons claiming or who shall thereafter claim all or any of the property sold from, through or under the Mortgagor, its successors and assigns; nevertheless he Mortgagor, if requested by the Bank so to do, shall join in the execution and delivery of all proper conveyances, assign... ats and transfers of the properties so sold.

Operation of the Mortgaged Property by the Bank. occurrence of an event of default, and in addition to all other rights herein conferred on the Bank, the Bank or its agent is baceby appointed a keeper of the Mortgaged Property pursuant to the terms and provisions of Louisiana Revised Statutes 9:5131, et seq. The keeper may operate the same without any liability to the Mortgagor in connection with such operations, except to use ordinary care in the operation of said properties, and the keeper, to the extent permitted by law, shall have the right (a) to enter into and upon and take possessica of the Mortgugad Property, to lease the same. collect and receive all rents, issues and profits there of and apply the same, less the necessary expenses of collection thereof, for the care, operation and preservation of the Mortgaged Property, including without limitation, the payment of dees, insurance premiums, cost of operation of the Mortgaged Property, taxes, assess ments, interest, penalties and water charges; (b) to collect, receive and receipt for all Hydrocarbons produced and sold from the Mortgaged Property, to make repairs, purchase machinery and equipment, con fuct work-over operations, and drill additional wells; (c) to exercise every power, right and privilege of the Mortgagor with respect to the Mortgaged Property; and (d) to exercise every power, right and privilege now or hereafter granted to a keeper of the Mortgaged Property under applicable law. The compensation for the services of the keeper are hereby fixed at five percent (5%) of the amount due or sold for or claimed or sought to be protected, preserved or enforced in the proceeding for the recognition of this Martgage and, together with all costs, expenses and liabilities of every character incurred by the Bank in managing, operating and maintaining the Mortgaged Property, as Keeper or otherwise, and interest thereon at the rate of eighteen percent (18%) per annum, shall be secured by the rivilege and mortgage he em granted.

ARTICLE V Miscellaneous Provisions

5.1 Defense of Claims. The Mortgagor will notify the Bank, in writing, promptly of the commencement of any legal proceedings affecting the lien hereof or the Mortgaged Property, or any part thereof, and will take such action, employing cromeys agreeable to the Bank, as may be necessary to preserve the Mortgagor's or the Bank's rights affected thereby; and should the Mortgagor fail or refuse to take any such action the Bank may take such action in behalf and in the name of the Mortgagor and at the Debton's expense. Moreover, the Bank may take such independent action in connection the with as it may in its discretion doesn proper, the Debtor hereby agreeing the all sums advanced or all expenses incurred in such actions, plus interest

at the rate of eighteen percent (18%) per annum, will, on demand, be reimbursed to the Bank.

- 5.2 Costs and Expenses. All costs, expenses and advances (including, without limitation, cost of defense of claims, attorneys' fees, taxes, insurance, and operation cost) incurred or paid by the Bank in preserving and protecting the Mortgaged Frozerty and in preserving, protecting and enforcing its right; hereunder, all to the extent the Bank is given such right herein, up to twenty percent 20%) of the face amount of the Mortgage Note, shall constitute a demand obligation owing by the Debtor to the Bank and shall draw interest at the rate of eighteen percent (15%) per annum, all of which shall constitute a portion of the Indebtedness.
- 5.3 Renewals, Amendments and Other Security. Renewals and extensions of the indebtedness may be given at any time and amendments may be made to agreements relating to any part of such Indebtedness or the Mortgaged Property and the Bank may take or may now hold other security for its Indebtedness without notice to or consent of the Mortgagor. The Bank may resort first to such other security or any part thereof or first to the security herein given or any part thereof, or from time to time to either or both, even to the partial or complete abandonment of either security, and such action shall not be a waiver of any rights conferred by this Mortgage, which shall continue as a first lien upon the Mortgaged Property not expressly released until the Mortgage Note and all other indebtedness secured hereby is fully paid.
- 5.4 <u>Limitation on Interest</u>. No provision of this instrument or of the Mortgage Note shall require the payment or permit the collection of interest in excess of the maximum permitted by law or which is otherwise contrary to law. If any excess of interest in such respect is herein or in the Mortgage Note provided for, or shall be adjudicated to be so provided for herein or in the Mortgage Note, the Debtor shall not be obligated to pay such excess.
- 5.5 <u>Unenforceable or Inapplicable Provisions</u>. If any provision hereof or of the Mortgage Note is invalid or unenforceable in any jurisdiction, the other provisions hereof or of the Mortgage Note shall remain in full force and effect in such jurisdiction, and the remaining provisions hereof shall be liberally construed in favor of the Ferix in order to effectuate the provisions hereof, and the invalidity of any provisions hereof in any jurisdiction shall not affect the validity or enforceability of any such provision in any other jurisdiction.
- 5.6 Advances by the Bank. Each and every covenant herein contained shall be performed and kept by the Mortgagor solely at Debtor's expense. If the wortgagor shall fail to perform or keep any of the covenants of whatsoever kind or nature contained in this instrument, the Bank or any receiver or keeper appointed hereunde may, but shall not be obligated to, make advances to perform the same in the Mortgagor's behelf, and the Debtor hereby agrees to repay such sums upon demand plus interest at the rate of eighter percent (18%) per annum or, in the event any promissory note evidences such indebtedness, upon the terms and conditions thereof. No such advance shall be deemed to relieve the Mortgagor from any default hereunder.

- Rights Cumulative. Each and every right, power and remedy herein given to the Bank shall be cumulative and not exclusive; and each and every right. power and remedy whether specifically herein given or otherwise existing may be exercised from time to time and so often and in such order as may be deemed expedient by the Bank, and the exercise, or the beginning of the exercise, of any such right. power and remedy shall not be deemed a waiver of the right to exercise, at the same time or thereafter, any other right, power or remedy. No delay or omission by the Bank in the exercise of any right, power or remedy shall impair any such right, power or re nedy or operate as a waiver thereof or of any other right, power or remedy then or thereafter existing. The Bank may resort to any security given by this Mortgage or to any other security now existing or hereafter given to secure the payment of the Indebtedness secured hereby or any indebtedness secured by a pledge of the Mortgage, in whole or in part, and in such portions and in such order as may seem best to the Bank in its sole discretion, and any such action shall not in waywise be considered as a waiver of any of the rights, benefits, or liens evidenced by this instrument.
- 5.8 Waiver by the Bank. Any and all covenants in this instrument may from time to time by instrument in writing signed by the Bank be waived to such extent and in such manner as the Bank may desire, but no such waiver shall ever affect or impair the Bank's rights or liens hereunder, except to the extent specifically stated in such written instrument.
- 5.9 <u>Successors and Assigns</u>. This instrument is binding upon the Mortgagor, the Mortgagor's successors and assigns, and shall inure to the benefit of the Bank, its successors and assigns, and the provisions hereof shall likewise be covenants running with the land.
- 5.10 Article and Section Headings. The article and section headings in this instrument are inserted for convenience of reference and shall not be considered a part of this instrument or used in its interpretation.
- 5.11 Construction. This instrument is in all respects to be construed under the laws of the State of Louisiana as a special mortgage, hypothecation, piedge and assignment and confession of judgment by the Mortgagor in favor of and for the benefit of the Bank, to secure the payment of all Indebtedness and the performance of all obligations of the Mortgagor.
- 5.12 Waiver. The parties hereto waive the production of any mortgage, conveyance, and tax certificates and agree to hold the undersigned Notary harmless in the premises.
- 5.13 Notices Any notice, request, demand or other instrument which may be required or permitted to be given or served upon the Mortgagor shall be sufficiently given when mailed by first class mail, addressed to the Mortgagor at the address shown below its signature at the end of this instrument or to such different address as the Mortgagor shall have designated by written notice received by the Bank.

5.14 Non-Resource Liability of Partnership and Limited Partners. Nothing herein contained shall be construed as an assumption of the Mortgage Note by the Partnership, Texaco, Inc. ("Texaco") or Sequoia Petroleum, Inc. ("Sequoia"), as limited partners of the Partnership, or as otherwise causing the Partnership, Texaco or Sequoia to incur any liability for or arising out of the Mortgage Note or the Execution of this Mortgage. Anything contained herein to the contrary notwithstanding, the Bank shall look solely to the Debtor and the Mortgaged Property for payment of the Indebtedness and no recourse shall be had for the payment of the indebtedness or for any claim based thereon or otherwise in respect thereof or based on or in respect of this Mortgage against the Partnership, Texaco and/or Sequoia or their respective successors and assigns for any deficiency or any other sum owing in respect of the Indebtedness or arising under or with respect to this Mortgage; provided, however, that the foregoing provisions of this Section 5.14 shall not prevent recourse to the Mortgaged Property nor limit the right of any person to name the Partnership or any transferee of any interest in the Mortgaged Property as a party defendant in any action or suit for a judicial foreclosure of or in the exercise of any other remedy under this Mortgage, so long as no judgment in the nature of a deficiency judgment or seeking personal liability shall be asked for or (if obtained) enforced against such Partnership or against such transferee.

NOW, PERSONALLY INTERVENES

acting on behalf of any future holder or holders of the Mortgage Note and hereby accepts this Mortgage.

THUS DONE AND PASSED, in multiple originals, before me, the undersigned Notary Public, in and for the County of Harris, State of Texas, in the presence of the undersigned competent witnesses, who have hereunto signed their names with said appearers and me, said Notary Public, after due reading of the whole, on the date hereinabove written.

MESA PETROLEUM CO.

By David H. Batchelder,

Vice President - Finance and Treasurer

[Corporate Seal]

ATTEST:

MTS LIMITED PARTNERSHIP

By: Mesa Petrolcum Co., General Partner

By: Oavid H. Batchelle

David H. Batchelder, Vice President - Finance and Treasurer

[Corporate Seal]

ATTEST:

James J. Davis, Assistant Secretary

Address: One Mesa Square Amarillo, Texas 79189

A Lecred Hivier, K., Intervenor

WITNESSES TO ALL SIGNATURES:

Notary Public

In and for the State of Texas, Commission Expiring

DEBORAH SERNA

This instrument was prepared by:

Thomas L. Healey Andrews & Kurth

4200 Texas Commerce Tower

Houston, Texas 77002

EXHIBIT A-1
to
Act of Collateral Mortgage,
Collateral Chattel Mortgage,
Pledge and Assignment of Production

CERTIFIED RESOLUTIONS OF MESA PETROLEUM CO.

I hereby certify that attached hereto is a multiple original of certified resolutions of MESA PETROLEUM CO. (the "Company"), certified by the Assistant Secretary of the Company, and attached to an Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production (Louisiana), passed before me this 2772 day of September, 1984.

Notary Public

In and for the State of Texas, Commission Expiring

DEBORAH SERNA MY COMMISSION EXPIRES

EXHIBIT A-1
to
Act of Collateral Mortgage,
Collateral Chattel Mortgage.

Collateral Chattel Mortgage, Pledge and Assignment of Production

Certificate of Mesa Petroleum Co.

- I, the undersigned, Assistant Secretary of MESA PETROLEUM CO., a Delaware corporation (the "Company"), DO HEREBY CERTIFY that:
- 1. This Certificate is furnished pursuant to Section 16.1(e) of that certain Mesa Credit Agreement dated as of September 20, 1984 (the "Credit Agreement"), among the Company, the banks parties thereto (collectively, the "Banks"), Texas Commerce Bank National Association ("TCB"), as collateral agent for the Banks, Mellon Bank, N.A. ("Mellon"), as administrative agent for the Banks, and TCB and Mellon, as agents for the Banks. Unless otherwise defined herein, capitalized terms used in this Certificate have the meanings assigned to such terms in the Credit Agreement.
- There have been no amendments to the Restated Certificate of Incorporation of the Company since July 31, 1984.
- Attached hereto as Exhibit A is a true, correct and complete copy of the by-laws of the Company (and any amendments thereto) as in effect on the day hereof.
- 4. Attached hereto as Exhibit B is a true, correct and complete copy of resolutions duly adopted at a meeting of the Board of Directors of the Company, convened and held on the 18th day of September, 1984, which resolutions have not been revoked, modified, amended or rescinded and are still in full force and effect, and the Credit Agreement, the Notes, the MTR Pledge Agreement and the Mortgages are in substantially the forms of those documents submitted to and approved by the Board of Directors of the Company.
- 5. David H. Batchelder has been duly elected, has duly qualified as and at all times since August 24, 1984 (to and including the date hereof) has been the Vice President Finance and Treasurer of the Company and the signature set forth below is his genuine signature.

Dar- Il Satelle

I know of no proceeding for the dissolution or liquidation of the Company or threatening its existence.

Certificate of Mesa Petroleum Co.

WITNESS my hand and the seal of the Company this 27th day of September, 1984.

James J. Davis, Assistant Secretary

[Corporate Seal]

I, the undersigned, Vice President of the Company, DO HEREBY CERTIFY that James J. Davis is a duly elected and qualified Assistant Secretary of the Company and the signature above is his genuine signature.

David H. Batchelder, Vice President-Finance and Treasurer

Exhibit A

MESA PETROLEUM CO.

(a Delaware corporation)

BYLAWS

As amended through April 26, 1983

MESA PETROLEUM CO.

(a Delaware corporation)

BYLAWS

As amended through April 26, 1983

ARTICLE I

Offices

Section 1. Principal Office. The principal office of Mesa Petroleum Co. (hereinafter called the Corporation) in the State of Delaware shall be in the City of Wilmington, County of New Castle, and the resident agent in charge thereof shall be The Corporation Trust Company.

Section 2. Other Offices. The Corporation may have an office or offices at such other place or places, either within or without the State of Delaware, as the Board of Directors may from time to time determine or as shall be necessary or appropriate for the conduct of the business of the Corporation.

ARTICLE II

Meetings of Stockholders

Section 1. Place of Meeting. All meetings of the stockholders of the Corporation shall be held at such place or places, within or without the State of Delaware, as may from time to time be fixed by the Board of Directors, or as shall be specified or fixed in the respective notices or waivers of notice thereof.

Section 2. Annual Meetings. The annual meeting of the stockholders of the Corporation for the election of directors and for the transaction of such other business as may come before the meeting shall be held on the fourth Wednesday in April of each year, if not a legal holiday, and if a legal holiday, then on the next succeeding business day not a legal holiday at such hour as may be specified in the notice of said meeting, or at such other date as the Board of Directors may direct. If the election of directors shall not be held on the day designated herein for any annual meeting, or at any adjournment thereof, the Board of Directors shall cause a special meeting of the stockholders to be held as soon thereafter as conveniently may be for the election of such directors. At such special meeting the stockholders may elect directors and transact other business with the same force and effect as at an annual meeting of the stockholders duly called and held.

Section 3. Special Meetings. A special meeting of the stockholders, or of any class entitled to vote, for any purpose or purposes, may be called at any time by the Chairman of the Board, the President or by order of a majority of the Board of Directors.

Section 4. Notice of Meetings. Except as otherwise expressly required by law, notice of each meeting of stockholders, whether annual or special, shall be given at least ten (10) days before the date on which the meeting is to be held, to each stockholder of record entitled to vote thereat by delivering a type-written or printed notice thereof to him personally, or by mailing such notice in a postage prepaid envelope directed to him at his address as it appears on the stock book of the Corporation, unless he shall have filed with the Secretary of the Corporation a written request that notices intended for him be directed to another address, in which case such notice shall be directed to him at the address designated in such request. Except when expressly required by law, no publication of any notice of a meeting of the stockholders shall be required. Every notice of a special meeting of the stockholders, besides stating the time and place of the meeting, shall state briefly the objects or purposes thereof. Notice of any meeting of stockholders shall not be required to be given to any stockholder who shall attend such meeting in person or by proxy; and, if any stockholder shall, in person or by attorney thereunto authorized, in writing or by telegraph, cable or wireless, waive notice of any meeting of the stockholders, whether prior to or after such meeting, notice thereof need not be given to him. Notice of any adjourned meeting of the stockholders shall not be required to be given, except where expressly required by law.

1

Section 5. List of Stockholders. It shall be the duty of the Secretary or other officer of the Corporation who shall have charge of the stock ledger, either directly or through a transfer agent appointed by the Board, to prepare and make, at least ten (10) days before every election of directors, a complet wist of the stockholders entitled to vote thereat, arranged in alphabetical order. Such list shall be open to the examination of any stockholder at the place where said election is to be held for said ten (10) days, and shall be produced and kept at the time and place of the election during the whole time thereof, and subject to the inspection of any stockholder who may be present. The original or a duplicate stock ledger shall be the only evidence as to who are the stockholders entitled to examine such list or the books of the Corporation or to vote in person or by proxy at such election.

Section 6. Quorum. At each meeting of the stockholders, the holders of renord of a majority of the issued and outstanding stock of the Corporation entitled to vote at such meeting, present in person or by proxy, shall constitute a quorum for the massaction of business, except where otherwise provided by law. In the absence of a quorum, any officer each at to preside at, or act as Secretary of, such meeting, shall have the power to adjourn the meeting from time to take until a quorum shall be constituted. At any such a djourned meeting at which a quorum shall be present any be massacted which might have been transacted at the meeting as orginally called.

Section 7. Voting. At each meeting of stockholders each holder of record of the issued and outstanding stock of the Corporation entitied to vote at such meeting shall be entitled to one vote in person or by proxy on each matter for each such share of stock of the Corporation registered in his name on the books of the Corporation.

- (a) on the date fixed pursuant to Se sion 5 of Article VI of these Bylaws as the record date for the determination of stockholders entitled to notice of a. ... sote a such meeting; or
 - (b) if no such record date shall have been fixed, then at the date of such meeting,

provided, however, that, except where the transfer books of the Corporation shall have been closed or such a record date fixed, no share of stock of the Corporation shall be voted on at any election of directors which shall have been transferred on the books of the Corporation within twenty (20) days next preceding such election of directors. Shares of its own capital stock belonging to the Corporation shall not be voted upon directly or indirectly. At all meetings of the stockholders, a quorum being present, all matters, except those the manner of deciding upon which is otherwise provided by Articles Tenth, Thirteenth and Fourteenth of the Restated Certificate of Incorporation or by law, shall be decided by the affirmative votes of the holders of record of a majority of the issued and outstanding shares of stock of the Corporation entitled to vote the eat. Unless demanded by a stockholder of the Corporation present in person or by proxy at any meeting of the stockholders and entitled to vote thereat or so directed by the Chairman of the meeting, the vote thereat on any question need not be by ballot. Upon a demand of any such stockholder, or by direction of the Chairman of the meeting, for a vote by ballot on any question, such vote shall be taken. On a vote by ballot, each ballot shall be signed by the stockholder voting, or in his name by his proxy, if there be such proxy, and shall state the number of shares voted by him.

ARTICLE III

Board of Directury

Section 1. General Powers. The projecty, business and affairs of the Corporation shall be managed by the Board of Directors.

Section 2. The Yumber and Term of Office. The number of directors which shall constitute the whole Board of Directors of the Corporation shall not be more than nine (9), except in the case of an increase in the number of directors by reason of any details provisions with respect to any outstanding series of Preferred Stock. The number of directors which small comprise the Board of Directors shall be set by resolution of the Board of Directors. The Board is divided into three classes. Class I, Class II and Class III. The number of directors in each class shall be the whole number contained in the quotient arrived at by dividing the authorized number of directors by three and if a fraction is also a mained in such quotient, then if such fraction is one-third (1/2) the extra director shall be a member of Class III and I the fraction is two-thirds (1/2) one of the directors shall be a member of Class III and the other shall be a member of Class II. Each director shall serve for a term ending on the third annual meeting following the annual meeting at which such director was elected, provided, however, that the directors first elected to Class II

shall serve for a term ending on the annual setting next ending, the director first elected to Class II shall serve for a term ending on the second annual meeting following the meeting at which such directors were first elected, and the directors first elected to Class II shall serve a full term as hereinabove provided. The foregoing notwithstanding, each director shall serve until his specessor thalf have been duly elected and qualified, unless he shall resign, become disqualified, disabled or shall other this experiment. For purposes of the preceding paragraph, reference to the first election of directors shall sign by the first election of directors concurrent with the approval by stockholders of the Restated Certificate of Incorporation. At each annual election held thereafter, the directors chosen to succeed those whose terms then expire shall be identified as being of the same class as the directors they succeed. If for any reason the number of directors in the various classes shall not conform with the formula set forth in the preceding paragraph, the Board of Directors may redesignate any director into a different class in order that the balance of directors in such classes shall conform thereto. The directors to be elected for the year shall be elected at the Annual Meeting of stockholders, except as provided in Section 10 of this Article, and each director elected shall hold office until his successor is elected and qualified. Directors need not be stockholders.

- Section 3. Quorum and Manner of Acting. The greater of (a) four directors, or (b) a majority of the directors at anytime in office, shall constitute a quorum for the transaction of business, and if at any meeting of the Board of Directors there shall be less than such a quorum, a majority of those present may adjourn the majority from time to time. Every act or decision done or made by a majority of the directors present at a meeting which a quorum is present shall be regarded as the act of the Board of Directors unless a greater number by law or by the Restated Certificate of Incorporation. Notice of any adjourned meeting need not be give.
- Section 4. Place of Meetings, Etc. The Board of Directors may hold its meetings, have one of more offices, and keep the books and records of the Corporation, at such place or places within or without the State of Delaware, as the Board may from time to time determine.
- Section 5. First Meeting. After each annual election of directors and on the same day or as soon thereafter as convenient, the Board of Directors shall meet for the purpose of organization, the election of officers and the transaction of other business at the place where such annual election is held. Notice of such meeting need not be given. Such meeting may be held at any other time or place as shall be specified in a notice given as hereinafter provided for special meetings of the Board of Directors or in a consent and waiver of notice thereof signed by all the directors.
- Section 6. Regular Meetings. Regular meetings of the Board of Directors shall be held at such places and at such times as the Board shall by resolution determine. Notice of regular meetings need not be given.
- Section 7. Special Meetings; Notice. Special meetings of the Board of Directors shall be held whenever called by the Chairman of the Board, the President or by any three (3) of the directors. Notice of each such meeting shall be mailed to each director, addressed to him at his residence or usual place of business, at least two (2) days before the day on which the meeting is to be held, or shall be sent to him at such place by telegraph, cable, radio or wireless, or be delivered personally or by telephone, not later than the day before the day on which such meeting is to be held. Each such notice shall state the time and place of the meeting but need not state the purposes thereof. Notice of any meeting of the Board need not be given to any director, however, if waived by him in writing or by telegram, cable, radio, or wireless whether before or after such meeting be held, or if he shall be present at such meeting, and any meeting of the Board shall be a legal meeting without any notice thereof having been given, if all the directors then in office shall be present thereat.
- Section 8. Resignation. Any director of the Corporation may resign at any time by giving written notice to the Chairman of the Board, the President or the Secretary of the Corporation. The resignation of any director shall take effect upon receipt of notice thereof or at such later time as shall be specified in such notice; and, unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.
- Section 9. Removal. Any director may be removed at any time, but only for cause (except as provided in subparagraph (B) of paragraph 7 of Part A of Article Fourth of the Restated Certificate of Incorporation), by the affirmative votes of the holders of record of a majority of the issued and outstanding stock of the Corporation,

entitled to vote thereon, given at a special meeting of the stockholders called for that purpose; and the vacancy in the Board of Directors caused by any such removal may be filled by the stockholders at such meeting.

Section 10. Vacancies. Any vacancy that shall occur in the Board of Directors by mason of death, resignation, disqualification or removal or any other cause whatever, and newly created directorships resulting from any increase in the authorized number of directors, may be filled by a majority of the remaining directors (though less than a quorum) or by the stockholders of the Corporation at the next annual meeting or any special meeting called for the purpose, and each director so chosen shall hold office until the next annual election at which directors of his class shall be elected and until his successor shall be newly elected and shall qualify, or until his death, resignation, or removal.

Section 11. Compensation of Directors. The directors shall be entitled to be reimbursed for any reasonable expenses paid by them on account of attendance at any regular or special meeting of the Board, and the Board of Director may provide to the Corporation shall pay each director such compensation for his services as such as may be fixed by resolution of the Board. Nothing herein contained shall be construed to preclude any director from serving the Corporation or any subsidiary thereof in any other capacity and receiving compensation therefor.

Section 12. Committees. The Board of Directors, by resolution or resolutions passed by a majority of the whole Board, may designate members of the Board to constitute one or more committees, which shall in each case consist of such number of directors and less than 2), and shall have and may exercise such powers of the Board in the management of the business and affairs of the Corporation (including the power to authorize the seal of the Corporation to be a fixed to all papers which may require it), as the Board of Directors may by resolution determine and specify in the respective resolutions appointing them, subject to such restrictions as may be contained in the Restated Certificate of Incorporation. Such committee or committees shall have such name or names as may be determined from time to time by resolution adopted by the Board of Directors. A majority of all the members of any such committee may fix its rules of procedure, determine its action and fix the time and place, whether within or without the State of Delaware, of its meetings and specify what notice thereof, if any, shall be given, unless the Board of Directors shall otherwise by resolution provide. The Board of Directors shall have power to change the membership of any such committee at any time, to fill vacancies therein and to discharge any such committee, either with or without cause, at any time. Each member of any such committee shall be paid such fee, if any, as shall be fixed by the Board of Directors for each meeting of such committee which he shall attend and, in addition, such transportation and other expenses actually incurred by him in going to the meeting of such committee and returning therefrom as the Board of Directors shall approve.

Section 13. Indemnification of Officers, Directors, Employees and Agents; Insurance.

- (a) The Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation) by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by him in connection with such action, suit or proceeding if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his conduct was unlawful.
- (b) The Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against expenses (including attorneys' fees) actually and reasonably incurred by him in connection with the defense or settlement of such action or suit if he acted in good faith and in a

manner he reasonably believed to be in or not opposed to the best interests of the Corporation and except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable for negligence or misconduct in the performance of his duty to the Corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

- (c) To the extent that a director, officer, employee or agent of the Corporation has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in subsections (a) and (b), or in defense of any claim, issue or matter therein, he shall be indemnified against expenses (including attorneys' fees) acrually and reasonably incurred by him in connection therewith.
- (d) Any indemnification under subsections (a) and (b) (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the director, officer, employee or agent is proper in the circumstances because he has met the applicable standard of conduct set forth in subsections (a) and (b). Such determination shall be made (1) by the Board of Directors by a majority vote of a quorum consisting of directors who were not parties to such action, suit or proceeding, or (2) if such a quorum is not obtainable, or, even if obtainable a quorum of disinterested directors so directs, by independent legal counsel in a written opinion, or (3) by the stockholders.
- (e) Expenses incurred in defending a civil or criminal action, suit or proceeding may be paid by the C my praction in advance of the final disposition of such action, suit or proceeding as authorized by the Board of Directors in the specific case upon receipt of an undertaking by or on behalf of the director, officer, employee or agent to repay such amount unless it shall ultimately be determined that he is entitled to be indemnified by the Corporation as authorized in this Section.
- (f) The indemnification provided by this Section shall not be deemed exclusive of any other rights to which those seeking indemnification may be entitled under any by-law, agreement, vote of stockholds or disinterested directors or otherwise, both as to action in his official capacity and as to action in another capacity while holding such office, and shall continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.
- (g) The Corporation shall have power to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against him and incurred by him in any such capacity, or arising out of his sixtus as such, whether or not the Corporation would have the power to indemnify him against such liability under the provisions of the Delaware Corporation law.
- (h) For the purposes of this Section, references to "the Corporation" include all constituent corporations absorbed in a consolidation or merger as well as the resulting or surviving corporation so that any person who is or was a director, officer, employee or agent of such constituent corporation or is or was serving at the request of such constituent corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise shall stand in the same position under the provisions of this Section with respect to the resulting or surviving corporation as he would if he had served the resulting or surviving corporation in the same capacity.
- Section 14. Action Without a Meeting; Participation by Telephone. Any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all members of the Board or of such committee, as the case may be, consent thereto in writing, and such written consent is filed with the minutes of proceedings of the Board or of such committee. Members of the Board of Directors, or any committee designated by such Board, may participate in a meeting of such Board or committee by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting pursuant to this Section shall constitute presence in person at such meeting.

ARTICLE IV

Officers

- Section 1. Number. The principal officers of the Corporation shall be a Chairman of the Board, a President, a Chairman of the Executive Committee, one or more Group Vice-Presidents and Vice-Fresidents (the number thereof to be determined by the Board), a Treasurer, a Secretary and such other officers as may be appointed in accordance with the provisions of Section 3 of this Article IV. One person may hold the offices and perform the duties of any two or more of said offices, except the offices and duties of President and Secretary.
- Section 2. Election and Term of Office. The principal officers of the Corporation shall be chosen annually by the Board of Directors at the first meeting thereof held after each annual meeting of stockholders for the election of directors. Each such officer shall hold office until his successor shall have been duly chosen and shall qualify, or until his death or until he shall resign or shall have been removed in the manner hereinafter provided. The Chairman of the Board and the President shall be chosen from among the directors of the Corporation.
- Subordinate Officers, Etc. In edition to the principal officers enumerated in Section 1 of this Article 1V, the Torporation may have such other officers, agents and employees as the Board may deem necessary, including a Controller, one or more Assistant Controllers, one or more Assistant Secretaries and one or more Assistant Treasurers, each of whom shall hold office for such period, have such authority, and perform such duties as the Board or the President may from time to time determine; provided, however, that no person may hold the offices and perform the duties of both Treasurer and Controller. The Board may delegate to any principal officer the power to appoint and to remove any such subordinate officers, agents or employees.
- Section 4. Removal. Any officer may be removed, either with or without cause, at any time, by resolution adopted by the Board at any regular meeting of the Board or at any special meeting of the Board called for the purpose at which a quorum is present.
- Section 5. Resignations. Any officer may resign at any time by giving written notice to the Board of Directors or to the Chairman of the Board, the President or the Secretary. Any such resignation shall take effect upon receipt of such notice or at any later time specified therein; and, unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.
- Section 6. Vacancies. A vacancy in any office because of death, resignation, removal, or any other cause whatever, shall be filled for the unexpired portion of the term in the manner prescribed in Sections 2 and 3 of this Article IV for election or appointment to such office for such term.
- Section 7. Chairman of the Board. The Chairman of the Board shall, if present, preside at all meetings of stockholders and all meetings of the Board and shall perform such other duties as may from time to time be assigned to trim by the Board.
- Section 8. President. The President shall be the chief executive officer of the Corporation and as such shall have general supervision of the affairs of the Corporation, subject to the control of the Board. In the absence of the Chairman of the Board, he shall, if present, preside at all meetings of stockholders and at all meetings of the Board. He shall be ex officio a member of all standing committees. He may sign, with any other proper officer of the Corporation thereunto authorized, certificates for stock of the Corporation. When authorized by the Board, he may enter into any contract or execute and deliver any instrument in the name and on behalf of the Corporation, except in cases in which the authority to enter into such contract or execute and deliver such instrument, as the case may be, shall be otherwise expressly delegated. In general, he shall perform all duties incident to the office of President, as herein defined, and all such other duties as from time to time may be assigned to him by the Board.
- Section 9. Group Vice-Presidents and Vice-Presidents. At the request of the President, or in his absence or inability to act, the Group Vice-President or, if there be more than one, the Group Vice-President designated by the Board, shall perform all the duties of the President and, when so acting, shall have all the powers of and

be subject to all the restrictions placed upon the President. Each Group Vice-President and each Vice-President shall perform such duties as from time to time may be assigned to him by the President or the Board. Any Group Vice-President or Vice-President may sign, with any other proper officer of the Corporation thereunto authorized, certificates for stock of the Corporation, and when authorized by the Board may enter into any contract or execute and deliver any instrument, in the name and on behalf of the Corporation, except in cases in which the authority to enter into such contract or execute and deliver such instrument, as the case may be, shall be otherwise expressly delegated.

Section 10. Treasurer. The Treasurer shall have charge and custody of, and be responsible for, all funds and securities of the Corporation and shall deposit all such funds in the name of the Corporation in such banks or other depositaries as shall be relected in accordance with Section 4 of Article V hereof. He shall exhibit at all reasonable times his books of account and records to any of the directors of the Corporation upon application during business hours at the office of the Corporation where such books and records shall be kept; shall render a statement of the condition of the finances of the Corporation at all regular meetings of the Board, and a full financial report at the annual meeting of stockholders, if called upon so to do; shall receive, and give receipt for, moneys due and payable to the Corporation from any source whatsoever; may sign, with any other proper officer of the Corporation thereunto authorized, certificates for stock of the Corporation; and in general, shall perform all the duties incident to the office of Treasurer and such other duties as from time to time may be assigned to him by the President or the Board. The Treasurer shall give such bond, if any, for the faithful discharge of his duties as the Board may require.

Section 11. Secretary. The Secretary if present, shall act as secretary at all meetings of the Board and of the stockholders and keep the minutes thereof in a book or book to be provided for that purpore; shall see that all notices required to be given by the Corporation are duly given and served; shall be custodian of the seal of the Corporation and shall affix the seal or cause it to be a fixed to all certificates of stock of the Corporation and to all documents the execution of which on behalf of the Corporation under its seal stall be duly authorized in accordance with the provisions of these Bylaws; shall have charge of the stock records of the Corporation; shall see that all reports, statements and other documents required by law are properly kept and filled; may sign, with any other proper officer of the Corporation thereunto authorized, certificates for stock of the Corporation; and in general, shall perform all the duxies incident to the office of Secretary and such other duties as from time to time may be assigned to him by the President or the Board.

Section 12. Salaries. The s. laries of the officers shall be fixed from time to time by the Board, and an officer shall not be prevented from receiving such a salary by reason of the fact that he is also a director of the Corporation.

ARTICLE V .

Contracts, Loans, Checks, Drafts, Bank Accounts, Etc.

Section 1. Execution of Contracts. The Board may authorize any officer or officers, agent or agents, or attorney or attorneys, to enter into any contract or execute and deliver any instrument in the name and on behalf of the Corporation, and such authority may be general or confined to specific instances; and, unless so authorized by the Board or expressly authorized by these Bylans, no officer, agent or employee shall have any power or authority to bind the Corporation by any contract or other engagement or to pledge its credit or to render it liable pecuniarily for any purpose or in any amount.

Section 2. Loans. No loan shall be contracted on behalf of the Corporation, and no negotiable paper shall be issued in its name, unless authorized by the Board or by a committee of the Board to whom the Board has delegated such power.

Section 3. Checks, Drafts, Etc. All checks, drafts, bills, notes and other negotiable instruments and orders for the payment of money issued in the name of the Corporation, shall be signed by such officer or officers, employee or employees, agent or agents, of the Corporation and in such manner as shall from time to time be determined by resolution of the Board.

Section 4. Deposits. All funds of the Corporation not otherwise employed shall be deposited from time to time to the credit of the Corporation in such banks, trust companies or other depositaries as the Board may designate, or as may be designated by any officer or officers, agent or agents, or attorney or attorneys, of the Corporation to whom power in that respect shall have been delegated by the Board. For the purpose of deposit and for the purpose of collection for the account of the Corporation, the President, or any Vice-President, or the Treasurer (or any other officer or agent or employee or attorney of the Corporation to whom such power shall be delegated by the Board) may endorse, assign and deliver checks, drafts and other orders for the payment of money which are payable to the order of the Corporation.

Section 5. General and Special Bank Accounts. The Board may from time to time authorize the opening and keeping of general and special bank accounts with such banks, trust companies or other depositaries as it may designate or as may be designated by any officer or officers, agent or agents, or attorney or attorneys, of the Corporation to whom power in that respect shall have been delegated by the Board. The Board may make such special rules and regulations with respect to such bank accounts, not inconsistent with the provisions of these Bylaw 1, as it may deem expedient.

ARTICLE VI

Shares and Their Transfer

Section 1. Certificates for Stock. Every owner of stock of the Corporation shall be entitled to a certificate or certificates, to be in such form as the Board shall prescribe, certifying the number and class of shares of the capital stock of the Corporation owned by him. The certificates for the respective classes of such stock shall be numbered in the order in which they shall be issued and shall be signed in the name of the Corporation by the President or any Vice-President and the Treasurer or an Assistant Treasurer, or the Secretary or an Assistant Secretary of the Corporation and its seal be affired thereto; provided, however, that, where such Certificate is signed by a tirpsfer agent or an actistant transfer agent or by a transfer clerk acting on behalf of the Corporation and a registrar, if the Board shall by resolution so authorize, the signature of such President, Vice-President, Treasurer, Secretary, Assistant Treasurer, or Assistant Secretary and the shall of the Corporation may be facsimile. In case any officer or officers of the Corporation who shall have signed, or whose facsimile signature or signatures shall have been used on, any such certificate or certificates shall cease to be such officer or officers, whether by reason of death, resignation or otherwise, before such certificate or certificates shall have been delivered by the Corporation, such certificate or certificates may nevertheless be adopted by the Corporation and be issued and delivered as though the person or persons who signed such certificate or certificates, or whose facsimile signature or signatures shall have been affixed the eto, had not ceased to be such officer or officers \ record shall be kept by the Secretary, transfer agent or by any other officer, employee or agent designated by the Board of the name of the person, firm, or corporation owning the stock represented by such certificates, the number and class of shares represented by such certificates, respectively, and the respective dates thereof, and in case of cancellation, the respective dates of cancellation. Every certificate surrendered to the corporation for exchange or transfer shall be cancelled, and no new certificate or certificates shall be issued in exchange for any existing certificate until such existing certificate shall have been so cancelled, except in cases provided for in Section 4 of this Article VI.

Section 2. Transfer of Stock. Transfers of shares of the capital stock of the Corporation shall be made on the books of the Corporation by the registered holder thereof, or by his attorney thereunto authorized by power of attorney duly executed and filed with the Secretary of the Corporation, or with a transfer clerk or a transfer agent appointed as in Section 3 of this Article VI provided, and on surrender of the certificate or certificates for such shares properly endorsed and the payment of all taxes thereon. The person in whose name shares of stock stand on the books of the Corporation shall be deemed the owner thereof for all purposes as regards the Corporation; provided, however, that whenever any transfer of shares shall be made for collateral security, and not absolutely, such fact, if known to the Secretary of the Corporation, shall be so expressed in the entry of transfer.

Section 3. Regulations. The Board may make such rules and regulations as it may deem expedient, not inconsistent with the Certificate of Incorporation or these Bylaws, concerning the issue, transfer and registration of certificates for shares of the stock of the Corporation. It may appoint, or authorize any principal officer or officers

to appoint, one or more transfer clerks or one or more transfer agents and one or more registrars, and may require all certificates of stock to bear the signature or signatures of any of them.

Section 4. Lost, Destroyed or Mutilated Certificates. In case of loss, destruction, or mutilation of any certificates of stock, another certificate or certificates may be issued in place thereof upon proof of such loss, destruction, or mutilation and upon the giving of a bond of indemnity to the Corporation in such form and in such sum as the Board may direct; provided, however, that a new certificate may be issued without requiring any bond when, in the judgment of the Board, it is proper so to do.

Section 5. Closing of Transfer Books. The Board of Directors may, by resolution, direct that the stock transfer books of the Corporation be closed for a period not exceeding fifty (50) days preceding the date of any meeting of the stockholders, or the date for the payment of any dividend, or the date for the allotment of any rights, or the date when any change or conversion or exchange of stock of the Corporation shall go into effect, or for a period not exceeding fifty (50) days in connection with obtaining the consent of stockholders for any purpose. In lieu of such closing of the stock transfer books, the Board may fix in advance a date, not exceeding fifty (50) days preceding the date of any meeting of stockholders, or the date for the payment of any dividend, or the date for the allotment of rights, or the date when any change or conversion or exchange of capital stock shall go into effect or a date in connection with obtaining such consent, as a record date for the determination of the stockholders entitled to notice of, and to vote at, such meeting, and any adjournment thereof, or to receive payment of any such dividend, or to receive any such allotment or rights, or to exercise the rights in respect of any such change, conversion, or exchange of capital stock or to give such consent, as the case may be, notwithstanding any transfer of any stock on the books of the Corporation after any record date so fixed.

ARTICLE VI

Miscellaneous Provisions

Section 1. Corporate Seal. The Board shall provide a corporate seal, which shall be in the form of a circle and shall bear the name of the Corporation and words and figures showing that it was incorporated in the State of Delaware in the year 1964.

Section 2. Fiscal Year. The fiscal year of the Corporation shall end at the close of business on the 31st day of December in each year.

ARTICLE VIII

Amendments

All Bylaws of the Corporation shall be subject to alteration or repeal, and new Bylaws may be made, either by the affirmative votes of the holders of record of a majority of the issued and outstanding stock of the Corporation, entitled to vote in respect thereof, given at any annual or special meeting, or by the affirmative votes of a majority of the whole Board of Directors given at any regular or special meeting of the Board of Directors, provided that notice of the proposal so to alter or repeal or to make such Bylaws be included in the notice of such meeting of the stockholders or the Board, as the case may be. Bylaws, whether made or altered by the stockholders or by the Board of Directors, shall be subject to alteration or repeal by the stockholders as in this Article VIII above provided.

RESOLUTIONS OF THE BOARD OF DIRECTORS OF MESA PETROLEUM CO.

WHEREAS, there has been presented to this meeting: (i) a form of Mesa Credit Agreement (the "Credit Agreement"), among this Corporation, the banks and other financial institutions parties thereto (collectively, the "Banks" Texas Commerce Bank National Association (herein called "TCB") and Mellon Bank, N.A. (herein called "Mellon"), as agents for the Banks, herein in such capacity, together with any successor(s) thereto in such capacity, collectively called the "Agents" and individually called an 'Agent"), TCB, as collateral agent for the Banks (herein in such capacity, together with any successor(s) thereto in such capacity, called the "Collateral Agent"), and Mellon, as administrative agent for the Banks (herein, in such capacity, together with any successor(s) thereto in such capacity, called the "Administrative Agent"), providing for the making by the Banks of Loans (as defined in the Credit Agreement) to this Corporation in an aggregate principal amount not to exceed \$1,100,000,000); (ii) a form of pledge agreement (the "MTR Pledge Agreement") pursuant to which this Corporation will pledge to the Collateral Agent, for the ratible and proportionate benefit of the Banks and in accordance with the terms of the Credit Agreement, the MTR Stock, the Intercompany Note (as such terms are defined in the Credit Agreement) and certain other claims of the Company against MTR (as defined in the Credit Agreement); and (iii) forms of mortgage and collateral mortgage documents (including a Mortgage, Deed of Trust, Assignment of Production and Security Agreement, an Act of Collateral Mortgage, Collateral Chattel Mortyage, Pledge and Assignment of Production, a Collateral Mortgage Note, and an Act of Pledge) (collectively, the "Security Documents") from this Corporation to the Collateral Agent and trustee for the Collateral Agent, pursuant to which this Corporation will, in accordance with the terms of the Credit Agreement, mortgage to the Collateral Agent, for the ratable and proportionate benefit of the Banks, certain oil and gas properties, for the purpose of securing repayment of the Loans:

NOW, THEREFORE, BE IT RESOLVED, that the President or any Vice President of this Corporation and each of them, be and he hereby is authorized to execute, in the name and on behalf of this Corporation, and deliver, a credit agreement among this Corporation, the Banks, the Administrative Agent, the Collateral Agent and the Agents, substantially in the form of the Credit Agreement presented to this meeting,

except for such changes, additions and deletions as to any or all of the terms and provisions thereof as the officer executing the Credit Agreement to be concept to be proper; additions and deletions as to any or all of the terms and deletions as to any or all of the compositions thereof to be proper;

FURTHER RESOLVED, that the President or any Vice President of this Corporation, and each of them, be and he hereby is authorized to borrow from time to time on behalf of this Corporation the amounts permitted or provided to be borrowed by this Corporation under the Credit Agreement executed by this Corporation pursuant to these resolutions, and to execute and deliver on behalf of this Corporation Notes (as defined in the Credit Agreement) payable to the order of each of the Banks, substantially in the form or forms provided for in exhibits to the Credit Agreement, evidencing such borrowings;

FURTHER RESOLVED, that the President or any Vice President of this Corporation, and each of them, be and he hereby is authorized, on behalf of the Company, to pledge and grant a security interest in the MTR Stock, the Inter-company Note and the other claims of the Company against MTR in the manner provided in the MTR Pledge Agreement and to mortgage and assign the Mortgaged Properties (as defined in the Credit Agreement), including interests in Hydrocarbons owned by any partnership and attributable to a partnership interest owned by the Company, in the manner provided in the Mortgages (as defined in the Credit ? reement), and to execute, in the name and on behalf of this Corporation, and deliver, a pledge agreement between this Corporation and the Collateral Agent, as Collateral Agent for the Banks under the Credit Agreement, substantially in the form of the MTR Pledge Agreement presented to this meeting, and financing statements substantially in the form presented to this meeting, except for such changes, additions and deletions as to any or all of the terms and provisions thereof as the officer executing the MTR Pledge Agreement or such financing statement on behalf of this Corporation shall deem proper, such execution by such officer of each of the MTR Pledge Agreement and such financing statements to be conclusive evidence that such officer deems all of the terms and provisions thereof to be proper;

FURTHER RESOLVED, that the President or any Vice President of this Corporation, and each of them, be and he hereby is authorized to execute, in the name and on behalf of this Corporation, and deliver, one or more mortgages,

deeds of trust, assignments of production, security agreements, financing statements and collateral mortgage documents from this Corporation to the Collateral Agent, as Collateral Agent for the Banks under the Credit Agreement, and trustees for the Collateral Agent, substantially in the forms of the Security Documents presented to this meeting, except for such changes, additions and deletions as to any or all of the terms and provisions thereof as the officer executing the Security Documents on behalf of this Corporation shall deem proper, such execution by such officer of the Security Documents to be conclusive evidence that such officer deems all of the terms and provisions thereof to be proper.

FURTHER RESOLVED, that any and all acts, transactions, or agreements undertaken prior to the date of these resolutions by any of the officers or representatives of this Corporation, in the name and on behalf of this Corporation, in connection with any of the foregoing matters, are hereby ratified, confirmed, and adopted by this Corporation; and

FURTHEP RESOLVED, that the tresident or any Vice President and any Secretary or Assistant Secretary of this Corporation, and each of them, be and he nereby is authorized in the name and on behalf of this Corporation from time to time to take such actions and to execute and deliver such certificates, instruments, notices and documents as may be required or as any such officer may deem necessary, advisable or proper in order to carry out and perform the obligations of this Curporation under the Credit Agreement, the MTR Pledge Agreement and the Security Documents executed by this Corporation pursuant to these resolutions, or under any other instrument or document executed pursuant to or in connection with the Credit Agreement, the MTR Pledge Agreement or the Security Documents; all such actions to be performed in such manner, and all such certificates, instruments, notices and documents to be executed and delivered in such form, as the officer performing or executing the same shall approve, the performance or execution thereof by such officer to be conclusive evidence of the approval thereof by such officer and by this Board of Directors.

EXHIBIT A-2 to Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production

COLLATERAL MORTGAGE NOTE

| \$3. | nr | n | n | n n | 0 | n | n |
|------|----|-----|---|-----|---|---|---|
| 20. | ·u | JU. | | υu | | ш | u |

September , 1984

FOR VALUE RECEIVED, on demand, the undersigned, Mesa Petroleum Co., a Delaware corporation promises to pay to the order of Bearer at the offices of Texas Commerce Bank National Association, at 712 Main Street, Houston, Texas 77002, the sum of THREE BILLION DOLLARS (\$3,000,000,000) with interest at the rate of eighteen percent (18%) per annum from the date until paid.

In the event this note should be placed in the hands of an attorney, after its maturity, to institute legal proceedings to recover the amount hereof, or any part hereof, in principal or interest, or to protect the interests of the holder hereof, or in the event the same should be placed in the hands of an attorney for collection, compromise or other action, the maker binds its if to pay the fees of the attorney who may be employed for that purpose, which fees are hereby fixed at ten percent (10%) of the amount due or sued for, in claimed or sought to be protected, preserved or enforced.

The parties here's whether is original signers or by guaranty or endorsement or otherwise, hereby waive poser ment for payment, demand, notice of non-payment, protest, and all pleas of dission and discussion, and agree that the time of payment hereof may be extended from time to time, one or more times, without notice of such extension or extensions and without previous consent, hereby binding themselves in solido, unconditionally are as original promissors, for the payment hereof in principal, interest, costs and attorneys' fees.

This note shall be construed in accordance with and governed by the laws of the State of Louisiana.

MESA PETROLEUM CO.

By ______David H. Batchelder

Vice President - Finance and Treasurer

Collateral Mortgage Note

"NE VARIETUR"

| Mortgage, of Septem | , Pledge and Assignment of | Production passed before me this | Chatte day |
|------------------------|----------------------------|---|---------------|
| | | | |
| | | Notary Public | |
| | | In and for the State of Texas, Commission Expiring | |

EXHIBIT B

to

Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production

PROPERTY DESCRIPTIONS

This Exhibit B contains the description of the "cil and gas leases" as defined in the Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production to which this Exhibit B is attached.

The oil and gas leases are expressly limited insofar and only insofar as they cover land and depth intervals in which Mortgagor owns or holds an interest and do not include land and depth intervals in which Mortgagor owns no interest.

As used in this Exhibit B, the term "working interest" specifies the leasehold or operating interest (expressed as a decimal fraction) of Mortgagor in an oil and gas lease, and the term "net interest in production" specifies the interest (expressed as a decimal fraction) of Mortgagor in the production of Hydrocarbons from an oil and gas lease after deducting all applicable royalties, overriding royalties and other burdens on production.

L Oil and Gas Lease OCS-G 2317, effective February 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering Block 333, Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A.

| Mesa | Working Interest | 13333 |
|------|----------------------------|-------|
| Mesa | Net Interest in Production | 11111 |

 Oil and Gas Lease effective July 19, 1971, granted by the State of Louisiana, as Lessor. to Mesa Petroleum Co., et al., as Lessee, covering entire Tract No. 11960 located in St. Martin and St. Mary Parishes, Louisiana, described as follows: the lands now or formerly constituting the beds and bottoms of all water bodies of every nature and description and all islands and other lands formed by accretion or reliction, except tax lands, owned by and not under mineral lease from the State of Louisiana on April 29, 1971 situated in St. Martin and St. Mary Parishes, Louisiana, within the following described boundaries: Beginning at a point having Lambert Plane Coordinates of X = 1.966,700.00 and L = 446,600.00, being the Northwest corner of the tract herein described; thence East 16,000.00 feet; thence South 13,500.00 feet to a point having Lambert Plane Coordinates of X = 1,982,700.00 and Y = 433,000.00; thence West 16,000.00 feet; thence North 13,600.00 feet to a point of beginning, estimated to contain approximately 4,620 acres, all more fully shown outlined in red on a plat on file in the State Land Office. All bearings are based on Louisiana Lambert Plane Coordinate System (South Zone), recorded in Book 652, Folio 115, Entry No. 154449 of the Conveyance Records of St. Martin Parish, Louisiana and recorded in Book 16-T, Entry No. 147,217 of the Conveyance Records of St. Mary Parish, Louisiana.

| Mesa Working Interest | 608540X II 6 | .18750 |
|------------------------------|--------------|--------|
| Mesa Net Interest in Product | ion | .13281 |

3. Oil and Gas Lease OCS-G 3286, effective September 1, 1975, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 613, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 18.

| Mesa | Working Interest | .18400 |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | .15333 |

4. Oil and Gas Lease OCS-G 2591, effective May 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 146, South Marsh Island Area, South Addition, Official Leasing Map, Louisiana Map No. 3C.

| Mesa | Working Interest | 16000 |
|------|----------------------------|-------|
| | Net Interest in Production | 13333 |

5. Oil and Gas Lease OCS-G 2910, effective December 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 327, Eugene Island Area, South Addition, OCS Official Leasing Map. Louisiana Map No. 4A.

| Mesa | Working Interest | 11500 |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | .09583 |

6. Oil and Gas Lease OCS-G 2912, effective December 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 329, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A.

| Mesa | Working Interest | 22500 |
|------|----------------------------|------------|
| Mesa | Net Interest in Production | .13749 |

7. Oil and Gas Lease OCS-G 3141, effective July 1, 1975, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 397, Vermilion Area, South Addition, as shown on OCS Official L asing Map, Louisiana Map No. 3B.

| Mesa | Working Interest | .60000 |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | .40) |

This property is subject to the Advance Payment Agreement dated December 16, 1975, as amended February 19, 1976 and May 10, 1977, between Mesa Petroleum Co., as Seller, and Michigan Wisconsin Pipe Line Company of Bluwer, and as opting of the pledged and Michigan Wisconsin Pipe Line Company of Bluwer, and as opting of the property of the United States of America, as Lesse, to Mesa Petroleum Co., et al., as Lessee, covering Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of Supplemental Decree of U.S. Supreme Court entered 12-13-65 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288) South Pass Area, South and East Addition, as shown on Official Leasing

| Mesa Working Interest | .25000 |
|---------------------------------|--------|
| Mesa Net Interest in Production | 20833 |

Map, Louisiana Map No. 9A.

This property is subject to the Exploration and Development Loan Agreement dated December 19, 1973, as amended December 12, 1974, February 28, 1975 and June 14, 1977, between Mesa Petroleum Co. and Natural Gas Pipeline Company of America.

 Oil and Gas Lease OCS-G 3299, effective September 1, 1975, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 263, East Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 2A.

| Mesa Working Interest | .10000 |
|---------------------------------|--------|
| Mesa Net Interest in Production | .08333 |

This property is subject to the Mortgage, Deed of Trust, Assignment of Production, Security Agreement and Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortgagor, and McMoRan Exploration Co. and to the Act of Mortgage,

Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Owner, and McMoRan Exploration Co.

10. Oil and Gas Lease OCS-G 3388, effective January 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessoe, covering all of Block 336, East Cameron Area, South Addition, as shown on OCS Official Lessing Map, Louisiana Map No. 2A.

This property is subject to the Mortgage, Dead of most, Assignment of Production, Security Agreement and Financing South and Make and Assignment of Production deted Jac 21, 1977, between Mesa Petroleum Co., as Owner, and McMoRan Exploration Co.

11. Oil and Gas Lease OCS-G 2000. Rective January 1, 1977, granted by the United States of America, as Lessor, to Mess Total Co., et al., as Lessee, covering all of Block 25, Vermilion Area, as shown on OCS O. and Leasing Map, Louisiana Map No. 3.

| Mesa | Working Interes | .04737 |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | .02947 |

This property is subject to the Mortgage, Deed of Trust, Assignment of Production, Security Agreer and and Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortgagor, and McMoRan Exploration Co. and to the Act of Mortgage, Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Owner, and McMoRan Exploration Co.

12. Oil and Gas Lease OCS-G 3400, effective January 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 310, Vermilion Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map. No. 3B.

| Mesa | Working Interest | .05000 |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | .04167 |

This property is subject to the Mortgage, Deed of Trust, Assignment of Production, Security Agreement and Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortgagor, and McMoRan Exploration Co. and to the Act of Mortgage, Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Owner, and McMoRan Exploration Co.

13. Oil and Gas Lease OCS-G 3414, effective January 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering the N/2 of Block 34, West Delta Area, as shown on OCS Official Leasing Map, Louisiana Map No. 8.

| Mesa Working Interest | | .05000 |
|---------------------------------|--|--------|
| Mesa Net Interest in Production | N 61 4104 - 103 19 (0 3206) 410 321438 (3 101) | .04167 |

This property is subject to the Mortgage, Deed of Trust, Assignment of Production, Security Agreement and Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortgagor, and McMoRan Exploration Co. and to the Act of Mortgage, Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Owner, and McMoRan Exploration Co.

14. Oil and Gas Lease OCS-G 3589, effective August 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 18, South Pelto Area, as shown on OCS Official Leasing Map, Louisiana Map No. 6.

| Mesa | Working Interest | ा करने साम्बन्ध करने रूपा प्रथम कारणा विकास विकास | 26173 |
|------|----------------------------|---|-------|
| Mesa | Net Interest in Production | | 21815 |

15. Oil and Gas Lease OCS-G 2271, effective February 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering Block 348, Vermilion Area, South Addition, on OCS Official Leasing Map, Louisiana Map No. 3B.

| Mesa | Working Interest | .75000 |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | .62499 |

16. Oil and Gas Lease No. 6310 effective May 13, 1974, granted by the State of Louisiana, as Lessor, to Mesa Petroleum Co. et al., as Lessee, covering those portions of Blocks 57 and 58, South Pass Area. Piaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974, described as follows:

Beginning at the Northwest corner of Block 57, South Pass Area, having Lambert Plane Coordinates of X = 2,615,928.00 and Y = 82,150.00; thence East 9,685.54 feet along the North line of said Block 57 to a point having Lambert Plane Coordinates of X = 2.625,613.54 and Y = 82.150.00; thence Southwesterly in a straight line to a point having Lamba. Plane Coordinates of X = 2,624,340.00 and Y = 80,576.00; thence Southwesterly along an arc which has a radius of 18,240.60 feet and a vertex having Lambert Plane Coordinates of X = 2,610,160.00 and Y = 92,050.00 to a point having Lambert Plane Coordinates of X = 2,621,555.00 and Y = 77,806.00; thence Southwesterly in a straight line to a point having Lambert Plane Coordinates of X = 2,621,180.00 and Y = 77,506.00; thence Southwesterly along an arc which has a radius of 18,240.60 feet and a vertex having Lambert Plane Coordinates of X = 2,609,785.00 and Y = 91,750.00 to a point having Lambert Plane Coordinates of X = 2,617,996.00 and Y = 75,462.00; thence Southwesterly in a straight line to a point having Lambert Plane Coordinates of X = 2,617,391.00 and Y = 75,157.00; thence Westerly along an arc which has a radius of 18,240.60 feet and a vertex having Lambert Plane Coordinates of X = 2,609,180.00 and Y = 91,445.00 to a point having Lambert Plane Coordinates of X = 2,610,702.97 and Y = 73,263.09; thence North 8,881.91 feet to a point of intersection with the North line of Block 58, South Pass Area; thence East 5,225.03 feet along the North line of said Block 58 to its Northeast corner, the same being the point of beginning, estimated to contain approximately 2,028.40 acres, all more fully shown outlined in red on a plat on file in the State Land Office. All bearings are based on Louisiana Lambert Plane Coordinate System (South Zone). Reference is made to a survey line described in a Fourth (4th) Supplemental Decree, entered on the 16th day of October, 1972, in that macter styled United States of America v. State of Louisiana, et al., No. 9 Original, on the docket of the Supreme Court of the United States, it is expressly understood that although a segment of that line is incorporated in the description of the property herein leased, that the State of Louisiana is leasing only that portion of the above described block or blocks lying more than one (1) foot landward of that line, all as provided for in said Fourth (4th) Supplemental Decree. Note: Portions of this tract are located within the restrictive area as set out by the U.S. Coast Guard and the Corps of Engineers, U.S. Army.

This property is subject to the Act of Mortgage and Security Agreement to Secure Future Advances dated November 22, 1978, between Mesa Petroleum Co., as Mortgagor, and Columbia Gas Transmission Corporation, as Mortgagee.

17. Oil and Gas Lease OCS-G 3416, effective January 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering that portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975, (United States v. Louisiana), 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 366, December 20, 1971), as shown on OC5 Official Leasing Map. Louisiana Map No. 9.

| Mesa | Working Interest | .33323 |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | 27777 |

This property is subject to the Act of Mortgage and Security Agreement to Secure Future Advances dated November 22, 1978, between Mesa Petroleum Co., as Mortgagor, and Columbia Gas Transmission Corporation, as Mortgagee.

18. Oil and Gas Lease OCS-G 2619, effective May 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 115, Ship Shoal Area, Official Leasing Map, Louisiana Map No. 5.

| Mesa | Working Interest | |
|------|----------------------------|--------|
| Mesa | Net Interest in Production | .16667 |

19. Oil and Gas Lease OCS-G 2887, effective December 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 173, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C.

20. Oil and Gas Lease OCS-G 2888, effective December 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 174, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C.

| Mesa Working | Interest | MONTH THE RESIDE THE LITTLE IN WHITE HER RE | | .10239 |
|--------------|----------|---|--|--------|
|--------------|----------|---|--|--------|

21. Oil and Gas Lease OCS-G 2607, effective May 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering Block 312, E gene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A.

 22. Oil, Gas, and Mineral Lease dated May 5, 1980 executed by Melba L. Harvey, Jr., et al., as Lessor, in favor of Franks Petroleum, Inc., as Lessee, and recorded in the Conveyance Records of East Baton Rouge Parish, Louisiana in Book 637 on page number 9375.

23. Oil, Gas, and Mineral Lease dated April 12, 1976 executed by Edward D. Annison, as Lessor, in favor of Forest Oil Corporation, as Lessee, and recorded in the Conveyance Records of East Baton Rouge Parish, Louisiana in Book 2505 on page number 200.

24. Oil, Gas, and Mineral Lease dated April 15, 1980 executed by James C. Celestin, et ux., as Lessor, in favor of Tomlinson Interest, Inc., as Lessee, and recorded in the Conveyance Records of Pointe Coupee Parish, Louisiana, in Book 188, on page number 62.

25. Oil and Gas Lease OCS-G 2102 effective February 1, 1971 granted by the United States of America, as Lessor, to Pennzoil, et al., as Lessee, covering all of Block 256, Eugene Island Area, Official Leasing Map, Louisiana Map No. 4.

26. Oil and Gas Lease OCS-G 31/1, effective July 1, 1975 granted by the United States of America, as Lessor, to Mera Petroleum Co., et al., as Lessee, covering all of Block 13, South Pelto Area, as shown on OCS Official Leasing Map, Louisiana Map No. 6.

27. Oil and Gas Lease OCS-G 3186, effective July 1, 1975 granted by the United States of America, as Lessor, to Mesa Petroleum Co., as Lessee, covering all of Block 61, West Delta Area, as shown on OCS Official Leasing Map, Louisiana Map No. 8.

The above described Lease is subject to an overriding royalty interest conveyed by an overriding royalty conveyance from Mesa Petroleum Co., as Assignor, to Mesa Offshore Royalty Partnership, as Assignee, dated as of December 1, 1982.

| 28. Oil and Gas Lease OCS-G 3332, effective March 1, 1976 granted by the United States of America, as Lessor, to Pennzoil, et al., as Lessee, covering all of Block 337, Eugene Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A. |
|---|
| MTS Working Interest |
| 29. Oil and Gas Lease OCS-G 3587, effective Aug.st 1, 1977 granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 8, South Pelto Area, as shown on OCS Official Leasing Map, Louisiana Map No. 6. |
| Mesa Working Interest |
| 30. Oil and Gas Lease OCS-G 3588, effective August 1, 1977 granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering the East Half of Block 12, South Pelto Area, as shown on OCS Official Leasing ap, Louisiana Map No. 6. |
| Mesa Working Interest |
| The above described Lease is subject to an overriding royalty interest conveyed by an overriding royalty conveyance from Mesa Petroleum Co., as Assignor, to Mesa Offshore Royalty Partnership, as Assignee, dated as of December 1, 1982. |
| 31. Oil and Gas Lease OCS-G 3601, effective August 1, 1977 granted by the United States of America, as Lessor, to Mesa Petroleum Co., as Lessee, covering all of Block 62, West Delta Area, as shown on OCS Official Leasing Map, Louisiana Map No. 8. |
| Mesa Working Interest |
| The above described Lease is subject to an overriding royalty interest conveyed by an overriding royalty conveyance from Mesa Petroleum Co., as Assignor, to Mesa Offshore Royalty Partnership, as Assignee, dated as of December 1, 1982. |
| 32. Oil and Gas Leare OCS-G 4102, effective Septem' or 1, 1979 granted by the United States of America, as Lessor, to McMoRan Exploration, et al., as Lessee, covering W/2 NE, NW, S/2 Block 36, Vermilion Area, as shown on OCS Official Leasing Map, Louisiana Map No. 3. |
| Mesa Working Interest |
| 33. Oil and Gas Lease OCS-G 4110, effective September 1, 1979 granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 155, South Marsh Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 3C. |
| Mesa Working Interest |

The above described Lease is subject to an overriding royalty interest conveyed by an overriding royalty conveyance from Mesa Petroleum Co., as Assignor, to Mesa Offshore Royalty Partnership, as Assignee, dated a of December 1, 1982.

34. Oil and Gas Lease OCS-G 5221, effective November 1, 1582 granted by the United States of America, as Lessor, to McMoRan Cffshore Exploration Co., et al., as Lessee, covering all of Block 176, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C.

35. Oil and Gas Lease OCS-G 2045, effective January 1, 1971 granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 270, East Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 2A.

36. Oil and Gas Lease OCS-G 2078, effective February 1, 1971 granted by the United States of America, as Lessor, to Pennzoil, et al., as Lessee, covering all of Block 228, Vermilion Area, Official Leasing Map, Louisiana Map No. 3.

37. Oil and Gas Lease OCS-G 2115, effective January 1, 1971, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 330, Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A.

38. Oil and Gas Lease OCS-G 2254, effective February 1, 1973 granted by the United States of America, as Lessor, to Burmah, et al., as Lessee, covering all of Block 322, East Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 2A.

39. Oil and Gas Lease OCS-G 2255, effective February 1, 1973 granted by the United States of America, as Lessor, to Burmah, et al., as Lessee, covering all of Block 323, East Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 2A.

40. Oil and Gas Lease OCS-G 4220, effective December 1, 1979 granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 156, South Marsh Island Area, South Addition, as shown on OCS Leasing Map, Louisiana Map No. 3C.

The above described Lease is subject to an overriding royalty interest conveyed by an overriding royalty conveyance from Mesa Petroleum Co., as Assignor, to Mesa Offshore Royalty Partnership, as Assignee, dated as of December 1, 1982. This property is subject to the Purchase Agreement - Development Production Payment dated June 6, 1980, as amended December 12, 1980, November 19, 1981 and September 2, 1983, between Mesa Petroleum Co., as Seller, and 110 Wacker Drive Foundation, Inc., as Buyer and the Conveyance of Development Production Payment dated June 6, 1980, as amended December 12, 1980, November 19, 1981, and September 2, 1983, between Mesa Petroleum Co., as W.I. Owner and 110 N. Wacker Drive Foundation as P.P. Owner.

41. Oil and Gas Lease OCS-G 4216, effective December 1, 1979 granted by the United States of America, as Lessor, to Mera Petroleum Co., et al., as Lessee, covering all of Block 381, Vermilion Area, South Addition, as shown on OCS Leasing Map, Louisiana Map No. 3B.

The above described Lease is subject to an overriding royalty interest conveyed by an overriding royalty conveyance from Mesa Petroleum Co., as Assignor, to Mesa Offshore Royalty Partnership, as Assignee, dated as of December 1, 1982.

42. Cil and Gas Lease OCS-G 2557, effective April 1, 1974 granted by the United States of America, as Leasor, to Burmah Development Co., Inc., et al., as Lessee, and covering all of Block 612, West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B.

43. Oil and Gas Lease OCS-G 2623, effective May 1, 1974 granted by the United States of America, as Lessor, to Mesa and Occidental, as Lessee, covering all of Block 31, South Timbalier Area, Official Leasing Map, Louisiana Map No. 6.

LEMLE, KELLEHER, KOHLMEYER & MATTHEWS

EIGHTEENTH FLOOR

LOUIS C . FULE I-tp CHAPLES :- . MEYER, JR H 468" N - . N.EY. . F. WILL AM 5*2'4. ... C. W. HOMY -: 55. - P AL . EY 9. FONTENOT GERRSE & FA LTT. T ALBERT H HAMEMANN. - R JAMES A. THURSHILL PAUL B. DEAL -ULIAN -. 3003 MILLAN S PENICS

JAMES M. DAISLE

FREDERICK F PREAUS W.L. AM G FORRESTER, .. R. MILLIAM E D'SEIL W LL AM J LARIELERE, JR. EARL S. EICH N. . 3. ERMEST L COMARDS ROMALD L MASUIN CANCO & CELLINI 45H"24 R J'CWTER. - R W E. NCE. VANCE & CLLEFSON .AMES M PETERSEN FOR # 3:5#483

CEORGE FRANCE W

FIRST NATIONAL SANK OF COMMERCE BUILDING

NEW ORLEANS 70112

TELEPHONE (504) 588- 14" CABLE ADDRESS: LE " " TELEX 584272

April 17, 1979

CERROR B. MATTHEWS (1313) 9701

MACE F BARNAM OF COUNSEL

MARS S. STEIS RICHARD B FOSTER MICHAEL A. NO SLONE CHARLES F THEMSTED RALPH 5. HUBBARD. III SHERRY MARCUS WISE HAL T. AFLEM STEWART P PECA DAVID 5. HELLT COUGLAS & MATTHEWS S. RODGER AMENTON, .R.

JAMES H. BROWN, JR. GARET & TRIBBLE ATRICS, CT MART A JOSEPH . MOLT RUSSELL D LEVER DAVID BUSCH MARK L 70 . DEBONAN F TENNER PETER L. AGERBER

Department of the Interior Bureau of Land Management Outer Continental Shelf Office 500 Camp Street New Orleans, LA 70130

Attention: Mr. John L. Rankin,

Offshore Manager

Leases Nos. OCS-G2224, 2225, 2317, 2177, <u>2185</u>, 2193, 2195, 2839, 3284, 2850, 2559, 2859, 2860, 2882, 2833, 2537, 2384, 2900, 3156, 2607, 3193, 3195, 2947, 3206, 3202, 3203, 3417, 3416, 3409, 3500, 3766, 3726.

Offshore Louisiana

Leases Nos. OCS-G2366, 2372, 3118, 2378, 2701, 2704, 2779, 2705, 2706, 2384, 2713, 2383, 2389, 2390, 2718, 2719, 2398, 2403, 3314, 2409, 2737, 2414, 2416, '18, 2422, 2739, 2426, 2429, 2745, '46, 2766, 2367, 3756, 3757. Offshore Texas

Gentlemen:

Enclosed are an original executed copy and 69 xerox copies of each of the following documents:

> 1. Amendment to Production Payment Agreement dated as of April 1, 1979 between Pennzoil Company and Pennzoil Louisiana and Texas Offshore, Inc.

Department of the Interior Page 2 April 17, 1979

- Amendment to Instrument of Conveyance of Production Payment dated as of April 1, 1979 between Pennzoil Louisiana and Texas Offshore, Inc. and Pennzoil Company.
- Confirmation of Assignment dated as of April 1, 1979 among Pennzoil Louisiana and Texas Offshore, Inc. Pennzoil Company and The First Natinal Bank of Chicago, as Agent for certain banks.

We request that you file the original executed copy of each of these documents, along with this letter, in your file applicable to Lease No. OCS-G2839. Please also place an additional copy of this letter and a xerox copy of these documents in your file applicable to each of the remaining captioned leases for cross-referencing purposes.

We understand that the payment of a fee in connection with this filing is not required by applicable regulations, and that no approval is necessary. Please return one enclosed copy of this letter to evidence such filing and the accomplishment of our request.

Very truly yours.

LEMLE, KELLEHER, KOHLMEYER & MATTHEWS

from Mr. dar ...

James M. Petersen

JMP/dca

Enclosures

AMENDMENT TO PRODUCTION PAYMENT AS REEMENT

THIS AMENDMENT TO PRODUCTION PAYMENT AGREEMENT (hereinafter called this "Amendment"), dated as of April 1, 1979 between Pennzoil Company, a Delaware corporation (hereinafter called "Pennzoil"), and Pennzoil Louisiana and Texas Offshoat Ing, a Delaware corporation (hereinafter called the "Company"),

WHEREAS, by means of the decentain Instrument of Conveyance of Production Payment dated as of March 1, 1971 between the Company and Pennzoil (such Instrument of Conveyance being hereinafter called the "Conveyance"), the Company conveyed to Pennzoil, and Pennzeil purchased from the Company, at the time the first loan was made to the Company under that certain Bank Crecit Agreement dated as of March 1, 1974, as thereafter amended and restated as of November 1, 1977, among the Company, The First National Bank of Chicago, and the other banks named thereir (such Bank Credit Agreement, as amended and restated being hereinarter called the "Bank Credit Agreement"), a production payment discl. urgeable from the proceed; of production from or applicable to the interests of the Company (hereinafter called the "Leasehold Interests"), in those certain oil and gas leases covering lands located offshore Louisiana and Tevas in the Gulf of Mixico, which interests, leases and lands are described in Exhibit "A" to the Conveyance and in Exhibit "A" to Appendix I to the Production Payment Agreement hereinafter referred to, in an amount equal to the aggregate principal amount of the loans, plus accrued interest and commitment fee, outstanding under the Bank Credit Agreement on the date, if any, upon which the purchase price therefor is paid by Pennzoil to the Company as provided in that certain Productica Payment Agreement dated as of March 1, 1974. as thereafter amended and restated as of November I. 1977, between Pennzoil and the Company (such Production Payment Agreement, as amended and restated, being hereinafter called the "Production Payment Agreement"; all as more particularly provided in the Production Payment Agreement;

Whereas, in order to induce the banks referred to in Section 2 of the Ba redit Agreement; (hereinafter called the "Banks") to make loans pursuant to the Bank Credit Agreement and to secure the performance by the Comp.ny of its obligations under the Eank Credit Agreement, the Company, that certain assignment dated as of March 1, 1974, and confirmed by the Confirmation of Assignment dated as of November 1, 1977 (hereinafter called the "Assignment") from the Company to The First National Bank of Chicago, as agent for the Banks (hereinafter called the "Agent"), assigned, appear all the right, title and interest of the Company in, to and under the Production Payment Agreement, subject to the exceptions set forth in the Assignment, and which Assignment was also out therein.

Whereas, the Bank Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of May 1, 1976 to reflect an extension of the maturity of the credit available to the Company under the Bank Credit. Agreement, an increase in the amount of that credit to \$200,000,000, the addition of certain braks as parties to the Bank Credit Agreement, the and for other purposes;

Whereas, the Bank Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of November 1, 1977 to reflect an extension of the maturity of the credit available to the Company under the Bank Credit Agreement, an increase in the amount of the additional deletion from the Leasehold Interests of the interests of the Company in certain and for other purposes;

WHEREAS, the Bank Credit Agreement is now proposed to be amended to reflect a further extension of the maturity of the credit available to the Company thereunder, an increase in the amount of that credit to \$275,090,000, and for other purposes; and

WHEREAS, pursuant to the request of the Company and Pennzeil, the Agent proposes to join herein for the limited purposes set forth below;

Now, Therefore, in order to reflect the Amendment. ... e Bank Credit Agreement proposed as aforesaid, to add to the Leasehold Interests the interests of the Company in certain oil and gas leases covering lands located offshore in the Gulf of Mexico adjacent to the United States of America, and to delete from the Leasehold Interests certain other leases, and in consideration of the premises and of the mutual covenants and agreements herein contained and intending to be legally bound, the parties hereto agree as follows:

- 1. The Production Payment Agreement is hereby amended in the following respects:
- (a) The introductory paragraph of the Production Payment Agreement is amended by changing the date "November 1, 1977" therein to "April 1, 1979".
- (b) Section 2.1 of the Production Payment Agreement is amended by changing the phrase "as amended and restated as of November 1, 1977" the into read "as amended, including by the amendments there to pursuant to the Amendment to b. k Credit Agreement dated as of April 1, 1979".
- (c) Section 2.2 of the Production Payment Agreement is amended by changing the figure "105%" therein to read "106%" and by deleting the words "plus 1/2%" therein.
- (d) Section 2.3 of the Production Payment Agreement is amended by changing the date "May 1, 1976" therein to read "November 1, 1977" and by changing the date "November 1, 1977" therein to read "April 1, 1979".
- (e) Section 2.4 of the Production Payment Agreement is amended by changing the date "July 1, 1986" therein to read "January 1, 1988", and by changing the pirase "as amended and restated as of November 1, 1977" therein to read "as amended, including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1979".
- (f) Section 2.5 of the Production Payment Agreement is amended by changing the date "June 30, 1986" in paragraph (b) thereof to read "Decemer 31, 1987", and by changing the phrase "as amended and restated as of November 1, 1977" therein to read "as amended, including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1979".
- (g) Section 3.3 of the Production Payment Agreement is amended by deleting the second occurrence of the word "area" therein and by changing the reference to "Federal Power Commission" therein to "Federal Energy Regulatory Commission".
- (h) The "IN WITNESS WHEREOF" paragraph of the Production Payment Agreement is amended by changing the date "November 1, 1977" therein to read "April 1, 1979".
- (i) The signatures and acknowledgements set forth in the Production Payment Agreement are amended to read in their entirety as set forth in A..nex I hereto as the signatures and acknowledgements thereto.
- (j) The first paragraph of Appendix 1 to the Production Payment Agreement is amended by changing the date "November 1, 1977" there... to "April 1, 1979".
- (k) Subpuragraph (f) of Paragraph 1 of Appendix 1 to the k-oduction Payment Agreement is amended by changing the date "November 1, 1977" therein to "April 1, 1979".
- (1) Subparagraph (a) of Paragraph 4 of Appendix 1 to the Production Payment Agreement is amended by changing the phrase "as thereafter amended and restated as of November 1, 1977" therein to read "as thereafter amended, including by the amendment's thereto pure to the Amendment to Bank Credit Agreement dated as of April 1, 1979".
- (m) Subparagraph (b) of Paragraph 4 of Appendix 1 to the Production Payment Agreement is amended by changing the figure "105%" therein to "106%" and by deleting the words "plus 1/2%".

- (n) Paragraph 9 of Appendix 1 to the Production Payment Agreement is amended by changing all references to "Federal Power Commission" therein to "Federal Energy Regulatory Commission".
- (o) The signatures and acknowledgments set forth in Appendix 1—the Production Payment Agreement are amended to read in their entirety as set forth in Annex I hereto as the signatures and acknowledgments thereto.
- (p) Exhibit "A" to Appendix 1 to the Production Payment Agreement is amended to read in its entirety as set forth in Annex I hereto as Exhibit "A" to Appendix 1 thereto.
- 2. Except as amended hereby, the provisions of the Production Payment Agreement shall remain in full force and effect. The Production Payment Agreement, as amended hereby, shall be restated to read in its entirety as set forth in Annex I hereto, it being agreed and understood that all references in the Production Payment Agreement as amended hereby and so restated to the Bank Credit Agreement shall be deemed to be references to the Bank Credit Agreement as amended, including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1979.
- 3. The Agent, for and on behalf and in the name of the Banks, hereby consents to the foregoing amendment to and restatement of the Production Payment Agreement; provided, however, that notwithstanding any provision hereof to the contrary, this Amendment shall not be construed or operate, except as expressly set forth above, to modify in any respect the rights and obligations of the Company or Pennzoil under the Production yment Agreement or the Assignment.

IN WITNESS WHEREOF the parties eto have caused this Amendment to be executed by their respective officers thereunto duly authorized as of the day and year first above written.

WITNESSES:

PENNEOIL LOUISIANA AND TEXAS OFFSHORE, INC.
[SEAL.]

By Klivetelly Tredsurer

Roger hurnette Pennzoe Company

[SEAL]

Au J. Prejean

Roger wirnette

Su J. Freyan

THE FIRST NATIONAL BANF OF CHICAGO, as Agent

Group Vice Presiden?

[SEAL]

K M. Kless

File Lamon

R. S. Glanile Tille: Vice President THE STATE OF TEXAS COUNTY OF HARRIS SS:

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 9th day of April, 1979:

(Louisiana)

Before me appeared J. N. AVERETT, JR., to me personally known, who, being by me duly sworn, did say that he is the Treasurer of Pennzoil Louisiana and Texas Offshore, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that J. N. AVERETT, JR. acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared HAROLD E. SORTOR, to me personally known, who, being by me duly sworn, did say that he is a Group Vice President of PENNZOIL COMPANY, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that HAROLD E. SORTOR acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared J. N. Averett, Jr., known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be the Treasurer of Pennzoil Louisiana and Texas Offshore, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on his day personally appeared HAROLD E. SORTOR, kind white to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of PENNZOR. COMPANY, a Delaware corporation, and admowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Couston, County of Harris, State of Texas, this 9th day or April, 1979.

[NOTARIAL SEAL]

Notary Public in and for Harris County, Texas My Commission Exp. es

Calia J. Laya

CELIA L GARZA

Notary Public In Harris County for the State of Taxes
My Commission Expires Sectember 30, 1990

THE STATE OF ILLINOIS SS.

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 14th day of April, 1979:

(Louisiana)

Before me appeared R.J. GLANVILLE to me personally known, who, being by me duly sworn, did say that he is a vice President of The First National Bank of Chicago, a national banking association, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that R.J. GLANVILLE acknowledged the instrument to be the free act and deed of the national banking association.

(Texas)

Before me on this day personally appeared R. J. GLAL "ILLE", known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a VICE FRESHOLD TO THE FIRST NATIONAL BANK OF C. TAGO, a national banking association, and acknowledged to me that he executed raid instrument for the purposes and consideration therein expressed and in the capacity therein stated, and as the act and deed of said national hanking association.

IN WITNESS WHEREOF, I have hereunto set my hazid and official notarial seal, in the City of Chicago, County of Cook, State of Illinois, this //#day of April, 1979.

Notary Public in and for Cook County, Illinois
My Commission Expires

My Commission Expires Dec. 5, 1982

[NOTARIAL SEAL]

PRODUCTION PAYMENT AGREEMENT

THIS AGREEMENT, dated as of March 1, 1974, as amended and restated as of April 1, 1979, between Pennzoil Company, a Delaware corporation (hereinafter called "Pennzoil"), and Pennzoil Louisiana and Texas Offshore, Inc., a Delaware corporation (hereinafter caued the "Company"),

WITNESSETH:

ARTICLE I

Recitals

- 1.1 The Company proposes herein to agree to make available to Pennzoil, or Pennzoil's designee, for purchase on the terms and conditions provided herein all gas produced from or applicable to the interests of the Company (hereinafter called the "I easehold Interests") in certain oil and gas leases covering lands located offshore in the Gulf of Mexico adjacent to the United States of America, which interests, leases and lands are described in Exhibit "A" to Appendix 1, attached hereto and hereby made a part hereof.
- 1.2 Pennzoil proposes to purchase from the Company, at the time the first loan is made to the Company under that certain Bank: Credit Agreement, dated as of the date hereof, among the Company, The First National Bank of Chicago and the other banks named therein (hereinafter called the "Bank Credit Agreement"), a production payment dischargeable from the proceeds of production from or applicable to the Leasehold Interests in an amount equal to the aggregate principal amount of loans, plus accrued interest and commitment fee, outstanding under the Bank Credit Agreement on the date, if any, upon which the purchase price therefor is paid by Pennzoil to the Company as provided herein (hereinafter called the "Purchase Price Payment Date"), all as more particularly provided herein.

Now, Therefore, in consideration of the mutual promises herein set forth, it is agreed as follows:

ARTICLE II

Sale and Purchase of Production Payment

2.1 The Company hereby agrees to sell and convey to Pennzoil, and Pennzoil hereby agrees to purchase from the Company, a production payment in an amount equal to the aggregate principal amount of the loans, plus accrued interest and commitment fee, outstanding under the Bank Credit Agreement, as amended, including by the amer liments there o pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1975, on the Purchase Price Payment Date, if any, which shall be specified by the Company as provided in Section 2.5. The purchase price of such production payment shall be an amount equal to the principal amount thereof.

2.2 The conveyance of production payment

- (a) shall entitle the holder thereof to an amount equal to interest (computed from the Purchase Price Payment Date) on the unpaid balance of the principal amount of the production payment from time to time remaining outstanding at a rate per annum (based on a year of 365 days) which shall be equal to 106% of the corporate base rate on 90-day commercial loans to its largest and most credit-worthy commercial borrowers in effect at The First National Bank of Chicago from time to time, such rate to change automatically and from time to time effective as of the date of each change in such corporate base rate.
- (b) shall provide that the production payment shall be dischargeable from eighty-five percent (85%) of the proceeds of production from or applicable to the Leasehold Interests.

- (c) shall be effective from and as of the day on which such production payment shall have been sold and conveyed to Pennzoil as provided in Section 2.3, and
- (d) shall grant to the holder thereof the rights provided for in, and shall be substantially in the form of, the form of conveyance of production payment attached hereto as Appendix 1.
- 2.3 Pursuant to the agreement of the Company and Pennzoil as hereinbefore provided, a production payment was sold and conveyed by the Company to Pennzoil by Instrument of Conveyance of Production Payment dated as of Merch 1, 1974 between the Company and Pennzoil, and thereafter amended and restated as of November 1, 1977. The terms of such production payment shall be further amended by an Amendment to Instrument of Conveyance of Production Payment to be dated as of April 1, 1979, whereby the aforesaid Instrument of Conveyance of Production Payment shall be amended and restated so as to be substantially in the form of Appendix 1 horoto.
- 2.4 Following the conveyance of such production payment, amended as aforesaid, by the Company to Pennzoil as provided in Section 2.3, the Company or its assignee shall have the right at any time prior to January 1, 1988, upon port of given to Pennzoil as provided in Section 2.5, to call upon Pennzoil aither
 - (a) to pay to the Company or its assignee the purchase price of such production payment, or
 - (b) to execute, acknowledge and deliver to the Company, at the Company's expense, such instruments or instruments as may be necessary or appropriate to release, relinquish and extinguish as fully discharged such production payment;

provided, however, that the right referred to in subsection (a) above may not be exercised by the Company unless all certificates of the value of Proven Hydrocarbon Reserves which are required to be delivered to The First National Bank of Chicago under the Bank Credit Agreement, as amended, including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1979, shall have been so delivered as required and an executed counterpart of each such certificate shall have been delivered to Pennzoil.

- 2.5 Notice of the election by the Company or its assignee to exercise either of the rights referred to in Section 2.4 shall specify the following:
 - (a) which of such rights the Company or its assignee shall have elected to exercise;
 - (b) a date for closing the transactions contemplated by such right which (i) shall be a business day not less than 30 days after the date upon which such notice is received by Pennzoi' unless Pennzoil shall agree in writing to an earlier closing date, and (ii) shall not be later than December 31, 1987; and
 - (c) if the Company or its assignee shall have elected to exercise its right to receive from Pennzoil the purchase price of the production payment, the aggregate principal amount of loans, plus accrued interest and commitment fee, outstanding under the Bank Credit Agreement, as amended, including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1979, as of the Purchase Price Payment Date, such amount to be subject to adjustment as required on and as of such date.

The election by the Company or its assignee, in the manner set forth above, of one of the rights referred to in Section 2.4 shall constitute a waiver of the other such right.

- 2.6 The closing of the transactions referred to in Section 2.5 shall be held at the general offices of Pennzoil located at Pennzoil Place, Houston, Texas, or at such other location in Houston, Texas as Pennzoil shall specify in writing to the Company or its assignee. The closing date for such transactions shall be the date specified in the notice to Pennzoil, and on such closing date
 - (a) if the Company or its assignee shall have elected to receive payment of the purchase price of the production payment, Pennzoil shall pay to the Company or its assignee such purchase price in immediately available funds; or

(b) if the Company or its assignee shall have elected to have the Company receive a reconveyance of the production payment, Pennzoil shall execute, have acknowledged and deliver to the Company such instrument or instruments as may be necessary or appropriate to effect such reconveyance to the Company.

ARTICLE III

Pennzoil's Right to Purchase Gas

- 3.1 The Company shall make available to Pennzoil for purchase by Pennzoil, or such other person, firm or corporation as Pennzoil may designate, all gaz produced from or applicable to the Leasehold Interests in accordance with the provisions set forth below.
- 3.2 When in Pennzoil's judgment gas reserves of sufficient quantity and deliverability to justify the construction of necessary facilities customarily utilized by purchasers of such gas in performance under gas purchase contracts have been developed on lands covered by any one or more of the Leasehold Interests, Pennzoil shall have the right to purchase such gas. Pennzoil's right to purchase gas hereunder may be exercised whenever and as often as in Pennzoil's judgment gas reserves of sufficient quantity and deliverability are developed on lands covered by one or more of the Leasehold Interests.
- 3.3 The price (hereinafter called the "Contract Price") to be paid by Pennzoil for gas purchased from the Company hereunder shall be fixed by the contract between the parties hereto for the sale and purchase thereof. Such contract shall contain specific prices competitive with the copprices then being included in similar contracts for the purchase of gas from the same ar prices shall not be less than the applicable ceiling rate set from time to time by the Fed Regulatory Commission.
- 3.4 The procedures to be employed by Penacoil and the Company on each exercise of such right to purchase gas are as follows:
 - (a) Pennzoil shall give the Company notice of its election to purchase gas, which notes shall also specify the purchaser if it be a person, firm or corporation other than Pennzoil, and shall transmit with such notice a form of gas purchase contract providing for the sale and purchase of such gas. After each such election to purchase, on or before the expiration of thirty (30) days following the date upon which the pertinent notice is received by the Company, the Company shall execute the gas purchase contract transmitted therewith and return such contract to Pennzoil, unless the Company shall give Pennzoil notice that the Company is of the opinion that the Contract Price provided for therein does not neet the requirements set forth in Section 3.3, in which event the parties shall promptly thereafter undertake in good faith negotiations to determine by agreement such Contract Price. If such Contract Price has not been determined within sixty (60) days after receipt of the initial notice and form of gas purchase contract by the Company, either the Company or Pennzoil shall have the right to cause such Contract Price to be determined by arbitration as provided in Section 3.7.
- 3.5 When (i) in the Company's judgment gas reserves of a sufficient quantity and deliverability to justify the construction of necessary facilities customarily utilized by purchasers of such gar in performance under gas purchase contracts have been developed on lands covered by any one or more of the Leasehold Interests, and (ii) Pennzoil shall not theretofore have given the Company notice of its election to purchase such gas, then and in such event the Company shall have the right to sell such gas in an artifact with and subject to the provisions of Section 3.6.
- correctures to be employed by the Company and Pennzoil on each exercise by the Comunder Section 3.5 are as follows:
 - tree entering into any contract for the sale of such gas, the Company shall give tice of its desire to sell such gas and shall transmit with each such notice a form release contract providing for the sale and purchase of such gas.

- (b) Pennzoil shall have a period of thirty (30) days from and after its receipt of such notice from the Company within which to elect to purchase such gas. If Pennzoil elects to purchase such gas, it shall give the Company notice to such effect and shall execute such gas purchase contract and return the same to the Company unless Pennzoil shall be of the opinion that the Contract Price provided for therein does not meet the requirements set forth in Section 3.3, in which event the parties shall promptly thereafter undertake in good faith negotiations to determine by agreement such Contract Price. If Pennzoil and the Company are unable to reach agreement on such Contract Price within sixty (60) days after receipt by Pennzoil of the aforesaid notice, either Pennzoil or the Company shall have the right to cause such Contract Price to be determined by arbitration as provided in Section 3.7.
- (c) In the event Pennzoil does not elect to purchase such gas on or before the expiration of thirty (30) days after receipt of such notice from the Company, the Company shall have the right for a period of sixty (60) days from and after the expiration of thirty (30) days next following the receipt by Pennzoil of such original notice from the Company within which to execute with a third party a contract for the sale of such gas in the form submitted to Pennzoil with the original notice from the Company. If such contract is not executed within such sixty (60) day period, such gas shall not thereafter be sold by the Company to a chird party, without first making the same available to Pennzoil in accordance with the provisions hereof.
- 3.7 If, pursuant to the provisions hereof, either party elects to cause the Contract Price to be provided in a gas purchase contract to be determined by arbitration, such party shall give the other party notice of such election, which notice shall also name one arbitrator. The party receiving such notice shall, within ten (10) days thereafter, by notice to the other, name the second arbitrator. The two (2) arbitrators so appointed shall name the third, or upon their failing to do so within ten (10) days, the third arl itrator may, upon request of either the Company or Pennzoil, be appointed by the person who at the time is the chief judge of the United States District Court for the Southern District of Texas. The arbitrators so appointed shall promptly hear and determine (after giving the parties due notice of hearing and a reasonable apportunity to be heard) the question submitted, and shall render their decision within sixty (80) days after completion of the hearing. The decision of the arbitrators, or a majority thereof, made in writing shall be final and binding upon Pennzoil and the Company. The expenses of arbitration, including reasonable compensation to the arbitrators, shall be borne equally by Pennzoil and the Company, except that each party shall bear the compensation and expenses of its counsel, witnesses and employees.

ARTICLE IV

General

- 4.1 All notices and other communications provided for herein shall be in writing and shall be delivered to the addressee thereof at Pennzoil Place, P. O. Box 2967, Houston, Texas 77001, or to such other address as the addressee shall have designated by written notice to the party giving such notice or making such communication. The date upon which any such notice or other communication is delivered shall be deemed to be the date the same is received.
- 4.2 The terms and provisions of Article III of this Agreement shall constitute covenants running with the lands and leases affected hereby and shall be birding upon and shall inure to the benefit of the parties hereto, their respective successors and assign.

IN WITNESS WHEREOF, this Restated Production Payment Agreement amends and restates the Production Payment Agreement dated as of March 1, 1974, as heretofore amended and restated, and has been executed in multiple counterparts as of April 1, 1979.

| Roger Wernette. Sue J. Prysan | Pennzoil Louisiana and Texas Offsmare, Inc. By Miresulf L. Treasurer ATTEST: |
|--|---|
| | Secretary Secretary |
| | [CORPORATE SEAL] |
| Witnesses: Roga Wernette Sue J. Pryson | By Company Group Vice President |
| | ATTEST: Secretory |

THE STATE OF TEXAS COUNTY OF HARRIS SS:

corpo:ation.

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 9th day of April, 1979:

(Lov

Before me appeared J. N. Averett, Jr., to me personally known, who, being by me duly sworn, say that he is the Treasurer of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corportion, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and to the instrument was signed and sealed in behalf of the corporation by authority of its Poard of

Directors and that J. N. AVERETT, Jr. acknowledged the instrument to be the free act and deed of the

(Louisiana)

Before me appeared HAROLD E. SORTOR, to me personally known, who, being by me duly sworn, did say that he is a Group Vice President of Pennzon. Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and seale in behalf of the corporation by authority of its Board of Directors and that HAROLD E. SORTOR acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared J. N. AVERETT, JR., known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be the Treasurer of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corp. ation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

Texas)

Before me on this day personally appeared HAROLD E. SORTOR, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of Pennzoil Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

In Witness Whereof, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 9th day of April, 1979.

[NOTARIAL SEAL]

Notary Public in and for Harris County, Texas My Commission Expires

Colie I Daya

CELIA I. GARZA

Notary Public in Herris County for the State of Texes

My Commission Expires September 30, 1380

APPENDIX 1

TO

PRODUCTION PAYMENT AGREEMENT BETWEEN

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC. AND

PENNZOIL COMPANY

CONVEYANCE OF PRODUCTION PAYMENT

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this 'Conveyance') dated as of March 1, 1974, as amended and restated as of April 1, 1979, between Pennzoil Louisiana and Texas Offshore, Inc., a Delaware corporation (the "Grantor"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas, and Pennzoil Company, a Delaware corporation (the "Grantee"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas.

WITNESSETH:

Whereas, the Grantor desires to sell and the Grantee desires to purchase the Production Payment hereinafter described.

Now, Therefore, it is agreed by and between the Grantor and the Grantee as follows:

- Certain Definitions. The following terms shall be used in this Conveyance with the meanings given below:
 - (a) "Accounting Month" shall mean any monthly period commencing with and including the 21st day of any calendar month and ending on and including the 20th day of the nest succeeding calendar month.
 - (b) "Effective Date" shall mean 7 o'clock, A.M., on the date of this Couveyance, according to the time then in effect at the location of each Subject Interest.
 - (c) "Exhibit A" shall mean Exhibit A attached to this Conveyance and hereby made a part hereof for all purposes.
 - (d) "Hydrocarbons" shall mean all oil, gas and other hydrocarbons or any combination of one or more of such substances appropriate to the context in which such term is used.
 - (e) "Lease" shall mean any oil and gas lease or sub-lease, described in Exhibit A.
 - (f) "Proceeds Commencement Date" shall mean 7 o'clock, A.M., according to the time then in effect at the location of each Subject Interest, on the date upon which the purchase price is paid by the Grantee to the Grantor is provided in the Production Payment Agreement dated as of March 1, 1974, as amended and restated as of April 1, 1979, between the Grantor and the Grantee.
 - (g) "Subject Hydrocarbons" shall mean all Hydrocarbons in and under, and which may be produced and saved from, and which shall accrue and be attributable to, the Subject Interests from and after the Proceeds Commencement Date and throughout the period specified in this Conveyance for the term of the Production Payment.
 - (h) "Subject Interests" shall mean the undivided percentage interests of the Grantor in the Leases, and, in addition thereto, each and every other kind and character of right, title, claim or interest which the Grantor now has in, to or under the Leases, which interests or Leases are

either specifically or generally described in Exhibit A, all as the same shall be enlarged by the discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and all renewals and extensions of any of the same, but there shall not be included within the Subject Interests any rights which the Grantor now has or may hereafter obtain with respect to production attributable to the interests of nonconsenting parties under any operating agreement, unit operating agreement, contract for development or similar instrument.

2. Conveyance of the Production Payment. The Grantor, for valuable consideration, the receipt adequacy and sufficiency of which are hereby acknowledged, by these presents does hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, as a production payment, an undivided eighty-five percent (85%) interest in the proceeds from the sale of the Subject Hydrocarbons; subject, however, to the provisions of Paragraph 3 of this Conveyance.

The aforesaid eighty-five percent (85%) interest in the proceeds from the sale of the Subject Hydrocarbons, as granted, bargained, sold, conveyed, assigned, transferred and delivered unto the Grantee by the terms of this Conveyance, shall be the "Production Payment."

To Have and to Houd the Production Payment, together with all and singular the rights, privileges and appurtenances thereunto in any way belonging, unto the Grantee, its successors and assigns, forever.

- 3. Certain Provisions Governing Subject Hydrocarbons. All the provisions of this Conveyance shall be subject to the following:
 - (a) For the discharge of the Production Payment the Grantee shall look exclusively to the proceeds of Subject Hydrocarbons, and the Grantor shall not be liable for such discharge.
 - (b) The Production Payment shall not be dischargeable from the proceeds of any products resulting from any manufacturing, processing or refining operation, except to the extent of that portion of such proceeds which represents the fair market value at the wellhead of the Subject Hydrocarbons used in such manufacturing, processing or refining operation.
 - (c) The Production Payment shall not be dischargeable out of any bonus which the Grantor shall receive for any Lease or assignment of any of the Subject Interests or out of any payments made to the Grantor in connection with the drilling or deferring of drilling of any well on any of the Subject Interests or in connection with any adjustment of any well and leasehold equipment upon unitization of any of the Subject Interests.
 - (d) There shall not be included in the Subject Hydrocarbons any Hydrocarbons unavoidably lost in the production thereof or produced and saved from any of the Subject Interests and used by the Grantor in conformity with good field practices for drilling and production operations (including gas injection, secondary recovery, pressure maintenance, repressuring or cycling operations) conducted for the purpose—producing Subject Hydrocarbons from the Subject Interests or from any unit to which the Subject interests are committed, but only so long as such Hydrocarbons are so used.
 - (e) So long as and to the extent that the same may be required by applicable laws and regulations, in the case of any Lease from the United States of America included in the Subject Laterests, from which the average production of oil per well per day averaged on the monthly basis is 15 barrels or less, the obligation to pay and the right of the Grantee to receive the proceeds of oil produced from such Lease shall be suspended until said average production of oil per well per day exceeds said minimum amount, and such suspension shall apply separately to any zone or portion of such Lease segregated for computing government royalty.
- 4. Amount and Term of the Production Payment. The Production Payment shall continue and remain in full force and effect until such time as the Grantee shall have received and realized,

out of the proceeds from the sale of the Subject Hydrocarbons, free and clear of all development, operating, mining, producing, treating, processing, handling, storing, marketing, transporting and other costs and expenses of every land whatsoever, the full aggregate sum of the following amounts:

- (a) The aggregate principal amount of loans, plus accrued interest and commitment fee, outstanding on the Proceeds Commencement Date under the Bank Credit Agreement, dated as of March 1, 1974, as thereafter—ended including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1979, among the Grantor, The First National Bank of Chicago and the other banks named therein (hereinafter called the "Primary Sum"); plus
- (b) An amount computed at a rate per minum (on the basis of a 365-day year) which shall be equal to 106% of the corporate base rate on 90-day commercial loans to its largest and most credit-worthy commercial borrowers in effect at The First National Bank of Chicago from time to time, such rate to change automatically and from time to time effective as of the date of each change in such corporate base rate, from and including the Proceeds Commercement Date on the unliquidated balance from time to time of the Primary Sum (the first such computation to be made on the 21st day of the mont, next following the Proceeds Commencement Date for the period from and including the Proceeds Commencement Date to and including the 20th day of the month next fc!!owing the Proceeds Commencement Date on the amount of such unliquidated balance on the Proceeds Commencement Date, and subsequent computations to be made monthly on the first day of each Accounting Morath for the preceding Accounting Month on the amount of such unliquidated balance on the first day of such preceding Accounting Month); plus
- (c) An amount equal to the aggregate of all amounts which are paid by the Grantee to any state or political subdivision thereof on account of ad valorem, transfer, mortgage, gross production, gross receipts, income, profits, severance, occupation, sales, use, franchise and other taxes and assessments of any kind whatsoever, including penalties and interest, imposed or assessed with respect to or measured by or charged against or attributable to the Production Payment, the Subject Hydrocarbons or the proceeds thereof, together with an amount equal to interest on the unliquidated amount thereof at the rate of 7% per annum from the date of payment of each such amount by the Grantee; provided, however, that the Grantor shall have the right at its expense to contest any such taxes or assessments in good faith on behalf of the Grantee.

IT BEING THE INTENTION OF THE GRANTOR AND THE GPANTEE IN ENTERING INTO THIS CONVEYANCE that the Grantee shall be entitled to receive and realize out of the Subject Hydrocarbons the full arguegate sum of the amounts described in subparagraphs (a) and (b) of this Paragraph 4, free and clear of all costs and expenses with respect to the Subject Interests and the Subject Hydrocarbons (which expenses and costs shall be borne by the Grantor) and over and above all taxes, expenses and costs of the character described, specified or referred to in subparagraph (c) of this Paragraph 4.

- 5. Termination of the Production Payment. When the full aggregate sum of the amounts specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance (as increased as expressly provided by the terms of this Conveyance) has been received by the Grantee, the Production Payment shall forthwith terminate and all interest therein shall immediately revert to and become vested in the Grantor to the same extent and with the same force and effect as if this Conveyance had not been made. The Grantee agrees that, upon such termination, it shall execute or cause to be executed, upon the request and at the expense of the Grantor, such instruments as may be necessary or appropriate to evidence the termination of the Production Payment.
- 6. Application of the Proceeds of Subject Hydrocarbons. For all purposes of this Conveyance, the proceeds of Subject Hydrocarbons actually received by the Grantee prior to the close of business on 20th day of the month next following the Proceeds Commencement Date shall be deemed to have been received and applied immediately after the opening of business on the 21st day of such month, and thereafter, such proceeds actually received by the Grantee after the opening of business on the first day of each Accounting Month and prior to the close of business on the last day of such Account-

ing Month shall be decined to have been received and applied immediately after the opening of business on the first day of the next succeeding Accounting Month; provided, however, that, if any date of application specified above shall be a Saturday, Sunday or legal holiday under the law of the jurisdiction in which such proceeds are actually received by the Grantee, such proceeds shall be deemed to have been received and applied on the last business day next preceding such Saturday, Sunday or legal holiday in such jurisdiction, but the amount of such proceeds to be applied pursuant to subparagraph (a) of this Paragraph 6 shall nevertheless be the amount under subparagraph (b) of Paragraph 4 of this Conveyance accrued up to (but not including) such Saturday, Sunday or legal holiday. Such proceeds shall be deemed to have been applied on the first day of each Accounting Month as follows:

- (a) First, to the amount referred to in subparagraph (b) of Paragraph 4 of this Conveyance accrued up to (but not including) the date of such as plication, and
- (b) Second, to the amount referred to in subparage with (c) of said Paragraph 4, to the extent then ascertained, and
 - (c) Third, to the reduction of the unliquidated balance of the Primary Sum;

provided, however, u.u., in the event such proceeds so applied on the first day of any Accounting Month shall be insufficient to cover the full amount specified in subparagraph (a) of this Paragraph 6, the unliquidated balance of the Primary Sum shall forthwith be increased (to the extent permitted by law) by an amount equal to the amount of such deficiency.

Within 10 days after the date of each such application by the Grantee of any proceeds of Subject Hydrocarbons, the Grantee will furnish or cause to be furnished to the Grantor a written statement showing the application of such proceeds in accordance with the provisions of this Paragraph 6.

- 7. Marketing the Subject Hydrocarbons. The Grantor, at its expense, shall market, or cause to be marketed, the Subject Hydrocarbons for the best price reasonably available when marketed, and it shall not be necessary for the Grantee to join in any Sales Contracts (as hereinafter defined) or any amendment thereof. In the event the Grantor shall take any of the Subject Hydrocarbons for its own use, the Grantor shall pay to the Grantee, on or before the last day of the Accounting Month during which such taking occurs, a price equal to the fair market value at the wellhead of the Subject Hydrocarbons so taken.
- 8. Sales Contracts. The Grantor will duly perform all obligations binding on it under all contracts and agreements for the sale of Subject Hydrocarbons or any portion thereof, whether presently existing or hereafter entered into (all such contracts and agreements being the "Sales Contracts"), in accordance with the terms hereof and will take all appropriate measures to enforce the performance under each of the Sales Contracts of the obligations of the purchaser thereunder. All Subject Hydrocarbons sold by the Grantor, whether pursuant to the terms of the Sales Contracts or otherwise, shall be delivered by the Grantor to the purchasers thereof into the pipelines to which the wells producing such Subject Hydrocarbons may be connected or to such other point of purchase as is reasonably required in the marketing of such Subject Hydrocarbons.
- 9. Withholding and Restitution of Proceeds of Subject Hydrocarbons. All obligations of the Grantor hereunder shall be subject to the applicable provisions of the Natural Gas Act and all applicable rules and regulations of the Federal Energy Regulatory Commussion. Rates permitted under the Natural Gas Act and said rules and regulations to be paid for gas included in Subject Hydrocarbons shall be controlling if varying from prices established in Sales Contracts or if different from the fair market price at the wellhead. The Grantor shall be entitled to use its reasonable discretion in making filings with the Federal Energy Regulatory Commission affecting Subject Hydrocarbons. If any proceeds of Subject Hydrocarbons shall be withheld for any reason whatsoever, including, without limitation, withholding by the Grantor pending final approval of rate increases filed with the Federal Energy Regulatory Commission, the Grantee shall not be deemed to have received or realized such proceeds until, and only to the extent that, such proceeds shall actually have been received by the

Grantee. If, at any time before or after receipt of the full aggregate sum specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance, the Grantee shall be compelled, for any reason whatsoever, to make any payment or restitution, or such payment or restitution shall be made for the account of the Grantee, on account of any proceeds of Subject Hydrocarbons, which proceeds have already been applied pursuant to the provisions of Paragraph 6 of this Conveyance, the unliquidated balance of the Primary Sum shall forthwith be increased by the amount of the proceeds so paid over by or for the account of the Grantee plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties, provided, however, that if at the time of such payment or restitution the entire Primary Sum shall have been liquidated, the Primary Sum shall be deemed not to have been so liquidated and the unliquidated balance of the Primary Sum shall be thereupon deemed to be in the amount of the proceeds so paid over plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties.

- 10. Protection to Purchasers of Production. No person purchasing or taking or processing any Subject Hydrocarbons shall be required to take notice of, or keep informed as to, the discharge or termination of the Production Payment until the actual receipt by such person of written notice advising such person of such discharge or termination.
- 11. Payment of Taxes. The Grantor will tender and pay, or cause to be paid, punctually, before they shall become delinquent (or as to any thereof which are being contested in good faith, promptly after the final determination of such contest), all ad valorem taxes (or taxes imposed in lieu thereof) and all transfer, mortgage, gross production, mining, severence, occupation, excise, sales, use, franchise, income and other taxes and assessments of every kind and character whatsoever (other than federal income taxes of the Grantee) imposed or assessed with respect to or measured by or charged against or attributable to the Subject Interests, the Production Payment or the Subject Hydrocarbons, or the proceeds thereof, or against the Grantee by reason of its acquisition or ownership of the Production Payment or against any assignee of the Production Payment by reason of his or its interest as assignee, together with any interest or popularly payable in connection therewith.
- 12. Covenants of the Grantor. Until such time as the Production Payment has terminated as provided in Paragraph 5 of this Conveyance, the Grantor will, at its expense and regardless of who may be the operator or operators of the Subject Interests, cause:
 - (a) The Subject Interests to be maintained and developed and continuously operated for the production of Hydrocarbons in a good and workmanlike manner and in accordance with sound field practices and all applicable federal, state and local laws, rules and regulations (except those being contested in good faitn); provided, however, if the Grantor in good faith and in conformity with sound field practices elects not to participate in any operation which is to be conducted under the terms of an operating agreement, unit operating agreement, contract for development or similar instrument entered into by the Grantor or its predecessors in title pertaining to the development and operation of any Subject Interest (or any portion thereof) and which allows any consenting party or parties to conduct nonconsent operations, then, during the period or periods in which the consenting party or parties are entitled to receive Subject Hydrocarbons attributable to such Subject Interest (or portion thereof) by the terms of such instrument pending the recoupment or receipt of the sums determined thereunder, the Grantee shall not be entitled to receive or be deemed to have received (but only during such period or periods) proceeds from Subject Hydrocarbons attributable to such Subject Interest (or portion thereof) to the extent affected thereby.
 - (b) All rentals, royalties, and all liabilities of any kind or nature, including, without limitation, all liabilities for labor, material, supplies and equipment incurred in, or arising from, the administration, operation or development of, or the gathering, producing, treating, processing, storing, marketing or transporting of, the Subject Hydrocarbons, to be paid punctually when due, or, as to any thereof which are being contested in good faith, promptly after the final determination of such contest.

- (c) All machinery, equipment and facilities of any kind now or hereafter located on the Subject Interests and necessary or useful in the operation thereof for the production of Hydrocarbons therefrom to be provided and to be kept in good and effective operating condition and all repairs, renewals, replacements, additions and improvements thereof or thereto needed for such purpose to be duly made or provided.
- (d) All necessary and proper steps to be taken diligently to protect and defend the Subject Interests, the Subject Hydrocarbons and the proceeds thereof against any adverse claim or demand, including, without limitation, the employment or use of counsel for the prosecution or defense of litigation and the contest, settlement, release or discharge of any such claim or demand.
- (e) Written notice to be given to the Grantee of every adverse claim or demand made by any person (including any government or governmental agency) affecting the Subject Interests, the Subject Hydrocarbons or the proceeds thereof in any manner whatsoever, or of any proceeding instituted with respect thereto.
- (f) The interest of the Grantor in each of the Subject Interests to be kept free and clear of liens, charges and encumbrances of every character, other than those which are of a character customarily found in connection with comparable drilling and producing operations and which do not materially adversely affect the operation of the Subject Interests and other than those becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Conveyance.
- (g) Subject to the provisions of Paragraph 19 of this Conveyance, all Leases included in or relating to the interests described or referred to in Exhibit A and under which the Grantor is lessee, sub-lessee, or assignee to be maintained in full force and effect without amendment, modification or waiver of any provision of any such Lease which might materially adversely affect the rights of the Grantee under this Conveyance.

Anything contained in this Paragraph 12 to the contrary not withstanding, the Grantor shall not be obligated to perform undertakings performable only by the operator and which are beyond the control of the Grantor, with respect to individual Subject Interests of which it is not the operator. In such case, however, the Grantor will use its best efforts to secure the performance of any such undertakings.

- 13. Access to the Subject Interests. The Grantor shall permit any one or more representatives designated by the Grantee to make any inspection of the Subject Interests at any reasonable time as the Grantee may reasonably request. In addition the Grantor shall furnish to the Grantee such detailed information as the Grantee may reasonably request in writing concerning the Subject Interests, the operation thereof and the production and sale of the Subject Hydrocarbons.
- 14. Pooling and Unitization. Anything herein to the contrary notwithstanding, the Grantor may, without the consent of the Grantee, pool, combine, communitize or unitize any Subject Interest (or any portion thereof) from time to time with other leasehold, mineral or other interests to form units when, in the judgment of the Grantor, it is necessary or advisable to do so in order to facilitate the orderly development or operation (including repressuring, pressure maintenance, cycling, water flood and secondary recovery operations) of any Subject Interest (or any portion thereof). Forthwith, after the formation of any unit, the Grantor shall furnish to the Grantee a true copy of the pooling agreement, unit designation or any other instrument creating such unit, together with, upon request of the Grantee, any operating agreement relating thereto. The interest in any such unit accruing or attributable to any Subject Interest (or any portion thereof) included therein shall be subject to the Production Payment in the same manner and with the same effect as though such unit were in existence on the Effective Data. The signature or joinder of the Grantee shall not be necessary to the instruments evidencing the pooling and unitization herein authorized.

- 15. Default by the Grantor and the Remedies of the Grantæ. Should the Grantor default in the performance or observance of any covenant or agreement provided in this Conveyance to be performed or observed by the Grantor, and should such default continue unremedied for 30 days (or for such longer period as may be necessary to complete performance diligently commenced within 30 days) from the delivery to the Grantor of written notice of such default, the Grantee may, in addition to pursuing any or all other remedies available at law or in equity:
 - (a) Effect performance or observance of such covenant or agreement on behalf of and at the expense of the Grantor, in which event the Grantee may expeud funds reasonably necessary to effect such purposes and shall be entitled to reimbursement therefor from the Grantor, together with interest on such amounts at the rate of 7% per annum from the date of such expenditure to the date of reimbursement.
 - (b) Upon written notice to the Grantor, succeed to all rights of the Grantor with respect to the possession, operation and development of the Subject Interests, and may use in connection therewith all of the appurtenant property, both real and personal, situated upon or used or held for future use in connection therewith, or in connection with the production, storing, transporting or sale of the Subject Hydrocarbons, and the Grantee shall have the right, on behalf of and for the account of the Grantor, to sell the Subject Hydrocarbons and the Grantor shall, upon demand, reimburse the Grantee for all amounts so expended by the Grantee, together with interest on such amounts at the rate of 7% per annum from the date of such expenditure to the date of reimbursement.
 - (c) Apply to a court of equity for the specific performance or observance of such covenant or in aid of the execution of any power herein granted and for the appointment of a receiver of the Subject Interests and the Subject Hydrocarbons.
- 16. Exercise and Termination of the Remedies of the Grantee. All rights and remedies granted by the provisions of Paragraph 15 of this Conveyance and otherwise belonging to the Grantee shall be cumulative and no exercise of any right or remedy shall preclude the exercise of any other right or remedy. Any right or remedy may be exercised on one or more occasions without exhaustion thereof. All right and remedies shall terminate:
 - (a) When the Production Payment is discharged and all amounts then due and payable to the Grantee pursuant to the terms of Paragraph 15 of this Conveyance (including amounts payable for interest) shall have been paid in full; or
 - (b) At any earlier time when all defaults of the Grantor shall have been remedied and all such amounts shall have been duly paid in full, without prejudice, however, to the exercise of any rights and remedies conferred upon the Grantee upon any subsequent failure of the Grantor to perform or observe any of the covenants or agreements herein provided to be performed or observed by the Grantor.
- 17. Reports to the Grantee. While the Production Payment remains in force and effect, the Grantor will, at its expense, furnish to the Grantee within 45 days after the end of each calendar month, a report showing for such calendar month (i) the quantity of Subject Hydrocarbons produced and sold or taken by the Grantor for its use and (ii) the gross proceeds received from the sale, or attributable to such taking, of such Subject Hydrocarbons.
- 18. Warranty of Title. The Grantor covenants and warrants that at the time of the execution and derivery of this Conveyance, the Grantor has good and marketable title to the Subject Interests, rubject only to such exceptions, charges, liens or encumbrances, if any, as are set forth or referred to in Exhibit A; that the Grantor has the power and right to create and confer upon the Grantee the

rights of the Grantee granted by the terms of this Conveyance; and that there are no suits or proceedings pending or, to the knowledge of the Grantor, threatened, against or affecting the Subject Interests before any court or by or before any governmental commission, bureau or any regulatory authority which if decided adversely to the interests of the Grantor shall materially affect the rights of the Grantee pursuant to this Conveyance. The Grantor hereby binds itself, its successors and assigns, to warrant and defend forever the rights under this Conveyance of the Grantee, its successors and assigns, against the lawful claims and demands of every person who shall claim the same or any part thereof, and the Grantor will indemnify and hold the Grantee harmless against and from all loss suffered by the Grantee resulting from any failure of or defect in the Grantor's title to the Subject Interests and the production of Hydrocarbons as warranted as aforesaid. The covenants and warranties of this Paragraph 18 shall survive the discharge of the Production Payment.

19. Transfer of the Subject Interests. Except for the conveyance of production payments becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Corveyance, prior to such termination the Grantor shall not, without the written consent of the Grantee, make or permit any surrender, abandonment, sale, conveyance, assignment, leaso or sublease, in whole or in part, of any of the Subject Interests; provided, however, that the Grantor shall have the right to surrender, abandon or otherwise tenninate (either voluntarily or involuntarily) any of the Subject Interests (or any portion thereof) when there is no well located thereon which is capable of producing Hydrocarbons in paying quantities and when the Grantor has reasonably determined that (i) no existing well may be recompleted thereon as a well capable of producing Hydrocarbons in paying quantities, (ii) no additional well could be drilled and completed thereon which would be capable of producing Hydrocarbons, over and above the cost of drilling and completing same, in paying quantities, and (iii) no pressure maintenance or secondary recovery operations may be instituted with respect to any well thereon which would cause such well to be capable of producing Hydrocarbons, over and above the cost of instituting and completing such pressure maintenance or secondary recovery operation, in paying quantities. For the purposes of the Paragraph 19, the words "in paying quantities" as applied at any time to the producing capability of any well shall mean that, in the light of conditions existing at the time of determination and which reasonably appear to be not temporary, such well is producing or is or will be capable of producing Subject Hydrocarbons whose aggregate value exceeds or will exceed the Grantor's share (as the owner of the Subject Interest on which such well is located) of the direct cost of operating such well.

In each case where the preceding paragraph of this Paragraph 19 permits the Grantor to surrender, abandon or otherwise terminate any of the Subject Interests (or any portion thereof), then (a) any of the Subject Interests (or any portion thereof) which are so surrendered, abandoned or terminated shall upon surrender, abandonment or termination cease to be a Subject Interest, and (b) the Grantor, in lieu of surrendering, abandoning or terminating any of the Subject Interests (or any portion thereof), may sell, assign, sublease, farmout or convey the same free and clear of the Production Payment without the consent of the Grantee, and the Grantee shall not have the right to any proceeds from such sale, assignment, sublease, farmout or conveyance; provided, however, that any interest in any such Subject Interest (or any portion thereof) which may be excepted from such sale, assignment, sublease, farmout or conveyance and retained by the Grantor shall not be released thereby from the Production Payment.

- 20. Exception to Payments. The acceptance by the Grantee of any payment hereunder shall not preclude the Grantee from taking exception to the correctness of the amount thereof; provided, however, that any such exception must be specified by the Grantee in a notice given to the Grantowithin 12 months after the date such payment is made.
- 21. Transfer of the Production Payment. Nothing contained in this Conveyance shall be construed to limit or restrict in any way the right of the Grantee to sell, conver, assign or mortgage the Production Payment, in whole or in part.

- 22. Rights of Mortgagee or Tristee. If the Grantee shall at any time execute a mortgage or deed of trust covering all or part of the Production Payment, the mortgagee or trustee therein named or the holder of any obligation secured thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to exercise all of the rights, remedies, powers and privileges conferred upon the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained hereunder by the Grantor, but the provisions of this Paragraph 22 shall in no way be deemed or construed to impose upon the Grantor any obligation or liability undertaken by the Grantee under such mortgage or deed of trust or under the obligation secured thereby.
- 23. Notices. Any notice, request, demand, report or other instrument which may be required or permutted to be given or furnished to or served upon either party hereto or other person succeeding to any interest of either party hereto shall be deemed sufficiently given or furnished and served if in writing and delivered to such party or person or to an officer of such party or person or deposited in the United States mail in a sealed envelope, registered, with postage prepaid, addressed if to the Grantor to Pennzoil Place, P.O. Box 2967, Houston, Texas, and if to the Grantee to Pennzoil Place, P.O. Box 2967, Houston, Texas, or to such other address as the party or person to be addressed shall have designated by written notice to the party or person giving such notice or furnishing such report or making such request or demand.
- 24. Further Assurances. So long as permitted by applicable law so to do, the Grantor will execute and deliver all such other and additional instruments, notices, releases, acquittances and other documents, and will do all such other acts and things, as may be necessary more fully to assure to the Grantee all the rights and interests herein and here! granted or intended so to be.
- 25. Binding Effect. All the covenants and agreements of the Grantor contained in this Conveyance shall be deemed to be covenants running with the land and shall be binding upon all successors in interest to, and all assigns of, the Grantor and shall inure to the benefit of all successors in interest to, and all assigns of, the Grantee. The provisions of this Conveyance shall inure to the benefit of and be binding upon the respective successors and assigns of the Grantor and the Grantee from and after acceptance by the Grantee. All references herein to either the Grantor or the Grantee shall include their respective successors and assigns.
- 26. Release or Failure of Title. No release by the Grantee of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect or reducing the amount of the proceeds from the sale of production of Hydrocarbons out of which the Production Payment is dischargeable or reducing the amount of the Production Payment or creating any offset or other prejudice to the Production Payment, and the Production Payment shall continue in full force and effect as to the remainder of the Subject Interest until the full amount of the Production Payment has been received by the Grantee as herein provided.
- 27. Final Termination. Notwithstanding the other provisions hereof, the Production Payment shall, in any event, terminate 1 day prior to the expiration of 21 years after the death of the survivor of all the descendants of Theodore Roosevelt, late President of the United States, who are living on the date hereof.
- 28. Headings. Paragraph headings used in this Conveyance are for convenience only and shall not affect the construction of this Conveyance.

IN WITNESS WHEREOF, this Restated Instrument of Conveyance of Production Payment amends and restates the Instrument of Conveyance of Production Payment dated as of March 1, 1974, as heretofore amended and restated, and the parties have caused this Restated Instrument of Conveyance of Production Payment to be duly executed in the presence of the undersigned witnesses on the dates of their respective acknowledgments annexed hereto, in several counterparts, each of which is an original and all of which are identical. Each of such counterparts shall for all purposes be deemed

| Witnesses: | PETINZOIL LOUISIANA AND TEXAS OFFICHORE, INC. |
|---|---|
| ES EXPLICATE ES ES TRUBBES : ES EXPLICATE ES ES EXPLICATES ASSET | By Treasurer |
| EX TAIL 19 ET - TELEBROOM OF THE RESERVE TO A STREET THE | ATTEST: |
| | Secretary |
| | [CORPORATE SEAL] |
| Witnesses: | |
| | Pennzoil Company |
| TO A TO A DESCRIPTION OF THE PARTY OF THE PA | By Group Vice President |
| \$100 (000,000 to \$100,000 to \$ | ATTEST: |
| | Secretary |
| | [CORPORATE SEAL] |

(Louisiana)

Before me appeared J. N. AVERETT, JR., to me personally known, who, being be me duly sworn, did say that he is the Treasurer of PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that J. N. AVERETT, JR. acknowledged the instrument to be the free aut and deed of the corporation.

in and for the County and State aforesaid, hereby certify that, on this 9th day of April, 1979:

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting

(Louisiana)

Before me appeared HAROLD E. SORTOR, to me personally known, who, being by me duly sworn, did say that he is a Group Vice President of PENNZOIL COMPANY, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Lirectors and that HAROLD E. SORTOR acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared J. N. AVERETT, JR., known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be the Treasurer of PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared HAROLD E. SCRTOR, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of PENNZOIL COMPANY, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

In WITNESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 9th lay of April, 1979.

> Notary Public in and for Harris County, Texas My Commission Expires

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT

| BLM or Stere Lease No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
|---------------------------|-------------------|--|--|
| | S | Federal leases are as shown on Official er Continental Shelf Leasing Maps) | |
| OCS-G2839 | 12-1-74 | All of Block 352, West Cameron Area, West Addition, OCS Official Leasing Map, Louisiana Map No. 1A | 25.00% |
| OCS-G2224 | 2-1-73 | All of Block 532 West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 26.67% |
| OCS-G2225 | 2-1-73 | All oi Block 533 West Cameron Area, South Addition, Official Lessing Map, Louisiana Map No. 1B | 26.67% |
| OC\$-G3284 | 8-1-75 | All of Block 563, West Cameron Area, South Addition, as shown on OCS Official Leasing | |
| OCS-G2850 | 12-1-74 | Map, Louisiana Map No. 1B All of Block 608, West Cameron Area, South Addition, OCS Official Leasing Map, | 25.00% |
| OCS-G2559 | 5-1-74 | Louisiana Map No. 1B All of Block 617 West Cameron Area — South Addition, Official Lessing Map, | 6.66667% |
| OCS-G2859 | 12-1-74 | Louisiana Map No. 1B All of Block 236, East Cameron Area, South Addition, OCS Official Leasing Map, | 6.66667% |
| OCS-G2860 | 12-1-74 | Louisiana Map No. 2A All of Block 237, East Cameron Area, South Addition, CCS Official Leasing Map, | 16.66667% |
| OCS-G2882 | 12-1-74 | All of Block 125, South Marsh Lland Area, South Addition, OCS Official Losing Map, | 16.66667% |
| OCS-G2883 | 12-1-74 | Louisiana Map No. 3C All of Block 127, South Marsh Island Area, South Addition, OCS Official Leasing Map. | 6.66667% |
| OCS-G2587 | 5-1-74 | Louisiana Map No. 3C All of Block 128 South Marsh Island Area, South Addition, Official Leasing Map, | 6.65367% |
| OCS-G2884 | 12-1-74 | Louisiana Map No. 3C All of Block 129, South Marsh Island Area, South Addition, OCS Official Leasing Map, | 6.66667% |
| OCS-G2900 | 12-1-74 | Louisiana Map No. 3C All of Block 261, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 6.66667% 16.66667% |
| OC5-G3156 | 7-1-75 | All of Block 262, Eugene Island Area, as shown on OCS Official Leasing Map, Louisiana Map No. 4 | |
| OCS-G2607 | 5-1-74 | All of Block 312 Eugene Island Area, South Addition, Official Leasing Map, Louisiana | 10.66667% |
| OCS-G2317 | 2-1-73 | Map No. 23 All of Block 333 Eugene Island Area, South Addition, Official Leasing Map, Louisiana | 26.66667% |
| | | Map No. 4A | 26.66667% |

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| BLM or State Lease No. | Effective Date | Descripti u | Grantor's Undividen Percentage, Interest |
|---------------------------|-------------------|--|---|
| OCS-G2177 | 11-1-72 | All of Block 49 South Pass Area as shown on chicia. leaving map, Louisiana May 2000, 9 | 10.00% |
| State Lease No. 6316 | 5-13-7-2 | Tract 15018, to a classe portions of Blocks 57 and 58, So all Eass Area, Plaquemines Parish, Louisiana, belouing to the State of Louisiana and not under it have a lease on March 4, 1974 and being more particularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Louisiana | 20,0,4 |
| OCS-G2185 | 10-1 72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court micred December 13, 1965 in the U.S. v. Louisians No. 9 Original (382 U.S. 188), South Pass Area, South and East Addition, as shown on official leasing map, Louisiana Map No. 9A | 15,10% |
| OCS-C3193 | 775 | S½ of Block 59, Main Pass Area, that portion located in Zone 2 is that zone was defined in the agreement between the United States and the State of Louisiana, October 12, 1956 and landward of the Third Supplemental Decree I ine (404 U.S. 388 (December 20, 1971), OCS Official Leasing Map, misiana Map No. 10 | 25.90% |
| OCS-G3195 | 7-1-75 | Block 72, Main Pass Area, that period located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 21.00% |
| OCS-G2947 | 12-1-74 | All of Block 73, Main Pais Area CCS Official Leasing Map, Louisiana Map No. 10. That perion located more than one foot seaward of the Tuird Supplemental Decree Line (404 U.S. 388 (December 20, 1971)) | 21.00% |
| OCS-G2193 | 10-1-72 | All of Block 110 Main Pass Area as shown on official leasing map, Louisiana Map No. 10 | 10.00% |
| OCS-G3202 | 7-1-75 | All of Black 947. Ewing Bank Area, as shown on OU3 Official Protraction Diagram NH-15-12 | 21.00% |
| OCS-G3203 | 7-1-75 | All or block 903, Ewing Bank Area, as shown on Ocs Official Protraction Diagram MH-15-12 | 21.00% |
| OCS-G3206 | 7-1-75 | Al' of Block N663E63, Mobile South No. 2, as shown on OCS Official Leasing Map, NH-16-10 (Also known as Block 63, Mississippi Canyon Area, as shown on OCS Official Protraction Diagram N1 16-10) | 10.00% |

EXHIBIT A TO CONVEYANCE OF TODUCTION PASSMENT (Continued)

| OCS-G2366 8-1-73 All i Block A-474, High Island Area, South | |
|--|---------|
| OCS-G2366 8-1-73 All i Block A-474, High Island Area, South Addition. Official Leasing Map, Texas Mar No. 7B | 2.00% |
| OCS-G2372 8-1-73 All of Block A-489, High Island Area, South Addition. Official Least 17 Map, Texas M. No. 7B | 5.00% |
| | 6 40% |
| | 2.00% |
| OCS-G2701 7-1-74 All of Block A-40. Figh Island Area - South Addition, OCS Onti-ial Leasing Map, Texas Map No. 7B. | 15.00% |
| OCS-G2704 7-1-7/ All of Block A-54% high Island Area - South Addition, OCS (ficial Leasing Map, Texas | 10.00% |
| OCS-G2779 10-1-74 All of Block A-546, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCE 135 7-1-74 All of Block A-547, High Island - South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2706 7-1-74 All of Block A-548, High Island Area – South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2384 8-1-73 All of Block A-555, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 14500 |
| OCS-G2713 7-1-74 Al. of Block A-562, High Island Area – South Addition, OCS Official Leasing Map, Texas Map No. 7B | 12.715% |
| OCS-G2388 8-1-73 All of Block A-563, High Liand Area, South Addition. Official Leasing Map. Texas Map | 12.00% |
| OCS-G2389 8-1-73 All of Block A-564, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 1.200% |
| GCS-G2390 8.1-73 All of Block A-570, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B 2 | 20.00% |
| OCS-G2718 7-1-74 All of Block A-575 High Island Area — South Addition, OCS Official Leasing | 20.00% |
| Cause G2719 7-1-74 All of Block A 582 High Island Area — South Addition, OCS Official Learning Map, Texas Map No. 7B | 14.118% |

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT (Continued)

Grantor's

| PLM or State Lease No. | Fif-ctive | Description | Undividen Percentage Interest |
|---------------------------|-----------|--|-------------------------------------|
| OCS-G2398 | 8-1-73 | All of Block A-273, High Island Area, East Addition, South Extension Official Leasing Map, Texas Map | 13.50% |
| OCS-G2403 | 8-1-7? | All of Block A-279, High Island Arra. East Addition, South Extension. Official Leasing Map, Texas Map No. 702 | 17 00% |
| OC\$ G3314 | 479 | All of Block A-288, High Leand Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C | 17.00% |
| CCS-G2409 | 8-1-73 | All of Block A-312, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 18.11538% |
| OCS G2737 | 7-1-74 | All of Block A-316 High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 21.132% |
| OCS-G2414 | 8-1-73 | All of Mock A-323, High Island Area, Flast Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 14.00% |
| OCS-G2416 | 8-1-7? | Ali of Plock A-325, High Island Area, East Addition South Extension. Official Leasing View, Texas Map No. 7C | 68.57143% |
| OCS-G2418 | 8-1-73 | All of Block A 327, High Liland Area, East A Jdition, South Extension. Offic al Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-C2422 | 8-1-73 | All of Bask A-332, High Island Area, East Addition, Jouth Extension. Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G27% | 7-1-74 | All of Block A-339 11 th Island Area — East Addition — Buth Extension, OCS Official Leasing Map, Texas Map No. 7C | 34.00% |
| OCS-G2426 | 8-1-73 | All of Block A- 10, High Island Area, East Addition South Ext. 13, 10, Official Leusing May, Texas Map No. 7C | J4.00% |
| OCS-G2429 | 8-1-73 | All of Block A-301 1.5 h Island Area, East Addition, 3 oth Extension. Official Leasing Map. Totas Map No. 7C | 4.i.)462 5% |
| OCS-G2745 | 7-1-74 | Il of Block A-355, High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 11.33% |
| OCS-G2746 | 7-1-74 | A.' of Plock A-356, High Island Area – Bas' Addition – South Extension, OCS Official Leasing Map, Texas Map No. 7C | |
| OCS-G2766 | 7 1-74 | Ali of Block A-403 High Island Area — East Addition — South Extent In, OCS Official Leasing May, Texas May No. 70 | 10.00% |

EXHIBIT A
TO
CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| BLM or State Lease No. | Effective Date | Descripti, n | Grantor's Undivided Percentage Interest |
|---------------------------|-------------------|---|--|
| OCS-G3417 | 1-1-77 | That portion of Main Piss Blocks 72 and 74 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975, (U.S. vs Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971), as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 25.00% |
| OCS-G3409 | 1-1-77 | All of E. & 351, Eugene Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | 25 0046 |
| OCS-G3416 | 1-1-77 | That portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (U.S. vs. Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (44 U.S. 388, December 20, 1971), as shown on OCS Official Leasing Map, Louisiana Map No. 9 | 20.00% |
| OCS-G3500 | 8-1-77 | All of Block 253, West Cameron Area, as shown on OCS Official Leasing Map, Louisiana Map No. 1 | 30.00% |
| OCS-G2367 | 8-1-73 | S ½ of NE ¼ and SE ¼ of Block A-475, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G3756 | 6-1-78 | All of Block A-522, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 30.00% |
| OCS-G3757 | 6-1-78 | All o' Block A-551, High Island Area, South Addition, OCS Official Leasing Map, Texas Map, No. 7B | 16.67% |
| OCS-G3766 | 6-1-76 | ul of Block 510 West Comeron Area, South Addition, OCS Official Leasing Map, Louisi- ana Map No. 1B | 16.66% |
| OCS-G3786 | 3 ' 78 | All of Block 372, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisi- ana Map No. 4A | 15.00% |

This Conveyance is subject to the term and provisions of the above-described oil and gas leases and accordingly, the Subject Hydroc, there, and the proceeds therefrom shall be calculated after demucting the proportionate share of the analysis reserved in each such lease a ributable to Grantor's said undivided percentage interest in surb lease.

Lease Nos. OC3-C ... C9. OCS-C 2/25, OCS-G 2850, OCS-G 2559, OCS-G 2607, OCS-G 2617, OCS-G 3193, OCS-G 3195, CCS-G 2947 and OC s-G 3417 referred to above are each subject to an Operating Agreement between Mobil Oil Corporation, as operator, and the Grantor, et al., as non-operators.

Lease No. CCS-6:2224 referred to above is subject to an Operating Agreement between Chevron Oil Company, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G3324, OCS-G2859, OCS-G2860, OCS-G2882, OCS-G2883, OCS-G2587, OCS-G2884, OCS-G2585, OCS-G3266, OCS-G3265, OCS-G3266, OCS-G32672, OCS-G3118, OCS-G2701, OCS-G2334, OCS-G2713, OCS-G2388, OCS-G2369, OCS-G2719, OCS-G2416, OCS-G2426, OCS-G266, OCS-G3409, OCS-G3416, OCS-G3500 and State Lease No. 6310 referred to above are each subject to an Operating Agreement between the Grantee, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2177, OCS-G2193 and OCS-G3206 referred to above are each subject to an Operating Agreement between Gulf Oil Corporation, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2403 and OCS-G3314 referred to above are each subject to an Operating Agreement between Marathon Oil Congress, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2378, OCS-G2-37, OCS-G241², OCS-G2745 and OCS-G2746 referred to above are each subject to an Operating Agreement between Cities Service Oil Company, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OC 'G2409, OCS-G2418 and OCS-G2422 referred to above are each subject to an Operating Agreement between Sun Oil Company (Delaware), as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2390 and OCS-G2718 referred to above are each subject to an Operating Agreement between Exxon Corporation, as operator, and the Grantor, et al., as non-operators.

Lease No. OCS-G2398 referred to above is subject to an Operating Agreement between Aminoil Development, Inc., as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2704, OCS-G2779, OCS-G2705 and OCS-2706 are each subject to an Operating Agreement between Texaco Inc., as operator, and the Grantor, et al., as non-operators.

The Grantor's interests in Lease N. is. OCS-G2224, OCS-G2225, OCS-G2228 and OCS-G2185 referred to above are subject to Advanc. I syment Agreements dated as of June 13, 1973, as amended, between Grantor and United Gas Pipe I. he Company.

The Grantor's interests in Lease Nos. OCS-G2883, OCS-G3417, OCS-G3195, OCS-G2947, OCS-G2414, OCS-G2978, OCS-C-2882, OCS-G2587, OCS-G2607, OCS-G2317 and OCS-G2193 are subject to liquids purchase agreements between Grantor and Atlas Processing Company.

The Crantor's interests in Lense Nos. OCS-G2839, OCS-G2224, OCS-G2225, OCS-G2177, OCS-G2185, OCS-G3193, OCS-G3194, OCS-G2947, OCS-G2193, OCS-G3202, OCS-G3203, OCS-G3206, OCS-G2366, OCS-G2372, OCS-G23118, OCS-G2378, OCS-G2701, OCS-G2704, OCS-G2779, OCS-G2705, OCS-G2706, OCS-G2734, OCS-G2713, OCS-G2388, OC3-G2389, OC3-G2390, OCS-G2718, OCS-G2719, OCS-G2398, OCS-G2384, OCS-G2719, OCS-G2414, OCS-G2716, OCS-G2418, OCS-G2418, OCS-G2739, OCS-G2426, OCS-G2745, OCS-G2746, OCS-G2416, OCS-G2417, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G2416, OCS-G2417, OCS-G2418, OCS-G2418, OCS-G2418, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G2418, OCS-G2416, OCS-G

The Grantor's interests in Lagra Nos. OCS-G3284, OCS-G2850, OCS-G2859, OCS-G2882, OCS-G2537, OCS-G2882, OCS-G3156, OCS-G2607, OCS-G287, OCS-G2883, OCS-G3500 are subject to gas purchase contracts between Grantor and Sea Robin Pipe Line Company.

The Grantor's interest in Lease No. OCS-G2701 is subject to a Partiel Assignment of Oil and Gas Lease dated as of July 1, 1977, among Grantor, Pennzoil Offshood to Operators, Inc. (now named Pogo Producing Company), et al., as assignors, and Mobil Oil Corporation, et al., as assignees.

The Grantor's interests in Lease Nos. OGS-C2366 and OCS-G2367 referred to above are subject to a Farmout Agreement dated April 27, 1978, as amended, among Kefr-McGee Corporation, et al., as owner or farmor, and Grantor, et al., as farmee. The Grantor's interest in Lease No. 2367 consists of operating rights arising from said Farmout Agreement.

The Grantor's interest in State Lease No. 6310 referred to above is subject to a gas purchase contract between Grantor and United Gas Pipe Line Company, under which sales are conditioned on the approval of or waiver by the State of Louisiana.

The Grantor's interests in Lease Nos. OCS-G2418 and OCS-G2422 are subject to liquids purchase agreements between Grantor and Sun Production Company.

The Grantor's interest in Lease No. OCS-G2429 referred to above is subject to a Unitization Agreement dated May 1, 1978, USGS Contract #14-08-0001-16930.

The Grantor interests in Lease Nos. OCS-G2224 and OCS-G2225 referred to above are each subject to a Unit Plan, a Unit Operating Agreement, and Plans of Operation among Mobil Oil Corporation, as unit operator, and Grantor, et al.

The Giantor's interest in OCS-G2409 referred to above is subject to an Assignment of Operating Rights among Grantor, et al., as assignors, and Mesa Petroleum Co., et al., as assignees, and is subject to a Gas Availability Agreement between Grantee and United Gas Pipe Line Company.

AMENDMENT

TO

INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT

THIS AMENDMENT TO INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Amendment") dated as of April 1, 1979, between Pennzon. Louisiana and Texas Official Inc., a Delaware corporation (the "Grantor"), having its principal office and post office address a Pennzoil Place, P.O. Box 2967, Houston, Texas, and Pennzoil Company, a Delaware corporation (the "Grantee"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967 Houston, Texas.

WITNESSETH:

WHEREAS, by means of that certain Instrument of Conveyance of Production Payment dated as of March 1, 1974 between the Grantor and Grantee (such Instrument of Conveyance being hereinafter called the "Conveyance"), the Grantor conveyed to the Grantee, and the Grantee purchased from the Grantor, at the time the first loan was made to the Grantor under that certain Bank Credit Agreement dated as of March 1, 1974, as thereafter amended and restated, among the Grantor, The First National Bank of Chicago, and the other banks named therein (such Bank Credit Agreement, as amended and restated as of November 1, 1977, being hereinafter called the "Bank Credit Agreement"), a production payment dischargeable from the proceeds of production from or applicable to the interests of the Grantor (hereinafte: called the "Leasehold Interests"), in those certain oil and gas leases covering lands located offshore Louisiana and Texas in the Gulf of Mexico, which interests, leases and lands are described in Exhibit "A" to the Conveyance and in Exhibit "A" to Appendix 1 to the Production Payment Agreement hereinafter referred to, in an amount equal to the aggregate principal amount of the loans, plus accrued interest and commitment fee, outstanding under the Bank Credit Agreement on the date, if any, upon which the purchase price therefor is paid by the Grantee to the Grantor as provided in that certain Production Payment Agreement dated as of March 1, 1974, as thereafter amended and restated as of November 1, 1977, between the Grantee and the Granter (such Production Payment Agreement, as so amended and restated, being hereinafter called the "Production Payment Agreement"), all as more particularly provided in the Froduction Payment Agreement;

Whereas, in order to induce the banks referred to in Section 2 of the Bank Credit Agreement (hereinafter called the "Banks") to make loans pursuant to the Bank Credit Agreement and to secure the performance by the Grantor of its obligations under the Bank Credit Agreement, the Grantor, by that certain assignment dated as of March 1, 1974 and confirmed by the Confirmation of Assignment dated as of November 1, 1977 (hereinafter called the "Assignment") from the Grantor to The First National Bank of Chicago, as agent for the Banks (hereinafter called the "Agent"), assigned, transferred and set over to the Agent for the proportionate benefit of the Banks as their interests may appear all the right, title and interest of the Grantor in, to and under the Production Payment Agreement, subject to the exceptions set forth in the Assignment, and which Assignment was also executed by the Grantee for the purpose of its making the representations and special agreements set out therein;

WHEREAS, the Bank Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of May I, 1976 to reflect an extension of the maturity of the credit available to the Company under the Bank Credit Agreement, an increase in the amount of that credit to \$200,000,000, the addition of certain banks as parties to the Bank Credit Agreement, the addition to the Leasehold Interests of the interests of the Company in certain oil and gas leases, and for other purposes;

Whereas, the Bank Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of November I, 1977 to reflect an extension of the maturity of the credit available to the Company under the Bank Credit Agreement, an increase in the amount of that credit to \$250,000,000, the addition of certain banks as parties to the Bank Credit Agreement, the

addition to and deletion from the Leasehold Interests of the interests of the Company in certain oil and gas leases, and for other purposes;

WHEREAS, the Bank Credit Agreement is now proposed to be amended to reflect a further extension of the maturity of the credit available to the Company thereunder, an increase in the amount of that credit to \$275,000,000, and for other purposes; and

WHEREAS, pursuant to the request of the Grantor and the Gra..tee, the Agent proposes to join herein for the limited purposes set forth below;

Now, Therefore, in order to add to the Leasehold Interests the interests of the Grantor in certain oil and gas leases covering lands located offshore in the Gulf of Mexico adjacent to the United States of America and to delete from the Leasehold Interests certain other leases, and in consideration of the premises and the mutual covenants and agreements herein contained and intending to be legally bound, the parties hereto agree as follows:

- The Conveyance is hereby amended in the following respects:
- (a) The first paragraph of the Conveyance is amended by changing the date "November 1, 1977" therein to "April 1, 1979".
- (b) Subparagraph (f) of Paragraph 1 of the Conveyance is amended by changing the date "November 1, 1977" therein to "April 1, 1979".
- (c) Subparagraph (a) of Paragraph 4 of the Conveyance is amended by changing the phrase "as thereafter amended and restated as of November 1, 197." therein to read "as thereafter amended, including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as of April 1, 1979".
- (d) Subparagraph (b) of Paragraph 4 of the Conveyance is amended by changing the figure "105%" therein to "106%" and by deleting the words "plus ½%" therein.
- (e) Paragraph 9 of the Conveyance is amended by changing all references to "Federal Power Commission" therein to "Federal Energy Regulatory Commission".
- (f) The signatures and acknowledgments set forth in the Conveyance are amended to read in their entirety as set forth in Annex I hereto as the signatures and acknowledgments thereto.
- (g) Exhibit "A" to the Conveyance is amended to read in its entirety as set forth in Annex I hereto as Exhibit "A" thereto.
- 2. Except as amended hereby, the provisions of the Conveyance shall remain in full force and effect. The Conveyance, as amended hereby, shall be restated to read in its entirety as set forth in Annex I hereto, it being agreed and understood that all references in the Conveyance as amended hereby and so restated to the Bank Credit Agreement shall be deemed to be references to the Bank Credit Agreement as amended, including by the amendments thereto pursuant to the Amendment to Bank Credit Agreement dated as ci April 1, 1979.
- 3. The Agent, for and on behalf and in the name of the Banks, hereby consents to the foregoing amendment to and restatement of the Conveyance; provided, however, that notwithstanding any provision hereof to the contrary, this Amendment shall not be construed or operate, except as expressly set forth above, to modify in any respect the rights and obligations of the Company or Peanzoil under the Production Payment Agreement or the Assignment.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective officers thereunto duly authorized as of the day and year first above written.

WITNESSES:

Pennzoll Louisiana and Texas Offshore, Inc.

Sue & Prixan

By

Treasister

Pennzoll Company

[Seal]

Sue & Prixan

Pennzoll Company

[Seal]

Sue & Prixan

By

Croup Vice President

The First National Bank of Chicago, as Agent

[Seal]

E.W. LaBolt

By

R. S. Glanille

THE STATE OF TEXAS
COUNTY OF HARRIS
SS:

Be It Remembered that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 9th day of April, 1979.

(Louisiana)

Before me appeared J. N. Averett, Jr., to me personally known, who, being by me duly sworn, did say that he is the Treasurer of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that J. N. Averett, Jr., acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared HAROLD E. SORTOR, to me personally known, who, being by me duly sworn, did say that he is a Group Vice President of Pennzoil Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that HAROLD E. SORTOR acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared J. N. Averett, Jr., known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be the Treasurer of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared HAROLD E. SORTOR, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of Pennzoil Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 9th day of April, 1979.

Notary Public in and for Harris County, Texas My Commission Expires

CELIA I. GARZA

Notary Public in Harris County for the State of Texas

My Commission Expires September 30, 1980

Codia J. Lana

[NOTARIAL SEAL]

THE STATE OF ILLINOIS SS

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this //// day of April, 1979:

(Louisiana)

Before me appeared R. J. GLAR VILLE , to me personally known, who, being by me duly sworn, did say that he is a VICE PRESIDENT of THE FIRST NATIONAL BANK OF CHICAGO, a national banking association, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that E. S. GLAN VILLE acknowledged the instrument to be the free act and deed of the national banking association.

(Texas)

Before me on this day personally appeared F.J. GLANVILLE, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a vice flex wear of The First National Bank of Chicago, a national banking association, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed and in the capacity therein stated, and as the act and deed of said national banking association.

IN WITNESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Chicago, Country of Cook, State of Illinois, this // th day of April, 1979.

Maria & Krupa

Notary Public in and for Cook County, Illinois
My Commission Expires

My Commission Expires Dec. 5, 1952

[NOTARIAL SEAL]

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC. TO

PENNZOIL COMPANY

CONVEYANCE OF PRODUCTION PAYMENT

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Conveyance") dated as of March 1, 1974, as amended and restated as of April 1, 1979, between Pennzoil Louisiana and Texas Offichore, Inc., a Delaware corporation (the "Grantor"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas, and Pennzoil Company, a Delaware corporation (the "Granter"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas.

WITNESSETH:

WHEREAS, the Grantor desires to sell and the Grantee desires to purchase the Production Payment Sereinaster described.

Now, Therefore, it is agreed by and between the Grantor and the Grantee as follows:

- Certain Definitions. The following terms shall be used in this Conveyance with the meanings given below:
 - (a) "Accounting Month" shall mean any monthly period commencing with and including the 21st day of any calendar month and ending on and including the 20th day of the next succeeding calendar month.
 - (b) "Effective Date" shall mean "o':lock, A.M., on the date of this Conveyance, according to the time then in effect at the 'ocation of each Subject Interest.
 - A" shall me___ Zkhibit A attached to this Conveyance and hereby made a part he
 - s" shall meen all oil, gas and other hydrocarbons or any combination of stances appropriate to the context in which such term is used.
 - nean any oil and gas lease or sub-lease, described in Exhibit A.
 - (1) Proceed commencement Date" shall mean 7 o'clock, A.M., according to the time then in effect at the location of each Subject Interest, on the date upon which the purchase price is paid by the Grantee to the Grance: as provided in the Production Payment Agreement dated as of March 1, 1974, as amended and restated as of April 1, 1979, between the Grantor and the Grantee.
 - (g) "Subject Hydrocarbons" shall mean all Hydrocarbons in and under, and which may be produced and saved from, and which shall accrue and be attributable to, the Subject Interests from and after the Proceeds Commencement Date and throughout the period specified in this Coveyance for the term of the Production Payment.
 - (h) "Subject Interests" shall mean the undivided percentage interests of the Grantor in the Leases, and, in addition thereto, each and every other kind and character of right, title, claim or interest which the Grantor now has in, to or under the Leases, which interests or Leases are

either specification or generally described in Exhibit A, all as the same shall be enlarged by the discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and all renewals and extensions of any of the same, but there shall not be included within the Subject Interests any rights which the Grantor now has or may hereafter obtain with respect to production attributable to the interests of nonconsenting parties under any of the agreement, unit operating agreement, contract for development or similar instrument.

2. Conveyance of the Production Payment. The Grantor, for valuable consideration, the receipt adequacy and sufficiency of which are hereby acknowledged, by these presents does hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, as a production payment, an undivided eighty-five percent (85%) interest in the proceeds from the sale of the Subject Hydrocarbons; subject, however, to the provisions of Paragraph 3 of this Conveyance.

The aforesaid eighty-five percent (85%) interest in the proceeds from the sale of the Subject Hydrocarbons, as granted, bargained, sold, conveyed, assigned, transferred and delivered unto the Grantee by the terms of this Conveyance, shall be the "Production Payment."

To Have and to Hold the Production Payment, together with all and singular the rights, privileges and appurtenances thereunto in any way belonging, unto the Grantee, its successors and assigns, forever.

- 3. Certain Provisions Governing Subject Hydrocarbons. All the provisions of this Conveyance shall be subject to the following:
 - (a) For the discharge of the Production Payment the Grantee shall look exclusively to the proceeds of Subject Hydrocarbons, and the Grantor shall not be liable for such discharge.
 - (b) The Production Payment shall not be dischargeable from the proceeds of any products resulting from any manufacturing, processing or refining operation, except to the extent of that portion of such proceeds which represents the fair market value at the wellhead of the Subject Hydrocarbons used in such manufacturing, processing or refining operation.
 - (c) The Production Payment shall not be dischargeable out of any bonus which the Grantor shall receive for any Lease or assignment of any of the Subject Interests or out of any payments made to the Grantor in connection with the drilling or deferring of drilling of any well on any of the Subject Interests or in connection with any adjustment of any well and leasehold equipment upon unitization of any of the Subject Interests.
 - (d) There shall not be included in the Subject Hydrocarbons any Hydrocarbons unavoidably lost in the production thereof or produced and saved from any of the Subject Interests and used by the Grantor in conformity with good field practices for drilling and production operations (including gas injection, secondary recovery, pressure maintenance, repressuring or cycling operations) conducted for the purpose of producing Subject Hydrocarbons from the Subject Interests or from any unit to which the Subject Interests are committed, but only so long as such Hydrocarbons are so used.
 - (e) So long as and to the extent that the same may be required by applicable laws and regulations, in the case of any Lease from the United States of America included in the Subject Interests, from which the average production of oil per well per day averaged on the monthly basis is 15 barrels or less, the obligation to pay and the right of the Grantee to receive the proceeds of oil produced from such Lease shall be suspended until said average production of oil per well per day exceeds said minimum amount, and such suspension shall apply separately to any zone or portion of such Lease segregated for computing government royalty.
- Amount and Term of the Production Payment. The Production Payment shall continue and remain in full force and effect until such time as the Grantee shall have received and realized.

out of the proceeds from the sale of the Subject Hydrocarbons, free and clear of all development, operating, mining, producing, treating, processing, handling, storing, marketing, transmovers and other costs and expenses of every kind whatsoever, the full aggregate sum of the tille ling amounts:

- (a) The aggregate principal amount of loans, plus accrued interest and commitment fee, outstanding on the Proceeds Communicement Date under the Bank Credi. Agreement, dated as of March 1, 1974, as ther an emended, including by the amendments hereto pursuant to the Amendment to Bank of air Agreement dated as of April 1, 1979, among the Grantor, The First National Bank of Chicago and the other banks named therein (hereinafter called the "Primary Sum"); plus
- (b) An amount computed at a rate per annum (on the basis of a 365-day year) which shall be equal to 106% of the corporate base rate on 90-day commercial loans to its largest and most credit-worthy commercial borrowers in effect at The First National Bank of Chicago from time to time, such rate to change automatically and from time to time effective as of the date of each change in such corporate base rate, from and adding the Proceeds Commencement Date on the unliquidated balance from time to time of any trima. Sum (the first such computation to be made on the 21st day of the month next following the Proceeds Commencement Date for the period from and including the Proceeds Commencement Date to and including the 20th day of the month next following the Proceeds Commencement Date on the amount of such unliquidated balance on the Proceeds Commencement Date, and subsequent computations to be made monthly on the first day of each Accounting Month for the preceding Accounting Month on the amount of such unliquidated balance on the first day of such preceding Accounting Month); plus
- (c) An amount equal to the aggregate of all amounts which are paid by the Grantee to any state or political subdivision thereof on account of ad valorem, transfer, mortgage, gross production, gross receipts, uncome, profits, severance, occupation, sales, use, franchise and other taxes and assessments of any kind whatsoever, including penalties and interest, imposed or assessed with respect to or measured by or charged against or attributable to the Production Payment, the Subject Hydrocarbons or the proceeds thereof together with an amount equal to interest on the unliquidated amount thereof at the rate of 7% per annum from the date of payment of each such amount by the Grantee, provided however, that the Granter shall have the right at its expense to contest any such taxes of assessments in good faith on behalf of the Grantee.

IT BEING THE INTENTION OF THE GRANTOR AND THE GRANTER IN ENTERING INTO THIS CONVEYANCE that the Grantee shall be entitled to receive and realize out of the Subject Hydrocarbons the full aggregate sum of the colorate described in subparagraphs (a) and (b) of this Paragraph 4, free and clear of all costs and expenses with respect to the Subject Interests and the Subject Hydrocarbons (which expenses and costs shall be home by the Granton' and over and above all taxes, expenses and costs of the character described, specified or referred to in subparagraph (c) of this Paragraph 4.

- 5. Termination of the Production Fayment. When the full aggregate sum of the amounts specified in subparagraphs (a), (b) and (c) of Faragraph, 4 of this Conveyance (as increased as expressly provided by the terms of this Conveyance has been received by the Grantee, the Production Fayment shall forthwith terminate and all interest therein shall comediately revert to and become vested in the Grantor to the same extent and with the same force and effect as if this Conveyance had not been made. The Grantee agrees that, upon such termination, it shall execute or cause to be executed upon the request and at the expense of the Grantor, such instruments as may be necessary or appropriate to expense the termination of the Production Payment.
- 6. Application of the Proceeds of Subject Hydrocarbons. For all purposes of this Conveyance, the proceeds of Subject Hydrocarbons actually received by the Grantee prior to the close of business on 20th day of the month next following the Proceeds Commencement Date shall be deemed to have been received and applied immediately after the opening of business on the 21st day of such month, and thereafter, such proceeds actually received by the Grantee after the opening of business on the first day of each Accounting Month and prior to the close of business on the last day of such Accounting

ing Month shall be deemed to have been received and applied immediately after the opening of business on the first day of the next succeeding Accounting Month; provided, however, that, if any date of application specified above shall be a Salad lay, Sunday or legal holiday under the law of the jurisdiction in which such proceeds are actually received by the Grantee, such proceeds shall be deemed to have been received and applied on the last business day next preceding such Caturday, Sunday or legal holiday in such jurisdiction, but the amount of such proceeds to be applied pursuant to a bparagraph (a) of this Paragraph 6 shall nevertheless be the amount under subparagraph (b) of Para, raph 4 of this Conveyance accrued up to (but not including) such Saturday, Sunday or legal holida. Such proceeds shall be deemed to have been applied on the first day of each Accounting Month, as follows:

- (a) First, to the amount referred to in subparagraph (b) of Paragraph 4 of this Conveyance accrued up to (but not including) the date of such application, and
- (b) Second, to the amount referred to in subparagraph (c) of said Paragraph 4, to the extent then ascertained, and
 - (c) Third, to the reduction of the unliquidated balance of the Primary Sum;

provided, however, that, in the event such proceeds so applied on the first day of any Accounting Month shall be insufficient to cover the rull amount specified in subparagraph (a) of this Paragraph 6, the unliquidated balance of the Primary Sum shall forthwith be increased (to the extent permitted by law) by an amount equal to the amount of such deficiency.

Within 10 days after the date of each such application by the Grantee of any proceeds of Subject Hydrocarbons, the Grantee will furnish or cause to be furnished to the Grantor a written statement showing the application of such proceeds in accordance with the provisions of this Paragraph 6.

- 7. Marketing the Subject Hydrocarbons. The Grantor, at its expense, shall market, or cause to be marketed, the Subject Hydrocarbons for the best price reasonably available when marketed, and it shall not be necessary for the Grantee to join in any Sales Contracts (as hereinafter defined) or any amendment thereof. In the event the Grantor shall take any of the Subject Hydrocarbons for its own use, the Grantor shall pay to the Grantee, on or before the last day of the Accounting Month during which such taking occurs, a price equal to the fair market value at the wellhead of the Subject Hydrocarbons so taken.
- 8. Sales Contracts. The Grantor will duly perform all obligations binding on it under all contracts and agreements for the sale of Subject Hydrocarbons or any portion thereof, whether presently existing or hereafter entered into (all such contracts and agreements being the "Sales Contracts"), in accordance with the terms hereof and will take all appropriate measures to enforce the performance under each of the Sales Contracts of the obligations of the purchaser thereunder. All Subject Hydrocarbons sold by the Grantor, whether pursuant to the terms of the Sales Contracts or otherwise, shall be delivered by the Grantor to the purchasers thereof into the pipelines to which the wells producing such Subject Hydrocarbons may be connected or to such other point of purchase as is reasonably required in the marketing of such Subject Hydrocarbons.
- 9. Withholding and Restitution of Proceeds of Subject Hydrocarbons. All obligations of the Grantor hereunder shall be subject to the applicable provisions of the Natural Gas Act and all applicable rules and regulations of the Federal Energy Regulatory Commission. Rates permitted under the Natural Gas Act and said rules and regulations to be paid for gas included in Subject Hydrocarbons shall be controlling if varying from prices established in Sales Contracts or if different from the fair market price at the wellhead. The Grantor shall be entitled to use its reasonable discretion in making filings with the Federal Energy Regulatory Commission affecting Subject Hydrocarbons. If any proceeds of Subject Hydrocarbons shall be withheld for any reason whatsoever, including, without limitation, withholding by the Grantor pending final approval of rate increases filed with the Federal Energy Regulatory Commission, the Grantee shall not be deemed to have received or realized such proceeds until, and only to the extent that, such proceeds shall actually have been received by the

Grantee. If, at any time before or after receipt of the full aggregate sum specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance, the Grantee shall be compelled, for any reason whatsoever, to make any payment or restitution, or such payment or restitution shall be made for the account of the Grantee, on account of any proceeds of Subject Hydrocarbons, which proceeds have already been applied pursuant to the provisions of Paragraph 6 of this Conveyance, the unliquidated balance of the Primary Sum shall forthwith be increased by the amount of the proceeds so paid over by or for the account of the Grantee plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties, provided, however, that if at the time of such payment or restitution the entire Primary Sum shall have been liquidated, the Primary Sum shall be deemed not to have been so liquidated and the unliquidated balance of the Primary Sum shall be thereupon deemed to be in the amount of the proceeds so paid over plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties.

- 10. Protection to Purchasers of Production. No person purchasing or taking or processing any Subject Hydrocarbons shall be required to take notice of, or keep informed as to, the discharge or termination of the Production Payment ur. I the actual receipt by such person of written notice advising such person of such discharge or termination.
- 11. Payment of Taxes. The Grantor will tender and pay, or cause to be paid, punctually, before they shall become delinquent (or as to any thereof which are being contested in good faith, promptly after the final determination of such contest), all ad valorem taxes (or taxes imposed in lieu thereof) and all transfer, mortgage, gross production, mining, severence, occupation, excise, sales, use, franchise, income and other taxes and assessments of every kind and characte: whatsoever (other than federal income taxes of the Grantee) imposed or assessed with respect to or measured by or charged against or attributable to the Subject Interests, the Production Fayment or the Subject Hydrocarbons, or the proceeds thereof, or against the Grantee 'by reason of its acquisition or ownership of the Production Payment or against any assignee of the Production Payment by reason of his or its interest as assignee, together with any interest or penalty payable in connection therewith.
- 12. Covenants of the Grantor. Until such time as the Production Payment has terminated as provided in Paragraph 5 of this Conveyance the Grantor will, at its expense and regardless of who may be the operator or operators of the Subject Interests, cause:
 - (a) The Subject Interests to be maintained and developed and continuously operated for the production of Hydrocarbons in a good and workmanlike manner and in accordance with sound field practices and all applicable federal, state and local laws, rules and regulations (except those being contested in good faith); provided, however, if the Grantor in good faith and in conformity with sound field practices elects not to participate in any operation which is to be conducted under the terms of an operating agreement, unit operating agreement, contract for development or similar instrument entered into by the Grantor or its predecessors in title pertaining to the development and operation of any Subject Interest (or any portion thereo?) and which allows any consenting party or parties to conduct nonconsent operations, then, during the period or periods in which the consenting party or parties are entitled to receive Subject Hydrocarbons attributable to such Subject Interest (or portion thereof) by the terms of such instrument pending the recoupment or receipt of the sums determined thereunder, the Grantee shall not be entitled to receive or be deemed to have received (but only during such period or periods) proceeds from Subject Hydrocarbons attributable to such Subject Interest (or portion thereof) to the extent affected thereby.
 - (b) All rentals, royalties, and all liabilities of any kind or nature, including, without limitation, all liabilities for labor, material, supplies and equipment incurred in, or arising from, the administration, operation or development of, or the gathering, producing, treating, processing, storing, marketing or transporting of, the Subject Hydrocarbons, to be paid punctually when due, or, as to any thereof which are being contested in good faith, promptly after the final determination of such contest.

- (c) All machinery, equipment and facilities of any kind now or hereafter located on the Subject Interests and necessary or useful in the operation thereof for the production of Hydrocarbons therefrom to be provided and to be kept in good and effective operating condition and all repairs, renewals, replacements, additions and improvements thereof or thereto needed for such purpose to be duly made or provided.
- (d) All necessary and proper steps to be taken diligently to protect and defend the Subject Interests, the Subject Hydrocarbons and the proceeds thereof against any adverse claim or demand, including, without limitation, the employment or use of counsel for the prosecution or defense of litigation and the contest, settlement, release or discharge of any such claim or demand.
- (e) Written notice to be given to the Grantee of every adverse claim or demand made by any person (including any government or governmental agency) affecting the Subject Interests, the Subject Hydrocarbons or the proceeds thereof in any manner whatsoever, or of any proceeding instituted with respect thereto.
- (f) The interest of the Grantor in each of the Subject Interests to be kept free and clear of liens, charges and encumbrances of every character, other than those which are of a character customarily found in connection with comparable drilling and producing operations and which do not materially adversely affect the operation of the Subject Interests and other than those become g effective only after termination of the Production Payment as provided in Paragraph 5 of this Conveyance.
- (g) Subject to the provisions of Faragraph 19 of this Conveyance, all Leases included in or relating to the interests described or referred to in Exhibit A and under which the Grantor is lessee, sub-lessee, or assignee to be maintained in full force and effect without amendment, modification or waiver of any provision of any such Lease which might materially adversely affect the rights of the Grantee under this Conveyance.

Anything contained in this Paragraph 12 to the contrary notwithstanding, the Grantor shall not be obligated to perform undertakings performable only by the operator and which are beyond the control of the Grantor, with respect to individual Subject Interests of which it is not the operator. In such case, however, the Grantor will use its best efforts to secure the performance of any such undertakings.

- 13. Access to the Subject Interests. The Grantor shall permit any one or more representatives designated by the Grantee to make any inspection of the Subject Interests at any reasonable time as the Grantee may reasonably request. In addition the Grantor shall furnish to the Grantee such detailed information as the Grantee may reasonably request in writing concerning the Subject Interests, the operation thereof and the production and sale of the Subject Hydrocarbons.
- 14. Pooling and Unitization. Anything herein to the contrary notwithstanding, the Grantor may, without the consent of the Grantee, pool, combine, communitize or unitize any Subject Interest (or any portion thereof) from time to time with other leasehold, mineral or other interests to form units when, in the judgment of the Grantor, it is necessary or advisable to do so in order to facilitate the orderly development or operation (including repressuring, pressure maintenance, cycling, water flood and secondary recovery operations) of any Subject Interest (or any portion thereof). Forthwith, after the formation of any unit, the Grantor shall furnish to the Grantee a true copy of the pooling agreement, unit designation or any other instrument creating such unit, together with, upon requist of the Grantee, any operating agreement relating thereto. The interest in any such unit accruing or attributable to any Subject Interest (or any portion thereof) included therein shall be subject to the Production Payment in the same manner and with the same effect as though such unit were in existence on the Effective Date. The signature or joinder of the Grantee shall not be necessary to the instruments evidencing the pooling and unitization herein authorized.

- 15. Default by the Grantor and the Remedies of the Grantee. Should the Grantor default in the performance or observance of any covenant or agreement provided in this Conveyance to be performed or observed by the Grantor, and should such default continue unremedied for 30 days (or for such longer period as may be necessary to complete performance diligately commenced within 30 days) from the delivery to the Grantor of written notice of such default, the Grantee may, in addition to pursuing any or all other remedies available at law or in equity:
 - (a) Effect performance or observance of such covenant or agreement on behalf of and at the expense of the Grantor, in which event the Grantee may expend funds reasonably necessary to effect such purposes and shall be entitled to reimbursement therefor from the Grantor, together with interest on such amounts at the rate of 7% per annum from the date of such expenditure to the date of reimbursement.
 - (b) Upon written notice to the Grantor, succeed to all rights of the Grantor vith respect to the possession, or and development of the Subject Interests, and may use in connection therewith all of the appurtenant property, both real and personal, situated upon or used or held for future use in connection therewith, or in connection with the production, storing, transpooning or sale of the Subject Hydrocarbons, and the Grantee shall have the right, on behalf of and for the account of the Grantor, to sell the Subject Hydrocarbons and the Grantor shall, upon demand, reimburse the Grantee for all amounts so expended by the Grantee, together with interest on such amounts at the rate of 7% per annum from the date of such expenditure to the date of reimbursement.
 - (c) apply to a court of equity for the specific performance or observance of such covenant or in aid of the execution of any power herein granted and for the appointment of a receiver of the Subject Interests and the Subject Hydrocarbons.
- 16. Exercise and Termination of the Remedies of the Grantee. All rights and remedies granted by the provisions of Paragraph 15 of this Conveyance and otherwise belonging to the Grantee shall be cumulative and no exercise of any right or remedy shall preclude the exercise of any other right or remedy. Any right or remedy may be exercised on one or more occasions without exhaustion thereof. All rights and remedies shall terminate:
 - (a) When the Production Payment is discharged and all amounts then due and payable to the Grantee pursuant to the terms of Paragraph 15 of this Conveyance (including amounts payable for interest) shall have been paid in full; or
 - (b) At any earlier time when all defaults of the Grantor shall have been remedied and all such amounts shall have been duly paid in full, without prejudice, however, to the exercise of any rights and remedies conferred upon the Grantee upon any subsequent failure of the Grantor to perform or observe any of the covenants or agreements herein provided to be performed or observed by the Grantor.
- 17. Reports to the Grantee. While the Production Payment remains in force and effect, the Grantor will, at its expense, furnish to the Grantee within 45 days after the end of each calendar month, a report showing for such calendar month (i) the quantity of Subject Hydrocarbons produced and sold or taken by the Grantor for its verified in the gross proceeds received from the sale, or attributable to such taking, of such Subject Hydrocarbons.
- 18. Warranty of Title. The Grantor covenants and warrants that at the time of the execution and delivery of this Conveyance, the Grantor has good and marketable title to the Subject Interests, subject only to such exceptions, charges, liens or encumbrances, if any, as are set forth or referred to in Exh bit A, that the Grantor has the power and right to create and confer upon the Grantee the

rights of the Grantee granted by the terms of this Conveyance; and that there are no suits or proceedings pending or, to the knowledge of the Grantor, threatened, against or affecting the Subject Interests before any court or by or before any governmental commission, bureau or any regulatory authority which if decided adversely to the interests of the Grantor shall materially affect the rights of the Grantee pursuant to this Conveyance. The Grantor hereby binds itself, its successors and assigns, to warrant and defend forever the rights under this Conveyance of the Grantee, its successors and assigns, against the lawful claims and demands of every person who shall claim the same or any part thereof, and the Grantor will indemnify and hold the Grantee harmless against and from all loss suffered by the Grantee resulting from any failure of or defect in the Grantor's title to the Subject Interests and the production of Hydrocarbons as warranted as aforesaid. The covenants and warranties of this Paragraph 18 shall survive the discharge of the Production Payment.

19. Transfer of the Subject Interests. Evcept for the conveyance of production payments becoming effective only after term ination of the Production Payment as provided in Paragraph 5 of this Conveyance, prior to such termination the Grantor shall not, without the written consent of the Grantee, make or permit any surrender, ab a. nment, sale, conveyance, assignment, lease or sublease, in whole or in part, of any of the Subject Interests; provided, however, that the Grantor shall have the right to surrender, abaudon or otherwise terminate (either voluntarily or involuntarily) any of the Subject interests (or any portion thereof) when there is no well located thereon which is capable of producing Hydrocarbons in paying quantities and when the Grantor has reasonably determined that (i) no existing well may be recompleted thereon as a well capable of p. oducing Hydrocarbons in paying quantities, (4) no additional well could be drilled and completed thereon which would be capable of producing Hydrocarbons, over and above the cost of drilling and completing same, in paying quantities, and (ii.) no pressure maintenance or secondary recovery operations may be instituted with respect to any well thereon which would cause such well to be capable of producing Hydrocarbons, over and above the cost of instituting and completing such pressure maintenance or secondary recovery operation, in paying quantities. For the purposes of this Paragraph 19, the words "in paying quantities" as applied at any time to the producing capability of any well shall mean that, in the light of conditions existing at the time of determination and which reasonably appear to be not temporary, such well is producing or is or will be capable of producing Subject aggregate value exceeds or will exceed the Grantor's share (as the owner of a which such well is located) of the direct cost of operating such well. the Subject Intere

In each case where the preceding paragraph of this Paragraph 19 permits the Grantor to surrender, abandon or otherwise terminate any of the Subject Interests (or any portion thereof), then (a) any of the Subject Interests (or any portion thereof) which are so surrendered, abandoned or terminated shall upon surrender, abandonment or termination cease to be a Subject Interest, and (b) the Grantor, in lieu of surrendering, abandoning or terminating any of the Subject Interests (or any portion thereof), may sell, assign, sublease, farmout or convey the same free and clear of the Production Payment without the consent of the Grantee, and the Grantee shall not have the right to any proceeds from such sale, assignment, sublease, farmout or conveyance; provided, however, that any interest in any such Subject Interest (or any portion thereof) which may be excepted from such sale, assignment, sublease, farmout or conveyance and retained by the Grantor shall not be released thereby from the Production Payment.

- 20. Exception to Payments. The acceptance by the Grantee of any payment hereunder shall not preclude the Grantee from taking exception to the correctness of the amount thereof; provided, however, that any such exception must be specified by the Grantee in a notice given to the Grantor within 12 months after the date such payment is made.
- 21. Transfer of the Production Payment. Nothing contained in this Conveyance shall be construed to limit or restrict in any way the right of the Grantee to sall, convey, assign or mortgage the Production Payment, in whole or in part.

- 22. Rights of Mortgagee or Trustee. If the Grantee shall at any time execute a mortgage or deed of trust covering all or part of the Production Payment, the mortgagee or trustee therein named or the holder of any obligation secured thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to exercise all of the rights, remedies, powers and privileges conferred upon the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained hereunder by the Grantor, but the provisions of this Paragraph 22 shall in no way be deemed or constitued to impose upon the Grantor any obligation or liability undertaken by the Grantee under such mortgage or deed of trust or under the obligation secured ther by.
- 23. Notices. Any notice, request, demand, report or other instrument which may be required or permitted to be given or furnished to or served upon either party hereto or othe, person succeeding to any interest of either party hereto shall be deemed sufficiently given or furnished and served if in writing and delivered to such party or person or to an officer of such party or person or deposited in the United States mail in a sealed envelope, registered, with postage prepaid, addressed if to the Grantor to Pennzoil Place, P.O. Box 2967, Houston, Texas, and if to the Grantee to Pennzoil Place, P.O. Box 2967, Houston, Texas, or to such other address as the party or person to be addressed shall have designated by written notice to the party or person giving such notice or furnishing such report or making such request or demand.
- 24. Further Assurances. So long as permitted by applicable law so to do, the Grantor will execute and deliver all such other and additional instruments, notices, releases, acquittances and other documents, and will do all such other acts and things, as may be necessary more fully to assure to the Grantee all the rights and interests herein and hereby granted or intended so to be.
- 25. Binding Effect All the coverages and agreements of the Grantor contained in this Conveyance shall be deemed to be coverants running with the land and shall be binding upon all successors in interest to, and all assigns of, the Grantor and shall inure to the benefit of all successors in interest to, and all assigns of, the Crantee. The provisions of this Conveyance shall inure to the benefit of and be binding upon the respective successors and assigns of the Grantor and the Grantee from and after acceptance by the Grantee. All references herein to either the Grantor or the Grantee shall include their respective successors and assigns.
- 26. Release or Failure of Title. No release by the Grantee of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect of reducing the amount of the proceeds from the sale of production of Hydrocarbons out of which the Production Payment is dischargeable or accluding the amount of the Production Payment or creating any offset or other prejudice to the Production Payment, and the Production Payment shall continue in full force and effect as to the remainder of the Subject Interest until the full amount of the Production Payment has been received by the Grantee as herein provided.
- 27. Final Termination. Notwithstanding the other provisions hereof, the Production Payment shall, in any event, terminate 1 day prior to the expiration of 21 years after the death of the survivor of all the descendants of Theodore Rocsevelt, late President of the United States, who are living on the date hereof.
- 28. Headings. Paragraph headings used in this Conveyance are for convenience only and shall not affect the construction of this Conveyance.

IN WITNESS WHEREOF, this Restated Instrument of Conveyance of Production Payment amends and restates the Instrument of Conveyance of Production Payment dated as of March 1, 1974, as heretofore amended and restated, and the parties have caused this Restated Instrument of Conveyance of Production Payment to be duly executed in the presence of the undersigned witnesses on the dates of their respective acknowledgments annexed hereto, in several counterparts, each of which is an original and all of which are identical. Each of such counterparts shall for all purposes be deemed

to be an original, and all such counter; rts together shall constitute but one and the same Instrument. The date of this Instrument shall be deemed for all purposes to be the date first above written.

| Roses Winste Sue & Prysan | By Treasurer ATTEST: Secretary |
|--------------------------------|---|
| | [CORPORATE SEAL] |
| Witnesses: | |
| Rose hernette Sue & Prejean | By Grou; Vice President ATTEST. Secretary |

THE STATE OF TEXAS
COUNTY OF HARRIS
SS

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 9th day of April, 1979:

(Louisiana)

Before me appeared J. N. Averett, Jr., to me personally known, who, being be me duly sworn, did say that he is the Treasurer of Pennzoil Louisiana and Texas Offshore, Inc. a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that J. N. Averett, Jr. acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared HAROLD E. SORTOR, to me personally known, who, being by me duty sworn, did say that he is a Group Vice President of Pennzon. Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that HAROLD E. SORTOR acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared J. N. AVERETT, JR., known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be the Treasurer of Pennzoil Louisiana and Texas Offshore, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared HAROLD E. SORTOR, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of Pennzon. Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 9th day of April, 1979.

Notary Public in and for Harris County, Texas My Commission Expires

Calia S. Barra

CELIA I. GARZA

Notary Public in Harris County for the State of Texas

My Commission Expires September 30, 1980

EXHIBIT A TO CONVEYANCE OF PROD' TON PAYMENT

| BLM or State Lease No. | Effective Date | <u>I</u> <u>e</u> | Granter's Undivided Percentage Interest |
|---------------------------|-------------------|---|--|
| | | er Continental Shelf Leasing Maps) | |
| | Out | er Conduction Shed Ceasing Maps) | |
| OCS-G2839 | 12-1-74 | All of Block 352, West Cameron Area, West Addition, OCS Official Leasing Map, Louisiana Map No. 1A | 25.00% |
| OCS-G2224 | 2-1-73 | All of Block 532 West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 26.67% |
| OCS-G2225 | 2-1-73 | All of Block 533 West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 26.67% |
| OCS-G3284 | 8-1-75 | All of Block 563, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 1B | 25.00% |
| OCS-G2850 | 12-1-74 | All of Block 609, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 6.66667% |
| OCS-G2559 | 5-1-74 | All of Block 617 West Cameron Area — South Addition, Official Leasing Map, Louisiana Map No. 1B | 6.66667% |
| OCS-G2859 | 12-1-74 | All of Block 236, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.66667% |
| OCS-G2860 | 12-1-74 | All of Block 237, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.66667% |
| OCS-G2882 | 12-1-74 | All of Block 125, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2883 | 12-1-74 | All of Block 127, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2587 | 5-1-74 | All of Block 125 South Marsh Island Area, South Addition, Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2884 | 12-1-74 | All of Block 129, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2900 | 12-1-74 | All of Block 261, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 16.66667% |
| OCS-G3156 | 7-1-75 | All of Block 262, Eugene Island Area, as shown on OCS Official Leasing Map, Louisiana Map No. 4 | 16.66667% |
| OCS-G2607 | 5-1-74 | All of Block 312 Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A | 26 66667% |
| OCS-G2317 | 2-1-73 | All of Block 333 Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A | 26.66667% |
| | | | |

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| BLM or State Lease No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
|---------------------------|-------------------|---|--|
| OCS-G2177 | 11-1-72 | All of Block 49 South Pass Area as shown on official leasing map, Louisiana Map No. 9 | 10.00% |
| State Lease No. 6310 | 5-13-74 | Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and being more particularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Louisiana | 23.00% |
| OCS-G2185 | 10-1-72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court entered December 13, 1965 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288), South Pass Area, South and East Addition, as shown on official leasing map, Louisiana Map No. 9A | 15.00% |
| OCS-G3193 | 7-1-75 | S½2 of Biock 59, Main Pass Area, that portion located in Zone 2 as that zone was defined in the agreement between the United States and the State of Louisiana, October 12, 1956 and landward of the Third Supplemental Decree Line (434 U.S. 386 (December), 1971), OCS Official Leasing Map, Louisiana Map No. 10 | 25.00% |
| OCS-G3195 | 7-1-75 | Block 72, Main Pass Area, that portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), DCS Official Leasing Map, Louisiana Map No. 10 | 21.00% |
| OCS-G2947 | 12-1-74 | All of Block 3, Main Pass Area, OCS Official Leasing Map, Louisiana Map No. 10. That portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)) | 21.00% |
| OCS-G2193 | 10-1-72 | All of Block 140 Main Pass Area as shown on official leasing map, Louisiana Map No. 10 | 10.00% |
| OCS-G3202 | 7-1-75 | All of Block 947, Ewing Bank Area, as shown on OCS Official Protraction Diagram NH-15-12 | 21.00% |
| OCS-G3203 | 7-1-75 | All of Block 903, Ewing Bank Area, as shown on OCS Official Protraction Diagram NH-15-12 | 21.00% |
| OCS-G3206 | 7-1-75 | All of Block N663E63, Moniith No. 2, as shown on OCS Official Leasing Map NH-16-10 (Also known as Block 63, Mississippi Canyon Area, as shown on OCS Official Protraction Diagram NH-16-10) | 10.00% |
| | | | |

EXHIBIT A
TO
CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| | | | Country's |
|---------------------------|-------------------|--|--|
| BLM or State Lease No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
| OCS-G2366 | 8-1-73 | All of Block A-474, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2372 | 8-1-73 | All of Block A-489, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G3118 | 4-1-75 | All of Block A-493, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 16.40% |
| OCS-G2378 | 8-1-73 | All of Block A-520, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2701 | 7-1-74 | All of Block A-540, High Island Area - South Addition, OCS Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2704 | 7-1-74 | All of Block A-545, High Island Area - South Addition, OCS fficial Leasing Map, Texas Map No. 7B | 10.00% |
| OCS-G2779 | 10-1-74 | All of Block A-546, High Island ea, South Addition, O. i Official Losing Map, Texas Map No. 77 | 14.00% |
| OCS-G2705 | 7-1-74 | All of Block A-54, High Island - South Addition, OCS Office Leasing Map "-xas M., No. 7B | 14.50% |
| OCS-G2706 | 7-1-74 | All of Block A-548, High 'slaul Area - South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2384 | 8-1-73 | All of Block A-555, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 14.50% |
| OCS-G2713 | 7-1-74 | All of Block A-562, High Island Area - South Addition, OCS Official Leasing Map, Texas | |
| OCS-G2388 | 8-1-73 | All of Block A-563, High Island Area, South Addition. Official Leasing Map, Texas Map | 12.715% |
| OCS-G2389 | 8-1-73 | All of Block A-564, High Island Area, South Addition. Official Leasing Man | 12.00% |
| OCS-G2390 | 8-1-73 | Tevas Map No. 7B All of Block A-570, High Island Area, South Addition. Official Leasing Map, | 12.00% |
| OCS-G2718 | 7-1-74 | Texas Map No. 7B All of Block A-575 High Island Area — South Addition, OCS Official Leasing Map Texas Map No. 7B | 20.00% |
| OCS-G2719 | 7-1-74 | Map, Texas Map No. 7B All of Block A-552 High Island Area — South Addition, OCS Official Leasing Map Texas Map No. 7B | 20.00% |
| | | Map, Texas Map No. 7B | 14.118% |

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| BLM or State Lease No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
|---------------------------|-------------------|---|--|
| OCS-G2398 | 8-1-73 | All of Block 4-273, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 13.50% |
| OCS-G2403 | 8-1-73 | All of Block A-279, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 17.00% |
| OCS-G3314 | 4-1-76 | All of Block A-288, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C | 17.00% |
| OCS-G2409 | 8-1-73 | All of Block A-312, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 18.11538% |
| OCS-G2737 | 7-1-74 | All of Block A-316 High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 21.132% |
| OCS-G2414 | 8-1-73 | All of Block A-323, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 14.00% |
| OCS-G2416 | 8-1-73 | All of Block A-325, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 68.57143% |
| OCS-G2418 | 8-1-73 | All of Block A-327, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G2422 | 8-1-73 | All of Block A-332, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G2739 | 7-1-74 | All of Block A-339 High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 34.00% |
| OCS-G2426 | 8-1-73 | All of Block A-340, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 34.00% |
| OCS-G2429 | 8-1-73 | All of Block A-351, High Island Area, East Addition. South Extension. Official Leasing Map, Texas Map No. 7C | 43.94625% |
| OCS-G2745 | 7-1-74 | All of Block A-355, High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 11.33% |
| OCS-G2746 | 7-1-74 | All of Block A-356, High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 15.00% |
| OCS-G2766 | 7-1-74 | All of Block A-403 High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 10.00% |
| | | | |

EXHIBIT A
TO
CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| BLM or State Lease No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
|---------------------------|-------------------|---|--|
| OCS-G3417 | 1-1-77 | That portion of Main Pass Blocks 72 and 74 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975, (U.S. vs Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971), as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 25.00% |
| OCS-G3409 | 1-1-77 | All of Block 351, Eugene Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | 25.00% |
| OCS-G3416 | 1-1-77 | That portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, Jur. 16, 1975 (U.S. vs. Louisiana, 422 U.S. 13) 11 one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971), as shown on OCS Official Leasing Map, Louisiana Map No. 9 | 20.00% |
| OCS-G3500 | 8-1-77 | All of Block 253, West Cameron Area, as shown on OCS Official Leasing Map, Louisiana Map No. 1 | 30.00% |
| OCS C2367 | 8-1-73 | S ½ of NE ¼ and SE ¼ of Block *.475, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G3756 | 6-1-78 | All of Block A-522, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 30.00% |
| OCS-G3757 | 6-1-78 | All of Block A-551. High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 16.67% |
| OCS-G3766 | 6-1-78 | All of Block 510 West Cameron Area, South Addition, OCS Official Leasing Map, Louisi- ana Map No. 1B | 16.66% |
| OCS-G3786 | 6-1-75 | All of Block 372, Eugene Island Area. South Addition, OCS Official Leasing Map, Louisi- ana Map No. 4A | 15.00% |

This Conveyance is subject to the terms and provisions of the above-described oil and gas leases and accordingly, the Subject Hydrocarbons and the proceeds therefrom shall be calculated after deducting the proportionate share of the royalties reserved in each such lease attributable to Grantor's said undivided percentage interest in such lease.

Lease Nos. OCS-G2839, OCS-G2225, OCS-G2850, OCS-G2859, OCS-G2807, OCS-G2317, OCS-G3193, OCS-G3195, OCS-G2947 and OCS-G3417 referred to above are each subject to an Operating Agreement between Mobil Oil Corporation, as operator, and the Grantor, et al., as non-operators.

Lease No. OCS-Gon24 referred to above is subject to an Operating Agreement between Chevron Oil Company, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G3284, OCS-G2859, OCS-G2860, OCS-G2882, OCS-G2883, OCS-G2587, OCS-G2584, OCS-G2885, OCS-G2900, OCS-G3156, OCS-G2185, OCS-G3202, OCS-G3203, OCS-G2366, OCS-G2372, OCS-G3118, OCS-G2701, OCS-G2384, OCS-G2713, OCS-G2388, OCS-G2389, OCS-G2719, OCS-G2416, OCS-G2426, OCS-G2429, OCS-G2766, OCS-G3409, OCS-G3416, OCS-G3500 and State Le PNo. 6310 referred to above are each subject to an Operating Agreement between the Grantee, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2177, OCS-G2193 and OCS-G3206 referred to above are each subject to an Operating Agreement between Gulf Oil Corporation, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2403 and OCS-G3314 referred to above are each subject to an Operating Agreement between Marathon Oil Company, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2378, OCS-G2737, OCS-G2414, OCS-G2745 and OCS-G2746 referred to above are each subject to an Operating Agreement between Cities Service Oil Company, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2409, OCS-G2418 and OCS-G2422 referred to above are each subject to an Operating Agreement between Sun Oil Company (Delaware), as operator, end the Grantor, et al., as non-operators.

Lease Nos. OCS-G2390 and OCS-G2718 referred to above are each subject to an Operating Agreement between Exxon Corporation, as operator, and the Grantor, et al., as non-operators.

Lease No. OCS-G2398 referred to above is subject to an Operating Agreement between Aminoil Development, Inc., as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2704, OCS-G2779, OCS-G2705 and OCS-2706 are each subject to an Operating Agreement between Texaco Inc., as operator, and the Grantor, et al., as non-operators.

The Grantor's interests in Lease Nos. OCS-G2224, OCS-G2225, OCS-G2228 and OCS-G2185 referred to above are subject to Advance Payment Agreements dated as of June 13, 1973, as amended, between Grantor and United Gas Pipe Line Company.

The Grantor's interests in Lease Nos. OCS-G2883, OCS-G3417, OCS-G3195, OCS-G2947, OCS-G2414, OCS-G2378, OCS-G-2882, OCS-G2587, OCS-G2607, OCS-G2317 and OCS-G2193 are subject to liquids purchase agreements between Grantor and Atlas Processing Company.

The Grantor's interests in Lease Nos. OCS-G2539, OCS-G2224, OCS-G2225, OCS-G2177, OCS-G2155, OCS-G3193, OCS-G3195, OCS-G2947, OCS-G2193, OCS-G3202, OCS-G3203, OCS-G3206, OCS-G2366, OCS-G2372, OCS-G3118, OCS-G2378, OCS-G2701, OCS-G2704, OCS-G2779, OCS-G2705, OCS-G2706, OCS-G2384, OCS-G2713, OCS-G2388, OCS-G2389, OCS-G2390, OCS-G2718, OCS-G2719, OCS-G2395, OCS-G2403, OCS-G2314, OCS-G2409, OCS-G2737, OCS-G2414, OCS-G2416, OCS-G2415, OCS-G2422, OCS-G2739, OCS-G2426, OCS-G2429, OCS-G2745, OCS-G2746, OCS-G2766, OCS-G3417 and OCS-G3416 are subject to gas purchase contracts between Grantor and United Gas Pipe Line Company.

The Grantor's interests in Lease Nos. OCS-G254, OCS-G2550, OCS-G2559, OCS-G2582, OCS-G2587, OCS-G2883, OCS-G2900, OCS-G3156, OCS-G2607, OCS-G2317, OCS-G2859, OCS-G2860, OCS-G2883, OCS-G3409 and OCS-G3500 are subject to gas purchase contracts between Grantor and Sea Robin Pipe Line Company.

The Grantor's interest in Lease No. OCS-G2701 is subject to a Partial Assignment of Oil and Gas Lease dated as of July 1, 1977, among Grantor, Pennzoil Offshore Gas Operators, Inc. (now named Pogo Producing Company), et al., as assignors, and Mobil Oil Corporation, et al., as assignees.

The Grantor's interests in Lease Nos. OGS-G2366 and OCS-G2367 referred to above are subject to a Farmout Agreement dated April 27, 1978, as amended, among Kerr-McGec Co. Hon, et al., as owner or farmor, and Grantor, et al., as farmee. The Grantor's interest in the No. 2367 consists of operating rights arising from said Farmout Agreement.

The Grantor's interest in State Lease No. 6310 referred to above is an a gas purchase contract between Grantor and United Gas Pipe Line Company, units what are conditioned on the approval of or waiver by the State of Louisians.

The Grantor's interests in Lease Nos. OCS-G2:. and GG2-C are subject to liquids purchase agreements between Grantor and Sun Production Company.

The Grantor's interest in Lease No OCS-G2429 referred to above is subject to a Unitization Agreement dated May 1, 1978, USGS Contract #14-05-050-16930.

The Grantor's interests in Lease Nos. OCS-G22.4 ACC-G2225 referred to above are each subject to a Unit Plan, a Unit Operating Agreement, and Plans of Operation among Mobil Oil Corporation, as unit operator, and Grantor, et al.

The Grantor's interest in OCS-G2409 referred to above is subject to an Assignment of Operating Rights among Grantor, et al., as assignors, and Mesa Petroleum Co., et al., as assignees, and is subject to a Gas Availability Agreement between Grantee and United Gas Pipe Line Company.

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC

CONFIRMATION OF ASSIGNMENT

THIS CONFIRMATION OF ASSIGNMENT dated as of April 1, 1979 (the "Confirmation"), between Pennzoil Louisiana and Texas Offshore, Inc. (the "Company"), and The First National Bank of Chicago, as Agent for the Panks mentioned below (the "Agent"),

WITNESSETH THAT:

WHEREAS, the Company entered into a Bank Credit Agreement (the "Credit Agreement"), dated as of March 1, 1974, with The First National Bank of Chicago, the other Banks referred to in Section 2 of the Credit Agreement and the Agent providing for loans (the "Loans") to the Company by the Banks then parties to the Credit Agreement in an aggregate principal amount not in excess of \$100,000,000, to be evidenced by the promissory notes of the Company (the "Notes"):

WHEREAS, the Company executed, acknowledged and delivered to Pennzoil Company ("Pennzoil") a conveyance of production payment dated as of March 1, 1974, substantially in the form of Appendix 1 (the "Conveyance") to that certain Production Payment Agreement, dated as of March 1, 1974, between Pennzoil and the Company (the "Production Payment Agreement"), all as provided in Section 2.3 of the Production Payment Agreement;

WHEREAS, in order to secure the full and complete performance by the Company of its obligations under the Credit Agreement, the Con pany agreed therein to assign its rights and options in, to and under the Production Payment Agreement to cause Pennzoil either (a) to pay the Company, as the purchase price for the production payment conveyed by the Company to Pennzoil as aforesaid, an amount equal to the principal amount of such production payment, such principal amount to equal the aggregate principal amount of the loans, plus accrued interest thereon and commitment fee applicable thereto, if any, outstanding under the Credit Agreement on the date, if any, upon which such payment is made, or (b) to execute, acknowledge and deliver to the Company, at its expense, such instrument or instruments as may be necessary or appropriate to release, relinquish and extinguish as fully discharged, such production payment;

Whereas, in order to induce the Banks then parties to the Credit Agreement to make loans pursuant to the Credit Agreement and to secure the performance by the Company of its obligations under the Credit Agreement, including its obligation to pay the Notes (the "Credit Agreement Obligations"), the Company executed, acknowledged and delivered to the Agent an Assignment dated as of March 1, 1974 (the "Assignment"), a conformed copy of which is attached hereto as Appendix I (in which Pennzoil joined for the limited purposes set forth therein), by which the Company assigned, transferred and set over to the Agent for the proportionate benefit of the Banks then parties to the Credit Agreement as their interests may appear all the right, title and interest of the Company in, to and under the Production Payment Agreement, save and except, however, those rights of the Company relating to the sale of gas produced from the Leasehold Interests provided for in Article III of the Production Payment Agreement (such right, title and interest of the Company in the Production Payment Agreement, save and except such rights relating to the sale of gas, being herein and in the Assignment referred to as the "Production Payment Rights"), together with the right, title and interest of the Company in, to and under any amendment of the Production Payment Rights or additions thereto;

Whereas, the Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of May 1, 1976 to reflect an extension of the maturity of the credit available to the Company under the Credit Agreement, an increase in the amount of that credit to \$200,000,000 and the addition of Manufacturers Hanover Trust Company, Security Pacific National Bank, Wells Fargo Bank National Association, The Northern Trust Company, European-American Bank & Trust Company and The First National Bank of Boston as parties to the Credit Agreement, and for other purposes;

WHEREAS, in connection with the aforesaid amendment and restatement of the Credit Agreement, the Production Payment Agreement and the Conveyance, the Assignment was confirmed by the Confirmation of Assignment dated as of May 1, 1976 between the Company and the Agent (in which Pennzoil joined for the limited purposes set forth therein) in order to confirm the continuing validity reunzou joined for the limited purposes set form therein) in order to confirm the continuing validity and effectiveness of the Assignment and to provide that all references in the Assignment as confirmed thereby to the Credit Agreement, the Production Payment Agreement or the Conveyance would be deemed to refer to such instruments as amended and restated as of May 1, 1978 and that all references deemed to refer to such instruments as amended and restated as or May 1, 1910 and that air references to the Banks would be deemed to refer to the Banks then parties to the Credit Agreement as so

WHEREAS, the Credit Agreement, the Production Payment Agreement and the Convernce WHEREAS, the Credit Agreement, the Production Payment Agreement and the Convernace were further amended and restated as of November 1, 1977 to reflect an extension of the maturity were rurner amended and restated as or November 1, 1971 to renect an extension of the imacunty of the credit available to the Company under the Credit Agreement, an increase in the amount of or the credit available to the Company under the Credit Agreement, an increase in the amount of that credit to \$250,000,000 and the addition of Morgan Guaranty Trust Company of New York and amended and restated; that credit to \$250,000,000 and the addition of Morgan Guaranty Trust Company of New Tork and First International Bank in Houston, N.A. as parties to the Credit Agreement, and for other purposes. WHEREAS, in connection with the amendment and restatement of the Credit Agreement, the

Production Payment Agreement and the Conveyance as of November 1, 1977, the Assignment was confirmed by the Confirmation of Assignment dated as of November 1, 1977, the Assignment was commed by the Communition of Assignment dated as of November 1, 1911, between the Company and the Agent (in which Pennzoil joined for the limited purposes set forth therein) in order to and the Agent (in which rennzon joined for the limited purposes set form therein) in order to confirm the continuing validity and effectiveness of the Assignment and to provide that all references confirm the continuing validity and effectiveness of the Assignment and to provide that all references in the Assignment as confirmed the eby to the Credit Agreement, the Production Payment Agreement in the Assignment as confirmed the epy to the Credit Agreement, the Production Payment Agreement or the Conveyance would be deemed to refer to such instruments as amended and restated as of or the Conveyance would be deemed to refer to such inscriments as amended and restated as of November 1, 1977 and that all references to the Banks would be deemed to refer to the Panks then

WHEREAS, the Credit Agreement, the Production Payment Agreement and the Conveyance WHEREAS, the Uredit Agreement, the Production Payment Agreement and the Conveyance are being further amended and restated as of April 1, 1979 to reflect an extension of the maturity of parties to the Credit Agreement as so amended and restated; are being further amended and restated as of April 1, 1919 to renect an extension of the maturity of the credit available to the Company under the Credit Agreement, an increase in the amount of that the credit available to the Company under the Credit Agreement, an increase in the amount of three credit to \$275,000,000 and the deletion as parties to the Credit Agreement of three of the Banks there.

WHEREAS, pursuant to the request of the Company and the Banks which are parties to the Credit WHEREAS, pursuant to the request of the Company and the Banks which are parties to the Credit Agreement as amended and restated as of April 1, 1979 as aforesaid (hereinafter called the "Banks"). tofore parties "hereto, and for other purposes, and

Now, Therefore, the Company and P. nzoil hereby again confirm the continuing validity and Pennzoil proposes to join herein for the limited purposes set forth below:

effectiveness of the Assignment, which has been heretofore made by the Company to enforce. enecuveness or the Assignment, which has been neretorore made by the Company to ensuree, secure and provide additional means of payment of the Credit Agreement Obligations. It is secure and provide stood that, effective on the Amendment Date as defined in the Amendment turiner agreed and up stood that, effective on the Amendment Date as denned in the Amendment to Bank Credit Agreement dated as of April 1, 1979 whereby the Credit Agreement is amended and to bank Crean Agreement dated as of April 1, 19,9 whereby the Crean Agreement is amended and restated as of April 1, 1979 as aforesaid, all references in the Assignment as confirmed hereby and in restated as of April 1, 1919 as aroresaid, an references in the Assignment as committee nereby and the the following paragraph to the Credit Agreement, the Production Payment Agreement and the Conveyance shall be deemed to refer to such instruments as amended and restated as if April 1, 1979, and that all references in the Assignment as confirmed hereby and in the following paragraph to the Banks shall be deemed to refer to the Banks which are parties to the Credit Agree. graph to the Banks shall be deemed to refer to the Banks which are parties to the Green Agreement of which is ment as so amended and restated. The Credit Agreement Obligations the payment of which is secured by the Assignment as confirmed hereby shall include the obligations of the Company under

The Assignment, as confirmed hereby, (i) shall remain in full force and effect until payment in the Credit Agreement as amended and restated as of April 1, 1979. full of all indebtedness and obligations of the Company to each of the Banks under and pursuant to the Credit Agreement and the Notes executed and which may be executed pursuant thereto, at which time the Banks will, upon request and at the expense of the Company, related and reassign to the Time the panks will, upon request and at the expense of the Company release and reassign to the Company and Pennzoil, as their interests may appear, the rights assigned to the Banks hereunder as soon as there has been sufficient elapse of time that, in the opinion of the Banks, the continuation of the security interest under the Assignment is no longer necessary to protect the Banks from loss or expense by reason of any claim that the payment might constitute a preference or otherwise be subject to being voided, (ii) shall be binding upon the Company and Pennzoil, their respective successors and assigns, (iii) shall inure to the benefit of and be enforceable by the Agent and the Banks, their respective successors, transferees and assigns, and (iv) shall be deemed to have been made under and shall be governed by the laws of the State of Texas.

IN WITNESS WHEREOF, the parties hereto have caused this Confirmation to be executed by their respective officers the eunto duly authorized as of the day first above written.

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

| ۰ | - | | | | | |
|---|----|-----|---|----|---|--|
| | 62 | 200 | - | | | |
| L | o | 45 | a | a. | ٠ | |
| | | | | | | |

y Title: Treasure

PENNZOIL COMPANY

[SEAL]

Title: Group Vice President

THE FIRST NATION L BANK OF CHICAGO, as Agent

[SEAL]

By R.S. Glanville Title: Vice Rusident

THE STATE OF TEXAS COUNTY OF HARRIS SS:

BE IT REMEMBERED that I, the undersigned, a Notary Public Carry qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 6th day of April, 1979.

(Lcuisiana)

Before me appeared J. N. Averett, Jr., to me personally known, who, being by me duly sworn, did say that he is the Treasurer of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that J. N. Averett, Jr. acknowledged the instrument to be the free and deed of the corporation.

(Louislana)

Before me appeared Harold E. Sortor, to me personally known, who, being by me duly sworn, did say that he is Group Vice President of Pennzon. Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that Harold E. Sortor acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this clay personally appeared J. N. Averett, Jr., known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be the Treasurer of Pennzoll Louislana and Texas Offshore, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared Harold E. Sortor, known to me to be the person whose name is subscribed to the egoing instrument and known to me to be a Group Vice President of Pennzon Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 9th day of April, 1979.

Notary Public in and for Harris County, Texas My Commission Expires

> CELIA I. GARZA Notary Public in Harris County for the State of Texas My Commission Expires September 30, 1980

Colie J. Danga

[NOTARIAL SEAL]

THE STATE OF ILLINOIS SS:

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this W day of April, 1979:

(Louisiana)

Defore me appeared S. S. GLANVILLE , to me personally known, who, being by me duly sworn, did say that he is a Vice President of The FIRST NATIONAL BANK OF Ch.CACO, a national banking association, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that R.S. GLANVILLE acknowledged the instrument to be the free act and deed of the national banking association.

(Texas)

Before me on this day personally appeared R.S. GLANJILLE, known to me to be the person whose name is subscribed to the foreging instrument and known to me to be a Vice Resident of The First National Bank of Chicaco, a national banking association, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed and in the capacity therein stated, and as the act and deed of said national banking association.

IN WITNESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Chicago, County of Cook, State of Illinois, this U day of April, 1979.

Notary Public in and for Cook County, Illinuis
My Commission Expires

Maria Krupa

My Commission Expires Dec. 5, 1982

[NOTARIAL SEAL]

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

ASSIGNMENT

THIS ASSIGN 'ENT dated as of March 1, 1974 (the "Assignment"), from Pennzoil Louisiana and Texas Offshore, (the "Company") to The First National Bank of Chicago, as Agent for the Banks mentioned below (the "Agent"),

WITNESSETH THAT:

WHEREAS, the Company has entered into a Bank Credit Agreement (the "Credit Agreement"), dated as of March 1, 1974, with The First National Bank of Chicago, the other Banks referred to in Section 2 of the Credit Agreement (the "Banks") and the Agent providing for loans (the "Loans") by the Banks to the Company in an aggregate principal amount not in excess of \$100,000,000, to be evidenced by the promissory notes of the Company (the "Notes");

WHEREAS, the Company has executed, acknowledged and delivered to Pennzoil Company ("Pennzoil") a conveyance of production payment, dated as of March 1, 1974, substantially in the form of Appendix 1 to that certain Production Payment Agreement, dated as of March 1, 1974, between Pennzoil and the Company (the "Production Payment Agreement"), all as provided in Section 2.3 of the Production Payment Agreement;

Whereas, in order to secure the full and complete performance by the Company of its obligations under the Credit Agreement, the Company has regreed therein to assign its rights and options in, to and under the Production Payment Agreement to cause Pennzoil either (a) to pay to the Company, as the purchase price for the production payment conveyed by the Company to Pennzoil as aforesaid, an amount equal to the principal amount of such production payment, which principal amount shall equal the aggregate principal amount of the loans, plus accrued interest thereon and commitment fee applicable thereto, if any, outstanding under the Credit Agreement on the date, if any, upon which such payment is made, or (b) to execute, acknowledge and deliver to the Company, at its expense, such instrument or instruments as may be necessary or appropriate to release, relinquish and extinguish as fully discharged, such production payment; and

WHEREAS, pursuant to the request of the Company and the Banks, Pennzoil proposes to join herein for the limited purposes set forth 'selow;

Now, THEREFORE, in order to induce the Banks to make loans pursuant to the Credit Agreement and to secure the performance by the Company of its obligations under the Credit Agreement, including its obligation to pay the Notes (the "Credit Agreement Obligations"), the Company hereby assigns, transfers and sets over to the Agent for the proportionate benefit of the Banks as their interests may appear all of the right, title and interest of the Company in, to and under the Production Payment Agreement, save and except, however, those rights of the Company relating to the sale of gas produced from the Leasehold Interests provided for in Article III of the Production Payment Agreement (such right, title and interest of the Company in the Production Payment Agreement, save and except such rights relating to the sale of gas, being hereinafter referred to as the "Production Payment Rights"), together with the right, title and interest of the Company in, to and under any amendment of the Production Payment Rights or additions thereto.

- 1. This Assignment is made by the Company to enforce, secure and provide an additional true-ins of payment of the Credit Agreement Obligations.
- 2. The Production Payment Pights may be exercised by the Agent for and on behalf of the Banks upon the occurrence of an Event of Default as described in Section 11 of the Credit Agreement and at any time during the continuance of such Event of Default. To exercise the right of the

Company to call upon Peruzoil to pay the purchase price of such production payment as provided in Section 2.4 of the Production Payment Agreement (the "Purchase Price Right"), the Agent shall give Pennzoil in ice to such effect which shall

- (a) specify the purchase price of the production payment (determined as provided in Section 2.1 of the Production Payment Agreement) to be paid by Pentroil pursuant to such request, and
- (b) specify the date upon which the purchase price of such production payment is to be paid.

Concurrently with the giving of such notice to Pennzoil, the Agent shall deliver or mail an executed counterpart of such notice to the Company. Giving of such notice by the Agent to Pennzoil shall constitute an effective notice of election by the Company to exercise the Purchase Price Right as provided for in Section 2.5 of the Production Payment Agreement. The full amount of the purchase price of such production payment shall be paid by Pennzoil, without setoff, counterclaim or recoupment whatsoever by reason of any claim of Pennzoil against the Company or otherwise, directly to the Agent on the date specified in such notice and shall be applied by the Agent to the payment of the Notes in the manner described in Section 6 of the Credit Agreement.

- 3. Notwithstanding this Assignment, the Company and Pennzoil shall have the right, without the consent of the Agent or the Banks, to amend the Production Payment Agreement and the conveyance of production payment executed pursuant thereto for the sole and limited purposes of releasing, or adding to, one or more of the Leasehold Interests or subjecting the same to gas sales, unit, operating or other agreements providing for the operation or development of the Leasehold Interests or the disposition of production therefrom; provided, however, that any such amendment shall not otherwise modify in any respect the rights and obligations of the Company or Pennzoil under the Production Payment Agreement or reduce the purchase price of said production payment determined as provided in Section 2.1 of the Production Payment Agreement.
- 4. No failure or delay on the part of the Agent or the Banks in exercising any power or right hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power preclude any other single or partial exercise thereof or the exercise of any other right or power hereunder.
- At the request of the Company and in order to induce the Banks to make the Loans to the Company, Pennzoil agrees
 - (a) with the Company that each certificate as to value of Proven Hydrocarbon Reserves delivered by the Company to the Agent pursuant to Section set the Credit Agreement shall be signed by Pennzoil to acknowledge its approval thereof and to be deemed to have been delivered to Pennzoil pursuant to Section 2.4 of the Production Payment Agreement, and
 - (b) with the Agent and the Banks that the breach or fadure of the Company to perform any of its obligations under this Assignment or the Production Payment Agreement shall not reduce, impair or adversely affect in any respect the enforceability by the Agent and the Banks of (i) the Production Payment Rights assigned to them hereunder and (ii) the agreement of Pennzoil to pay directly to the Agent upon the date specified in the notice given to Pennzoil by the Agent pursuant to Paragraph 2 above an amount could to the purchase price of the production payment specified in such notice after any required adjustment as provided in Section 2.5 of the Production Payment Agreement,

it being the purpose and intent of Pennzoil herein to confirm to the Agent and the Banks the continued effectiveness, validity and enforceability of the Production Payment Rights assigned to them hereunder throughout the term of this Assignment without regard to the performance or breach by the Company of its obligations under this Assignment or the Production Payment Agreement.

6. Pennzoil hereby represents and warrants that

- (a) It is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. It is duly qualified to do business in those states in which it owns a material amount of property or transacts a material amount of business.
- (b) It has corporate power and authority to make and carry out the Production Payment Agreement and this Assignment, and all such action has been duly authorized by all necessary corporate proceedings on its part.
- (c) Since December 31, 1973 there has been no material adverse change in the financial condition of Pennzoil or of Pennzoil and it; consolidated subsidiaries.
- (d) The Production Payment Agreement, the conveyance of production payment executed pursuant thereto and this Assignment have been duly and validly executed and delivered by Pennzoil and constitute valid and legally binding agreements of Pennzoil enforceable in accordance with their terms, except as enforcement thereof is limited by bankruptcy, insolvency or other laws of general application relating to or affecting the enforcement of creditors' rights.
- (e) While this Assignment continues unreleased Pennzoil shall, if notified of the exercise by the Agent of the right to have the purchase price paid to the Agent, be unconditionally obligated to pay such purchase price directly to the Agent in accordance with the terms hereof.
- 7. All notices and other communications provided for herein shall be in writing and shall be delivered or mailed (or in the case of telegraphic communication, if by telegram, delivered to the telegraph company and if by the telex, graphic scanning or other telegraphic communications equipment of the sending party, delivered by such equipment) addressed, if to the Company or Pennzoil, as appropriate, at 900 Southwest Tower, Houston, Texas 77002, and if to the Agent, to it at One First National Plaza, Chicago, Illinois 60670. All notices and other communications given to any party hereto in accordance with the provisions of this Assignment shall be deemed to have been given when sent by registered or certified mail, if by mail, or when delivered to the telegraph company, charges prepaid, if by telegram, or when receipt is acknowledged, it by any telegraphic communications equipment of the sender in each case addressed to sparty as provided in this Paragraph 7 or in accordance with the latest unrevoked direction from _ach party.
- 8. This Agreement (i) shall remain in full force and effect until payment in full of all indebtedness and obligations of the Company to each of the Banks under and pursuant to the Credit Agreement and the Notes, at which time the Banks will, upon request and at the expense of the Company, release and leassign to the Company and Pennzoil, as their interests may appear, the rights assigned to the Banks hereunder as soon as there has been sufficient elapse of time that, in the opinion of the Banks, the continuation of the security interest under the Assignment is no longer necessary to protect the Banks from loss or expense by reason of any claim that the payment might constitute a preference or otherwise be subject to being voided, (ii) shall be binding upon the Company and Pennzoil, their respective successors and assigns, (iii) shall inure to the bound and be enforceable by the Agent and the Banks, their respective successors, transferees and and (iv) shall be construed in accordance with the laws of the State of Texas.

In Witness Whereor, the parties hereto have caused this Assignment to be executed by their respective officers thereunto duly authorized, and in the presence of the undersigned witnesses, as of the day first above written.

Witt sses: PENNZOIL LGUISIANA AND TEXAS OFFSHORE, INC. R. C. BRYAN By s/ MARY FRANCES FORESTER Vice President GEORGIA SPIVEY ATTEST: (Louis'ana) SALLY HAZEN 5/ Assistant Secretary (Texas) [CORPORATE SEAL] Witnesses: PENNZOIL COMPANY MARY FRANCES FORESTER (Texas) W. E. GIPSON By s/ Vice President GEORGIA SPIVEY ATTEST: SALLY HAZEN 5/ Assistant Secretary

(Louisiana)

[CORPORATE SEAL]

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 21st day of March, 1974:

Before me appeared R. C. BRYAN, to me personally known, who, being by me di / sworr, did say that he is Vice President of Pennzoil Louisiana and Texas Offshore, Inc., a Delawise corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that R. G. Bryan acknowledged the instrument to be the free act and deed of the

Before me appeared W. E. Gipson, to me personally known, who, being by me duly swom, did say that he is Vice President of Pennzou Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said cooration and that the instrument was signed and sealed in behalf of the corporation by authority ts Board of Directors and that W. E. Gipson acknowledged the instrument to be the free act and deed of the corporation.

Before me on this day personally appeared R. C. BRYAN, known to me to be the person whose name 's subscribed to the foregoing instrument and known to me to be a Vice President of Pennzon. LOUISIANA AND TEXAS OFFSHORE, INC., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

Before me on this day personally appeared W. E. Gipson, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Pennzon. Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WINESS WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 21st day of March, 374.

3/

DOROTHY STEARMAN Dorothy Stearman

Notary Public in and for Harris County, Texas My Commission Expires June 1, 1975

LISKOW & LEWIS

ATTORNEYS AT LAW A PAR NERSHIP OF LAW CORPORAT CHS

NEW ORLEANS, LA 70139

ONE SHELL SQUARE FIFT. ETH F. 308 TELEPHONE :504: 581-7979 TAL BIO-951-5252

LAFAYETTE, LA 70505

SEL TRAVIS ST P 0 8C # 12008 0 C 8 TELEPHONE (3:8) 232-7424

New Orleans, 70139 July 20, 1981

Mr. John L. Rankin, M_nager New Orleans Outer Continental Shelf Office Bureau of Land Management 500 Camp Street, Suite 841 New Orleans, Louisiana 70130

THOMAS F DETTEN THOMAS F GETTEN
GEORGE HOTENISON.
GEORGE J DOMAS
MARLYN C MALONEY
POSERM C GIGLIO, JR
JARRY M ROFEEL
LAW CORPORATIONS

BRUCE J ORECK PATRICK W GRAY DEBORAH TAHN PR POBERT E HOLDEN JOE B NORMAN
THOMAS M MUNAMARA
JAMES M MANBFIELD III
BILLT J DOMINGUE
LAMBERT M LAPEROUSE
SUBAN S LASITER
PHILIP R JONES, JR OE B NORMAN ANNE E TATE PAUL & DAVID LYNN C HANTEL JULIE E SCHWARTS



Instrument of Conveyance of Production Payment Re: Pennzoil Company, Pennzoil Producing Company, Peni oil Oil & Gas, Inc. and Duval Corporation to New Dunstan, Inc.

Dear Mr. Rankin:

ULLEN R LIBHOW 1893-1970

VICTOR.

WILLIAM M MEYERS
COBERT T JORDEN
THARLES C GREMILION
DENE W LAFITTE
BILLT H HINES
HAMES L PELLETIER
THOMAS D HARDEMAN
LINTON MORGAN

STEPHEN T VICTOREDWARD J DAY III
ENNETHE CORDON, JR
ENNETHE CORDON, JR
ECON J REFMOND, JR
DONALD R ABAJINZA
CONN M WILSO!
CHARLES M STEEN
LAWRENCE P SIMON JR
FREDERICH W BRADLEY
REFREY M MASSAM
5 GENE FENDLER

OHN M KING

By instrument of Conveyance of Production Payment (the "Conveyance") dated July 16, 1981 executed by Pennzoil Company, Pennzoil Producing Company, Pennzoil Oil & Gas, Inc. and Duval Corporation (the "Grantors") in favor of New Dunstan Inc. (the "Grantee"), the Grantors conveyed to Grantee a production payment relating to the properties and interests described in Part One and Part Two of Exhibit A attached to the Conveyance, including the interests of Grantors in and to the federal OCS oil and gas leases which are described in Part One of Exhibit A to the Conveyance ("Part One of Exhibit A"). A copy of Part One of Exhibit A is attached to this letter as a part thereof.

Enc. sed is a Jully executed counterpart of the Conveyance with a complete copy of Part One and Part Two of Exhibit A attached thereto. In order to place third parties on notice as to the execution and efficacy of the Conveyance, please file this particular counterpart of the Conveyance, with a copy of this letter attached thereto, in the file maintained by your office with respect to Lease OCS-G 2115.

Alos enclosed are sixty-six (66) fully executed counterparts of the Conveyance. Insofar as Exhibit A is concerned, these counterparts have attached thereto only Part One of Exhibit A. In order that third parties will be placed on notice as to the execution and efficacy of the Conveyance, please file one of these counterparts of the Conveyance, with a copy of this letter attached thereto, in the file maintained by your office with respect to each of the Federal OCS leases listed in Part One of Exhibit A (other than Lease OCS-G 2115).

Enclosed is our check in the amount of \$1,675.00 in payment of filing fees.

It would be appreciated if you would acknowledge that filing as above requested has been accomplished by signing all copies of this letter in the space provided below.

Yours very truly,

LISKOW & LEWIS

. Jullan L. Joseph

WMM: eeb

FILED AS REQUESTED

BY: (Orig. Sgd.) Ruby J. Boehm

John L. Rankin, Manager New Orleans Outer Continental Shell Office Bureau of Land Management

Date: July 20, 1981

PART ONE

OFFSHORE SUBJECT INTERESTS

TEXAS

| BLM or State Lease No. | Effective Date | Description | Net Reve | nue Interest PPC |
|---------------------------|-------------------|---|----------|---------------------|
| OCS-G-2398 | 8- 1-73 | All of Block A-273, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 70 | 1.12639 | 11.24999 |
| OCS-G-2403 | 8- 1-73 | All of Block A-279, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.60124 | 14.16666 |
| OCS-G-2414 | 8- 1-73 | All of Block A-323, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 11.66662 |
| OCS-G-2416 | 8- 1-73 | All of Block A-325, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 7.22016 | 57.14286 |
| OCS-G-2419 | 9- 1-73 | All of Block A-327, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 8.3333 3 |
| OCS-G-2422 | 8- 1-73 | All of Block A-332, High Island Area, East Addition, South Extension, OCS Official Lessing Map, Texas Map No. 7C | 1.00124 | 8.33333 |
| OCS-G-2739 | 7- 1-74 | All of Block A-339, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 28.33332 |
| OCS-G-2426 | 9- 1-73 | All of Block A-340, High Island Area, East Addition, South Extension, OCS Official Leasing Map. Texas Map No. 7C | 1.00124 | 28.33332 |
| OCS-G-2429 | S- 1-73 | All of Block A-351, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 70 | 2.88528 | 36.62186 |
| OCS-G-2745 | 7- 1-74 | All of Block A-355, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 7.34246 | 9.44166 |

| BLM or State Lease No. | Effective Date | | | ue Interest |
|---------------------------|-------------------|--|----------|-------------|
| OCS-G-2746 | 7- 1-74 | All of Block A-356, High Island Area, East Addition, South Extension, OCS Official Leasing Map. Texas | 10.01245 | 12.49999 |
| OCS-G-2366 | 8- 1-73 | Map No. 7C All of Block A-474, Figh Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 9.99999 |
| OCS-G-2367 | 8- 1-73 | S 1/2 of NE 1/4 & SE 1/4 of Block A-475, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.001241 | 9,999996 |
| OCS-G-2372 | 8- 1-73 | All of Block A-489, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.49999 |
| OCS-G-3118 | 4- 1-75 | All of Block A-499, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.10137 | 13.66667 |
| OCS-G-2378 | 8- 1 73 | All of Block A-520, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 9.99999 |
| OCS-G-2704 | 7- 1-74 | All of Block A-545, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 6.67497 | 8.33333 |
| OCS-G-2779 | 10- 1-74 | All of Block A-546, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | S.67746 | 11.66666 |
| OCS-G-2705 | 7- 1-74 | All of Block A-547, High Island Area, South Addition, OCS Official Lensing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2706 | 7- 1-74 | All of Block A-548, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2384 | 8- 1-73 | All of Block A-555, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.08332 |
| OCS-G-2388 | S- 1-73 | All of Block A-563, High Island Area, South Addition, OCS Official Leasing Map, Texas Map, No. 7B | 1.00124 | 9.99999 |
| OCS-G-2389 | 8- 1-73 | All of Block A-564, High Island Area, South Addition, OCS Official Leasing Map, No. 7B | 1.00124 | 9.99999 |

| BLM or State | Effective Date | Description | Net Reven | ue Interest PPC |
|--------------|-------------------|---|-----------|--------------------|
| OCS-G-2390 | 8- 1-73 | All of Block A-570, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.01958 | 16.97190 |
| OCS-G-2719 | 7- 1-74 | All of Block A-582, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.17780 | 11.76499 |
| | | Louisiana | | |
| St. Ls. 7604 | 4-17-78 | TRACT 14836 — PORTION OF BLOCK 3 SABINF FASS AREA, Cameron Parish. Lasiana. That portion of Block 3, Salline Pass Area, Cameron Parish. Louisiana, belonging to the State of Louisiana and not under a mineral lease on January 23, 1978, described as follows: Beginning at a point on the North line of Block 3, Sabine Pass Area having Lambert Plane Coordinates of X = 1,217,855.14 and Y = 364,274.14; thence East 9,612.44 feet along said North line to a point having Lambert Plane Coordinates of X = 1,227,467.58 and Y = 364,274.14; thence Southerly and Southwesterly along an arc having a radius of 18,240.60 feet and a vertex of X = 1,209,227 and Y = 364,245 to a point on the South line of said Block 3, having Lambert Plane Coordinates of X = 1,219,987.06 and Y = 349,517.10; thence West 2,131.92 feet along said South line to a point having Lambert Plane Coordinates of X = 1,217,355.14 and Y = 349,516.10; thence North 14,758.04 feet to the point of beginning, estimated to contain approximately 2,497.59 acres. as shown outlined in red on a plat on file in the Office of the Secretary, Department of Natural Resources, LESS AND EXCEPT that portion thereof, if any, which is more than three nautical miles from the coast line as determined by the Report of the Special Master in the litigation in the Supreme Court of the United States styled United States v. State of Louisiana et al. No. 9 Original, | | 38.125 |

| BLM or State Lease No. | Effective Date | Description | PCGI | PPC |
|---------------------------|-------------------|---|----------|----------|
| | | said three mile line as set out in the June, 1975, decree of the Supreme Court. All bearings are based on Louisiana Lambert Plane Coordinate System (South Zone). | | |
| OCS-G-4144 | 9- 1-79 | That portion of Block 3 which is more than three geographical miles seaward from the line described in the supple- mental decree of the U.S. Supreme Court, June 16, 1975 (United States vs. Louisiana, 422 U.S. 13) Sabine Pass Area, as shown on OCS Official Leasing Map. Louisiana Map No. 12. | | 41.66663 |
| OCS-G-4379 | 11- 1-80 | All of Block 53, West Cameron Area, as shown on OCS Leasing Map, Louisiana Map No. 1 | | 41.66665 |
| OCS-G-2839 | 12- 1-74 | All of Block 352, West Cameron Area, West Addition, OCS Octial Leasing Map, Louisiana Map No. 1A | 8.34372 | 20.833 |
| OCS-G-2224 | 2- 1-73 | All of Block 532, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.224 |
| OCS-G-2225 | 2- 1-73 | All of Block 533, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.22499 |
| OCS-G-3284 | 8- 1-75 | All of Block 563, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 1B | 25.03115 | 20.83332 |
| OCS-G-2436 | 8- 1-73 | All of Block 586, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 16,63742 | |
| OCS-G-2021 | 2- 1-71 | All of Block 587, West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1 B | 22.80504 | |
| OCS-G-2850 | 12- 1-74 | All of Block 609, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.55555 |
| DCS-G-2559 | | All of Block 617, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.55555 |

| BLM or State | Effective | | Net Reven | ue Interest |
|--------------|-----------|---|-----------|-------------|
| Lease No. | Date | Description | POGI | PPC |
| OCS-G-2860 | 12- 1-74 | All of Block 237, Fast Cameron Area, South Ad. Ition, OCS Official Leasing Map, Louisiana Map No. 2A | 2.22499 | 5.55555 |
| OCS-G-2045 | 1- 1-71 | All of Block 270, East Cameron Area, South Addition. OCS Official Leasing Map, Louisiana Map No. 2A | 16.68741 | |
| OCS-G-2062 | 2- 1-71 | All of Block 334, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| OCS-G-2439 | 8- 1-73 | All of Biock 335, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| OCS-G-2078 | 2- 1-71 | All of Block 228, Vermilion Area, OCS Official Leasing Map, Louisiana Map No. 3 | 17.9273 | |
| OCS-G-2882 | 12- 1-74 | All of Block 125, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5.55555 |
| OCS-G-2883 | 12- 1-74 | All of Block 127, South Marsh Island Area, South Addition, CCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5.55555 |
| OCS-G-2587 | 5- 1-74 | All of Block 128, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5.55555 |
| OCS-G-0479 | 12- 1-54 | All of Block 53, Eugene Island Area as shown on Official Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisiana Offshore Operations) | | 83.33333 |
| OCS-G-0478 | 1- 1-55 | E 14 of Block 116. Eugene Island Area as shown on Official Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisiana Offshore Operations) | | 41.66665 |
| OCS-G-2102 | 2- 1-71 | All of Block 256, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 23.1054 | |
| OCS-G-2900 | 12- 1-74 | All of Block 261, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 5.56247 | 13.88888 |
| OCS-G-3156 | 7- 1-75 | All of Block 262. Eugene Island Area, as shown on OCS Official Leasing Map, Louisiana Map No. 4 | 5,56247 | 13.88888 |

| BLM or State | Effective | Net Revo | nue Interest |
|--------------|-----------|--|--------------|
| Lease No. | Date | Description POGI | PPC |
| OCS-G-2104 | 2- 1-71 | All of Block 295. Eugene Island Area, 6.76497 South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | |
| OCS-G-2607 | 5- 1-74 | All of Block 312, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisians Map No. 4A | 22.22221 |
| OCS-G-2115 | 1- 1-71 | All of Block 330, Eugene Island Area, 16.68741 South Addition, OCS Official Leasing Map No. 4A | |
| OCS-G-2317 | 2- 1-73 | All of Block 333, Eugene Island Area, South Addition, OCS Official Leasing Map No. 4A | 22.22201 |
| OCS-G-3332 | 3- 1-76 | All of Block 337, Eugene Island Area, 16.68741 South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | |
| OCS-G-3409 | 1- 1-77 | All of Block 351, Eugene Island Area, 8.34371 South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | 20.83333 |
| OCS-G-0823 | 5- 1-60 | All of Block 186, Ship Shoal Area, as shown on Official Leasing Map, Louis- iana Map No. 5, Outer Continental Shelf Leasing Map (Louisiana Off- shore Operations) | 28.33332 |
| OCS-G-1525 | 7- 1-67 | All of Block 222, Ship Shoal Area, OCS Official Leasing Map, Louisiana Map No. 5 | 15.83332 |
| OCS-G-1984 | 9- 1-70 | N 15 Block 225, Ship Shoal Area, OCS Official Leasing Map, Louisiana Map No. 5 | 19.79165 |
| OCS-G-2177 | 11- 1-72 | All of Block 49 South Pass Area, as 3.3.748 shown on OCS Official Leasing Map, Louisiana Map No. 9 | 8.33333 |
| OCS-G-31' * | 1- 1-77 | That portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (U.S. vs. Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971) as shown on OCS Official Leasing Map, Louisiana Map No. 9 | 16.66667 |

| BLM or State | Effective | | | nue Interest |
|--------------|-----------|---|-----------------|--------------|
| Lease No. | Date | Description | POGI | PPC |
| St. Ls. 10 | 5-13-74 | Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and being more particularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Low siana | 5.92738 | 14.80 |
| OCS-G-2185 | 10- 1-72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court entered December 13, 1965 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288), South Pass Area, South and East Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 9A | 5.00 <u>622</u> | 12.49999 |
| OCS-G-3195 | 7- 1-75 | Block 72. Main Pass Area, that portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map. Louisiana Map No. 10 | 7.54441 | 18.83758 |
| OCS-G-3417 | 1- 1-77 | That portion of Main Pass Blocks 72 and 74 located more than three geographs. I miles serward from the line described in the supplemental decree of the U.S. Supreme Court, J. ne 16, 1975 (U.S. vs. Louisiana, 422 U.S. 13) to a foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971) as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 7.54441 | 18.83758 |
| OCS-G-2947 | 12- 1-74 | All of Block 73, Main Pass Area, OCS Official Leasing Map, Louisiana Map No. 10. That portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)) | 7.54441 | 18.83758 |

| BLM or State Lease No. | Effective Date | Description | POGI POGI | P C |
|---------------------------|-------------------|--|-----------|----------|
| OCS-G-2193 | 10- 1-72 | All of Block 140. Main Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 3.33748 | 3,33333 |
| OCS-G-3206 | 7- 1-73 | All of Block N663E63, Mobile South No. 2, as shown on OCS Official Leasing Map, NH-16-10 (Also known as Block 63), Mississippi Canyon Area, as shown on OCS Official Protraction Diagram NH-16-10 | 3.33748 | S.J.P.E. |
| OCS-G-3500 | 8- 1-77 | All of Block 253, West Cameron Area, as shown on OCS Official Leasing Map, Louisiana Map No. 1 | 10.01245 | 25.6000 |
| OCS-G-3785 | 6- 1-78 | All of Block 372, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 12.4999 | 12.4999 |

PENNZOIL COMPANY, ET AL.

TO

NEW DUNSTAN, INC.

Instrument of Conveyance of Production Psyment

Dated July 16, 1981

Return recorded counterparts to:

PENNEOII. COMPANY Penneoil Place, P.O. Box 2967 Houston Texas 77001 Attention: Ms. Sally Hazen This instrument was prepared by:

Frank W. R. Hubert, Jr. 3000 One Shell Plaza Houston, Texas 77002

TABLE OF CONTENTS

| Paragr | aph | Page |
|--------|---|------|
| 1. | Certain Definitions | 1 |
| 2. | Conveyance of the Production Payment | 11 |
| | Certain Provisions Governing the Production Payment | 14 |
| 4. | Amount and Term of the Production Payment | 17 |
| 5. | Termination of the Production Payment | 20 |
| 6. | Application of the Proceeds of Production Payment Minerals | 21 |
| 7. | Payment of Consideration for the Production Payment | 23 |
| 8. | Marketing of Production Payment Minerals | 24 |
| 9. | Performance of Sales Contracts; Payments; Etc. | 26 |
| | Taking In Kind | 27 |
| 11. | Withholding and Restitution of Proceeds of Production Payment | |
| | Minerals | 27 |
| 12. | Protection to Purchasers of Production | 29 |
| 13. | Certain Payments by Grantors | 29 |
| 14. | Covenants of the Granters | 30 |
| 15. | Access to the Subject Interests | |
| 16. | | 38 |
| 17. | | 38 |
| 18. | | 40 |
| 19. | Reports and Payments to the Grantee | 41 |
| 20. | Representations, Warranties and Covenants | 42 |
| 21. | Transfer of the Subject Interests | 47 |
| 22. | Exception to Payments | 47 |
| 23. | | 48 |
| 24. | | 48 |
| 25. | Grantors' Rights to Make Payments to be Applied to Discharge of | |
| | Production Payment | 48 |
| 26. | Notices | 49 |
| 27. | Further Assurances | 49 |
| 28. | | 49 |
| 29. | | 50 |
| A 100 | Governing Law | 50 |
| 31. | Final Termination | 50 |
| 20 | Unadings | 50 |

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (the "Conveyance") dated July 16, 1981, between Pennzoil Company, a Delaware corporation, Pennzoil Producting Company, a Delaware corporation and Duval ("Correction, a Delaware corporation (the "Grantors"), and New Dunstan, Inc., a Delaware non-pin in corporation (the "Grantee"),

WITNESSETH :

Whereas, the Grantors desire to sell and the Grantee degrees to purchase the Production Payment hereinafter described,

Now, Therefore, it is agreed by and between the Grantors and the Grantee as follows:

1. Gertain Definitions. The following terms, when used in this Conveyance, have the meanings indicated below:

"Accounting Month" means any monthly period commencing at 7:00 A.M., local time in effect at the location of each Subject Interest, on the first day of any calendar month and ending at 7:00 A.M., local time in effect at the location of each Subject Interest, on the first day of the next succeeding calendar month.

"Affiliate" means, as to the party specified, any Person controlling, controlled by or under common control with such party, with the concept of control in such context meaning the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of another, whether through the ownership of voting securities, by contract or otherwise.

"Alternate Base Rate" means a fluctuating rate per annum equal at all times to the higher of:

- (a) the Base Rate; or
- (b) ½ of 1% above the latest three-week moving average of secondary market morning offering rates in the United States for three-month certificates of deposit of major United States money market banks, such three-week moving average

being determined weekly by Citibank on the basis of such rates reported by certificate of deposit dealers to and published by the Federal Reserve Bank of New York or, if such publication shall be suspended or terminated, on the basis of quotations for such rates received by Citibank from three New York certificate of deposit dealers of recognized standing, in either case adjusted to the nearest 1/4 of 1% or, if none, to the next hig' 1/4 of 1%,

each change in such fluctuating rate to take effect simultaneously with the corresponding change in such Base Rate or moving average.

"Applicable Rate" means, for each day, the weighted average of the Eurodollar Rates and the Domestic mate applicable to the unliquidated balance of the Primary Sum on such day determined on the basis of the proportion which that postion of such unliquidated balance to which each Eurodollar Rat and the Domestic Rate is applicable bears to the total amount of such unliquidated balance; provided, however, that in no event shall the Applicable Rate exceed the maximum rate which would be permitted by the laws of the State of New York if the Applicable Rate were being applied to a borrowing of money in an amount equal to the unliquidated balance of the Primary Sum. Except as provided hereinafter, the Grantors may from time to time select the Eurodollar Rate to be in effect with respect to all or any portion of the then unliquidated balance of the Primary Sum for a Rate Period, on the condition that notice of such selection (including notice of said portion of such unifoundated balance and of the duration of the Rate Period therefor) is delivered to the Grantee and to Citibank not later than 11:00 A.M. (New York City time) at least four Business Days prior to the commencement of such Rate Period, and provided that in no event shall the Grantors be permitted to have selected the Eurodollar Race to be in effect concurrently with respect to more than three portions of such unliquidated balance; and except to the extent that the Eurodollar Rate shall have been so selected by the Grantors in accordance herewith to be applicable to all or one or more portions of such unliquidated balance, the Domestic Rate shall be applicable to the

entire unliquidated balance of the Primary Sum. Notwithstanding the foregoing, and except as set forth in the next succeeding sentence hereto, if either

- (i) it shall become unlawful, or any central bank or other governmental authority shall assert that it is unlawful, for any lender to the Grantee to obtain funds in the London interbank market in order to fund or continue to fund or maintain its loan to the Grantee, upon such lender giving notice thereof to the Grantee (which shall give prompt notice thereof to the Grantors), or
- (ii) the Grantee shall demand payment of Incressed Financing Costs referred to in Section 4(d) hereof, and the Grantors shall have given the Grantee four Business Days' notice of their selection of the Domestic Rate in substitution for the Eurodollar Rate,

the Applicable Rate shall thereupon be the Domestic Rate as to those portions of the unliquidated balance of the Primary Sum previously subject to the Eurodollar Rate as to which clause (i) or (ii) is applicable, and the right of the Grantors to select the Eurodollar Rate hereunder shall thereupon terminate unless and until such funding is no longer unlawful or so asserted to be unlawful or the circumstances giving rise to such Increased Financing Costs have cear if to exist. Notwithstanding the foregoing sentence, in the event of the happening of any circumstance described in clause (i) of said sentence with respect to any such lender to the Grantee, or of demand for payment of any such Increased Financing Costs described in clause (ii) of said sentence attributable to any such lender, the Grantee covenants and agrees that, at the Grantors' request, it will consult with such lender in good faith to ascertain whether any arrangement (including, but not limited to, a change in such lender's Lending Office) satisfactory to such lender, the Grantors and the Grantoe can be made with a view to eliminating such Increased Financing Costs and, upon demand of the Grantors, the Grantee will cause such lender to be replaced by another financial institution designated by

the Granters, provided that such financial institution is acceptable to the Grantee and the lenders to the Grantee and is prepared to assume all of the rights and obligations of such original lender on terms acceptable to such lender; and if upon said replacement there shall not be continuing any circumstances described in either clause (i) or (ii) of the foregoing sentence, the Grantors shall thereupon once again be entitled to select the Eurodollar Rate hereunder. Any such financial institution which shall replace any such lender in connection with any loan in accordance with the preceding sentence shall be deemed to be a lender to the Grantec for all purposes hereof.

"Application Date" means a Regular Application Date or, if such Regular Application Date is not a Business Day, the Next Preceding Business Day. For purposes of this Conveyance, an Application Date shall be deemed to occur "at or near the end" of the Accounting Month which ends on such Application Date or, if such Application Date is a Next Preceding Business Day, the next following Regular Application Date.

"Base Rate" means a fluctuating interest rate per annum equal at all times to the rate of interest announced publicly by Citibank in New York, New York from time to time as Citibank's base rate.

"Business Day" means a day on which banks are not required or authorized to close in New York City and, if the Eurodotlar Rate is relevant, on which dealing are carried on in the London interbank market and banks are open for business in London.

"Citibank" means Citibank, N.A., a national banking associa-

"Discharge Schedule" means the tabulation set forth in Exhibit B attached to this Conveyance and hereby made a part hereof.

"Domestic Rate" means a fluctuating rate per annum (computed on the basis of a year of 365 or 366 days, as the case may be, for the a-ual number of days—including the first day but excluding the last day—occurring in the period for which such rate is computed) equal at all times to:

- (i) from the date hereof to and including July 31, 1983, 1/4 of 1% per annum above the Alternate Base Rate in effect from time to time;
- (ii) from and after August 1, 1983 to and including July 31, 1987, ½ of 1% per annum above the Alternate Base Rate in effect from time to time; and
- (iii) from and after August 1, 1987, 34 of 1% per annum above the Alternate Base Rate in effect from time to time.

"Effective Date" means 7:00 A.M., local time in effect at the location of each Subject Interest, on July 1, 1981.

"Eurodollar Rate" means, with respect to any portion of the unliquidated balance of the Primary Sum as to which such rate is applicable, a fluctuating rate per annum (computed on the basis of a year of 360 days for the actual number of days—including the first day but excluding the last day—cecurring in the period for which such rate is computed), subject to funds being available to the lenders to the Grantee on reasonable terms under routine interbank market conditions (all as determined in the discretion of the lenders to the Grantee), equal at all times to:

- (i) from the date hereof to and including July 31, 1983, 5% of 1% per annum above the LIBO Rate in effect from time to time for such portion of the Primary Sum;
- (ii) from and after August 1, 1983 to and including July 31, 1987, 34 of 1% per annum above the LIBO Rate in effect from time to time for such portion of the Primary Sum; and
- (iii) from and after August 1, 1987, 78 of 177 per annum above the LIBO Rate in effect from time to time for such portion of the Primary Sum.

"Excess Rate" has the meaning assigned to that term in Paragraph 4(d) hereof.

"Exhibit A" means Exhibit A attached to this Conveyance and hereby made a part hereof for all purposes of interpreting, construing and enforcing this Conveyance, provided, that if at any time said Exhibit A has been supplemented by the Grantors by

the execution and delivery of the Supplemental Instrument of Conveyance of Production Payment, then at such time Exhibit A shall mean said Exhibit A as so supplemented.

"First Production Period" means the period commencing with the Effective Date and ending at 7:00 A.M., local time in effect at the location of each Subject Interest, on August 1, 1983.

"Hydrocarbons" means all oil, gas and other gaseous and liquid hydrocarbons or any combination of one or more of such substances appropriate to the context in which such term is used.

"Increased Financing Costs" means any increase in the cost to the Grantee of obtaining or maintaining loans from any lender during any period in which the Eurodollar Rate is applicable to all or any portion of the unliquidated balance of the Primary Sum. and which is due to any increase in the cost to such lender to fund or continue to fund or maintain any loan to the Grantee through funds obtained in the London interbank market, arising from (i) the introduction of or any change (including, without limitation, any change by way of imposition or increase of reserve requirements) in, or in the interpretation of, any law or regulation since November 1, 1980 or (ii) the compliance by such lender with any request from any central bank or other governmental authority whether or not having the force of law, and if the Grantors shall exergise their option to select the Domestic Rate as the Applicable Rate prior to the last day of a Rate Period applicable to all or any portion of the Primary Sum, shall also include any costs to suc lender incurred as a result of the selection of such Rate. A certificate from such lender to the Grantee as to the amount of such increased costs shall be conclusive as to the amount of such increased cos's to the Grantee, absent manifest error,

"Lease" means any oil and/or gas lease or sub-lease, or any oil, gas and/or mineral lease or sub-lease, or any mining claim or award, or any fee or leasehold interest in, or any contractual right to produce or receive the proceeds of production of, any Mineral.

"LIBO Rate" means a rate per annum, if any, equal at all times during each Rate Period for any portion of the unliquidated balance of the Primary Sum to which the Eurodollar Rate is applicable to the average (rounded upward, if necessary, to the nearest whole multiple of 1/16 of 1% per annum) of the rate of interest per annum at which deposits in United States dollars in immediately available funds are offered by each of the Reference Banks to prime banks in the London interbank market at 11:00 A.M. (London time) two Business Days before the first day of such Rate Period for a period equal to such Rate Period and in an amount substantially equal to such portion of such unliquidated balance of the Primary Sum (such rate for each Rate Period being determined by Citibank on the basis of applicable quotations for the LIBO Rate furnished to and received by Citibank from the Reference Banks two Business Days prier to the first day of such Rate Period and, if any one or more of the Reference Banks shall not so furnish such a quotation for any Rate Period, the LIBO Rate for such Rate Period shall be based upon the quotations furnished by the other Reference Banks).

"Minerals" means Hydrocarbons and, in the event the Supplemental Instrument of Conveyance of Production Payment is executed and delivered (and from and after the effective date thereof), also includes sulphur.

"Next Preceding Business Pay" means, when used with reference to a Regular Application Date or Quarterly Regular Application Date, the first Business Day next preceding such Regular Application Date or Quarterly Regular Application Date if such Regular Application Date or Quarterly Regular Application Date is not a Business Day.

"Person" means an individual, corporation, partnership, joint venture, trust or unincorporated organization, or a government or any agency or political subdivision thereof.

"Primary Sum" means at any time the aggregate amount of the Purchase Price which shall at such time have been paid.

"Production Payment" has the meaning assigned to that term in Paragraph 2 hereof.

"Production Payment Minerals" has the meaning assigned to that term in Paragraph 2 hereof.

"Production Payment Percentage" has the meaning assigned to that term in Paragraph 2 hereof.

'I urchase Price" has the meaning assigned to that term in Paragraph 7 hereof.

"Quarterly Application Date" means a Quarterly Regular Application Date or, if such Quarterly Regular Application Date is not a Business Day, the next Preceding Business Day.

"Quarterly Regular Application Date" means the Regular Application Date falling on the first day of any January, April, July or October during the term of the Production Payment.

"Rate" means the Domestic Rate or the Eurodollar Rate.

"Rate Period" means any of the periods selected from time to time by the Grantors to apply in respect of the application of the Eurodollar Rate to all or any portion of the unliquidated balance of the Primary Sum. Each Rate Period shail be one, two, three or six months, as the Grantors shall, upon notice received by the Grantee not later than 11:00 A.M. (New York time) at least four Business Days prior to the commencement of such Rate Period. select: provided, however, that (a) if the Grantors shall fail on a timely basis to select a Rate Period for any portion of the Primary Sum to which the Eurodoilar Rate is applicable, the Domestic Rate shall apply to such portion of the Primary Sum as of the end of the then current Rate Period and (b) any Rate Period determined in accordance with Paragraph 4(d) hereof shall be for the period of time specified therein. Whenever the last day of any Rate Period would otherwise occur on a day other than a Business Day, the last day of such Rate Period shall occur on the next succeeding Business Day; provided, however, that if such extension would cause the last day of such Rate Period to occur in the next following calendar month, the last day of such Rate Period shall occur on the next preceding Business Day.

"Reference Banks" means the respective principal London offices of (a) Citibank, Morgan Guaranty Trust Company of New York and Mellon Bank, N.A. and (b) in addition to or in substitution for any such bank, any bank designated by the Grantors and approved by the Grantee.

"Regular Application Date" means the first day of any cal-

"Recerve Report" shall mean a report of Ryder Scott Company Petroleum Engineers or other independent petroleum engineers agreed upon by the Grantors and the Grantee furnished pursuant to subparagraph (c) of Paragraph 19 of this Conveyance.

"Sales Contracts" has the meaning assigned to that term in Paragraph 8 hereof.

"Second Production Period" means the period commencing at 7:00 A.M., local time in effect at the location of each Subject Interest, on August 1, 1983, and ending at 7:00 A.M., local time in effect at the location of each Subject Interest, on January 1, 1990, unless the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, in which case the Second Production Period shall end at 7:00 A.M., local time in effect at the location of each Subject Interest, on May 1, 1990.

"Subject Hydrocarbons" means all Hydrocarbons in and under, and which may be produced, saved and sold from, and which shall accrue and be attributable to, the Subject Interests from and after the Effective Date and throughout the period specified in this Conveyance for the term of the Production Payment, subject to the provisions of Paragraph 3 of this Conveyance.

"Subject Interests" shall mean each and every kind and character of right, title, claim or interest which any Grantor now has in the Leases or portions thereof which are described in Exhibit A and all unitization and pooling agreements and the units created thereby which cover or include such Leases or portions thereof, and all the right, title, claim or interest which any Grantor now has in and to the lands specifically described or referred to in Exhibit A or covered by the Leases which are described in Exhibit A, whether such right, title, claim or interest be under and by virtue of a Lease, a unitization or pooling agreement, a unitization or pooling order, a mineral deed, a royalty deed, an operating agreement, a division order, a transfer order or any other type of contract, conveyance or instrument or under any other type of claim or title, legal or equitable, recorded or unrecorded, even

though such Grantor's interests by necessary or incompletely described in, or a description thereof 's omitted from, Exhibit A, all as the same shall be enlarged by an reversionary interest or by the discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and cit reas wais and extensions of any of the same. but there slall not be included within the Subject Interests an, rights while ... Grantor now has or may hereafter obtain with resp. the production attributable to the interests of nonconsenting parties u. cer any operating agreement, unit operating agreement, contract for development or similar instrument. Without in any way live ting the foregoing, the Subject Interests shall include each and ever kind and character of right, title, claim or interest which any Contact has in any and all Leases covering lands situated in those counties in the States of New York, Pennsylvania or West Virginia named in Exhibit A, regardless of whether such Leases are described in Exhibit A.

"Subject Minerals" means Subject Hydrocart and Subject Sulphur.

"Subject Sulphur" means, in the event the Supplemental Instrument of Conveyance of Production Payment is executed and delivered, all sulphur in and under, and which may be produced, saved and sold from, and which shall accrue and be attributable to, the Subject Interests described in the Supplemental Instrument of Conveyans of Production Payment from and after the effective date thereof and throughout the remainder of the period specified in this Conveyance, or the term of the Production Payment, subject to the provisions of Paragraph 3 of this Conveyance.

"Subject Taxes" has the meaning assigned to that term in Paragraph 13 hereof.

"Supplemental Instrument of Conveyance of Production Payment" means the Supplemental Instrument of Conveyance of Production Payment in the form of Exhibit i attached herete and hereby made a part hereof. The execution and delivery of the Supplemental Instrument of Conveyance of Production Payment

shall not require the consent or joinder of the Grantee or any mortgage or trustee referred to in Paragraph 24 hereof.

"Paird Production Period" means the period commencing at 7:00 A.M., lees, "the in effect at the Preation of each Subject Interest, on an inary 1, 1990, unless the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, in which care the Third Preduction Period shall commence at 7:00 A.M., rocal time is effect at each on the Subject Interests, or May 1, 1990 and end on the date the Production Payment is discharged or terminated as herein provided.

2. Conveyance of the Production Payment. The Grantors, in consideration of the payment to them by the Grantee of the Purchase for e pursuant to the provisions of Paragraph 7 of this Conveyance and of other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, by these presents do hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, effective as of the Effective Date, subject to the limitations and encumbrances set forth in Paragraphs 14(f) and 20 of this Conveyance and in Exhibit A, as a production payment, the undivided percentages, or if 100%, the percentage, of Subject Minerals provided for in the following subparagraphs First through Fourth of this Paragraph 2:

First: The undivided percentage of the Subject Minerals conveyed under this subparagraph First during the First Production Period shall be that undivided percentage tuple of 100%, or, if 100%, the percentage, of the Subject Minerals as shall be required to cause the Grantee to be entitled to receive, out of the Subject Minerals produced and saved during each Accounting Month during the First Production Period, a sum which, when actually received by the Grantee in accordance with the terms of this Conveyance, would equal a sum hereby specified to be the aggregate amount referred to in subparagraphs (b), (c), (d), (e) and (f) of Paragraph 4 of this Conveyance to the extent such aggregate amount shall have been ascertained or accrued during (or, in the case of said subparagraphs (c), (c!), (e) and (f), prior to) such Accounting Month and remains unpaid;

- Second. Subject to the provisions of subparagraph Third of this Paragraph 2, the undivided percentages of the Subject Minerals conveyed under this subparagraph Second during each Accounting Month during the Second Production Period shall be the undivided percentages specified in the following clause (a) or clause the whichever entitles the Grandee to receive the greater amount of proceeds out of Subject Minerals produced and saved during such Accounting Month:
 - (a) an undivided 65% of the Subject Hydrocarbons and an undrawed 35% of the Subject Sulphur produced and saved during such Accounting Menth, provided, however, that in the event the Princary Sum at the close of business on July 31, 1983 is less than \$2,500,000 000, then (1) if the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, sam undivided percentages of Subject Hydrecarbons and Subject Sulphur shall be adjusted to the percentages determined by multiplying each of 65% and 35% I a traction the numerator of which shall be the inliquidated balance of the Primary Sun at the close of business on July 31, 1023 and the denominator of which shall be \$2,500,000,500 or (2 If the Supplemental Instrument of Conveyance of Production Payment has not been ever to I and delivered, said undivided percentage of Subject abdrocarbons shall be adjusted to the percentage determined by multiplying 65% by a fraction the numerator of whice shall be the unliquidated balance of the Primary Sum on July 51, 1983 and the denoninator of which shall be \$1,865,000,000 at
 - that undivided percent up (u., to 90%) of the Subject Hydrographons and (but only it said andivided percentage of the project Hydrographons for such Accounting Month is 90% and it insufficient to entitle the Grantee to receive the sum specialed in this splat agraph (b)) that undivided percentage (up to 40%) of the Subject Sulphur produced and saved during such Accounting Menth as shall be required to cause the Grantee to be entitled to receive, out of the Subject Mineral produced and saved during such Accounting Manth, a sum which, when actually received by the Grant c in accordance with the terms of this Conveyance, would equal a sum breeby

specified to be the total of (i) the aggregate amount referred to in subparagraphs (b), (c), (d) (e) and (f) of Paragraph 4 of this Conveyance to the externs the aggregate amount shall have been ascertained or accrued along (or, in the case of said subparagraphs (e), (d), (e) and (f), prior to) such Accounting Month and remains unpaid, plus (ii) an amount determined by multiplying the unliquidated balance of the Primary Sum at the close of business on July 31, 1983 by the percentage shown on the Discharge Schedule under the heading "Hydrocarbons Monthly Percentage" or, in the event the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, the percentage shown on the Discharge Schedule under the heading "Minerals Monthly Percentage", opposite the Regular Application Date occurring at or near the end of such Accounting Month;

Third: If on any Application Date during the Second Production Period the aggregate amount of reductions in the unliquidated balance of the Prinary Sum made on such A; plication Date and prior thereto is less than as amount determined by multiplying 95% times the cumulative total of the amounts determined for such Application Date and all an ounts determined for all previous Application Dates ar provided in subclause (ii) of clause (b) of subparagraph Second of this Paragraph 2 by reference to the "Hydrocarboas Monthly Per entere" or, in the event the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, by reference to the "Minerals Monthly t'ercentage", t': t etwichstanding the provisions of subparagraph Second above high shall not be effective during any period during which the precisions of the subject agraph Third are opera-(2.1) during the Accounting Month is ginning at or near such Application Date, the undivid doc centages of the Subject Minerals conveyed under this subpartmanh Third shall be an undivided 90% of the Subject Hydrocar sons and an undivided 40% of the Subject Sulphur .

and

Fourth: The percentage of the Subject Minerals conveyed under this subparagraph. Fourth Juring the Third Production

Period shall be one hundred percent (100%) of the Subject Minerals;

together with an easement of ingress and egress to remove the same from the applicable Subject Interests to the extent Grantor Prive the right to grant such an easement (the unalvided percentages of the Subject Minerals so granted, bargained, seld, conveyed, assigned, transferred and delivered and at any time in effect being herein called the "Production Payment Percentages", the Subject Minerals attributable to the Production Payment Percentages being herein called the "Production Payment Minerals", and the Production Payment Minerals, together with such easement of ingress and egress and all other interests hereby granted, bargained, sold, conveyed, assigned, transferred and delivered, or intended so to be, being herein called the "Production Payment").

To Have And To Hold the Production Payment, together with all and singular the rights, tistes, interests, estates, remedies, powers, privileges and appartmances thereunto in any way belonging, unto the Grantee, its successors and assigns, forever.

- 3. Certain Provisions Governing the Production Payment. All the provisions of this Conveyance shall be subject to the following principles:
 - (a) Except as provided in Paragraphs 14(b)(ii) and 14(c) hereof, for the discharge of the Production Payment the Grantee shall look exclusively to the proceeds of Production Payment Minerals, and the Grantors shall not be liable for such discharge.
 - (b) Except as provided in the provise clause below, the Production Payment shall not be dischargeable out of any advance payments made to the Grantors for Production Payment Minerals to be produced in the toture and covered by, and made in advance of such production under the terms of, any advance payment or take or pay agreement, provided, however, that the Production Payment shall be dischargeable out of, and the Country shall (except to the extent that payment is made directly to the Grantee by the purchaser or unless the Production Payment shall have been fully discharged) make payment to the Grant eight as on account of,

any Production Payment Minerals covered by any such advance payment or take or pay agreement when such Production Payment Minerals shall be produced and saved, on the basis of the full purchase price under the Sales Contract for such Production Payment Minerals so produced and saved (such full purchase price to be without regred to any deduction from or credit against the payment of such price due to any previous payment made in advance of production).

- (c) The Production Payment shall not be dischargeable out of any bonus which the Grantors shall receive for any Lease or assignment of any of the Subject Interests or out of any payments made to any of the Grantors in connection with the drilling or deferring of drilling of any well on any of the Subject Interests or in connection with any adjustment of any well and leasehold equipment upon unitization of any of the Subject Interests.
- (d) There shall not be included in the Subject Minerals any Minerals unavoidably lost in the production thereof or produced and saved from any of the Subject Interests and used by the Grantors in conformity with good field practices for drilling, mining and production operations (including gas injection, secondary recovery, pressure maintenance, repressuring or cycling operations) conducted for the purpose of producing Minerals from the Subject Interests or from any unit to which the Subject Interests are committed, but only so long as such Minerals are so used.
- (e) So long as and to the extent that the same may be required by applicable laws and regulations, in the case of any Lease from the United States of America included in the Subject Interests from which the average production of oil per well per day averaged on the monthly basis is 15 barrels or less, the obligation to pay and the right of the Grantee to receive the proceeds of oil produced from such Lease shall be suspended until said average production of oil per well per day exceeds said minimum amount, and such suspension shall apply separately to any zone or portion of such Lease segregated for computing government royalties; provided, 2 wever, that during the period of any such suspension, the Production Payment Percentage applicable to production from

each other Lease included in the Subject Interests, or portion or zone the eof, shall (subject to the provisions of Paragraph 2 hereof) be increased to that percentage necessary to cause the Grantee to receive, in addition to the production from each such other Lease (or portion or zone thereof) attributable to the Production Payment Percentages otherwise applicable under Paragraph 2 hereof, proceeds of production from each such other Lease, or portion or zone thereof, equivalent in the aggregate to the amount of proceeds of production the Grantee would have received from the Lease, or portion or zone thereof, as to which the Grantee's rights to production have been suspended had such rights not been so suspended. The increases in the Production Payment Percentages pursuant to the foregoing provisions of this subparagraph (e) are hereby conveyed by the Granters to the Grantee effective as of the Effective Date as a part of the Production Payment.

- (f) There shall not be included in the Subject Minerals any Minerals attributable to nonconsent operations conducted with respect to the Subject Interests (or any portion thereof) as to which any Grantor shall be a nonconsenting party and dedicated to the recoupment or reimbursement of costs and expenses of the consenting party or parties by the terms of the relevant operating agreement, unit agreement, contract for development or other instrument providing for such nonconsent operations, provided such Grantor's election not to participate in such operations is made in conformity with the provisions of Paragraph 14(a) of this Conveyance.
- (g) Whenever reference is made in this Conveyance to the term "fair market value at the wellhead or mine" in determining the price to be paid or obtained for Production Payment Minerals sold for the account of the Grantee, or purchased by any Grantor, or taken in kind by the Grantee, such term shall be understood to mean the price paid or payable for such Minerals at the wellhead or mine in the case of Minerals customarily sold at the wellhead or mine or, in the case of Minerals customarily sold or disposed of otherwise than at the wellhead or mine, the price paid or payable for such Minerals at the nearest point where such Minerals may be marketed and sold after production, less all costs of transporting such Minerals to such point from the wellhead or mine.

- (h) In the case of Production Payment Minerals sold or disposed of other than at the wellhead or mine, the amount of all reasonable costs incurred and paid by the Grantors in transporting such Minerals to the point of sale or dispositio—shall be deducted from the proceeds of sale of the Production Payment Minerals, and the amount so deducted shall be deemed not to be proceeds of the Production Payment Minerals received or realized by the Grantee.
- (i) Cognizant of Sections 171,26(d), 172,22(d), 173,21(d) and 171.78(b) of Title 25 of the Code of Federal Regulations, it is agreed, with respect to any Lease from Indian tribes, bands or groups, and any Lease on lands alletted to Indians in severalty, included in the Subject Interests, that nothing in this Conveyance shall be construed as modifying any of the obligations of the lessee under such Lease, including, but not limited to, obligations for diligent development and operation, protection against drainage, compliance with oil and gas operating regulations (30 CFR Part 221), and the requirement for departmental approval before abandonment of any well situated on such Lease. All such obligations are to remain in full force and effect, the same as if free of the Production Payment. The existence of the Production Payment, whether or not actually paid, shall not be considered as justification for the approval of abandonment of any well situated on such Lease.
- 4. Amount and Term of the Production Payment. The Production Payment shall continue and remain in full force and effect until such time as the Grantee shall have received and realized out of the proceeds of Production Payment Minerals, free and clear of all development, operating, mining, producing, treating, processing, handling, storing, marketing, transporting and other costs and expenses of every kind whatsoever, the full aggregate sum of the following amounts:
 - (a) The Primary Sum; plus
 - (b) An amount computed at the Applicable Rate on the unliquidated balance from time to time of the Primary Sum,
 - (i) The first such computation to be made on September 1, 1981, for the period commencing with and including the date hereof up to and including August 31, 1981, on the amount of such unliquidated balance of the Primary Sum from time to time during such period, and

(ii) Subsequent computations to be made monthly or the first day of each calendar month for the preceding calendar month on the amount of the unliquidated balance from time to time of the Primary Sum during such preceding calendar month;

plus

- (c) The sum of \$100,000 payable on September 1, 1981, and, in the event the Primary Sum at any date exceeds \$1,000,060,000, the additional sum of \$100,000 payable on the Application Date next following such date, and, in the event the Primary Sum at any date exceeds \$1,865,000,000, the additional sum of \$50,000 payable on the Application Date next following such date; plus
- (d) A sum equal to the aggregate of (i) all Increased Financing Costs from time to time of the Grantee and (ii) with respect to each portion of the unliquidated balance of the Primary Sum to which the Eurodollar Rate is applicable which may be paid on other than the last day of the Rate Period therefor (including without limitation any payments made pursuant to subparagraph (f) of Paragraph 6 hereof or Paragraph 25 hereof), an amount computed on the basis of the Excess Rate (as defined below) for such portion of the unisquidated balance of the Primary Sum on the amount of such portion so paid, for the period commencing on the day of such payment and ending on the last day of the current Rate Period for such portion (for the purpose of computing such amount, the "Excess Rate" in respect of any portion of the Primary Sum so paid shall be a rate per annum equal to the excess (if any) of (x) the LIBO Rate in effect for the current Rate Period for such portion over (y) the LIBO Rate computed on the basis of a Rate Period commencing on the day of such payment and ending on the last day of the current Rate Period for such portion) and (iii) interest on the amounts specified in clauses (i) and (ii) at the Applicable Rate (but not to exceed the maximum rate permitted by replicable law);

plus

(e) A sum equal to the aggregate of all the Subject Taxes to the extent that such taxes have not been paid by the Grantors pursuant to Paragraph 13 hereof, together with interest on the unliquidated amount thereof at a rate equal to the Applicable Rate (but in no event in excess of the maximum rate permitted by applicable law) in effect during the period from and including the date of payment of each such amount by the Grantee (notice of each such payment to be given promptly to the Grantors) to the date the Grantee is reimbursed therefor; plus

(f) A sum equal to the aggregate of all reasonable expenses of the Grantee incidental to the acquisition, ownership, mortgaging and transfer of the Production Payment or any part thereof, the obtaining of any loan secured thereby and the receipt and disbursement of funds on account thereof (including, without limitation, any commitment fees, any withholding or other similar taxes (together with any interest, penalties and expenses in connection therewith), any costs and expenses of any lender making such loan incurred as a result of any installment of the Purchase Price not being paid on the date specified in a notice by the Grantors pursuant to Paragraph 7(b) of this Conveyance or failure to make a payment on the date and before the time specified in a notice provided pursuant to Paragraph 6 with respect to any portion of the unliquidated balance of the Primary Sum to which the Eurodollar Rate is applicable and all fees and expenses of accountants and counsel for the Grantee and of counsel for the agent for any lenders making such loan, all expenses and taxes imposed by any state within which any Subject Interest is wholly or partially located on such lender attributable to the mortgaging of the Production Payment, all fees incurred in connection with, and all other expenses and costs of litigation and the contest, release or discharge of, any adverse claim or demand made or proceedings instituted by any person affecting in any manner whatsoever the Production Payment, the Production Payment Minerals or the proceeds thereof and all other fees, costs and expenses of the Grantee and any leader making such loan in connection with or arising out of any investigation, litigation or proceeding related to any acquisition or proposed acquisition of all or any portion of the stock or substantially all of the assets of any Person, directly or indirectly, with any of the proceeds of the Purchase Price, which shall have been paid or incurred by the Grantee or any lender making such loan, together with interest on the unliquidated amount thereof at a rate equal to the Applicable Rate (but not to exceed the maximum rate permitted by applicable law) in effect during the period from and including the date of payment of each such amount by the Grantee (notice of each such payment to be given promptly to the Granters) to the date the Grantee is reimbursed therefor.

IT BEING THE INTENTION OF THE GRANTORS AND THE GRANTEE IN ENTERING INTO THIS CONVEYANCE that the Grantee shall be entitled to receive and realize out of the Production Paymert Minerals the full aggregate sum 6. the amounts described in subparagraphs (a), (b) and (c) of this Paragraph 4, free and clear of all costs and expenses with respect to the Subject Interests and the Subject Minerals (which expenses, costs and fees shall be borne by the Grantors) and over and above all taxes, assessments, fees, expenses and costs of the character and amount described, specified or referred to in subparagraphs (d), (e) and (f) of this Paragraph 4.

All of the Subject Taxes may be deducted from the proceeds of the Production Payment Minerals and paid for the account of the Grantee and, in the event of any such deduction, the amount so deducted shall be deemed not to be proceeds of the Production Payment Minerals received or realized by the Grantee. Similarly (and anything in this Conveyance to the contrary notwithstanding), the tax under the Crude Oil Windfall Profit Tax Act of 1980 and any amendments or supplements thereto (or any tax enacted in substitution therefor) imposed or assessed with respect to or measured by or charged against or attributable to the Production Payment Minerals or the proceeds thereof and for which the Grantors are liable may be deducted from the proceeds of the Production Payment Minerals and paid by the Grantors and, in the event of any such deduction, the amount so deducted shall be deemed not to be proceeds of the Production Payment Minerals received or realized by or payable to the Grantee.

5. Termination of the Production Payment. When the full aggregate sum of the amounts specified in subparagraphs (a), (b), (c), (d), (e) and (f) of Paragraph 4 of this Conveyance (as expressly provided by the terms of this Conveyance) has been received by the Grantee, the Production Payment shall forthwith terminate, and all interest therein shall immediately revert to and become vested in the

Grantors to the same extent and with the same force and effect as if this Conveyance had not been made. The Grantee agrees that, upon such termination and upon indemnification against loss by reason of such acts and by rea on of the restitution of proceeds of Production Payment Minerals as provided in Paragraph 11 of this Conveyance, it shall execute or cause to be executed, upon the request and at the expense of the Grantors, such instruments as may be necessary or relate to evidence the termination of the Production Payment.

- 6. Application of the Proceeds of Production Payment Minerals. For all purposes of this Conveyance, the proceeds of Production Payment Minerals actually received by the Grantee at or before 11:00 A.M. (local time in effect in The City of New York) on any Application Date (commencing with September 1, 1981) and not theretofore applied pursuant to this Paragraph 6, shall be deemed to have been received and applied on such Application Date as follows:
 - (a) First, to the amount referred to in subparagraph (b) of Paragraph 4 of this Conveyance to the extent such amount has been ascertained or accrued up to (but not including) such Application Date and which has not been added to the Primary Sum pursuant to the proviso to this Paragraph 6 or paid by any previous application of proceeds of Production Payment Minerals pursuant to the provisions of this Paragraph 6, and
 - (b) Second, to the amount referred to in subparagraph (d) of said Paragraph 4 to the extent such amount has been ascertained or accrued up to (but not including) such Application Date and which has not been paid by any previous application of proceeds of Production Payment Minerals pursuant to the provisions of this Paragraph 6, and
 - ird, to the amount referred to in subparagraph (c) of th 4 and which has not been paid by any previous roceeds of Production Payment Minerals pursuant of this Paragraph 6, and
 - said Paragraph 4, to the extent then ascertained and which has not been plid by any previous application of proceeds of Produc-

tion Payment Minerals pursuant to the provisions of this Paragraph 6, and

- (e) Fifth, to all other amounts referred to in subparagraph (f) of said Paragraph 4, to the extent then ascertained and which have not been paid by any previous application of proceeds of Production Payment Minerals pursuant to the provisions of this Paragraph 6, and
- (f) Sixth, to the reduction of the unliquidated balance of the Primary Sum, initially by reduction of that portion of the unliquidated balance of the Primary Sum to which the Domestic Rate is applicable and then by reduction of those one or more portions of the unliquidated balance of the Primary Sum to which a Eurodollar Rate is applicable in such order as may be selected by the Grantors by not less than four Business Days' prior written notice to the Grantee or, if and to the extent not so selected, in the order in which the last days of the respective Rate Periods then in effect are scheduled to occur;

provided, however, that, in the event such proceeds so applied on the relevant Application Date shall be insufficient to cover the full amount specified in subparagraph (a) of this Paragraph 6, the unliquidated balance of the Primary Sum shall forthwith be increased by an amount equal to the amount of such deficiency.

Within 15 Business Days after the date of each such application by the Grantee of any proceeds of Production Payment Minerals, the Grantee will furnish or cause to be furnished to the Grantors a written statement showing the application of such proceeds in accordance with the provisions of this Paragraph 6.

If the Eurodollar Rate is applicable to any portion of the unliquidated balance of the Primary Sum to which proceeds are to be applied on an Application Date and such Application Date is on a date other than the last day of the Rate Period for such Eurodollar Rate, Grantors shall give written notice to the Grantee (and any lenders to the Grantee) at least four Business Days prior to such Application Date that the Grantors will pay such portion of the unliquidated balance of the Primary Sum prior to 11:00 A.M. (local time in effect in The City of New York) on such Application Date.

7. Payment of Consideration for the Production Payment.

- (a) In consideration of the conveyance by the Grantors to the Grantee of the Production Payment pursuant to this Conveyance, the Grantee hereby agrees to pay to the Grantors the m of \$2,500,000,000 (such sum, less the aggregate amount of reductions ther pursuant to the provisions of subparagraph (d) of this Paragraph 7, being herein called the "Purchase Pri e"), of which \$10,000,000 is being paid by the Grantee to the Grantors concurrently with the execution and delivery . this Conveyance and the remaining \$2,490,000,000 is payable in one or more installments, each to be in an amount not less than \$50,000,000 (except that any installment which causes the aggregate amount of the Purchase Price then paid to equal \$1,865,000,000 or. if the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, \$2,500,000,000, may be in such lesser amount which shall cause the aggregate amount of the Purchase Price then paid to equal \$1.865,000,000 or \$2.500,000,000 as the case may be), payable by the Grantee at the times and in (1 e amounts specified by the Grantors as hereinafter in this Paragraph 7 provided.
- (b) Each installment of the remaining \$2,490,000,000 of the Purchase Price shall become payable on the Business Day as to which the Grantee shall have received from the Granters at Last four Business Days' prior written notice specifying the amount of such installment and the Business Da., for payment of such installment. Any provision of this Paragraph 7 to the contrary notwithstanding, the obligation of the Grantee to pay each installment of the Purchase Price on the Business Day on which such installment shall otherwise be payable hereunder shall be subject to the conditions precedent that on such Business Day (i) the Grantee shall have received a certificate, substantially in the form of Exhibit D attached hereto and hereby made a part hereof, of the Grantors signed by duly authorized officers of the Grantors and dated such Business Day (the statements made in which certificate shall be true and correct on such Business Day), and (a) as to the installment which, when paid, we did cause the aggregate amount of the Perchase Price then paid to exceed \$1,865,000,000, the Grantors shall have executed and delivered to the Grantee the Supplemental Instrument of Conveyance of Production Payment.
- (c) Any provision of this Paragraph 7 to the contrary notwithstanding, the obligation of the Grantee to pay any further installments

of the Purchase Price pursuant to this Paragraph 7 shall terminate upon the occurrence of an Event of Default specified in clause (c) or (d) of Paragraph 17.

- (d) The obligation of the Grantee to pay any installment or installments of the Purchase Price which is not or are not payable on or prior to (i) the date of termination of the Production Payment pursuant to Paragraph 5 hereof or (ii) July 31, 1983, whichever date first occurs, shall terminate on such date and thereupon the amount of the Purchase Price shall be reduced by the amount of the unpaid portion of the Purchase Price which the Grantee is not obligated to pay.
- (e) All payments made by the Grantee on account of the Purchase Price shall be paid in lawful money of the United States of America and in immediately available funds by a deposit to the credit of the Grantors' account (No. 39019761) with Citibank at 399 Park Avenue, New York, New York 10043.
- (f) The Grantors hereby irrevocably and unconditionally waive and relinquish (i) any express or implied vendor's lien, and any other lien, security interest, charge or encumbrance, which would otherwise be imposed on or affect the Production Payment or the Production Payment Minerals on account of any unpaid installment or installments of the Purchase Price and (ii) with respect to the obligations of the Grantors to make payments under this Conveyance (including, without limitation, payments of proceeds of Production Payment Minerals under Paragraphs 3, 8, 9 and 11 hereof and payments under Paragraphs 13, 17 and 25 hereof), any set-off, counterclaim, credit, recoupment, defense, abatement, suspension, reduction and other right or claim which the Grantors may have against the Grantee as a result of or arising out of the failure of the Grantee to pay any amount on account of the Purchase Price.
- 8. Marketing of Production Payment Minerals. The Grantors, at their own expense, shall market or cause to be marketed the Production Payment Minerals, on behalf of and for the account of Grantee, on the same basis as the Grantors market their share of Subject Minerals, and for such purpose the Grantee agrees that sales of Production Payment Minerals may continue to be made by the Grantors with their share of Subject Minerals pursuant to existing contracts and agree-

ments entered into by the Grantor, ? the sale of, or commitment to sell, Subject Minerals and that the Grantors may enter into one or more contracts and agreements in the future for the sale of Subject Minerals (including Production Payment Minerals included therein) at the best prices and on the best terms as the Grantors shall determine in their good faith judgment are reasonably obtainable in the circumstances (all such present and future contracts and agreements for the sale of or commitment to sell Subject Minerals being herein called "Sales Contracts"), and it shall not be necessary for the Grantee to join in any Sales Contract or any amendment thereof; provided, however, that the Grantors will not agree to, or permit, any amendment, modification or termination in any manner adverse to the interests of the Grantee of any Sales Contract with an Affiliate of any Grantor, nor will the Grantors enter into any Sales Contract in the future with an Affiliate of any Grantor (except pursuant to the terms of any presently existing Sales Contract) in any manner adverse to the interests of the Grantee, without the prior written consent of the Grantee, and the Grantors will give the Grantee at least 15 days' prior written notice of the terms of any proposed amendment, modification or termination of any Sales Contract with an Affiliate of any Grantor, or any proposed future Sales Contract with an Affiliate of any Grantor. Subject to the rights of purchasers under then existing Sales Contracts and to any existing right of the Grantee to take in kind pursuant to notice given in accordance with Paragraph 10 hereof, and (so long as the Grantors are not in default in the performance or observance of any covenant or agreement on the Grantors' part to be performed or observed hereunder) the Grantors shall be entitled at any time and from time to time to purchase all or any part of the Production, Payment Minerals (including Production Payment Minerals which the Grantee has elected to take in kind provided that the Grantors shall have, within 30 days of the giving of any notice by the Grantee in accordance with Paragraph 10 hereof, given written notice to the Grantee that the Grantors elect to purchase all or part of the Production Payment Minerals specified in such notice given by the Grantee and specifying when the Grantors' purchases shall commence and the duration thereof), and, in such case, the Grantors shall pay to the Grantee, on or before 11:00 A.M. (local time in effect in the City of New York) on the Application Date at or near the end of the Accounting Month next succeeding the Accounting Month in which such purchase by the Grantors occurs, in immediately available funds, a price

equal to the fair market value at the wellhead or mine at the time of purchase of the Production Payment Minerals so purchased, less the amount of all severance taxes and assessments, payable by the Grantee or the Granters, of the character referred to in Paragraph 4(e) or the last sentence of Paragraph 4, attributeble to the production and sale of such Production Payment Minerals, and the amount of such taxes and assessments so deducted shall be deemed not to be proceeds of Production Payment Minerals received or realized by the Grantee.

9. Perform tice of Sales Contracts; Payments; Etc. The Grantors will duly perform all obligations binding on them under all Sales Contracts in accordance with the terms thereof and will take all appropriate measures to enforce the performance under each of the Sales Contracts of the obligations of the purchaser thereunder. All Production Payment Minerals sold by the Grantors, whether pursuant to Sales Contracts or otherwise, shall be delivered by the Grantors to the purchasers thereof, for the credit of the Grantee, to such point of purchase as is reasonably required in the marketing of such Production Payment Minerals. So long as no Event of Default has occurred and is continuing, the proceeds of the sale of all Subject Minerals (whether pursuant to Sales Contracts or otherwise) shall be paid to the Grantors by the purchasers thereof or other parties obligated to make payment therefor. All proceeds of Production Payment Minerals shall constitute trust funds in the possession of the Grantors and all such proceeds of Production Payment Minerals received by the Grantors on or before the next to last Business Day of each calendar month which are proceeds of Production Payment Minerals produced and saved during any Accounting Month prior to the Accounting Month ending on the first day of such calendar month and not theretofore paid over shall be paid over to the Grantee at or before 11:00 A.M. (local time in effect in The City of New York) on the Application Date which next follows the next to last Business Day of such calendar month in lawful money of the United States of America and in immediately available funds by a deposit to the credit of the Grantee's account (No. 39019753) with Citibank at 399 Park Avenue, New York, New York 10043. Upon the occurrence and during the continuance of any Event of Default (but not otherwise), the Grantee shall have the right to receive directly from the purchasers of the Production Payment Minerals, and to direct

such purchasers to pay directly to the Grantee, amounts to be paid for purchases of Production Payment Minerals, whether pursuant to Sales Contracts or otherwise.

- 10. Taking In Kind. From time to time upon the giving of at least 120 days' prior written ice to the Grantors (except that no such notice shall be required if an Event of Default shall have occurred and is cor nuing), the Grantee shall have the right, at its option, to take in kind all or any part of the Production Payment Minerals; subject, however, to (i) the rights of any purchasers under then existing Sales Centracts, and (ii) the right of the Grantors to purchase Production Payment Minerals pursuant to Paragraph 8 hereof. Each such notice from the Grantee shall specify the Minerals to be taken in kind, the date such taking is to commence and the proposed duration thereof The Grantee shall market any such Production Payment Minerals promptly at the best price and on the best terms the Grantee shall determine in its good faith judgment are reasonably obtainable in the circumstances, and the Production Payment shall be credited with the proceeds of any such sale, in the same manner as if such Production Payment Minerals had been marketed by the Grantors. No such taking by the Grantee shall obligate the Grantors to incur additional expenses by reason of such taking, but the Grantors shall continue to pay all costs and expenses incurred by them in the same manner and to the same extent hereunder as if there had been no such taking, provided that the Grantors shall be reimbursed out of the proceeds of sale of such Production Payment Minerals marketed by the Grantee pursuant to this Paragraph 10, to the extent actually incurred by the Grantors, for the amount of all the transportation costs of the character referred to in Paragraph 3(h) and for the amount of all Subject Taxes attributable to the production and sale of the Production Payment Minerals so taken and marketed by the Grantee and the amount of any such transportation costs and Subject Taxes so reimbursed to the Grantors shall be deemed not to be proceeds of Production Payment Minerals received or realized by the Grantee.
- 11. Withholding and Restitution of Proceeds of Production Payment Minerals. All obligations of the Grantors hereunder shall be subject to the applicable provisions of each statute purporting to provide regulation of the sale of Minerals or establishing maxi-

mum prices at which the same may be sold and all applicable rules and regulations thereunder of each governmental agency, board or commission having jurisdiction. Rates permitted under each such statute and the rules and regulations thereunder to be paid for natural gas or other Minerals included in the Production Payment Minerals shall be controlling if lower than prices established in Sales Contracts or if lower than the fair market value at the wellhead or mine. The Grentors shall be entitled to use their reasonable discretion in making filings, for themselves and on behalf of the antee, with any governmental agency, board or commission having jurisdiction, affecting the price or prices at which Subject Minerals may be sold. If any proceeds of Production Payment Minerals shall be with eld for any reason whatsoever, including, without limitation, withholding by the Grantors pending a final, non-appealable order on any application for certificates or determinations with respect to a sale, or a rate increase, filed with any governmenta! agency board or commission having jurisdiction (and the Grantors are hereby authorized to make such withholding pending such final determination), the Grantee shall not be deemed to have received or realized such proceeds until, and only to the extent that, such proceeds shall actually have been received by the Grantee; provided, however, that promptly (and in no event later than 30 days) following the receipt by the Grantors of a final non-appealable order in any such rate proceeding, the Grantors shall promptly pay over to the Grantee the amount vy proceeds of Production Payment Minerals so withheld by the Grantors, in conformity with such non-appealable order. If, at any time before or after receipt of the full aggregate sum specified in subparagraphs (a), (b), (c), (d), (e) and (f) of Paragraph 4 of this Conveyance, the Grantee shall be compelled, for any reason whatsoever, to make any payment or restitution, or such payment or restitution shall be made for the recount of the Grantee, on account of any proceeds of Production Payment Minerals, which proceeds have already been applied pursuant to the provisions of Paragraph 6 of this Conveyance, the unliquidated balance of the Primary Sum shall forthwith be increased by the amount of the proceeds so paid over by or for the account of the Grantee plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties; provided, however, that, if, at the time of such payment or restitution, the entire Primary Sum shall have been liquidated, the Primary Sum shall be deemed not to have been so liquidated and the unliquidated balance of the Primary Sum shall thereupon be deemed to be in the amount of the proceeds so paid over plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties.

- 12. Protection to Purchasers of Production. No person purchasing or taking or processing any Subject Minerals shall be required to take notice of, or keep informed as to, the discharge or termination of the Production Payment until the actual receipt by such person of written notice advising such person of such discharge or termination.
- 13. Certain Payments by Grantors. The Grantors will tender and pay, or cause to be paid, punctually, before they shall become delinquent (or as to any thereof which are being contested in good faith, promptly after the final determination of such contest), the following:
 - (a) all ad valorem taxes (or taxes imposed in lieu thereof), all gross production, severance, excise and sales taxes, the excise tax imposed on any windfall profit from taxal a crude oil by the Crude Oil Windfall Profit Tay Act of 1980 and any amenuments or supplements thereto or tax enacted in substitution therefor, and any similar tax imposed or assessed with respect to or measured by or charged against or attributable to the Production Payment or the Production Payment Minerals, or the proceeds thereof, or against the Grantee by reason of its ownership of the Production Payment or against any assignce or mortgagee of the Production Payment by reason of his or its interest as assignee or mortgagee, together with any interest or penalty payable in connection therewith, but any other tax, including, but not limited to, any tax imposed on net income or the right to do business shall be payable only to the extent provided in subparagraph (b) below:
 - (b) all taxes or other governmental charges of any kind or character whatsoever imposed on the Grantee (but not an assignee of Grantee) by any jurisdiction, including franchise and income taxes, other than income taxes imposed on (i) the amount received pursuant to subparagraph (c) of Paragraph 4 hereof, and (ii) any gain attributable to the sale or other disposition, in whole or in part, of the Production Payment to the extent such gain is attrib-

utable to consideration received by the Grantee in excess of the unliquidated balance of the Primary Sum allocable to the portion of the Production Payment so sold or disposed of; and

(c) obligations of the character referred to in subparagraph (f) of Paragraph 4 hereof.

Grantors shall have the right at their expense to contest any tax in good faith. The taxes which Grantors have agreed to pay or cause to be paid in subparagraphs (a) and (b) above are herein called "Subject Taxes".

- 14. Covenants of the Grantors. Until such time as the Production Payment has terminated as provided in Paragraph 5 of this Conveyance, the Grantors will, at their expense and regardless of who may be the operator or operators of the Subject Interests:
 - (a) subject to the provisions of subparagraph (c) of this Paragraph 14 and the provisions of Paragraph 21 of this Conveyance, cause the Subject Interests to be maintained and developed and continuously operated for the production of Minerals in a good and workmanlike marner and in accordance with sound field or mining practices and all applicable federal, state and local laws, rules and regulations (except those being contested in good faith), all such operations to be carried on, to the extent reasonably practicable, without any preference being given to Minerals from sources other than the Subject Interests because of the fact that Grantors have created and conveyed the Production Payment; provided, however, that nothing contained in this subparagraph (a) shall be deemed to prevent or restrict any Grantor from electing not to participate in any operation which is to be conducted under the terms of any operating agreement, unit operating agreement, contract for development or similar instrument affecting or pertaining to the Subject Interests (or any portion thereof) and allowing consenting parties to conduct nonconsent operations thereon, if such election is made by such Grantor in good faith and in conformity with sound field or mining practices:
 - (b) jointly and severally assume liability for, and hereby agree to indemnify, protect, defend, save and keep harmless the

Grantee, and any lender to the Grantee and their representatives. and any successors and assigns from and against any and all royalties and all liabilities, obligations, losses, damages, penalties, fines, claims, actions, suits, costs, expenses and disbursements (including, without limitation, fees and disbursements of counsel and other legal and investigative fees and expenses) of whatsoever kind and nature, which may at any time be imposed on, be incurred by or on behalf of or be asserted against the Grantee or any such lender and in any way related to or arising out of (i) the ownership, producing, handling, storing, transporting, refining, controlling or other disposition of any Subject Minerals or any Production Payment Minerals (including, without limitation, the leakage or spillage of, or other pollution by, any Subject Minerals or Production Payment Minerals, or any tort claim or claim for damages of any kind or nature and any claim or liability in respect of any adverse environmental impact or effect) or (ii) any investigation, litigation or proceeding related to any acquisition or proposed acquisition of all or any portion of the stock or substantially all of the assets of any Person directly or indirectly with any of the proceeds of the Purchase Price, whether or not the Grantee or any such lender is a party to such acquisition. The obligations of the Grantors under this subparagraph (b) shall survive the termination of the Production Payment. All payments required to be made pursuant to clause (ii) of this subparagraph (b) shall be made directly to the Grantee or such lender at Citibank New York, upon written demand by the Grantee or such lender. No provision of this subparagraph (b) or any other provision of this Conveyance shall be construed to be a waiver on the part of the Grantee or such lender of the benefit of any limitation of, or exemption from, liability accorded by any statute or rule of law:

(c) cause all machinery, equipment and facilities of any kind (including production platforms) now or hereafter located on the Subject Interests (or any portion thereof) and necessary to the effective operation thereof for the production of Minerals therefrom to be provided and to be kept in good and effective operating condition, and all repairs, renewals, replacements, additions and improvements thereof or thereto needed for such purpose to be duly made or provided for; provided, however, that the Grantors

shall not be required to repair, replace or reb such machinery, equipment or facilities as to Subject from which Minerals are produced (i) when such Subject to ests are jointly owned and are operated by a party other than any Grantor or an Affiliate of any Grantor and the operator decides not to repair. replace or rebuild such equipment, (ii) when failure to do so would not materially adversely affect the rate of production of Minerals. or (iii) if, as to any production platform located on the Subject Interests (or any portion thereof), including without limitation. equipment, machinery and facilities of any kird thereon, or any well or wells located on any such production platform, the Grantors shall pay over to the Grantee, for application to the discharge of the Production Payment, (x) quarterly, on or before 11:00 A.M. tlocal time in effect in The City of New York) on each Quarterly Application Date (herein in this Paragraph 14(c) called a "relevant Quarterly Application Date") commencing with the second such Quarterly Application Date next following the date of loss of or damage to the relevant platform, that amount which, when added to all other amounts received by the Grantee pursuant to the provisions of Paragraph 6 hereof during the quarterly period commencing with (but not including) the last preceding Quarterly Application Date and ending with and including the relevant Quarterly Application Date, will cause the Grantee to receive on or before the relevant Quarterly Application Date a sum equal to (A) the aggregate of the amounts to be applied in accordance with subparagraphs (a), (b), (c), (d) and (e) of Paragraph 6 hereof during said quarterly period, plus, after the commencement of the Second Production Period, (B) an amount to be applied in accordance with subparagraph (f) of said Paragraph 6 sufficient to reduce the unliquidated balance of the Primary Sum on or before the relevant Quarterly Application Date to an amount equal to (1) the unliquidated balance of the Primary Sum on the last preceding Quarterly Application Date (after the application made on that date) less (2) an amount determined by multiplying the Primary Sum at the close of business on July 31, 1983 by the percentage shown on the Discharge Schedule under the heading "Hydrocarbons Monthly Percentage" or, in the event the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, under the heading "Minerals Monthly Percentage", for each of the Application Dates in said quarterly period, and aggregating the products of such calculations (the aggregate of all such quarterly payments made under this clause (x) in respect of such platform (including all wells located thereon) not to exceed the aggregate fair market value at the wellhead, as reasonably determined by the Grantors with the concurrence of the Grantee, or, if the Grantee shall not concur with such determination by the Grantors, as determined by Ryder Scott Company Petrole: Engineers or other independent petroleum engin ers satisfac by to the Grantors and the Grantee, as of the date of loss of or damage to the relevant platform, of all Subject Minerals that would have been produced from wells previously completed on such platform had the necessary repairs, replacements or rebuilding of such production platforn, and any neces sary redrilling or reworking of such wells, been carried out), or, at the sole option of the Grantors, (v) within thirty (30) days after the first to occur of (A) the receipt by the Grantors of the proceeds of any insurance recovered by the Grantors on account of loss of or damage to such r'atform or (B) the expiration of one (1) year from the date of such loss or damage, an amount equal to the aggregate fair market value at the wellhead of the Subject Minerals referred to, and determined as provided in, clause (x) above, and the amount of any such payments to the Grantee shall be deemed to be proceeds of Production Payment Minerals for all purposes of this Conveyance and shall be included in the written statement to be furnished to the Grantors by the Grantee pursuant to the last paragraph of Paragraph 6 Fereof, but may be reflected as a payment received from the Grantors pursuant to this subparagraph(e):

- (d) cause all necessary and proper steps to be taken disigntly to protect and defend the Subject Interests, the Subject Minerals and the proceeds thereof against any adveragalam or demand, including, without limitation, the employment or use of counsel for the prosecution or defense of litigation and the contest, settlement, release or discharge of any such claim or demand;
- (e) cause written notice to be given to the Grantee of every adverse claim or demand made by any Person materially affecting the Subject Interests, the Subject Minerals or the proceeds there-

of in any manner whatsoever, and of any proceeding instituted in respect thereof;

- (f) cause the interest of the Grantors in each of the Subject Interests to be kept free and clear of liens, charges and encumbrances of every character, other than those which are of a character customarily found in connection with comparable drilling, mining and producing operations and which do not meterially adversely affect the operation of the Subject Interests and other than those becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Conveyence;
- (g) subject to the provisions of Paragraph 21 of this Conveyance, cause all Leases included in or relating to the interests described or referred to in Exhibit A and added which the Grantor is lessee, sub-lessee or assigned to be maintained in full force and effect without amendment, no lifectic or assiver of any provision of any such Lease which might materially adversely affect the rights of the Grantee under the Conveyance;
- (h) (i) maintain, or cause to be muchtained, insurance insuring the Grantors, to the extent of their respective interests therein, against loss of or damage to any and all machinery, equipment and facilities of any kind (including all production platforms) now or hereafter located on the Subject Interests (or any portion thereof) in amounts equal to the full insurable value of the Grantors' interest (as if this Conveyance had not been executed and delivered) in such machinery, equipment and facilities and covering loss or damage due to fire and windstorms, hurricanes and such other events, including blowout and cratering, usually covered by insurance carried by companies engaging in similar business and owning similar properties in the same general area or areas in which the relevant Subject Interests are located, and (ii) in addition to the insurance coverage required under clause (i) above, maintain, or cause to be maintained, insurance in such amounts and covering such risks as is usually carried by companies engaged in similar business and owning similar properties in the same general area or areas in which the relevant Subject Interests are located and as may be required by applicable law, which coverage shall in any event include the following:

- (A) general public liability and property damage insurance, endorsed to include offshore operations, insuring the Grantors against liability for injury to persons or property as a result of or growing out of the ownership, operation or development of the Subject Interests (or any portion thereof) or the production or transportation of Minerals therefrom, with a limit of liability of not less than \$500,000 for each occurrence;
- (B) automobile public liability and property damage insurance, insuring the Grantor against liability for injury to persons or property growing out of the ownership or operation of automobiles (including leased automobiles) in connection with the activities described in clause (A) above, with a limit of liability of not less than \$500,000 for each occurrence;
- (C) aviation public liability and property damage insurance, insuring the Grantors against liability for injury to persons or property growing out of the ownership or operation of aircraft (including leased aircraft) in connection with the activities described in clause (A) above, with a limit of liability of not less than \$10,000,000 for each occurrence; and
- (D) marine public liability and property insurance, insuring the Grantors against liability for injury to persons or property growing out of the ownership or operation of barges or vessels (including chartered barges or vessels) in connection with the activities described in clause (A) above, with a limit of liability of not less than \$500,000 for each occurrence.

All insurance required by this subparagraph (h) shall be maintained with responsible and reputable insurance companies or associations.

(i) In the event the Supplemental Instrument of Conveyance of Production Payment is executed and delivered, cause all contracts and rights (as in effect on the date hereof), with respect to the gas and water supply used in connection with the mining of the Subject Sulphur, to be maintained in full force and effect without any amendment, modification or waiver which might materially adversely affect the rights of the Grantee.

Notwithstanding the provisions of subparagraphs (a) and (g) of this Paragraph 14, the Grantors shall have the right to abandon any Subject Interest or any portion thereof from which Minerals have been produced and to discontinue performance of their obligations under said subparagraphs with respect to such Subject Interest or such portion thereof when (A) with respect to Subject Interests in Hydrocarbons there is no well located on such Subject Interest or such portion thereof which is capable of producing Hydrocarbons in paying quantities and when the Grantors have reasonably determined (i) that no existing well may be recompleted, and no additional well may be completed, thereon which would be capable of producing Hydrocarbons in paying quantities, and (ii) that no pressure maintenance or other recovery techniques or methods then currently in use by prudent operators may be instituted with respect to any existing well which would result in the production from such well of Hydrocarbons in paying quantities (for purposes of this sentence, any Subject Interest or any portion thereof which is committed to a unit or a joint operation and which is thereafter permanently abandoned under the pplicable unit or operating agreement prior to termination of the nit or joint operation, shall be deemed an abandonment of such Subject Interest or such portion, as the case may be, by the Grantors which is permitted); and (B) with respect to Subject Interests in Sulphur, there is no mine located on such Subject Interest or such portion thereof which is capable of producing Culphur in paying quantities.

For all purposes of this Paragraph 14, an existing well or mine located on any Subject Interest shall be deemed to be capable of producing Minerals "in paying quantities" unless a condition not of a temporary nature shall exist whereby the aggregate value of the Minerals produced from such well or mine, as the case may be, is less than the direct costs of operating such well or mine and preparing the production thereof for marketing. Further, a well on any Subject Interest in Hydrocarbons which may result from additional drilling or which may be the subject of recompletion, pressure maintenance or other recovery techniques or methods shall not be deemed to be capable of producing Hydrocarbons "in paying quantities" if, in the light of conditions existing at the time of the determination and which are not of a temporary nature, it is estimated that the aggregate value of the Hydrocarbons to be produced from such well shall not exceed the then

estimated direct costs of operating such well and preparing the production therefrom for marketing, as well as the direct costs of providing the necessary installations for, and carrying out, any pressure maintenance or other recovery techniques or methods properly allocable to such well and the direct costs of recompleting such existing well or drilling, completing and equipping such additional well, in each case to the extent such direct costs are allocable to such Subject Interest.

As used herein, "direct costs" shall mean and include only (i) the cost of labor, transportation, other services, materials, equipmer* and supplies, (ii) amounts paid or to be paid independent contractors and (iii) amounts charged under applicable operating or unit agreements for or in lieu of general or administrative overhead or, with respect to those Subject Interests which are not subject to operating agreements, a reasonable amount for general and administrative overhead (which latter amount shall not in any event exceed 5% of the costs and amounts referred to in clauses (i) and (ii) above). Payments on account of the Production Payment or on account of royalties, overriding royalties or any other payments out of production shall not be deemed to constitute "direct costs".

Anything contained in this Paragraph 14 to the contrary notwithstanding, the Grantors shall not be obligated to perform undertakings performable only by the operator and which are beyond the control of the Grantors with respect to individual Subject Interests of which none of them is the operator. In such case, however, the Grantors will use their best efforts to secure the performance of each of such undertakings.

15. Access to the Subject Interests. The Grantors shall permit any one or more representatives designated by the Grantee to make any inspection of the Subject Interests at any reasonable time as the Grantee may reasonably request. In addition, the Grantee shall furnish to the Grantee such detailed information as the Grantee may reasonably request in writing concerning the Subject Interests, the operation thereof and the production and sale of the Subject Minerals therefrom, provided that the Grantee shall conform to such confidentiality and secrecy requirements respecting such information as the Grantors shall in writing reasonably request.

- 16. Pooling and Unitization. An him herein to t conv notwithstanding, the Grantors may, without the consent e and Grantee (unless any party having an interest in the proposed unit is an Affiliate of any Grantor holding an interest in the unit not attributable to a Subject Interest, in which case the man rifter, consent of the Grantee shall be required to be obtained (e. pt as to a unit required by governmental authority, for which no consent shall be required), which consent shall not be unreas, ably withheld), pool, combine, communitize or unitize any Subject Interest (or any portion chereof) from time to time with other leasehold, mineral or other interests to form units, or to reform, expand or modify units theretofore formed. when, in the judgment of the Grantors, it is necessary or advisable to do so in order to facilitate the orderly development or operation (including repressuring, pressure maintenance, cycling, water flood and secondary recovery operations) of any Subject Interest (or any portion thereof). The interest in any such unit accruing or attributable to any Subject Interest (or any portion thereof) included therein shall be subject to the Production Payment in the same manner and with the same effect as though such unit were specifically described in Exhibit A. The signature or joinder of the Grantee shall not be necessary to the instruments evidencing the pooling and unitization herein authorized.
- 17. Default by the Grantors and the Remedies of the Grantee. Any of the following events shall be an Event of Default (each such event being herein called an "Event of Default") under this Conveyance:
 - (a) any representation or warranty contained in Paragraph 20 of this Conveyance or in any certificate or statement delivered by the Grantors (or any of their officers) pursuant to subparagraph (b) of Paragraph 7 of this Conveyance shall be untrue when made in any respect which adversely affects the rate of discharge of the Production Payment or otherwise materially adversely affects the interests of the Grantee hereunder; or
 - (b) any Grantor shall default in the performance or observance of any term, covenant or condition on such Grantor's part to be performed or observed hereunder which default adversely affects the rate of discharge of the Production Payment or

otherwise adversely affects the interests of the Grantee hereunder, and such default shall continue unremedied for a period of 30 days after written notice shall have been given to such Grantor by the Grantee's demanding the performance or observance of the relevant term, covenant or condition (or, if such default is capable of being remedied by such Grantor but cannot be remedied within such 30-day period, such Grantor shall not have commenced to remedy such default within such period after receipt of such notice and thereafter shall not continuously and diligently pursue the remedying of the same until remedied); or

- (c) any Grantor shall become insolvent, or be adjudicated a bankrupt, or make a general assignment for the benefit of its creditors, or shall generally not pay its debts as such debts become due, or admit in writing its inability to pay its debts generally as they mature, or voluntarily commence any proceeding or file any petition under any bankruptey, insolvency or similar law or seeking dissolution or reorganization or the appointment of a receiver, trustee or liquidator for it or for a substantial portion of its assets, or file any answer admitting the jurisdiction of the court and the material allegations of an involuntary petition filed against it under any bankruptey, insolvency or similar law; or
- (d) an involuntary proceeding shall be commenced or an involuntary petition shall be filed against any Grantor under any bankruptcy, insolveney or similar law or seeking the dissolution or reorganization of such Granter or the appointment of a receiver, trustee or liquidator for such Grantor or a substantial part of the property of such Grantor and such involuntary proceeding or involuntary petition shall not be dismissed within 60 days after the commencement or filing thereof, as the case may be.

Upon the occurrence and during the continuance of any Event of Default, the Grantee may, in addition to pursuing any or all other remedies available at law or in equity, do any of the following:

(i) effect performance or observance of any such term, covenant or condition referred to in subparagraph (b) above on behalf of and at the expense of the Grantors, in which event the Grantee may expend funds reasonably necessary to effect such purposes

and shall be entitled to reimbucement therefor from the Grantors, together with interest on such amounts at a rate equal to the lesser of (x) the maximum rate of interest that may be lawfully charged on such amounts under applicable law or (y) the Applicable Rate in effect during the period from and including the date of such expenditure to the date of reimbursement; or

- (ii) upon written notice to the Grantors, succeed to all rights of the Grantors with respect to the possession, operation and development of the Subject Interests, and may use in connection therewith all of the appurtenant property, both real and personal, situated upon the Subject Interests (or any portion thereof) or used or held for future use in connection therewith, or in connection with the production, storing, transporting or sale of the Subject Minerals, and the Grantee shall have the right, on behalf of and for the account of the Grantors, to sell the Subject Minerals, and the Grantors shall, upon demand, reimburse the Grantee for all amounts so expended by the Grantee, together with interest on such amounts at a rate equal to the Applicable Rate in effect during the period from and including the date of such expenditure to the date of reimbursement; or
- (iii) apply to a court of equity for the specific performance or observance of any, set term, covenant or condition referred to in subparagraph (b) above or in aid of the execution of any power herein granted and for the appointment of a receiver of the Subject Interests and the Subject Minerals.
- 18. Exercise and Termination of the Remedies of the Grantee. All rights and remedies granted by the provisions of Paragraph 17 of this Conveyance and otherwise belonging to the Grantee shall be cumulative, and no exercise of any right or remedy shall preclude the exercise of any other right or remedy. Any right or remedy may be exercised on one or more occasions without exhaustion thereof. All such rights and remedies shall terminate:
 - (a) when the Production Payment is discharged and all amounts then due and payable to the Grantee pursuant to the terms of Paragraph 17 of this Conveyance (including amounts payable on account of interest) shall have been paid in full, or

- (b) at any earlier time when all defaults of the Grantors shall have been remedied and all such amounts shall have been duly paid in full, without prejudice, however, to the exercise of any rights conferred upon the Grantee upon any subsequent failure of the Grantors to perform or observe any of the terms, covenants or conditions herein provided to be performed or observed by the Grantors.
- 19. Reports to the Grantee. While the Production Payment remains in force and effect, the Grantors will, at their expense, furnish to the Grantee:
 - (a) Monthly, on or before the 25th day of each Accounting Month, or if such information is not available to the Grantors by such date, within five days after the required information is received by the Grantors, a report, in form and substance satisfactory to the Grantee, showing (i) the gross quantity of oil, condensate, plant products and gas and, if the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, of sulphur, included in the Subject Minerals sold by the Grantors, or taken by the Grantors for their own use, during the preceding Accounting Month, (ii) the net quantity of oil, condensate, plant products and gas (after deduction of oil and gas used in connection with the operation of the Subject Interests) included in the Subject Minerals produced and seld by the Grantors, or taken by the Grantors for their own use, during such preceding Accounting Month, (iii) the gross proceeds received from such sale, or attributable to such taking, of Subject Minerals (to the extent of the Grantors' net revenue interest therein) during such preceding Accounting Month, separately stated as to oil, condensate, plant products, gas and, if the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, as to sulphur, (iv) the amount of all costs incurred by the Grantors in transporting Subject Minerals so sold to the point of sale if sale is not made at the wellhead or mine, (v) the amount of all taxes and assessments attributable to the production and sale of such Subject Minerals, (vi) that portion of the foregoing quantities, proceeds, transportation costs, taxes and assessments allocable to the Production Payment Minerals and (vii) all

direct costs and operating costs incurred by the Grantors in producing and processing Subject Minerals.

- (b) Annually, within 90 days after the end of each fiscal year of the Grantors, a reserve report of Ryder Scott Company Petroleum Engineers (or other independent petroleum engineers satisfactory to the Grantee and the lenders to the Grantee), substantially in the form and containing the same information (as of the end of such fiscal year) and using the same definitions and assumptions with respect to the Hydrocarbons recoverable from proved reserves attributable to the Subject Interests, as the letter report of Ryder Scott, dated June 29, 1981, delivered to the Grantee and the lenders to the Grantee prior to the date hereof.
- (c) Such other information and reports (including financial statements of the Grantors and or reports in respect of sulphur) as the Grantee may from time to time reasonably request in writing.
- 20. Representations, Warranties and Covenants. (a) The Grantors jointly and severally represent and warrant as follows:
 - (i) the Grantors have good and marketable title to the Subject Interests, which are presently valid, subsisting and in full force and effect, subject only to this Conveyance and the exceptions, restrictions, reservations, liens, charges, encumbrances and other matters, if any, as are set forth in Exhibit A relative thereto, and such Sales Contracts, operating agreements, unitization and pooling agreements and other agreements as are customarily found in connection with comparable drilling, mining and producing operations, none of which, singly or in the aggregate, materially adversely affects the Production Payment or the Production Payment Minerals or the proceeds thereof, or the rights of the Grantee therein or adversely affects the rate of discharge of the Production Payment;
 - (ii) the Grantors' interest in production of Hydrocarbons produced from each Subject Interest specifically described in Exhibit A hereto (after deducting all applicable royalties, overriding royalties, production payments and other payments out of

production not owned by the Grantors and before giving effect to the Production Payment hereby conveyed to the Grantee) is at the date hereof no less than that interest shown under the heading "Net Revenue Interest" opposite the description of each such Subject Interest;

- (iii) the Grantors have the legal right, power and authority to create and confer upon the Grantee the rights of the Grantee granted by the terms of this Conveyance;
- (iv) the Production Payment is, and at the time of proper recording of this Conveyance in each place as may be required by law in order fully to preserve and protect the interest of the Grantee in the Production Payment will be, free and clear of all liens, charges and encumbrances of every character, except the liens, charges and encumbrances permitted by Paragraph 14(f) hereof and except such liens, charges and encumbrances as may be vested in persons claiming the Production Payment or any interest therein by, through or under the Grantee;
- (v) all taxes and assessments of any kind whatsoever (other than those contested by the Grantor in good faith) imposed or assessed with respect to or measured by or charged against or attributable to the Subject Interests have been duly paid or provided for;
- (vi) there are no suits or proceedings pending or, to the knowledge of the Grantors, threatened against or affecting the Subject Interests, or any part thereof, or the sale of Subject Minerals produced therefrom, before any court or by or before any governmental commission, bureau or any regulatory authority, which if decided adversely to the interests of the Grantors would materially adversely affect the Production Payment, or the Production Payment Minerals or the proceeds thereof or rights of the Grantee the ein or would adversely affect the rate of discharge of the Production Payment;
- (vii) to Grantors' knowledge, all rents and royalties that have become due prior to the Effective Date with respect to the Subject Interests or production therefrom, and all liabilities of any kind or nature incurred with respect to the Subject Interests prior

to such date, have been paid or have been provided for, and no Grantor has received any notice of default or claimed default with respect to the Subject Interests or any part thereof or any interest in production therefrom; and

- (viii) all prices received by Grantors under presently existing Sales Contracts are collected in accordance with all existing rules, regulations, and orders of the Federal Energy Regulatory Commission, or its predecessor, establishing maximum lawful prices for the sale of natural gas.
- (b) The Grantors hereby jointly and severally bind themselves, their successors and assigns, to warrant and forever defend their title to the Subject Interests and the Grantee's title to the Production Payment against the lawful claims and demands of every person who shall claim the same or any part thereof, and the Grantors will indemnify and hold the Grantee harmless from and against all loss suffered by the Grantee resulting from any failure of or defect in the title to the Subject Interests or the Production Payment as warranted above in this Paragra in 20. This Conveyance is made with full substitution and subrogation of the Grantee in and to all covenants and warranties by others heretofore given or made in respect of the Subject Interests or any part thereof.
- (c) The Grantors jointly and severally represent and warrant as follows:
 - (i) Each of the Grantors is a corporation duly organized and validly existing in good standing under the laws of the State of Delaware, is duly qualified and in good standing as a foreign corporation in each State in which such Grantor owns Subject Interests and has the legal right, power and authority, corporate or otherwise, to conduct its business and to own its properties (including the Subject Interests) and to convey to the Grantee the Production Payment and all of the rights, titles, interests, estates, remedies, powers and privileges appurtenant or incident thereto in the manner hereby contemplated, and to execute and deliver, and to perform all of its obligations under, this Conveyance.
 - (ii) The making and performance by the Grantors of this Conveyance, and the conveying by the Grantors to the Grantee of

the Production Payment, have been daly authorized by all necessary corporate action (including any si archolder approval) the part of each Grantor and do not and will not (A) viol any provision of any law, rule, regulation, order, writ, judgment, decree or determination presently in effect having applicability to any Grantor or of the Certificate of Incorp. ation or By-laws of any Grantor, or (B) result in a breac's of or constitute a default under any indenture or bank loan or credit agreement or any other agreement or instrument to waich any Grantor is a party or by which it or its properties may be presently bound or affected or (C) result in or require the creation or imposition of any mortgage, lien, pledge, security interest, charge or other encumbrance on, upon or of any of the properties or assets of any Grantor (including the Subject Interests) under any such indenture, bank loan or credit agreement or other agreement or instrument; and no Grantor is in default under any such order, writ, judgment, decree, determination, indenture, agreement or instrument in any way which materially adversely affects, or may materially adversely affect, the ability of the Grantors to perform their obligations under this Conveyance; and all consents or approvals under such indentures, agreements and instruments necessary to permit the valid execution, delivery and performance by the Grantors of this Conveyance and the conveying of the Production Payment by the Grantors to the Grantee have been obtained.

- (iii) Assuming the due authorization and execution thereof on the part of the Grantee, this Conveyance constitutes the legal, valid and binding obligation of the Grantors enforceable against the Grantors in accordance with its terms, except as limited by bankruptcy, insolvency or other similar laws of ge application relating to or affecting the enforcement of cred: rights.
- (iv) All consents and waivers necessary to permit the valid conveying of the Production Payment to the Grantee have been obtained or the time for giving such consents or waivers has expired following a written request therefor given by the Grantor.
- (v) No authorization, consent, approval, license or exemption of, and no filing or registration with, any court or governmental

department, commission, board, bureau, agency or instrumentality, domestic or foreign, is or was necessary to the valid execution and delivery by the Grantors of this Conveyance or to the conveying by the Grantors to the Grantee of the Production Payment or to the performance by the Grantors of their obligations under this Conveyance (other than, in the State of Louisiana, with regard to leases from the State of Louisiana or political subdivisions thereof, certain governmental approvals which are not obtainable prior to the execution and delivery hereof and which are normally routine and will be obtained in due course) and in the case of the performance by the Grantors of such obligations, such authorizations, consents, approvals, I'censes or exemptions which are routine and either have been obtained or will be obtained in due course.

- (vi) No fire, explosion, accident, earthquake, act of the public enemy or other casualty (whether or not covered by insurance) materially adversely affecting any material portion of the Subject Interests or the operation thereof, or materially adversely affecting the ability of the Grantors to perfect their obligations under this Conveyance, has occurred.
- (vii) The offices where the Grantors keep their respective records concerning the Subject Interests and the Production Payment and the payments to be made to the Grantors in respect of production from the Subject Interests and in respect of the Production Payment are located in the State of Texas. The Grantors hereby agree that such records will be kept in such State or, after 60 days' prior written notice to the Grantee in respect thereof, in any other State or States of the United Control
- (viii) No information, exhibit, mem or report (excluding estimates contained therein) furnished writing by the Grantors to the Grantee in connection with the sale of the Production Payment or by the Grantors to Ryder Scott Company Petroleum Engineers or DeGolyer and MacNaughton in connection with their evaluation of the Subject Interests contained any material misstatement of fact or mitted to state a material fact relative to the subject matter thereof or any fact necessary to me the statements contained therein not misleading. All estimates is furnished were prepared on the basis of assumptions, data, information,

tests or conditions believed to be valid or accurate or to exist at the time such material was so furnished.

- (ix) All of the properties and interest with respect to which value was given in the financial and engineering reports submitted by Ryder Scott Company Petroleum Engineers and the Grantors to the Grantee and its lenders in connection with the negotiations for the purchase and financing of the Production Payment are Subject Interests (or, if less than all, the net effect of such difference is negligible).
- (x) Since March 31, 1981 there has not been such a significant adverse change in the financial condition of the Grantors, viewed as a whoie, as to materially impair their ability to perform the obligations of the Grantors set forth in this Conveyance.

The covenants, representations and warranties contained in this Paragraph 20 shall survive the discharge of the Production Payment.

- 21. Transfer of the Subject Interests. Except to the extent required by the terms of the relevant operating agreement, unit operating agreement, contract for development or other instrument in connection with nonconsent operations conducted with respect to any Subject Interest (or any portion thereof) as to which any Grantor's election not to participate shall have been made in conformity with the provisions of Paragraph 14(a) of this Conveyance, and except as otherwise permitted by Paragraphs 14 and 16 of this Conveyance, the Grantors shall not, prior to the termination of the Production Payment, without the prior written consent of the Grantee, which consent shall not be unreasonably withheld, make or permit (i) any surrender, abandonment, sale, conveyance, assignment, lease, sublease or farmout, in whole or in paof any of the Subject Interests insofar as they include or cover Hydrocarbon) after the e ecution and delivery of the Supplemental Instru Conveyance of Production Payment, any surrender, sale, conveyance, assignment, lease or sublease, in whole or in part, of any of the Subject Interests insofar as they include or cover sulphur.
- 22. Exception to Payments. The acceptance by the Grantee of any payment hereunder shall not preclude the Grantee from taking excep-

tion to the correctness of the amount thereof; provided, however, that any such exception must be specified by the Grange in a notice given to the Grantors within 12 months after the date such payment is made.

- 23. Transfer of the Production Payment. Nothing contained in this Conveyance shall be construed to limit or restrict in any way the right of the Grantee to sell, convey, assign or mortgage the Production Payment, in whole or in part.
- 24. Rights of Mortgagee or Trustee. If the Grantee shall at any time execute a mortgage or deed of trust covering all or part of the Production Payment, the mortgagee(s) or trustee(s) therein named or the holder of any obligation secured thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to exercise all the rights, remedies, powers and privileges of ferred upon the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained hereunder by the Grantors, but the provisions of this Paragraph 24 shall in no way be deemed or construed to impose upon the Grantors any obligation or liability undertaken by the Grantee under such mortgage or deed of trust or under the obligation secured thereby.
- 25. Grantors' Rights to Make Payments to be Applied to Discharge of Production Payment. Anything in this Conveyance to the contrary notwithstanding, the Grantors shall have the right, at any time and from time to time, upon written notice furnished to the Grantee and each of the mortgagees and trust es referred to in Paragraph 4 hereof not less than 15 days before the Application Date on which the counters exercise such right, to make payment (out of any funds of the Grentors whether or not such funds are proceeds from the Subject Minerals' to the Grantee on any Application Date, which payments shall be deer. I to be proceeds of Production Payment Minerals for purposes of the. Conveyance and shall be applied to the discharge of the Production Payment pursuant to the provisions of Paragraph 6 hereof, provided, leavever, that the Grantee has been provided with the indennity specified in Paragraph 5. Any such payment shall be in an amount sufficient to reduce the Primary Sum by \$50,000,000, or a multiple thereof (unless the full aggregate sum of the amount specified in subparagraphs (a), (b), (c), (d), (e) and (f) of Paragraph 4 hereof, after application of

the proceeds of Production Payment Minerals on such Application Date, is less than \$50,000,000, in which event such payment shall be an amount equal to such full aggregate sum), and shall be made by a deposit to the credit of Grantee's account (No. 39019753) with Citibank at 399 Park Avenue, New York, New York 10043, in lawful currency of the United States of America and in includiately available funds. No payment made by the Grantors pursuant to this Paragraph 25 shall be added to, or have the effect of increasing, the Purchase Price.

- 26. Notices. Any notice, request, demand, report or other instrument which may be required or permitted to be given or furnished to or served upon either party hereto or other Person succeeding to any interest of either party hereto shall be deemed sufficiently given or furnished and served if in writing and delivered to such party or Person or to an officer of a chiparty or Person or deposited in the United States mail in a scaled envelope, registered, with postage prepaid, addressed if to the Grantors to Pennzoil Company, Pennzoil Place, P.O. Box 2967, Houston, Texas 77002, Attention: Harold E. Sorter, and if to the Grantee to 100 West Tenth Street, Wilmington, Delaware 19801, Attention: Joseph A. Barbera, or to such other address as the party or Person to be addressed shall have designated by written notice to the party or Person giving such notice or furnishing such report or making such request or demand.
- 27. Further Assurances. So long as permitted by applicable law so to do, the Grantors will execute and deliver all such other and additional instruments, notices, releases, acquittances and other documents, and will do all such other acts and things, as may be necessary more fully to assure to the Grantee all the rights and interests herein and hereby granted or intended so to be.
- 28. Binding Extect. All the overants and agreements of the Grantors contained in this Conveyance shall be deemed to be covenants running with the land and the Subject Exterests and shall be binding upon all successors in interest to, and all assigns of, the Grantors and shall inure to the benefit of all successors in interest to, and all assigns of, the Grantee. All references herein to either the Grantors or the Grantee shall include their respective successors and assigns, except that the term "Grantee" as used in this Conveyance shall not

include any collection agent of any mortgagee or assignee for security purposes of the Grantee appointed by such mortgagee or assignee without the consent of the Grantors.

- 29. Release or Failure of Title. No release by the Grantee of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect of reducing or increasing the amount of production of Subject Minerals out of which the Production Payment is dischargeable or reducing the amount of the Production Payment or creating any offset or other prejudice to the Production Payment, and the Production Payment shall continue in full force and effect as to the remainder of the Subject Interests until the full amount of the Production Payment has been received by the Grantee as herein provided.
- 30. Governing Law. The velidity, effect and construction of this Conveyance insofar as it relates to any conveyance or other disposition affecting any Subject Interest, any warranties of title relating thereto and the remedies herein provided for or provided for by law with respect to such Subject Interest shall be governed by the laws of the jurisdiction in which such the ject Interest is located. With respect to all other matters (including, without limitation, matters of construction, validity and performance other than as set forth above in this Paragraph 30), this Conveyance shall be governed by and construed in accordance with the laws of the State of Texas.
- 31. Final Termination. Notwithstanding the other provisions hereof, the Production Payment shall, in any event, terminate one day prior to the expiration of 21 years after the death of the survivor of all the descendants of Joseph P. Kennedy, father of the late President of the United States, who are living on the Effective Date.
- 32. Headings. Paragraph headings used in this Conveyance are for convenience only and shall not affect the construction of this Conveyance.

IN WIGNESS WHEREOF, the Grantors and the Grantee have executed this instrument in the presence of the undersigned witnesses on the date set forth in the acknowledgments annexed hereto, in several counterparts (one of which with all of the property descriptions included in Exhibit A is on file at the office of the Grantee referred to in Paragraph 26 hereof, one of which with all said property descriptions is to be recorded in Terrebonne Parish, Louisiana, and one of which with all said property descriptions is to be recorded in Culberson County, Texas), each of which is an original and all of which are identical, except that, to facilitate recordation, there are omitted from certain counterparts those property descriptions in Exhibit A which contain descriptions of property located in recording jurisdictions other than the jurisdiction (County or Parish) in which the particular counterpart is to be recorded. Each of the counterparts of this Conveyance so executed shall for all purposes be deemed to be an original, and all such counterparts together shall constitute but one and the same Conveyance.

PENNZOIL COMPANY
PENNZOIL PRODUCING COMPANY
PENNZOIL OIL & GAS, INC.

DUVAL CORPORATION

Group Vice President,
Pennzoil Company

[CORPORATE SEAL]

Vice President, Pennzoil Producing Company

[COBPORATE SEAL]

Vice President, Pennzoil Oil & Gas, Inc.

[CORPORATE SEAL]

Vice President, Duval Corporation

[CORPORATE SEAL]

| A | TT | 6.0 | 6.0 |
|---|----|-----|-----|

Sally Hazen, Secretary, Pennzoil Company Secretary, Pennzoil Producing Company Secretary, Pennzoil Oil & Gas, Inc. Assistant Secretary, Duval Corporation

Grantors

The address of Pennzoil Company, Pennzoil Producing Company, Pernzoil Oil & Gas, Inc. and Duval Corporation is:

> c/o Pennzoii Company Pennzoil Place P.O. Box 2967 Houston, Texas 77002

NEW DUNSTAN, INC.

Grantee

The address of New Dunstan, Inc. is:

> 100 West Tenth Street Wilmington, Delaware 19801

[CORPORATE SEAL]

Attest:

Andrew M. Baker. Assistant Secretary

Executed by all parties in the presence of the undersigned

Conne Daroley

Cone Elling S

Witness

STATE OF NEW YORK
COUNTY OF NEW YORK

BE IT REMEMBERED, that I, Douglas A, Stuart, a Notary Public duly qualified, commissioned, sworn and acting in and for the County and State aforesaid, levely certify that on this 16th day of July, 1981:

There appeared before me severally each of the following persons, each residing at the address set forth below his name, and each the designated officer of the corporations set opposite his name, each such corporation being a corporation of the state indicated and being a party to the foregoing instrument:

Harold E. Sortor 910 Oak Valley Houston, Texas 77024 Group Vice President of Pennzoil Company, a Delaware corporation.

Vice President of Pennzoil Producing Company, a Delaware corporation.

Vice President of Pennzoil Oil & Gas, Inc., a Delaware corporation.

Vice President of Duval Corporation, a Delaware corporation.

Joseph Nalle 3024 Chevy Chase Houston Texas Vice President of New Dunstan, Inc., a Delaware corporation. (Alabama)

Before me personally appeared each such person, whose names are signed to the foregoing conveyance, and who are known to me, and acknowledged before me on this day that, being informed of the contents of the conveyance, each, as such officer of each of the corporations set opposite his name and with full authority, executed the same voluntarily for and as the act of said corporation.

(Arkansas)

Before me appeared each such person, to me personally well known, and each stated that he is the designated officer of each of the corporations set opposite his name, and he is duly authorized in his capacity to execute the foregoing instrument for and in the name of said corporation, and further stated and acknowledged that he had so signed, executed and delivered said foregoing in trument for the consideration, uses and purposes therein mentioned and set forth.

(Louisiana)

Before me appeared each such person, to me personally known, who being by me duly sworn, did say that he is the designated officer of each of the corporations set opposite his name, and that the seal of such corporation affixed to the foregoing instrument is the corporate seal of the corporation and that the instrument was signed and sealed in behalf of the corporation by authority of the Board of Directors of such corporation and that each acknowledged the instrument to be the free act and deed of the corporation.

(Mississippi)

Before me personally appeared each such person and acknowledged that he signed and delivered the foregoing instrument on the day and year therein mentioned as the act and deed of each of the corporations set opposite his name, having first been authorized to do so.

(Montana)

Before me personally appeared each such person, known to me to be the designated officer of each of the corporations set opposite his name that executed the within instrument and acknowledged to me that such corporation executed the same.

(New Mexico)

Before the think date appeared each such person, to me personally know the design of whom, being by me doly swern, did say that he is the design of each of the corporations set opposite his name, fixed to said instrument was signed and sealed in corporation by authority of its Board of Directors, and each to a "knowledged said instrument to be the free act and the triangle of the corporation by authority of its Board of Directors, and each to a "knowledged said instrument to be the free act and the triangle of the corporation by authority of its Board of Directors, and each to be the free act and the triangle of the corporation by authority of its Board of Directors, and each to be the free act and the triangle of the corporations are to opposite his name, fixed to said instrument was signed and sealed in the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of its Board of Directors, and the triangle of the corporation by authority of the corporation by authority of the corporation by author

(New York)

Before me personally came each such person, to me personally known, and each, being by me duly sworn, did depose and say that he resides at the address set forth below his name and is the designated officer of each of the corporations set opposite his name and that said corporation is described in and executed the above instrument; that he knows the seal of the corporation; that the seal affixed to such instrument is such corporate seal; that it was so affixed by order of the Board of Directors of the corporation; and that he signed his name thereto by like order and that the instrument was the free act and deed of the corporation.

(North Palota)

Before me on this date personally appeared each such person, known to me to be the designated officer of each of the corporations set opposite his name, which corporation is described in and executed the within instrument, and each acknowledged to me that such corporation executed the same.

(Pennsylvania)

Before me personally appeared each such person and acknowledged himself to be the designated efficer of each of the corporations set opposite his name, and that he as such officer, being authorized so to do, executed the foregoing instrument for the purposes therein contained, by signing the name of the corporation by himself as such officer.

(Texas)

Before me on this day personally appeared each such person, each of whom is known to me to be the person whose name is subscribed to the foregoing instrument, and known to me to be the designated officer of each of the corporations set opposite his name, and each acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, in the capacity therein stated and as the act and deed of said corporation.

(West Virginia)

Each such person has this day in my said County, before me, acknowledged the said instrument to be the act and deed of each of the corporations set opposite his name.

(Wyoming)

Before me on this day appeared each such person to me personally known, each of whom, being by me duly sworn, did say that he is the designated officer of each of the corporations set opposite his name, and that the seal affixed to said ninstrument is the corporate seal of said corporation and that said instrument was signed and sealed in behalf of said corporation by authority of its Board of Directors, and each such person acknowledged it to be the free act and deed of said corporation.

In WITNESS WHEREOF, I have hereunto set my hand and official seal, in the City, County and State of New York, this 16th day of July, 1981.

Notary Public

Porylo A. Stact Notary Public, State of New York

No. 31-4729635

Qualified in New York County Commission Expires March 30, 1983

My Commission expires March 30, 1983.

to nent of Conveyance

Instrument of Conveyance of Production Payment

ATTACHED TO AND FORMING A PART OF THAT CERTAIN INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT DATED JULY 16, 1981 BETWEEN PENNZOIL COMPANY, ET AL. AND NEW DUNSTAN, INC.

This Exhibit A contains the specific descriptions of the Subject Interests which are referred to in Paragraph 1 of this Instrument of Conveyance of Production Payment as being specifically described or referred to in this Exhibit A.

PART ONE of this Exhibit A sets forth those Subject Interests located offshore Texas and Louisiana, and PART TWO of this Exhibit A sets forth those Subject Interests located in the States of Alabama, Arkansas, Louisiana, Mississippi, Montana, New Mexico, New York, North Dakota, Pennsylvania, Texas, West Virginia and Wyoning.

The Subject Interests are expressly limited to the Leoses set forth in this Exhibit A insofar and only insofar as they cover lands and depth intervals in which one or more of the Grantors own or hold an interest and do not include lands and depth intervals in which none of the Grantors own an interest, even though such lands or depth intervals are covered by the Leases set forth in this Exhibit A.

PART ONE of this Exhibit A refers to the "BLM or State Lease No." for each of the Leases lister, therein. Those Leases so listed having a number with the prefix "OCS-G" are Leases issued by the United States of America as Lessor through the Bureau of Land Management (the "BLM") of the United States Department of the Interior. Those Leases listed i. " ort One of this Exhibit A having a number with the prefix "St. Ls." are Leases issued by the State of Louisiana as Lessor.

In the case of each Subject Interest set forth in PART ONE of this Exhibit A, there is shown under the heading "Net Revenue Interest" Pennzoil Oil & Gas, Inc.'s (under the sub-heading "POGI") and Pennzoil Producing Company's (under the sub-heading "PP(") respective interest (expressed in percentages) in total production of Hydrocarbons produced and saved from and that accrues or is at ribur of the relevant Subject Interest after deducting all applicable roy, (i.e., overriding royalties, production payments and other payments out of production not owned by Pennzoil Oil & Gas, Inc. or Pennzoil' roducing Company, as the case may be.

In PART TWO of this Exhibit A, there is shown opposited) the description of each Subject Interest, or (ii) the well located up 1 or from which production is attributable to such Subject Interest or (iii) the unit which contains such Subject Interest or a portion thereof, under the heading "Net Revenue Interest" the aggregate interest of Grantors (expressed in decimals) in total production of Hydrocarbons produced and saved from and that accross or is attributable to the relevant Subject Interest, well or unit, as the case may be after deducting all applicable coyalties, overriding royalties, production payments and other payments out of production—of owned by the Grantors or any of them. Where the Net Revenue interest is listed opposite a well or unit, the relevant Subject Interests are listed below such well or unit.

The Leases set forth in PART TWO of this Exhibit A may be committed to more than one pool or unit, or may cover lands situated in more than one county or parish; consequently, such Leases may be listed more than once in this Exhibit A. Such repetition is not intended to nor shall it be construed as enlarging or otherwise affecting the Subject Interest or Subject Interests which appertain to such Lease.

All recording references in PART TWO of this Exhibit A are to the official records in the office of the county or parish clerk or recorder of the counties or parishes in which the relevant Subject Interests are situated.

On the pages of PART TWO of this Exhibit A relating to lands located in various Parishes within the State of Louisiana, the State of Louisiana is identified in the first line of the upper left corner of each of said pages by the letters "LA" and the respective Parishes are identified in the second line of the upper left corner of each of said pages. The column appearing on each of said pages under the heading "Recording Book-Page" or "Recorded Book-Page" sets forth recording references to the public records of the particular Parish concerned. The "Book" referred to is the book maintained by each Parish as a part of its official Conveyance Records, including the Conveyance

Book or any book which the Parish may maintain for the recordation of documents evidencing oil, gas and mineral transactions. In some instances, the recording reference set forth may also include (or consist solely of) the entry number, file number, original act number, or registry number, etc. under which a particular document is filed.

In the description of the Subject I terests for the states of New York, Pennsylvania and West Virginia, the Net Revenue Interest of Grantors is iisted opposite a "Farm" number and name. The Farm designation denotes the interest of the Grantors with reference to a single and unique royalty interest originating from one grantor or lessor (either leasehold or fee). All of the Leases to which the Farm designations relate are described in the pages immediately following the listing of Farms, by county, and the description of such Leases, standing alone, r arstitutes a complete description of all of the Leases intended to be covered by this Conveyance.

PART ONE

OFFSHORE SUBJECT INTERESTS

TEXAS

| BLM or State Lease No. | Effective Date | Description | Net Lever | PPC |
|---------------------------|-------------------|--|-----------|----------|
| OCS-G-2398 | 8- 1-7- | All of Block A-273, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.12639 | 11.24999 |
| OCS-G-2403 | 8- 1-73 | All of Block A-279, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 14.16666 |
| OCS-G-2414 | 8- 1-73 | All of Block A-323, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 11.66662 |
| OCS-G-2416 | 8- 1-73 | All of Block A-325, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 7.22016 | 57.14286 |
| OCS-G-2418 | 8- 1-73 | All of Block A-327, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 8.33333 |
| OCS-G-2422 | 8- 1-73 | All of Block A-332, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Te. as Map No. 7C | 1.00124 | 8.33333 |
| OCS-G-2739 | 7- 1-74 | All of Block A-339, High Island Area, East Addition, South Extension, OCS Official Leasing Map. Texas Map No. 7C | 1.00124 | 28.33332 |
| OCS-G-2426 | 8- 1-73 | All of Block A-340, High Island Area, East Addition, Sou h Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 28.33332 |
| OCS-G-2429 | 8- 1-73 | All of Block A-351, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 2.88528 | 36.62186 |
| OCS-G-2745 | 7- 1-74 | All of Block A-355, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 7.34246 | 9.44166 |

| BLM or State Lease No. | Effective Date | Description | Net Reven | PPC PPC |
|---------------------------|-------------------------|--|-----------|----------|
| OCS-G-2746 | 7 - 1-7 4 | All of Block A-35ô, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 10.01245 | 12.49999 |
| OCS-G-2366 | 8- 1-73 | All of Block A-474, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 9.99999 |
| OCS-G-2367 | 8- 1-73 | S ½ of NE ¼ & SE ¼ of Block A-475, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.001241 | 9.999996 |
| OCS-G-2372 | 8- 1-73 | All of Block A-489, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.49999 |
| OCS-G-3118 | 4- 1-75 | All of Block A.499. High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.10137 | 13.66667 |
| OCS-G-2378 | 8- 1-73 | All of Block A-520, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 9.99999 |
| OCS-G-2704 | 7- 1-74 | All of Block A-545, High Island Area South Addition, OCS Official Leasing Map, Texas Map No. 7B | 6.67497 | 8.33333 |
| OCS-G-2779 | 10- 1-74 | All of Block A-546, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2705 | 7- 1-74 | All of Block A-547, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2706 | 7- 1-74 | All of Block A-548, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2384 | 8- 1-73 | All of Block A-555, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.08332 |
| OCS-G-2388 | 8- 1-73 | All of Block A-563, High Island Area, South Addition, OCS Official Leasing Map, Texas Map, No. 7B | 1.00124 | 9,99999 |
| OCS-G-2389 | 8- 1-73 | All of Block A-564, High Island Area, South Addition, OCS Official Leasing Map, No. 7B | 1.00124 | 9,999999 |

| BLM or State | Effective Date | Description | Net Reven | ue Interest PPC |
|--------------|-------------------|--|-----------|--------------------|
| OCS-G-2390 | 8- 1-73 | All of Block A-570, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.01958 | 16.97190 |
| OCS-G-2719 | 7- 1-74 | All of Block A-582, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.17780 | 11.76499 |
| | | Louisiana | | |
| St. Ls. 7604 | 4-17-78 | TRACT 14836 — PORTION OF BLOCK 3 SABINE PASS AREA, Cameron Parish, Louisiana. That portion of Block 3, Sabine Pass Area, Cameron Parish, Louisiana, belong- ing to the State of Louisiana and not under a mineral lease on January 23, 1978, described as follows: Beginning at a point on the North line of Block 3, Sabine Pass Area having Lambert Plane Coordinates of X = 1,217,855.14 and Y = 364,274.14; thence East 9,612.44 feet along said North line to a point having Lambert Plane Coor- dinates of X = 1,227,467.58 and Y = 364,274.14; thence Southerly and Southwesterly along an arc having a radius of 18,240.60 feet and a vertex of X - 1,209,227 and Y = 364,245 to a point on the South line of said Block 3, having Lambert Plane Coordinates of X = 1,219,987.06 and Y = 349,516.10; thence West 2,131.92 feet along said South line to a point having Lambert Plane Coordinates of X = 1,217,855.14 and Y = 349,516.10; thence North 14,758.04 feet to the point of beginning, estimated to contain approximately 2,497.59 acres, as shown outlined in red on a plat on file in the Office of the Secretary, Department of Natural Resources, LESS AND EXCEPT that portion thereof, if any, which is more than three nautical miles from the coast line as determined by the Report of the Special Master in the litigation in the Supreme Court of the United States styled United States v. State of Louisiana et al. No. 9 Original, | | 38.125 |

| BLM or State Lease No. | Effective Date | Description | POCI | PPC |
|---------------------------|-------------------|---|----------|-----------------|
| | | said three mile line as set out in the June, 1975, decree of the Supreme Court. All bearings are based on Louisiana Lambert Plane Coordinate System (South Zone). | | |
| OCS-G-4144 | 9- 1-79 | That portion of Block 3 which is more than three geographical miles seaward from the line described in the supple- mental decree of the U.S. Supreme Court, June 16, 1975 (United States vs. Louisiana, 422 U.S. 13) Sabine Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 12. | | 41.66663 |
| OCS-G-4379 | 11- 1-80 | All of Block 53, West Cameron Area, as shown on OCS Leasing Map, Louisiana Map No. 1 | | 41.66665 |
| OCS-G-2839 | 12- 1-74 | All of Block 352, West Cameron Area, West Addition, OCS Official Leasing Map, Louisiana Map No. 1A | 8.34372 | 20.83333 |
| OCS-G-2224 | 2- 1-73 | All of Block 532, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.22499 |
| OCS-G-2225 | 2- 1-73 | All of Block 533, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.22499 |
| OCS-G-3284 | 8- 1-75 | All of Block 563, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 1B | 25.03113 | 20.83332 |
| OCS-G-2436 | 8- 1-73 | All of Block 586, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 16.68742 | |
| OCS-G-2021 | 2- 1-71 | All of Block 587, West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1 B | 22.80504 | |
| OCS-G-2850 | 12- 1-74 | All of Block 609, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.55555 |
| DCS-G-2559 | 5- 1-74 | All of Block 617, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.555 55 |

| BLM or State | Effective | | Net Reven | ue Interest |
|--------------|-----------|--|-----------|-------------|
| Lease No. | Date | Description | POCI | PPC |
| OCS-G-2860 | 12- 1-74 | All of Block 237, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 2.22499 | 5.55555 |
| OCS-G-2045 | 1- 1-71 | All of Block 270, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.68741 | |
| OCS-G-2062 | 2- 1-71 | All of Block 334, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| OCS-G-2439 | 8- 1-73 | All of Block 335, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| OCS-G-2078 | 2- 1-71 | All of Block 228, Vermilion Area, OCS Official Leasing Map, Louisiana Map No. 3 | 17.9273 | |
| OCS-G-2882 | 12- 1-74 | All of Block 125, South Marsh Island Area, South Addition, OC.3 Official Leasing Map, Louisiana Lap No. 3C | 8.89995 | 5.55555 |
| OCS-G-2883 | 12- 1-74 | All of Block 127, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5,55555 |
| OCS-G-2587 | 5- 1-74 | All of Block 128, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Μαρ No. 30 | 8.89995 | 5.55555 |
| OCS-G-0419 | 12- 1-54 | All of Block 53. Eugene Island Area as shown on Official Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisiana Offshore Operations) | | 83.33333 |
| OCS-G-0478 | 1- 1-55 | E 1/2 of Block 116, Eugene Island Area as shown on Official Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisiana Offshore Operations) | | 41.66665 |
| OCS-G-2102 | 2- 1-71 | All of Block 256, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 23,1054 | |
| OCS-G-2900 | 12- 1-74 | All of Block 261, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 5.56247 | 13.88889 |
| OCS-G 3156 | 7- 1-75 | All of Block 262. Eugene Island Area, as shown on OCS Official Leasing Map, Louisiana Map No. 4 | 5.56247 | 13.88888 |

| BLM or State Lease No. | Effective Date | € scription | Not Reven | PPC |
|---------------------------|-------------------|--|-----------|------------------|
| OCS-G-2104 | 2- 1-71 | All of Block 295, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 6.76497 | |
| OCS-G-2607 | 5- 1-74 | All of Block 312, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 2.22499 | 22.22221 |
| OCS-0-2115 | 1- 1-71 | All of Block 330, Eugene Island Area, South Addition, OCS Official Leasing Map No. 4A | 16.68741 | |
| OCS-G-2317 | 2- 1-73 | All of Block 333, Eugene Island Area, South Addition, OCS Official Leasing Map No. 4A | 2.22499 | 22.22 221 |
| OCS-G-3332 | 3- 1-76 | All of Block 337, Eugene Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | 16.68741 | |
| OCS-G-3409 | 1- 1-77 | All of Block 351, Eugene Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | 8.34371 | 20.83333 |
| OCS-G-0823 | 5- 1-60 | All of Block 186, Ship Shoal Area, as shown on Official Leasing Map, Louis- iana Map No. 5, Outer Continental Shelf Leasing Map (Louisiana Off- shore Operations) | | 28.33332 |
| OCS-G-1525 | 7- 1-67 | All of Block 222, Ship Shoal Area, OCS Official Leasing Map, Louisiana Map No. 5 | | 15.83332 |
| OCS-G-1984 | 9- 1-70 | N 14 Block 225, Ship Shoal Area, OCS Official Leasing Map, Leuisiana Map No. 5 | | 19.79165 |
| OCS-G-2177 | 11- 1-72 | All of Block 49 South Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 9 | 3.33748 | 8.33333 |
| OCS-G-3416 | 1- 1-77 | That portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (U.S. vs. Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971) as shown on CCS Official Leasing Map, Louisiana Map No. 9 | 6.67497 | 16.66667 |

| BLM or State | Effective | | Net Rever | nue Interest |
|---------------|-----------|--|-----------|--------------|
| Lease No. | Date | Description | POGI | PPC |
| St. I.s. 6310 | 5-13-74 | Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and being more par- ticularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Louisiana | 5.92738 | 14.80 |
| OCS-G-2185 | 10- 1-72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court entered December 13, 1965 in the U.S. v. Louisiana No 9 Original (382 U.S. 288), South Pass Area, South and East Addition, as shown on GCS Official Leasing Map, Louisiana Map No. 9A | 5.00622 | 12.49999 |
| OCS-G-3195 | 7- 1-75 | Block 72, Main Pass Area, that portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 7.54441 | 18.83758 |
| OCS-G-3417 | 1- 1-77 | That portion of Main Pass Blocks 7 and 74 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (U.S. vs Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971) as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 7.54441 | 18.83758 |
| OCS-G-2947 | 12- 1-74 | All of Block 73, Main Pass Area, OCS Official Leasing Map, Louisiana Map No. 10. That portion located more than one foot seaward of the Third Supplemental Decree Line (404 S. 388 (December 20, 1971)) | 7.54441 | 18.83758 |

| BLM or State | Effective | | Net Revenu | se Interest |
|--------------|-----------|--|------------|-------------|
| Luase No. | Date | Description | POGI | PPC |
| OCS-G-2193 | 10- 1-72 | All of Block 140, Main l'ass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 3.33748 | 8.33333 |
| OCS-G-3206 | 7- 1-75 | All of Block N663E63, Mobile South No. 2, as shown on OCS Official Leasing Map, NH-16-10 (Also known as Block 63), Mississippi Canyon Area, as shown on OCS Official Protraction Diagram NH-16-10 | 3.33748 | 8.33333 |
| OCS-G-3500 | 8- 1-77 | All of Block 253, West Cameron Area, as shown on OCS Official Leasing Map, Louisiana Map No. 1 | 10.01245 | 25.0000 |
| OCS-G-3785 | 6- 1-78 | All of Block 372, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 12.4999 | 12.4999 |

PART TWO

STATES (F ALABAMA, ARKANSAS, LOUISIANA, MONTANA, MISSISSIPPI, NEW MEXICO, NORTH DAROTA, TEXAS, WYOMING NEW YORK, PENNSYLVA HA AND WEST VIRGINIA

EXHIBIT B to Instrument of Conveyance of Production Payment

DISCHARGE SCHEDULE

| Aegular Application Date | Hydrocerbons Monthly Percentage | Minerals Monthly Percentage |
|--------------------------------|---------------------------------------|-----------------------------------|
| October 1, 1983 | 1.2011% | 0.9320% |
| November 1, 1983 | 1.2011% | 0.9320% |
| December 1, 1983 | 1.2011% | 0.9320% |
| January 1, 1984 | 1.201158 | 0.9320% |
| February 1, 1984 | 1.2279% | 0.9800% |
| March 1, 1984 | 1.227956 | 0.9800% |
| April 1, 1984 | 1.2279% | 0.9800% |
| May 1, 1984 | 1.22797 | 0.9800% |
| June 1, 1984 | 1.227967 | 0.9800% |
| July 1, 1984 | 1.227977 | 0.9800% |
| August 1, 1984 | 1.227976 | 0.9800% |
| September 1, 1984 | 1.2279% | 0.9800% |
| October 1, 1984 | 1.227976 | 0.9800% |
| November 1, 1984 | 1.227957 | 0.9800% |
| December 1, 1984 | 1.22797 | 0.9800% |
| January 1, 1985 | 1.227957 | 0.9800% |
| February 1, 1985 | 1.270858 | 1.0600% |
| March 1, 1985 | 1.2708% | 1.0600% |
| April 1, 1985 | 1.270857 | 1.0600% |
| May 1, 1985 | 1.2708% | 1.0600% |
| June 1, 1985 | 1.270807 | 1.0600% |
| July 1, 1985 | 1.27080 | 1.0600% |
| Augus: 1, 1985 | 1.270867 | 1.0600% |
| September 1, 1985 | 1.270897 | 1.0600% |
| October 1, 1985 | 1.2708% | 1,0600% |
| November 1, 1985 | 1.27080 | 1.0600% |
| December 1, 1985 | 1.27080 | 1.0600% |
| January 1, 1986 | 1.2708% | 1.0600% |
| February 1, 1986 | 1.3834% | 1.2440% |
| March 1, 1986 | 1.3834% | 1.24:0% |
| April 1, 1986 | 1.3834% | 1.2440% |
| May 1, 1986 | 1.383457 | 1.2440% |
| June 1, 1986 | 1.3834% | 1.2440% |
| July 1, 1986 | 1.3834% | 1.2440% |
| August 1, 1986 | 1.3834% | 1.2440% |
| September 1, 1986 | 1.3834% | 1.2440% |
| October 1, 1986 | 1.3834% | 1.2440% |

DISCI'ARGE SCHEDULE - Continued

| Regular Application Date | Hydrocarbons Monthly Percentage | on rals onthly Percentage |
|--|---------------------------------------|---------------------------------|
| November 1, 1986 | 1.3834% | 1.2440% |
| December 1, 1986 | 1.3834% | 1.2440% |
| January 1, 1987 | 1., 834% | 1.2440% |
| February 1, 1987 | 1 31277 | 1.2720% |
| March 1, 1987 | 1.31376 | 1.2720% |
| April 1, 1987 | 1.31374 | 1.2720% |
| May 1, 1987 | 1.3137 % | 1.2720% |
| June 1, 1987 | 1.3137% | 1.272068 |
| July 1, 1987 | 1.3137% | 1, 720% |
| August 1, 1987 | 1.31377 | 1.2720% |
| September 1, 1987 | 1.3137% | 1.2720% |
| October 1, 1987 | 1.3137% | 1.272000 |
| November 1, 1987 | 1.31375 | 1.2720 |
| December 1, 1987 | 1.3137% | 1.272 50 |
| January 1, 1988 | 1 3137% | 17200 |
| Februa v 1, 1988 | 1.3244% | 1.5720 , 6 |
| March 1, 1988 | 1.3244% | 1.3720% |
| April 1, 1988 | 1.3244% | 1.3720% |
| May 1, 1988 | 1.3244 8 | 1.3720% |
| June 1, 1989 | 1.4244% | 1.3729% |
| July 1, 1988 | 4 0 14 4 6 4 | 1.3720% |
| August 1, 1989 | 1.3244% | 1 20% |
| September 1, 1738 | 1.3244% | 1.37. 1% |
| October 1, 1958 | | 1.3720% |
| 31 1 1 1000 | 1.3244% | 1.3720% |
| December 1, 1988 | 1.3244% | 1.3720% |
| January 1, 1989 | 1.32447 | 1.3720% |
| Februar: 1, 1989 | 1.3351% | 1.4920% |
| | 1.3351% | 1.4020% |
| April 1, 1989 | 1.3351% | 1.492356 |
| May 1, 1989 | 1.3351% | 1.4920% |
| June 1, 1989 | 1.3351% | 1.4020% |
| July 1, 1989 | 1.3351% | 1.4990% |
| August 1, 1989 | 1.3351% | 1.4920% |
| September 1, 1989 | 1.3351% | 1.49205% |
| October 1, 198) | 1.3351% | 1.4920% |
| November 1, 1989 | 1.3351% | 1.4920% |
| December 1, 1989 | 1.33516 | 1.49297 |
| January 1, 1990 | 1.3351% | 1.4920 2 |
| February 1, 1990 | 0.9320% | 1,4464% |
| March 1, 1990 | 0 to 2011/02/2015 AN | 1.4464% |
| April 1, 1990 | | 1.4464% |
| May 1, 1990 | es un secs ó | 1.4464% |
| June 1, 1990 | | |
| SECTION OF THE PROPERTY AND ADDRESS OF THE PARTY OF THE P | | 1.446476 |

EXHIBIT C

to

Instrument of Conveyance of Production Payment

PENNZOIL COMPANI, ET AL.

TO

NEW DUNSTAN, INC.

Supplemental Instrument of Conveyance of Production Payment

Dated as of, 198

Supplementing the
Instrument of Conveyance of Production Payment
Dated July 16, 1981

WITNESSETH:

Whereas the Grantors and the Grantee have heletofore en erest into an astrument of Conveyance of Production Payment dated July 15, 1985 (the "Conveyance") and an executed counterpart of the Conveyance was filed for record in Volume. Page of the Deed Reards of Culberson County, Texas, to which recorded counterpart afference is made for all purposes; and

Whereas, the Grantors desire to supplement Exhibit A to the Conveyance pursuant to and in the manner prescribed by subparagraph (b) of Paragraph 7 of the Conveyance;

Now, Therefore, the Grantors hereby execute and deliver this Supplemental Conveyance as follows:

 Additions to Exhibit A. Exhibit A to the Conveyance is hereby supplemented by adding hereto the following:

PART THREE

SULPHUR CULCUSON COUNTY, TEXAS

PATENTED MANAGE CLAIM

563.6 acres of land out of Section 2. Block 111, PSL Survey, Culberson County, Tevas, and being the some land covered by Prose certain thirty-four (34) Mineral Patents delayer of as (1) Nos. 358-268. Volume 40-B, recorded in Volume 6, Proges and C6; (ii) No. 212, Volume 40-B, recorded in Volume 7, Page 584; and (44) Pros. 370-391, Volume 40-B,

recorded in Volume 6, Pages 468-489, (all recordation in the Patent ords of Culber on County, Texas), subject, however, to (i) the ovalty of the State of Texas equal to six and one-fourth percent (614%) of the value of the production of sulphur and certain other unerals from said unpatented claims and awards as shown by the net smelter, mill, mint or refinery returns or of the gross sums arising from the sale of the ore or products from such claims and awards and received by the owner thereof, and (ii) a perpetual royalty interest of at undivided one-thirty second of the gross value of all production from said mining claims of sulphur, salt, potash and any other minerals covered by said mining a aims, being an undivided one-half of that certain royalty reserved in Pat certain Conveyance of Unpatented Mining Claims dated September 1, 1967, and recorded in Volume 3, page 217, of the Mineral Records of Culberson County, Texas from To, at Oil Corporation to Duval Corporation, insofa, as the same relates to the above-described claims.

IV.

LEASEHOLD

An undivided one-half (19) interest in and to 160 acres of land out of Sections 2 and 3, Block 111, PSL Survey, Culberson County, Texas, and being further identified as the "Dot" Mining Claim, Mineral Patent No. 70, covered by Sulphur Lease dated September 29, 1967, from R. B. McGowen, Jr., et al. to Duva! Corporation, and recorded in Volume 3, page 212, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty on all sulphur mined, saved and marketed from said land by Lessee of one-sixth (16th) (reduced to one twelfth (1/12th) by reason of proportionate reduction clause), f.e.b., mine site.

III.

LEASEHOLD

An undivided one-half (12) interest in and to 160 acres of land out of Section 2 and 3, Block 111, PSL Survey, Culberson County, Texas, and he re, the "Dot" Mining Claim, Mineral Patent No. 70, covered by Subline Lease dated September 14, 1967, from Toyah Valley Sulphur Co. of Inolders, Inc. to Fred Prickett, and recorded in Volume 3, Page

309, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty of one-sixth (1₆th) (reduced to one twelfth (1/12th) by reason of proportionate reduction clause), of the gross production of sulphur produced from the leased premises.

IV.

LEASEHOLD

555 acres 6° land out 6° Section 5, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by Sulphur Lease dated November 7, 1957, from the State of Texas to W. C. Tillett, General Land Office Sulphur Lease No. 61880, and recorded in Volume 2, page 242, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty reserved to the State of Texas in said Lease of one-sixth (16th) of the gross reduction, or the value thereof, of the sulphur; but in no event, to be less than Four and 50/100 Dollars (\$4.50) per long ton.

V.

PATENTED MINING CLAIMS

639.96 acres of land out of Section 4, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those thirtysix (36) Mineral in lents identified as Nos. 256-291, Volume 38-B, recorded in Volume 5. Pages 326-361, inclusive, of the Patent Records of Culberson County, Texas, subject, however, to the royalty of the Stale of Texas equal to six and one-fourth person (614%) of the value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mid, mixt, or refinery returns or of the gress sums arising from the sale of the ore or products from such claims and awards by the owner thereof.

VI.

LEASEHOLD

214.27 acres of land out of Section 9, Block 111, PSL Survey, Cuiberson County, Texas, and being the same land covered by Sulphur Lease dated April 15, 1968, from R. B. McGovern, Jr., et al., to Duval Corporation, General Land Office Mineral File No. M-61178, and

recorded in Volume 5, price 72, of the Mineral Records of Culberson County, Texas, subject, powever, to a royalty of one-sixth (44th) of the gross production of sulphur produced from said leased premises, reserved in said lease.

VII.

PATENTED MINING CLAIMS

425.58 acres of land out of Section 9, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain twenty-three (23) Mineral Patents identified as Nos. 173-195, Volume 39-B, recorded in Volume 6, Pages 52-74, inclusive, of the Patent Records of Culberson County, Texas, subject, however, to (i) the royalty of the State of Texas equal to six and ene-fourth per cent (614%) of the value of the production of sulphur and certain other minerals from said patented claims and awards as shown by the net smelter, mill, mint, or refinery returns or of the gross sums arising from the sale of ore or products from such claims or awards and received by the owner thereof, and (ii) a perpetual royalty interest of an undivided one-thirty second of the gross value of all production from said mining claims of sulphur, salt, potash and any other minerals covered by said mining claims, being an undivided one-half of that certain royalty reserved in that certain Conveyance of Unpatented Mining Claims dated September 1, 1967, and recorded in Volume 3, page 217, of the Mineral Records of Culberson County, Texas, from Topat Oil Corporation to Duval Corporation, insofar as the same relates to the above-described claims and awards.

VIII.

PATENTED MINING CLAIMS

640.56 acres of land out of Section 10, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain thirty-six (36) Mineral Patents identified as (i) Nos. 220 and 221, Volume 38-B, recorded in Volume 5, Pages 290-291; (ii) Nos. 223-234, Volume 38-B, recorded in Volume 5, Pages 293-304; (iii) Nos. 236-240, Volume 38-B, recorded in Volume 5, Pages 306-310; (iv) Nos. 242-255, Volume 38-B, recorded in Volume 5, Pages 312-325; (v) No.

96, Volume 40-B, recorded in Volume 6, Page 76; (vi) Nos. 255 and 256, Volume 39-B, recorded in Volume 5, Pages 550 and 551 (all recordation in the Patent Records of Culberson County, Texas), subject, however, to the royalty of the State of Texas equal to six and one-fourth per cent (614%) of the value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or the gross sums arising from the sale of the ore or products from such claims and received by the owner thereof.

IX.

LEASEHOLD

480 acres of land out of Section 15, Block 111, PSL Survey, Culberson County, Texas, being the E-12, S-12 SW-14, N-12 NW-14, of said Section, and being the same land covered by Sulphur Lease dated November 7, 1967, from the State of Texas to W. C. Tillett, General Land Office Sulphur Lease No. M-61881, and recorded in Volume 3, page 246, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty reserved to the Strong of Texas in said Lease of one-sixth (16th) of the gross production or the value thereof, of the sulphur; but in no event, to be less than Four and 50/100 Dollars (\$4.50) per long ton.

X.

PATENTED MINING CLAIM

380.79 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain twenty-six (26) Mineral Patents identified as Nos. 47-72, Volume 38-B, recorded in Volume 5, Pages 209-232, inclusive, of the Patent Records of Culberson County, Texas, subject, however, to the royalty of the State of Texas equal to six and one-fourth per cent (614%) of the value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or the gross sums arising from the sale of the ore or products from such claims and received by the owner thereof.

XI.

PATENTED MINING CLAIM

31.54 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain two (2) Mineral Patents identified as Nos. 93 and 94, Volume 40-B, recorded in Volume 6, Pages 270-271 of the Patent Records of Culberson County, Texas, subject, however, to the royalty of the State of Texas equal to six and one-fourth per cent (614%) of the value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or the gross sums arising from the sale of the ore or products from such claims and received by the owner thereof.

XII.

LEASEHOLD ON PATENTED MINING CLAIM

80 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, being the E-1/2 of the "Virginia" Mining Claim, Mineral Patent No. 68, Volume 1, below the elevation of 100 feet measured vertically from the surface thereof on July 1, 1968, and being the same land covered by Sulphur Lease dated March 26, 1968, from Violet G. O'Fiel to Jack R. Fraser, and recorded in Volume 4, page 267, of the Mineral Records of Culberson County, Texas, insofar only as the same covers below the elevation of 100 feet measured vertically from the surface of said land on July 1, 1968, subject, however, to royalty and overriding royalty aggregating four-forty seconds (4/42nds) of all sulphur produced and marketed.

XIII.

MINERAL DEED AND PATENTED MINING CLAIM

80 acres of land out of Section 16, Block 111, PSL Survey, Culberson County. Texas, being the W-12 of the "Virginia" Mining Claim, Mineral Patent No. 68, Volume 1, below the depth of 100 feet measured vertically from surface level thereof on February 28, 1968, and being the same land conveyed by Deed dated February 28, 1968, from Jack R. Fraser to Duval Corporation, and recorded in Volume 5, page

91, of the Mineral Records of Culberson County, Texas, and being the same land conveyed by Deed dated February 28, 1968, from Jack R. Fraser, Successor Trustee of Michigan Sulphur and Oil Co., to Duyal Corporation, and recorded in Volume 5, page 89, of the Mineral Records of Culberson County, Texas.

XIV.

LEASEHOLD

40 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by Sulphur Lease dated May 7, 1968, from the State of Texas to Duval Corporation, General Land Office Sulphur Lease No. M-62083, and recorded in Volume 5, page 128, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty reserved to the State of Texas in said Lease of one-sixth (16th) of the gross production, or the value thereof, of the sulphur; but in no event, to be less than Four and 50/100 Dollars (\$4.50) per long ton.

XV.

LEASEHOLD

33.02 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by Sulphur Lease dated May 7, 1968, from R. B. McGowen, Jr., et al, to Duval Corporation, General Land Office Mineral File No. M-61996, and recorded in Volume 5, page 75, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty of one-sixth (1/6th) of the gross production of sulphur produced from said leased premises, reserved in said lease.

XVI.

PATENTED MINING CLAIMS

160.44 acres of land out of Section 11, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those nine (9) Mineral Patents identified as Nos. 438-446, Volume 38-B, recorded in Volume 5, Pages 455-463, inclusive, of the Patent Records of Culberson County, Texas, subject, however, to (i) the royalty of

.

the State of Texas equal to six and one-fourth per cent (64%) of the value of the production of sulfur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or of the gross sum arising from the sale of the ore or products from such claims and received by the owner thereof, and (ii) a perpetual royalty interest of an undivided one-thirty second of the gross value of all production from said mining claims of sulfur, salt, potash and any other minerals covered by said mining claims, being an undivided one-half of that certain royalty reserved in that certain Conveyance of Unpatented Mining Claims dated September 1, 1967 and recorded in Volume 3, page 217, of the Mineral Records of Culberson County, Texas, insofar as the same relates to the above described claims.

XVII.

FEE TITLE AND LEASEHOLD

- (1) An undivided one-half (1½) fee interest in and to the 160 acres comprising the south one-half of the northwest one-quarter and the north one-half of the southwest one-quarter of Section 15, Block 111; PSL Survey, Culberson County, Texas, being the same land covered by Deed dated July 29, 1971, and recorded in Volume 92, Page 206 of the Deed Records of Culberson County, Texas; and
- (2) An undivided one-half (1½) interest in and to the 160 acres comprising the south one-half of the northwest one-quarter and the north one-half of the southwest one-quarter of Section 15, Block 111, PSL Survey, Culberson County, Texas, and being the same and covered by a Mineral Lease dated July 22, 1971, from Jack Russell to Duval Corporation and recorded in Volume 8, Page 48, Mineral Records of Culberson County, Texas, subject, however, to an initial royalty of one-sixteenth (1/16) on sulphur, either in kind or market value at the railhead at the mine site, for the first 2,750,000 long tons whether produced or not and a subsequent royalty of one-sixteenth (1/16) on all sulphur actually produced in excess of 2,750,000 long tons reduced to .001% when 3,000,000 long tons have been produced.

XVIII

LEASEHOLD

2.256 acres of land out of Sections 14, 15, 22 and 23, Block 110, PSI, Survey, Culberson County, Texas, being the same land covered by Sulphur Lease dated August 26, 1975, from Hillary Phillips, Jr. and Benny Gail T. Hunnicutt to Fred T. Armstrong, General Land Office Mineral File No. M-72194 and recorded in Volume 8, Page 414 of the Mineral Records of Culberson County, Texas, subject, however, to a royalty reserved to the State of Texas in said Lense of 16.67% of the gross production of sulphur, and subject to an overriding royalty interest of .00332 of all sulphur that may be produced from the land covered by said Lease reserved in Assignment dated September 15, 1975, from Fred T. Armstrong to Duval Corporation, recorded in Volume 9, Page 51 of the Mineral Records of Culberson County, Texas.

XIX

FEE TITLE

160 acres of land, being all of Pre-emption Survey No. 86 as described in Patent dated November 14, 1945, recorded in Volume 7, Page 75 of the Deed Records of Culberson County, Texas, and being the same land covered by Deed dated January 31, 1969, from R. B. McGowen, Jr., et al., to Dayal Corporation, recorded in Volume 86, Page 121 of the Deed Records of Culberson County, Texas.

2. Conveyance of the Production Payment. The Grantors, in consideration of the payments made and agreed to be made to them by the Grantee of the Purchase Price pursuant to the provisions of Paragraph 7 of the Conveyance and of other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, hereby confirm and ratify the estates and rights created by the Conveyance, as supplemented by this Supplemental Conveyance, and, in furtherance of such confirmation and ratification, the Grantors by these presents do hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, effective as of 1, 198 [first day of the calendar month in which the Supplemental Instrument of Conveyance of Production Payment is executed as a production payment, that undivided percentage, or if 100%, the percentage, of the Subject Minerals (including, without limitation, those Subject Minerals in and under and which may be produced, saved and sold and which shall accrue and be attributable to, the Subject Interests described in Paragraph 1 of this Supplemental Conveyance) provided for in Subparagraphs First, Second, Third and Fourth of Paragraph 2 of the Conveyance, together with an easement of ingress and egrees to remove the same from the applicable Subject Interests to the extent Grantors have the right to grant such an easement, for the purposes, upon the terms and conditions, and subject to the encumbrances, reservations and limitations contained in the Conveyance, as supplemented by this Supplemental Conveyance.

3. Warranty of Title. The Grantors hereby make, as to the Subject Interests described and referred to in Paragraph 1 hereof, all of the representations and warranties contained in Paragraph 20 of the Conveyance.

4. Miscellaneous Provisions.

- (a) From and after the execution and delivery of this Supplemental Conveyance, the Conveyance shall be deemed to be supplemented as herein provided (and the term "Conveyance" shall thereafter mean the Conveyance as so supplemented), and as so supplemented shall continue in full force and effect, and none of the rights, titles or interests existing or to exist thereunder shall be released, diminished or impaired.
- (b) The Conveyance and this Supplemental Conveyance shall be read, taken and construed as one and the same instrument.
- (c) Unless otherwise specifically provided herein, all terms contained in this Supplemental Conveyance which are defined in the Conveyance shall for all purposes hereof have the meanings given to such terms in the Conveyance

In Witness Wheneof, the Grantors have executed this instrument to presence of the undersigned witnesses on the date set forth in the weldgment agreed hereto, in multiple originals.

| | PENNZOIL COMPANY |
|-------------------|----------------------------|
| CAL] | |
| | By Vice President |
| Attest: | |
| | |
| [CORPORATE SEAL] | PENNZOIL PRODUCING COMPANY |
| [Contonate Diana] | By Vice President |
| Attest: | |

| ************************ | |
|---|--------------------------|
| [C C] | PENNZOIL OIL & GAS, INC. |
| [CORPORATE SEAL] | |
| Attest: | ByVice Presia ent |
| ittest. | |
| FO 12 COMM CX 10 FO 10 10 10 10 10 10 10 10 | |
| | DUVAL CORPORATION |
| [Corporate Seal] | |
| | By Vice President |
| Attest: | |
| *************** | |

STATE OF NEW YORK COUNTY OF NEW YORK BEIT REME: .d. D, that I, Notary Public dulg a salified, commissioned, sworn and acting in ard for the County and St. 's aforesaid, hereby certify that on this day of ,198 : There appeared before me severally each of the following persons, each residing at the address set opposite his name, and each the designated officer of the corporation set opposite his name, each such corporation being a corporation of the sta dicat and being a party to the foregoing instrument: , Vice President of Pennzoil Corporation, a Delaware corporation , Vice President of Pennzoil Producing Company, a Delaware corporation , Vice President of Pennzoil Oil & Gas, Inc., a Dela vare corporation , Vice President of Duval Corporation, a Delaware corporation , Vice President of New Dunstan, Inc., a Delaware corporation.

Before the personally came each such person, to me personally known, and each, being by me duly sworn, did depose and say that he reside at the address set opposite his name and is the designated officer of the corneration set opposite his name and that said corporation is described usual executed the above instrument; that he knows the seal of the corporation; that the seal affixed to such instrument is such corporate seal; that it was so affixed by order of the Board of Directors of the surporation; and that a signed has name ther to by like order and that the instrument was the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared each of such person, each of whom is known to me to be the person whose name is subscribed to the foregoing instrument, and known to me to be the designated officer of the corporation set opposite his name, and each acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, in the capacity therein stated and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and official seal, in the City, County and State of New York, this day of, 198...

Notary Public

Notary Public, State of New York

Qualified in New York County
Commission Expires, 19...

My commission expires, 19 ...



to
Instrument of Conveyance
of Production Payment

CERTIFICATE

Penazil Company. Delaware corporation, Penazoil Producing Company, a Delaware corporation, Penazoil Oil & Gas, Inc., a Delaware corporation and Duval Corporation, a Delaware Corporation (the "Grantors"), do hereby state and certify as follows:

- 1. This Certificate is furnished pursuant to subparagraph (b) of Paragraph 7 of that—rtain Instrument of Conveyance of Production Payment dated as of July 16, 1981 between the Grantors and New Dunstan, Inc., a Delaware corporation (the "Grantee"), (said Instrument of Conveyance of Production Payment being herein called the "Conveyance"). Unless otherwise defined herein, defined terms used herein have the meanings assigned thereto in the Conveyance.
- 2. Except to the extent that the representations and warranties contained in Paragraph 20 of the Conveyance may not be true as a result of transactions or states of fact permitted by provisions of the Conveyance, the representations and warranties contained in Paragraph 20 of the Conveyance are true on and as of the date hereof as through made on and as of such date.
- 3. No Event of Default, or an event which with the giving of notice or the lapse of time, or both, would become an Event of Default, has occurred and is continuing, or would result from the payment of the installment of the Purchase Price to be made on the date hereof.

| r, the Grantors have caused this certificate to rs of the Grantors thereunto duly authorized , 19 °. |
|--|
| PENNZOIL COMPANY |
| By |
| Pennzoil Producing Company |
| By Title: |
| Pennzoil Oil & Gas, Inc. |
| By |
| DUVAL CORPORATION |
| By Title: |

Insert the date of payment of the applicable installment of the Purchase Price.

(" . ·) . . rs

LISKOW & LEWIS

ATTORNEYS AT LAW

NEW ORLEANS, LA 70139

THE SHELL BOUARE

FIFTIETH FLOOR

TELEPHONE (804) 581-7979

TWF 810-951-9252

LAFAYETTE LA 70505

321 TRAVIS ST

New Orleans, 70139 July 20, 1981

Mr. John L. Rankin, Manager N v Orleans Outer Continental Shelf Office Bureau of Land Management 500 Camp Street, Suite 841 New Orleans, Louisiana 70130 THOMAS F GETTEN
GEORGE H ROBINSON, JA
GEORGE J DOMAS
MARILYN C MALONEY
POBERT W BOOKSH, JR
JOSEPH C GIGLIO, JR,
LARRY M ROEDEL
LAW GORPORATIONS

LAW CORPORATIONS
BRUCE JOBECK
PATRICK W GRAY
DESCRATE HOLDEN
JOE B NORMAN
THOMAS M MEVANMANA
JAMES N MONTHELD III
BILLY J DOMINOUE
LAMBERT M LAPEROUSE
SUSAN S LASITER
PHILLP HOYES, JR
ANGEL TAYENOM
PAUL B DAVID
LYNN C HANTEL
JULIE STEMPAL

JULIE



Re: Mortgage, Deed of Trust and Assignment of Production New Dunstan, Inc. to F. D. Thompson and W. T. Fox III, Trustees, and Citibank, N.A., as Agent

Dear Mr. Rankin:

CULLEN R LISEOW (1893-1971) AUSTIN W LEWIS ((910-1974)

WILLIAM M MEYERS ROBERT Y JORDEN CMARLES C GREMILLION GENE & LAPITTE BILL, M MINES

DENE & LAPITY
JAMES L PELLETIER
THOMAS D HARDEMAN
LLITTON WORDAN
TOWNED WITH
EDWARD J DAY III
RENNETH E GORDON JR
NILLIAM R PITTS
J BERRY ST. JOHN JR
J BERRY ST. JOHN JR
JOHN M WILSON
LOWER DENEMBRY
THE ST. JOHN JR
FREDERICK W. BRADLEY
FREDERICK W. BRADLEY
FREDERICK W. BRADLEY
FREDRY MASSAR
S GENE FENDLER

by instrument of Conveyance of Production Payment (the "Conveyance") dated July 16, 1981 executed by Pennzoil Company, Pennzoil Producing Company, Pennzoil Oil & Gas, Inc. and Duval Corporation (the "Grantors") in favor of New Dunstan Inc. (the "Grantee"), the Grantors conveyed to Grantee a production payment relating to the properties and interests described in Part One and Part Two of Exhibit A attached to the Coveyance, including the interests of Grantors in and to the federal OCS oil and gas leases which are described in Part One of Exhibit A to the Conveyance ("Part One of Exhibit A"). A copy of Part One of Exhibit A is attached to this letter as a part thereof.

The production payment referred to above is subject to, and affected by, a Mortgage, Deed of Trust and Assignment of Production (the "Mortgage") dated July 16, 1981 executed by Grantee in favor of F. D. Thompson and W. T. Fox III, Trustees, and Citibank, N.A.., as Agent.

Enclosed is a fully executed counterpart of the Mortgage which has a copy of the Conveyance attached thereto

as Exhibit B and such copy of the Conveyance has attached thereto a complete copy of Part One and Part Two of Exhibit A to the Conveyance. In order to place third parties on notice as to the execution and efficacy of the Mortgage, please file this particular counterpart of the Mortgage, with a copy of this letter attached thereto, in the file maintained by your office with respect to Lease OCS-G 2115.

Also enclosed are sixty-six (66) fully executed counterparts of the Mortgage. Insofar as Exhibit A to the Conveyance is concerned, these counterparts of the Mortgage have attached thereto only Part One of Exhibit A. In order that third parties will be placed on notice as to the execution and efficacy of 1. Mortgage, please file one of these counterparts of the Mortgage, with a copy of this letter attached thereto, in the file maintained by your office with respect to each o, the federal OCS leases listed in Part One of Exhibit A (other than Lease OCS-G 2115).

Onclosed is our check in the amount of \$1,675.00 in payment of filing fees.

It would be appreciated if you would acknowledge that filing as above requested has been accomplished by signing all copies of this letter in the space provided below.

Yours very truly,

LISKOW & LEWIS

BY: Willow In Jorge

WMM: eeb

FILED AS REQUESTED

BY: (Orig. Sgd.) Ruby I. Boehm

New Orleans Outer Continental Shelf Office

Bureau of Land Management

Date: July 20, 1981

PART ONE

OFFSHORE SUBJECT INTERESTS

TEXAS

| BLM or State | Effective | | Charles of the County of the C | nue Interest |
|--------------|-----------|---|--|--------------|
| Lease No. | Date | Description | POGI | PPC |
| OCS-G-2398 | 8- 1-73 | All of Block A-273, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.12639 | 11.24999 |
| OCS-G-2403 | 8- 1-73 | All of Block A-279, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 14.16666 |
| OCS-G-2414 | 8- 1-73 | All of Block A-323, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 11.66662 |
| OCS-G 2416 | 8- 1-73 | All of Block A-325, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 7.22016 | 57.14286 |
| OCS-G-2418 | 8- 1-73 | All of Block A 327, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 8.33333 |
| OCS-G-2422 | 8- 1-73 | All of Block A-332, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 8.33333 |
| OCS-G-2739 | 7- 1-74 | All of Block A-339, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 28.33332 |
| OCS-G-2426 | 8- 1-73 | All of Block A-340, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 28,33332 |
| OCS-G-2429 | 8- 1-73 | All of Block A-351. High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 2.88528 | 36.62186 |
| OCS-G-2745 | 7- 1-74 | All of Block A-355, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 7.34246 | 9.44166 |

| BLM or State Lease No. | Effective Date | Description | Net Reven | ue Interest PPC |
|---------------------------|-------------------|--|-----------|--------------------|
| OCS-G-2746 | 7- 1-74 | All of Block A-356, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 10.01?45 | 12.49999 |
| OCS-G-2366 | 8- 1-73 | All of Block A-474, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 9.99999 |
| OCS-G-2367 | 8- 1-73 | S ½ of NE ¼ & SE ¼ of Block A-475, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.001241 | 9.999996 |
| OCS-G-2372 | 8- 1-73 | All of Block A-489, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.49999 |
| OCS-G-3118 | 4- 1-75 | All of Block A-499, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.10137 | 13.66667 |
| OCS-G-2378 | 8- 1-73 | All of Block A-520, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 9.99999 |
| OCS-G-2704 | 7- 1-74 | All of Block A-545, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 6.67497 | 8.33333 |
| OCS-G-2779 | 10- 1-74 | All of Block A-546, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2705 | 7- 1-74 | All of Block A-547, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2706 | 7- 1-74 | All of Block A-548, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2384 | 8- 1-73 | All of Block A-555, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.08332 |
| OCS-G-2388 | S- 1-73 | All of Block A-563, High Island Area, South Addition, OCS Official Leasing Map, Texas Map, No. 7B | 1.00124 | 9.99999 |
| OCS-G-2389 | 8- 1-73 | All of Block A-564, High Island Area, South Addition, OCS Official Leasing Map, No. 7B | 1.00124 | 9.99999 |

| BLM or State | Effective | | Net Reven | ue Interest |
|--------------|-----------|--|-----------|-------------|
| Lease No. | Date | Description | POGI | PPC |
| OCS-G-2390 | 8- 1-73 | All of Block A-570, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.01958 | 16.97190 |
| OCS-G-2719 | 7- 1-74 | All of Block A-582, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.17780 | 11.76499 |
| | | Louisiana | | |
| St. Ls. 7604 | 4-17-78 | TRACT 14836—PORTION OF BLOCK 3 SABINE PASS AREA, Cameron Parish, Louisiana. That portion of Block 3. Sabine Pass Area, Cameron Parish, Louisiana, belong- ing to the State of Louisiana and not under a mineral lease on January 23, 1978, described as follows: Beginning at a point on the North line of Block 3, Sabine Pass Area having Lambert Plane Coordinates of X = 1,217,855.14 and Y = 364,274.14; thence East 9,612.44 feet along said North line to a point having Lambert Plane Coor- dinates of X = 1,227,467.58 and Y = 364,274.14; thence Southerly and Southwesterly along an arc having a radius of 18,240.60 feet and a vertex of X = 1,209,227 and Y = 364,245 to a point on the South line of said Block 3, having Lambert Plane Coordinates of X = 1,219,987.06 and Y = 349,516.10; thence West 2,131.92 feet along said South line to a point having Lambert Plane Coordinates of X = 1,217,855.14 and Y = 349,516.10; thence North 14,758.04 feet to the point of beginning, estimated to contain approximately 2,497.59 acres, as shown outlined in red on a plat on file in the Office of the Secretary, Department of Natural Resources, LESS AND EXCEPT that portion thereof, if any, which is more than three nautical miles from the coast line as determined by the Report of the Special Master in the litization in the Supreme Court of the United States styled United States v. State of Louisiana et al. No. 9 Original, | | 38.125 |

| BL.M or State Lease No. | Effective Date | Description | POGI | PPC |
|----------------------------|-------------------|--|----------|----------|
| | | Said three mile lin sut in the June, 1975, decree upreme Court. All bearing used on Louisiana Lamba Coordinate System (South Zone). | | |
| OCS-G-4144 | 9- 1-79 | That portion of Block 3 which is more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (United States vs. Louisiana, 422 U.S. 13) Sabine Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 12. | | 41.66663 |
| OCS-G-4379 | 11- 1-80 | All of Block 53, West Cameron Area, as shown on OCS Leasing Map, Louisiana Map No. 1 | | 41.66665 |
| OCS-G-2839 | 12- 1-74 | All of Block 352, West Cameron Area, West Addition, OCS Official Leasing Map, Louisiana Map No. 1A | 8.34372 | 20.83333 |
| OCS-G-2224 | 2- 1-73 | All of Block 532. West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.22499 |
| OCS-G-2225 | 2- 1-73 | All of Block 533, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.22499 |
| OCS-G-32S4 | 8- 1-75 | All of Block 563, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 1B | 25.03113 | 20.83332 |
| OCS-G-2436 | 8- 1-73 | All of Block 586, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 16.68742 | |
| OCS-G-2021 | 2- 1-71 | All of Block 587, West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1 B | 22.80504 | |
| OCS-G-2850 | 12- 1-74 | All of Block 609, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.55555 |
| DCS-G-2559 | 5- 1-74 | All of Block 617. West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.55555 |

| BLM or State | Effective | | Net Reven | ue Interest |
|--------------|-----------|---|-----------|-------------|
| Lease Ne. | Date | Description | POG1 | PPC |
| OCS-G-2860 | 12- 1-74 | Ail of Block 237, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 2.22499 | 5.55555 |
| OCS-G-2045 | 1- 1-71 | All of Block 270, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.68741 | |
| UUS-G-2062 | 2- 1-71 | All of Block 334, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| € CS-G-2439 | 8- 1-73 | All of Block 335, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| OCS-G-2078 | 2- 1-71 | All of Block 228, Vermilion Area, OCS Official Leasing Map, Louisiana Map No. 3 | 17.9273 | |
| OCS-G-2882 | 12- 1-74 | All of Block 125, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5.55555 |
| C 1S-G-2883 | 12- 1-74 | All of Block 127, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5.55555 |
| OCS-G-2587 | 5- 1-74 | All of Block 128, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.39995 | 5.55555 |
| OCS-G-0479 | 12- 1-54 | All of Block 53. Eugene Island Area as shown on Official Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisians Offshore Operations) | | 83.33333 |
| OCS-G-0478 | 1- 1-55 | E 14 of Block 116, Eugene Island Area as shown on Official Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisiana Offshore Operations) | | 41.66665 |
| OCS-G-2102 | 2- 1-71 | All of Block 256, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 23.1054 | |
| OCS-G-2900 | 12- 1-74 | All of Block 261, Eugene Island Arc OCS Official Leasing Map, Louisiana Map No. 4 | 5.56247 | 13.88888 |
| OCS-G-3156 | 7- 1-75 | All of Block 262. Eugene Island Area, as shown on OCS Official Leasing Map, Louisiana Map No. 4 | 5.56247 | 13.88888 |

| BLM or State | Effective | | | ue Interest |
|--------------|-----------|--|-----------|-------------|
| Lease No. | Date | Description | POCI | PPC |
| OCS-G-2104 | 2- 1-71 | All of Block 205, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 6.76497 | |
| OCS-G-2607 | 5- 1-74 | All of Block 312, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 2.22499 | 30.00001 |
| OCS-G-2115 | 1- 1-71 | All of Block 330, Eugene Island Area, South Addition, OCS Official Leasing Map No. 4A | . 6.68741 | |
| OCS-G-2317 | 2- 1-73 | All of Block 333, Eugene Island Area, South Addition, OCS Official Leasing Map No. 4A | 2.22499 | 22.22221 |
| OCS-G-3332 | 3- 1-76 | All of Block 337, Eugene Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | 16.68741 | |
| OCS-G-3409 | 1- 1-77 | All of Block 351, Eugene Island Area, South Addition, as shown on OCS Ocicial Leasing Map, Louisiana Map No. 4A | | 20.83333 |
| OCS-G-0823 | 5- 1-60 | All of Block 186, brip Shoal Area, as shown on Official Leasing Map, Locisiana Map No. 7. Outer Continue al Shelf Leasing Map (Luisiana Offshore Operations) | | 28.33332 |
| OCS-G-1525 | 7- 1-67 | All of Block 222, Ship Sheal Area, OCS Official Leasing Map, Louisiana Map No. 5 | | 15.83332 |
| OCS-G-1984 | 9- 1-70 | N 1 . Block 225, Shin Shoal Area, OCS Official Leasing Map, Louisiana Map No. 5 | | 19.79165 |
| OCS-G-2177 | 11- 1-72 | All of Block 49 South Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 9 | 3.33748 | 8.33333 |
| OCS-G-3416 | 1- 1-77 | That portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court. June 16, 1975 (U.S. vs. Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971) as shown on OCS Official Leasing Map. Louisiana Map No. 9 | 6.67497 | 16.66667 |

| BLM or State Lease No. | Effective Date | Description | POGI | PPC |
|---------------------------|-------------------|---|---------|----------|
| St. Ls. 6310 | 5-13-74 | Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish. Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and being more par- ticularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Louisiana | 5.92738 | 14.80 |
| OCS-G-2185 | 10- 1-72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court entered December 13, 1965 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288), South Pass Area, South and East Addition, as shown on OCS Official Leasing Map. Louisiana Map No. 9A | 5.00622 | 12.49999 |
| OCS-G-310° | 7- 1-75 | Block 72. Main Pass Area, that portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 7.54441 | 18.83758 |
| OCS-G-3417 | 1- 1-77 | That portion of Main Pass Blocks 72 and 74 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (U.S. vs Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971) as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 7.54441 | 18.83758 |
| OCS.G-2947 | 12- 1-74 | All of Block 73, Main Pass Area, OCS Official Leasing Map, Louisiana Map No. 10. That portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)) | 7.54441 | 15.83758 |

| BLM or State Lease No. | Effective Date | Description | Net Revenue Interest | |
|---------------------------|-------------------|--|----------------------|---------|
| | | | POGI | PPC |
| OCS-G-2193 | 10- 1-72 | All of Block 140, Main Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 3.33748 | 8.33333 |
| OCS-G-3206 | 7- 1-75 | All of Block N663E63, Mobile South No. 2, as shown on OCS Official Leasing Map, NH-16-10 (Also known as Block 63), Mississippi Canyon Area, as shown on OCS Official Protraction Diagram NH-16-10 | 3.33748 | 8.33333 |
| OCS-G-3500 | 8- 1-77 | All of Block 253, West Cameron Area, as shown on OCS Official Leasing Map, Louisiana Map No. 1 | 10.01245 | 25.0000 |
| OCS-G-3785 | 6- 1-78 | All of Block 372, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 12.4999 | 12.4999 |

THIS INSTRUMENT IS A MORTGAGE OF AN INTEREST IN BOTH REAL AND PERSONAL PROPERTY, AND IS, AMONG OTHER THINGS, A MORTGAGE OF CHATTELS AND A SECURITY AGREEMENT.

Mortgage, Deed of Trust and Assignment of Production

Dated July 16, 1981

NEW DUNSTAN, INC.

CT

F. D. THOMPSON

AND

W. T. FOX III.

Trustees

AND

CITIBANK, N.A.,

As Agent

Return recorded counterparts to:

SHEARMAN & STERLING 153 E. 53rd Street New York, New York 10022 Attention: Michael A. Ross, Esq. This instrument was prepared by:

Frank W. R. Hubert, Jr. 3000 One Shell Plaza Houston, Texas 77002

TABLE OF CONTENTS

| Articl | <u>e</u> | | | |
|--------|--|----|--|--|
| I. | Indebtedness Secured | 3 | | |
| П. | Representations, Warranties and Covenants | | | |
| III. | Assignment of Production | | | |
| IV. | Defeasance | 20 | | |
| V. | Remedies in Event of Default | | | |
| VI. | Special Mortgage of Property in Louisiana | | | |
| VII. | Concerning the Trustees | | | |
| VIII. | Miscellaneous Provisions | 38 | | |
| | Exhibit A Credit Agreement | | | |
| | Exhibit B Instrument of Conveyance of Production Payment | | | |
| | Exhibit C Supplemental Mortgage. Deed of Trust and Assignment of Production | | | |
| | Exhibit D Resolutions of the Board of Directors of the Company | | | |

THIS MORTGAGE, DEED OF TRUST AND ASSIGNMENT OF PRODUCTION dated July 16, 1981, from New Dunstan, Inc., a Delaware non-profit corporation (the "Company"), to F. D. Thompson of Houston, Texas and W. T. Fox II; of Houston, Texas (herein collectively the "Trustees" and individually a "Trustee"), and CITIBANK, N.A., a national banking association (the "Agent"),

WITNESSETH :

Whereas, concurrently with the execution of this instrument, the Company is purchasing from Pennzoil Company, a Delaware corporation, Pennzoil Producing Company, a Delaware corporation, Pennzoil Oil & Gas, Inc., a Delaware corporation, and Duval Corporation, a Delaware corporation (the "Grantors"), the Production Payment defined and described in that certain Instrument of Conveyance of Production Payment dated July 16, 1981, from the Grantors to the Company (said Instrument of Conveyance of Production Payment being herein called the "Conveyance" and said Production Payment, as so defined and described, being herein called the "Production Payment"), having an initial primary sum of \$10,000,000; and

WHEREAS, the aforesaid primary sum of the Production Payment may be increased to an amount not to exceed \$2,500,000,000 in the manner provided in the Conveyance (the primary sum, as the same may at any time be so increased, being herein called the "Primary Sum"); and

Whereas, the Production Payment is dischargeable out of and consists of the fixed and varying percentages specified in the Conveyance (the applicable percentages at any time being herein and in the Conveyance called the "Production Payment Percentages") of the oil, gas and other gaseous ar invide hydrocarbons (herein and in the Conveyance called "Hydror pons") and, in the event the Supplemental Instrument of Convertice of Production Payment is executed and delivered, sulphur (herein and in the Conveyance together with Hydrocarbons, called "Mine Is") in and under, and which may be produced, mined and saved from and which shall accrue or be attributable to the properties and interests described and referred to in the Conveyance

and Exhibit A thereto (said properties and interests being herein and in the Conveyance called the "Subject Laterests" and the Minerals therein and thereunder, and which may be so produced, mined and saved therefrom, and which shall accrue or be attributable thereto, being herein and in the Conveyance called the "Subject Minerals") (the Production Payment Percentages of the Subject Minerals being herein and in the Conveyance called the "Production Payment Minerals"); and

Whereas, the Company has entered into a Credit Agreement dated as of July 16, 1981 (said Credit Agreement, as the same may at any time be amended and modified and in effect, being herein called the "Credit Agreement") with the banks named therein (the "Banks") and the Agent, providing for the making of advances ("Advances") to which the Domestic Rate or the Eurodollar Rate may be applicable by the Banks to the Company from time to time in an aggregate principal amount not to exceed \$2,500,000,000 for the purpose of enabling the Company to make installment payments on account of the purchase price of the Production Payment under the Conveyance; and

Whereas, in order to evidence its obligation to repay each and all of the Advances of each Bank to which the Domestic Rate may be applicable, the Company has executed its 25 promissory notes dated July 16, 1981 each substantially in the form of Exhibit A to the Credit Agreement, of which one is payable to the order of each Bank listed on the signature page of the Credit Agreement and is in the stated principal sum equal of the Total Commitment set forth opposite the name of such Bank of such signature page (such promissory notes of the Company, and any promissory notes issued in replacement thereof being called the "A Notes"); and

Whereas, in order to evidence its obligation to repay each and all of the Advances of each Bank to which the Eurodollar Rate may be applicable, the Company has executed its 25 promissory notes dated July 16, 1981, each substantially in the form of Exhibit B to the Credit Agreement, of which one is payable to the order of each Bank listed on the signature page of the Credit Agreement and is in the stated principal sum equal to the Total Commitment set forth opposite the name of such Bank on such signature page (such promissory notes of the Company, and any promissory notes issued in replacement thereof

being called the "B Notes", and together with the A Notes, collectively the "Notes" and individually a "Note"; and

Whereas, the actual Total Commitment (as defined in the Credit Agreement) of each Bank is the stated principal amount of each Note payable to the order of such Bank, and not the aggregate stated principal amount of all Notes payable to the order of such Bank; and

Whereas, the form of the Credit Agreem int, with the forms of the Notes attached thereto, is attached hereto as Exhibit A, and hereby made a part nereof for all purposes; and attached hereto as Exhibit B and made a part hereof for all purposes is the form of the Conveyance, together with all of the property descriptions included in Exhibit A to the Conveyance as te that counterpart to be recorded in Terrebonne Parish, Louisiana and Culberson County, Texas and in the files maintained by the New Orleans Outer Continental Shelf Office of the Bureau of Land Management, U.S. Department of the Interior, as to Lease OCS-G-2115; and, as to those counterparts to be recorded in all + cording jurisdictions (country or parish) other than in Terrebonne Parish, Louisiana and Culberson County, Texas, there is omitted from Exhibit A to the Conveyance attached to such counterparts of this Mortgage, the descriptions of property located in recording jurisdictions other than the jurisdiction (county or parish) in which the particular counterpart is to be recorded; and

Whereas, all conditions necessary to authorize the execution and delivery of this Mortgage, Deed of Trust and Assignment (as the same may at any time be amended and modified and in effect, being herein called the "Mortgage") and the issuance of the Notes have been complied with; and

Whereas, for all purposes of this Mortgage, unless the context otherwise requires or unless otherwise defined or stated herein:

- (a) All terms in this Mortgage which are defined in the Conveyance have the respective meanings assigned to them by the Conveyance.
- (b) All reference herein to "Articles", "Sections", "Subsections", "Clauses" or other subdivisions refer to the corresponding Articles, Sections, Subsections, Clauses and other subdivisions of this instrument; and the words "this instrument", "this Mortgage",

"herein", "hereof", "hereby", "hereunder" and words of similarimport refer to this instrument as a whole and not to any particular Article, Section, Subsection, Clause or other subdivision hereof.

(c) The term "Event of Default" has the meaning stated in Section 5.1 hereof.

Now, Therefore, to secure the payment of the indebtedness hereinafter in Section 1.1 hereof described and the performance of the covenants herein and in the Notes contained, and in consideration of the loans made to the Company by the Banks pursuant to the Credit Agreement, the Company by these presents does grant, bargain, sell, mortgage, create a security interest in, pledge, assign, transfer, convey, set over and deliver unto the Trustees and their successors in this trust and unto the ssigns, the Production Payment, including all the Company's interest in the Production Payment Minerals and the proceeds thereof, and all rights, titles, interests, estates, remedies, powers and privileges vested in the Company as the owner of the Production Payment or which the Company now has or may become entitled to under or by virtue of the terms and provisions of the Conveyance (the Production Payment, and all such rights, titles, interests, estates, remedies, powers and privileges, being herein collectively called the "Mortgaged Property").

Subject, However, to the pledge and assignment of production and other amounts contained in Article III hereof, but only insofar and so long as the same are not inoperative under the provisions of Section 3.3 hereof, and to the liens, charges and encumbrances set forth or referred to in Subsection II of Section 2.1 hereof. Reference is hereby made to Exhibit B attached hereto for a more complete description and statement of the Subject Interests, the Production Payment Minerals and the other rights, titles, interests, estates, remedies, powers and privileges constituting and appertaining to the Production Payment.

To Have and To Hold the Mortgaged Property, together with all rights, estates, powers and privileges appurtenant or incident thereto, unto the Trustees, their successors in this trust and their assigns forever. BUT IN TRUST, NEVERTHELESS, for the benefit and security of the holders of the Notes and the other indebtedness secured hereby and upon the trusts and subject to the covenants and conditions hereinafter set forth.

ARTICLE I.

INDEBTEDNESS SECURED

- Section 1.1. This instrument is made by the Company irrevocably in trust, with power of sale, to secure and enforce its indebtedness below described:
 - A. The indebtedness of the Company evidenced by the Notes, including any extensions or renewals thereof, in an aggregate principal amount not to exceed \$2,500,000,000 with interest as therein provided (such Notes having a final maturity date of February 1, 1990 unless the Supplemental Instrument of Conveyance of Production Payment is executed and delivered by the Grantors in which case the final maturity date of the Notes shall be June 1, 1990).
 - B. All other indebtedness of the Company arising pursuant to the provisions of the Credit Agreement and this Mortgage (including, without limitation, all amounts payable by the Company under the Credit Agreement on account of commitment fees).
 - C. All costs of enforcing the Notes, the Credit Agreement and this Mortgage.
 - D. All fees and expenses payable to the Trustees under this Mortgage and the Credit Agreement.

The maximum indebtedness of the Company secured hereby in addition to that set forth in Section 1.1 A hereof shall not exceed 75% of the aggregate principal amount of the Notes.

ARTICLE II.

REPRESENTATIONS, WARRANTIES AND COVENANTS

Section 2.1. The Company hereby represents and warrants to the Trustees, the Banks and the holders from time to time of the Notes as follows:

- A. The Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware and has all requisite power and authority, corporate and otherwise, to conduct its business and to own its properties (including the Mortgaged Property), to enter into the Credit Agreement and to borrow thereunder and to execute and deliver the Credit Agreement, the Conveyance, this Mortgage and the Notes and to perform its obligations under each of the foregoing documents and instruments except to the extent that it may be necessary or advisable to qualify to do business in certain states in which the Mortgaged Property is located.
- B. The making and performance by the Company of the Credit Agreement, the Conveyance, this Mortgage and the Notes and the purchase and ownership by the Company of the Mortgaged Property have been duly authorized by all necessary corporate action (including any necessary shareholde: action) and do not and will not (i) violate any provision of any law, rule, regulation, order, writ, judgment, decree, determination or award presently in effect having applicability to the Company or of the Certificate of Incorporation or By-Laws of the Company, or (ii) result in a breach of or constitute a default under any indenture or bank loan or credit agreement or any other agreement or instrument to which the Company is a party or by which it or its property may be presently bound or affected, or (iii) result in or require the creation or imposition of any mortgage, lien, pledge, security interest or other charge or other encumbrance in, upon or of any of its properties or assets under any such indenture or bank loan or credit agreement or other agreement or instrument, and all consents and approvals necessary to permit the Company to execute and deliver, and to perform its obligations under, the Credit Agreement, the Conveyance, this Mortgage and the Notes and to purchase the Mortgaged Property have been obtained, and the Company is not in violation or default under any suc'. order, writ, judgment, decree, determination, award, indenture, agreement or instrument.
- C. (i) Assuming the due authorization and execution thereof by each other necessary party thereto, the Credit Agreement, the Conveyance, this Mortgage and the Notes, when issued by the Com-

pany for value, constitute the legal, valid and birding obligations of the Company enforceable against the Company in accordance with their terms and (ii) subject to the accomp' binent of all requisite filings and recordings in the States of Alabama, Arkansas, Louisiana, Mississippi, Montana, New Mexico, New York, North Dakota, Pennsylvania, Texas, West Virginia and Wyoming with respect to the property covered thereby prior to the intervention of rights of third parties therein, this Moregage constitutes a legal, valid and effective first mortgage lien on the Mortgaged Property and a legal, valid and effective first security interest in and assignment of Production Payment Minerals and the proceeds of sale thereof.

- D. The balance sheet of the Company as of July 16, 1981 (certified by the chief financial officer of the Company), copies of which have been furnished to the Banks, correctly and fairly sets forth the financial condition of the Company at said date, and since said date there has been no material adverse change in its condition, financial or otherwise.
- E. There are no actions, suits or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company or any of its properties (including the Mortgaged Property) before any court or governmental department, commissior, board, bureau, agency or instrumentality, domestic or foreign, which, if determined adversely to the Company, would have a material adverse effect on the financial condition, properties (including the Mortgaged Property) or a tions of the Company or its ability to perform its obligations ander or under the Credit Agreement, the Conveyance, this Mortgage or the Notes.
- F. No authorization, consent, approval, license or exemption of, and no filing or registration with, any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, is or was necessary to the valid execution, delivery and performance by the Company of the Credit Agreement, the Conveyance, this Mortgage or the Notes, except to the extent that it may be necessary or advisable to qualify to do business in certain states in which the Mortgaged Property is located.

- G. The Company has the legal right and authority to grant, bargain, sell, mortgage, create a security interest in pledge, assign, transfer, convey, set over and deliver the Mortgaged Property, and all rights, titles, interests, estates, remedies, powers and privileges appurtenant or incident thereto, in the manner and form hereby done or intended; the entire amount of the Production Payment initially sold to the Company is, on the date of this Mortgage, outstanding and undischarged; the Company has received, on the date of this Mortgage, nothing whatsoever towards the satisfaction of the Production Payment; and the Production Payment is not subject to any set-off, counterclaim, recoupment or defense.
- H. The Company has good and indefeasible title to the Mortgaged Property initially sold to it free and clear of all lieus, security interests, charges and encumbrances of every character other than (a) the lieu and security interest of this Mortgage and (b) such lieus, charges and encumbrances as are set forth in Exhibit A to the Conveyance or as are permitted by subparagraph (f) of Paragraph 14 of the Conveyance to the extent the same are applicable to the Production Payment.
- I. There is no effective financing statement or other instrument similar in effect on file against the Company in any of the States of Alabama, Arkansas, Louisiana, Mississippi, Montana, New Mexico, New York, North Dakota, Pennsylvania, Texas, West Virginia and Wyoming covering the Mortgaged Property or any part thereof, or the proceeds of the Morte red Property or any part thereof, other than these related to as Mortgage and the Conveyance.
- Section 2.2. The Company covenants and agrees with the Trustees, the Banks and the holders from time to time of the Notes that, so long as any indebtedness secured hereby remains unpaid, unless (except as to Subsection D of this Section) the holders of not less than 6624% in aggregate unpaid principal amount of the Notes shall have a cherwise consented in writing:
 - A. The Company will maintain in good standing existence, rights, privileges and franchises in the State and will qualify and remain due, qualified to do business. I own

property in each jurisdiction in which any of the Mortgaged Property shall be located if such qualification is necessary in view of its business and operations, and will promptly pay all frenchise, income and other taxes owning by it; and the Company will, as and when necessary, qualify and pay franchise and other taxes (subject to the right of the Grantors under the Conveyance to contest such taxes in good faith) in any jurisdiction in which any of the Subject Interests shall be located if required by the law of such jurisdiction for the full exercise of its rights or the full performance of its obligations under this Mortgage or the Conveyance.

- P. The Company will promptly pay or discharge, and cause the same promptly to be paid or discharged, when the same shall become due (or, as to any thereof which are being contested in good faith, promptly following final determination of such contest), all taxes, assessments and governmental charges or levies imposed upon it or upon its income or profits or any property belonging to it, including all ad valorem, severance, gross income, gross receip gross profits, cross production, drilling, occupation, excise, so franchise, income and other taxes and assessments of any kind whatsoever (including, without limitation, any excise tax imposed on any windfall profit from taxable crude oil by the Crude Oil Windfall Profit Tax of 1980 and any amendments or supplements thereto or tax enacted in substitution therefor and any interest and penalties payable thereon or with respect thereto), imposed upon or assessed with respect to or measured by or charged against or attributable to the Mortgaged Property (including the interest therein of the Trustees or of the holders of the Notes or other indebtedness secured hereby), the Production Payment, the Production Payment Minerals or the proceeds of sale thereof or the drilling for or extraction of any Production Payment Minerals or the acquisition, ownership, sale, assignment, transfer or other Asposition by the Company in whole or in part of the Production yment or the Production Payment Minerals and will maintain and preserve the lien and security interest created by this Morigage.
 - C. The Company will, on request of any Trustee or of the of any of the Notes, promptly correct any defect, error or

omission which may be discovered in the contents of this instrument or in the execution hereof and will execute, acknowledge and deliver such further instruments and do such further nets and things as may be necessary or as may be reasonably requested by the Trustees or the holder of any of the Notes to carry out more effectively the purposes of this Mortgage and to subject to the lien and security interest hereby created any of the Mortgaged Property or the rights or interests covered or intended to be covered nereby, and to perfect and maintain—ach lien and security interest.

- D. The Company will duly and punctually pay each and every obligation owing hereunder and under the Credit Agreement and on account of each Note in accordance with the terms hereof and thereof, unless the holders of all the Notes shall have otherwise consented in writing.
- E. The Company will promptly and (insofar as not contrary to applicable law) at its own expense record and re-record, file and re-file and register and re-register the Conveyance, this Mortgage, any financing or continuation statements and every other instrument additional or supplemental hereto—thereto that shall be required by law in order to perfect and maintain the validity and effectiveness of the Conveyance and the lien and security interest intended to be created hereby, in such manner and places and within such times as may be necessary to perfect and maintain such lien and security interest and preserve and protect the rights and remedies of the Trustees and of the holders of the Notes, and will furnish satisfactory evidence of every such recording, filing and registration to the Trustees and the holders of the Notes.
- F. The Company will keep proper books of record and account in which complete and correct entries will be made of all of its business and financial transactions, such entries to be made in accordance with generally accepted a counting principles consistently applied in the case of financial transactions. All such books of record and account shall be kept, and the chief place of business of the Company's lall be located, in the State of Delaware at Accempany's address stated at the end of this Mortgage or at such

other place within the continental limits of the United States as the Trustes and the holders of 66°3% or more in aggregate unpaid principal amount of the Notes shall have approved in writing prior to the change of the place of keeping of books of record and account or the location of its chief place of business.

G. The Company will furnish to the Trustees, the Agent and each of the holders of the Notes (a) promptly after the Company shall have obtained actual knowledge of the occurrence of an Event of Default under the Credit Agreement or under this Mortgage, and/or an event which with giving of notice o. the lapse of time (or both) would become uch an Event of Default, a statement of the chief financial officer of the Company setting forth details of such Event of Default or event, (b) within ninety (90) days after the end of each fiscal year of the Company, the balance sheet of the Company as of the end of such fiscal year, and the statement of income and retained earnings of the Company for such fiscal year, certified by an authorized financial officer of the Company or, if requested by the Trustees or the holders of not less than 6624% in as regate principal amount of the Notes, by independent public accountants of re-organized standing acceptable to the Trustees or the holders of not less than 5624% in aggregate principal amount of the Notes, (c) within thirty (30) days after the end of each of the first three quarters of each such fiscal year, the balance sheet of the Company as of the end of such quarter and statements of income and retained earnings of the Company for the period commencing at the end of the previous fiscal year and ending with the end of such quarter, certified by an authorized financial officer of the Company, (d) promptly following receipt thereof by the Company, copies of all information and each report or other statement or document furnished to the Company pursuant to the provisions of Paragraph 19 of the Conveyance, and (e) such other information respecting the business affairs, financial condition and/or operations of the Company as the Agent may from time to time reasonable request.

H. The Company will not,

(1) amend, modify, supplement or otherwise revise the Conveyance or consent to or permit any such amendment, modification, supplement or other revision, except by execution and delivery of the Supplemental Instrument of Conveyance of Production Payment and pursuant to Subsection C of this Section 2.2:

- (2) except with respect to actions as to which consent of the Company is not required by the terms of the Conveyance, consent to or permit any surrender, abandonment, release, sale, conveyance, assignment, lease or sublease, farmout or any participation in any pooling, communitization or unitization, by the Grantors under the Conveyance of all or any part of the Subject Interests;
- (3) surrender, abandon, release, sell, convey, assign, lease or sublease the Mortgaged Property or any part thereof;
- (4) liquidate or merge into or consolidate with any other corporation or permit any other corporation to be merged into the Company;
- (5) sell, lease, assign, transfer or otherwise dispose of all or a substantial portion of its assets, including its accounts receivable, except as contemplated by the Conveyance and this Mortgage;
- (6) issue or sell cry shares of its capital stock having ordinary voting pewar the election of its Board of Directors or other governing body; or
- (7) declare or pay any dividends, purchase, redeem or otherwise acquire for value any of its capital stock now or hereafter outstanding, return any capital to stockholders or make any distribution of assets to stockholders, except that the Company may (i) declare and pay dividends to its stockholders in an amount not to exceed the amount received by it pursuant to Paragraph 4(c) of the Conveyance; provided, however, that no such dividends or distributions shall be permitted pursuant to this clause (i) if an Event of Default shall have occurred and be continuing and (ii) declare and pay dividends to its stockholders in amounts equal to interest and dividends, if any, received on the securities referred to in Subsection O of this Section 2.2; provided, however, tha'

no such dividends or distributions shall be permitted pursuant to this clause (ii) which shall reduce the current market value of the Company's assets (other than the Production Payment) below the amount set forth in Subsection O of this Section 2.2.

- I. The Company will promotly notify the Trustees and the Banks in writing of any default, of which it has knowledge, in the observance or performance of any term, covenant, agreement or condition on the Grantors' part to be performed or observed under the Conveyance.
- J. The Company will, on the request of the Trustees or of the holders of not less than 66°3% in aggregate unpaid principal amount of the Notes, and at the expense of the Company, promptly take all such action as may be so requested to enforce or secure the observance or performance of any term, covenant, agreement or condition of the Conveyance to be observed or performed by the Grantors thereunder, or to exercise any of its rights, remedies, powers and privileges under the Conveyance, all in accordance with the terms thereof, provided, however, that the Trustees shall not be obligated to make any such request to the Company under this Subsection J unless directed to do so in writing by the holders of not less than 66°3% in aggregate unpaid principal amount of the Notes.
- K. If the title or rights—f the Trustees to the Mortgaged Property or any part thereof or the lien of or security interest created by this Mortgage shall be in danger, or shall be attacked directly or indirectly, or if any legal proceedings are instituted against the Company, the Trustees or the holders of the Notes with respect thereto, the Company will promptly give written notice thereof to the Trustees and the holders of the Notes and at its own cost and expense will undertake diligently to cure any defect that may have developed or may be claimed and will take all necessary and proper steps for the protection and defense thereof and will take such action as is reasonably appropriate to the detense of any such legal proceedings, including, but not limited to, the employment of counsel, the prosecution and defense of litigation and, if approved by the holders of not less than 66% 3%

in aggregate unpaid principal amount of the Notes, the compromise or release and discharge of any adverse claims made. The Trustees shall not be obligated to exercise any rights under this Subsection K unless directed to do so in writing by the holders of not less than 66% in aggregate unpaid principal amount of the Notes.

- L. Except as permitted by Subsection O of this Section 2.2. the Company will not conduct any activity unrelated to the ownership of the Mortgaged Property or to the exercise of any rights appertaining thereto and, in any event, will not open or maintain an office in any state other than the State of Delaware. The Company will not qualify to do business as a foreign corporation in any state other than the state or states the laws of which shall require such qualification for the full exercise of the Company's rights or the full performance of the Company's obligations under this Mortgage or the Conveyance.
- M. The Company will not create, assume or incur, or in any manner be or become liable in respect of, any indebtedness, whether current or funded, direct or contingent, other than
 - (1) the Notes and other indebtedness of the Company secured hereby;
 - (2) taxes, assessments and governmental charges or levies not due and delinquent or the validity of which is currently being contested in good faith by appropriate proceedings; and
 - (3) current accounts, payable or accrued, necessary to the conduct of the ordinary course of business of the Company, provided that all such indebtedness shall from time to time be promptly paid and discharged when due.

For the purposes of this Subsection M, the term "indebtedness" shall mean (i) any obligation, by whomsoever incurred, properly shown on the liability side of a balance sheet in accordance with generally accepted accounting principles (other than items properly shown in the capital account of a corporate balance sheet), (ii) any obligation secured by a mortgage, conditional sales agreement or other title retention or security instrument, whether or not any person has personal responsibility to discharge the same, and (iii) any obligation for the payment of rent or hire for property of any kind whatsoever, whether real, personal or mixed.

N. The Company will not create or assume, or suffer to be created or assumed or to any mortgage, pledge, security interest, charge or encambrate. Any kind upon or in any of its property of any character, whether owned at the date hereof or hereafter acquired, as security for any indebtedness or acquire or agree to acquire any property subject to any conditional sale agreement or other title retention agreement, excluding, however, from the operation of this Subsection

- (1) the lien of this Mortgage;
- (2) other liens securing indebtedness of the kinds permitted by clauses (1) and (2) of Subsection M of this Section 2.2; and
- (3) such liens, charges and encumbrances as are set forth in Exhibit A to the Conveyance or are permitted by Paragraph 14(f) of the Conveyance to the extent the same are applicable to the Production Payment.
- O. The Company will at all times maintain assets (other than the Production Payment) in the form of cash or readily marketable securities, of a current market value or principal amount of not less than \$10,000 in the aggregate.
- P. The Company will not file or join in the filing of any consolidated federal or state income or franchise iax return with any other corporation or entity.
- Q. Upon receipt of evidence satisfactory to the Company of the loss, theft, destruction or mutilation of any Note and, in the case of any such loss, theft or destruction, upon delivery of a bond of indemnity (or in the case of any Bank, upon written indemnification Ly such Bank) satisfactory to the Company, or, in the case of any such mutilation, upon surrender and cancellation of such Note, the Company will issue, in replacement of such lost, stolen, destroyed or mutilated Note, a new Note of like tenor, datal the date to which interest has been paid on such Note and not d to show any payments or prepayments of principal thereof.

- R. Upon execution and delivery by the Grantors of the Supplemental Instrument of Conveyance of Production Payment, the Company will execute and deliver to the Trustees the Supplemental Mortgage, Deed of Trust and Assignment of Production substantially in the form attached hereto as Exhibit C;
- S. The Company will not assume, guarantee, endorse or otherwise be or become directly or indirectly liable in respect of or in effect responsible for (whether by guarantee or by agreement to purchase securities, to contribute capital or to make lease payments or otherwise) any obligation of any other person, except by endorsement of negotiable instruments for deposit or collection or similar transactions necessary to the conduct of the ordinary business of the Company.
- T. The Company will not make any loan or advance to any person.
- U. Subject to the provisions of Subsection H of Section 2.1 hereof, the Company hereby binds itself to warrant and forever defend unto the Trustees, their successors in this trust and their assigns, the Production Payment against every person whomsoever lawfully claiming or to claim the same or any part thereof. This Mortgage is made with full substitution and subrogation of the Trustees in and to all covenants and warranties heretofore given or made by others in respect of the Production Payment or the Subject Interests, or any part thereof.
- Section 2.3. To the full extent that it lawfully may, the Company hereby agrees that the Trustees or the holders of not less than 66% in aggregate unpaid principal amount of the Notes shall be entitled at any time and from time to time (i) to exercise all of the rights, remedies, powers and privileges vested in the Company as Grantee under the Conveyance and by virtue of its ownership of the Production I ayment, (ii) to give or withhold all consents, requests, directions entices, approvals and waivers required or permitted by which the Company as such Grantee would otherwise the Co or withhold thereunder or pursuant thereto and (iii) he cor. use to be performed any act which the Company is to per pe orm under Subsections J and K of Section 2.2 hereof rec . . in the manner and extent of the performance by the Com-BI

pany of any such act. The Company will, upon request of the Trustees or the holders of not less than 66%. in aggregate unpaid principal amount of the Notes, execute and deliver to such person or persons as may be designated by the Trustees or such holders appropriate powers of attorney to act for and on behalf of the Company in all transactions of the Company with an agency, department or subdivision of any state or of the United States relating to the Mortgaged Property or any part thereof. Except in connection with enforcement of the Trustees' right to be paid by the Company any sums to be paid to them by the Company under this Mortgage, the Trustees shall take any action which they are authorized to take pursuant to the provisions of this Section 2.3 only in accordance with the written directions of the holders of not less than 66%. in aggregate unpaid principal amount of the Notes, and the Trustees shall be fully protected in so doing.

Section 2.4. The Company acroes that, if it fails to perform any act which it is required to perform hereunder or to pay any money which it is required to pay hereunder, the Trustees or the holder of any Note may (subject to the provisions of Section 2.3 with respect to performance of the Company's obligations under Subsection J or K of Section 2.2), but shall not be obligated to, perform or cause to be performed such act and may pay such money. The Trustees shall not be obligated to exercise any rights under this Section 2.4 unless directed to do so in writing by the holders of not less than 66%% in aggregate unpaid principal amount of the Notes.

Section 2.5. The Company agrees that any Trustee and any holder of a Note, at its own expense, shall have the right to visit and inspect any of the properties belonging to the Company, to examine and make copies of and take abstracts from its books and records and to discuss its affairs, finances or accounts with the Company's officers, all at such times and intervals as the Trustees or such holder may reasonably request and shall have the right to make all inspections, and to receive all reports, certificates, statements and other documents and all other information which the Company shall be entitled to make or receive under Paragraphs 15, 16 and 19 of the Conveyance, subject, however, to the limitations and conditions provided in said Paragraphs 15, 16 and 19.

ARTICLE III.

ASSIGNMENT OF PRODUCTION

Section 3.1. The Company, to the extent in this Article provided, effective as of 7 o'clock A.M., local time in effect at the location of each of the Subject Interests, on July 1, 1981, hereby bargains, sells, assigns, transfers, conveys, pledges, sets over and delivers unto the Agent, its successors and assigns, all the Company's right, titles, interests, estates, remedies, powers and privileges in and with respect to the production of Production Payment Minerals, together with all proceeds derived from the sale of such Minerals, including all moneys due and to become due under any of the Sales Contracts and all of the Company's rights to receive all amounts payable to or otherwise receivable by the Company on account of payments pursuant to Paragraphs 14(c) and 25 of the Conveyance. All parties producing, purchasing, taking, processing or receiving Minerals from any properties from which the Production Payment is dischargeable, or having in their possession such Minerals, or proceeds therefrom, or making any payment pursuant to Paragraphs 14(c) and 25 of the Conveyance, for which they or others are accountable to the Agent by virtue of the provisions of this Article, are authorized and directed to treat and regard the Agent as the assignee and transferee of the Company and entitled in the piace and stead of the Company to receive such proceeds accruing or attributable to the Production Payment, and said parties and each of them shall be fully protected in so treating and regarding the Agent and shall be under no obligation to see to the application by the Agent of any such proceeds received by it. Without in any way limiting the effectiveners of the authorization and direction in the next preceding sentence, if the Company shall receive any such proceeds which under this instrument are receivable by the Agent, the Company will hold the same in timet and will remit such proceeds or amounts, or cause such proceeds or amounts to be remitted, immediately to the Agent. The Company will, at the request of the Agent, furnish it with or obtain for it the names of all parties purchasing or receiving Subject Minerals from any properties from which the Production Payment is dischargeable and the names of all parties having in their possession such Minerals, or proceeds therefrom, and the Company will execute and deliver any and all instruments that may be requested by the Agent for the pureffectuating the assignment hereby made by the Company and the payment to the Agent of the proceeds so assigned.

Section 3.2. All monies received by the Agent pursuant to the assignment made by the Company in Section 3.1 hereof which are deemed to be proceeds of Production Payment Minerals under Paragraph 4 of the Conveyance shall be held by the Agent as security for the payment of the Notes and shall be applied as provided in Paragraphs First through Firth of this Section. Monies so received by the Agent pursuant to Section 3.1 hereof up to 11:00 A.M. (local time in effect in the City of New York) on any Application Date (commencing with September 1, 1981) and not theretofore applied pursuant to this Section 3.2, shall be deemed to have been received and shall be applied as follows:

FIRST: There shall be paid to the holders of the Notes the aggregate amount then applied with respect to the Company pursuant to, and in the order of priority specified in subparagraphs (a) and (b) of Paragraph 6 of the conveyance.

Second: There shall be paid to the Company the amount deemed to have been then applied with respect to the Company pursuant to subparagraph (c) of Paragraph 6 of the Conveyance; provided, however, that if an Event of Default (as set forth in Section 5.1 hereof) hereunder shall have occurred and then be continuing, the Agent shall have the right to apply such amount to the payment and/or prepayment without penalty or premiu. 1 of the indebtedness secured arreby; and

There shall be paid to the Company (or to the Trustees or the holders of the relevant Notes to the extent that such Trustees or holders have incurred any expense or made any payment on behalf of the Company pursuant to Section 2.4 hereof for which the Company would be entitled to be reimbursed under subparagraphs (d) and (e) of Paragraph 6 of the Conveyance) the aggregate amount deemed to have been then applied with respect to the Company pursuant to, and in the order of priority specified in, said subparagraphs (d) and (e) of said Paragraph 6 of the Conveyance; provided, however, that, in the case of the taxes, assessments, costs and expenses referred to in said subparagraphs of the Conveyance, the Agent shall have received satisfactory

evidence of the Company's payment or obligation to pay sucitaxes, assessments, costs and expenses in respect of which such amount is deemed to have been so applied and shall have approved the same; and

FOURTH: The balance shall be applied to the payment or prepayment of the unpaid principal of the Notes; and

FIFTH: After payment in full of the Notes and all other indebtedness secured hereby, there shall be pc.1 to the Company an amount equal to all amounts (if any) previously applied by the Agent pursuant to the proviso of clause Second above;

Section 3.3. Notwithstanding the other provisions of this Article, the Trustees or any receiver appointed in judicial proceedings for the enforcement of this instrument shall have the right to receive all the income, revenues, rents, profits and proceeds from the Mortgaged Property (including the Production Payment Minerals and all proceeds therefrom) and all other amounts payable to or receivable by the Company pursuant to the Conveyance. After the Notes have been declared due and payable in accordance with the provisions of Section 5.1 hereof and to apply all said income, revenues, rents, profits and proceeds in the order of priority set forth in Section 3.2 hereof and to exercise all the rights, privileges, powers and remedies conferred upon the Company by the Conveyance in the sole interest of the holders of the Notes without regard to the effect such action may have upon the rights of any other person under the assignment contained in Section 3.1 hereof. Upon any sale of the Mortgaged Property or any part thereof pursuant to Article V hereof, the Production Payment Minerals thereafter produced from that portion of the Mortgaged Property so sold, and the proceeds therefrom, shall be included in such sale and shall pass to the purchaser free and clear of the assignment contained in this Article.

Section 3.4. The Agent shall not be liable for failure to collect or for failure to enforce collection of any proceeds assigned pursuant to this Article and shall have no other responsibility in connection therewith, except the responsibility to account for funds actually received. For all purposes of this Mortgage, any proceeds, amounts or moneys shall be deemed to have been "actually received" by the Agent when

such proceeds, amounts or moneys shall be immediate! available to the Agent at the office of the Agent at 399 Park Avenue, New York, New York 10043 or at such other office of the Agent in New York City as shall be designated from time to time by the Agent. The Agent, if and to the extent that the Agent receives and applies funds under this Article III, shall account to the Grantors, the Company and the holders of the Notes for funds so received and applied under this Article III by furnishing to the Grantors, the Company and the holders of the Notes, quarterly within 15 Business Days after each Application Date (or the Next Preceding Business Day, as applicable), the written statement referred to in the last sentence of Paragraph 6 of the Conveyance with respect to any funds applied on such Application Date (or the Next Preceding Business Day, as applicable).

Section 3.5. Nothing herein contained shall detract from or limit the absolute obligation of the Company to make payment of the Notes at maturity and of all amounts owing hereunder by the Company at the time and in the manner provided herein and in the Notes, regardless of whether the proceeds of production assigned by this Article are sufficient to pay the same, and the rights under this Article shall be in addition to all other security of any and every character now or hereafter existing to secure the indebtedness of the Company secured hereby.

Section 3.6. The Companagrees to indemnify the Trustees, the Banks, the Agent and the holders of the Notes against all claims, actions, liabilities, judgments, costs, charges and attorney's fees made against or incurred by them, or any of them, with respect to the Production Payment, as a consequence of an assertion that they received funds from the production of Minerals claimed by third persons either before or after the payment in full of the Notes; and the Trustees, the Banks, the Agent and such holders shall have the right to defend against any such claims, actions and judgments, employing their attorneys therefor, and if they are not furnished with reasonable indemnity, they shall have the right to compromise and adjust all such claims, actions and judgments; and the Company agrees to indemnify and pay to the Trustees, the Banks, the Agent or such holders any and all such claims, judgments, costs, charges and attorneys' fees as may be paid in any adjustment thereof or as may be adjudged against any of them.

Section 3.7. Without limiting the foregoing provisions of this Article III, the Company agrees that this Article III is intended to grant to the Agent for the benefit of the holders of the Notes a security interest in Minerals to be extracted from or attributable to the Subject Interests which are or may become subject to the Production Payment and in and to the proceeds resulting from the sales thereof (including, but not limited to, sales at the wellhead or mine), such security interest to attach to such Minerals as extracted and to the accounts resulting from such sale. The Company, as owner of the Production Payment, has an interest of record in the Subject Interests which are now subject to the Production Payment and, upon execution and delivery of the Supplemental Instrument of Conveyance of Production Payment referred to in the Conveyance, will have an interest of record in the Subject Interests described in said Supplemental Instrument of Conveyance of Production Payment, Pennzoil Company, Pennzoil Producing Company, Pennzoil Oil & Gas, Inc. and Duval Corporation, each a Delaware corporation, are the record owners of said Subject Interests subject to the Production Payment to the extent provided in the Conveyance.

ARTICLE IV.

DEFEASANCE

Section 4.1. If all indebtedness of the Company secured hereby shall be paid, then, upon receipt by the Trustees of evidence satisfactory to them of the payment in full of the indebtedness secured hereby, this Mortgage shall become null and void and the interest of the Company in the Mortgaged Property shall become wholly clear of the lien, security interest, conveyance and assignment created hereby, and such lien and security interest shall be released in due course at the cost of the Company. Otherwise this Mortgage, Deed of Trust and Assignment of Production shall remain and continue in full force and effect.

ARTICLE V.

REMEDIES IN EVENT OF DEFAULT

Section 5.1. In case any one or more of the following "Events of Default" shall occur and shall not have been remedied:

- A. Default by the Company in any payment of the principal of or interest on the Notes when due or in any payment of any costs or other amounts required by or under the Cradit Agreement or to lotes when such costs of other amounts become due and paya and or to company to the costs of the cos
- B. Any representation or warranty contained in Section 2.1 hereof, or in any certificate furnished by the Congany in connection with the Credit Agreement, shall prove to have been untrue in any material respect when made; or
- C. Default by the Company in the due performance or observance of any covenant or condition herein or in the Credit Agr. ement contained, which shall have continued unremedied for a period of 30 days after it shall have become known to the Company; or
- D. At any time the unliquidated balance of the Primary Sum shall be less than the aggregate unpaid principal amount of the Notes; or
- E. The Company shall (1) become insolvent or be unable to pay its debts as they mature or generally not pay its debts as such debts become due, (2) apply for or consent to the appointment of a receiver, trustee or liquidator of the Company or of all or a substantial part of its assets, (3) be adjudicated a bankrupt or insolvent, or file a voluntary petition in bankruptey, (4) make a general assignment for the benefit of creditors, (5) file a petition or an asswer seeking reorganization or an arrangement with creditors or to take advantage of any insolvency law, (6) file an answer admitting the material allegations of, or consent to, or default in answering, reorganization or insolvency proceedings or proceedings or (7) take any corporate action to authorize any of the actions set forth in this Subsection E; or
- F. An "Event of Default" under Paragraph 17(c) or (d) of the Conveyance shall occur; or
- G. An order, judgment or decree shall be entered by any court of competent jurisdiction, approving a petition seeking reorganization of the Company or appointing a receiver, trustee or liquidator of the Company or of all or a substantial part of

its assets, and such order, judgment or decree snall continue unstayed and in effect for any period of 30 consecutive days;

then, and in any such event, the Agent may with the consent, or shall at the direction, of the holders of not less than 66% 1% in aggregate unpaid principal amount of the Notes, by notice in writing to the Company, the Trustees and the holders of the Notes, declare the entire unpaid principal of and interest on all the Notes and all other amounts then owing by the Company hereunder and under the Credit Agreement and the Notes to be forthwith due and payable, whereupon the same shall become and be forthwith due and payable without other notice or demand of any kind and the liens and security interests evidenced hereby shall be subject to foreclosure in the manner provided for herein or provided for by law as the Trustees may elect.

Section 5.2. All costs and expenses (including attorneys' fees) incurred by the Trustees, or by the bolders of the Notes in protecting and enforcing their rights hereunder shall constitute a demand obligation owing by the Company and shall bear interest at a rate equal to the Applicable Rate (but in no event to exceed the maximum rate permitted by applicable law) in effect during the period such obligation is outstanding, all of which shall constitute a portion of the indebtedness secured by this instrument.

Section 5.3. If the entire unpaid principal of and interest payable on the Notes and all other amounts owing by the Company hereunder and under the Credit Agreement and the Notes shall have become due and payable, whether at the stated date of maturity thereof or prior thereto, and shall not have been paid, the Trustees, upon and only upon the written request of the holders of not less than 66% in aggregate unpaid principal amount of the Notes, shall have the following rights and powers:

A. To sell, to the extent permitted by law, the Mortgaged Property, at one or more sales as an entirety or in parcels, as the Trustees may elect, at such place or places and otherwise in such manner and upon such notice as may be required by law, or, in the absence of any such requirement, as the Trustees may deem appropriate, and to make conveyance to the purchaser or purchasers; or

B. To proceed by a suit or suits in equity or at law, whether for the specific performance of any covenant or agreement of the Company herein contained or in and of the execution of any power herein granted, or for any forcelosure or the sale of the Mortgaged Property under the judgment or decree of any court of competent jurisdiction, or for the appointment of a receiver or receivers of the Company as to the Mortgaged Property, or for the enforcement of any other appropriate legal or equitable remedy; or

C. To take such other action as may be authorized or permitted by any applicable law including without limitation the Uniform Commercial Code.

Section 5.4. Any sale by the Trustees of any part of the Mort. gaged Property located in the State of Texas may be made in any county in which any part of the Mortgaged Property to be sold at such sale may be situated, provided that the notice hereinafter mentioned shall designate the county where the item of property will be sold. All items of property may be sold singly, or together, or in lots or parcels, and in such order, as to the Trustees may seem expedient. The items of property shall be sold at the door of the county courthouse in the county in which the sule is held, and netween the hours of 10:00 a.m. and 4:00 p.m. on the first Tuesday in any nonth, after having given rotice of such sale in accordance with the applicable statute of Texas now in force (Article 3810 of the Regissed Civil Statutes of Texas of 1925) governing sales of real estate under powers of sale conferred by a deed of trust. If such applicable statutes should be hereafter amended to provide a different notice of sale applicable to sales of real estate under provide a univerent notice of sale applicance to sales of real estate under power of sale in a deed of trust, then the Trustees may, in their sole discretion and to the extent permitted by law, give either the notices of ale now provided to be given under said Article 3810 or that prescribed by the amended statute. Any one or more such sales may be conducted in the same month, or in specessive or different months, as the Trustees may deem expedient. The Trustees may appoint or dele. gate any one or more persons as agent to perform any act or acts necessary or incident to any and each such sale, including the posting of notices and the conduct of sale, but the Trustees shall execute any and all deeds as Trustees. In the case of each such sale, the Trustees are authorized to execute and deliver to the purchaser or purchasers of the

property sold good and sufficient deeds or other instruments of conveyance therefor, in fee simple, with covenants of general warranty, or otherwise, as the Trustees may deem expedient.

Any sale by the Trustees of any part of the Mortgaged Property located in any county in the State of Alabama may be made at public outcry at the front door of the courthouse of said county, for cash to the highest bidder, after first giving thirty days' notice of the time, place and terms of sale, together with a description of the property to be sold, by publication once a week for three consecutive weeks, in a newspaper published in said county.

Section 5.5. In exercising the power of sale hereby given, the Trustees may conduct any number of sales from time to time to the extent permitted by law. The power of sale shall not be exhausted by any one or more such sales as to any part of the Mortgaged Property not lawfully sold, but shall continue unimpaired until all the Mortgaged Property shall have been sold or the Notes and all indebtedness of the Company secured hereby shall have been paid.

The sale of all or any portion of the Mortgaged Property may, to the extent permitted by law, be postponed by public announcement at the time and place of such sale, and from time to time thereafter may be further postponed by public announcement made at the time of sale fixed by the preceding postponement.

Upon any sale, whether under the power of sale hereby given or by virtue of judicial proceedings, no necessity shall exist for the Trustees or any public officer acting under execution or order of court to have physically present or constructively in his possession any of the Mortgaged Property.

Upon the completion of any sale made pursuant to this Article, the Trustees or Trustee making such sale shall be entitled to execute and deliver to the purchaser or purchasers a good and sufficient deed or deeds of conveyance, sale and transfer of the Mortgaged Property or any part thereof, and the Trustees or Trustee (as the case may be) making such sale and their or his successors are hereby irrevocably appointed the true and lawful attorneys for the Company, in its name and stead, to make and execute all necessary deeds and conveyances of the property sold. Notwithstanding the foregoing, the Company

will, if so requested by the Trustees or Trustee (as the case may be) making such sale or by any purchaser, confirm any such sale or transfer by executing and delivering to the Trustees or such Trustee or to such purchaser all proper instruments of conveyance and transfer and releases as may be specified in any such request.

Section 5.6. Upon any sale, whether made under the power of sale hereby given or by virtue of judicial proceedings, the receipt of any Trustee, or of the officer making a sale under judicial proceedings, shall be a sufficient discharge to the purchaser or purchasers at any sale for his or their purchase money, and such purchaser or purchasers, or his or their assigns or personal representatives, shall not, after paying such purchase money and receiving such receipt of a Trustee or of such officer therefor, be obliged to see to the application of such purchase price, or be in anywise answerable for any loss, misapplication or non-application thereof.

Section 5.7. The proceeds of sale of the Mortgaged Property and all other monies received by the Trustees hereunder or in any proceedings for the enforcement hereof, the application of which has not elsewhere herein been specifically provided for, shall be applied as follows:

First: To the payment of all necessary costs and expenses incident to such sale and the enforcement of this instrument, including, but not limited to, a reasonable compensation to the agents, attorneys and counsel of the Trustees and of the holders of the Notes:

SECOND: To the payment of any indebtedness of the Company secured hereby, other than indebtedness referred to in the following Paragraphs Third and Fourth;

THIRD: To the payment of accrued interest payable on the Notes (ratably according to amount accrued and unpaid thereon), costs and other amounts (excluding principal) due under the Notes;

FOURTH: To the payment of the unpaid principal of the Notes (ratably according to the unpaid principal amounts thereof); and

FIFTH: The remainder, if any, shall be paid to the Company, its successors or assigns, or to whomsoever may be la rfully entitled

to receive the same or as a court of competent jurisdiction may direct.

Except as otherwise provided by law, the recitals contained in any conveyance made by any Trustee to any purchases at any sale made under this Article shall conclusively establish the truth and accuracy of the matters therein stated, including, without limiting the generality of the foregoing, nonpayment of the unpaid principal of and interest payable on the Notes after the same have become due and payable, whether at the stated date of maturity or otherwise, advertisement and conduct of such sale in the manner provided in this Article and appointment of any successor Trustee hereunder, and all prerequisites to such sale shall be presumed to have been satisfied and performed.

Section 5.8. The holder or holders of any Note or Notes shall have the right to become the purchaser at any sale held by a Trustee or by any receiver or public officer, and such holder purchasing at any such sale shall have the right to credit upon the amount of the bid made therefor the amount payable to such holder out of the net proceeds of such sale.

Section 5.9. The Company agrees that, after any Event of D fault hereunder shall have occurred, the holders of not less than 65% in aggregate unpaid principal amount of the Notes shall be entitled as a matter of right, without regard to the adequacy of the security for the Notes or the solvency of the Company, to the appointment of a receiver or receivers of such holders' choice, except as may be prohibited by law, of all or any part of the Mortgaged Property and the rents, issues, profits, revenues and other income thereof, whether such receivership be incident to a proposed sale thereof or otherwise, and the Company does hereby irrevocably consent to the appointment of such receiver or receivers and agrees not to oppose any application therefor by such holder or any appointment pursuant thereto.

Section 5.10. All remedies herein expressly provided for are in addition to any and all other remedies now or hereafter existing at law or in equity, and the holder of any Note shall, in addition to the remedies herein expressly provided for, be entitled to avail itself or himself of all such other remedies as may now or hereafter exist at

law or in equity for the collection of the indebtedness hereby secured and the enfercement of the covenants herein contained and the fore-closure of the lien and security interest created hereby and the resort to any remedy provided for hereunder or provided for by law shall not prevent the concurrent or subsequent employment of any other appropriate remedy or remedies.

Section 5.11. The holders of the Notes may resort to any security given by this instrument or to any other security now existing or hereafter given to secure the payment of any of the indebtedness secured "ereby, in whole or in part, and in such portions and in such order as may seem best to such holders in their sole and uncontrolled discretion and any such action shall not in anywise be considered as a waiver of any of the rights, benefits or living created by this instrument.

Section 5.12. To the full extent that it may lawfully so agree, the Company agrees that it will not at any time insist upon, plead, claim or take the benefit or advantage of any appraisement, valuation, stay, extension or redemption law, now or hereafter in force, in order to prevent or hinder the enforcement of this instrument or the absolute sale of the Mortgaged Property, or any part thereof, or the possession thereof by any purchaser at any such sale, and the Company, for itself and all who may chi im under it, insofar as it now or hereafter lawfully may, hereby waives the benefit of all such laws. The Company, for itself and all who may claim under it, waives, to the extent that it lawfully may, all right to have the Mortgaged Property marshalled upon any foreclosure hereof.

If any law in this Section referred to and now in force, of which the Company or its successors might take advantage despite this Section, shall hereafter be repealed or cease to be in force, such law shall not thereafter be deemed to constitute any part of the contracts herein contained or to preclude the operation or application of this Section.

Section 5.13. Pursuant to Section 39-5-19 New Mexico Statutes, Annotated, 1978, Comp., the Company agrees that, as to the Mortgaged Property situated in the State of New Mexico, the redemption period shall be shortened to one (1) month.

ARTICLE VI.

SPECIAL MORTGAGE OF PROPERTY IN LOUISIANA

Section 6.1. With respect to this instrument, in so far as the portions of the Mortgaged Property situated in the State of Louisiana are concerned, or as to which laws of the State of Louisiana are applicable as a matter of Federal law, the general language of conveyance to the Trustee in this instrument is intended and shall be construed as words of hypothecation and not of conveyance, and this instrument shall be considered under the laws of the State of Louisiana as an Act of Mortgage and Pledge and Assignment of Production in authentic form by the Company for the benefit of the holders of the Notes. Whenever the terms "mortgage" and "pledge" appear herein, such terms shall be construed to entitle any holder of a Note to, without limitation, all of the rights and remedies provided for in LSA-R.S.9:4301-04 and LSA-R.S.31:197, et seq.

Section 6.2. In order that the Act of Mortgage and Pledge and Assignment of Production shall be an authentic act under the laws of the State of Louisiana, the Company does hereby declare, before the undersigned Notary Public of the State of New York and in the presence of the undersigned witnesses to the execution of this instrument by the Company that, in order to secure full and punctual payment of the Notes at maturity, or when otherwise due hereunder (as provided in Article V hereof), or as the same may be extended, and any and all its other indebtedness secured by this instrument, it does, by these presents, mortgage, pledge and specially hypothecate in favor of the the holders of the Notes represented herein by the Trustees and the Agent, such portion of the Mortgaged Property as is situated in the State of Louisiana or as to which laws of the State of Louisiana are applicable as a matter of Federal law. The Company does further declare that attached hereto as Exhibit D and made a part hereof is a certified copy of the resolutions adopted by the Board of Directors of the Company on July 9, 1981, and pursuant to which this instrument is executed on behalf of the Company.

Section 6.3. The Company hereby acknowledges the obligations secured hereby whether now existing or arising hereafter, and con-

fesses judgment in favor of the Trustees, for the benefit of each holder of the Notes, up to the full arount of the unpaid principal thereof and interest thereon, together with all costs, charges expenses, taxes, assessments, fees and other amounts payable by the Company in accordance with the terms hereof in the event that the same is not paid at maturity, and does by these presents consent, agree and stipulate that, upon the occurrence of an Event of Default hereunder, the holders of not less than 6624% in aggregate unpaid principal amount of the Notes, may at the option of such holders, without making demand and without notice or putting in default, all the same being hereby expressly waived, cause all and singular the property of the Company herein mortgaged to be seized and sold by executory process issued by any competent court, or to proceed with the enforcement of this Act of Mortgage and Pledge and Assignment of Production in any manner prescribed by law, the Company hereby waiving the benefit of any laws or parts of laws relating to the appraisement of the property seized and sold under executory process or other legal process, and consenting that the Mortgaged Property be sold without appraisement to the highest bidder for cash. Each Trustee is irrevocably appointed special attorney-in-fact for the holders of the Notes and vested with full power in their behalf to effect and enforce the mentgage and pleage hereby constituted for their benefit, or otherwise to act as herein provided for.

Section 6.4. The Company agrees that its interest in the Mortgaged Property shall remain mortgaged, hypothecated and pledged until full and final payment of unpaid principal of and interest payable on the Notes and any and all other amounts owing by the Company and secured by this instrument, and the Company further agrees and binds itself, its successors and assigns, not to sell, alienate or encumber the Mortgaged Property or any part thereof to the prejudice of this Act of Mortgage and Pledge and Assignment of Production.

Section 6.5. The Notes have been paraphed "Ne Varietur" for identification with this Act of Mortgage and Pledge and Assignment of Production by the undersigned Notary Public of the State of New York.

Section 6.6. The Company hereby expressly waives: (a) the benefit of appraisement, as provided in Articles 2332, 2336, 2723 and 2724;

Louisiana Code of Civil Procedure, and all other laws conferring the same; (b) the demand and three (3) days delay accorded by Articles 2639 and 2721, Louisiana Code of Civil Procedure, (c) the notice of relative required by Articles 2293 and 2721, Louisiana Code of Civil Procedure; (d) the three (b) days delay provided at Articles 2331 and 1722, Louisiana Code of Civil Procedure; and (c) the benefit of the other provisions of Articles 2331, 2722 and 2722, Louisiana Code of Civil Procedure; and the benefit of any other Articles or laws relating to rights of appraisement, notice, or delay not specifically mentioned above; and Company expressly to sees to the immediate seizure of the Mortgaged Property in the event of suit hereon.

Secret 6.7. If any Note transferred by notarial Act of Assignment, in accordance with the laws of the State of Louisiana then obtaining, such transferred Note "all and entire subrogation of title in and to the Note so transfer et and to any and all rights and privileges under this instrument hereby, anted to the Trustees and the present holder of such Note.

Section 6.8. The Company hereby expressly authorizes and agrees that the Trustees shall have the right to appear a keeper of the Mortgaged Property pursuant to the terms an 'provisions of La S. 9:51 of et seq.

Section 6.9. The execution by the Trusce's of this instrument shall be deemed to constitute acceptance by the Agent.

ARTICLE VII

CONCERNING THE TUCKS EES

Surpos 7.1. The Trustees hereby accept the trus's hereunder, but only upon the terms set forth in this Mortgage, including the following:

A. It shall be no part of the Caty of the Trustees to see to any recording or rerecording, filing or refiture, or registration or reregistration of this Mortgage or any in transmit in addition or supplemental here, to or to see to the payment of or be under any

duty in respect of any tax or assessment or other governmental charge which may be levied or assessed on the Mortgaged Property, or any part thereof, or against the Company, or to see to the performance or observance by the Company of any of the constants or agreements contained in this Mortgage.

- B. Except with respect to the matters set forth in their respective acknowledgments hereto, the Trustees shall of buttes onsible for the execution, acknowledgment or valid ty of this Mortgame, the Conveyance or any instrument in addition or a pipeline a hereto or thereto or of the Notes, or for the truth of the statements contained herein or therein, or for the sufficiency of the security purported to be created hereby or thereby, or for the proper as chorization hereof or thereof by corporate or public action, and the Trustees make no representation in respect hereof or thereof or its respect of the rights of the holders of the Note.
- C. The Trustees shall have the right to employ and obtain the advice of counsel selected by them, who may also be counsel for the Company, the Agent or the holders of the Notes, and shall not be liable hereunder for any action taken, suffered or omitted by them hereunder in good faith and in reliance upon such advice.
- D. The Company will indemnify and save harmless each of the Trustees from and against any liability or damage which he may incur in the exercise and performance of any of his powers or duties hereunder except as a result of his own gross negligence or wilful misconduct.
- E. The Trustees shall be protected hereunder in relying on any instruction, document or signature I, dieved by them to be genuine and to have been signed by the proper party or parties or by any person or person authorized to act on its or their behalf. The Trustees may accept a copy of a resolution of the Dord of Directors of any corporation certified by the Secretary or in Assistant Secretary of such corporation to have been duly adopted by such Board and to be in full force and effect on the date of such certification as conclusive evidence that such resolution has been duly adopted by such Board and that the same is still in full force and effect. A certificate signed by the President or one or more vice Presidents or the Treasurer of the Company as to the

existence or non-zistence of any fact pertinent to the light of the Trustees to take or refrain from taking any action index this Mortgage may be recepted by the Trustees as conclusive evidence of the facts that in stated, and shall constitute full protection hereunder to the Trustees for any action taken or omitted to be taken by them in good faith in reliance thereon.

- F. The Trustees shall be fully protected hereunder in any action taken or omitted to be taken by them at the request or on the direction of the holders of not less than 66% in aggregate unpaid principal amount of the Notes, unless the request or direction is required or permitted by the terms of this Mortgage to be made by the Lolder or holders of a lesser or greater amount of the Notes and, in that event, the Trustees shall be fully protected hereunder in any action taken or omitted to be taken at the request or on the direction of such holders.
- G. Unless and until the Time ses shall have received written notice to the contrary from the holders of not less than 66%% in aggregate unpaid principal amount of the Notes, as provided in Section 51, they may for all purposes conclusively assume that the Company has paid the principal of and interest on the Notes according to the terms thereof and that no circumstances exist or events have coursed by reason of which the holders of the Notes have or will have the right to declare the united principal amounts thereof and interest thereon to be immediately due and payable, as provided therein and in Section 5.1.
- H. The Trustees shall not be required to take any action for the enforcement of this Mortgage or the energie of any rights or remedies hereunder or to appear in or defending action, suit or other proceeding in connection therewith, which is in the opinion of the Trustees, such action will be likely to involve them in expense or liability, these the Trustees have been tendered security or indemnity, and asset to the defending against the costs, expenses and liabilities which might be incurred by them in connection therewith.
- I. Any moneys received by or on behalf of the Trustees under any provision of this Mortgage need not be segregated from other funds except to the extent required by law. The Trustees shall not be under any liability for interest on any moneys received by

or on behalf of them hereunder except as otherwise agreed with the Company.

- J. The Trustees shall not be personally liable hereunder, in the case of entry by them upon any of the Subject Interests or any of the appurtenant and other projectly situated upon or used or useful or held for future use in connection with the exploration, development or operation of the Subject Interests or for the production, gathering, treating, storing or transporting of Subject Minerals or other minerals pursuant to the provisions hereof or of the Conveyance, for debts contracted or liabilities or damages incurred in connection with the enforcement of any of the rights, remedies, powers or other interests of the Trustees in, to or under the Subject Interests or such property.
- K. The Trustees shall not be personally liable hereunder for any action taken or omitted by them in good faith and without gross negligence, and believed by them to be authorized or within the discretion or rights or powers conferred upon them by this Mortgage and which is not contrary to any of the express provisions of this Mortgage.
- L. The Trustees shall not be liable hereunder except for the performance of duties which are required to be performed by the Trustees by the express terms of this Mortgage, and no implied covenants or obligations shall be read into this Mortgage against the Trustees.
- M. The Trustees shall not valve the exercise of any rights, powers, privileges or remedies prented to them under this Mortgage or any default by the Company under this Mortgage or give any consent permitted to be given by them under this Mortgage, unless expressing directed to do so in writing by the holders of not less than 664 for the appropriate unpaid principal amount of the Notes.

Section 7 2. The Company agrees:

A. Except as otherwise expressly provided herein, to reimburse the Trustees upon their request for all reasonable expenses, disbursements and advances accurred or torue by the Trustees in accordance with any provision of this Mortgage (including the reasonable compensation and the expenses and disbursements of

their agents and counsel), except any such expense, disbursement or advance as may be attributable to their gross negligence or bad faith.

2. To indemnify the Trustees for and to held them harmless against, any loss liability or expense incurred without grass negligence or bad faith to their part, arising out of or in connection with the acceptance of administration of the trusts hereunder, including the costs and expenses of defending themselves against any claim or liability in connection with the exercise or performance of any of their powers or duties hereunder.

All such payments and reimbursements which are not made within 10 days of the date when due or requested hereunder shall be made with interest at a rate per annum equal to 2% over the Applicable Rate in effect from time to time (but in no event in excess of the maximum rate permitted under applicable law.) from and including the eleventh day after the date when due or requested hereunder to (but not including) the date on which such payments and reimbursements shall be made.

As security for the performance of the obligations of the Company under this Section, the Trushas shall have a lien prior to the Notes on the Mortgaged Property and upon all property and mone; assigned pursuant to Section 5.1. The obligations of the Company under this Section shall be payable from proceeds of Production Payment Minerals and, if such proceeds shall be insufficient, payment of such obligations may be made from and out of any moneys held by or upon behalf of the Trustees, or either of them, hereunder and from and out of the Mortgaged Property prior to any payment therefrom on account of the Notes or to open on the order of the Company.

Section 7.3. Any Trustee may resign at any time by giving written notice there if to the other Trustee, the Company ar i the holders of the Notes or be renaived at any time with or without cause by an instrument in writing dad, executed by or on behalf of the Agent. In case of the death, resignation or removal of a Trustee the remaining Trustee may act alone or a successor Trustee may be appointed by the Agent, by an instrument of substitution complying with applicable requirements of law, or, in the absence of any such requirements, without other formality than appointment and designation in a viting. Such

appointment and designation by the Agent shall be full evidence of the right and authority of the Agent to make the same and of all facts therein recited, and upon the making of any such appointment and designation, and compliance by the successor Trustee with the requirements for a successor Trustee set forth in Section 7.4, the successor Trustee named in the appointment and designation shall without further act deed or conveyance forthwith become vested with all the estate and title of the Trustees named herein in and to all the Mortgaged Property, and shall thereupon succeed to all the rights, powers, privileges, immunities and duties hereby conferred upon the Trustees named herein, and one such appointment and designation shall not exhaust the right to appoint and designate successor Trustees hereunder but such right may be exercised repeatedly as long as any indebtedness secure its reby remains unpaid.

If no successor Trustee shall have been appointed as contemplated by the foregoing provisions of this Section, or if a successor Trustee shall have been appointed but shall not have accepted the appointment, within 30 days after the resignation of the last remaining Trustee or the occurrence of a vacancy in the office of the Trustees, the holder of any Note or such retiring Trustee may apply to any court of competent jurisdiction to appoint a successor Trustee or Trustees.

Section 7.4. Every successor Trustee appointed hereunder shall execute, acknowledge and deliver to the Company and to any retiring Trustee an instrument accepting such appointment, and thereupon such successor Trustee, without any further act, deed or conveyance, shall become vested with all the rights, powers, privileges, immunities and duties of the Trustee being succeeded; but, at the request of the Company or the successor Trustee, any retiring Trustee who has resigned or been removed shall, upon payment of any amounts then due him under Section 7.2, execute and deliver an instrument transferring to such successor Trustee all the rights, powers, privileges, immunities and duties of the retiring Trustee, and shall duly assign, transfer and deliver to such successor Trustee all property and moneys held by such retiring Trustee hereunder, subject nevertheless to his lien, if any, provided for in Section 7.2. Upon request of any such successor Trustee, the Company shall execute any and all instruments for more fully and certainly vesting in and confirming to such successor Trustee all such rights, powers, privileges, immunities and duties.

Section 7.5. In this Mortgage, unless the context otherwise requires, the term "Trustees" shall mean the Trustees hereinbefore named, a Trustee acting alone under the provisions of Section 7.3 or any successor Trustee appointed pursuant to the provisions of Sections 7.3 and 7.4.

Section 7.6. Any Trustee from time to time serving hereunder shall have the absolute right, acting independently, to take any action and to give any consent and to exercise any right, remedy, power or privilege conferred upon the Trustees, or upon any individual Trustee, hereunder, and any action taken by any Trustee from time to time serving hereunder shall be binding upon all other Trustees, and no person dealing with any Trustee from time to time serving hereunder shall be obligated to confirm the power and authority of such Trustee to act without the concurrence of the other Trustee.

Section 7.7. The powers and duties of Trustees conferred or imposed by Section 126 of the Real Property Law of the State of New York, which powers and duties are hereby incorporated by reference herein, shall be in addition to those conferred or imposed by this Mortgage, and, in the case of a conflict, the provision of such Section 126 shall prevail, provided, however, that if such Section 126 shall not apply to this Mortgage or shall be repealed or shall cease to apply to this Mortgage, the incorporation by reference hereinabove contained in this Section 7.7 shall have have no force and effect.

ARTICLE VIII.

MISCELLANEOUS PROVISIONS

Section 8.1. Whenever in this Mortgage it is provided that the holders of a specified percentage in aggregate unpaid principal amount of the Notes may take any action (including the making of any demand or request, the giving of any notice, consent or waiver or the taking of any other action), the fact that at the time of taking any such action the holders of such percentage have joined therein may be evidenced by any instrument or any number of instruments of similar tenor executed by such holders in person or by agent or proxy appointed in writing.

Section 8.2. All options and rights of election herein precided for the benefit of the Trustees and any of the holders of the Notes are continuing, and the failure to exercise any such option or right of election upon a particular default or breach or upon any subsequent default or breach shall not be construed as waiving the right to exercise such option or election at any later date. By the acceptance of payment of any indebtedness secured hereby after its due date, the holder of such indebtedness does not waive the right either to require prompt payment when due of all other sums so secured or to regard as a default the failure to pay any other sums due which are secured hereby. No exercise of the rights and powers herein granted and no delay or omission in the exercise of such rights and powers shall be held to exhaust the same or be construed as a waiver thereof, and every such right and power may be exercised at any time and from time to time.

Section 8.3. No release of any part of the Mortgaged Property shall in anywise alter, vary or diminish the force, effect or lien of this instrument on the balance of the Mortgaged Property.

Section 8.4. Any notice herein provided to be mailed to the Company by the Trustees or by the holders of the Notes shall be mailed, first class postage prepaid, addressed to the Company at 100 West Tenth Street, Wilmington, Delaware 19801, Attention: Joseph A. Barbera, or to such other address as the Company may have furnished to the Trustees and the Agent in writing for the purpose. Any notice hereunder to be mailed to the Trustees or the Agent shall be mailed, first class postage prepaid, addressed to the Trustees or the Agent, as the case may be, to the Trustees c/o Citibank, N.A., 399 Park Avenue, New York, New York 10043, Attention: North American Banking Group/Energy Central, with a copy thereof sent as well e o Citicorp (USA), Inc., 2100 Two Allen Center, 1200 Smith Street, Houston, Texas 77002, Attention: Houston Energy Department.

Section 3.5. If any provision hereof is invalid or unenforceable in any jurisaiction, the other provisions hereof shall remain in full force and effect in such jurisdiction, and the remaining provisions hereof shall be liberally construed in favor of the Trustees, the Agent and the holders of the Notes in order to effectuate the provisions

hereof, and the invalidity or unenforceability of any provision hereof in any jurisdiction shall not affect the validity or enforceability of any such provision in any other jurisdiction.

Section 8.6. This instrument is made with full substitution and subrogation of the Trustees and their successors in this trust and his assigns in and to all covenants and warranties by others heretofore given or made in respect of the Mortgaged Property or the Subject Interests, or any part of either thereof.

Section 8.7. The terms, provisions, covenants and conditions hereof shall bind and inure to the benefit of the respective successors and assigns of the Company, the Trustees, the Agent and the holders from time to time of the Notes.

Section 8.8. In addition to any supplement or amendment hereof entered into pursuant to the provisions of Subsection R of Section 2.2, the Company and the Trustees, with the written consent of the holders of not less than 66%% in aggregate unpaid principal amount of the Notes, may amend or supplement this Mortgage for the purposes of adding any provisions to or changing in any manner or eliminating any of the provisions of this Mortgage or of modifying in any manner the rights of the holders of the Notes under this Mortgage, except that, without the written consent of the holders of all the Notes at the time outstanding, no amendment or supplement to this Mortgage shall extend the maturity of any Note or reduce the rate or extend the time of payment of interest payable with respect to any Note or reduce the principal amount of any Note or change the percentage of the aggregate unpaid principal amount of the Notes required for any amendment or supplement hereof or any consent or other action hereunder.

Section 8.9. No recourse shall be had for the payment of the principal of or interest on any of the Notes or for any claim based thereon or on this Mortgage or any instrument amendatory thereof or supplemental thereto or based upon any representation, warranty or statement contained herein or therein or in any certificate or other document delivered in connection with the transactions contemplated hereby, against any incorporator or against any member, beneficiary, distributee, grantee, director or officer, past, present or future, of

the Company or any successor corporation, as such, either directly or through the Company or any such successor corporation, whether by virtue of any constitution, statute or rule of law or by the enforcement of any assessment or penalty or otherwise, any and all such liability, whether at common law, in equity, by any constitution, statute or otherwise, of incorporators, members, beneficiaries, distributees, grantees, directors or officers, as such, being released as a condition of and in consideration for the execution hereof and the issue of the Notes.

Section 8.10. The validity, effect and construction here of as a mortgage, deed of trust and assignment affecting any Mortgaged Property and the remedies herein provided for or provided for by law with respect to any Mortgaged Property shall be governed by the laws of the States where the Mortgaged Property is located or which are made applicable to the Mortgaged Property as a matter of Federal law. With respect to all other matters (including, without limitation, matters of construction, validity and performance other than as set forth above in this Section 8.10), this Morgtage shall be governed by and construed in accordance with the laws of the State of Texas.

Section 8.11. For purposes of any legal action or proceeding against the Company arising out of or based upon this Mortgage, the Notes or the Credit Agreement, the Company hereby irrevocably submits to the jurisdiction of the courts of the State of New York and of the United States for the Southern District of New York, and hereby irrevocably designates and appoints CT Corporation System 1633 Broadway, New York, New York 10019, as the Company's agent for service of summons and other legal process on the Company in any such action or proceeding, which service may be made by mailing a copy of any summons or other legal process in any such action or proceeding to the Company, in care of such agent, at the address of CT Corporation System set forth above; and such agent is hereby authorized and directed to accept for and on behalf of the Company service of summons and other legal process in any such action or proceeding against the Company. The mailing, as herein provided, of such summons or other legal process in any such action or proceeding shall be deemed personal service and accepted by the Company for all the purposes of such action or proceeding. Final judgment (a certified or exemplified copy of which shall be conclusive evidence of the fact and of the amount of any indebtedness of the Company therein described) against the Company in any such legal action or proceeding shall be conclusive and may be enforced in other jurisdictions, by suit on the judgment, provided that any legal action or proceeding against the Company arising out of or based upon this Mortgage, the Notes or the Credit Agreement may be brought against the Company in the courts of any other appropriate jurisdiction and that summons or other legal process may be served upon the Company or any agent designated to receive such summons or service of process in such other jurisdiction in any manner authorized by the laws of such jurisdiction. The Company further agrees to take any and all action, including the execution and filing of all such instruments and documents as may be necessary, to continue the designation and appointment of CT Corporation System as its agent for service of summons and other legal process, as aforesaid, until six years from the satisfaction and discharge of this Mortgage pursuant to Article IV.

Section 8.12. No provision of this Mortgage shall require the payment or permit the collection of interest in excess of the rate permitted by applicable law. If any excess of interest in such respect has been provided for herein, or shall be adjudicated to be so provided for herein, neither the Company nor its successors or assigns shall be obligated to pay such interest in excess of the amount permitted by applicable law.

Section 8.13. This Mortgage, Deed of Trust and Assignment of Production is being executed in multiple originals.

In Witness Whereof, the Company and the Trustees have executed this is strument before the undersigned Notary Public in the presence of the undersigned competent witnesses after a due reading of the whole on the date set forth in the acknowledgments annexed hereto.

NEW DUNSTAN, INC.

[CORPORATE SEAL]

By: Joseph Baller.

"net:

Assistant Secretary

The address of New Dunstan, Inc. is:

> 100 West Tenth Street Wilmington, Delaware 19801

The address of F. D. Thompson and W. T. Fox III is: c/o Citicorp (USA), Inc. 2106 Two Allen Center 1200 Smith Street Houston, Texas 77002 Attention: Houston Energy Department

Executed in the presence of:

Before me:

Notary Public of New York

2.D. Thompson

(F. D. Thompson)

Trustee

Trustee

Notary Public, State of New York No. 31-4729635 Qualified in New York County Commission Expires March 30, 1983 STATE OF NEW YORK
COUNTY OF NEW YORK
SS.

BE IT REMEMBERED THAT, I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that on this 16th day of July, 1981:

There appeared before me the following person, residing at the address set forth below his name, and being the designated officer of the corporation set opposite his name, such corporation being a corporation of the designated jurisdiction and being a party to the foregoing instrument:

Joseph Nalle Vice President of New Dunstan, Inc., 3024 Chevy Chase a Delaware corporation Houston, Texas

Before me personally appeared such person, whose name is signed to the foregoing conveyance, and who is known to me, and acknowledged before me on this day that, being informed of the contents of the conveyance, he, as such officer of the corporation set opposite his name and with full authority, executed the same voluntarily for and as the act of said corporation.

Before me appeared such person, to me personally well known, and stated that he is the designated officer of the corporation set opposite his name, and he is duly authorized in his capacity to execute the foregoing instrument for and in the name of said corporation, and further stated and acknowledged that he had so signed, executed and delivered said foregoing instrument for the consideration, uses and purposes therein mentioned and set forth.

Before me appeared such person, to me personally known, who being by me duly sworn, did say that he is the designated officer of the corporation set opposite his name, and that the seal of such corporation affixed to the foregoing instrument is the corporate seal of the corporation set opposite his name and that the instrument was signed and sealed in behalf of the corporation set opposite his name by authority of the Board of Directors of such corporation and that he acknowledged the instrument to be the free act and deed of the corporation set opposite his name.

(Alabama)

(Arkansas)

(Louisiana)

(Mississippi)

Before me personally appeared such person and acknowledged that he signed and delivere of e foregoing instrument on the day and year therein mentioned the act and deed of the corporation se opposite his name, having first been authorized to do so.

(Montana)

Before me percenary appeared such person, known to me to be the designated officer of the corporation set opposite his name that executed the within instrument, and acknowledged to me that such corporation set opposite his name executed the same.

(New Mexico)

Before me on this date appeared such person, 'me personally known, who being by me duly sworn, did say that he is the designated officer of the corporation set opposite his name, and that the seal affixed to said instrument was signed and scaled in behalf of said corporation by authority of its Board of Directors, and such person acknowledged said instrument to be the free act and deed of said corporation.

(New York)

Before me personally came such person, to me personally known, and he, being by me duly sworn, did depose and say that he resides at the address set opposite his name and is the designated officer of the corporation set opposite his name and that said corporation is described in and executed the above instrument; that he knows the seal of the corporation; that the seal affixed to such instrument is such corporate seal; that it was so affixed by order of the Board of Directors of the corporation; and that he signed his name thereto by like order and that the instrument was the free act and deed of the corporation.

(North Dakota)

Before me on this date personally appeared such person, known to me to be the designated officer of the corporation set opposite his name, which corporation is described in and executed the within instrument, and he acknowledged to me that such corporation executed the same.

("ennsylvania)

Before me personally approved so his person and acknowledged himself to be the designated other of the corporation set opposite his name, and that he, as such officer, being authorized so to do, executed the foregoing instrument for the purposes therein contained, by signing the name of the corporation by his self as such officer.

(Texas)

Before me on this day personally appeared such person, who is known to me to be the person whose named is subscribed to the foregoing instrument, and known to me to be the designated officer of the corporation set opposite his name, and each acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, in the capacity therein stated and as the act and deed of said corporation.

(West Virginia)

Such person has this day in my said County, before me, acknowledged the said instrument to be the act and deed of the corporation set opposite his name.

(Wyoming)

Before me on this day appeared such person to me personally known, who, being by me duly sworn, did say he is the designated officer of the corporation set opposite his name, and that the seal affixed to said instrument is the corporate seal of said corporation and that said instrument was signed and sealed in behalf of said corporation by authority of its Board of Directors, and such person acknowledged it to be the free act and deed of said corporation.

There appeared before me the following persons, each residing at the address set opposite his name and each being a party to the foregoing instrument:

- F. D. Thompson, 2100 Two Allen Center, 1200 Smith Street, Houston, Texas 77002
- W. T. Fox III, 2100 Two Allen Center, 1200 Smith Street, Houston, Texas 77002

(Alabama)

Each such person, known to me, acknowledged before me on this day that, being informed of the contents of the conveyance, he executed the same voluntarily on the day the same bears date.

(Arkansas)

Each such person personally appeared, and being to me personally well known, each stated that he executed and delivered said foregoing instrument as Trustee and for the consideration, uses and purposes therein mentioned and set forth. (Louisiana)

Each such person before me personally appeared, and each to me known to be the person described in and who executed the foregoing instrument as Frustee, and each acknowledged that he executed it as his free act and deed.

(Mississippi)

Personally appeared before me each such person and acknowledged that he signed and delivered the foregoing instrument as Trustee on the day and the year therein mentioned

(Mentana)

Before me each such person personally appeared, each known to me to be the person whose name is subscribed to the within instrument as Trustee, and acknowledged to me that he executed the same.

(New Mexico)

The foregoing instrument was acknowledged before me this day by each such person as Trustee.

(New York)

Before me personally came each such person, to me known to be the person described in and who executed the foregoing instrument as Trustee, and acknowledged that he executed the same.

(North Dakota)

Before me personally appeared each such person, known to me to be the person who is described in and who executed the within and foregoing instrument as Trustee, and acknowledged to me that he executed the same.

(Pennsylvania)

Before me, personally appeared each such person, known to me to be the person whose name is subscribed to the within instrument as Trustee and acknowledged that he executed the same for the purposes therein contained.

(Texas)

Before me personally appeared each such person, known to me to be the person whose name is subscribed to the foregoing instrument as Trustee, and acknowledged to me that he executed the same for the purposes and consideration therein expressed.

(West Virginia)

Each such person, whose name is signed to the writing above as Trustee, has this day acknowledged the same before me, in my said county. (Wyoming)

The foregoing instrument was acknowledged before me by each such person, as Trustee, this day.

IN WITNESS WHEREOF, I have hereunio set my hand and official seal, in the City, County and State of New York, this 16th day of July, 1981.

Notary Public Notary Public, State of New York No. 31-47:9635

Qualified in New York County Commission Expires March 30, 1983.

My commission expires March 30, 1983

EXHIBIT A to Mortgage, Deed of Trust and Assignment of Production

...dit Agreement

Dated as of July 16, 1981

AMONG

NEW DUNSTAN, INC.

THE BANKS NAMED HEREIN

AND

CITIBANK, N.A. as Agent

TABLE OF CONTENTS

ARTICLE I

CERTAIN DEFINITIONS AND ACCOUNTING TERMS

| | | Page |
|--------------|--|------|
| Section | 1.01 - Certain Defined Terms | 1 |
| Section | 1.02 - Terms Defined in the Conveyance | 3 |
| Section | 1 03 - Accounting Terms | 3 |
| | | |
| | ARTICLE II | |
| | AMOUNTS AND TERMS OF THE ADVANCES | |
| Section | 2.01 - Advances and Commitment | 3 |
| Section | 2.02 - Making the Advances | 4 |
| Section | 2.03 - L. Juction of the Commitments | 5 |
| Section | 2.04 = Use of Proceeds of Advance; | 5 |
| Section | 2.05 - Ceramitment Fee | 5 |
| Section | 2.06 - Repayment of the Advances: Interest | 5 |
| Section | 2 07 - Prepayments | 6 |
| Section | 2 08 - Payments and Computations | 6 |
| Section | 2 09 - Payment on Non Business Days | 6 |
| Section | 2.10 - Taxes | 6 |
| Section | 2 11 - Sharing of Payments, Etc. | 8 |
| Section | 2.12 - Evidence of Debt | 8 |
| | ARTICLE III | |
| | CONDITIONS OF LENDING | |
| Section | 3.01 - Conditions Precedent to the Initial A Borrowing | 9 |
| | 3.02 - Conditions Precedent to the Initial B Borrowing | 12 |
| | 3.03 - Conditions Precedent to Borrowings Other than the Initial | 80. |
| | A Forrowing and the Initial B Borrowing | 14 |
| Section | 3.04 — Additional Conditions Precedent | 15 |
| | ARTICLE IV | |
| | EVENTS OF DEFAULT | |
| Section | 4.01 - Events of Default | 16 |
| OCCUPATION . | TO SECURITY DESCRIPTION OF THE PROPERTY OF THE | 10 |

TABLE OF CONTENTS - Continued

ARTICLE V

THE AGENT

| Section 5.01 – Authorization and Action Section 5.02 – Agent's Reliance, Etc. Section 5.03 – Citibank and Affiliates Section 5.04 – Bank Credit Decision Section 5.05 – Indemnification Section 5.06 – Successor Agent | Page 18 18 18 19 19 |
|--|--|
| ARTICLF VI | |
| Section 6.09 – Headings | 20 20 20 21 21 22 22 22 22 |

List of Exhibits

- A. Domestic Note.
- B. Eurodollar Note.
- C. Instrument of Conveyance of Production Payment.
- D. Mortgage. Deed of Trust and Assignment of Production.
- E. Notice of Borrowing.
- F. Certificate of Officers of the Company.
- G. Certificate of Officers of the Grantors.
- H. Letter of Process Agent.
- I. Opinion of Messrs. Baker & Botts, counsel to the Grantors.
- Opinion of Messrs. Baker & Botts, counsel to the Company.
- K. Opinion of Messrs. Lyons, Pipes & Cook, special Alabama counsel.
- L. Opinion of Messrs. Rose, Nash, Williamson, Carroll, Clay & Giroir, special Arkansas counsel.
- M. Opinion of Messrs. Liskow & Lewis, special Louisiana counsel.
- N. Opinion of Messrs. Brunini, Grantham. Grower & Hewes, special Mississippi counsel.
- Opinion of Messrs. Crowley, Haughey, Hanson, Toole & Dietrich, special Montana counsel.
- P. Opinion of Messrs. Hinkle, Cox, Eaton, Coffield & Hensley, special New Mexico counsel.
- Q. Opinion of Messrs. Shearman & Sterling, special New York counsel.
- R. Opinion of Messrs. Fleck, Mather, Strutz & Mayer, Ltd., special North Dakota counsel.
- Opinion of Messrs, Reed Smith Shaw & McClay, special Pennsylvania counsel.
- T. Opinion of Messrs. Baker & Botts, special Texas counsel.
- U. Opinion of Messrs. Jackson, Kelley, Holt & O'Farrell, special West Virginia counsel.
- V. Opinion of Messrs. Brown, Drew, Apostolos, Massey & Sullivan, special Wyoming counsel.
- W. Opinion of Messrs. Shearman & Sterling, counsel to the Agent.
- X. Opinion of Messrs. Baker & Botts, counsel to the Company.
- Y. Opinion of Messrs. Baker & Botts, counsel to the Grantors.
- Z. Opinion of Messrs. Baker & Botts, special Texas counsel.
- AA. Opinion of Messrs. Shearman & Sterling, special counsel to the Agent.
- BB. Opinion of Messrs. Shearman & Sterling, special New York counsel.

CREDIT AGREEMENT

Dated as of July 16, 1981

New Dunstyn, Inc., a Delaware non-profit corporation (the "Company"), the banks (the "Banks") listed on the signature pages hereof, and Cuthank, N.A. ("Citibank"), as agent (the "Agent") for the Banks hereunder, agree as follows:

ARTICLE I

CERTAIN DEFINITIONS AND ACCOUNTING TERMS

Section 1.01. Certain Defined Terms. As used in this Credit Agreement, unless otherwise defined herein, the tollowing terms have the following meanings (such meanings to be equally applicable to both the singular and plural forms of the terms defined):

- "A Advance" has the meaning specified in Section 2.01(a).
- "A Borrowing" means a Borrowing comprised of A Advances.
- "A Commitment" means, with respect to any Bank, at any time from the date of this Credit Agreement to and including the Termination Date, the amount set opposite such Bank's name on the signature pages hereof under the heading "A Commitment", as such amount may be, or may have been, reduced at or prior to such time pursuant to Section 2.03.
- "A Loan" means, at any time for the determination thereof, the aggregate principal amount of all A Advances made by the Banks.
 - "Advance" means an A Advance or a B Advance.
 - "B Advance" has the meaning specified in Section 2.01(b).
 - "B Borrowing" means a Borrowing comprised of B Advances.
- "B Commitment" means, with respect to any Bank, at any time from the date of this Credit Agreement to and including the Termination Date, the amount set opposite such Bank's name on the signature pages hereof under the heading "B Commitment", as such amount may be, or may have been, reduced at or prior to such time pursuant to Section 2.03.
- "B Loan" means, at any time for the determination thereof, the aggregate principal amount of all B Advances made by the Banks.
 - "Borrowing" means a borrowing by the Company of simultaneous Advances.
- "Citibank New York" means the principal office of Citibank, N.A. in New York City, located on the date hereof at 399 Park Avenue, New York, New York 10043.
- "Conveyance" means the Instrument of Conveyance of Production Payment, dated the date hereof, among the Grantors and the Company, substantially in the form attached hereto as Exhibit C texcluding for such purposes Exhibit A thereto), including, from and after the date of the execution and delivery thereof, the Conveyance Supplement.
- "Conveyance Supplement" means the Supplemental Instrument of Conveyance of Production Payment, dated on or prior to the date of the initial B Borrowing, by the Grantors, substartially in the form attached to the Conveyance as Exhibit C.

"Domestic Note" has the meaning specified in Se-

"Drilling Operations Certificate" means the proceedings of the date hereof of the Granters (other than Duval Corporation) with respect to, among other things, the operation of the wells located on the Subject Interests in Hydrocarbons in accordance with applicable laws, rules and regulations.

"Eurodollar Lending Office" means, with respect to any Bank, the office of such Bank located at its address specified on the signature pages of this Agreement under the heading "Eurodollar Lending Office" or any other office or affiliate of such Bank hereafter selected and notified to the Company and the Agent from time to time by such Bank.

"Eurodollar Note" has the meaning specified in Section 2.12.

"Event of Default" has the meaning specified in Section 4.01.

"Evoluted Traves" has the meaning specified in Section 2.10.

"Grantors" means Pennzoil Company, Pennzoil Producing Company, Pennzoil Oil & Gas, Inc. and Duvai Corporation, each a Delaware corporation.

"Leading Office" means (except as used otherwise in Section 2.10), with respect to any Bank, the Domestic Leading Office, Eurodollar Leading Office or affiliate of such Bank located at the address specified on the signature pages of this Agreement or any other office or affiliate of such Bank hereafter selected and notified to the Company and the Agent from time to time by such Bank.

"Letters in Lieu of Transfer and Division Orders" means those undated Letters in Lieu of Transfer and Division Orders signed by each of the Grantors (except Duval Cerporation), the Company, the Trustees and the Agent, to the purchasers of production from the Subject Interests in Hydrocarbons, notifying said purchasers of the interests of the Company, the Trustees, the Agent and the Banks in and to the sales proceeds and authorizing said purchasers to continue to pay the sales proceeds to such Grantors until further notice by the Agent.

"Majority Banks" means at any time Banks holding at least 6623% of the then aggregate unpaid principal amount of the Notes held by all of the Banks, or, if no such principal amount is then outstanding, Banks having at least 6673% of the Total Commitments.

"Mortgage" means the Mortgage, Deed of Trust and Assignment of Production dated the date hereof, from the Company to the Trustees and the Agent, substantially in the form attached hereto as Exhibit D, providing, among other things, for the mortgage by the Company to the Trustees of the Production Payment and the assignment by the Company to the Agent of Production Payment Minerals and the proceeds thereof, as security for the repayment of the Notes and certain other indebtedness of the Company specified in Section 1.1 thereof, including from and after the execution and delivery thereof, the Mortgage Supplement.

"Mortgage Supplement" means the Supplemental Mortgage, Deed of Trust and Assignment of Production, dated on or prior to the date of the initial B Borrowing, from the Company to the Trustees and the Agent, substantially in the form attached to the Mortgage as Exhibit A.

- "Notes" has the meaning specified in Section 2.12.
- "Notice of Borrowing" has the meaning specified in Section 2.02.
- "Tracess Agent" means CT Corporation System in its capacity as process agent for the Company pursuant to Section 8.11 of the Mortgage.
- "Tax In a mity Letter cans that certain Letter Agreement, dated as of the date hereof, among the Granters a the Company, providing, among other things, for the indemnification of the Company from all liabilities for the payment of certain taxes referred to therein.
 - "Taxes" has the meaning specified in Section 2.10.
- "Termination Date" means the earliest of (a) July 31, 1983, (b) the date on which the Commitment of eac! Bank shall be terminated pursuant to Section 2.03 or 4.01, or (c) the date of a notice from the Grantors to the Company under Paragraph 25 of the Conveyance, to the effect that they intend to make a payment to the Company spacient to discharge the Production Payment in full pursuant to said Paragraph 25.
- "Total Commitment" means, for each Bank, the sum of its A Commitment and its B Commitment.
 - "Trustees" means the trustees acting under the Mortgage.
- Szerios 1.02. Terms Defined in the Conveyance, "Applicable Rate", "Application Date", "Rusiness Day", "Domestic Rate", "Encodellar Rate", "Excess Rate", "Exhibit A", "Hydrocarbons", "LIBO Rate", "Monerals", "Next Preceding Business Day", "Person", "Permany Sure", Production Payment", "Production Payment Minerals", "Purchase Prov", "Rate", "Rate Period", "Subject Interests" and "Subject Sulphur" have the respective meanings specified in the Conveyance
- Section 1.03. Accounting Terms. All accounting terms not specifically defined herein shall be construed in accordance with generally accepted accounting principles consistently applied.

ARTICLE II

Amounts and Terms of the Advances

- SECTION 2.01. Advances and Commitment. (a) Each Bank: "ally agrees, on the terms and conditions hereinafter set forth, to make advances ("A news") to the Company from time to time during the period from the date hereof to and including the Termonation Date, in an aggregate principal amount not to exceed at any time outstanding such Bank's A Commitment; proseded, however, that no Bank shall be obligated to make an A Advance if the sum of such A Advance and all other A Advances of such Bank outstanding hereunder would be in excess of such Bank's A Commitment.
- (b) Each Bank severally agrees, on the terms and conditions hereinafter set forth, to make advances and Bath success?) to the Company from time to time during the period from the day, hereof to and helpfling the Termination Date, in an aggregate principal

principal amount not to exceed at any time outstanding such Bank's B Commitmen provided, however, that no Bank shall be obligated to make a B Advance if the sum of such B Advance and all other B Advances of such Bank outstanding hereunder would be in excess of such Bank's B Commitment.

- (c) Each Borrowing shall be in an aggregate amount of not less than \$50,000,000 (except as provided in paragraph (d) of this Section 2.01 and except that any Borrowing which would cause the aggregate amount of the A Loan to equal \$1,000,000,000 or \$1,865,000,000 or the aggregate amount of the B Loan to equal \$635,000,000 may be made in such lesser amount as shall cause the aggregate amount of the A Lean to equal \$1,000,000,000 or \$1,865,000,000 or the aggregate amount of the B Loan to equal \$635,000,000, as the case may be) and shall consist of Advances made on the same day by each of the Banks ratably according to their respective Commitments; provided, however, that no Bank shall be obligated to make an Advance if the sum of such Advance and all other Advances of such Bank outstanding hereunder would be in excess of such Bank's Total Commitment.
- (d) The initial Borrowing shall be in the aggregate amount of \$10,000,000 and no further Borowings shall be made until evidence that the requirements of Section 4.01(b) have been met has been received by the Company and the Agent.

Section 2.02. Making the Advances. (a) Each Borrowing . . !! be made on at least four Business Days' notice texcept in the case of the initial A Advance too which two Business Days' notice shall be given) from the Company to tra Agent, which thell give prompt notice thereof, and of each other notice received from the Company at entering tender, to each Bank. Each such notice (a "Nother of Borrowing") shall a substantially the form of Exhibit E hereto, specifying the date and aggregate amount of the propo ed Borr wing. Subject to the respective provision of the definitions of Applicable Rat., L/30 Rat. and Rate Period referred to in Section 1.02, the Company shall select the Ray which is to be applicable to the Advances to be made as part of such Borrowing and it and to the extent that the Eurodollar Rate is to be applicable thereto, the Rate Period therefor, Not later than 11-00 A.M. (New York City time) on the date of such Berrowing, each Bank (through the Eurodollar Lending Office of such Bank to the extent that the Eurodollar Rate is to be applicable to the Advances to be made as part of such Borrowing) shall make available to the Agent at Citibank New York such Bank's ratable share of such Borrowing, in immediately available funds. After the Agent's receipt of such funds and upon fulfillment of the applicable conditions set forth in Article III, the Agent will make such funds available to the Company at Citionak New York, in like funds as so received

(b) Each Notice of Borrowing shall be irrevocable and binding on the Company, and, in respect of any proposed Porrowing to be comprised of Advances bearing interest at the Eurodollar Rate, the Company shall indemnify each Bank against any loss or expense incurred by such Bank as a result of any failure to fulfill on or before the date specified for such Borrowing the applicable conditions set forth in Article III, including, without limitation, any loss (including loss of anticipated profits) or expense incurred by reason of the liquidation or reemployment of deposits or other funds acquired by such Bank to fund the Advance to be made by such Bank as part of such Borrowing when such Advance, as a result of such failure, is not made on such date.

- (c) Unless the Agent shall have received notice from a Bank prior to the date of any Borrowing that such Bank will not make available to the Agent such Bank's ratable portion of such Borrowing, the Agent may assume that such Bank has made such portion available to the Agent on the date of such Borrowing in accordance with and to the extent provided in subsection (a) of this Section 2.02 and the Agent may, in reliance upon such assumption, make available to the Company on such date a corresponding amount. If and to the extent cach Bank shall not have so made such ratable portion available to the Agent, such Bank and the Company severally agree to repay to the Agent forthwith on demand such corresponding amount together with interest thereon, for each day from the date such amount is made available to the Company until the date such amount is repaid to the Agent, at the Rate applicable at the time to the Advances comprising such Borrowing. If such Bank shall repay to the Agent such corresponding amount, such amount so repaid shall constitute such Bank's Advance as part of such Borrowing for purposes of this Agreement.
- (d) The failure of any Bank to make an Advance hereunder shall not relieve any other Bank of its obligation to do so.

Section 2.03. Reduction of the Commitments. The Company may at any time after the date of the initial Borrowing, upon at least four Business Days' notice to the Agent, terminate in whole or reduce ratably in part the respective Λ Commitments or the B Commitments of $\dot{}^+$ Banks then in effect; provided, however, that no reduction in the Λ Commitments shall be made unless the B Commitments shall, at or prior to the date of such reduction, have been terminated in whole, and provided further, that any partial reduction in either the Λ Commitments or the B Commitments shall be in the aggregate amount of \$50,000,000 or an integral multiple thereof.

Section 2.04. Use of Proceeds of Advances. The Company covenants and agrees to use the proceeds of each Advance solely for the purpose of making installment payments on account of the Purchase Price of the Production Payment pursuant to Paragraph 7 of the Conveyance.

Section 2.95. Commitment Fee: The Company shall pay to the Agent for the account of each Bank a commitment fee at the rate of 3₈ of 1% per annum, on the average daily unused portion of such Bank's Total Commitment in effect from time to time, payable monthly in arrears from July 16, 1981 until the Termination Date, on each Application Date during such term and on the Termination Date (or, if the Termination Date is not an Application Date, on the Application Date next succeeding the Termination Date). All computations of commitment fee becaunder shall be made by the Agent on the basis of a year of 365-366 days for the actual number of days (including the first day but excluding the last day) elapsed.

Section 2.06. Repairment of the Advances; Interest. (a) Unless sooner prepaid in accordance with the provisions of Section 2.07 hereof, or unless the maturity of the Notes is accelerated in accordance with the provisions of Section 4.01 hereof, the aggregate principal amount of the Advances made by each Bank shall be repayable in a lump sum on June 1, 1990, unless the Conveyance Supplement has not been executed and delivered by the Grantors on or prior to July 31, 1983, in which case, the aggregate principal amount of the Advances made by each Bank shall be repayable in a lump sum on February 1, 1990.

- (b) Each Advance or portion thereof shall bear interest, payable on the respective dates specified in the appropriate Note for such Advance or portion thereof, at the Rate in effect under the Conveyance from time to time for that portion of the Primary Sum funded with such Advance or such portion thereof. The Company shall deliver to the Agent, promptly after receipt thereof by the Company, a copy of any notice from the Grantors required by the Conveyance with respect to the Applicable Rate thereunder.
- (c) The Notes and accrued interest thereon, and certain other amounts payable hereunder and under the Notes, are subject to payment or prepayment from time to time by the application of monies actually received by or on behalf of the Banks under the Mortgage by virtue of the assignment in Article III of the Mortgage, in the manner, at the time and to the extent set forth in the Mortgage.

Section 2.07. Prepayments. In the event that any Grantor shall make a payment to the Company to be applied to the discharge of the Production Payment pursuant to Paragraphs 9 or 25 of the Conveyance, the Company shall, on the date on which such payment is received, make a prepayment in the amount of such payment.

Section 2.08, Payments and Computations, (a) The Company shall make each payment to be made by it hereunder and under the Notes not later than 11:00 A.M. (New York City time) on the day when due in United States dollars to the A ort at Citibank New York in immediately available funds. The Agent will promptly after its receipt thereof distribute to each Bank (for the account of the Eurodollar Lending Office of such Bank in the case of any such payment in respect of an Advance (or portion of an Advance) made by such Bank to which the Eurodollar Rate is applicable) its ratable share of each such payment received by the Agent for the account of the Banks.

- (b) If for any leason the Company makes any payment hereunder with respect to which the Eurodol. Rate is applicable on a date other than the last day of the Rate Period applicable ereto, the Company agrees to pay the Banks on the Application Date next so ceeding any demand therefor an amount computed on the basis of the Excess Rate applied to the amount of such payment for the period commencing with the date of such payment and ending on the last day of the current Rate Period applicable theret, with interest at the Applicable Rate from the date of the payment to which the Eurodollar Rate was applicable.
- (c) Each notice pursuant to the last sentence of Paragraph 6 of the Conveyance of a payment with respect to which the Eurodollar Rate is applicable shall be given to the Agent and shall be irrevocable and binding with respect to the corresponding payment to be made by the Company pursuant to Section 2.67 hereof. The Company agrees to pay the Banks on the Application Date next succeeding any demand therefor as amount sufficient to indemnify each Bank against any loss or expense incurred by snow Bank as a result of the Company's failure to make such payment on the date specified in such notice.

Section 2.09. Payment on Non-Business Days. Whenever any payment to be made hereunder or under the Notes shall be stated to be due on a day other than a Business Day, such payment shall be made on the Next Preceding Business Day.

Section 2.10. Taxes. (a) The Company agrees to bear all present or future taxes, levies, imposts, deductions, charges, withholdings or other governmental charges

- ("Taxes") imposed on a Bank or the Agent (the term "Bank" being used hereinafter in this Section 2.10 also to encompass the Agent in its capacity as such) which are attributable to the Credit Agreement, the Notes, payments thereunder or any transactions or other document contemplated therein, other than the Taxes excluded in clauses (i) and (ii) below ("Leeladed Taxes"):
 - (i) Except as projected in clause (iii) below, the Company shall not bear any type of income, withher ling, property or frame'r'se Taxes imposed on a Bank by a Jurisdiction if such Bank is organized under the laws of, or maintains a Lending Office in, such Jurisdiction or any income or withholding Taxes imposed by the United States or a political subdivision thereof on any Bank organized outside the United States which is unable, as of the date of execution hereof, validly to furnish the representations and forms provided for in paragraph (d)(ii) of this Section 2.10. For purposes of this Section 2.10, "Lending Office" means an office or other location which in connection which has physical custody of the Notes or the activities of documents contemplated therein constitute the engaging in, or doing of, bu iness in such Jurisdiction; and "Jurisdiction", when applied to a political subdivision, includes the nation of which it is a part and any overlapping political subdivision, but nation, includes only the nation.
 - (ii) The Company shall not bear any stamp, documentary or other excise Taxes which may arise from the sale or other transfer of a Note by a Bank other than a transfer to the Company or a sale made pursuant to Section 6.05(b).
- (iii) The Company shall bear any property Taxes imposed on a Bank by a Jurisdiction to the extent such property Taxes are imposed as a result of Subject Interests being located in such Jurisdiction.
- (b) If the Company shall be required by law to deduct any Taxes (other than Excluded Taxes) from o, in respect of, any sum payable hereunder or under any Note to any Bank. (i) the sum payable shall be increased as may be necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section 2.10) such Bank receives an amount equal to the sum it would deductions and (iii) the Company shall pay the fell amount deducted to the appropriate taxation authority or other authority in accordance with applicable law and furnish appropriate receipts therefor to the Agent.
- (c) The Company will indemnify each Bank for the full amount of Taxes (other than Excluded Taxes) paid by such Bank or any liability (including penalties, interest and expenses and any increase in Taxes, including Excluded Taxes, to the extent such increase is attributable to amounts payable under this Section 2.10) arising from, or with respect to, such Taxes whether or not such Taxes were correctly or legally asserted. This indemnification payment shall be made within 30 days from the date such Bank makes written demand therefor.
- (d) Each Bank organized outside of the United States represents either (i) that it conducts business in the United States and that interest payable to it hereunder is and will be effectively connected with the conduct of a trade or business in the United States

within the meaning of Section 864(c) of the United States Internal Revenue Code as amended ("Code") and the Regulations thereunder as they currently exist and at interpreted, and that the Bank will execute and deliver to the Company as required from time to time copies of Internal Revenue Service Form 4224 (or any succession of the Code in accordance with applicable U.S. Treasury Regulations, and (ii) payable hereunder to such Bank is not subject to withholding under Section 184 (c) and of the Code by virtue of an applicable meaning text treats and that the Code in accordance with applicable U.S. Treasury Regulations of in figure there are a fixed in accordance with applicable U.S. Treasury Regulations of in figure there are a fineate to be furnished in accordance with any income tax treaty to which the United States is a party.

(e) Without prejudice to the survival of any other arms ment of the Company hereunder, the agreements and obligations of the Company—mained in this Section 2.10 shall survive the payment in full of principal and interes, hereunder and under the Notes.

Section 2.11. Sharing of Pauments, Etc. If any Bank shall obtain any payment (whether voluntary, involuntary, through the exercise of any right of set-off, or otherwise) on account of the Notes held by it in excess of its retable share of payments on account of the Notes obtained by all the Banks, such Bank shall purchase from the other Banks such participations in the Notes held by them as shall be necessary to cause such purchasing Bank to share the excess payment ratably with each of them; provided, however, that if all or any portion of such excess payment is thereafter recovered from such purchasing Bank, the purchase shall be rescinded and the purchase price restored to the extent of such recovery, together with interest if interest thereon is required to be paid by such Bank pursuant to court order. The Company agrees that any Bank so purchasing a participation from another Bank pursuant to this Section 2.11 may, to the fullest extent permitted by law, exercise all its rights of payment (including the right of set-off) with respect to such participation as fully as if such Bank were a direct creditor of the Company in the amount of such participation.

Section 2.12. Evidence of Debt. (a) The indebtedness of the Company resulting from each Advance made by each Bank shall, so long as and to the extent that such Advance hears interest at the Domestic Rate, be evidenced by a promissory note of the Company payable to the order of such Bank, substantially in the form of Exhibit A hereto (a "Domestic Note").

- (b) The indehtedness of the Company resulting from each Advance made by each Bank shall, so long as and to the extent that such Advance bears interest at the Eurodollar Rate, be evidenced by a promissory note of the Company payable to the order of such Bank, substantially in the form of Exhibit B hereto (a "Eurodollar Note", the Eurodollar Notes, together with the Domestic Notes, being collectively the "Notes" and individually a "Note").
- (e) All Advances made by each Bank shall be recorded by such Bank and all adjustments required by changes in the Rate or Rates applicable to such Advances, and all payments made on account of principal thereof, shall be recorded by such Bank. Prior to the transfer by any Bank of its Notes, a record of the Advances evidenced thereby shall be endorsed by such Bank on the grids attached thereto and forming a part of such Notes, Said adjustments to a Note evidencing Advances bearing interest at a particular Rate

shall consist of additions of principal amounts due to changes in the interest rate applicable to such Advances from the other Rate to such Rate, and shall consist of subtractions of principal amounts due to changes in the interest rate applicable to such Advances from such Rate to the other Rate.

ARTICLE III

CONDITIONS OF LENDING

Section 3.01. Conditions Precedent to the Install | Borrowing. The obligation of each Bank to make its initial A Advance is subject to the conditions precedent that (a) all conditions precedent contained in Paragraph 7 of the Conveyance to the obligation of the Company to pay the installment of the Purchase Price of the Production Payment to be paid with the proceeds of the Borrowing of which such Advance is a part shall have been satisfied, (b) the Grantors shall have paid the Agent a management fee in the amount of \$750,000 and (c) the Agent shall have received at Citibank New York all of the following, each dated (unless otherwise indicated below) the date of the initial A Borrowing and each in form and substance satisfactory to the Agent and (except for the Notes) in sufficient copies for each Bank:

- (i) A Domestic Note and a Eurodollar Note drawn to the order of each Bank.
- (ii) The Conveyance, duly executed by the Grantors and the Company and delivered.
- (iii) The Mortgage, duly executed and delivered by the Company, the Trustees and the Agent.
- (iv) The Letters in Lieu of Transfer and Division Orders, duly signed on behalf of the Grantors (except Daval Corporation), the Company, the Trustees and the Agent.
 - (v) The Tax Indemnity Letter, duly executed by the Grantors and the Company.
- (vi) The balance sheet of the Company as of the date of this Credit Agreement, certified by a duly authorized officer of the Company.
- (vii) The written confirmatory Notice of Borrowing with respect to the initial A Borrowing, dated at least two Business Days prior to the initial Λ Borrowing.
- (viii) Each opinion, certificate, notice and other document required pursuant to the Conveyance to be furnished to the Company by the Grantors in connection with the payment of the installment of the Purchase Price of the Production Payment to be paid with the proceeds of the initial A Borrowing, dated as provided in the Conveyance.
- (ix) The Certificate of Lacorporation of the Company and each amendment thereto, certified (as at a date reasonably near the date of the initial A Borrowing) by the Secretary of State of the State of Delaware as being a true and correct copy thereof.
- (x) A certificate of the Secretary of State of the State of Delaware, dated reasonably near the date of the initial A Borrowing, listing the Certificate of Incorporation of the Company and each amendment thereto on file is his office and certifying that (Λ) such amendment—are the only amendments to the Certificate of Incorporation of the

Company on file in his office, (B) the Company has paid all franchise taxes, if any, to the date of such certificate and (C) the Company is duly incorporated and validly existing in good standing under the laws of the State of Delaware.

- (xi) The Certificate of Incorporation of each Grantor and each amendment thereto, certified (as at a date reasonably near the date of the initial A Borrowing) by the Secretary of State of the State of Delaware as being a true and correct copy thereof.
- (xii) A certificate of the Secretary of State of the State of Delaware, dated reasonably near the date of the initial A Borrowing, listing the Certificate of Incorporation of each Grantor and each amendment thereto on file in his office and certifying that (A) such amendments are the only amendments to the Certificate of Incorporation of such Grantor on tile in his office, (B) such Grantor has paid all franchise taxes, if any, to the date of such certificate and (C) such Grantor is duly incorporated and in good standing under the laws of the State of Delaware.
 - (xiii) A certificate of the President or any Vice President of the Company and the Secretary or any Assistant Secretary of the Company, the statements in which shall be true and correct, substantially in the form of Exhibit F hereto.
 - (xiv) A certificate of the President or any Vice President or the Treasurer of each of the Grantors and the Secretary or any Assistant Secretary of each of the Grantors, the statements in which shall be true and correct, substantially in the form
 - (xv) A Certificate of Ryder Scott Company Petroleum Engineers, to the effect of Exhibit G hereto. that all of the properties and interests with respect to which value was given in their report dated June 29, 1981, entitled "Estimated Future Net Production and Income Data Attributable to the Leasehold and Royalty Interests of Pennzoil Exploration and Production Company U.S.A. Onshore and Offshore Beginning January 1, 1981." submitted to the Company and the Bank in connection with the negotiations for the purchase and finance of the Production. Payment are Subject Interests (or, if less than all, specifying - differences and confirming that the net effect of such differences is negligible).
 - (xvi) A Certificate of the Grantors to the effect that all of the properties and interests with respect to which value was given in engineering and financial reports submitted by the Grantors to the Company and the Banks with respect to sulphur in connection with the negotiations for the purchase and financing of the Production Payment are included in the Subject Sulphur (or, if less than all, such certificate shall confirm that the net effect of the difference is negligible); and a letter of DeGolyer and MacNaughton to the effect that all of the properties and interests included in their letter report dated July 2, 1981 regarding proved sulphur reserves as of January 1, 1981 are included in the Subject Sulphur (or, if less than all, such certificate shall confirm that the net effect of the difference is negligible).
 - (xvii) A copy of the Drilling Operations Certificate duly signed on behalf of each of the Grantors (except Duval Corporation).
 - (xviii) Λ letter from the Process Agent, substantially in the form of Exhibit H hereto, agreeing to act as Process Agent and to forward forthwith all process received by at to the Company. 10

- (xix) An opinion of Messrs, Baker & Botts, counsel to the Grantors, addressed to the Banks and the Agent, substantially in the form of Exhibit I bereto and as to such other matters as any Bank, through the Agent, may reasonably request.
 - (xx) An opinion of Messrs, Baker & Botts, counsel to the Company, addressed to the Banks and the Agent, substantially in the form of Exhibit J hereto and as to such other matters as any Bank, through the Agent, may reasonably request,
 - (xxi) An opinion of Messrs, Lyons, Pipes & Cook, special Alabama counsel, addressed to the Bauks and the Agent, substantially in the form of Exhibit K hereto (with the blanks therein appropriately filled in) and as to such other matters as any
 - (xxii) An opinion of Mosses, Rose, Nash, Williamson, Carroll, Clay & Giroir, Bank, through the Agent, may reasonably request. special Arkansas counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit L hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
 - (xxiii) Au opinion of Messrs, Liskow & Lewis, special Louisiana counsel, addressed to the Bauks and the Agert, substantially in the form of Exhibit M hereto (with the blanks therein appropriately filled in) and as to such other matters as any
 - (xxiv) An opinion of Messrs, Brunini, Grantham, Grower & Howes, speci-Bank, through the Agent, may reasonably request. Mississippi counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit N hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
 - (xxv) An opinion of Messrs, Crowley, Haughey, Hanson, Toole & Dietrich, special Montana counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit O hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
 - (XXVI) An opinion of Messrs, Hinkle, Cox, Eaton, Coffield & Hensley, special New Mexico rounsel, addressed to the Banks and the Agent, substantially in the form of Exhibit P hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
 - (xxvii) An opinion of Messrs, Shearman & Sterling, special New York counsel. addressed to the Banks and the Agent, substantially in the form of Exhibit Q hereto (with the blanks therein appropriately filled in) and as to such other matters as any
 - (xxviii) An opinion of Messrs, Fleck, Mather, Strutz & Mayer, Ltd., special North Bank, through the Agent, may reasonably request. Dakota counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit R hereto (with the blanks therein appropriately filled in) and as to such
 - other matters as any Bank, through the Agent, may reasonably request. (XXIX) An opinion of Messers, Rocal Smith Shaw & McClay, special Ponnsylvania counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit S
 - hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
 - (XXX) An opinion of Messes, Baker & Botts, special Texas counsel, addresses to the Banks and the Agent, substantially in the form of Exhibit T hereto (with the

blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.

- (xxxi) An opinion of Messrs, Jackson, Kelley, Holt & O'Farrell, special West Virginia counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit U hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
- (xxxii) An opinion of Messrs. Brown, Drew, Apostolos, Massey & Sullivan, special Wyoming counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit V hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
- (xxxiii) An opinion of Messrs. Shearman & Sterling, counsel to the Agent, addressed to the Banks and the Agent, substantially in the form of Exhibit W hereto.
- (xxxiv) Such other instruments and documents (including, without limitation, documents evidencing any necessary governmental authorizations, consents, approvals, licenses and exemptions) as any Bank, through the Agent, may reasonably request.

Section 3.02. Conditions Precedent to the Initial B Borrowing. The obligation of each Bank to make its initial B Advanc — subject to the coaditions precedent that (a) all conditions precedent contained in Paragraph 7 of the Conveyance to the obligation of the Company to pay the installment of the Purchase Price of the Production Payment to be paid with the proceeds of the Borrowing of which such B Advance is a part shall have been satisfied, (b) the A Commitments of the Banks shall have been or, contemporaneously with the initial B Borrowing will be, fully utilized, and (c) the Agent shall have received at Citibank New York all of the following, each dated on or prior to the date of the initial B Borrowing and each in form and substance satisfactory to the Agent and in sufficient copies for each Bank:

- (i) The Conveyance Supplement, duly executed by the Grantors and delivered, together with:
 - (A) evidence that the Conveyance Supplement has been duly filed for record in the State of Texas where necessary to establish, preserve and protect the rights thereunder of the parties thereto (it being understood that the opinion of Messrs, Baker & Botts provided for in clause (ix) of this Section 3.02 shall constitute such evidence);
 - (B) acknowledgment copies of proper Financing Statements duly filed under the Uniform Commercial Code of the State of Texas with respect to certain interests of the Company under the Conveyance Supplement; and
 - (C) certified copies of Requests for Information or Copies (Form UCC-11), or certificates from a reliable commercial Uniform Commercial Code search service, listing the Financing Statements referred to in subclause (B) of this clause (i) and all other effective financing statements which name the Grantors (under their present names and any previous names) as debtors and which are filled in the jurisdiction referred to in said subclause (B) together with copies of such other financing statements (none of which shall cover the interests intended to be covered by said Financing Statements).
- (ii) The Mortgage Supplement, duly executed by the Company and delivered, together with:

- (A) evidence that the Mortgage Supplement has been filed for record in the State of Texas where necessary in order to establish and perfect the security interests intencied to be created thereby (if being understood that the opinion of Messrs, Baker & Botts provided for in clause (ix) of this Section 3.02 shall constitute such evidence);
- (B) acknowledgment copies of proper Financing Statements duly filed under the Uniform Commercial Code of the State of Texas with respect to certain interests of the Agent and the Trustees under the Mortgage Supplement; and
- (C) certified copies of Requests for Information or Copies (Form UCC-11), or certificates from a reliable commercial Uniform Commercial Code search service, listing the Financing Statements referred to above in subclause (B) of this clause (ii) and all other effective financing statements which name the Company (under its present name and any previous name) as debtor and which are filed in the jurisdiction referred to in said subclause (B), together with copies of such financing statements (none of which shall cover the interests intended to be covered by said Financing Statements).
- (iii) The written confirmatory Notice of Borrowing with respect to the initial B Borrowing, dated at least four Business Days prior to the initial B Borrowing.
- (iv) Each opinion, certificate, notice and other document required pursuant to the Conveyance to be furnished to the Company by the Grantors in connection with the payment of the installment of the Purchase Price of the Production Payment to be paid with the proceeds of the initial B Borrowing, dated as provided in the Conveyance.
- (v) The Certificate of Incorporation of Duval Corporation and each amendment thereto, certified (as at a date reasonably near the date of the initial B Borrowing) by the Secretary of State of the State of Delaware as being a true and correct copy thereof.
- (vi) A certificate of the Secretary of State of the State of Delaware, dated reasonably near the date of the initial B Borrowing, listing the Certificate of Incorporation of Duval Corporation and each amendment thereto on file in his office and certifying that (A) such an adments are the only amendments to the Certificate of Incorporation of Duval Corporation on file in his office, (B) Duval Corporation has paid all franchise taxes, if any, to the date of such certificate and (C) Duval Corporation is duly incorporated and in good standing under the laws of the State of Delaware.
- (vii) A certificate of the President or any Vice President of the Company and the Secretary or any Assistant Secretary of the Company, dated the date of the initial B Borrowing, the statements in which shall be true and correct, substantially in the form of Exhibit F hereto.
- (viii) A certificate of the President or any Vice President or the Treasurer of each of the Grantors and the Secretary or any Assistant Section of each of the Grantors, dated the date of the initial B Borrowing, the states is in which shall be true and correct, substantially in the form of Exhibit G hereto.
- (ix) An opinion of Messrs, Baker & Botts, counsel to the Company, addressed to the Banks and the Agent, dated the date of the initial B Borroving, substantially

in the form of Exhibit X herete and as to such other matters as any Bank, through the Agent, may reasonably request.

- (x) An opinion of Messrs, Baker & Botts, counsel to the Grantors, addressed to the Banks and the Agent, dated the date of the initial B Borrowing, substantially in the form of Exhibit Y hereto and as to such other matters as any Bank, through the Agent, may reasonably request.
- (xi) An opinion of Messrs, Baker & Botts, special Texas counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit Z hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request.
- (xii) An opinion of Messrs, Shearman & Sterling, special counsel to the Agent, addressed to the Banks and the Agent, substantially in the form of Exhibit AA hereto.
- (xiii) Such other instruments and documents (including, without limitation, documents evidencing any necessary governmental authorizations, consents, approvals, licenses and exemptions) as any Banks, through the Agent, may reasonably request.

Section 3.03. Conditions Precedent to Borrowings Other than the Initial A Borrowing and the Initial B Borrowing. The obligation of each Bank to make an Advance on the occasion of each Borrowing subsequent to the initial A Borrowing (other than its initial B Advance) is subject to the conditions precedent that (a) all conditions precedent contained in Paragraph 7 of the Conveyance to the obligation of the Company to pay the installment of the Purchase Price of the Production Payment to be paid with the proceeds of the Borrowing of which such Advance is a part shall have been satisfied, (b) the statements made in the certificates delivered pursuant to clauses (iii) and (iv) below shall be true and correct on the date of such Borrowing and (c) the Agent shall have received at Citbank New York all of the following, each dated (unless otherwise indicated below) the date of such Borrowing and each in form and substance satisfactory to the Agent and in sufficient copies for each Bank:

- (i) The written confirmatory Notice of Borrowing with respect to such Borrowing, dated at least four Business Days prior to such Borrowing.
- (ii) Each opinion, certificate, notice and other document required to be furnished pursuant to the Conveyance to the Company by the Grantors in connection with the payment of the installment of the Purchase Price of the Production Payment to be to 34 with the proceeds of such Advance, dated as provided in the Conveyance.
- (iii) A certificate of the President or any Vice President and the Secretary or Assistant Secretary of the Company, substantially in the form of Exhibit F nereto.
- (iv) A certificate of President or any Vice President or the Treasurer of each of the Grantors and the Secretary or any Assistant Secretary of each of the Grantors substantially in the form of Exhibit G hereto.
- (v) in the event of any change after the date hereof in any applicable law, rule or regulation which might materially adversely affect the rate of discharge of the Production Payment or the rights of the Banks hereunder, such other instruments and documents related to such change (including, without limitation, documents evidencing

any necessary governmental authorizations, consents, approvals, lie and exemptions necessitated by such change) as the Majority Banks, through Agent, may reasonably request.

(vi) such other documents and evidence as the Majority Panks, through the Agent, may reasonably request, in order to establish compliane with the conditions set forth herein and in the Conveyance.

Section 3.04. Additional Conditions Precedent. The obligation of each Bank to make an A Advance on the occasion of any Borrowing which would increase the A Loan to an amount equal to or greater than \$1,000,000,000 is subject to the conditions precedent that (a) each of the Grantors having Subject Interests located in the State of New York, and, if necessary or advisable, the Company, are duly qualified to do business as foreign corporations in, and are in good standing under the laws of, the State of New York, (b) the Grantors' representations and warranties contained in Paragraph 20 of the Conveyance as to their title to the Subject Interests located in the State of New York and the Production Payment were true on the date of the Conveyance, (c) none of the Granters has executed any document purporting to convey title to or any interest in the Subject Interests located in the State of New York or the Production Payment to the extent attributable to such Subject Interests, the Grantors have paid all taxes that are due and payable with respect to the New York Subject Interests and no rights of any third party have intervened with respect ther to since the execution and delivery of the Cenveyance, (d) the representations and warranties of the Company contained in the Morigage as to factual matters were true on the date of the Mortgage, (c) the Company has not executed any documents purporting to convey title to or any interest in the Subject Interests located in the State of New York or the Production Payment to the extent attributable to such Schoot Interests, the Company has paid ail taxes that are due and payable with respect to the Mortgaged Property (as defined in the Mortgage), and no rights of any third pacty have intervened with respect thereto since the execution and delivery of the Mortgage, and (f) the Agent shall have received at Citibank New Yor' . of the following, each dated on or prior to the date of such Borrowing and each in m and substance satisfactory to the Agent and in sufficient copies for each Bank:

- (i) evidence that the Conveyance has been filed for record in the State of New York;
- (ii) acknowledgment copies of proper Financing Statements duly files under the Uniform Commercial Code of the State of New York with respect to certain interests of the Company under the Conveyance;
- (iii) certified copies of Requests for Information or Copies (Form UCC-11) or certificates from a reliable commercial Uniform Commercial Code search service listing the Financing Statements referred to in clause (ii) above and all other effective financing statements which name the Grantors (under their present names and any previous names) as debtors and which are filed in the State of New York, together with copies of such other financing statements (none of which shall cover the interests intended to be covered by said Financing Statements);
- (iv) evidence that the Mortgage has been duly filed for record in the State of New York where necessary to establish, preserve and protect the rights thereunder of the parties thereto;

- (v) acknowledgment copies of proper Financing Statements duly filed under the Uniform Commercial Code of the State of New York with respect to certain interests of the Agent under the Mortgage;
- (vi) certified copies of Requests for Information or Copies (Fo.m UCC-11) or certificates from a reliable connecreial Uniform Commercial Code search service listing the Financing Statements referred to in clause (v) above and all other effective financing statements which name the Company (under its present name and any previous name) as debtor and which are filed in the State of New York, together with copies of such financing statements (none of which cover the interests intended to be covered by said Financing Statements); and
- (vii) An opinion of Messrs, Shearman & Sterling, special . Tew York counsel, addressed to the Banks and the Agent, substantially in the form of Exhibit BB hereto (with the blanks therein appropriately filled in) and as to such other matters as any Bank, through the Agent, may reasonably request,

ARTICLE IV

EVENTS OF DEFAULT

Section 4.01. Events of Default. Any one or more of the following events shall be an Event of Default (each such event being called an "Event of Default") hereunder and under the Notes:

- (a) an Event of Default under and as defined in the Mortgage;
- (b) a failure by the Company and the Agent to receive:
- (i) within 20 days after the date of the initial Λ Borrowing, evidence that the Conveyance has been duly filed for record in the States of Alabama, Arkansas, Louisinaa, Mississippi, Montana, New Mexico, North Dabaa, Pennsylvania, Texas, West Virginia and Wyoming with respect to the Sub_{cont} Interests located in such states where necessary (as indicated in the opinions of counsel delivered pursuant to clauses (xxi) (xxxii) of Section 3,01(c)) to establish, preserve and protect the rights thereunder of the parties thereto;
- (a) within 30 days after the date of the initial A Borrowing, acknowledgment copies of proper Financing Statements duly filed under the Uniform Commercial Code of each of the States of Alabama, Arkansas, Mississippi, Montana, New Mexico, North Dakota, Pennsylvania, Texas, West Virginia and Wyoming with respect to certain interests of the Company under the Conveyance with respect to the Subject Interests located in such states where neces indicated in the opinions of counsel delivered pursuant to clauses (xy) of Section 3.01(c)) to establish, preserve and protect the rights thereunder are parties thereto;
- (ii) within 45 days after the dates i the initial A Borrowing, certified copies of Requests for Information or Copies (Form VCC II) or certificates from a reliable commercial Uniform Commercial Code search service listing over Financiag Statements referred to above in subclause (ii) of this clause (b) and all other effective financing statements which name any of the Grantors (under their present names and any previous name) as debtors and which are filed in the jurisd 'ions referred to in said subclause (ii), together with copies of such other

financing statements (none of which shall ever the interests intended to be covered by said Financing Statements to any extent which would cause the representations and warranties of the Granton contained in subparagraph (a) of Paragraph 20 of the Conveyance to be untrue);

- (iv) within 30 days after the date of the mitial A Borrowing, evidence that the Mortgage has be duly filed for record in the states of Alabama, Arkansas, Louisiana, Mississippo, Montana, New Mexico, North Dakota, Pennsylvania, Texas, West Virginia and Wyoming with respect to the Mortgaged Property located in such states where necessary (as indicated in the opinions of counsel delivered pursuant to clauses (xxi) (xxxii) of Section 3,01(c)) in order to establish and perfect the security interests to be created thereby;
- (v) within 30 days after the date of the initial A Borrowing, acknowledgment copies of proper Financing Statements duly filed under the Uniform Commercial Code of each of the States of Alabama, Arkansas, Mississippi, Montana, New Mexico, North Dakota, Pennsylvania, Texas, West Virginia and Wyoning with respect to certain interests of the Agent under the Mortgage with respect to the Mortgaged Property beated in such states where necessary as indicated in the opinions of counsel delivered pursuant to clauses (XXI) (XXXII) of Section 3.01(c)) in order to establish and perfect the security interests to be created thereby; and
- (vi) within 45 days after to state of the initial A Borrowing, certified copies of Requests for Information or Copies (Form UCC 11), or certificates from reliable commercial Uniform Commercial Code search service, listing the Financing Statements referred to above in subclause (v) of this clause (b) and all other effective financing statements which name the Company (under its present name and any previous name) as debtor and which are filed in the jurisdictions referred to in said subclause (v), together with copy of such other financing statements (none of which shall cover the interests intended to be covered by said Financing Statements to any extent which would cause the representations and warranties of the Grantors contained in subparagraph (a) of Paragraph 20 of the Conveyance to be untrue);

if such non-receipt shall continue for a period of 30 days after written notice thereof shall have been given to the Company; and

(e) a default by the Company in the performance or observance of any term, covenant or cond- on on its part to be performed or observed becoming, and such default shall remain unremoded for a period of 30 days after written notice thereof shall have been given to the Company.

ARTICLE V

THE AGENT

Section 5.01. Authorization and Action. Each Bank hereby appoints and authorizes the Agent to take such action as agent on its behalf and to exercise such powers under this Credit Agreement, the Notes, the Conveyance and the Mortgage as are delegated to the Agent by the terms hereof and thereof, together with such powers as are reasonably incidental thereto. As to any matters not expressly provided for by this Credit Agreement (including, without limitation, enforcement or collection of the Notes), the Agent shall Il be required to act not be required to exercise any discretion or take any action, but or to refrain from acting (and sha be fully protected in so act or refraining from acting) upon the instructions of held rs of a least 662 3% in principa, amount of the Notes then outstanding for if no Notes are at the time outstanding, upon the instructions of Banks having at least 662 get of the Total Commitments), and such instructions shall be binding upon all Banks and all holders of Notes; provided, however, that the Agent shall not be required to take any action which exposes the Agent to personal liability or which is contrary to this Credit Agreement or apdicable law.

Section 5.02. Agent's Reliance, Etc. Neither the Agent nor any of its directors, officers, agents or employees shall be liable for any action taken or omitted to be taken by it or them under or in connection with this Credit Agreement, the Notes, the Conveyance or the Mortgage, except for its or their own gross neglicence or wilful misconduct. Without limiting the generality of the foregoine, the Agent; (a) may treat the passes of any Note as the holder thereof until the Agent receives written notes of the assignment or transfer thereof signed by such pavee and in form satisfactory to the Agent; (b) may consult with legal counsel (including counsel for the Company and counsel for the Grantors), independent public accountants and other experts selected by it and shall not be liable for any action taken or omitted to be taken in good faith by it in accordance with the advice of such counsel, accountants or experts; (c) makes no warranty or representation to any Bank and shall not be responsible to my Bank for any statements, warranties or representations made in or in connection with this Credit Agreement, the Notes, the Conveyance or the Mortgage; (d) shall not have any duty to ascertain or to inquire as to the performance or observance of any of the terms, covenants or conditions of this Credit Agreement, the Notes, the Conveyance or the Mortgage or to inspect the property tincluding the books and records) of the Company or any of the Grantors; (c) shall not be responsil - to any Bank for the due execution, legality, validity, enforceability, genuineness, suffice ney or value of this Credit Agreement, C. Notes, the Conveyance or the Mortgage or any other instrument or document furnished pursuant hereto or thereto; and (f) shall mene no fiability under or in respect of this Credit Agreement, the Notes, the Conveyance or the Mortgage by acting upon any notice, consent, certificate or other instrument or writing (which may be by telegram, cable or telex) believed by it to be genuine and signed or sent by the proper party or parties.

Species 5.03, Citibank and Affiliates, With respect to its A. B and Total Commitments, the Advances made by it and the Notes issued to it, Citibank shall have the same rights and powers under this Credit Agreement, its Notes, the Conveyance and the Mortgane as any other Bank and may exercise the same as though it were not the Agent; and the term "Bank" or "Banks" shall, unless otherwise expressly indicated,

include Citibank in its individual capacity. Cit bank and its affiliates may accept deposits from, lend money to, act as trustee under indentures of, and generally engage in any kind of business with, the Company, the Grantors, any of their respective subsidiaries and any person or entity who may do business with or own securities of the Company, any Grantor or any of their respective subsidiaries, all as if Citibank were not the Agent and without any duty to a count therefor to the Banks.

Suction 5.04. Bank Credit Decision. Each Bank acknowledges that it has, independently and without reliance upon the Agent or any other Bank and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Credit Agreement. Each Bank also acknowledges that it will, independently and without reliance upon the Agent or any other Bank and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Credit Agreement.

Section 5.05. Indomnification. The Banks agree suden aif's the Agent (to the extent not reimbursed by or on behalf of the Company), ratably according to the respective aggregate outstanding principal amounts of the Notes then held by each of them (or, if no Notes are at the time outstanding, ratably according to the respective amounts of their Total Commitments), from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever which may be imposed on, incurred by, or asserted against the Agent in any way relating to or arising out of this Credit Agreement, the Notes, the Conveyance or the Mortgage r any action taken or omitte by the Agent under this Credit Agreement, the Notes, the Conveyance or the Mortgage, provided that to Bank shall be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting from the Agent's gross negligence or wilful misconduct. Without limitation of the foregoing, each Bank agrees to rehiburse the Agent promptly upon demand for its rapable share of any out-of-pocket expenses (including counsel fees) incurred by the Agent in connection with the preparation, execution, filing, recording, registration (or refiling, rerecording or reregistration), administration or enforcement of, or legal advice in respect of rights or responsibilities under. this Credit Agreement, the Notes, the Conveyance or the Mortgage, to the extent that the Agent is not reimbursed for such expenses by or on behalf of the Company.

Section 5.06, Successor Aleent. The Agent may resign at any time by giving written notice thereof to the Banks and the Company and may be removed at any time with or without cause by the Majority Banks, upon appointment by the Majority Banks of a successor Agent or, if Majority Banks are unable to agree, the appointment of a successor Agent by the retiring Agent, which successor. Tent in any case shall be a commercial bank organized under the laws of the United States of America or of any State thereof and having a combined capital and surplus of at least \$50,500,000. Upon the acceptance of any appointment as Agent hereunder by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent, and the retiring Agent shall be discharged from its duties and obligations under and in connection with this Credit Agreement. After any retiring Agent's resignation or removal hereunder as Agent, the provisions of this Article V shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Credit Agreement.

ARTICLE VI

MISCELLANEOUS

Section 6.01. No Waiver: Madifications in Writing; Remedies. No failure or delay on the part of any Bank or the Agent is exercising any right, power or remedy under this Credit Agreement, the Notes, the Conveyance or the Mortgage shall operate as a waiver thereof, nor shall any single or partial exercise of any such right, power or remedy preclude any other or further exercise thereof or the exercise of any other right, power or remedy hereunder or thereunder. No amendment, modification or waiver of any provision of this Credit Agreement or the Notes, nor consent to any departure therefrom, shall in any event be effective unless the same shall be in writing and signed on behalf of the Majority Banks, and then such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, however, that no amendment, modification, waiver or consent shall, unless in writing and signed by all the Banks, do any of the following: (a) waive any of the conditions specified in Article HI; (b) increase the Λ, B or Total Commitments of the Banks or subject the Banks to any additional obligations; (c) reduce the principal of, or interest on, the Notes or any fees bereunder; (d) postpone any date fixed for any payment of principal of, or interest on, the Notes or any fees hereunder; or (e) change the percentage of the Total Commitments or of the aggregate unpaid principal amount of the Notes, or the number of Banks, which shall be required for the Agent or the Banks or any of them to take any action under or in connection with this Credit Agreement. No notice to or demand on the Compare in any case shall entitle the Company to any other or further notice or demand in similar or other circumstates. The remedies herein provided are cumulative and not exclusive of any remedies provided by law,

Section 6.02. Addresses for Notices. All vol.—and other communications provided for hereunder shall, unless otherwise stated herein, ac in writing or by telex, telegram or cable and mailed or sent or delivered, if to the Company, at 100 West Tenth Street, Wilmington, Delaware 19801, Attention: Joseph A. Barbera; if to any Bank, at its address set forth under its name under the heading "Address for Notices" on the signature pages hereof; and if to the Agent, at its address at 399 Park Avenue, New York, New York 10043, Attention; North American Banking Group Energy Central, with a copy to Citicorp (USA), Inc., 2100 Two Allen Center, 1200 Smith Street, Houston, Texas 77002, Attention; Heaston Energy Department, or, as to each such party, at such other address as shall be designated by such party in a written notice to each other party complying as to delivery with the terms of this Section 6.02. All such notices and communications, when mailed or sent by telex, telegram or cuble, shall be effective when deposited in the mails or sent by telex, telegram or cuble, shall be effective when deposited in the mails or sent by telex, telegram or cuble, respectively, addressed as aforesaid, except that notices and communications to the Agent pursuant to any of the previsions of Article II or V shall not be effective until received by the Agent.

Section 6.03, Users and Expenses—a) The Company agrees to pay (i) the reasonable fees and distursement—of Messes, Shearman & Sterling in connection with the matters contemplated by this Credit Agreement; (ii) the reasonable disbursements of the Trustees; (iii) the reasonable fees and disbursements of each of the counsel referred to in Section 5.01(2) (2) (2) through (xxxiii) in connection with the matters contemplated by this Credit Agreement; (iv) all reasonable printing and other expenses in connection with this Credit Agreement, the Notes, the Conveyance and the Mortgage, and all documents re-

ferred to herein or therein or contemplated hereby or thereby or delivered in connection herewith or therewith; (v) all documentary stamp and other taxes (including penalties and interest assessed on any such taxes), and all toos payable in connection with the execution, delivery, filing, recording and registration (and refiling, rerecording, and reregistration) of this Credit Agreement, the Notes, the Conveyance, the Mortgage, financing statements, any continuation statements and any and all other documents referred to herein or therein or contemplated hereby or thereby or delivered in connection herewith or therewith; (vi) the reasonable expenses incurred by the Agent in connection with the transactions contemplated hereby, including the reasonable fees and out-of-pocket expenses of legal counsel, independent public accountants and other outside experts retained by the Agent in connection with the administration of this Credit Agreement, the Notes, the Conveyance and the Mortgage, and any and all other documents referred to herein or therein or contemplated hereby or thereby or delivered in connection herewith or therewith; and (vii) all reasonable costs and expenses, of the Agent and the Banks, if any, in connection with the enforcement of this Credit Agreement, the Notes, the Conveyance and the Mortgage and any and all other documents referred to herein or therein or contemplated hereby or thereby or delivered in connection herewith or therewith.

- (b) The Company agrees to indemnify and hold harmless the Agent and each Bank from and against any and all claims, damages, liabilities and expenses (including, without limitation, fees and disbursements of counsel) which may be incurred by or asserted against the Agent or such Bank in connection with or arising out of any investigation, litigation or not acquisition or proposed acquisition of all or any portion of the stock or substantially all of the assets of any Person directly or indirectly with the proceeds of any Advances hereunder, whether or not the Agent or such Bank is a party to such acquisition.
- (c) The Company covenants and agrees to pay to the Trustees or any holder of a Note, as the case may be, on the Application Date next succeeding any demand therefor, any monies paid and any expenses incurred by the Trustees or such holder of a Note pursuant to Section 2.4 of the Mortgage with interest at a rate equal to 2% over the Applicable Rate in effect (but in no event in excess of the maximum rate permitted by law) during the period from and including the date of the incurring of such expenses or the making of such payment to the date payment thereof is made in full and any monies so paid shall be part of the indebtedress secured by the Mortgage and the Trustees or such holder making such payment shall be subrogated to all the rights of the person, corporation or body politic receiving such payment.

Section 6.04. Execution in Counterparts. This Credit Agreement may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed to be an original and all of which taken together shall constitute but one and the same instrument.

Section 6.05 Dinding Effect; Assignment; Substitution of Banks. (a) This Credit Agreement shall become effective when it shall have been executed by the Company and the Agent and when the Agent shall have been notified by each Bank that such Bank has executed it and thereafter shall be binding upon, and inure to the benefit of, the Company, the Agent and each Bank and their respective successors and assigns, provided that the Company may not assign its rights or obligations hereunder or any interest herein without the prior written consent of the Banks; and provided further that no Bank may assign

its rights or obligations hereunder or any interest herein or under a Note to any Person other than a financial institution which shall, upon the effectivents of such assignment, make to the Company the representation contained in subsection (d) of Section 2.10.

(b). In the event that it shall become unlawful for any Bank to fund or continue to fund or maintain any loan hereunder carrying interest at the Eurodollar Rate, or any Bank shall demand payment of Increased Financing Costs or the Company shall be required to make a payment on account of Taxes with respect to such Bank or the Company shall be required to bear any Taxes with respect to such Bank, such Bank shall, at the Company's request, consult with the Company and the Agent in good faith to ascertain whether any arrangement (including, but not limited to, a change in such Bank's Eurodollar Lee ling Office) satisfactory to the Company, such Bank and the Agent and the Grantors can be made with a view to eliminating such illegality or Increased Financing Costs or Taxes and, if no satisfactory arrangement is available, such Bank may be replaced by another financial institution designated by the Company through an assignment and transfer by such Bank to such financial institution of its Notes and its rights and obligations bereunder, provided that such financial institution is acceptable to the Banks and is prepared to assume all of the rights and obligations of such original Bank on terms acceptable to such Bank. Any such francial institution which shall replace such Bank shall be deemed to be a Bank for all purposes hereof.

Section 6.06. Governing Law. This Credit Agreement and the Notes shall be governed by and construed in accordance with the laws of the State of New York.

Section 6.07. Severability of Provisions. Any provision of this Credit Agreement or any Note which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof or of the Notes or affecting the validity or enforceability of such provision in any other jurisdiction.

Section 6.08. Limitation on Interest. No provision of this Credit Agreement or the Notes shall require the payment or permit the collection of interest in excess of the maximum rate permitted by applicable law. If any excess of interest in such respect has been provided for herein or in the Notes, or shall be adjudicated to be so provided for herein or in the Notes, neither the Company nor its successors or assigns shall be obligated to pay such interest in excess of the amount permitted by applicable law.

Section 6.09, Headings, Article and Section headings used in this Credit Agreement are for convenience only and shall not affect the construction of this Credit Agreement.

In Witness Whereof, the parties hereto have caused this Credit Agreement to be executed and delivered by their respective officers thereunto duly authorized, as of the date first above written.

| New | Dunstan, Inc. | |
|------------|----------------------|--|
| By $Title$ | e: Vice President | |
| Cm | BANK, N.A., as Agent | |
| у | Vice President | |
| 3/3 | | |

| A Commitment | B Commitment To | al Commitment | |
|-------------------|------------------|---------------|---|
| \$ 171,580,000 \$ | \$ 58,420,000 \$ | 230,000,000 | CITIBANK, N.A. |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: |
| | | | Citibani, N.A. 399 F - Avenue New York, New York 10043 |
| | | | Eurodollar Lending Office: |
| | | | Citibank, N.A. 399 Park Avenue New York, New York 10043 |
| 149,200,000 | 50,800,000 | 200,000,000 | CHEMICAL BANK |
| | | | By |
| | | | Domestic Lending Office and Address for Notices: Chemical Benk 277 Park Avenue New York, New York 10017 |
| | | | Eurodollar Lending Office: Chemical Bank P. O. Box 4872 Norfolk Heuse Frederick Street Nassau, Bahamas |
| 149,200,000 | 50,800,000 | 200,000.000 | MANUFACTURERS HANOVER TRUST COMPANY |
| | | | By |
| | | | Domestic Lending Office and Address for Notices: Manufacturers Hanover Trust Company 350 Park Avenue New York, New York 10022 |
| | | | Eurodollar Lending Office: Manufacturers Hanover Trust Company 350 Park Avenue New York, New York 10022 |

| Commitment | B Commitment | To | al Commitment | |
|-------------------|---------------|----|---------------|--|
| \$ 149,200,000 | \$ 50,800,000 | \$ | 200,000,000 | MORGAN GUARANTY TRUST COMPANY OF NEW YORK |
| | | | | By |
| | | | | Domestic Lending Office and Address for Notices: Morgan Guaranty Trust Company of New York 23 Wall Street New York, New York 10015 |
| | | | | Eurodollar Lending Office: Morgan Guaranty Crust Company of New York St. Helier Office Queensway House Queen Street Jersey, Channel Islands |
| 130,550,000 | 44,450,000 | | 175,000,000 | BANK OF AMERICA NATIONAL TRUST AND SAVINGS ASSOCIATION |
| | | | | By |
| | | | | Domestic Lending Office and Address for Notices: |
| | | | | Bank of America National Trust and Savings Association MAD Corporate Service Center South 1000 West Temple Los Angeles, California 90012 |
| | | | | Eurodollar Lending Office: |
| | | | | Bank of America National Trust and Savings Association Grand Cayman Branch West Wind Building Fort and South Church Street |

P. O. Box 1078

B.W.I.

Georgetown, Grand Cayman

A Commitment B Commitment To'al Commitment 59,680,000 \$ 20,320,000 \$ 80,000,000 Mellon Bank, N.A. By Title: Domestic Lending Office and Address for Notices: Mellon Bank, N.A. Mellon Square Pittsburgh, Pa. 15230 Eurodollar Lending Office: Mellon Bark, N.A. P.O. Pox 710 Grand Cayman Cayman Islands, B.W.I 130,000,000 THE BANE OF NOVA SCOTIA 96,980,000 33,020,000 By

Eurodollar Lending Office and Address for Notices:

BNS International (UNITED KINGDOM) Limited c/o The Bank of Nova Scotia International, Ltd. P.O. Box N7545 Nassau, Bahanas

Domestic Lending Offices:

The Bank of Nova Scotia Suite 909 First National Bank Beilding 2 Peachtree Street S.W. Atlanta, Georgia 30383

| A | Commitment | B Crmm'tment | Total Commitment | |
|----|------------|---------------|------------------|--|
| \$ | 96,980,000 | \$ 33,020,000 | \$ 130,000,000 | CANADIAN IMPERIAL BANK OF COMMERCE |
| | | | | By |
| | | | | Domestic Lending Offic and Address for Notices: |
| | | | | Canadian Imperial Bank of Commerce Atlanta Agency Suite 1400 First Atlanta Tower Atlanta, Georgia 30383 |
| | | | | Eurodollar Lending Office: |
| | | | | Canadian Imperial Bank of Commerce Grand Cayman Branch c/o Suite 1400 First Atlanta Tower Atlanta, Georgia 30383 |
| | 96,980,000 | 33,020,000 | 130,000,000 | Toronto Dominion Bank |
| | | | | By Title: |
| | | | | Domestic Lending Office and Address for Notices: |
| | | | | Toronto Dominion Bank Peachtree Center South Tower, Suite 1600 225 Peachtree Street, N.E. Atlanta, Georgia 30303 |
| | | | | Eurodollar Lending Office: |

Toronto Dominion Bank Peachtree Center South Tower, Suite 1600 225 Peachtree Street, N.E.

Atlanta, Georgia 30303

| A | Commitmes. | <u>F</u> | t | То | al Commitment | |
|----|------------|----------|-----------|----|---------------|--|
| #: | 85,790,000 | \$ 25 | 9,210,000 | 4. | 115,000 000 | THE FIRST NATIONAL BANK OF CHICAGO |
| | | | | | | By Tille: |
| | | | | | | Domestic Lending Office at 1 Address for Notices: |
| | | | | | | The First National Bank of Chicago One First National Plaza Chicago, Illinois 66670 |
| | | | | | | Eurodollar Lending Office: |
| | | | | | | The First National Bank of Chicago Panama Branch Via Espana con Venezuela P.O. Box 8051 Panama 7, Republic of Panama |
| | 82,060,000 | 27 | ,940,000 | | 110,000,000 | Security Pacific National Bank |
| | | | | | | By |
| | | | | | | Domestic Lending Office and Address for Notices: |
| | | | | | | Security Pacific National Bank 333 South Hope Street Los Angeles, California 90051 |

Eurodollar Lending Office:

Security Pacific National Bank 333 South Hope Street Los Angeles, California 90051

| \$ 67,140,000 \$ | 22,860,000 \$ | 90,000,000 | CREDIT SUISSE |
|---------------------|---------------|------------|--|
| | | | By Title: |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: |
| | | | Credit Suisse 100 Wall Sweet New York, New York 10005 |
| | | | Eurocallar Lending Office: Credit Suisse 100 Wall Street New York, New York 10005 |
| 67,140,000 | 22,860,000 | 90,000,000 | DEUTSCHE BANK, A.G. |
| | | | By Title: |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: |
| | | | Deutsche Bank, A.G. New York Branch |

Eurodollar Lending Office:

9 West 57th Street

Deutsche Bank, A.G. Cayman Islands Branch c/o New York Branch 9 West 57th Street New York, New York 10019

New York, New York 10019

| Commitment | B Commitment | Total | Commitment | |
|------------|---------------|-------|-------------|--|
| 67,140,000 | \$ 22,860,000 | \$ | ≎00,000,00° | NATIONAL WESTMINSTER BANK LAMITED |
| | | | | By |
| | | | | Domestic Lending Office and Address for Notices: |
| | | | | National Westminster Bank Limited 100 Wall Street New York, New York 10005 |
| | | | | Eurodollar Lending Office: |
| | | | | International Westminster Bank Limited 41 Threadneedle Street London EC2 38AP, England |
| | | | | |
| 67,140,000 | 22,860,000 | | 90,000,000 | TEXAS COMMERCE BANK NATIONAL ASSOCIATION |
| | | | | By |
| | | | | Domestic Lenning Office and Address for Notices: |
| | | | | Texas Commerce Bank National Association |
| | | | | P.O. Box 2558 Houston, Texas 77001 |
| | | | | Eurodollar Lending Office: |
| | | | | Texas Commerce Bank National Association |
| | | | | Nassau Branch Nassau, Bahamas |
| | | | | e/o P.O. Box 2558 |
| | | | | Houston, Texas 77001 |

| A Commitment | B Commitment | Total Commitment | |
|--------------|-----------------|------------------|---|
| \$ 7,140,00 | 0 \$ 22,860,000 | \$ 90,000,000 | Wells Farge Bank, N.A. |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: |
| | | | Wells Fargo Bank, N.A. 770 Wilshire Blvd., 7th Floor Los Angeles, California 90017 |
| | | | Eurodollar Lending Office: Wells Fargo Bank, N.A. 770 Wilshire Blyd., 7th Floor Los Angeles, California 90017 |
| 52,220,00 | 0 17,780,000 | 70,000,000 | FIRST CITY NATIONAL BANK OF HOUSTON |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: |
| | | | First City National Bank of Houston 1001 Main Street Houston, Texas 77001 |
| | | | Eurodollar Lending Office: First City National Bank of Houston 99 Bishopsgate, 20th Floor London EC2M 3XD, England |
| 33,570,00 | 0 11,430,000 | 45,000,000 | Am-terdam-Rotterdam Bank, N.V., New York Branch |
| | | | By $Title$: |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: |
| | | | Amsterdam-Rotterdam Bank, N.V. Amro London Branch 29/30 King Street London EC2V SEQ, England |
| | | | Eurodollar Lending Office: Amsterdam Rotterdam Bank, N.V. Amro London Branch 29-30 King Street London EC2V SEQ, England |

| A Commitment | B Commitment | Total Commi | |
|---------------|---------------|---------------|---|
| \$ 29,840,000 | \$ 10,160.000 | \$ 40,000,000 | European American Banking Corporation |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: European American Banking Corperation 10 Hanover Square New York, New York 10015 |
| | | | Eurodollar Lending Office: European American Banking Corporation Nassau (Bahamas) Branch 10 Hanover Square New York, New York 10015 |
| 29,840,000 | 10,160,000 | 40,000,000 | THE FIRST NATIONAL BANK OF BOSTON |
| | | | By Title: |
| | | | Domestic Lending Office and Address for Notices: The First National Bank of Boston 100 Federal Street Boston, Massachusetts 02110 |
| | | | Eurodollar I. ung Office: The First National Bank of Boston 100 Federal Street 10 fon, Massachusetts 02110 |
| 29,840,000 | 10.160,000 | 40,000,000 | NATIONAL BANK OF DETROIT |
| | | | By $Title$: |
| | | | Domestic Lending Office and Address for Notices: National Bank of Detroit 611 Woodward Avenue Detroit, Michigan 48226 |
| | | | Eurodollar Lending Office: National Bank of Detroit 611 Woodward Avenue Detroit, Michigan 48226 |

| | A Commitment E | Commitment To | otal Comm'tment | |
|-----|--|------------------------------------|-----------------|--|
| \$ | 22,380,000 \$ | 7,620,000 \$ | 30,000,000 | AMERITRUST COMPANY |
| | | | | By Title: |
| | | | | Domestic Lending Office and Address for Notices: |
| | | | | Ameritrust Company 900 Euclid Avenue Cleveland, Ohio 44101 |
| | | | | Eurodollar Lending Office: |
| | | | | Ameritrust Company Nassau, Bahamas Branch e o 900 Euclid Avenue Cleveland, Ohio 44101 |
| | 22,380,000 | 7,620,000 | 30,000,000 | Compagnie Luxembourgeoise de la Dresdner Bank, A.G. — Dresdner Bank Internationale — |
| | | | | By Title: |
| | ic Lending Offi | ce and | | Eurodollar Lending Office: |
| Com | s for Notices; pagnie Luxem Presdner Bank, resdner Bank e Postale 355 L Que du Marché emburg | A.G. Internationals ux Ville | | Compagnie Luxembourgeoise de la Dresdner Bank, A.G. — Dresdner Bank Internationale — Case Postale 355 Lux Ville 26, Rue du Marché-aux-Herbes Luxemburg |
| | 22,380,000 | 7,620,000 | 30,000,000 | THE NORTHERN TRUST COMPANY |
| | | | | By Title: |
| | | | | Domestic Lending Office and Address for Notices: The Northern Trust Company 50 South LaSalle Street Chicago, Illinois 60675 |
| | | | | Eurodollar Lending Office: The Northern Trust Company 50 South LaSalle Street Chicago, Illinois 60675 |

| A | Commitment | В | Commitment | 70 | al Commitment | |
|------|-------------|------|------------|-----|---------------|---|
| \$ | 18,650,000 | \$ | 6,350,000 | # | 25,000,000 | Banque Nationale de Paris |
| | | | | | | By Title: |
| | | | | | | Domestic Lending Office and Address for Notices: |
| | | | | | | Banque Nationale de Paris New York Branch 499 Park Ayenue New York, New York 10022 |
| | | | | | | Eurodollar Lending Office: |
| | | | | | | Banque Nationale de Paris New York Branch 499 Park Avenue New York, New York 10022 |
| \$1, | 865,000,000 | \$6: | 35,000,000 | +2, | 500,000,000 | |

DOMESTIC NOTE

U.S. \$

New York, New York July 16, 1981

For Value Received, New Dunstan, Inc., a Delaware non-profit corporation (the "Company"), hereby promises to pay to (the "Bank"), or order, the principal sum of [amount of the Bank's Total Commitment in words] United States Dollars (U.S.\$ [amount of the Bank's Total Commitment in figures]) or, if less, the aggregate unpaid principal amount of all Domestic Advances (as defined below) of June 1, 1990, unless the Conveyance Supplement (as defined in the Credit Agreement referred to below) has not been executed and delivered on or prior to July 31, 1983, in which case, the Company promises to pay such principal sum on February 1, 1990.

The Company hereby also promices to pay interest at the Domestic Rate in effect from time to time as provided in the Credit Agreement hereinafter referred to (a) on September 1, 1981, on the aggregate principal amount of all Domestic Advances outstanding from time to time from the date hereof to and including August 31, 1981, (b) on the first day of each calendar month thereafter (commencing October 1. 1981), on the aggregate principal amount of all Domestic Advances outstanding from time to time during the calendar month next preceding such date, (c) on the date the aggregate unpaid principal amount of all Domestic Advances shall be paid in full (to the extent not paid pursuant to clauses (a) and (b) above), on the aggregate principal amount of all Domestic Advances outstanding from time to time during the period from the first day of the calendar month next preceding such final payment date for which interest was paid pursuant to clause (b) above to such final payment date and (d) (to the extent permitted by applicable law) on demand, on any amount of interest not paid on the due date therefor referred to in clause (a), (b) or (c) above.

As used herein, the term "Domestic Advances" means those advances (or portions of advances) made by the Bank to the Company pursuant to the Credit Agreement which bear interest at the Domestic Rate as provided in the Credit Agreement. As used herein, the term

"Domestic Rate" has the meaning specified in the Conveyance (as defined in the Mortgage referred to below).

Both principal and interest are payable in lawful money of the United States of America to the Agent (as defined below), in immediately available funds, at the office of Citibank, N.A., at 399 Park Avenue, New York, New York 10043, for the account of the Bank, free and clear of and without deduction for any and all present and future Taxes (other than Excluded Taxes) as set forth in the Credit Agreement.

The grid attached hereto is a part of this Note and, prior to any transfer hereof, the Bank shall endorse the relevant information on such grid as provided in the Credit Agreement.

This Note evidences certain advances (or portions of advances) made to the Company by the Bank pursuant to a Credit Agreement, dated as of July 16, 1981 (the "Credit Agreement"), among the Company, the Bank, certain other banks parties thereto and Citibank, N.A., as agent (the "Agent"), and is secured by a Mortgage, Deed of Trust and Assignment of Production, dated July 16, 1981 (the "Mortgage"), from the Company to F. D. Thompson and W. T. Fox III, as Trustees, and the Agent, to which Mortgage reference is hereby made for a description of the property thereby mortgaged, pledged, assigned, transferred and conveyed, the nature and extent of the security and the rights of said Trustees, the Agent, the holder of this Note and the Company in respect of such security.

This Note is subject to prepayment from time to time prior to its maturity by the application of monies actually received by or on behalf of the Bank by virtue of the assignment in Article III of the Mortgage, in the manner, at the time and to the extent set forth in the Mortgage.

Upon the occurrence of an Event of Default specified in the Mortgage, the unpaid principal hereof and the interest accrued hereon may be declared to be forthwith due and payable in the manner and with the effect provided in the Mortgage.

Should the indebtedness represented by this Note or any part thereof be collected at law or in equity or in bankruptey, receivership or other court proceedings or should this Note be placed in the hands of attorneys for collection after default, the Company agrees to pay, in addition to the principal and interest due and payable hereon and other amounts payable hereunder, reasonable attorneys' and collection fees.

The Company, and any endorser hereof, hereby waives presentation for payment, demand, notice of nonpayment, protest and notice of protest. No delay on the part of the holder hereof in exercising any rights hereunder shall operate as a waiver of such rights.

No holder of this Note shall have recourse for the payment of the principal of or interest on this Note or for any amounts payable hereunder or for any claims based hereon against any incorporator or against any v_1 alors, beneficiary, grantee, distributee, director or officer, past, present or future, of the Company or any successor corporation, as such, either directly or through the Company or any such successor corporation, whether by virtue of any constitution, statute or rule of law or by the enforcement of any assessment or penalty or otherwise, any and all such liability, whether at common law, in equity, by any constitution, statute or otherwise, of incorporators, members, beneficiaries, grantees, distributees, directors or officers, as such, being released as a condition of and in consideration for the issue of this Note.

This Note shall be governed by and construed in accordance with the laws of the State of New York.

NEW DUNSTAN, INC.

| By Vice President | 608 F 100 B |
|--|-------------|
| Ne Varietur | |
| To identify this Note with the above mentioned Mortgage, De Trust and Assignment of Production, passed before me this 16t of July, 1981. | |
| Notary Public | |

Domestic Note Grid

ADVANCES, ADJUSTMENTS AND PAYMENTS OF PRINCIPAL

| Date | Amount of Advance Made at the Domestic Rate | Amount of Change to the Dometic Rate | Amount of Principal Paid or Prepaid | Amount of Change to Eurodollar Rate | Unpaid Principal Balance | Notation Made By |
|---|---|---|---|--|--------------------------------|---------------------|
| | | | | | | |
| | - | | | | | |
| | | | | | | |
| | - | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | - | | | | | |
| | | | | | | - |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| | - | | | | | |
| | - | | | | | - |
| *************************************** | + | | | | | |

Credit Agreement

EURODOLLAR NOTE

U.S. \$

New York, New York July 16, 1981

FOR VALUE RECEIVED, New DUNSTAN, Inc., a Delaw. e non-profit corporation (the "Company"), hereby promises to pay to (the "Bank"), or order, the principal sum of [amount of the Bank's Total Commitment in words] United States Dollars (U.S. \$ [amount of the Bank's Total Commitment in figures]) or, if less, the aggregate unpaid principal amount of all Eurodollar Advances (as defined below) on June 1, 1990, unless the Conveyance Supplement (as defined in the Credit Agreement referred to below) has not been executed and delivered on or prior to July 1, 1983, in which case the Company premises to pay such principal sum on February 1, 1990.

The Company hereby also promises to pay interest at the Eurodollar Rates in effect from time to time as provided in the Credit Agreement hereinafter refer 1 to (a) on September 1, 1981, on the aggregate principal amount of all Eurodollar Advances outstanding from time to time from the date hereof to - .d including August 31 1981, (b) on the first day of each calendar month thereafter (commencing October 1. 1081), on the aggregate principal amount of all Eurodollar Advances outstanding from time to time during the calendar month next preceding such date, (c) on the date the aggregate unpaid principal amount of all Eurodollar Advances shall be paid in full (to the extent not paid pursuant to clause (a) and (b) above), on the aggregate principal amount of all Eurodollar Advances outstanding from time to time during the period from the first day of the calendar month next preceding such final payment date for which interest was paid pursuant to clause (b) above to such final payment date and (d) (to the extent permitted by applicable law) on demand, on any amount of interest not paid on the due date therefor referred to in clause (a), (b) or (c), above.

As used herein, the term "Eurodollar Advances" means those advances (or portions of advances) made by the Bank to the Company

pursuant to the Credit Agreement which bear interest at the Eurodollar Rate as provided in the Credit Agreement. As used herein, the term "Eurodollar Rate" has the meaning specified in the Conveyance (as defined in the Mortgage referred to below).

If due to (a) the introduction of or any change (including, without limitation, any change by way of imposition or increase of reserve requirements) in, or in the interpretation of, any law or regulation since November 1, 1980, (b) the compliance by the Bank with any request from any central bank or other governmental authority (whether or not having the force of law) or (c) the inability of the Company to comply with a Notice of Borrowing (as defined in the Credit Agreement), there shall be any increase in the cost to the Bank of funding or continuing to fund or maintaining this Note through funds obtained in the London interbank market then the Company shall from time to time upon the Application Date next succeeding any demand therefor by the Bank, pay to the Bank additional amounts sufficient to indemnify the Bank against such increr sed costs together with interest at the rate provided in the Conveyance. A certificate as to the amount of such increased costs, submitted to the Company by or on behalf of the Bank, shall be conclusive as to all such costs, absent manifest error.

Both principal and interest are payable in lawful money of the United States of America to the Agent (as defined below), in immediately available funds at the office of Citibank, N.A., at 399 Park Avenue, New York, New Ye k 10043, for the account of the Bank's Eurodollar Lending Office (as defined in the Credit Agreement), free and clear of and without deduction for any and all present and future Taxes (other than Excluded Taxes) as set forth in the Credit Agreement.

The grid attached hereto is a part of this Note and, prior to any transfer hereof, the Bank shall endorse the relevant information on such grid as provided in the Credit Agreement.

This Note evidences certain advances (or portions of advances) made to the Company by the Bank pursuant to a Credit Agreement, dated as of July 16, 1981 (the "Credit Agreement"), among the Company, the Bank, certain other banks parties thereto and Citibank, N.A., as agent (the "Agent"), and is secured by a Mortgage, Deed of Trust and Assignment of Production, dated July 16, 1981 (the "Mortgage"), from the Company to F. D. Thompson and W. T. Fox III, as Trustees, and the Agent, to which Mortgage reference is hereby made

for a description of the property thereby mortgaged, pledged, assigned, transferred and conveyed, the rather and extent of the security and the rights of said Trusteer, the Agent, the holder of this Note and the Company in respect of such security.

This Note is subject to prepayment from time to time prior to its maturity by the application of monies actually received by or a benalf of the Bank by virtue of the assignment in Article III is the Mortgage, in the manner at the time and to the extent set forthe Mortgage, if, the manner at the time and to the extent set forthe Mortgage. If, due to any prepayment of the principal event of virtue is assignment in Article III of the Mortgage or due to accept the maturity of this Note or due to any other reason, the Bank receives payments of principal of this Note on other than the forthey of a Rate Period (as defined in the Conveyance), the Company shall pay to the Agent for the account of the Bank on the Approach of the ause (ii) of subparagraph (c) of Paragraph 4 of the Conveyance to compensate the Bank for any osses, costs or expenses which the Bank may incur as a result of such payment, with interest from the date of such prepayment at the rate provided in the Conveyance.

Upon the occurrence of an Event of Def: specified in the Mortgage, the unpaid principal hereof and the interest secrued hereon may be declared to be forthwith due and payable in the manner and with the effect provided in the Mortgage.

Should the indebtedness represented by this Note or any part thereof be collected at law or in equity or in bankruptcy, receivership or other court proceedings or should this Note be placed in the hands of attorneys for collection after default, the Company agreed to pay, in addition to the principal and interest due and payable kereon as a other amounts payable her and ler, reasonable attorneys and collection fees.

The Company and any endorser hereof, hereby waives presentation for paymen' d mand, notice of nonpayment, protest and notice of protest. No delay at the part of the holder hereof in exercising any rights hereunder shall operate as a waiver of such rights.

No holder of this Note shall have recourse for the payment of the principal of or interest on this Note or for any amounts payable hereunder or on any craims based hereon against any incorporator or

against any member, beneficiary, grantee, distributee, director or officer, past, present or future, if the Company or any present or future, if the Company or any present or corporation, as such, either directly or through the Company of any such successor corporation, whether by victue of any constitution, statute or rule of law or by the enforcement of any assessment or remailty of otherwise, any and all such liability, whether at common law, in equity, by any constitution, statute or otherwise, of incorporators, per deep, beneficiaries, grantees, distributees, a rectors or officers, as such either released as a condition of and in consideration for the issue of this Note.

This Note shall be governed by and construed in accordance with the laws of the State o New York.

NEW DUNSTAN, INC.

| | By Vice President |
|---------------|---|
| Ne Varietur: | |
| | he above-mentioned Mortgage, Deed of Trust tion, passed before me this 16th day of July, |
| Notary Public | |

Eurodollar Note Grid

ADVANCES, ADJUSTMENTS AND PAYMENTS OF IRINCIPAL

| Date | Amount of Advance Made at the Eurodollar Rate | Amount of Change to the Eurodollar Rate | Amount of Principal Paid or Prepaid | Amount of Change to Domestic Rate | Unpaid Principal Balance | Notation |
|------|---|--|---|--|--------------------------------|----------|
| | | | | | | |
| | an | | | | | |
| | | | | | | |
| | | | | | | - |
| | _1 | | | | | |
| | | | | | | |
| | | | <u> </u> | | | |
| | | | | | | - |
| | | | | | | |
| | | | | | | |
| -w-m | | F | | | | |
| | | | | | | - |
| | | | | | | - |
| | | | | | | - |
| | | | | | | 1 |
| | | | | | | |
| | | - | | | | <u> </u> |

[EXHIBITS C THROUGH BB TO THE CREDIT ACREEMENT ARE NOT INCLUDED IN THE FORM OF THE CREDIT AGREEMENT ATTACHED TO THE MORTGAGE, DEED OF TRUST AND ASS/GNMENT OF PRODUCTION]

EXHIBIT B

Mortgage, Deed of Trust and Assignment of Production

PENNZOIL COMPANY, ET AL.

TO

YEW DUNSTAN, INC.

Instrument of Conveyance of Production Payment

Dated July 16, 1981

leturn recorded counterparts to:

Pr. s. 204. Company Pe.mzo Place P.O. Box 2967 P. ..., Texas 77001 Attendon: Ms. Sally Hazen

This instrument was prepared by:

Frank W. R. Hubert, Jr. 3000 One Shell Plaza Houston, Texas 77002

TABLE OF CONTENTS

| Paragr | aph | Page |
|--------|---|------|
| 1. | Certain Definitions | 1 |
| 2. | Con ance of the Production Payment | 11 |
| 3. | Certain Provisions Governing the Production Payment | 14 |
| 4. | | 17 |
| 5. | | 20 |
| €. | 그 그 그리는 이 그를 내 그리는 그리는 그리고 있었다. 그리고 | 21 |
| 7. | | 23 |
| 8. | Marketing of Production Payment Minerals | 24 |
| 9. | Performance of Sales Contracts; Payments; Etc | 26 |
| | Taking In Kind | 27 |
| 11. | Withholding and Restitution of Proceeds of Production Payment | |
| | Minerals | 27 |
| 12. | Protection to Purchasers of Production | 29 |
| | Certain Payments by Grantors | 29 |
| 14. | Covenants of the Grantors | 30 |
| 15. | Access to the Subject Interests | 37 |
| 16. | Pooling and Unitization | 38 |
| 17. | Default by the Grantors and the Remedies of the Grantee | 38 |
| 18. | Exercise and Termination of the Remedies of the Grantee | 40 |
| 19. | Reports and Payments to the Grantee | 41 |
| 20. | Representations, Warranties and Covenants | 42 |
| 21. | Transfer of the Subject Interests | 47 |
| 22. | | 47 |
| 23. | | 48 |
| 24. | Rights of Mortgagee or Trustee | 48 |
| 25. | Grantors' Rights to Make Payments to be Applied to Discharge of | |
| | Production Payment | 48 |
| 26. | Notices | 49 |
| 27. | Further Assurances | 49 |
| 28. | Binding Effect | 49 |
| 29. | Release or Failure of Title | 50 |
| 30. | Governing Law | 50 |
| 31. | Final Termination | 50 |
| 32. | Headings | 50 |

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (the "Conveyance") dated July 16, 1981, between Pennzoil Company, a Delaware corporation, Pennzoil Producing Company, a Delaware corporation and Duval Corporation, a Delaware corporation (the "Grantors"), and New Dunstan, Inc., a Delaware non-profit corporation (the "Grantee"),

WITNESSETH:

Whereas, the Grantors desire to sell and the Grantee desires to purchase the Production Payment hereinafter described,

Now, Therefore, it is agreed by and between the Grantors and the Grantee as follows:

1. Certain Definitions. The following terms, when used in this Conveyance, have the meanings indicated below:

"Accounting Month" means any monthly period commencing at 7:00 A.M., local time in effect at the location of each Subject Interest, on the first day of any calendar month and ending at 7:00 A.M., local time in effect at the location of each Subject Interest, on the first day of the next succeeding calendar month.

"Affiliate" means, as to the party specified, any Person controlling, controlled by or under common control with such party, with the concept of control in such context meaning the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of another, whether through the ownership of voting securities, by contract or otherwise.

"Alternate Base Rate" means a fluctuating rate per annum equal at all times to the higher of:

- (a) the Base Rate; or
- (b) ½ of 1% above the latest three-week moving average of secondary market morning offering rates in the United States for three-month certificates of deposit of major United States money market banks, such three-week moving average

being determined weekly by Citibank on the basis of such rates reported by certificate of deposit dealers to and published by the Federal Reserve Bank of New York or, if such publication shall be suspended or terminated, on the basis of quotations for such rates received by Citioank from three New York certificate of deposit dealers of recognized standing, in either case adjusted to the nearest 1/4 of 1% or, if none, to the next higher 1/4 of 1%.

each change in such fluctuating rate to take effect simultaneously with the corresponding change in such Base Rate or moving average.

"Applicable Rate" means, for each day, the weighted average of the Eurodollar Rates and the Domestic Rate applicable to the unliquidated balance of the Primary Sum on such day determined on the basis of the proportion which that portion of such unliquidated balance to which each Eurodollar Rate and the Domestic Rate is applicable bears to the total amount of such unliquidated balance; provided, however, that in no event shall the Applicable Rate exceed the maximum rate which would be permitted by the laws of the State of New York if the Applicable Rate were being applied to a borrowing of money in an amount equal to the unliquidated balance of the Primary Sum. Except as provided hereinafter, the Grantors may from time to time select the Eurodollar Rate to be in effect with respect to all or any portion of the then unliquidated balance of the Primary Sum for a Rate Period, on the condition that notice of such selection (including notice of said portion of such unliquidated balance and of the duration of the Rate Period therefor) is delivered to the Grantee and to Citibank not later than 11:00 A.M. (New York City time) at least four Business Days prior to the commencement of such Rate Period, and provided that in no event shall the Grantors be permitted to have selected the Eurodollar Rate to be in effect concurrently with respect to more than three portions of such unliquidated balance; and except to the extent that the Eurodollar Rate shall have been so selected by the Grantors in accordance herewith to be applicable to all or one or more portions of such unliquidated balance, the Domestic Rate shall be applicable to the

entire unliquidated balance of the Primary Sum. Notwithstanding the foregoing, and except as set forth in the next succeeding sentence hereto, if either

- (i) it shall become unlawful, or any central bank or other governmental authority shall assert that it is unlawful, for any lender to the Grantee to obtain funds in the London interbank market in order to fund or continue to fund or maintain its loan to the Grantee, upon such lender giving notice thereof to the Grantee (which shall give prompt notice thereof to the Grantors), or
- (ii) the Grantee shall demand payment of Increased Financing Costs referred to in Section 4(d) hereof, and the Grantors shall have given the Grantee four Business Days' notice of their selection of the Domestic Rate in substitution for the Eurodollar Rate,

the Applicable Rate shall thereupon be the Domestic Rate as to those portions of the unliquidated balance of the Primary Sum previously subject to the Eurodollar Rate as to which clause (i) or (ii) is applicable, and the right of the Grantors to select the Eurodollar Rate bereunder shall thereupon terminate unless and until such funding is no longer unlawful or so asserted to be unlawful or the circumstances giving rise to such Increased Financing Costs have ceased to exist. Notwithstanding the foregoing sentence, in the event of the happening of any circumstance described in clause (i) of said sentence with respect to any such lender to the Grantee, or of demand for payment of any such Increased Financing Costs described in clause (ii) of said sentence attributable to any such lender, the Grantee covenants and agrees that, at the Grantors' request, it will consult with such lender in good faith to ascertain whether any arrangement (including, but not limited to, a change in such lender's Lending Office) satisfactory to such lender, the Grantors and the Grantee can be made with a view to eliminating such Increased Financing Costs and, upon demand of the Grantors, the Grantee will cause such lender to be replaced by another financial institution designated by

the Grantee and the lenders to the Grantee and is prepared to assume all of the rights and obligations of such original lender on terms acceptable to such lender; and if upon said replacement there shall not be continuing any circumstances described in either clause (i) or (ii) of the foregoing sentence, the Grantors shall thereupon once again be entitled to select the Eurodollar Rate hereunder. Any such financial institution which shall replace any such lender in connection with any loan in accordance with the preceding sentence shall be deemed to be a lender to the Grantee for all purposes hereof.

"Application Date" means a Regular Application Date or, if such Regular Application Date is not a business Day, the Next Preceding Business Day. For purposes of this Conveyance, an Application Date shall be deemed to occur "at or near the end" of the Accounting Month which ends on such Application Date or, if such Application Date is a Next Preceding Business Day, the next following Regular Application Date.

"Base Rate" means a fluctuating interest rate per annum equal at all times to the rate of interest announced publicly by Citibank in New York, New York from time to time as Citibank's base rate.

"Business Day" means a day on which banks are not required or authorized to close in New York City and, if the Eurodollar Rate is relevant, on which dealings are carried on in the London interbank market and banks are open for business in London.

"Citibank" means Citibank, N.A., a national banking association.

"Discharge Schedule" means the tabulation set forth in Exhibit B attached to this Conveyance and hereby made a part hereof.

"Domestic Rate" means a fluctuating rate per annum (computed on the basis of a year of 365 or 366 days, as the case may be, for the actual number of days — including the first day but excluding the last day — occurring in the period for which such rate is computed) equal at all times to:

- (i) from the date hereof to and including July 31, 1983, ¼ of 1% per annum above the Alternate Base Rate in effect from time to time;
- (ii) from and after August 1, 1983 to and including July 31, 1987, ½ of 1% per annum above the Alternate Base Rate in effect from time to time; and
- (iii) from and after August 1, 1987, 34 of 1% per annum above the Alternate Base Rate in effect from time to time.

"Effective Date" means 7:00 A.M., local time in effect at the location of each Subject Interest, on July 1, 1981.

"Eurodollar Rate" means, with respect to any portion of the unliquidated balance of the Primary Sum as to which such rate is applicable, a fluctuating rate per annum (computed on the basis of a year of 360 days for the actual number of days—including the first day but excluding the last day—occurring in the period for which such rate is computed), subject to funds being available to the lenders to the Grantee on reasonable terms under routine interbank market conditions (all as determined in the discretion of the lenders to the Grantee), equal at all times to:

- (i) from the date hereof to and including July 31, 1983, 5% of 1% per annum above the LIBO Rate in effect from time to time for such portion of the Primary Sum;
- (ii) from and after August 1, 1983 to and including July 31, 1987, 34 of 1% per annum above the LIBO Rate in effect from time to time for such portion of the Primary Sum; and
- (iii) from and after August 1, 1987, 7% of 1% per annum above the LIBO Rate in effect from time to time for such portion of the Primary Sum.

"Excess Rate" has the meaning assigned to that term in Paragraph 4(d) hereof.

"Exhibit A" means Exhibit A attached to this Conveyance and hereby made a part hereof for all purposes of interpreting, construing and enforcing this Conveyance, provided, that if at any time said Exhibit A has been supplemented by the Grantors by

the execution and delivery of the Supplemental Instrument of Conveyance of Production Payment, then at such time Exhibit A shall mean said Exhibit A as so supplemented.

"First Production Period" means the period commencing with the Effective Date and ending at 7:00 A.M., local time in effect at the location of each Subject Interest, on August 1, 1983.

"Hydrocarbons" means all oil, gas and other gaseous and liquid hydrocarbons or any combination of one or more of such substances appropriate to the context in which such term is used.

"Increased Financing Costs" means any increase in the cost to the Grantee of obtaining or maintaining loans from any lender during any period in which the Eurodollar Rate is applicable to all or any portion of the unliquidated balance of the Primary Sum, and which is due to any increase in the cost to such lender to fund or continue to fund or maintain any loan to the Grantee through funds obtained in the London interbank market, arising from (i) the introduction of or any change (including, without limitation, any change by way of imposition or increase of reserve requirements) in, or in the interpretation of, any law or regulation since November 1, 1980 or (ii) the compliance by such lender with any request from any central bank or other governmental authority whether or not having the force of law, and if the Grantors shall exercise their option to select the Domestic Rate as the Applicable Rate prior to the last day of a Rate Period applicable to all or any po: 'm of the Primary Sum, shall also include any costs to such lenger incurred as a result of the selection of such Rate. A certificate from such lender to the Grantee as to the amount of such increased costs shall be conclusive as to the amount of such increased costs to the Grantee, absent manifest error.

"Lease" means any oil and/or gas lease or sub-lease, or any oil, gas and/or mineral lease or sub-lease, or any mining claim or award, or any fee or leasehold interest in, or any contractual right to produce or receive the proceeds of production of, any Mineral.

"LIBO Rate" means a rate per annum, if any, equal at all times during each Rate Period for any pertion of the unliquidated balance of the Primary Sum to which the Eurodollar Rate is applicable to the average (rounded upward, if necessary, to the nearest whole multiple of 1/16 of 1% per annum) of the rate of interest per annum at which deposits in United States dollars in immediately available funds are offered by each of the Reference Banks to pro- banks in the London interbank market at 11:00 A.M. (London time) two Business Days before the first day of such Rate Period for a period equal to such Rate Period and in an amount substantially equal to such portion of such unliquidated balance of the Primary Sum (such rate for each Rate Period being determined by Citibank on the basis of applicable quotations for the LIBO Rate furnished to and received by Citibank from the Reference Banks two Business Days prior to the first day of such Rate Period and, if any one or more of the Reference Banks shall not so furnish such a quotation for any Rate Period, the LIBO Rate for such Rate Period shall be based upon the quotations furnished by the other Reference Banks).

"Minerais" means Hydrocarbons and, in the event the Supplemental Instrument of Conveyance of Production Payment is executed and delivered (and from and after the effective date thereof), also includes sulphur.

"Next Preceding Business Day" means, when used with reference to a Regular Application Date or Quarterly Regular Application Date, the first Business Day next preceding such Regular Application Date or Quarterly Regular Application Date if such Regular Application Date or Quarterly Regular Application Date is not a Business Day.

"Person" means an individual, corporation, partnership, joint venture, trust or unincorporated organization, or a government or any agency or political subdivision thereof.

"Primary Sum" means at any time the aggregate amount of the Purchase Price which shall at such time have been paid.

"Production Payment" has the meaning assigned to that term in Paragraph 2 hereof.

"Production Payment Minerals" has the meaning assigned to that term in Paragraph 2 hereof.

"Production Payment Percentage" has the treating assigned to that term in Paragraph 2 hereof.

"Quarterly Application Date" means a Quarterly Regular Application Date or, if such Quarterly Regular Application Date is not a Business Day, the next Preceding Business Day.

"Quarterly Regular Application Date" mer as the Regular Application Date falling on the first day of any January, April, July or October during the term of the Production Payment.

"Rate" means the Dog stic inte or the Eurodollar Rate.

"Rate Feriod" means any of the periods selected from time to time by the Grantors to apple in respect of the application of the Eurodollar Rate to all or any portion of the unliquidated balance of the Primary Sum. Each Rate Period shall be one, two, three or six months, as the Grantors shall, upon notice received by the Grantee not later than 11:00 A.M. (New York time) at least four Business Days prior to the commencement of such Rate Period. select; provided, however, that (a) if the Grantors shall fail on a timely basis to select a Rate Period for any portion of the Primary Sum to which the Eurodollar Rate is applicable, the Domestic Rate shall apply to such portion of the Primary Sum as of the end of the then current Rate Period and the any Rate Period determined in accordance with Paragraph 4(d) hereof shall be for the period of time specified therein. Whenever the last day of any Rate Perior would otherwise occur on a day other than a Business Day. the last day of such Rate Period shall occur on the next succeeding Business Day; provided, however, that if such extension would cause the last day of such Rate Period to occur in the next following calendar month, the last day of such Rate Period shall occur on the next preceding Business Day,

"Reference Banks" means the re-pective principal London offices of (a) Citibath, Morgan Guaranty Trust Company of New York and Melbon Bank, N.A. and (b) in addition to or in substitution for any such bank any bank a signated by the Grantors and approved by the Grantee.

"Regular Application Date" means the first day of any calendar mouth.

"Reserve Report" shall mean a report of Ryder Scott Company Petroleum Engineers or other independent petroleum engineers agreed upon by the Grantors and the Grantee furnished pursuant to subparagraph (c) of Paragraph 19 of this Conveyance.

"Sales Contracts" has the meaning assigned to that term in Paragraph 8 hereof.

"Second Production Period" means the period commencing at 7:00 A.M., local time in effect at the location of each Subject Interest, on August 1, 1983, and ending at 7:00 A.M., local time in effect at the location of each Subject Interest, on January 1, 1990, unless the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, in which case the Second Production Period shall end at 7:00 A.M., local time in effect at the location of each Subject Interest, on May 1, 1990.

"Subject Hydrocarbons" means all Hydrocarbons in and under, and which may be produced, saved and sold from, and which shall accrue and be attributable to, the Subject Interests from and after the Effective Date and throughout the period specified in this Conveyance for the term of the Production Payment, subject to the provisions of Paragraph 3 of this Conveyance.

"Subject Interests" shall mean each and every kind and character of right, title, claim or interest which any Grantor now has in the Leases or portions thereof which are described in Exhibit A and all unitization and pooling agreements and the units created thereby which cover or include such Leases or portions thereof, and all the right, title, claim or interest which any Grantor now has in and to the lands specifically described or referred to in Exhibit A or covered by the Leases which are described in Exhibit A, whether such right, title, claim or interest be under and by virtue of a Lease, a unitization or pooling agreement, a unitization or pooling order, a mineral deed, a royalty deed, an operating agreement, a division order, a transfer order or any other type of contract, conveyance or instrument or under any other type of claim or title, legal or equitable, recorded or unrecorded, even

though such Grantor's interests be incorrectly or incompletely described in, or a description thereof be omitted from, Exhibit A. all as the same shall be enlarged by any reversionary interest or by the discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and all renewals and extensions of any of the same, but at a shall not be included within the Subject Interests any rights which any Grantor now has or may hereafter obtain with respect to production attributable to the interests of nonconsenting parties under any operating agreement, unit operating agreement, contract for development or similar instrument. Without in any way limiting the foregoing, the Subject Interests shall include each and every kind and character of right, title, claim or interest which any Grantor has in any and all Leases covering lands situated in those counties in the States of New York, Penrsylvania or West Virginia named in Exhibit A, regardless of whether such Leases are described in Exhibit A.

"Subject Minerals" means Subject Hydrocarbons and Subject Sulphur.

"Subject Sulphur" means, in the event the Supplemental Instrument of Conveyance of Production Payment is executed and delivered, all sulphur in and under, and which may be preduced, saved and sold from, and which shall accrue and be attributable to, the Subject Interests described in the Supplemental Instrument of Conveyance of Production Payment from and after the effective date thereof and throughout the remainder of the period specified in this Conveyance for the term of the Production Payment, subject to the provisions of Paragraph 3 of this Conveyance.

"Subject Taxes" has the meaning assigned to that term in Paragraph 13 hereof.

"Supplemental Instrument of Conveyance of Production Payment" means the Supplemental Instrument of Conveyance of Production Payment in the form of Exhibit C attached hereto and be reby made a part hereof. The execution and delivery of the Supplemental Instrument of Conveyance of Production Payment shall not require the consent or joinder of the Grantee or any mortgagee or trastee referred to in Paragraph 24 hereof.

"Third Production Period" means the period commencing at 7:00 A.M., local time in effect at the location of each Subject Interest, on January 1, 1990, unless the Supplemental Instrument of Conveyance of Production Payment has been excuted and delivered, in which case the Third Production Period shall commence at 7:00 A.M., local time in effect at each of the Subject Interests, on May 1, 1990 and end on the date the Production Payment is discharged or terminated as herein provided.

2. Conveyance of the Production Payment. The Grantors, in consideration of the payment to them by the Grantee of the Purchase Price pursuant to the previsions of Paragraph 7 of this Conveyance and of other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, by these presents do hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, effective as of the Effective Date, subject to the limitations and encumbrances set forth in Paragraphs 14(f) and 20 of this Conveyance and in Exhibit A. as a production payment, the undivided percentages, or if 100%, the percentage, of Subject Minerals provided for in the following subparagraphs First through Fourth of this Paragraph 2:

First: The undivided percentage of the Subject Minerals conveyed under this subparagraph First during the First Production Period shall be that undivided percentage (up to 100%) or, if 100%, the percentage, of the Subject Minerals as shall be required to cause the Grantee to be entitled to receive, out of the Subject Minerals produced and saved during each Accounting Month during the First Production Period, a sum which, when actually received by the Grantee in accordance with the terms of this Conveyance, would equal a sum hereby specified to be the aggregate amount referred to in subparagraphs (b), (c), (d), (e) and (f) of Paragraph 4 of this Conveyance to the extent such aggregate amount shall have been ascertained or accound during (or, in the case of said subparagraphs (e), (d), (e) and (f), prior to) such Accounting Month and remains unpaid;

- Second: Subject to the provisions of subparagraph Third of this Paragraph 2, the undivided percentages of the Subject Minerals conveyed under this subparagraph Second during each Accounting Month during the Second Production Period shall be the undivided percentages specified in the following clause (a) or clause (b) whichever entitles the Grantee to receive the greater mount of proceeds out of Subject Minerals produced and saved ring such Accounting Month:
 - (a) an undivided 65% of the Subject 'Lydrocarbons and an undivided 35% of the Subject Sulphur produced and saved during such Accounting Month, provided, however, that in the event the Primary Sum at the close of business on July 31, 1983 is less than \$2,500,000,000, then (1) if the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, said undivided percentages of Subject Hydrocarbons and Subject Sulphur shall be adjusted to the percentages determined by multiplying each of 65% and 35% by a fraction the numerator of which shall be the unliquidated balance of the Primary Sum at the close of business on July 31, 1983 and the denominator of which shall be \$2,500,000,000 or (2) if the Supplemental Instrument of Conveyance of Production Payment has not been executed and delivered, said undivided percenage of Subject Hydrocarbons shall be adjusted to the percentage determined by multiplying 65% by a fraction the numerator of which shall be the unuquidated balance of the Primary Sum on July 31, 1983 and the denominator of which shall be \$1,865,009,000; or
 - (b) that undivided percentage (up to 90%) of the Subject Hydrocarbons and (but only if said undivided percentage of the Subject Hydrocarbons for such Accounting Month is 90% and is insufficient to entitle the Grantee to receive the sum specified in this subparagraph (b)) that undivided percentage (up to 40%) of the Subject Sulphur produced and saved during such Accounting Month as shall be required to cause the Grantee to be entitled to receive, out of the Subject Minerals produced and saved during such Accounting Month, a sum which, when actually received by the Grantee in accordance with the terms of this Conveyance, would equal a sum hereby

specified to be the total of (i) the aggregate amount referred to in subparagraphs (b), (c), (d), (e) and (f) of Paragraph 4 of this Conveyance to the extent such aggregate amount shall have been ascertained or accrued during (or, in the case of said subparagraphs (e), (d), (e) and (f), prior to) such Accounting Month and remains unpaid, plus (ii) an amount determined by multiplying the unliquidated balance of the Primary Sum at the close of business on July 31, 1983 by the percentage shown on the Discharge Schedule under the heading "Hydrocarbons Monthly Percentage" or, in the event the Supplemental Instrument of Conv e of Production Payment has been executed and delivered, the percentage shown on the Discharge Schedthe under the heading "Minerals Monthly Percentage", opposite the Regular Applies tion Date occurring at or near the end of such Accounting Month;

Third: If on any Application Date during the Second Production Period the aggregate amount of reductions in the unliquidated balance of the Primary Sum made on such Application Date and prior thereto is less than an amount determined by multiplying 95% times the cumulative total of the amounts determined for such Application Date and all amounts determined for all previous Application Dates as provided in subclause (ii) of clause (b) of subparagraph Second of this Paragraph 2 by reference to the "Hydrocarbons Monthly Percentage" or, in the event the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, by reference to the "Minerals Monthly Percentage", then rotwithstanding the provisions of subparagraph Second above (which shall not be effective during any period during which the provisions of this subparagraph Third are operative), during the Accounting Month beginning at or near such Application Date, the undivided percentages of the Subject Minerals conveyed under this subparagraph Third shall be ar undivided 90% of the Subject Hydrocarbons and an undivided 40% of the Subject Sulphur;

and

Fourth: The percentage of the Subject Minerals conveyed under this subparagraph Fourth during the Third Production

Period shall be one hundred percent (100%) of the Subject Minerals;

together with an easement of ingress and egress to remove the same from the applicable Subject Interests to the extent Grantors have the right to grant such an easement (the undivided percentages of the Subject Minerals so granted, bargained, sold, conveyed, assigned, transferred and delivered and at any time in effect being herein called the "Production Payment Percentages", the Subject Minerals attributable to the Production Payment Percentages being herein called the "Production Payment Minerals", and the Production Payment Minerals, together with such easeme. of ingress and egress and all of er interests hereby granted, bargained, sold, conveyed, assigned, transferred and delivered, or intended so to be, being herein called the "Production Payment").

To HAVE AND To Hold the Production Payment, together with all and singular the rights, citles, interests, estates, remedies, powers, privileges and appurtenances thereunto in any way belonging, unto the Grantee, its successors and assigns, forever.

- 3. Gertain Provisions Governing the Production Payment. All the provisions of this Conveyance shall be subject to the following principles:
 - (a) Except as provided in Paragraphs 14(b)(ii) and 14(c) hereof, for the discharge of the Production Payment the Grant.c shall look exclusively to the proceeds of Production Payment Minerals, and the Grantors shall not be liable for such discharge.
 - (b) Except as provided in the proviso clause below, the Production Payment shall not be dischargeable out of any advance payments made to the Grantors for Production Payment Minerals to be produced in the future and covered by, and made in advance of such production under the terms of, any advance payment or take or pay agreement, provided, however, that the Production Payment shall be dischargeable out of, and the Grantors shall (except to the extent that payment is made directly to the Grantee by the purchaser or unless the Production Payment shall have been fully discharged) make payment to the Grantee hereunder on account of,

any Production Payment Minerals covered by any such advance payment or take or pay agreement when such Production Payment Minerals shall be produced and saved, on the basis of the full purchase price under the Sales Contract for such Production Payment Minerals so produced and saved (such full purchase price to be without regard to any deduction from or credit against the payment of such price due to any previous payment made in advance of production).

- (c) The Production Payment shall not be dischargeable out of any bonus which the Grantors shall receive for any Lease or assignment of any of the Subject Interests or out of any payments made to any of the Grantors in connection with the drilling or deferring of drilling of any well on any of the Subject Interests or in connection with any adjustment of any well and leasehold equipment upon unitization of any of the Subject Interests.
- (d) There shall not be included in the Subject Minerals any Minerals unavoidably lost in the production thereof or produced and saved from any of the Subject Interests and used by the Grantors in conformity with good field practices for drilling, mining and production operations (including gas injection, secondary recovery, pressure maintenance, repressuring or cycling operations) conducted for the purpose of producing Minerals from the Subject Interests or from any unit to which the Subject Interests are committed, but only so long as such Minerals are so used.
- (e) So long as and to the extent that the same may be required by applicable laws and regulations, in the case of any Lease from the United States of America included in the Subject Interests from which the average production of oil per well per day averaged on the monthly basis is 15 barrels or less, the obligation to pay and the right of the Grantee to receive the proceeds of oil produced from such Lease shall be suspended until said average production of oil per well per day exceeds said minimum amount, and such suspension shall apply separately to any zone or portion of such Lease segregated for computing government royalties; provided, however, that during the period of any such suspension, the Production Payment Percentage applicable to production from

each other Lease included in the Subject Interests, or portion or zone thereof, shall (subject to the provisions of Paragraph 2 hereof) be increased to that percentage necessary to cause the Grantee to receive, in addition to the production from each such other Lease (or portion or zone thereof) attributable to the Production Payment Percentages otherwise applicable under Paragraph 2 hereof, proceeds of production from each such other Lease, or portion or zone thereof, equivalent in the aggregate to the amount of proceeds of production the Grantee would have received from the Lease, or portion or zone thereof, as to which the Grantee's rights to production have been suspended had such rights not been so suspended. The increases in the Production Payment Percentages pursuant to the foregoing provisions of this subparagraph (e) are hereby conveyed by the Granters to the Grantee effective as of the Effective Date as a part of the Production Payment.

- (f) There shall not be included in the Subject Minerals any Minerals attributable to nonconsent operations conducted with respect to the Subject Interests (or any portion thereof) as to which any Grantor shall be a nonconsenting party and dedicated to the recoupment or reimbursement of costs and expenses of the consenting party or parties by the terms of the relevant operating agreement, unit agreement, contract for development or other instrument providing for such nonconsent operations, provided such Grantor's election not to participate in such operations is made in conformity with the provisions of Paragraph 14(a) of this Conveyance.
- (g) Whenever reference is made in this Conveyance to the term "fair market value at the wellhead or mine" in determining the price to be paid or obtained for Production Payment Minerals sold for the account of the Grantee, or purchased by any Grantor, or taken in kind by the Grantee, such term shall be understood to mean the price paid or payable for such Minerals at the wellhead or mine in the case of Minerals customarily sold at the wellhead or mine or, in the case of Minerals customarily sold or disposed of otherwise than at the wellhead or mine, the price paid or payable for such Minerals at the nearest point where such Minerals may be marketed and sold after production, less all costs of transporting such Minerals to such point from the wellhead or mine.

- (h) In the case c ... tion Payment Minerals sold or disposed of other than at one wellhead or mine, the amount of all reasonable costs incurred and paid by the Grantors in transporting such Minerals to the point of sale or disposition shall be deducted from the proceeds of sale of the Production Payment Minerals, and the amount so deducted shall be deemed not to be proceeds of the Production Payment Minerals received or realized by the Grantee.
- (i) Cognizant of Sections 171.26(d), 172.22(d), 173.21(d) and 174.38(b) of Title 25 of the Code of Federal Regulations, it is agreed, with respect to any Lease from Indian tribes, bands or groups, and any Lease on lands allotted to Indians in severalty, included in the Subject Interests, that nothing in this Conveyance shall be construed as modifying any of the obligations of the lessee under such Lease, including, but not limited to, obligations for diligent development and operation, protection against drainage, compliance with oil and gas operating regulations (30 CFR Part 221), and the requirement for departmental approval before abandonment of any well situated on such Lease. All such obligations are to remain in full force and effect, the same as if free of the Production Payment. The existence of the Production Payment, whether or not actually paid, shall not be considered as justification for the approval of abandonment of any well situated on such Lease.
- 4. Amount and Term of the Production Payment. The Production Payment shall continue and remain in full force and effect until such time as the Grantee shall have received and realized out of the proceeds of Production Payment Minerals, free and clear of all development, operating, mining, producing, treating, processing, handling, storing, marketing, transporting and other costs and expenses of every kind whatsoever, the full aggregate sum of the following amounts:
 - (a) The Primary Sun; plus
 - (b) An amount computed at the Applicable Pate on the unliquidated balance from time to time of the Primary Sum,
 - (i) The first such computation to be made on September 1, 1981, for the period commencing with and including the date hereof up to and including August 31, 1981, on the amount of such unliquidated balance of the Primary Sum from time to time during such period, and

(ii) Subsequent computations to be made monthly on the first day of each calendar month for the preceding calendar month on the amount of the unliquidated balance from time to time of the Primary Sum during such preceding calendar month;

plus

- (c) The sum of \$100,000 payable on September 1, 1981, and, in the event the Primary Sum at any date exceeds \$1,000,000,000, the additional sum of \$100,000 payable on the Application Date next following such date, and, in the event the Primary Sum at any date exceeds \$1,865,000,000, the additional sum of \$50,000 payable on the Application Date next following such date; plus
- (d) A sum equal to the aggregate of (i) all Increased Financing Costs from time to time of the Grantee and (ii) with respect to each portion of the unliquidated balance of the Primary Sum to which the Eurodollar Rate is applicable which may be paid on other than the last day of the Rate Period therefor (including without limitation any payments made pursuant to subparagraph (f) of Paragraph 6 bereof or Caragraph 25 hereof), an amount co-aputed on the basis of the Excess Rate (as defined below) for such portion of the unfiquidated balance of the Primary Sam on the amount of such portion so paid, for the period commencing on the day of such payment and ending on the last day of the current Rate Period for such portion (for the purpose of computing such amount, the "Excess Rate" in respect of any portion of the Primary Sum so id shall be a rate per annum equal to the excess (if any) of (x) the LIBO Rate in effect for the current Rate Period for such portion over (v) the LIBO Rate computed on the basis of a Rate Period commencing on the day of such payment and ending on the last day of the current Rate Period for such portion) and (iii) interest on the amounts specified in clauses (i) and (ii) at the Applicable Rate (but not to exceed the maximum rate permitted by applicable law);

plus

(e) A sum equal to the aggregate of the Subject Taxes to the extent that such taxes have not becaude by the Grantors

pursuant to Paragraph 13 hereof, together with interest on the unliquidated amount thereof at a rate equal to the Applicable Rate (but in no event in excess of the maximum rate permitted by applicable law) in effect during the period from and including the date of payment of each such amount by the Grantee (notice of each such payment to be given promptly to the Grantors) to the date the Grantee is reimbursed therefor; plus

(1) A sum equal to the aggregate of all reasonable expenses of the Grantee incidental to the acquisition, ownership, mortgaging and transfer of the Preduction Payment or any part thereof. the obtaining of any loan secured thereby and the receipt and disbursement of funds on account thereof (including, without limitation, any commitment fees, any withholding or other similar taxes (together with any interest, penalties and expenses in connection therewith), any costs and expenses of any lender making such loan incurred as a result of any installment of the Purchase Price not being paid on the date specified in a notice by the Grantors pursuant to Paragraph 7() of this Conveyance or failure to make a payment on the date and before the the specified in a notice provided pursuant to P ragraph 6 with respect to any portion of the unliquidated balan cof the Princry Sum to which the Eurodollar Rate is applicable and a, fees and expense of accountants and counsel for the Grantee and of consel for sel for sel for any lenders making such loan, all exp uses and taxes imposed by any state within which any Subject laterest is wholly or partially located on such lender attributable to the mortgaging of the Production Payment, all fees incurred in connection with, and all other expenses and costs of litigation and the contest, release or discharge of, any adverse claim or demand made or proceedings instituted by any person affecting in any manner whatsoever the Production Payment, the Production Payment Minerals or the proceeds thereof and all other fees, costs and expenses of the Grantee and any lender making such loan in connection with or crising out of any investigation, litigation or proceeding related to an, acquisition or proposed acquisition of all or any portion of the stock or substantially all of the assets of any Person, directly or indirectly, with any of the proceeds of the Purchase Price, which shall have been paid or incurred by the Grantee or any lender making such loan, together with interest on the unliquidated amount thereof at a rate equal to the Applicable Rate (but not to exceed the maximum rate permitted by applicable law) in effect during the period from and including the date of payment of each such amount by the Grantee (notice of each such payment to be given promptly to the Granters) to the date the Grantee is reimbursed therefor.

IT BEING THE INTENTION OF THE GRANTORS AND THE GRANTEE IN ENTERING INTO THIS CONVEYANCE that the Grantee shall be entitled to receive and realize out of the Production Payment Minerals the full aggregate sum of the amounts described in subparagraphs (a). (b) and (c) of this Paragraph 4, free and clear of all costs and expenses with respect to the Subject Interests and the Subject Minerals (which expenses, costs and fees shall be borne by the Grantors) and over and above all taxes, assessments, fees, expenses and costs of the character and amount described, specified or referred to in subparagraphs (d), (e) and (f) of this Paragraph 4.

All of the Subject Taxes may be deducted from the proceeds of the Production Payment Minerals and paid for the account of the Grantee and, in the event of any such deduction, the amount so deducted shall be deemed not to be proceeds of the Production Payment Minerals received or realized by the Grantee. Similarly (and anything in this Conveyance to the contrary notwithstanding), the tax under the Crude Oil Windfall Profit Tax Act of 1980 and any amendments or supplements thereto (or any tax enacted in substitution therefor) imposed or assessed. In respect to or measured by or charged against or attributable to the Production Payment Minerals or the proceeds thereof and for which the Grantors are liable may be deducted from the proceeds of the Production Payment Minerals and paid by the Grantors and, in the event of any such deduction, the amount so deducted shall be deemed not to be proceeds of the Production Payment Minerals received or realized by or payable to the Grantee.

5. Termination of the Production Payment. When the full aggregate sum of the amounts specified in subparagraphs (a), (b), (c), (d), (e) and (f) of Paragraph 4 of this Conveyance (as expressly provided by the terms of this Conveyance) has been received by the Grantee, the Production Payment shall forthwith terminate, and all interest therein shall impediately revert to and become vested in the

Grantors to the same extent and with the same force and effect as if this Conveyance had not been made. The Grantee agrees that, upon such termination and upon indemnification against loss by reason of such acts and by reason of the restitution of proceeds of Production Payment Minerals as provided in Paragraph 11 of this Conveyance, it shall execute or cause to be executed, upon the request and at the expense of the Grantors, such instruments as may be necessary or appropriate to evidence the termination of the Production Payment.

- 6. Application of the Proceeds of Production Payment Minerals. For all purposes of this Conveyance, the proceeds of Production Payment Minerals actually received by the Grantee at or before 11:00 A.M. (local time in effect in The City of New York) on any Application Date (commencing with September 1, 1981) and not theretofore applied pursuant to this Paragraph 6, shall be deemed to have been received and applied on such Application Date as follows:
 - (a) First, to the amount referred to in surparagraph (b) of Paragraph 4 of this Conveyance to the extent such amount has been ascertained or accrued up to (but not including) such Application Date and which has not been added to the Primary Sum pursuant to the proviso to this Paragraph 6 or paid by any previous application of proceeds of Production Payment Minerals pursuant to the provisions of this Paragraph 6, and
 - (b) Second, to the amount referred to in adaptargraph (d) of said Paragraph 4 to the extent such amount has been ascertained or accrued up to (but not including) such Application Date and which has not been paid by any previous application of proceeds of Production Payment Minerals pursuant to the provisions of this Paragraph 6, and
 - (c) Third, to the amount referred to in subparagraph (c) of said Paragraph 4 and which has not been paid by any previous application of proceeds of Production Payment Minerals pursuant to the provisions of this Paragraph 6, and
 - (d) Fourth, to the amount referred to in subparagraph (e) of said Paragraph 4, to the extent then as a sained and which has not been paid by any previous application of proceeds of Produc-

tion Payment Minerals pursuant to the provisions of this Paragraph 6, and

- (e) Fifth, to all other amounts referred to in subparagraph (f) of said Paragraph 4, to the extent then ascertained and which have not been paid by any previous application of proceeds of Production Payment Minerals pursuant to the provisions of this Paragraph 6, and
- (f) Sixth, to the reduction of the unliquidated balance of the Primary Sum, initially by reduction of that portion of the unliquidated balance of the Primary Sum to which the Domestic Rate is applicable and then by reduction of those one or more portions of the unliquidated balance of the Primary Sum to which a Eurodollar Rate is applicable in such order as may be selected by the Grantors by not less than four Business Days' prior written notice to the Grantee or, if and to the extent not so selected, in the order in which the last days of the respective Rate Periods then in effect are scheduled to occur;

provided, however, that, in the event such proceeds so applied on the relevant Application Date shall be insufficient to cover the full amount specified in subparagraph (a) of this Paragraph 6, the unliquidated balance of the Primary Sum shall forthwith be increased by an amount equal to the amount of such deficiency.

Within 15 Business Days after the date of each such application by the Grantee of any proceeds of Production Payment Minerals, the Grantee will furnish or cause to be furnished to the Grantors a written statement showing the application of such proceeds in accordance with the provisions of this Paragraph 6.

If the Eurodollar Rate is applicable to any portion of the unliquidated balance of the Primary Sum to which proceeds are to be applied on an Application Date and such Application Date is on a date other than the last day of the Rate Period for such Eurodollar Rate Grantors shall give written notice to the Grantee (and any lenders to the Grantee) at least four Business Days prior to such Application Date that the Grantors will pay such portion of the unliquidated balance of the Primary Sum prior to 11:00 A.M. (local time in effect in The City of New York) on such Application Date.

7. Payment of Consideration for t.

- (a) In consideration of the conveyan-The sale. Grantee of the Production Paymer' bursuant or this Conveyance, the in Crantors to the Grantee hereby agrees to pay to the tirus, ors the sem of \$2,500,000,000 (such sum, less the aggregate a roant of reductions thereto pursuant to the provisions of subparagr d) of dis Paragraph 7, being herein called the "Purchase Pr., ", of which \$10,000,000 is being paid by the Grantee to Grantors concurrently with the execution and delivery of this Conveyance and the remaining \$2,490,000,000 is payable in one or more installments, each to be in an amount not less than \$50,000,000 (except that any installment which causes the aggregate amount of the Purchase Price then paid to equal \$1,865,000,000 or, if the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, \$2,500,000,000, may be in such lesser amount which shall cause the aggregate amount of the Purchase Price then paid to equal \$1,865,000,000 or \$2,500,000,000 as the case may be), payable by the Grantee at the times and in the amounts specified by the Grantors as hereinafter in this Paragraph 7 provided.
- (b) Each installment of the remaining \$2,490,000,000 of the Purchase Price shall become payable on the Business Day as to which the Grantee shall have received from the Grantors at least four Business Days' prior written notice specifying the amount of such installment and the Business Day for payment of such installment. Any provision of this Paragraph 7 to the contrary notwithstanding, the obligation of the Grantee to pay each installment of the Purchase Price on the Business Day on which such installment shall otherwise be payable hereunder shall be subject to the conditions precedent that on such Business Day (i) the Grantee shall have received a certificate, substantially in the form of Exhibit D attached hereto and hereby made a part hereof, of the Grantors signed by duly authorized officers of the Grantors and dated such Business Day (the statements made in which certificate shall be true and correct on such Business Day), and (ii) as to the installment which, when paid, would cause the aggregate amount of the Purchase Price then paid to exceed \$1,865,000,000, the Grantors shall have executed and delivered to the Grantee the Supplemental Instrument of Conveyance of Production Payment.
- (c) Any provision of this Paragraph 7 to the contrary notwithstanding, the obligation of the Grantee to pay any further installments

of the Purchase Price pursuant to this Paragraph 7 shall terminate upon the occurrence of an Event of Default specified in clause (c) or (d) of Paragraph 17.

- (d) The obligation of the Grantee to pay any installment or installments of the Purchase Price which is not or are not payable on or prior to (i) the date of termination of the Production Payment pursuant to Paragraph 5 hereof or (ii) July 31, 1, 83, whichever date first occurs, shall terminate on such date and thereupon the amount of the Purchase Price shall be reduced by the amount of the unpaid portion of the Purchase Price which the Grantee is not obligated to pay.
- (e) All payments made by the Grantee on account of the Purchase Price shall be paid in lawful money of the United States of America and in immediately available funds by a deposit to the credit of the Grantors' account (No. 39019761) with Citibank at 399 Park Avenue, New York, New York 10043.
- (f) The Grantors hereby irre ocably and unconditionally waive and relinquish (i) any express or implied vendor's lien, and any other lien, security interest, charge or encumbrance, which would otherwise be imposed on or affect the Production Payment or the Production Payment Minerals on account of any unpaid installment or installments of the Purchase Price and (ii) with respect to the obligations of the Grantors to make payments under this Conveyance (including, without limitation, payments of proceeds of Production Payment Minerals under Paragraphs 3, 8, 9 and 11 hereof and payments under Paragraphs 13, 17 and 25 hereof), any set-off, counterclaim, credit, recoupment, defense, abatement, suspension, reduction and other right or claim which the Grantors may have against the Grantee as a result of or arising out of the failure of the Grantee to pay any amount on account of the Purchase Price.
- 8. Marketing of Production Payment Minerals. The Grantors, at their own expense, shall market or cause to be marketed the Production Payment Minerals, on behalf of and for the account of Grantee, on the same basis as the Grantors market their share of Subject Minerals, and for such purpose the Grantee agrees that sales of Production Payment Minerals may continue to be made by the Grantors with their share of Subject Minerals pursuant to existing contracts and agree-

ments entered into by the Grantors for the sale of, or commitment to sell, Subject Minerals and that the Grantors may enter into one or more contracts and agreements in the future for the sale of Subject Mine als (including Production Payment Minerals included therein) at the best prices and on the best terms as the Grantors shall determine in their good faith judgment are reasonably obtainable in the circumstances (all such present and future contracts and agreements for the sale of or commitment to sell Subject Minerals being herein called "Sales Contracts"), and it shall not be necessary for the Grantee to join in any Sales Contract or any amendment thereof; provided, however, that the Grantors will not agree to, or permit, any amendment, modification or termination in any manner adverse to the interests of the Grantee of any Sales Contract with an Affiliate of any Grantor, nor will the Grantors enter into any Sales Contract in the future with an Afhiiate of any Grantor (except pursuant to the terms of any presently existing Sales Contract) in any manner adverse to the interests of the Grantee, without the prior written consent of the Grantee, and the Grantors will give the Grantee at least 15 days' prior written notice of the terms of any proposed amendment, modification or termination of any Sales Contract with an Affiliate of any Grantor, or any proposed future Sales Contract with an Affiliate of any Grantor. Subject to the rights of purchasers under then existing Sales Contracts and to any existing right of the Grantee to take in kind pursuant to notice given in accordance with Paragraph 10 hereof, and (so long as the Grantors are not in default in the performance or observance of any covenant or agreement on the Granters' part to be performed or observed hereunder) the Grantors all be entitled at any time and from time to time to purchase all or any part of the Production Payment Minerals (including Production Payment Minerals which the Grantee has elected to take in kind provided that the Grantors shall have, within 30 days of the giving of any notice by the Grantee in accordance with Paragraph 10 hereof, given written notice to the Grantee that the Grantors elect to purchase all or part of the Production Payment Minerals specified in such notice given by the Grantee and specifying when the Grantors' purchases shall commence and the duration thereof), and, in such case, the Grantors shall pay to the Grantee, on or before 11:00 A.M. (local time in effect in the City of New York) on the Application Date at or near the end of the Accounting Month next succeeding the Accounting Month in which such purchase by the Grantors occurs, in immediately available funds, a price

equal to the fair market value at the wellhead or mine at the time of purchase of the Production Payment Minerals so purchased, less the amount of all severance taxes and assessments, payable by the Grantee or the Grantors, of the character referred to in Paragraph 4(e) or the last sentence of Paragraph 4, attributable to the production and sale of such Production Payment Minerals, and the amount of such taxes and assessments so deducted shall be deemed not to be proceeds of Production Payment Minerals received or realized by the Grantee.

9. Performance of Sales Contracts; Payments; Etc. The Grantors will duly perform all obligations binding on them under all Sales Contracts in accordance with the terms thereof and will take all appropriate measures to enforce the performance under each of the Sales Contracts of the obligations of the purchaser thereunder. All Production Payment Minerals sold by the Grantors, whether pursuant to Sales Contracts or otherwise, shall be delivered by the Grantors to the purchasers thereof, for the credit of the Grantee, to such point of purchase as is reasonably required in the marketing of such Product. Payment Minerals. So long as no Event of Default has occurred and is continuing, the proceeds of the sale of all Subject Minerals (whether pursuant to Sales Contracts or otherwise) shall be paid to the Grantors by the purchasers thereof or other parties obligated to make payment therefor. All proceeds of Production Payment Minerals shall constitute trust funds in the possession of the Grantors and all such proceeds of Production Payment Minerals received by the Grantors on or before the next to last Business Day of each calendar month which are proceeds of Production Payment Minerals produced and saved during any Accounting Month prior to the Accounting Month ending on the first day of such calcudar month and not theretofore paid over shall be paid over to the Grantee at or before 11:00 A.M. (local time in effect in The City of New York) on the Application Date which next follows the next to last Business Day of such calendar month in lawful money of the United States of America and in immediately available funds by a deposit to the credit of the Grantee's account (No. 39019753) with Citibank at 399 Park Avenue, New York, New York 10043. Upon the occurrence and during the continuance of any Event of Default (but not otherwise), the Grantee shall have the right to receive directly from the purchasers of the Production Payment Minerals, and to direct

such purchasers to pay directly to the Grantee, amounts to be paid for purchases of Production Payment Minerals, whether pursuant to Sales Contracts or otherwise.

- 10. Taking In Kind. From time to time upon the giving of at least 120 days' prior written notice to the Grantors (except that no such notice shall be required if an Event of Default shall have occurred and is continuing), the Grantee shall have the right, at its option, to take in kind all or any part of the Production Payment Minerals; subject, however, to (i) the rights of any purchasers under then existing Sales Contracts, and (ii) the right of the Grantors to purchase Production Payment Minerals pursuant to Paragraph 8 hereof. Each such notice from the Grantee shall specify the Minerals to be taken in kind, the date such taking is to commence and the proposed duration thereof. The Frantee shall market any such Production Payment Minerals promptly at the best price and on the best terms the Grantee shall determine in its good faith judgment are reasonably obtainable in the circumstances, and the Production Payment shall be credited with the proceeds of any such sale, in the same manner as if such Production Payment Minerals had been marketed by the Grantors, No such taking by the Grantee shall obligate the Grantors to incur additional expenses by reason of such taking, but the Grantors shall continue to pay all costs and expenses incurred by them in the same manner and to the same extent hereunder as if there had been no such taking, provided that the Grantors shall be reimbursed out of the proceeds of sale of such Production Payment Minerals marketed by the Grantee pursuant to this Paragraph 10, to the extent actually incurred by the Grantors, for the amount of all the transportation costs of the character referred to in Paragraph 3(h) and for the amount of all Subject Taxes attributable to the production and sale of the Production Payment Minerals so taken and marketed by the Grantee and the amount of any such transportation costs and Subject Taxes so reimbursed to the Grantors shall be deemed not to be proceeds of Production Payment Minerals received or realized by the Grantee.
- 11. Withholding and Restitution of Proceeds of Production Payment Minerals. All obligations of the Grantors hereunder shall be subject to the applicable provisions of each statute purporting to provide regulation of the sale of Minerals or establishing maxi-

mum prices at which the same may be sold and all applicable rules and regulations thereunder of each governmental agency, board or commission having jurisdiction. Rates permitted under each such statute and the rules and regulations thereunder to be paid for natural gas or other Minerals included in the Production Payment Minerals shall be controlling if lower than prices established in Sales Contracts or if lower than the fair market value at the wellhead or mine. The Grantors shall be entitled to use their reasonable discretion in making filings, for themselves and on behalf of the Grantee, with any governmental agency, board or commission having jurisdiction, affecting the price or prices at which Subject Minerals may be sold. If any proceeds of Production Payment Minerals shall be withheld for any reason whatsoever, including, without limitation, withholding by the Grantors pending a final, non-appealable order on any application for certificates or determinations with respect to a sale, or a rate increase, filed with any governmental agency, board or commission having jurisdiction (and the Grantors are hereby authorized to make such withholding pending such final determination), the Grantee shall not be deemed to have received or realized such proceeds until, and only to the extent that, such proceeds shall actually have been received by the Grantee; provided, however, that promptly (and in no event later than 30 days) following the receipt by the Grantors of a final non-appealable order in any such rate proceeding, the Grantors shall promptly pay over to the Grantee the amount of any proceeds of Production Payment Minerals so withheld by the Grantors, in conformity with such non-appealable order. If, at any time before or after receipt of the full aggregate sum specified in subparagraphs (a), (b), (c), (d), (e) and (f) of Paragraph 4 of this Conveyance, the Grantee shall be compelled, for any reason whatsoever, to make any payment or restitution, or such payment or restitution shall be made for the account of the Grantee, on account of proceeds of Production Payment Minerals, which proceeds have a teady been applied pursuant to the provisions of Paragraph 6 of this Conveyance, the unliquidated balance of the Primary Sum shall forthwith be increased by the amount of the proceeds so paid over by ar for the account of the Grantee plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties; provided, however, that, if, at the time of such payment or restitution, the entire Primary Sum shall have been liquidated, the Primary Sum shall be deemed not to have been so liquidated and the unliquicated balance of the Primary Sum shall thereupon be deemed to be in the amount of the proceeds so paid over plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties.

- 12. Protection to Purchasers of Production. No person purchasing or taking or processing any Subject Minerals shall be required to take notice of, or keep informed as to, the discharge or termination of the Production Payment until the actual receipt by such person of written notice advising such person of such discharge or termination.
- 13. Certain Payments by Grantor. The Grantors will tender and pay, or cause to be paid, punctually, before they shall become delinquent (or as to any thereof which are being contested in good gaith, promptly after the final determination of such contest), the following:
 - (a) all ad valorem taxes (or taxes imposed in lieu thereof), all gross production, severance, excise and sales taxes, the excise tax imposed on any windfall profit from taxable crude oil by the Crude Oil Windfall Profit Tax Act of 1980 and any amendments or supplements thereto or tax enacted in substitution therefor, and any similar tax imposed or assessed with respect to or measured by or charged against or attributable to the Production Payment or the Production Payment Minerals, or the proceeds thereof, or against the Grantee by reason of its ownership of the Production Payment or against any assignee or mortgagee of the Production Payment by reason of his or its interest as assignee or mortgagee, together with any interest or penalty payable in connection therewith, but any other tax, including, but not limited to, any tax imposed on net income or the right to do business shall be payable only to the extent provided in subparagraph (b) below:
 - (b) all taxes or other governmental charges of any kind or character whatsoever imposed on the Grantee (but not an assignee of Grantee) by any jurisdiction, including franchise and income taxes, other than income taxes imposed on (i) the amount received pursuant to subparagraph (c) of Paragraph 4 hereof, and (ii) any gain attributable to the sale or other disposition, in whole or in part, of the Production Payment to the extent such gain is attrib-

utable to consideration received by the Grassian excess of the unliquidated balance of the Primary Sum allesable to the portion of the Production Payment so sold or disposed of; and

(c) obligations of the character referred to in subparagraph (f) of Paragraph 4 hereof.

Grantors shall have the right at their expense to contest any tax in good faith. The taxes which Grantors have agreed to pay or cause to be paid in subparagraphs (a) and (b) above are herein called "Subject Taxes".

- 14. Covenants of the Grantors. Until such time as the Production Payment has terminated as provided in Paragraph 5 of this Conveyance, the Grantors will, at their expense and regardless of who may be the operator or operators of the Subject Interests:
 - (a) subject to the provisions of subparagraph (c) of this Paragraph 14 and the provisions of Paragraph 21 of this Conveyance, cause the Subject Interests to be maintained and developed and continuously operated for the production of Minerals in a good and workmanlike manner and in accordance with sound field or mining practices and all applicable federal, state and local laws, rules and regulations (except these being contested in good faith), all such operations to be carried on, to the extent reasonably practicable, without any preference being given to Minerals from sources other than the Subject Interests because of the fact that Grantors have created and conveyed the Production Payment: provided, however, that nothing contained in this subparagraph (a) shall be growed to prevent or restrict any Grantor from electing not to part to in any operation which is to be conducted under the terms of the operating agreement, unit operating agreement, contract for development or similar instrument affecting or pertaining to the Subject Interests (or any portion thereof) and allowing consenting parties to conduct nonconsent operations thereon, if such election is made by such Grantor in good faith and in conformity with sound feld or mining practices:
 - (b) jointly and severally assume liability for, and hereby agree to indemnify, protect, defend, save and keep harmless the

Grantee, and any lender to the Grantee and their representatives, and any successors and assigns from and against any and all royalties and all liabilities, obligations, losses, damages, penalties, fines, claims, actions, suits, costs, expenses and disbursements (including, without limitation, fees and disbursements of counsel and other legal and investigative fees and expenses) of whatsoever kind and nature, which may at any time be imposed on, be incurred by or on behalf of or be asserted against the Grantee or any such lender and in any way related to or arising out of (i) the ownership, producing, handling, storing, transporting, refining, controlling or other disposition of any Subject Minerals or any Production Payment Minerals (including, without limitation, the leakage or spillage of, or other pollution by, any Subject Minerals or Production Payment Minerals, or any tort claim or claim for damages of any kind or nature and any claim or liability in respect of any adverse environmental impact or effect) or (ii) any investigation. litigation or proceeding related to any acquisition or proposed acquisition of all or any portion of the stock or substantially all of the assets of any Person directly or indirectly with any of the proceeds of the Purchase Price, whether or not the Grantee or any such lender is a party to such acquisition. The obligations of the Grantors under this subparagraph (b) shall survive the termination of the Production Payment. All payments required to be made pursuant to clause (ii) of this subpregraph (b) shall be made directly to the Grantee or such lender at Citibank New York, upon written demand by the Grantee or such lender. No provision of this subparagraph (b) or any other provision of this Conveyance shall be construed to be a waiver on the part of the Grantee or such lender of the benefit of any limitation of, or exemption from, liability accorded by any statute or rule of law;

(c) cause all machinery, equipment and facilities of any kind (including production platforms) now or hereafter located on the Subject Interests (or any portion thereof) and any arry to the effective operation thereof for the production of the condition, and all repairs, renewals, replacements, accitions and improvements thereof or thereto needed for such purpose to be duly made or provided for; provided, however, that the Grantors

shall not be required to repair, replace or rebuild any such machinery, equipment or facilities as to Subject Interests from which Minerals are produced (i) when such Subject Interests are jointly owned and are operated by a party other than any Grantor or an Affiliate of any Grantor and the operator decides not to repair, replace or rebuild such equipment, (ii) when failure to do so would not materially adversely affect the rate of production of Minerals, or (iii) if, as to any production platform located on the Subject Interests (or any portion thereof), including without limitation, equipment, machinery and facilities of any kind thereon, or any well or wells located on any such production platform, the Grantors shall pay over to the Grantee, for application to the discharge of the Production Payment, (x) quarterly, on or before 11:00 A.M. (local time in effect in The City of New York) on each Quarterly Application Date (herein in this Paragraph 14(c) called a "relevant Quarterly Application Date"; commencing with the second such Quarterly Application Date next following the date of loss of or damage to the relevant platform, that amount which, when added to all other amounts received by the Grantee pursuant to the provisions of Paragraph 6 hereof during 11 period commencing with (but not including) the la., preceding Quarterly Application Date and ending with and including the relevant Quarterly Application Date, will cause the Grantee to receive on or before the relevant Quarterly Application Date a sum equal to (A) the aggregate of the amounts to be applied in accordance with subparagraphs (a), (b), (c), (d) and (e) of Paragraph 6 hereof during said quarterly period, plus, after the commencement of the Second Production Period, (B) an amount to be applied in accordance with subparagraph (f) of said Paragraph 6 sufficient to reduce the unliquidated balance of the Primary Sum on or before the relevant Quarterly Application Date to an amount equal to (1) the unliquidated balance of the Primar Sum on the last preceding Quarterly Application Date (after the application made on that date) less (2) an amount determined by multiplying the Primary Sum at the close of business on July 31, 1983 by the percentage shown on the Discharge Schedule under the heading "Hydrocarbons Monthly Percentage" or, in the event the Supplemental Instrument of Conveyance of Production Payment has been red, under the heading "Minerals Monthly executed and de

Percentage", for each of the Application Dates in said quarterly period, and aggregating the products of such calculations (the aggregate of all such quarterly payments made under this clause (x) in respect of such platform (including all wells located thereon) not to exceed the aggregate fair market value at the wellhead, as reasonably determined by the Grantors with the concurrence of the Grantee, or, if the Grantee shall not concur with such determination by the Grantors, as dermined by Ryder Scott Company Petroleum Engineers or othe Independent petroleum engineers satisfactory to the Grantors and the Grantee, as of the date of loss of or damage to the relevant platform, of all Subject Minerals that would have been produced from wells previously completed on such platform had the necessary repairs, replacements or rebuilding of such production platform, and any neces sary redrilling or reworking of such wells, been carried out), or, at the sole option of the Grantors, (y) within thirty (30) days after the first to occur of (A) the receipt by the Grantors of the proceeds of any insurance recovered by the Grantors on account of loss of or damage to such platform or (B) the expiration of one (1) year from the date of such loss or damage, an amount equal to the aggregate fair market value at the wellher of the Subject Minerals referred to, and determined as provided in, clause (x) above, and the amount of any such payments to the Grantee shall be deemed to be proceeds of Production Payment Minerals for all purposes of this Convey nee and shall be included in the written statement to be furnished to the Grantors by the Grantee pursuant to the last paragraph of Paragraph 6 hereof, but may be reflected as a payment received from the Grantors pursuant to this subparagraph(c):

- (d) cause all necessary and proper s, ups to be taken diligently to protect and defend the Subject Interests, the Subject Minerals and the proceeds thereof against any adverse claim or demand, including, without limitation, the employment or up of counsel for the prosecution or defense of litigation and the contest, settlement, release or discharge of any such claim or demand.
- (e) cause written notice to be given to the Grantee of every adverse claim or demand made by any Person materially affecting the Subject Interests, the Subject Minerals or the proceeds trees-

of in any manner whatsoever, and of any proceeding instituted in respect thereof;

- (f) cause the interest of the Grantors in each of the Subject Interests to be kept free and clear of liens, charges and encumbrances of every character, other than those which are of a character customarily found in connection with comparable drilling, mining and producing operations and which do not materially adversely affect the operation of the Subject Interests and other than those becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Conveyance;
- (g) subject to provisions of Paragraph 21 of this Conveyance, cause all Leases included in or relating to the interests described or referred to in Exhibit A and under which the Grantor is lessee, sub-lessee or assignee to be maintained in full force and effect without amendment, modification or waiver of any provision of any such Lease which might materially adversely affect the rights of the Grantee under this Conveyance;
- (h) (i) maintain, or cause to be maintained, insurance insuring the rantors, to the extent of their respective interests therein. against loss of or damage to any and all machinery, equipment and facilities of any kind (including all production platforms) now or hereafter located on the Subject Interests (or any portion thereof) in amounts equal to the full insurable value of 'he Grantors' interest (as if this Conveyance had : at been executed and delivered) in such machinery, equipment and facilities and covering loss or damage due to fire and windstorms, hurricanes and such other events, including blowout and cratering, usually covered by insurance carried by companies engaging in similar business and owning similar properties in the same general area or areas in which the relevant Subject Interests are located, and (ii) in addition to the insurance coverage required under clause (i) above, maintain, or cause to be maintained, insurance in such amounts and covering such risks as is usually carried by companies engaged in similar business and owning similar properties in the same general area or areas in which the relevant Subject Interest, are located and a may be required by applicable law. which coverage shail in any event include the following:

- (A) general public liability and property damage insurance, endorsed to include offshore operations, insuring the Grantors against liability for injury to persons or property as a result of or growing out of the ownership, operation or development of the Subject Interests (or any portion thereof) or the production or transportation of Minerals therefrom, with a limit of liability of not less than \$500,000 for each occurrence:
- (B) automobile public liability and property damage insurance, insuring the Grantor against liability for injury to persons or property growing out of the ownership or operation of automobiles (including leased automobiles) in connection with the activities described in clause (A) above, with a limit of liability of not less than \$500,000 for each occurrence;
- (C) aviation public liability and property damage insurance, insuring the Grantors against liability for injury to persons or property growing out of the ownership or operation of aircraft (including leased aircraft) in connection with the activities described in clause (A) above, with a limit of liability of not less than \$10,000,000 for each occurrence; and
- (D) marine peod linbility and property insurance, insuring the Grantors against liability for injury to persons or property growing out of the ownership or operation of barges or vessels (including chartered barges or vessels) in connection with the activities described in clause (A) above, with a limit of liability of not less than \$500,000 for each occurrence.

All insurance required by this subparagraph (h) shall be maintained with responsible and reputable insurance companies or associations.

(i) In the event the Supplemental Instrument of Conveyance of Production Payment is executed and delivered, cause all contracts and rights (as in effect on the date hereof), with respect to the gas and water supply used in connection with the mining of the Subi et Subplier, to be no utained in full force and effect without any amondo and, prodification or waiver which might materially adversely affect the rights of the Grantee.

Notwithstanding the provisions of subparagraphs (a) and (g) of this Paragraph 14, the Grantors shall have the right to abandon any Subject Interest or any portion thereof from which Moncrals have been produced and to discontinue performance of their obligations under said subparagraphs with respect to such Subject Interest or such portion thereof when (A) with respect to Subject Interests in Hydrocarbons there is no well located on such Subject Interest or such portion thereof which is capable of producing Hydrocarbons in paying quantities and when the Grantors have reasonably determined (i) that no existing well may be recompleted, and no additional well may be completed, thereon which would be capable of producing Hydrocarbons in paving quantities, and (ii) that no pressure maintenance or other recovery techniques or methods then currently in use by prudent operators may be instituted with respect to any existing well which would result in the production from such well of Hydrocarbons in paying quantities (for purposes of this sentence, any Subject interest or any portion thereof which is committed to a unit or a joint operation and which is thereafter permanently abandoned under the applicable unit or operating agreement prior to termination of the unit or joint operation, shall be deemed an abandonment of such Subject Interest or such portion, as the case may be, by the Grantors which is permitted); and (B) with respect to Subject Interests in Sulphur. there is no mine located on such Subject Interest or such portion thereof which is capable of producing Sulphur in paying quantities.

For all purposes of this Paragraph 14, an existing well or mine located on any Subject Interest shall be deemed to be capable of producing Minerals "in paying quantities" unless a condition not of a temporary nature shall exist whereby the aggregate value of the Minerals produced from such well or mine, as the case may be, is less than the direct costs of operating such well or mine and preparing the production thereof for marketing. Further, a well on any Subject Interest in Hydrocarbons which may result from additional drilling or which may be the subject of recompletion, pressure maintenance or other recovery techniques or sethods shall not be deemed to be capable of producing Hydrocarbons "in paying quantities" if, in the light of conditions existing at the time of the determination and which are not of a temporary nature, it is estimated that the aggregate value of the Hydrocarbons to be produced from such well shall not exceed the then

estimated direct costs of operating such well and preparing the production therefrom for marketing, as well as the direct costs of providing the necessary installations for, and carrying out, any pressure maintenance or other recovery techniques or methods properly allocable to such well and the direct costs of recompleting such existing well or drilling, completing and equipping such additional well, in each case to the extent such direct costs are allocable to such Subject Interest.

As used herein, "direct costs" shall mean and include only (i) the cost of labor, transportation, other services, materials, equipment and supplies, (ii) amounts paid or to be paid independent contractors and (iii) amounts charged under applicable operating or unit agreements for or in lieu of general or administrative overhead or, with respect to those Subject Interests which are not subject to operating agreements, a reasonable amount for general and administrative overhead (which latter amount shall not in any event exceed 5% of the costs and amounts referred to in clauses (i) and (ii) above). Payments on account of the Production Payment or on account of royalties, overriding royalties or any other payments out of production shall not be deemed to constitute "direct costs".

Anything contained in this Paragraph 14 to the contrary notwithstanding, the Grantors shall not be obligated to perform undertakings performable only by the operator and which are beyond the control of the Grantors with respect to individual Subject Interests of which none of them is the operator. In such case, however, the Grantors will use their best efforts to secure the performance of each of such undertakings.

15. Access to the Subject Interests. The Grantors shall permit any one or more representatives designated by the Granter to make any inspection of the Subject Interests at any reasonable time as the Grantee may reasonably request. In addition, the Grantee shall furnish to the Grantee such detailed information as the Grantee may reasonably request in writing concerning the Subject Interests, the operation thereof and the production and sale of the Subject Minerals therefrom, provided that the Grantee shall conform to such confidentiality and secrecy requirements respecting such information as the Grantors shall in writing reasonably request.

- 16. Pooling and Unitization. At vthing herein to the contrary notwithstanding, the Grantors may, without the consent of the Grantee (unless any party having an interest in the proposed unit is an Affiliate of any Grantor holding an interest in the unit not attributable to a Subject Interest, in which case the prior written consent of the Grantee shall be required to be obtained (except as to a unit required by governmental authoristic for which no consent shall be required), which consent shall not be unreasonably withheld), pool, combine, communitize or unitize any Subject Interest (or any portion thereof) from time to time with other leasehold, mineral or other interests to form units, or to reform, expand or modify units theretofore formed, when, in the judgment of the Grantors, it is necessary or advisable to do so in order to facilitate the orderly development or operation (including repressuring, pressure maintenance, cycling, water flood and secondary recovery operations) of any Subject Interest (or any portion thereof). The interest in any such unit accruing or attributable to any Subject Interest (or any portion thereof) included therein shall be subject to the Production Payment in the same manner and with the same effect as though such unit were specifically described in Exhibit A. The signature or joinder of the Grantee shall not be necessary to the instruments evidencing the pooling and unitization herein authorized.
- 17. Default by the Granters and the Remedies of the Grantee. Any of the following events shall be an Event of Default (each such event being herε n called an "Event of Default") under this Conveyance:
 - (a) any representation or warranty contained in Peragraph 20 of this Cenveyance or in any certificate or statement delivered by the Grantors (or any of their officers) pursuant to subparagraph (b) of Paragraph 7 of this Conveyance shall be untrue when made in any respect which adversely affects the rate of discharge of the Production Payment or otherwise materially adversely affects the interests of the Grantee hereunder; or
 - (b) any Granter shall default in the performance or observance of any term, covenant or condition on such Granter's part to be performed or observed hereunder which default adversely affects the rate of discharge of the Production Payment or

otherwise adversely affects the interests of the Grantee hereunder, and such default shall continue unremedied for a period of 30 and such derinit shall continue differential to a period of sold days after written notice shall have been given to such Grantor by the Grantee's demanding the performance or observance of the relevant term, covenant or condition (or, if such default is capable relevant term, covenant or condition for, it such derault is capable of being remedied by such Grantor but cannot be remedied within or being remedica by Such Grantor our cannot be remedica within such 30-day period, such Grantor shall not have commenced to such ou-may period, such transfer such not may commenced to remedy such default within such period after receipt of such remeny such derium within such period and; Teseipe of such notice and thereafter shall not continuously and diligently pursue the remedying of the same until remedied); or

- (e) any Grantor shall become insolvent, or be adjudicated a for the benefit of its is as such debts become creators, or snau generally not pay its debts generally due, or admit in writing its inacidity to pay its debts generally bankrupt, or make a general assignment as they mature, or voluntarily commence any proceeding or file any ereditors, or shall generally not pay its petition under any bankruptey, insolveney or similar law or pertion ander any manacopies, insorveing or summa naw of seeking dissolution or reorganization or the appointment of a receiver, trustee or liquidator for it or for a substantial portion of receiver, trustee or administration for it or to a substantial portion of the its assets, or file any answer admitting the jurisdiction of the as assets, or me any answer admiring the introduction of the court and the material allegations of an involuntary potition fied against it under any bankruptey, insolveney or similar law; or
 - (d) an involuntary proceeding shall be commenced or an involuntary petition shall be filed against any Grantor under any involuntary periron shan or med egainst any grantor under any bankruptey, insolvency or similar law or seeking the dissolution or pankrapicy, insurency or summar my or seeking one dissolution of receiver, reorganization of such Grantor or the appointment of a receiver, reorganization of Such Mitantor of the appointment of a substantial part of the trustee or liquidator for such Grantor or a substantial part of the property of such Granter and such involuntary proceeding or property or such virance; and such involuntary proceeding or involuntary petition shall not be dismissed within 60 days after

the commencement or filing thereof, as the case may be.

Upon the occurrence and during the continuance of any Event of Default, the Grantee may, in addition to pursuing any or all other a lable at law or in equity, do any of the following

freet performaree or observance of any such term, covendition referred to in subparagraph (b) above on behalf the expense of the Grantors, in which event the Grantee expend funds reasonably necessary to effect such purposes TOT, IC

and shall be entitled to reimbursement therefor from the Grantors, together with interest on such amounts at a rate equal to the lesser of (x) the maximum rate of interest that may be lawfully charged on such amounts under applicable law or (y) the Applicable Rate in effect during the period from and including the date of such expenditure to the date of reimbursement; or

- (ii) upon written notice to the Grantors, succeed to all rights of the Grantors with respect to the possession, operation and development of the Subject Interests, and may use in connection therewith all of the appurtenant property, both real and personal, situated upon the Subject Interests (or any portion thereof) or used or held for future use in connection therewith, or in connection with the production storing, transporting or sale of the Subject Minerals, and the Grantee shall have the right, on behalf of and for the account of the Granters, to sell the Subject Minerals, and the Grantors shall, upon demand, reimburse the Grantee for all amounts so expended by the Grantee, together with interest on such amounts at a rate equal to the Applicable Rate in effect during the period from and including the date of such expenditure to the date of reimbursement; or
- (iii) apply to a court of equity for the specific performance or observance of any such term, covenant or condition referred to in subparagraph (b) above or in aid of the execution of any power herein granted and for the appointment of a receiver of the Subject Interests and the Subject Minerals.
- 18. Exercise and Termination of the Remedies of the Grantee. All rights and remedies granted by the provisions of Paragraph 17 of this Conveyance and otherwise belonging to the Grantee shall be cumulative, and no exercise of any right or remedy shall preclude the exercise of any other right or remedy. Any right or remedy may be exercised on one or more occasions without exhaustion thereof. All such rights and remedies shall terminate:
 - (a) when the Production Payment is discharged and all amounts then due and payable to the Grantee pursuant to the terms of Paragraph 17 of this Corveyance (including amounts payable on account of interest) shall have been paid in full, or

- (b) at any earlier time when all defaults of the Grantors shall have been remedied and all such amounts shall have been duly paid in full, without prejudice, however, to the exercise of any rights conferred upon the Grantee upon any subsequent failure of the Grantors to perform or observe any of the terms, covenants or conditions herein provided to be performed or observed by the Grantors.
- 19. Reports to the Grantee. While the Production Payment remains in force and effect, the Grantors will, at their expense, furnish to the Grantee:
 - (a) Monthly, on or before the 25th day of each Accounting Month, or if such information is not available to the Grantors by such date, within five days after the required information is received by the Grantors, a report, in form and substance satisfactory to the Grantee, showing (i) the gross quantity of oil, condensate, plant products and gas and, if the Supplemental Instrument of Conveyance of Production Payment has been executed and delivered, of sulphur, included in the Subject Minerals sold by the Grantors, or taken by the Grantors for their own use, during the preceding Accounting Month, (ii) the net quantity of oil, condensate, plant products and gas (after deduction of oil and gas used in connection with the operation of the Subject Interests) included in the Subject Minerals produced and sold by the Grantors, or taken by the Grantors for their own use, during such preceding Accounting Month, (iii) the gross proceeds received from such sale, or attributable to such taking, of Subject Minerals (to the extent of the Grantors' net revenue interest therein) during such preceding Accounting Month, separately stated as to oil, condensate, plant products, gas and, if the Supplemental Lastrument of Conveyance of Production Payment has been excuted and delivered, as to sulphur, (iv) the amount of all costs incurred by the Grantors in transporting Subject Minerals so sold to the point of sale if sale is not made at the wellhead or mine, (v) the amount of all taxes and assessments attributable to the production and sale of such Subject Minerals, (vi) that portion of the foregoing quantities, proceeds, transportation costs, taxes and assessments allocable to the Production Payment Minerals and (vii) all

lirect costs and operating costs incurred by the Grantors in proucing and processing Subject Minerals.

- (b) Annually, within 90 days after the end of each fiscal year of the Grantors, a reserve report of Ryder Scott Company Petroleum Eng. ears (or other independent petroleum engineers satisfactory to the Grantee and the lenders to the Grantee), substantially in the form and containing the same information (as of the end of such tiscal year) and using the same defluitions and assumptions with respect to the Hydrocarbons recoverable from proved reserves attributable to the Subject Interests, as the letter report of Ryder Scott, dated June 29, 1981, delivered to the Grantee and the lenders to the Grantee prior to the date hereof.
- (c) Such other information and reports (including financial statements of the Crantors and or reports in respect of sulphur) as the Grantee may from time to time reasonably request in writing.
- 20. Representations, Warranties and Covenants. (a) The Grantors jointly and severally represent and warrant as follows:
 - (i) the Grantors have good and marketable title to the Subject Interests, which are presently valid, subsisting and in full force and effect, subject only to this Conveyance and the exceptions, restrictions, reservations, liens, charges, encumbrances and other matters, if any, as are set forth in Exhibit A relative thereto, and such Sales Contracts operation agreements, unitization and pooling agreements and other agreements as are customarily found in connection with comparable drilling, mining and producing operations, none of which, singly or in the aggregate, materially adversely affect the Production Payment Monerels or the proceeds thereof, or the rights of the Grantee therein or adversely affects the rate of discharge of the Production Payment;
 - (ii) the Grantors' interest in production of Hydrocarbons produced from each Subject Interest specifically described in Exhibit A hereto (after deducting all applicable royalties, overriding royalties, production payments and other payments out of

production not owned by the Grantors and before giving effect to the Production Payment hereby conveyed to the Grantee) is at the date hereof no less than that interest shown under the here, ing "Net Revenue Interest" opposite the descriptor of each such Subject Interest:

- (iii) the Grantors have the legal right, power and a shorit; to create and confer upon the Grantee the rights of the Grantee granted by the terms of this Conveyance;
- (iv) the Production Payment is, and at the time of proper recording of this Conveyance in each place as may be required by law in order fully to preserve and protect the interest of the Grantee in the Production Payment will be, free and clear of all liens, charges and encumbrances of every character, except the liens, charges and encumbrances permitted by Paragraph 14(f) hereof and except such liens, charges and encumbrances as may be vested in persons claiming the Production Payment or any interest therein by, through or under the Grantee;
- (v) all taxes and assessments of any kind whatsoever (other than those contested by the Grantor in good faith) imposed or assessed with respect to or measured by or charged against or attributable to the Subject Interests have been duly paid or provided for:
- (vi) there are no suits or proceedings pending or, to the knowledge of the Grantors, threatened against or affecting the Subject Interests, or any part thereof, or the sale of Subject Minerals produced therefrom, before any court or by or before any governmental commission, bureau or any regulatory authority, which if decided adversely to the interests of the Grantors would materially adversely affect the Production Payment, or the Production Payment Minerals or the proceeds thereof or rights of the Grantee therein or would adversely affect the rate of discharge of the Production Payment;
- (vii) to Grantors' knowledge, all rents and royalties that have become due prior to the Effective Date with respect to the Subject Interests or production therefrom, and all liabilities of any kind or nature incurred with respect to the Subject Interests prior

to such date, have been paid or have been provided for, and no Grantor has received any: dice of default or claimed default with respect to the Subject Interests or any part thereof or any interest in production therefrom; and

- (viii) all prices received by Grantors under presently existing Sales Contracts are collected in accordance with all existing rules, regulations, and orders of the Federal Energy Regulatory Commission, or its predecessor, establishing maximum lawful prices for the sale of natural gas.
- (b) The Grantors hereby jointly and severally bind themselves, their successors and assigns, to warrant and forever defend their title to the Subject Interests and the Grance's title to the Production Payment against the lawful claims and demands of every person who shall claim the same or any part thereof, and the Grantors will indemnify and hold the Grantee harmless from and against all loss suffered by the Grantee resulting form any failure of or defect in the title to the Subject Interests or the Production Payment as warranted above in this Paragraph 20. This Conveyance is made with full substitution and subrogation of the Grantee in and to all covenants and warranties by others heretofore given or made in respect of the Subject Interests or any part thereof.
- (c) The Grantors jointly and severally represent and warrant as follows:
 - (i) Each of the Grantors is a corporation duly organized and validly existing in good standing under the laws of the State of Delaware, is luly qualified and in good standars as a foreign corporation in each State in which such Grantor owns Subject Interests and has the legal right, power and authority, corporate or otherwise, to conduct its business and to own its properties (including the Subject Interests) and to convey to the Grantee the Production Payment and all of the Lights, titles, interests, estates, remedies, powers and privileges appurtenant or incident thereto in the manner hereby contemplated, and to execute and deliver, and to perform all of its obligations under, this Cenveyance.
 - (ii) The making and performance by the Grantors of this Conveyance, and the conveying by the Grantors to the Grantee of

the Production Payment, have been duly authorized by all necessary corporate action (including any shareholder approval) on the part of each Grantor and do not and will not (A) violate any provision of any law, rule, regulation, order, writ, judgment, decree or determination presently in effect having applicability to any Grantor or of the Certificate of Incorporation or By-laws of any Grantor, or (B) result in a breach of or constitute a default under any indenture or bank loan or credit agreement o, any other agreement or instrument to which any Grantor is a party or by which it or its properties may be presently bound or affected or (C) result in or require the creation or imposition of any mortgage, lien, pledge, security interest, charge or other encumbrance on, upon or of any of the properties or assets of any Grantor (including the Subject Interests) under any such indenture, bank loan or credit agreement or other agreement or instrument; and no Grantor is in default under any such order, writ, judgment, decree, determination, indenture, agreement or instrument in any way which materially adversely affects, or may materially adversely affect, the ability of the Grantors to perform their obligations under this Conveyance; and all consents or approvals under such indentures, agreements and instruments necessary to permit the valid execution, delivery and performance by the Grantors of this Conveyance and the conveying of the Production Payment by the Grantors to the Grantee have been obtained

- (iii) Assuming the due authorization and execution thereof on the part of the Grantee, this Conveyance constitutes the legal, valid and binding obligation of the Grantors enforceable against the Grantors in accordance with its terms, except as limited by bankruptey, insolvency or other similar laws of general application relating to or affecting the erforcement of creditors' rights.
- (iv) All consents and waivers necessary to permit the valid conveying of the Production Payment to the Grantee have been obtained or the time for giving such consents or waivers has expired following a written request therefor given by the Grantor.
- (v) No authorization, consent, approval, license or exemption of, and no filing or registration with, any court or governmental

department, commission, board, bureau, agency or instrumentality, domestic or foreign, is or was recessary to the valid execution and delivery by the Grantors of this Conveyance or to the conveying by the Grantors to the Grantee of the Production Payment or to the performance by the Grantors of their obligations under this Conveyance (other than, in the State of Louisiana, with regard to leases from the State of Louisiana or political subdivisions thereof, certain governmental approvals which are not obtainable prior to the execution and delivery hereof and which are normally routine and will be obtained in due course) and in the case of the performance by the Grantors of such obligations, such authorizations, consents, approvals, licenses or exemptions which are routine and either have been obtained or will be brained in due course.

- (vi) No fire, explosion, accident, earthquake, act of the public enemy or other casualty (whether or not covered by insurance) materially adversely affecting any material portion of the Subject Interests or the operation thereof, or materially adversely affecting the ability of the Grantors to perform their obligations under this Conveyance, has occurred.
- (vii) The offices where the Grantors keep their respective records concerning the Subject Interests and the Production Payment and the payments to be made to the Grantors in respect of production from the Subject I cross and cospect of the Production Payment are located in the State of coxas. The Grantors hereby agree that such records will be kept in such State or, after 60 days' prior written notice to the Grantee in respect thereof, in any other State or States of the United States.
- (viii) No information, exhibit, memorandum or report (excluding estimates contained therein) furnished in writing by the Grantors to the Grantee in connection with the sale of the Production Payment, or by the Grantors to Ryder Scott Company Petroleum Engineers or DeGolyer and MacNaughton in connection with their evaluation of the Subject Interests contained any material misstatement of fact or omitted to state a material fact relative to the subject matter thereof or any fact necessary to make the statements contained therein not misleading. All estimates so furnished were prepared on the basis of assumptions, data, information,

tests or conditions believed to be valid or accurate or to exist at the time such material was so furnished.

- (ix) All of the properties and interest with respect to which value was given in the financial and engineering reports submitted by Ryder Scott Company Petroleum Engineers and the Grantors to the Grantee and its lenders in connection with the negotiations for the purchase and financing of the Production Payment are Subject Interests (or, if less than all, the net effect of such difference is negligible).
- (x) Since March 31, 1981 there has not been such a significant adverse change in the financial condition of the Grantors, viewed as a whole, as to materially impair their ability to perform the obligations of the Grantors set forth in this Conveyance.

The covenants, representations and warranties contained in this Paragraph 20 shall survive the discharge of the Production Payment.

- 21. Transfer of the Zubject Interests. Except to the extent required by the terms of the relevant operating agreement, unit operating agreement, contract for development or other instrument in connection with nonconsent operations conducted with respect to any Subject Interest (or any portion thereof) as to which any Granter's election not to participate shall have been made in conformity with the provisions of Paragraph 14(a) of this Conveyance, and except as otherwise permitted by Paragraphs 14 and 16 of this Conveyance, the Grantors shall not, prior to the termination of the Production Payment, without the prior written consent of the Grantee, which consent shall not be unreasonably withheld, make or permit (i) any surrender, abandonment, sale, conveyance, assignment, lease, sublease or farmout, in whole or in part, of any of the Subject Interests insofar as they include or cover Hydrocarbons, or (ii) after the execution and delivery of the Supplemental Instrument of Conveyance of Production Payment, any surrender, abandonment, sale, conveyance, assignment, lease or sublease, in whole or in part, of any of the Subject Interests insofar as they include or cover sulphur.
- 22. Exception to Payments. The acceptance by the Grantee of any payment hereunder shall not preclude the Grantee from taking excep-

tion to the correctness of the amount tuereof; provided, however, that any such exception must be specified by the Grantee in a notice given to the Grantors within 12 months after the date such payment is made.

- 23. Transfer of the Production Payment. Nothing contained in this Conveyance shall be construed to limit or restrict in any way the right of the Grantee to sell, convey, assign or mortgage the Production Payment, in whole or in part.
- 24. Rights of Mortgagee or Trustee. If the Grantee shall at any time execute a mortgage or deed of trust covering all or part of the Production Payment, the mortgagee(s) or trustee(s) therein named or the holder of any obligation secured thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to exercise all the rights, remedies, powers and privileges conferred upon the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained hereunder by the Grantors, but the provisions of this Paragraph 24 shall in no way be deemed or construed to impose upon the Grantors any obligation or liability undertaken by the Grantee under such mortgage or deed of trust or under the obligation secured thereby.
- 25. Grantors' Rights to Make Payments to be Applied to Discharge of Production Payment. Anything in this Conveyance to the contrary notwithstanding, the Grantors shall have the right, at any time and from time to time, upon written notice furnished to the Grantee and each of the mortgagees and trustees referred to in Paragraph 24 hereof not less than 15 days before the Application Date on which the Grantors exercise such right, to make payment (out of any funds of the Grantors whether or not such funds are proceeds from the Subject Minerals) to the Grantee on any Application Date, which payments shall be deemed to be proceeds of Production Payment Minerals for purposes of this Conveyance and shall be applied to the discharge of the Production Payment pursuant to the provisions of Paragraph 6 hereof, provided, however, that the Grantee has been provided with the indemnity specified in Paragraph 5. Any such payment shall be in an amount sufficient to reduce the Primary Sum by \$50,000,000, or a multiple thereof (unless the full aggregate sum of the amount specified in subparagraphs (a), (b), (c), (d), (e) and (f) of Paragraph 4 hereof, after application of

the proceeds of Production Payment Minerals on such Application Date, is less than \$50,000,000, in which event such payment shall be an amount equal to such full aggregate sum), and shall be made by a deposit to the eredit of Grantee's account (No. 39019753) with Citibank at 399 Park Avenne, New York, New York 10043, in lawful currency of the United States of America and in immediately available funds. No payment made by the Grantors pursuant to this Paragraph 25 shall be added to, or have the effect of increasing, the Purchase Price.

- 26. Notices. Any notice, request, demand, report or einer instrumont which may be required or permitted to be given or furnished to or served upon either party hereto or other Person succeeding to any interest of either party hereto shall be deemed sufficiently given or furnished and served if in writing and delivered to such party or Person or to an officer of such party or Person or deposited in the United States mail in a scaled envelope, registered, with postage prepaid, addressed if to the Grantors to Pennzoii Company, Pennzoil Place, P.O. Box 2967, Houston, Texas 77002, Attention: Harold E. Sortor, and if to the Grantee to 100 West Tenth Street, Wilmington, Delaware 19801, Attention: Joseph A. Barbera, or to such other address as the party or Person to be addressed shall have designated by writien notice to the party or Person giving such notice or furnishing such report or making such request or demand.
- 27. Further Assurances. So long as permitted by applicable law so to do, the Grantors will execute and deliver all such other and additional instruments, notices, releases, acquittances and other documents, and will do all such other acts and things, as may be necessary more fully to assure to the Grantee all the rights and interests herein and hereby granted or intended so to be.
- 28. Binding Effect. All the covenants and agreements of the Grantors contained in this Conveyance shall be deemed to be covenants running with the land and the Subject Interests and shall be binding upon all successors in interest to and all assigns of, the Grantors and shall inure to the benefit of all successors in interest to, and all assigns of, the Grantee. All references herein to either the Grantors or the Grantee shall include their respective successors and assigns, except that the term "Grantee" as used in this Conveyance shall not

include any collection agent of any mortgagee or assignee for security purposes of the Grantee appointed by such mortgagee or assignee without the consent of the Grantors.

- 29. Release or Failure of Title. No release by the Grantee of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect of reducing, or increasing the amount of production of Subject Minerals out of which the Production Payment is dischargeable or reducing the amount of the Production Payment or creating any offset or other prejudice to the Production Payment, and the Production Payment shall continue in full force and effect as to the remainder of the Subject Interests until the full amount of the Production Payment has been received by the Grantee as herein provided.
- 30. Governing Law. The validity, effect and construction of this Conveyance insofar as it relates to any conveyance or other disposition affecting any Subject Interest, any warranties of title relating thereto and the remedies herein provided for or provided for by law with respect to such Subject Interest shall be governed by the laws of the jurisdiction in which such Subject Interest is located. With respect to all other matters (including, without limitation, matters of construction, validity and performance other than as set forth above in this Paragraph 30), this Conveyance shall be governed by and construed in accordance with the laws of the State of Texas.
- 31. Final Termination. Notwithstanding the other provisions hereof, the Production Payment shall, in any event, terminate one day prior to the expiration of 21 years after the death of the survivor of all the descendants of Joseph P. Kennedy, father of the late President of the United States, who are living on the Effective Date.
- 32. Ecadings. Paragraph headings used in this Conveyance are for convenience only and shall not affect the construction of this Conveyance.

In Witness Whereof, the Grantors and the Grantee have executed this instrument in the presence of the undersigned witnesses on the date set forth in the acknowledgments and sed hereto, in several counterparts (one of which with all of the property descriptions included in Exhibit A is on file at the effice of the Grantee referred to in Paragraph 26 hereof, one of which with all said property descriptions is to be recorded in Terrebonne Parish, Louisiana, and one of which with all said property descriptions is to be recorded in Culberson County. Texas) each of which is an original and all of which are identical, except that, to facilitate recordation, there are omitted from certain counterparts those property descriptions in Exhibit A which contain descriptions of property located in recording jurisdictions other than the jurisdiction (County or Parish) in which the particular counterpart is to be recorded. Each of the counterparts of this Conveyance so executed shall for all purposes be deemed to be an original, and all such counterparts together shall constitute but one and the same Conveyance.

> PENNZOIL COMPANY PENNZOIL PRODUCING COMPANY PENNZCIL OIL & GAS, INC. DUVAL CORPORATION

By Group Vice President, Pennzoll Company

[CORPORATE SEAL]

Vice President. Pennzoil Producing Company

[CORPURATE SEAL]

Vice President. Pennzoil Oil & Gas, Inc.

[CORPORATE SEAL]

Vice President, **Duval Corporation**

[CORPORATE SEAL]

| Attest: | |
|---|--|
| SALLY HAZEN, Secretary, Pennzoil (Secretary, Pennzoil (Secretary, Pennzoil (Assistant Secretary, | Producing Company Dil & Gas, Inc. |
| | Grantors |
| | The address of Pennzoil Com- pany, Pennzoil Producing Com- pany, Pennzoil Oil & Gas, Inc. and Duval Corporation is: |
| | c/o Pennzoil Company Pennzoil Place P.O. Box 2967 Houston, Texas 77002 |
| [CORPORATE SEAL] | New Dunstan, Inc. |
| Attest: | By Vice President Grantee |
| | |
| | The address of New Dunstan, Inc. is: |
| | 100 West Tenth Street Wilmington, Delaware 19801 |
| | *************************************** |
| Andrew M. Baker, Assistant Secretary | |
| Executed by all parties in the presence of the undersigned competent witnesses: | |
| Witness | |
| Witness | |

STATE OF NEW YORK Sounty OF NEW YORK St.

BE IT REMEMBERED, that I, Douglas A, Stuart, a Notary Public duly qualified, commissioned, sworn and acting in and for the County and State aforesaid, hereby certify that on this 16th day of July, 1981:

There appeared before me severally each of the following persons, each residing at the address set forth below his name, and each the designated officer of the corporations set opposite his name, each such corporation being a corporation of the state indicated and being a party to the foregoing instrument:

Harold E. Sortor 910 Oak Valley Houston, Texas 77024 Group Vice President of Pennzoil Company, a Delaware corporation.

Vice President of Penrzoil Producing Company, a Delaware corporation.

Vice President of Pennzoil Oil & Gas, Inc., a Delaware corporation.

Vice President of Duval Corporation, a Delaware corporation.

Joseph Nalle 3024 Chevy Chase Houston Texas Vice President of New Dunstan, Inc., a Delaware corporation. (Alabama)

Before me personally appeared each such person, whose the are signed to the foregoing conveyance, and who are known to me, and acknowledged before me on this day that, being informed of the contents of the conveyance, each, as such officer of each of the corporations set opposite his name and with full authority, executed the same voluntarily for and as the act of said corporation.

(Arkansas)

Before me appeared each such person, to me personally well known, and each stated that he is the designated officer of each of the corporations set opposite his name, and he is duly authorized in his capacity to execute the foregoing instrument for and in the name of said corporation, and further stated and acknowledged that he had so signed, executed and delivered said foregoing instrument for the consideration, uses and purposes therein mentioned and set forth.

(Louisiana)

Before me appeared each such person, to me personally known, who being by me duly sworn, did say that he is the designated officer of each of the corporations set opposite his name, and that the seal of such corporation affixed to the foregoing instrument is the corporate seal of the corporation and that the instrument was signed and sealed in behalf of the corporation by authority of the Board of Directors of such corporation and that each acknowledged the instrument to be the free act and deed of the corporation.

(Mississippi)

Before me personally appeared each such person and acknowledged that he signed and delivered the foregoing instrument on the day and year therein mentioned as the act and deed of each of the corporations set opposite his name, having first been authorized to do so.

(Montana)

Enforce me personally appeared each such person, known to me to be the designated officer of each of the corporations set opposite his name that executed the within instrument and acknowledged to me that such corporation executed the same.

(New Mexico)

Before me on this date appeared each such person, to me personally known, each of whom, being by me duly sworn, did say that he is the designated officer of each of the corporations set opposite his name, and that the seal affixed to said instrument was signed and sealed in behalf of said corporation by authority of its Board of Directors, and each such person acknowledged said instrument to be the free act and deed of said corporation.

(New York)

Before ms personally came each such person, to me personally known, and each, being by me duly sworn, did depose and say that he resides in the address set forth below his name and is the designated officer of an hoof the corporations set opposite his name and that said corporation is described in and executed are across instrument; that he knows the soil of the corporation; that the seal affixed to such instrument is such corporate seal; that it was so affixed in order of the Board of Directors of the corporation; and that he is a seal and deed of the corporation.

('iorth Dakota)

Before me on this date personally appeare each such person, known to me to be the designated officer of each of the corporations set opposite his name, which corporation is described in and executed the within instrument, and each acknowledged to me that such corporation executed the same.

(Penusylvania)

Before me personally appeared each such person and accrowledged himself to be the designated officer of each of the contractions set opposite his name, and that he, as such officer, being authorized so to do, executed the foregoing instrument for the purposes therein contained, by signing the name of the corporation by himself as such officer.

(Texas)

Before me on this day, are nally appeared each such person, each of whom is known to me to be the person whose name to subscribed to the foregoing instrument, and known to me to be the designated officer of each of the responsitions set opposite his name, and each acknowledge, to me that he executed said instrument for the purposes and consideration therein expressed, in the capacity therein stated and as the act and deed of said corporation.

(West Virginia)

Each such person has this day in my said County, before me, acknowled, and the said instrument to be the act and deed of each of the corporations set opposite his name.

(Wyoming)

Before me on this day appeared each such person to me personally known, each of whom, being by me duly sworn, did say that he is the designated officer of each of the corporations set oppose a his name, and that the seal affixed to said nin-trument is the corporate seal of said corporation and that said instrument was signed and sealed in schalf of said corporation by authority of its Board of Directors, and each such person acknowledged it to be the free act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and official seal, in the City, County and State of New York, this 16th day of July, 1981.

Notary Publ.

Notary Public St. te of New York No. 4729635 Qualified in the v York County Commission Expires March 30, 1983

My Commission expires Myrch 30, 1983.

EXHIBIT A

to

Justrument of Conveyance of Production Payment

ATTACHED TO AND FORMING A PART OF THAT CERTAIN INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT DATED JULY 16, 1981 BETWEEN PENNZOIL COMPANY, ET AL. AND NEW DUNSTAN, INC.

This Exhibit A contains the specific descriptions of the Subject Interests which are referred to in Paragraph 1 of this Instrument of Corveyance of Production Payment as being specifically described or referred to in this Exhibit A.

PART ONE of this Exhibit A sets forth those Subject Interests tocated or shore Texas and Leuisiana, and PART TWO of this Vehibit A sets forth those Subject Interests located in the States of Alaman, Arkansa , Louisiana, Mississippi, Montana, New Mexico, Tew York, North Dakota, Pennsylvania, Texas, West Virginia and Wysming.

The Subject Interests are expressly limited to the Lauses set forth in this Euribit A insofar and only insofar as they cover lands and lepth intervals in vaich one or more of the Grantors own or hold an interest and do not include lands and depth intervals in which none of the Grantors own a latterest, even though such lands or depth intervals are covered by the Lause set forth in this Exact Set A.

PART ONE of the Fixhibit A refers to the "BLM or State Lease No." for each of the Lease, listed therein. Those Leases so listed having a number with the prefer "OCS G" are these sissued by the United States of A. or on as Lessor through the burnau of Land Management (the "BLA", of the United States Deportment of the Interior. Those Leases listed in "art One of this Exhibit A having a number with the prefix "St. Ls." are Leases issued by the State of Louisiana 5.1 issor.

"the case of each Subject Interest set with in PART ONE of this Exhibit A, there is shown under the heading "Net Revenue interest" Pennzoil Oil & Gas, Inc's (under the sub-heading "POGI") and Pennzoil Producing Company's (under the sub-heading "PPC") respective interest (expressed in perce lages) in total production of Hydrocarbons produced and saved from and that accrues or is attributable to the relevant Subject Interest, after deducting all applicable royalties, overriding royalties, production payments and other payments out of production not owned by Pennzoil Oil & Gas. Inc. or Pennzoil Producing Company, as the case may be.

In PART TVO of this Exhibit A, there is shown opposite (i) the description of each Subject Interest, or (ii) the well leasted upon or from which production is attributable to such Subject Interest or (iii) the unit which contains such Subject Interest or a portion the reof, under the heading "Net Revenue Interest" the aggregate interest of Grantors (expressed in decimals) in total production of Hydrocarbons produced and saved from and that accrues or is attributable to the relevant Subject Interest, well or unit, as the case may be, after deducting all applicable royalties, overriding royalties, production payments and other payments out of production not owned by the Grantors or any of them. Where the Net Revenue Interest is listed opposite a well or unit, the relevant Subject Interests are listed below such well or unit.

The Leases set forth in FART TWO of this Exhibit A may be committed to more than one pool or unit, or may cover lands situated in more than one county or parish; consequently, such Leases may be listed more than once in this Exhibit A. Such repetition is not intended to ner shall it be construed as enlarging or otherwise affecting the Subject Interest or Subject Interests which appertain to such Lease.

All recording references in PART TWO of this Exhibit A are to the offic a regords in the office of the county or parish clerk or recorder of the counties or parishes in which the relevant Subject Interests are situated.

On the pages of PART TWO of this Exhibit A relating to lands located in various Parishes within the State of Louisiana, the State of Louisiana is identified in the first line of the upper left corner of each of said page by the letters "LA" and the respective Parishes are identified in the second line of the upper left corner of each of said pages. The column appearing on each of said pages under the heading "Bounding Book Lage" or "Recorded Book-Page" sets forth recording references to the public records of the particular Parish concrete. The "Book" referred to is the book maintained by each Parish as a part of its official Conveyance Records, including the Conveyance

Book or any book which the Parish may maintain for the recordation of documents evidencing oil, gas and mineral transactions. In some instances, the recording reference set forth may also include (or consist solely of) the entry number, file number, original act number, or registry number, etc. under which a particular document is filed.

In the description of the Suoject Interests for the states of New York, Pennsylvania and West Virginia, the Net Revenue Interest of Grantors is bived opposite a "Farm" number and name. The Farm designation denotes the interest of the Grantors with reference to a single and unique royalty interest originating from one grantor or lessor (either leasehold a, fee). All of the Leases to which the Farm designations relate are described in the pages immediately collowing the listing of Farms, by county, and the description of such Leases, standing alone, constitutes a complete description of all of the Leases intended to be covered by this Conveyance.

PART ONE

OFFSHORE SUBJECT INTERESTS

TEXAS

| BLM or State | Effective | | Net Rever | nue Interest |
|--------------|-----------|---|-----------|--------------|
| Lease No. | Date | Description | POGI | PPC |
| OCS-G-2398 | 8- 1-73 | All of Block A-273, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.12639 | 11.24999 |
| OCS-G-2403 | 8- 1-73 | All of Block A-279, High Island Area, East Addition, South Extension Co'S Official Leasing Map, Texas Map No. 7C | 1.00124 | 14.16666 |
| OCS-G-2414 | 8- 1-73 | All of Block A-323, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 11.66662 |
| OCS-G-2416 | 8- 1-73 | All of Block A-325, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 7.22016 | 57.14286 |
| OCS-G-2418 | 8- 1-73 | All of Block A-327, High Island Area, East Addition, South Extension OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 8.33333 |
| OCS-G-2422 | 8- 1-73 | All of Block A-332, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 8.33333 |
| OCS-G-2739 | 7- 1-74 | All of Block A-339, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 28.33332 |
| OCS-G-2426 | 8- 1-73 | All of Block A-340, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 1.00124 | 28.33332 |
| OCS-G-2429 | 8- 1-73 | All of Block A-351, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 2.88528 | 36,62186 |
| OCS-G-2745 | 7- 1-74 | All of Block A-355, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 7.34246 | 9.44166 |

| BLM or State Lease No. | Effective Date | Description | Net Rever | PPC |
|---------------------------|-------------------|--|-----------|----------|
| OCS-G-2746 | 7- 1-74 | All of Block A-356, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C | 10.01245 | 12.49999 |
| OCS-G-2366 | 8- 1-73 | All of Block A-474, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 9.99999 |
| OCS-G-2367 | 8- 1-73 | S ½ of NE ¼ & SE ¼ of Block A-475, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.001241 | 9.999996 |
| OCS-G-2372 | 8- 1-73 | All of Block A-489, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.49999 |
| OCS-G-3118 | 4- 1-75 | All of Block A-499, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 1.10137 | 13.66667 |
| OCS-G-2378 | 8- 1-73 | All of Block A-520, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.60.24 | 9.99999 |
| OCS-G-2704 | 7- 1-74 | All of Block A-545, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 6.67497 | 8.33333 |
| OCS-G-2779 | 10- 1-74 | All of Block A-546, High Island Area, South Addition, OCS Official Lessing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2705 | 7- 1-74 | All of Block A-547, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2706 | 7- 1-74 | All of Block A-548, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 8.67746 | 11.66666 |
| OCS-G-2384 | 8- 1-73 | All of Block A-555, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.00124 | 12.08332 |
| OCS-G-2388 | 8- 1-73 | All of Block A-563, High Island Area, South Addition, OCS Official Leasing Map, Texas Map, No. 7B | 1.00124 | 9.99999 |
| OCS-G-2389 | 8- 1-73 | All of Block A-564, High Island Area, South Addition, OCS Official Leasing Map, No. 7B | 1.00124 | 9.99999 |

| BLM or State | Effective | D. Later | Net Reven | |
|--------------|-----------|--|-----------|----------|
| OCS-G-2396 | 8- 1-73 | All of Block A-570, High Island Area, South Addition, OCS Official Leasing | 1.01958 | 16.97190 |
| OCS-G-2719 | 7- 1-74 | Map, Texas Map No. 7B All of Block A-582, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 1.17780 | 11.76499 |
| | | Louisiana | | |
| St. Ls. 7604 | 4-17-78 | TRACT 14836 — PORTION OF BLOCK 3 SABINE PASS AREA, Cameron Parish, Louisiana. That portion of Block 3, Sabine Pass Area, Cameron Parish, Louisiana, belong- ing to the State of Louisiana and not under a mineral lease on January 23, 1978, described as follows: Beginning at a point on the North line of Block 3, Sabine Pass Area having Lambert Plane Coordinates of X = 1,217,855 14 and Y = 364,274.14; thence East 9,612.44 feet along said North line to a point having Lambert Plane Coor- dinates of X = 1,227,467.58 and Y = 364,274.14; thence Southerly and Southwesterly along an arc having a radius of 18,240.60 feet and a vertex of X = 1,209,227 and Y = 364,245 to a point on the South line of said Block 3, having Lambert Plane Coordinates of X = 1,219,987.06 and Y = 349,516.10; thence West 2,131.92 feet along said South line to a point having Lambert Plane Coordinates of X = 1,217,855.14 and Y = 349,516.10; thence North 14,758.04 feet to the point of beginning, estimated to contain approximately 2,497.59 acres, as shown outlined in red on a plat on file in the Office of the Secretary, Department of Natural Resources, LESS AND EXCEPT that portion thereof, if any, which is more than three nautical miles from the coast line as determined by the Report of the Special Master in the litigation in the Supreme Court of the United States styled United States v. State of Louisiana et al. No. 9 Original, | | 38.125 |

| BLM or State Lease No. | Effective Date | Description | POCI | PPC |
|---------------------------|-------------------|---|----------|-----------------|
| | | said three mile line as set out in the June, 1975, decree of the Supreme Court. All bearings are based on Louisiana Lambert Plane Coordinate System (South Zone). | | |
| OCS-G-4144 | 9- 1-79 | That portion of Block 3 which is more than three geographical miles seaward from the line described in the supple- mental decree of the U.S. Supreme Court, June 16, 1975 (United States vs. Louisiana, 422 U.S. 13) Sabine Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 12. | | 41,66663 |
| OCS-G-4379 | 11- 1-80 | All of Block 53, West Cameron Area, as shown on OCS Leasing Map, Louisiana Map No. 1 | | 41.66665 |
| OCS-G-2839 | 12- 1-74 | All of Block 352, West Cameron Area, West Addition, OCS Official Leasing Map, Louisiana Map No. 1A | 8.34372 | 20.83333 |
| OCS-G-2224 | 2- 1-73 | All of Block 532, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.22499 |
| OCS-G-2225 | 2- 1-73 | All of Block 533, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 2.22609 | 22.22499 |
| OCS G-3284 | 8- 1-75 | All of Block 563, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 1B | 25.03113 | 20.83332 |
| OCS-G-2436 | 8- 1-73 | All of Block 586, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 16.68742 | |
| OCS-G-2021 | 2- 1-71 | All of Block 587, West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1 B | 22.80504 | |
| OCS-G-2850 | 12- 1-74 | All of Block 609, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.55555 |
| OCS-G-2559 | 5- 1-74 | All of Block 617, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 8.89996 | 5.555 55 |

| WY 1.4 C. | | | Net Revenu | ie Interest |
|---------------------------|-------------------|---|------------|-------------|
| BLM or State Lease No. | Effective Date | Description | POCI | PPC |
| OCS-G-2860 | 12- 1-74 | All of Block 237, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 2.22499 | 5.55555 |
| OCS-G-2045 | 1- 1-71 | All of Block 270, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.68741 | |
| OCS-G-2062 | 2 1-71 | All of Block 334, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| OCS-G-2439 | 8- 1-73 | All of Block 335, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 23.36239 | |
| OCS-G-2078 | 2- 1-71 | All of Block 228, Vermilion Arca, OCS Official Leasing Map, Louisiana Map No. 3 | 17.9273 | |
| OCS-G-2882 | 12- 1-74 | All of Block 125, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5.55555 |
| OCS-G-2883 | 12- 1-74 | All of Block 127, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5.55555 |
| OCS-G-2587 | 5- 1-74 | All of Block 128, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 8.89995 | 5,55555 |
| OCS-G-0479 | 12- 1-54 | All of Block 53, Eugene Island Area as shown on Official Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisiana Offshore Operations) | | 83.33333 |
| OCS-G-0478 | 1- 1-55 | E 14, of Block 116, Eugene Island Area as shown on Oficial Leasing Map, Louisiana Map No. 4 Outer Continental Shelf Leasing Map (Louisiana Offshore Operations) | | 41.66665 |
| OCS-G-2102 | 2- 1-71 | All of Block 256, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 23.1054 | |
| OCS-G-2900 | 12- 1-74 | All of Block 261, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 5.56247 | 13.88888 |
| OCS-G-3156 | 7- 1-75 | All of Block 262. Eugene Island Area, as shown on OCS Official Leasing Map, Louisiana Map No. 4 | 5.56247 | 13.88888 |

| BLM or State | Effective | | _ | ue Interest |
|--------------|-----------|--|----------|-------------|
| Lease No. | Date | Description | POCA | PPC |
| OCS-G-2104 | 2- 1-71 | All of Block 295, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 6.76497 | |
| OCS-G-2607 | 5- 1-74 | All of Block 312, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 2.22499 | 22.22221 |
| OCS-G-2115 | 1- 1-71 | All of Block 330, Eugene Island Area, South Addition, OCS Official Leasing Map No. 4A | 16.68741 | |
| OCS-G-2317 | 2- 1-73 | All of Block 333, Eugene Island Area, South Addition, OCS Official Leasing Map No. 4A | 2.22499 | 22.22221 |
| OCS-G-3332 | 3- 1-76 | All of Block 337, Eugene Island Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | 16.68741 | |
| OCS-G-3409 | 1- 1-77 | All of Block 351, Eugene Island Area South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 4A | | 20.83333 |
| OCS-G-0823 | 5- 1-60 | All of Block 186, Ship Shoal Area, as shown on Official Leasing Map, Louis- iana Map No. 5, Outer Continental Shelf Leasing Map (Louisiana Off- shore Operations) | | 28.33332 |
| OCS-G-1525 | 7- 1-67 | All of Block 222, Ship Shoal Area, OCS Official Leasing Map, Louisiana Map No. 5 | | 15.83332 |
| OCS-G-1984 | 9- 1-70 | N 14 Block 225, Ship Shoal Area, OCS Official Leasing Map, Louisiana Map No. 5 | | 19.79165 |
| OCS-G 2177 | 11- 1-72 | All of Block 49 South Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 9 | 3.33748 | 8.33333 |
| OCS-(1-3416 | 1- 1-77 | That portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (U.S. vs. Lovisiana, 422 U.S. 13) to one foot seaward of the Third Supple- mental Decree Line (404 U.S. 388, December 20, 1971) as shown on OCS Official Leasing Map, Louisiana Map No. 9 | 6.67497 | 16.66667 |

| BLM or State | Effective | - | | nue Interest |
|--------------|-----------|--|---------|--------------|
| St. Ls. 6310 | 5-13-74 | Description Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and being more particularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of | 5,92738 | 14.S0 |
| OCS-G-2185 | 10- 1-72 | Plaquemines Parish, Louisiana Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court entered December 13, 1965 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288), South Pass Area, South and East Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 9A | 5.00622 | 12.49999 |
| OCS-G-3195 | 7- 1-75 | Block 72, Main Pass Area, that portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 7.54441 | 18.83758 |
| OCS-G-3417 | 1- 1-77 | That portion of Main Pass Blocks 72 and 74 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975 (U.S. vs Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971) as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 7.54441 | 18.83758 |
| (28-G-2947 | 12- 1-74 | All of Block 73, Main Pass Area, OCS Official Leasing Map, Louisiana Map No. 10. That portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)) | 7.54441 | 18.83758 |

| BLM or State | Effective | | Net Rever | .e Interest |
|--------------|-----------|--|-----------|-------------|
| Lease No. | Date | Description | POCI | PPC |
| OCS-G | 10- 1-72 | All of Block 140, Main Pass Area, as shown on OCS Official Leasing Map, Louisiana Map No. 10 | 3.33748 | 8.33333 |
| OCS-G-3206 | 7- 1-75 | All of Block N663E63, Mobile South No. 2, as shown on OCS Official Leasing Map, NH-16-10 (Also known as Block 63), Mississippi Canyon Area, as shown on OCS Official Protraction Diagram NH-16-10 | 3.33748 | 8.33333 |
| OCS-G-3500 | 8- 1-77 | All of Block 253, West Camero 1 Area, as shown on OCS Official Leasing Map, Louisiana Map No. 1 | 10.01245 | 25.0000 |
| OCS-G-3785 | 6- 1-78 | All of Block 372, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A | 12.4999 | 12.4959 |

PART TWO

STATES OF ALABAMA, ARKANSAS, LOUISIANA, MONTANA, MISSISSIPPI, NEW MENICO, NORTH DAKOTA, TENAS, WYOMING, NEW YORK, PENNSYLVANIA AND WEST VIRGINIA

EXHIBIT B to Instrument of Conveyance of Production Payment

DISCHARGE SCHEDULE

| Regular Application Date | Hydrocarbons Monthly Percentage | Minerals Monthly Percentage |
|--------------------------------|---------------------------------------|-----------------------------------|
| October 1, 1983 | 1.2011% | 0.9320% |
| November 1, 1983 | 1.201177 | 0.9320 % |
| December 1, 1983 | 1.2011% | 0.9320% |
| January 1, 1984 | 1.2011% | 0.9320% |
| February 1, 1984 | 1.2279% | 0.9800% |
| March 1, 1984 | 1.227972 | 0.9800% |
| April 1, 1984 | 1.227977 | 0.9800% |
| May 1, 1984 | 1.227967 | 0.9800% |
| June 1, 1984 | 1.227957 | 0.9800% |
| July 1, 1984 | 1.227977 | 0.9800% |
| August 1, 1984 | 1.227967 | 0.9800% |
| September 1, 1984 | 1.2275% | 9,9800% |
| October 1, 1984 | 1.227957 | 0.980006 |
| November 1, 1984 | 1.227958 | 0.9800% |
| December 1, 1984 | 1.227967 | 0.980056 |
| January 1, 1985 | 1.2279% | 0.9800% |
| February 1, 1985 | 1.270857 | 1.0600% |
| March 1, 1985 | 1.2708% | 1.0600% |
| April 1, 1985 | 1.2708% | 1.0600% |
| May 1, 1985 | 1.2708% | 1,0600% |
| June 1, 1985 | 1.2708% | 1.0600% |
| July 1, 1985 | 1.270877 | 1.0600% |
| August 1, 1985 | 1.2708% | 1.0600% |
| September 1, 1985 | 1.270857 | 1.0600% |
| October 1, 1985 | 1.270877 | 1.0600% |
| November 1, 1985 | 1.270867 | 1.0600% |
| December 1, 1985 | 1.27080 | 1.0600% |
| January 1, 1986 | 1.270877 | 1.060056 |
| February 1, 1986 | 1.383477 | 1.2440% |
| March 1, 1986 | 1.383477 | 1.2440% |
| April 1, 1986 | 1.383477 | 1.2440% |
| May 1, 1986 | 1.33 34 6 | 1.244076 |
| June 1, 1986 | 1.383477 | 1.2440% |
| July 1, 1986 | 1,3834% | 1.2440% |
| August 1, 1986 | 1.383477 | 1.2440% |
| September 1, 1986 | 1.383497 | 1.2440% |
| October 1, 1986 | 1.3854% | 1.2440% |

DISCHARGE SCHEDULE - Continued

| Regular Application Date | | Hydrocarbons Monthly Percentage | Minerals Monthly Percentage |
|--------------------------------|--|---------------------------------------|-----------------------------------|
| November 1, 1986 | | 1.3834% | 1.2440% |
| December 1, 1986 | | 1.3834% | 1.2440% |
| January 1, 1987 | | 1.3834% | 1.2440% |
| February 1, 1987 | | 1.3137 | 1.2720% |
| March 1, 1987 | | 1.31:1777 | 1.2720% |
| April 1, 1987 | | 130370 | 1.2720% |
| May 1, 1987 | | 1.31374 | 1.2720% |
| June 1, 1987 | | 1.3137 % | 1.272656 |
| July 1, 1987 | | 1.3137% | 1.2720% |
| August 1, 1987 | | 1.313777 | 1.2720% |
| September 1, 1987 | | 1.3137% | 1.2720% |
| October 1, 1987 | | 1.3137% | 1.2720% |
| November 1, 1987 | | 1.3137% | 1.272076 |
| December 1, 1987 | | 1.3137 € | 1.2720% |
| January 1, 1988 | | 1.3137% | 1.2720% |
| February 1, 1988 | | 1.3244% | 1.372000 |
| March 1, 1988 | | 1.3244% | 1.3720% |
| April 1, 1988 | | 1.3244% | 1.3720% |
| May 1, 1988 | | 1.324477 | 1.3720% |
| Jun 1, 1988 | | 1.324476 | 1.3720% |
| July 1, 1988 | | 1.3244% | 1.3720% |
| August 1, 1988 | | 1.3244% | 1.3720% |
| September 1, 1988 | | 1.3244% | 1.3720% |
| October 1, 1988 | TOTAL TOTAL STATE OF THE STATE | 1.3244% | 1.3720% |
| November 1, 1988 | | 1.324456 | 1.3720% |
| December 1, 1988 | | 1 3244% | 1.3720% |
| January 1, 1989 | | 1.3244% | $1.3720\gamma e$ |
| February 1, 1989 | | 1.3351% | 1.4920% |
| March 1, 1989 | | 1.5351% | 1.4920% |
| April 1, 1989 | | 1.3351% | 1.4920% |
| May 1, 1989 | | 1.3351% | 1.4920% |
| June 1, 1989 | | 1.3351% | 1.4920% |
| July 1, 1989 | | 1.335156 | 1.4920% |
| August 1, 1989 | | 1.3351% | 1.4920% |
| September 1, 1989 | | 1.3351 % | 1.4920% |
| October 1, 1989 | | 1.3351% | 1.4920% |
| November 1, 1989 | | 1.3351% | 1.4920% |
| December 1, 1989 | | 1,3351% | 1.4920% |
| January 1, 1990 | | 1.3351% | 1.4920% |
| February 1, 1990 | | 0.932017 | 1.4464% |
| March 1, 1990 | | | 1.4464% |
| April 1, 1990 | | | 1.4464% |
| May 1, 1390 | | 1 777 8 | |
| June 1, 1990 | | - | 1.4464% |
| Dune 1, 1930 | The state of the s | - | 1.4464% |

EXHIBIT C

Instrument of Conveyance of Production Payment

PENNZOIL COMPANY, ET AL.

TO

NEW DUNSTAN, INC.

Supplemental Instrument of Conveyance of Production Payment

Dated as of , 198

Supplementing the Instrument of Conveyance of Production Payment Dated July 16, 1981

Return recorded counterparts to:
PENNZOH, COMPANY
PennZoil Place, P.O. Box 2967
Houston, Texas 77002
Attentions Mr.

WITNESSETH:

Whereas, the Grantors and the Grantee have heretofore entered into _ Instrument of Conveyance of Production Paymert dated July 16, 1981 (the "Conveyance") at an executed counterpart of the Conveyance was filed for record in Volume _, Page _ of the Deed Records of Culberson County, Texas, to which recorded counterpart reference is made for all purposes; and

Whereas, the Grantors desire to supplement Exhibit A to the Conveyance pursuant to and in the manner prescribed by subparagraph (b) of Paragraph 7 of the Conveyance;

Now, Therefore, the Grantors hereby execute and deliver this Supplemental Conveyance as follows:

 Additions to Exhibit A. Exhibit A to the Conveyance is hereby supplemented by adding thereto the following:

PART THREE

Sulphur Culberson County, Texas

T.

PATENTED MINING CLAIM

563.6 acres of land out of Section 2, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain thirty-four (34) Mineral Patents identified as (i) Nos. 358-368, Volume 40-B, recorded in Volume 6, Pages 456-466; (ii) No. 212, Volume 42-B, recorded in Volume 7, Page 584; and (iii) Nos. 370-391, Volume 40-B,

recorded in Volume 6, Pages 468-489, (all recordation in the Patent Records of Culberson County, Texas), subject, however, to (i) the royalty of the State of Texas equal to six and one-fourth percent (614%) of the value of the production of sulphur and certain other minerals from said unpatented claims and awards as shown by the net smelter, mill, mint or refinery returns or of the gross sums arising from the sale of the ore or products from such claims and awards and received by the owner thereof, and (ii) a perpetual royalty interest of an undivided one-thirty second of the gross value of all production from said mining claims of suiphur, salt, potash and any other minerals covered by said mining claims, being an undivided one half of that certain royalty reserved in that certain Conveyance of Unpatented Mining Clams dated September 1, 1967, and recorded in Volume 3, page 217, of the Mineral Records of Culberson County, Texas from Topat Oil Corporation to Duval Corporation, insofar as the same relates to the above-described claims.

Ħ.

LEASEHOLD

An undivided one-half (42) interest in and to 160 acres of land out of Sections 2 and 3, Block 111, PSL Survey, Culberson County, Texas, and being further identified as the "Dot" Mining Claim, Mineral Patent No. 70, covered by Sulphur Lease dated September 29, 1967, from R. B. McGiowen, Jr., et al. to Duval Corporation, and recorded in Volume 3, page 212, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty on all sulphur mined, saved and marketed from said land by Lessee of one-sixth (46th) (reduced to one twelfth (1/12th) by reason of proportionate reduction clause), f.o.b., mine site.

III.

LEASEHOLD

An undivided one-half ('2) interest in and to 160 acres of land cut of Sections 2 and 3, Block 111, PSL Survey, Culberson County, Texas, and being the "Dot" Mining Claim, Mineral Patent No. 70, covered by Sulphur Lease dated September 14, 1967, from Toyah Valley Sulphur Co. Bondholders, Inc. to Fred Prickett, and recorded in Volume 3, Page

309, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty of one-sixth (13th) (reduced to one-twelfth (1/12th) by reason of proportionate reduction clause), of the gross production of sulphur produced from the leased premises.

IV.

LEASEHOLD

555 acres of land out of Section 3, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by Sulphur Lease dated November 7, 1967, from the State of Texas to W. C. Tillett, General Land Office Sulphur Lease No. 61880, and recorded in Volume 3, page 242, of the Mineral Becords of Culberson County, Texas, subject, however, to a coyalty reserved to the State of Texas in said Lease of one-sixth (16th) of the cross production, or the value thereof, of the sulphur; but in no event, to be less than Four and 56/100 Dollars (\$4.50) per long ten.

٧.

PATENTED MINING CLAIMS

639.96 acres of land out of Section 4, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those thirtysix (36) Mineral Patents identified as Nos. 256.291, Volume 38 B, recorded in Volume 5, Pages 326.361, inclusive, of the Patent Records of Culberson County, Texas, subject, however, to the royalty of the State of Texas equal to six and one fourth per cent (614%) of one value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or of the gross sums arising from the sale of the ore or products from such claims and awards by the owner thereof.

VI.

LEASEHOLD

214.27 acres of land out of Section 9, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by Sulphur Lease dated April 15, 1968, from R. B. McGowen, Jr., et al, to Duval Corporation, General Land Office Mineral File No. M-61178, and

recorded in Volume 5, page 72, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty of one-sixth (15th) of the gross production of sulphur produced from said leased premises, reserved in said lease.

VII.

PATENTED MINING CLAIMS

425.58 acres of land out of Section 9, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain twenty-three (23) Mineral Patents identified as Nos. 173-195, Volume 39-B, recorded in Volume 6, Prices 52-74, inclusive, of the Patent Records of Culberson County, Texas, subject, however, to (i) the royalty of the State of Texas equal to six and one-fourth per cent (61477) of the value of the production of sulphur and certain other minerals from said patented claims and awards as shown by the net smelter, mill, mint, or refinery returns or of the gross sums arising from the sale of ore or products from such claims or awards and received by the owner thereof, and (ii) a perpetual royalty interest of an undivided one-thirty second of the gross value of all production from said mining claims of sulphur, salt, potash and any other minerals covered by said mining claims, being an undivided one-half of that certain royalty reserved in that certain Conveyance of Unpatented Mining Claims dated September 1, 1967, and recorded in Volume 3, page 217, of the Mineral Records of Culberson County, Texas, from Topat Oil Corporation to Duval Corporation, insofar as the same relates to the above-described claims and awards.

VIII.

PATENTED MINING CLAIMS

640.56 acres of land out of Section 10, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain thirty-six (36) Mineral Patents identified as (i) Nos. 220 and 221, Volume 38-B, recorded in Volume 5, Pages 290-291; (ii) Nos. 223-234, Volume 38-B, recorded in Volume 5, Pages 293-304; (iii) Nos. 236-240, Volume 38-B, recorded in Volume 5, Pages 306-310; (iv) Nos. 242-255, Volume 38-B, recorded in Volume 5, Pages 312-325; (v) No.

96, Volume 40-B, recorded in Volume 5, Page 76; (vi) Nos. 255 and 256, Volume 39-B, recorded in Volume 5, Pages 550 and 551 (all recordation in the Patent Records of Culierson County, Texas), subject, however, to the royalty of the State of Texas equal to six and one-fourth per cent (614%) of the value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or the gross sums arising from the sale of the ore or products from such claims and received by the owner thereof.

IX.

LEASEHOLD

480 acres of land out of Section 15, Block 111, PSL Survey, Culberson County, Texas, being the E-12, S-12 SW-14, N-12 NW-14, of said Section, and being the same land covered by Sulphur Lease dated November 7, 1967, from the State of Texas to W. C. Tillett, General Land Office Sulphur Lease No. M-61881, and recorded in Volume 3, page 246, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty reserved to the State of Texas in said Lease of one-sixth (16th) of the gross production or the value thereof, of the sulphur; but in no event, to be less than Four and 50/100 Dollars (\$4.50) per long ton.

X.

PATENTED MINING CLAIM

380.79 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain twenty-six (26) Mineral Patents identified as Nos. 47-72, Volume 38-B, recorded in Volume 5, Pages 209-232, inclusive, of the Patent Records of Culberson County, Texas, subject, however, to the royalty of the State of Texas equal to six and one-fourth per cent (614%) of the value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or the gross sums arising from the sale of the ore or products from such claims and received by the owner thereof.

XI.

PATENTED MINING CLAIM

31.54 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those certain two (2) Mineral Patents identified as Nos. 93 and 94, Volume 40-B, recorded in Volume 6, Pages 270-271 of the Patent Records of Culberson County. Texas, subject, however, to the royalty of the State of Texas equal to six and one-fourth per cent (614%) of the value of the production of sulphur and certain other minerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or the gross sums arising from the sale of the ore or products from such claims and received by the owner thereof.

XII.

LEASEHOLD ON PATENTED MINING CLAIM

80 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, being the E-12 of the "Virginia" Mining Claim, Mineral Patent No. 68, Volume 1, below the elevation of 100 feet measured vertically from the surface thereof on July 1, 1968, and being the same land covered by Sulphur Lease dated March 26, 1968, from Violet G. O'Fiel to Jack R. Fraser, and recorded in Volume 4, page 267, of the Mineral Records of Culberson County, Texas, insofar only as the same covers below the elevation of 100 feet measured vertically from the surface of said land on July 1, 1968, subject, however, to royalty and overriding royalty aggregating four-forty seconds (4/42nds) of all sulphur produced and marketed.

XIII.

MINERAL DEED AND PATENTED MINING CLAIM

80 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, being the W-12 of the "Virginia" Mining Claim, Mineral Patent No. 68, Volume 1, below the depth of 100 feet measured vertically from surface level thereof on February 28, 1968, and being the same land conveyed by Deed dated February 28, 1968, from Jack R. Fraser to Duval Corporation, and recorded in Volume 5, page

91, of the Mineral Recents of Culberson County, Texas, and being the same land conveyed by Died inted February 28, 1968, from Jack R. Fraser, Successor Trustee of Michigan Sulphur and Oil Co., to Duval Corporation, and recorded in Volume 5, page 89, of the Mineral Records of Culberson County, Texas.

XIV.

LEASEHOLD

40 acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by Sulp'ear Lease dated May 7, 1968, from the State of Texas to Duval Corporation, General Land Office Sulphur Lease No. M-62083, and recorded in Volume 5, page 128, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty reserved to the State of Texas in said Lease of one sixth (1, th) of the gross production, or the value thereof, of the sulp'ear; but in no event, to be less than Four and 50/100 Dollars (\$4.50) per long ton.

XV.

LEASER OLD

33.0? acres of land out of Section 16, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by Sulphur Lease dated May 7, 1968, from R. B. McGewen, Jr., et al. to Duval Corporation, General Land Office Mineral File No. M-61996, and recorded in Volume 5, page 75, of the Mineral Records of Culberson County, Texas, subject, however, to a royalty of one-sixth (16th) of the gross procession of sulphur produced from said leased premises, reserved in said lease.

XVI.

PATENTED MINING CLAIMS

160.44 acres of land out of Section 11, Block 111, PSL Survey, Culberson County, Texas, and being the same land covered by those nine (9) Mineral Patents identified as Nos. 438-446, Volume 38-B, recorded in Volume 5, Pages 455-463, inclusive, of the Patent Records of Culberson County, Texas, sui however, to (i) the royalty of

.

the State of Texas equal to six and one-fourth per cent (614%) of the value of the production of sulfur and certain other mirerals from said patented claims as shown by the net smelter, mill, mint, or refinery returns or of the gross sum arising from the sale of the ore or products from such claims and received by the owner thereof, and (ii) a perpetual royalty interest of an undivided one thirty second of the gross value of all production from said mining claims of sulfur, salt, potash and any other minerals covered by said mining claims, being an undivided one-half of that certain royalty reserved in that certain Conveyance of Unpetented Mining Claims dated September 1, 1967 and recorded in Volume 3, page 217, of the Mineral Records of Culberson County, Texas, insofar as the same relates to the above described claims

XVII.

FEE TITLE AND LEASEHOLD

- (1) An undivide I one-half () fee interest in and to the 160 acres comprising the south one-half of the northwest one-quarter and the north one-half of the southwest one-quarter of S. tion 15, Block 111; PSL Survey, Culberson County, Fexas, being the same land covered by Deed dated July 29, 1971, and to reded to Venume 92, Page 176 of the Deed Records of Culberson County, Fexas; and
- (2) An undivided one-half (12) interests and to the 160 acres compassing the south one-half of the northwest one-quarter and the north one-half of the southwest one-quarter of Section 15 Block 111, PSL Survey, Culberson County, Texas, and being the same and covered by a Mineral Lease dated July 22, 1971, from Jack Russell to Duval Corporation and recorded in Volume 8, Page 48, Mineral Records of Culberson County, Texas, subject, however, to an initial royalty of one-sixteenth (1/16) on sulphur, either in kind or market value at the railhead at the mine site, for the first 2,750,000 long tons whether produced or not and a subsequent royalty of one-sixteenth (1/16) on all sulphur certially produced in excess of 2,750,000 long tons reduced to .001% when 3,900,000 long tons have been produced.

XVIII

LEASEHOLD

2.256 acres of land out of Sections 14, 15, 22 and 23, Block 110, PSL Servey, Cullerson County, Texas, being the same land covered by Sulphur Lease dated August 26, 1975, from Hillary Phillips, Jr.

and Benny Gail T. Hunnicutt to Fred T. Armstrong, General Land Office Mineral File No. M-72194 and recorded in Volume 8, Page 414 of the Mineral Records of Culberson County, Texas, subject, however, to a royalty reserved to the State of Texas in said Lease of 16.67% of the gross production of sulphur, and subject to an overriding royalty interest of .00332 of all sulphur that may be produced from the land covered by said Lease reserved in Assignment dated September 15, 1975, from Fred T. Armstrong to Duval Corporation, recorded in Volume 9, Page 51 of the Mineral Records of Culberson County, Texas.

XIX

FEE TITLE

160 acres of land, being all of Pre-emption Survey No. 86 as described in Patent dated November 14, 1945, recorded in Volume 7, Page 75 of the Deed Records of Culberson County, Texas, and being the same land covered by Deed dated January 31, 1969, from R. B. Jetlowen, Jr., et al., to Duyal Corporation, recorded in Volume 86, Page 121 of the Deed Records of Culberson County, Texas.

2. Conveyance of the Production Payment. The Grantors, in consideration of the payments made and agreed to be made to them by the Grantee of the Purchase Price pursuant to the provisions of Paragraph 7 of the Conveyance and of other good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, hereby confirm and ratify the estates and rights created by the Convevance, as supplemented by this Supplemental Conveyance, and, in furtherance of such confirmation and ratification, the Grantors by these presents do hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, effective as of 1, 198 | first day of the calendar month in which the Supplemental Instrument of Conveyance of Production Payment is executed as a production payment, that undivided percentage, or it 100%, the percentage, of the Subject Minerals (including, without limitation, those Subject Minerals in and under and which may be produced, saved and sold and which shall accrue and be attributable to, the Subject Interests described in Paragraph I of this Supplemental Conveyance) provided for in Subparagraphs First, Second, Third and Fourth of Paragraph 2 of the Conveyance, together with an easement of ingress and egrees to remove the same from the applicable Subject Interests to the extent Grantors have the right to grant such an easement, for the purposes, upon the terms and conditions, and subject to the encumbrances, reservations and limitations contained in the Conveyance, as supplemented by this Supplemental Conveyance.

3. Warranty of Title. The Grantors hereby make, as to the Subject Interests described and referred to in Paragraph 1 hereof, all of the representations and warranties contained in Paragraph 20 of the Conveyance.

4. Miscellaneous Provisions.

Attest:

- (a) From and after the execution and delivery of this Supplemental Conveyance, the Conveyance shall be deemed to be supplemented as herein provided (and the term "Conveyance" shall thereafter mean the Conveyance as so supplemented), and as so supplemented shall continue in full force and effect, and none of the rights, titles or interests existing or to exist thereunder shall be released, diminished or impaired.
- (b) The Conveyance and this Supplemental Conveyance shall be read, taken and construed as one and the same instrument.
- (c) Unless otherwise specifically provided herein, all terms contained in this Supplemental Conveyance which are defined in the Conveyance shall for all purposes hereof have the meanings given to such terms in the Conveyance.

IN WITNESS WHEREOF, the Grantors have executed this instrument in the presence of the undersigned witnesses on the date set forth in the acknowledgment annexed hereto, in multiple originals.

| [Corporate Seal] | Pennzoil Company |
|------------------|----------------------------|
| | By Vice President |
| Attest: | |
| ****** | Pennzoil Producing Company |
| [CORPORATE SEAL] | By Vice President |

| [CORPORATE SEAL] | PENNZOIL OIL & GAS, INC. |
|---|--------------------------|
| Attest: | By |
| | |
| [CORPORATE SEAL] | DUVAL CORPORATION |
| Attest: | By Vice President |
| IN AND ADDRESS AND AND ADDRESS AND AND AND AND ADDRESS. | |

STATE OF NEW YORK COUNTY OF NEW YORK BE IT REMEMBERED that ". Notary Public duly qualified, compared, sworn and acting in and for the County and State aforesaid, a artify that on this day of ,198 : There appeared before a severally each of the following persons, each residing at the address set opposite his name, and each the designated officer of the corporation set opposite his name, each such corporation being a corporation of the state indicated and being a party to the foregoing instrument: , Vice President of Pennzoil Corporation, a Delaware corporation , Vice President of Pennzoil Producing Company, a Delaware corporation . Vice President of Pennzoil Oil & Gas, Inc., a Delaware corporation , Vice President of Duval Corporation, a Delaware corporation , Vice President of New Dunstan, Inc., a Delaware corporation.

Before me personally came each such person, to me personally known, and each, being by me duly sworn, did depose and say that he resides at the address set opposite his name and is the designated officer of the corporation set opposite his name and that said corporation is described in and executed the above instrument; that he knows the seal of the corporation; that the seal affixed to such instrument is such corporate seal; that it was so affixed by order of the Board of Directors of the corporation; and that he signed his name thereto by like order and that the instrument was the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared each of such person, each of whom is known to me to be the person whose name is subscribed to the foregoing instrument, and known to me to be the designated officer of the corporation set opposite his name, and each acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, in the capacity therein stated and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my band and official seal, in the City, County and State of New York, this day of, 198.

Notary Public

Notary Public, State of New York

Qualified in New York County Commission Expires , 19 . . . , 19

My commission expires

EXHIBIT D

Instrument of Convergance of Production Parament

CERTIFICATE

Pennzoil Company, a Delaware corporation, Pennzoil Producing Company, a Delaware corporation, Pennzoil Oil & Gas, Inc., a Delaware corporation and Duval Corporation, a Delaware Corporation (the "Grantors"), do hereby state and certify as follows:

- 1. This Certificate is furnished pursuant to subparagraph (b) of Paragraph 7 of that certain Instrument of Conveyance of Production Payment dated as of July 16, 1981 between the Grantors and New Danstan, Inc., a Delaware corporation (the "Grantee"), (said Instrument of Conveyance of Production Payment being herein called the "Conveyance"). Unless otherwise defined herein, defined terms used herein have the meanings assigned thereto in the Conveyance.
- 2. Except to the extent that the representations and warranties contained in Paragraph 20 of the Conveyance may not be true as a result of transactions or states of fact permitted by provisions of the Conveyance, the representations and warranties contained in Paragraph 20 of the Conveyance are true on and as of the date bereof as through made on and as of such date.
- 3. No Event of Default, or an event which with the giving of notice or the passe of time, or both, would become an Event of Default, has occurred and is continuing, or would result from the payment of the installment of the Purchase Price to be made on the date hereof.

| IN WITNESS | Whereor, the Grantors have caused this certificate to |
|------------------|--|
| be duly executed | by officers of the Grantors thereunto duly authorized, |
| as of this da | y of ,19 *. |

| PE | NNZOIL COMPANY |
|-----|--------------------------|
| Ву | Title: |
| PEN | NNZOIL PRODUCING COMPANY |
| Ву | Title: |
| PEN | NIZOIL OIL & GAS, INC. |
| Ву | Title: |
| Du | VAL CORPORATION |
| Ву | Title: |

Insert the date of payment of the applicable installment of the Purchase Price.

Supplemental

Mortgage, Deed of Trust

and

Assignment of Production

Dated, 198

NEW DUNSTAN, INC.

TO

F. D. THOMPSON

AND

W. T. FOX III,

Trustees

AND

CITIBANK, N.A.,

As Agent

Return recorded counterparts to:

SHEARMAN & STERLING 399 Park Avenue New York, New York 10022 Attention: Michael A. Ross, Esq. This instrument was prepared by:

Frank W. R. Hubert, Jr. 3000 One Shell Plaza Houston, Texas 77002 THIS SUPPLEMENTAL MORTGAGE, DEED OF TRUST AND ASSIGNMENT OF PRODUCTION (the "Supplemental Mortgage") dated 198, from New Denstan, Inc., a Delaware non-profit corporation (the "Company"), to F. D. Thompson of Houston, Texas and W. T. Fox III of Houston, Texas (herein collectively the "Trustees" and individually a "Trustee"), and CITIBANK, N.A., a national banking association (the "Agent"),

WITNESSETH:

Whereas, the Company, the Trustees and the Agent have heretofore entered into that certain Mortgage, Deed of Trust and Assignment of Production dated July 16, 1981 (the "Mortgage"), an executed counterpart of which is recorded in Volume , Page of the Deed of Trust Records of Culberson County, Texas, to which recorded counterpart reference is made for all purposes;

Whereas, concurrently with the execution of this Supplemental Mortgage, the Company and the Grantors have executed and delivered a Supplemental Instrument of Conveyance of Production Payment, in the form attached hereto as Exhibit A; and

Whereas, the Company desires to supplement the Mortgage by executing this Supplemental Mortgage as provided in Subsection R of Section 2.2 of the Mortgage;

Now, Therefore, the Company hereby executes and delivers this Supplemental Mortgage as follows:

Addition to Mortgaged Property. The Mortgaged Property hereby includes the Production Payment as defined and described in the Conveyance, as supplemented by the Supplemental Conveyance.

2. Confirmation of Mortgage. The Company hereby confirms and ratifies the estates and rights created by the Mortgage, as supplemented by this Supplemental Mortgage, and, in furtherance of such confirmation and ratification and to secure the payment of the indebtedness described in Section 1.1 of the Mortgage and the performance of the covenants therein and in the Notes contained, results consideration of the loans made to the Company by the Banks pursuant to the Credit

Agreement, the Company by these presents does grant, bargain, sell, mortgage, create a security interest in, pledge assign, transfer, convey, set over and deliver unto the Trustees and their successors in this trust and unto their assigns, the Production Payment, including all the Company's interest in the Production Payment Minerals and the proceeds thereof, and all rights, titles, interests, estates, remedies, powers and privileges vested in the Company as the owner of the Production Payment or which the Company now has or may become entitled to under or by virtue of the terms and provisions of the Conveyance as supplemented by the Supplemental Conveyance.

3. Miscellaneous Provisions.

- (a) From and after the execution and delivery of this Supplemental Mortgage, the Mortgage shall be deemed to be supplemented as herein provided (and the term "Mortgage" shall thereafter mean the Mortgage as so supplemented and the term "Conveyance" as used in the Mortgage shall thereafter mean the Conveyance as supplemented by the Supplemental Conveyance), and as so supplemented shall continue in full force and effect, and none of the rights, titles or interests existing or to exist thereunder shall be released, diminished or impaired.
- (b) The Mortgage and this Supplemental Mortgage shall be read, taken and construed as one and the same instrument.
- (c) Unless otherwise specifically provided herein, all terms contained in this Supplemental Mortgage which are defined in the Mortgage shall for all purposes hereof have the meanings given to such terms in the Mortgage.

IN WITNESS WHEREOF, the Company has executed this instrument before the undersigned Notary Public in the presence of the undersigned competent witnesses after a due reading of the whole on the date set forth in the acknowledgment annexed hereto.

| [Corporate Seal] | New Dunstan, Inc. |
|--|--|
| Attest: | By: President |
| Assistant Secretary | × |
| The address of New Dunstan, Inc is: | |
| STATE OF NEW YORK COUNTY OF NEW YORK SS | 5. |
| BE IT REMEMBERED THAT, I | I, the undersigned, a Notary Public duly and "or the County and State aforesaid, of, 198: |
| address set opposite his nan the corporation set opposite l | e the rollowing person, residing at the c. and being the designated officer of its name, s. In corporation being a corrisdiction and being a party to the fore |
| a Delaware corporation. | Vice President of New Dunstan, Inc. |

(New York)

Before me personally came such person, to me personally known, and he, being by me duly sworn, did depose and say that he resides at the address set opposite his name and is the designated officer of the corporation set opposite his name and that said corporation is described in and executed the above instrument; that he knows the seal of the corporation; that the seal affixed to such instrument is such corporate seal; that it was so affixed by order of the Board of Directors of the corporation; and that he signed his name thereto by like order and that the instrument was the free act and deed of the corporation.

Before me on this day personally appeared such person, who is known to me to be the person whose name is subscribed to the foregoing instrument, and known to me to be the designated officer of the corporation set opposite his name, and he acknowledged to me that he executed said instrument for the jurposes and consideration therein expressed, in the capacity therein stated and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have hereunto set my hand and official seal, in the City, County and State of New York, this _____ day of ______, 198 _.

Notary Public
Notary Public, State of New York
Qur lifted in New York County
Comession Expires,
198...

My commission expires

(Texas)



RESOLUTIONS ADOPTED BY THE BOARD OF DIRECTORS OF NEW DUNSTAN, INC.

Resolved, that the Company purchase from Pennzoil Company, Pennzoil Producing Company, Pennzoil Gil & Gas, Inc. and Daval Corporation, each a Delaware corporation, (collectively referred to herein as the "Grantors"), for a cash consideration of up to \$2,500,000,000 (subject to reduction as provided in the below-mentioned Instrument of Conveyance of Production Payment ("Conveyance")), payable in installments as provided in the Conveyance, a production payment ("Production Payment"), which is dischargeable out of the varying and fixed percentages specified in the Conveyance of the Minerals (as that term is defined in the Conveyance) in and under, and which may be produced and saved from, and which may accrue or be attributable to, the Subject Interests, as defined or described in the Conveyance, commencing on the Effective Date referred to in the Conveyance executed and delivered on behalf of the Company pur-

Resolved, that the form, terms and provisions of the Instrumen Production Payment, as submitted with this Written Consent, pur Production Payment is to be conveyed to the Company by the Grantors, be and the Conveyance of same is hereby approved, and that the President or any Vice President of the Company be and each of them hereby is authorized to execute and deliver to the Grantors, 2 behalf of and in the name of the Company and under its corporate seal attested by the Secretary or any Assistant Secretary of the Company, a conveyance of production payment substantially in the form of the Conveyance, with such changes therein as may be approved by the officer of the Company elecuting the same, his execution thereof to be conclusive evidence of such approval; and further

Resolved, that the form, terms and provisions of the Credit Agreement ("Credit Agreement") among the Company, the banks maned therein ("Banks") and Citibank, N.A., as Agent, providing for the borrowing by the Company of 2p to \$2,590,000,000 from time to time, together with the form of promissory notes attached to the Credit Agreement as Exhibit A ("A Notes"), and the form of the promissory notes attached to the Credit Agreement as Exhibit B ("B Notes"), (all such promissory notes and any promissory notes of the Company issued in exchange or substitution therefor here-, all as submitted with this Written Consent, be and the same kereby are approve,, and the President or any Vice President of the Company is each hereby authorized to execute and deliver on behalf of and in the name of we a credit agreement substantially in the form of the Credit Agreement, with nges therein as such officer executing the same may doesn necessary or is execution thereof to be conclusive evidence of his approval for and in the name or and on behalf of the Company of any changes therein; and further

Resolver, that in order to make payments on account of the purchase price of the Production Payment, the Company borrow from time to time an aggregate principal amount not to exceed \$2,500,000,000 under the Credit Agreement and that, as evidence of

such borrowing, the President, or any Vice President of the Company be and each of them is hereby authorized and directed to execute and deliver, on behalf and in the name of the Company, promessory notes payable to the order of the Banks, substantially in the form of the A Notes and B Notes, respectively, dated the date of the initial corrowing under the Credit Agreement and payable as to principal and bearing interest thereon as therein provided, with such changes therein as such officer executing the same shall approve, his execution thereof to be conclusive evidence of such approval; and further

Resouver, that as security for the repayment of the indebtedness ϵ idenced by the Notes and the other indebtedness of the Company referred to in Section 1+ of the belowmentioned Mortrage, the Company mortrage to F. D. Thompson, Houston, Texas, and W. T. Fox, III, Houston, Texas as Trustees, the Production Payment is cluding all the Company's interests in the Production Payment Minerals (as that term is defined in the Conveyance) and all rights, titles, interests, estates, remedies, powers and privileges of the Company therein, and assign to Citibank, N. V. as Accent for the Account of the Bank, the Company's increst in the Production Payment Minerals and the proceeds thereof; and that the Company take all recessary actions of record, file and perfect the security interests created thereby; and further

Resolver, that the form and terms of the Mortgage, Deed of Trust and Assignment of Production ("Mortgage") accompanying this Written Consent by which the Company is to mortgage the Production Payment to the Trustees and assign to Citibank, N.A., as Agent for the Vecount of the Banks, all production of the Production Payment Minerals and the proceeds thereon, as aforesaid, be and the same are hereby approved, and that the President or any Vice President of the Company be and each of them hereby is authorized and directed to execute and deliver to the Trustees and Citibank, N.A., as Agent, on behalf of and in the name of the Company and under its corporate seal attested by the Secretary or any Assistant Secretary of the Company, a mortgage, deed of trust and assignment of production in substantially be form of the Mortgage, with such changes therein as may be approved by the officer of the Company executing the same, his approved thereof being conclusive evidence of such approval; and further

RUSOLVED, that the Conveyance, the Mortgage, the Credit Agreement and copies of the Notes be filed by the Secretary or any Assistant Secretary of the Company among the records of the Company; and further

RESOLVED, that the officers of the Company be and they hereby are authorized to do and perform such acts and things and to sign such documents and certificates and to take all such other action, as they may deem to be necessary or advisable or convenient or proper to carry out the intent of the foregoing resolutions, including, without fimitation, in the event the Supplemental Instrument of Conveyance of Poduction Payment, as defined in the Conveyance, is excepted and delivered, the executio, and delivery to the Trustees and Citibank, X.A., as Agent, of a Supplemental Morecage, Deed of Trust and Assignment of Production substantially in the form of Exhibe C to the Mortgage as executed, with such changes therein as may be approved by the officer of the Commany executing the same, his execution thereof to be correlative evidence of such approval, the execution of division orders, transfer orders and payment orders and of any other instruments or documents which they may deem to be necessary or proper to place the Company or any assigner of the Commany or mortgages of the Production Payment in line for taxment of all or any part of the proceeds of production accruing or attributable to the Production Payment and fully to perform the provisions of the conveyance, the credit agrees ment, the mortgage, the notes and the other agreements executed and delivered on behalf of the Company pursuant to these resolutions.

CERTIFICATE

I, the undersigned, Assistant Secretary of New Dunstan, Inc., a Delaware non-profit corporation (the "Company"), do hereby certify that the foregoing is a true and correct copy of Resolutions duly adopted by the Board of Directors of the Company by unanimous consent on July 9, 1981, which Resolutions have not been revoked, rescinded or modified and are currently in full force and effect, and that the Mortgage, Deed of Trust and Assignment dated July 16, 1981 to which this Exhibit D is at the head and made a part, executed and delivered by the Company, is abstantially the form of that instrument referred to in said Resolutions submitted to and approved by the Board of Directors by unanimous consent as aforesaid.

Witness my hand and the seal of the Company this 16th day of July, 1981.

> Andrew M. Baker, Assistant Secretary New Dunstan, Inc.

[CORPORATE SEAL]

CULLEN R LIBROW (893-197) AUSTIN W LEWIS (1910-1974)

WILLIAM II METERS
POSERT T JODGEN
TOSERT T JODGEN
THREES C GEEMILION
SENE W LAFITTE
THOMAS I HARDENAN
L LITTON MORGAN
TOWNED GAR TOWNED
STEPHEN T JUTOBRA
ECON J RETMOND JR.
JERRY ST JOHN
DONALD R ARAUNIA
JOHN M WILSON
LAWRENCE R SIMON JR.
FREDERIC M SECON JR.
FREDERIC M SEGON JR.
ECON JR. GETTEN
ECON JR. GETTEN
ECONGE M ROSINSON JR.
ECONGE M ROSINSON JR.
ECONGE M ROSINSON JR.
ECONGE D DOMAS
MARITN C MALONEY
MARITN M BOOKSH JR.

LISKOW & LEWIS

NEW ORLEANS, LA. 70139

FIFTIETH FLOOR

TELEPHONE (804) 881-7979

TWX 610-981-9282

LAFAYETTE, LA. 70505
321 TRAVIS ST
A O BOX 52008
TELEPHONE 3181 232-1424

New Orleans, 70139-5099 February 28, 1983 COSEPH C GIGLIO JR.

JARRY M ROEDEL*

BRUCE J ORECR

PATRICH W SHAY

OEBOTAN BANN BRICE

THOMES M MCHAMARA

JANES N MANSPIELD III

BILLY J DOMINGUE

JAMES M MCHAMARA

AND JARIES N JARIES

WILLIAM W LAREFOURE

NICHAN B LABITER

WILLIAM W HUGH

PAUL B DAVID

LYNN C NANTEL

JULIE E SCHWARTZ

JULIE E SCHWARTZ

JULIE E SCHWARTZ

JOREN H SEDOUN

ROUGE V SCHKWE

JANE C SCHWE

JANE C S

United States Department of the Interior Minerals Management Service Gulf of Mexico OCS Region Imperial Ofice Building Metairie, Louisiana 70010 Attention: LE-3-1 Ms. Boehm

| Re: | Lease | Numbers | OCS-G | 2317 | OCS-G | 2912 |
|-----|-------|---------|--------|------|-------|------|
| | | | ncs-G | 2398 | OCS-G | 3141 |
| | | | OCS-G | 2403 | OCS-G | 2185 |
| | | | OCS-G | 2410 | OCS-G | 3299 |
| | | | 07 - G | 2412 | OCS-G | 3388 |
| | | | OUJ G | 2421 | OCS-G | 3390 |
| | | | OCa G | 2426 | OCS-G | 3400 |
| | | | OCS-G | 2607 | OCS-G | 3414 |
| | | | OCS-G | 2739 | OCS-G | 3589 |
| | | | OCS-G | 2743 | OCS-G | 2271 |
| | | | OCS-G | 2786 | ocs-G | 3416 |
| | | | OCS-G | 3286 | OCS-G | 2619 |
| | | | OCS-G | 2591 | OCS-G | 2887 |
| | | | OCS-G | 2910 | OCS-G | 2888 |

Gentlemen:

Mesa Petroleum Co. ("Mesa") has executed an Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production (Louisiana) dated February 17, 1983 ("Collateral Mortgage") securing a Collateral Mortgage Note in the amount of \$3,000,000,000 which has been pledged to Citibank, N.A. The Collateral Mortgage affects Mesa's interest in the above-captioned leases and further pledges and assigns production from those leases.

In order that third persons may be placed on notice as to the execution and efficacy of the Collateral Mortgage,

February 28, 1983

please file one copy of this letter, with an attached duplicate original of the Collateral Morrgage in the files in your office relating to each of the above-captioned leases.

By your signature in the space provided on the photocopies of this letter, please acknowledge that filing has been accomplished pursuant to this request.

Yours very truly,

LISKOW & LEWIS

By Marlyn C. Malorey

MCM:mb Ficlosures

Filing accomplished as requested

MINERALS MANAGEMENT SERVICE

February 28, 1983

ACT OF COLLATERAL MORTGAGE, COLLATERAL CHATTEL MORTGAGE, PLEDGE AND ASSIGNMENT OF PRODUCTION (LOUISIANA)

Dated February 17, 1983

BE IT KNOWN, that on this seventeenth day of February, 1983, before me, the undersigned authority, a Notary Public in and for the County of New York, State of New York, and in the presence of the two undersigned competent witnesses, personally came and appeared MESA PETROLEUM Co., a Delaware co: poration (herein called the "Mortgagor"), being herein represented by John F. Boros, its Vice President, said person being duly authorized so to act by resolutions of the Board of Directors of his corporation, a certified copy of which is attached hereto as Exhibit A-1, and said person declared unto me, said Notary, in the presence of said witnesses, that the Mortgagor desires to obtain funds from any person, firm or corporation willing to lend the same; and the Mortgagor is desirous of securing the performance of any obligation it may have previously incurred; and for such purpose, the Mortgagor by these presents declares and acknowledges an indebtedness in the sum of THREE BILLION DOLLARS (\$3,000,000,000) and to evidence such indebtedness the Mortgagor has executed a demand promissory note, being dated even date herewith said promissory note being in the principal amount of \$3,000,000,000, made payable to the order of Bearer, due on demand at the offices of Citibank, N.A., at 399 Park Avenue, New York, New York 10043, and said promissory note is stipulated to bear interest at the rate of eighteen percent (18%) per annum from date thereof until paid, and to provide for attorneys' fees, fixed at ten percent (10%) of the amount due or sued for or claimed or sought to be protected, preserved or enforced thereunder (said promissory note being hereinafter referred to as the "Mortgage Note"), all as set forth in the unexecuted copy of the Mortgage Note which is attached hereto as a part hereof and identified as Exhibit A-3, said Mortgage Note having been paraphed "Ne Varietur" by me, Notary, for identification with this Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production (herein referred to as this "Mortgage"), was delivered to the Mortgagor, which hereby acknowledges receipt thereof, and Mortgagor further declared that the Mortgage Note would be pledged to CITIBANK, N.A. for the purpose of raising funds, as here-ofore stated; and the Mortgagor hereby acknowledges that it is justly indebted unto any future holder or holders of all or any part of the indebtedness evidenced by the Mortgage Note in the full amount thereof, together with interest, attorneys' fees, and collection fees, and all other costs and indebtedness provided for herein, including the compensation of a keeper, should any accrue (all such future holder or holders of the Mortgage Note being hereinafter referred to as the "Bank", whether one or more).

In the event that the Mortgage Note should be placed in the hands of an attorney, after its maturity, to institute legal proceedings to recover the amount thereof, or any part thereof, in principal or interest, or to protect the interests of the holder or holders thereof, or in the event that the same should be placed in the hands of an attorney for collection, compromise, or other action, the Mortgagor hereby binds itself to pay the fees of the attorney who may be employed for that purpose, which fees are hereby fixed at ten percent (10%) of the amount due or sued for, or claimed or sought to be protected, preserved, or enforced.

For all purposes of this instrument, unless the context otherwise requires:

- A. "oil and gas leases" shall include oil, gas and mineral leases and shall also include subleases and assignments of operating rights.
- B. "Hydrocarbons" shall mean oil, gas, casinghead gas and other liquid or gaseous hydrocarbons.
- C. "Production Sale Contracts" shall mean contracts now in effect, or hereafter entered into by the Mortgagor, or the Mortgagor's predecessors in interest, for the sale, purchase, exchange or processing of Hydrocarbons produced from the lands under the oil and gas leases, mineral, overriding royalty, royalty and other interests described in Exhibit B attached hereto and made a part hereof.
- D. "lands described in Exhibit B or the described in Exhibit B or the description of which is incorpated in Exhibit B by reference to another instrument or document or which are covered or affected by an oil and gas lease, mineral, overriding royalty, royalty and other interests, and shall also include any lands now or hereafter unitized or pooled with lands which are either described in Exhibit B or the description of which is incorporated in Exhibit B by reference.
- E. "Related Contracts" shall mean all unitization, operating and pooling agreements and other contracts and agreements that relate to the Mortgagee's interest in the lands described in Exhibit B or the oil and gas leases described on Exhibit B.
- F. "Commercial Movables" shall mean all corporeal movable property of whatever kind or character now or hereafter owned by the Mortgagor and placed upon the lands described in Exhibit B or on a unit including all or part of the lands described in Exhibit B at which location the Mortgagor operates as a commercial and industrial facility oil and gas wells, and which corporeal movables are used by the Mortgagor in the conduct of such activity, or are dedicated to the use and exploitation of any mineral right, it being the intention of the Mortgagor to mortgage and affect all corporeal movables now or hereafter owned by the Mortgagor that are located from time to time on the lands described in Exhibit B or on a unit including all or part of the lands described in Exhibit B for use in the conduct of the foregoing commercial and industrial activity or are dedicated to the use and exploitation of any mineral right.
- G. "Operating Equipment" shall mean all surface or subsurface machinery, equipment, facilities, supplies or other property of whatsoever kind or nature (excluding drilling ries, drill pipe, mud pumps, trucks, automotive equipment or other property taken to the premises to drill a well or for other similar temporary uses) now or hereafter located on or under any of the lands described in Exhibit B or on a unit including all or part of the lands described in Exhibit B which are used or useful for the production, treatment, storage or transportation of or exploration for hydrocarbons, including, but not by way of limitation, all oil wells, gas wells, platforms, water wells, injection wells, casing, tubing, rods, pumping units and engines, christmas trees, derricks, separators, gun barrels, flow lines, tanks, gas systems (for gathering, treating, compression, disposal and injection), chemicals, solutions, water systems (for treating, disposal and injection), pipe, pipelines, boilers, compressors, connectors, power plants, poles, lines, transformers, starters and controllers, machine shops, tools, storage yards and equipment stored therein, buildings and camps, telegraph, telephone and other communication systems, roads, loading docks, loading racks and shipping facilities, together with all improvements, betterments and additions thereto and replacements thereof.
- H. "Mortgaged Property" shall mean the properties, rights and interests hereinafter described and defined as the Mortgaged Property.
 - I. "Indebtedness" shall have the meaning stated below.

And now, in order to secure the full, due, and punctual payment of all indebtedness evidenced by the Mortgage Note, and/or any extension or renewal thereof, as well as all costs, taxes, assessments, charges, insurance premiums, attorneys' and collect' 'es, and other costs and indebtedness incurred or paid hereunder, including the compensat a keeper, and any sums advanced or expenses or costs incurred by the Bank (or any receiver or keeper appointed hereunder) which are made or incurred pursuant to, or permitted by, the terms hereof, plus interest at the rate herein specified or otherwise agreed upon, from the date of the advances or the incurring of such expenses or costs until reimbursed, and to secure the faithful performance and observance of all obligations, agreements, covenants and stipulations contained herein and in the Mortgage Note (all of the preceding being hereinafter sometimes collectively referred to as the "Indebtedness"), the Mortgagor declares that it does by these presents mortgage, effect, pledge, assign and hypothecate to the Bank, whether the Mortgage Note may be held by the Bank as an original obligation or in pledge, all of the Mortgagor's right, title and interest, whether now owned or hereafter acquired, in and to the following properties, rights and interests:

- (a) the lands described in Exhibit B, and the oil and gas leases, the fee, mineral, overriding royalty, royalty and other interests which are specifically described in Ex' ... B,
- (b) the presently existing unitization and pooling agreements and the covered and the units created thereby (including all units formed under orders, to rules or other official acts of any federal state or other governmental against jurisdiction) which are specifically described in Exhibit B or which relate to properties and interests specifically described in Exhibit B.
- (c) the Hydrocarbons which are in, under, upon, produced or to be produced from ω... lands described in Exhibit B.
 - (d) the Production Sale Contracts.
 - (e) the Related Contracts.
 - (f) the Commercial Movables.
 - (g) the Operating Equipment,
- (h) all subleases, farmout agreements, assignments of interest, assignments of operating rights, contracts, operating agreements, rights of way, franchises, privileges, permits, licenses, easements, tenements, hereditaments, appurtenances and benefits now existing or in the future obtained and incident to or in the future obtained and incident and appurtenant to any of the foregoing,
- (i) all rights in and to all lease records well records and production records which relate to any of the foregoing, and
- (j) all other rights, titles and interests of the Mortgager in, to and under or derived from the lands, leases and properties described in Exhibit B, even though the Mortgagor's interest therein may be incorrectly or insufficiently described or referred to in Exhibit B.

(all the aforesaid properties, rights and interests, together with any additions thereto which may bjected to the lien of this instrument by means of supplements hereto, being here realled the "Mortgaged Property"),

owever, to (i) the restrictions, exceptions, reservations, conditions, limitations, ther matters, if any, set forth or specified to in the specific descriptions of such as and interests in Exhibit B (including all presently existing royalties, overriding properties out of production and other burdens which are specified in Exhibit B and the a into consideration in computing any decimal interests set forth in Exhibit B), and a pirage as assignment of production contained in Article II hereof, but only insofar and so low add a signment of production is not inoperative under the provisions of Section 2.4 hereof and the condition that the Bank shall not be liable in any respect for the

performance of any covenant or obligation of the Mortgagor in respect of the Mortgaged Property;

TO HAVE AND TO HOLD the Mortgaged Property unto the Bank forever to secure the payment of the Indebtedness and to secure the performance of the obligations of the Mortgagor herein contained.

The Mortgagor, in consideration of the premises and to induce the Bank to make the loan above described, hereby covenants and agrees with the Bank as follows:

ARTICLE I Particular Covenants and Warranties of the Mortgagor

- 1.1 Payment of the Indebtedness. The Mortgagor will duly and punctually pay the Indebtedness, including, without limitation, each and every obligation owing on account of the Mortgage Note.
- The Mortgagor warrants that (a) the oil and gas leases described in Exhibit B are valid, subsisting leases, superior and paramount to all other oil and gas leases respecting the properties to which they pertain, (b) the Mortgagor, to the extent of the inverest specified in Exhibit B, has and will have good and marketable title to each property right or interest constituting the Mortgaged Property and has a good and legal right to mortgage the same to the Bank, it being understood that the Mortgagor's interest in each oil and gas lease shall exceed the Mortgagor's net interest in production from such oil and gas lease to the extent of the Mc 'gagor's proportionate share of the burden of all royalties, overriding royalties and other such payments out of production, (c) the Mortgaged Property is free from all encumbrances or liens whatsoever, except as may be specifically set forth in Exhibit B or as permitted by the provisions of Section 1.5(e) hereof. (d) the Mortgagor is not obligated, by virtue of any prepayment under any contract providing for the sale by the Mortgagor of Hydrocarbons which contains a "take or pay" clause or under any similar arrangement, to deliver Hydrocarbons at some future time without then or thereafter receiving full payment therefor, (e) each of the oil and gas leases described in Exhibit B is in full force and effect, and all royalties due and payable have been timely and correctly paid, and all severance and production taxes with respect thereto have been timely and correctly paid, and (f) all producing wells located thereon have been drilled, operated, and produced in conformity with all applicable laws, and rules, regulations, and orders of all regulatory authorities having jurisdiction, and are subject to no penalties on account of past production, and that such wells are in fact bottomed under and are producing from, and the well borc3 are wholly within, lands described in Exhibit B. The Mortgagor hereby covenants that it will warrant and fore er defend the Mortgaged Property unto the Bank, against every person whomsoever lawfully claiming the same or any part thereof, will maintain and preserve the Mortgaged Property free from all encumbrances or liens whatsoever, except as may be specifically set forth in Exhibit B or as permitted by the provisions of Section 1.5(e) hereof and will maintain and preserve the lien hereby created so long as any of the Indebtedness remains unpaid. The Mortgagor also warrants that (w) the Mortgagor is a coporation duly organized, validly existing and in good standing under the laws of the State of Delaware and is duly licensed or qualified and in good standing as a forcign corporation authorized to do business in each jurisdiction in which any part of the Mortgaged Property is located, (x) the execution and delivery of this instrument and the Mortgage Note are within the Mortgagor's corporate powers and have been duly authorized by the Mortgagor by all necessary corporate action (no shareholder action of the Mortgagor being required for the Mortgager to execute and deliver this in trument), (y) the Mortgagor is now in a solvent condition and no bankruptcy or insolvency proceedings are pending or contemplated by the Mortgagor and (z) this instrument and the Mortgage Note each is the legal, valid and binding obligation of the Mortgagor enforceable against the Mortgagor in accordance with its terms, except as enforceability thereof may be limited by bankruptcy, insolvency, reorganization or

moratorium or other similar laws relating to creditors' rights and by general equitable principles which may limit the right to obtain equitable remedies.

- 1.3 Further Assurances. The Mortgagor will execute and deliver such other and further instruments and will do such other and further acts as in the opinion of the Bank n ay be necessary or desirable to carry out more effectively the purposes of this instrument including, without limiting the generality of the foregoing, (a) prompt correction of any defect which may hereafter be discovered in the title to the Mortgaged Property or in the execution and acknowledgment of this instrument, the Mortgage Note, or any other note or document executed in connection herewith, and (b) prompt execution and delivery of all division or transfer orders which in the opinion of the Bank are needed to transfer effectually to the Bank the assigned proceeds of production from the Mortgaged Property.
- 1.4 Taxes. Subject to the Mortgagor's right to contest the same, the Mortgagor will promptly pay all taxes, assessments and governmental charges (including but not limited to severance taxes) legally imposed upon this instrument or upon the Mortgaged Property or production therefrom or upon the interest of the Bank therein, or upon the income and profits thereof.
- 1.5 Operation of the Mortgaged Property. So long as the Indebtedness or any part thereof remains unpaid, and whether or not the Mortgagor is the operator of the Mortgaged Property, the Mortgagor shall, at the Mortgagor's own expense:
 - (a) Do all things necessary to keep unimpaired the Mortgagor's rights in the Mortgaged Property and not, except in the ordinary course of business, abandon any well or forfeit, surrender or release any oil and gas lease or any rights in the Mortgaged Property with respect to the Mortgaged Property without the prior written consent of the Bank;
 - (b) Cause the lands described in Exhibit B to be maintained, developed, protected against drainage, and continuously operated for the production of Hydrocarbons in a good and workmanlike manner as would a prudent operator, and in accordance with generally accepted practices, applicable operating agreements, and all applicable federal, state and local laws, rules and regulations, excepting those being contested in good faith;
 - (c) Pay, or cause to be paid, promptly as and when due and payable, all rentals and royalties payable in respect of the Mortgaged Property, and all expenses incurred in or arising from the operation or development of the Mortgaged Property and furnish the Bank as and when requested with full information with respect to such payments;
 - (d) Cause the Operating Equipment to be kept in good and effective operating condition, and all repairs, renewals, replacements, additions and improvements there if or thereto, needful to the production of Hydrocarbons from the lands cescribed in Exhibit B. to be promptly made;
 - (e) Cause the Mortgaged Property to be kept free and clear of liens, charges and encumbrances of every character, other than (1) the lien hereof, (2) taxes constituting a lien but not due and payable. (3) defects or irregularities in title, and liens, charges or encumbrances, which are not such as to interfere materially with the development, operation or value of the Mortgaged Property and not such as to affect materially title thereto, (4) those set forth in Exhibit B, (5) those being contested by the Mortgagor in good faith in such manner as not to jeopardize the Bank's rights in and to the Mortgaged Property, and (6) those consented to in writing by the Bank; and
 - (f) Carry in standard insurance companies and in amounts satisfactory to the Bank the following insurance: (1) workmen's compensation insurance and public liability and property damage insurance in respect of all activities in which the Mortgagor might incur personal liability for the death of or injury to an employee or third person, or damage to or destruction of another's property, and (2) to the extent such insurance is generally carried by others engaged in similar undertakings in the same general areas in which the Mortgaged Property is located, insurance in respect of the Operating Equipment (with loss

payable clauses in favor of the Bank), against loss or damage by fire, lightning, hail, tornado, explosion and other similar risks.

The Bank may from time to time, in its discretion, perform any agreement of the Mortga. Finder this Section 1.5 or Section 1.4 hereof which the Mortgagor shall fail to perform any tall any other action which the Bank deems necessary for the maintenance or preservation is a che Mortgaged Property or its interest therein, and the Mortgagor agrees forthwith to remotive the Bank for any and all expenses of the Bank in connection with such performance or action together with interest thereon at the rate of eighteen percent (18%) per annumber the date such expenses are incurred until such expenses are reimbursed by the Mortgagor.

- 1.6 Recording etc. The Mortgagor will promptly, and at its own expense, record, register, deposit and file this and every other instrument in addition or supplemental hereto in such offices and places, at such times and with such frequency as may be necessary to preserve, protect and renew the lien hereof as a first lien on real and personal property, as the case may be, and the rights and remedies of the Bank, and otherwise will do and observe all things or matters necessary or expedient to be done or observed by reason of any law or regulation of any State or of the United States of America or of any other competent authority, for the purpose of effectively creating, maintaining and preserving the lien hereof on the Mortgaged Property.
- 1.7 Sale or Mortgage of the Mortgaged Property. Except as provided in Section 1.5(e) hereof and except for sales of severed Hydrocarbons in the ordinary course of the Mortgagor's business and the lien and security interest created by this instrument, the Mortgagor will not sell, convey, mortgage, pledge or otherwise dispose of or encumber the Mortgaged Property nor ary Porcion thereof, nor any of the Mortgagor's right, title or interest therein, without first securing the written consent of the Bank.
- 1.8 Records, Statements and Reports. The Mortgagor will keep proper books of record and account in which complete and correct entries will be made of the Mortgagor's transactions in accordance with generally accepted accounting principles and will furnish or cause to be furnished to the Bank (a) upon its request, but not more than once a year, reports prepared by an independent person or firm acceptable to the Bank concerning (1) the quantity of Hydrocarbons recoverable from the Mortgaged Property, (2) the projected income and expense attributable to the Mortgaged Property, and (3) the expediency of any change in methods of treatment or operation of all or any wells productive of Hydrocarbons, any new drilling or development, any method of secondary recovery by repressuring or otherwise, or any other action with respect to the Mortgage. Proper y, the decision as to which may increase or reduce the quantity of Hydrocarbons ultimately recoverable, or the rate of production thereof, (b) monthly, a report showing the gross proceeds from the sale of Hydrocarbons produced from the lands described in Exhibit 5 (including any thereof taken by the Mortgagor for the Mortgagor's own use), the quantity of . h Hydrocarbons sold, the severance, gross production, occupation, or gathering taxes deducted from or paid out of such proceeds, the number of wells operated, drilled or abandoned, and such other information as the Bank may reasonably request (youn request of the Bank, such reports referred to in clauses (a) and (b) above shall set fortl, such information on a lease or unit basis), and (c) such other information concerning the business and affairs and financial condition of the Mortgagor as the Bank may from time to time reasonably request.
- 1.9 No Governmental Approvals. The Mortgagor warrants that no approval or consent of any regulatory or administrative commission or authority, or of any other governmental body, is necessary to authorize the execution and delivery of this instrument or of the Mortgage Note, or to authorize the observance or performance by the Mortgagor of the covenants contained herein or in the Mortgage Note.
- 1.10 Right of Entry. The Mortgagor will permit the Bank, or its agents, to enter upon the Mortgaged Property, and all parts thereof, for the purpose of investigating and inspecting the condition and operation thereof.

- 1.11 Corporate Status. The Mortgagor (a) will continue the design of the State of Delaware, duly qualified to transactive the conduct of its business requires it to be qualified and duly ant to the first one perform its obligations under the Mortgage Note and this instrument and the will not with at the prior written consent of the Bank, consolidate or merge with any other corporation.
- 1.12 Legal Proceedings. The Mortgagor will notify the Bank in writing, promy by of the commencement of any legal proceedings materially and adversely affecting the foregaged Property or any part thereof and will take such action as may be nowed sary to mesor elegand the Bank's rights affected thereby; and should the Mortgagor fail of the second take any sach action, the Bank may at its election take such action in behalf and in the name of the Mortgagor and at the Mortgagor's expense. Moreover, the Bank at its option may called such independent action on its own behalf as holder of rights or lieus on said properties as it may in its discretion deem proper, all expenses incurred in such actions, together with interest thereon at the rate of eighteen percent (187) per annum, also being reimbursable to it by the Mortgagor and being payable by the Mortgagor to the Bank on demand.

ARTICLE II Pledge and Assignment of Production

- Pledge and Assignment. As further security for the payment of the Indebtedness and to facilitate the discharge of all indebtedness for which the Mortgage Note may be pledged as security, the Mortgagor hereby transfers, pledges, assigns, warrants and conveys to the Bank, effective as of the date hereof, at 7:00 A.M., local time, all Hydrocarbons which are thereafter produced from and which accrue to the Mortgaged Property, and all proceeds therefrom. The Mortgagor will immediately cause the title to all Hydrocarbons produced from and which accrue to the Mortgaged Property, together with the proceeds therefrom, to be delivered to the credit of the Bank, to be applied to the indebtedness for which the Mortgage Note may be pledged as security. All parties producing, purchasing or receiving any such Hydrocarbons, or having such, or proceeds therefrom, in their possession for which they or others are accountable to the Bank by virtue of the provisions of this Article, are authorized and directed to treat and regard the Bank as the assignee and transferee of the Mortgagor and entitled in the Mortgagor's place and stead to receive such Hydrocarbons and all proceeds therefrom; and said parties and each of them shall be fully protected i.. so treating and ag the Bank, and shall be under no obligation to see to the application by the Bank of any such proceeds or payments received by it. The Mortgagor will furnish the Bank the names of all parties purchasing or receiving any of such Hydrocarbons or proceeds therefrom, and will promptly execute and deliver any and all transfer orders, division orders, and other instruments that may be requested by the Bank for the purpose of effectuating the pledge and assignment hereunder.
- 2.2 No Liability of the Bank in Collecting. The Bank is hereby absolved from all liability for failure to enforce collection of any proceeds so assigned and from all other responsibility in connection therewith, except the responsibility to account to the Mortgagor for funds actually received.
- 2.3 Assignment Not a Restriction on the Bank's Rights. Nothing herein contained shall detract from or limit the absolute obligation of the Mortgagor to make payment of the Indebtedness and of all obligations for which the Mortgage Note may be pledged as security regardless of whether the proceeds pledged and assigned by this Article are sufficient to pay the same, and the rights under this Article shall be in addition to all other security now or hereafter existing to secure the payment of the Indebtedness and of all obligations for which the Mortgage Note may be pledged as security.
- 2.4 Status of Pledge and Assignment. Notwithstanding the other provisions of this Article, the Bank or any receiver or keeper appointed in judicial proceedings for the enforcement of this instrument shall have the right to receive all of the Hydrocarbons herein pledged.

and assigned and the proceeds therefrom after the Mortgage Note or any other promissory notes of the Mortgagor held by the Bank have been declared due and payable in accordance with the provisions of Section 3.1 hereof. Upon any sale of the Mortgaged Property or any part thereof pursuant to Article IV hereof, the Hydrocarbons thereafter produced from the property so sold, and the proceeds therefrom, shall be injuded in such sale and shall pass to the purchaser free and clear of the pledge and assignment contained in this Article.

2.5 Indemnity. The Mortgagor agrees to indemnify the Bank against all claims, actions liabilities, judgments, costs, attorneys' fees or other charges of whatsoever kind or nature (hereinafter in this Section 2.5 collectively called "claims") made against or incurred by the Bank as a consequence of the assertion, either before or after the payment in full of the Indebtedness and of all obligations for which the Mortgage Note may be pledged as security, that the Bank received Hydrocarbons herein assigned or the proceeds thereof claimed by third persons, and the Eank shall have the right to defend against any such claims, employing attorneys therefor, and unless furnished with reasonable indemnity, the Bank shall have the right to pay or compromise and adjust all such claims. The Mortgagor will indemnify and pay to the Bank any and all such amounts as may be paid in respect thereof or as may be successfully adjudged against the Bank. The obligations of the Mortgagor as hereinabove set forth in this Section 2.5 shall survive the release of this instrument.

ARTICLE III Events of Default

- 3.1 Certain Events of Default. In case any one or more of the following "events of default" shall occur and shall not have been remedied.
 - (a) default, and the continuance thereof for five (5) days, in the payment of principal of or interest on any Indebtedness when due, whether by demand or otherwise;
 - (b) any warranty or representation made herein shall prove to be untrue in an material respect;
 - (c) failure by the Mortgagor within thirty (30) days after notice thereof from the Bank, to cure a default in the due performance or observance of any covenant or agreement contained in this instrument and not constituting a default in the payment of principal of or interest upon the Mortgage Note or in the payment of any other Indebtedness; or
 - (d) failure by the Mortgas or within five (5) days after demail from the Bank to pay over to the Bank any proceeds of Hydrocarbons which are paid to the Mortgagor rather than the Bank as provided in Section 2.1 hereof;
 - (e) the title of the Mortgagor to the Mortgaged Property or any substantial part thereof shall become the subject matter of litigation which would or might, in the Bank's opinion, upon final determination result in substantial impairment or loss of the security provided by this instrument and upon notice by the Bank to the Mortgagor such litigation is not dismissed within thirty (30) days,

then, and in any such event, the Bank, at its option, may declare the entire unpaid principal of and the interest accrued on the Mortgage Note and all other Indebtedness secured hereby to be forthwith due and payable, without any notice or demand of any kind, both notice and demand being hereby expressly waived.

ARTICLE IV Enforcement of the Security

4.1 Remedies on Default. The Mortgagor for itself, its successors and assigns, does by these presents stipulate that it shall be lawful for, and the Mortgagor hereby authorizes the Bank, upon the occurrence and during the continuance of any event of default, to cause all and singular of the Mortgaged Property to be seized and sold by executory process, without appraisement, either in its entirety or in lots or parcels as the Bank may determine, to the

highest bidder for cash, or on such terms as the Bank in such proceedings may direct; and the Mortgagor for itself, its—ressors and assigns, hereby acknowledges the obligations secured hereby whether now existing or to arise hereafter and confesses judgment for the full amount of the Mortgage Note, together with all costs, attorneys' and collection fees and all other charges and Indebtedness secured hereby if the same are not paid at maturity.

- 4.2 Waiver of Appraisement, Other Rights, etc. To the extent allowed by law, the Mortgagor hereby waives: (a) the benefit of appraisement, as provided in Articles 2332, 2336, 2723 and 2724 of the Louisiana Code of Civil Procedure, and all other laws conferring the same; (b) the demand and three (3) days delay accorded by Articles 2639 and 2721 of the Louisiana Code of Civil Procedure; (c) the notice of seizure required by Articles 2293 and 2721 of the Louisiana Code of Civil Procedure; (d) the three (3) days delay provided by Articles 2331 and 2722 of the Louisiana Code of Civil Procedure; and (e) the benefit of the other provisions of Articles 2331, 2722, and 2723 of the Louisiana Code of Civil Procedure and the benefits of any other articles or laws relating to rights of appraisement, notice or delay not specifically mentioned above; and the Mortgagor expressly agrees to the immediate seizure of the Mortgagod Property in the event of suit hereon.
- 4.3 Judicial Proceedings. Upon the occurrence of any event of default and if such event shall be continuing, the Bank may proceed by a suit or suits in equity or at law, whether for a foreclosure hereunder, or for the sale of the Mortgaged Property, or for the specific performance of any covenant or agreement herein contained or in aid of the execution of any power herein granted, or for the appointment of a receiver or a keeper pending any foreclosure hereunder or the sale of the Mortgaged Property, or for the enforcement of any other appropriate legal or equitable remedy.
- 4.4 Certain Aspects of a Sale. The Bank shall have the right to become the purchaser at any sale held by the court, receiver or public officer, and the Bank so purchasing shall have the right to credit the amount of the bid made therefor upon the amount payable out of the net proceeds of such sale to it. Recitals contained in any conveyance made to any purchaser at any sale made hereunder shall conclusively establish the truth and accuracy of the matters therein stated, including, without limiting the generality of the foregoing, nonpayment of the unpaid principal sum of, and the interest accrued on, the Mortgage Note after the same shall have become due and payable, and advertisement and conduct of such sale in the manner provided herein.
- 4.5 Receipt to Purchaser. Upon any sale, the receipt of the Bank, or of the officer making sale under judicial proceedings, shall be sufficient discharge to the purchaser or purchasers at any sale for his or their purchase money, and such purchaser or purchasers and his or their assigns or personal representatives shall not, after paying such purchase money and receiving such receipt of the Bank or of such officer therefor, be obliged to see to the application of such purchase money, or be in anywise answerable for any loss, misapplication or nonapplication thereof.
- 4.6 Effect of Sale. Any sale or sales of the Mortgaged Property shall operate to divest all right, title, interest, claim and demand whatsoever either at law or in equity, of the Mortgagor of, in and to the premises and the property sold, and shall be a perpetual bar, both at law and in equity, against the Mortgagor, its successors and assigns, and against any and all persons claiming or who shell thereafter claim all or any of the property sold from, through or under the Mortgagor, its successors and assigns; nevertheles—the Mortgagor, if requested by the Bank so to do shall join in the execution and delivery of all proper conveyances, assignments and transfers of the properties so sold.
- 4.7 Operation of the Mortgaged Property by the Bank. Upon the occurrence of an event of default, and in addition to all other rights herein conferred on the Bank, the Bank or its agent is hereby appointed a keeper of the Mortgaged Property: resuant to the terms and provisions of Louisiana Revised Statutes 9:5131, et seq. The keeper may operate the same without any liability to the Mortgagor in connection with such operations, except to use

ordinary care in the operation of said properties, and the keeper, to the extent permitted by law, shall have the right (a) to enter into and upon and take possession of the Mortgaged Property, to lease the same, collect and receive al! rents, issues and profits thereof and apply the same, less the necessary expenses of collection thereof, for the care, operation and preservation of the Mortgaged Property, including without limitation, the payment of fees, insurance premiums, cost of operation of the Mort; gaged Property, taxes, assessments, interest, penalties and water charges; (b) to collect, receive and receipt for all Hydrocarbons produced and sold from the Mortgaged Property, to make repairs, purchase machinery and equipment, conduct work-over operations, and drill additional wells; (c) to exercise every power, right and privilege of the Mortgagor with respect to the Mortgaged Property; and (d) to exercise every power, right and privilege now or hereafter granted to a key per of the Mortgaged Property under applicable law. The compensation for the services of the e-per are hereby fixed at five percent (5%) of the amount due or sold for or claimed or sou, it to be protected, preserved or enforced in the proceeding for the recognition of this Mortgage and, together with all costs, expenses and liabilities of every character incurred the Bank in managing, operating and maintaining the Mortgaged Property, as Keeper or otherwise, and interest thereon at the rate of eighteen percent (18%) per annum, shall be secured by the privilege and mortgage herein granted.

ARTICLE V Miscellaneous Provisions

- 5.1 Defense of Claims. The Mortgagor will notify the Bank, in writing, promptly of the commencement of any legal proceedings affecting the lien hereof or the Mortgaged Property, or any part thereof, and will take such action, employing attorneys agreeable to the Bank, as may be necessary to preserve the Mortgagor's or the Bank's rights affected thereby; and should the Mortgagor fail or refuse to take any such action, the Bank may take such action in behalf and in the name of the Mortgagor and at the Mortgagor's expense. Moreover, the Bank may take such independent action in connection therewith as it may in its discretion deem proper, the Mortgagor hereby agreeing that all sums advanced or all expenses incurred in such actions, plus interest at the rate of eighteen percent (18%) per annum, will, on demand, be reimbursed to the Bank.
- 5.2 Costs and Expenses. All costs, expenses and advances (including, without limitation, cost of defense of claims, attorneys' fees, taxes, insurance, and operation cost) incurred or paid by the Bank in preserving and protecting the Mortgaged Property and in preserving, protecting and enforcing its rights hereunder, all to the extent the Bank is given such right herein, up to twenty percent (20%) of the face amount of the Mortgage Note, shall constitute a demand obligation owing by the Mortgagor to the party incurring such costs and expenses and shall draw interest at the rate of eighteen percent (18%) per annum, all of which shall constitute a portion of the Indebtedness.
- 5.3 Renewals, Amendments and Other Security. Renewals and extensions of the Indebtedness may be given at any time and amendments may be made to agreements relating to any part of such Indebtedness or the Mortgaged Property and the Bank may take or may now hold other security for its Indebtedness without notice to or consent of the Mortgagor. The Bank may resort first to such other security or any part thereof or first to the security herein given or any part thereof, or from time to time to either or both, even to the partial or complete abandonment of either security, and such action shall not be a waiver of any rights conferred by this Mortgage, which shall continue as a first lien upon the Mortgaged Property not expressly released until the Mortgage Note and all other Indebtedness secured hereby is fully paid.
- 5.4 Limitation on Interest. No provision of this instrument or of the Mortgage Note shall require the payment or permit the collection of interest in excess of the maximum permitted by law or which is otherwise contrary to law. If any excess of interest in such respect is herein or in the Mortgage Note provided for, or shall be adjudicated to be so provided for herein or in the Mortgage Note, the Mortgagor shall not be obligated to pay such excess.

- 5.5 Unenforceable or Inapplicable Provisions. If any provision hereof or of the Mortgage Note is invalid or unenforceable in any jurisdiction, the other provisions hereof or of the Mortgage Note shall remain in full force and effect in such jurisdiction, and the remaining provisions hereof shall be liberally construed in favor of the Bank in order to effectuate the provisions hereof, and the invalidity of any provision hereof in any jurisdiction shall not affect the validity or enforceablility of any such provision in any other jurisdiction.
- 5.6 Advances by the Bank. Each and every covenant herein contained shall be performed and kept by the Mortgagor solely at the Mortgagor's expense. If the Mortgagor shall fail to perform or keep any of the covenants of whatsoever kind or nature contained in this instrument, the Bank or any receiver or keeper appointed hereunder may, but shall not be obligated to, make advances to perform the same in the Mortgagor's behalf, and the Mortgagor hereby agrees to repay such sums upon demand plus interest at the rate of eighteen percent (18%) per annum or, in the event any promissory note evidences such indebtedness, upon the terms and conditions thereof. No such advance shall be deemed to relieve the Mortgagor from any default hereunder.
- 5.7 Rights Cumulative. Each and every right, power and remedy herein given to the Bank shall be cumulative and not exclusive; and each and every right, power and remedy whether specifically herein given or otherwise existing may be exercised from time to time and so often and in such order as may be deemed expedient by the Bank, and the exercise, or the beginning of the exercise, of any such right, power or remedy shall not be deemed a waiver of the right to exercise, at the same tine or thereafter, any other right, power or remedy. No delay or emission by the Bank in the exercise of any right, power or remedy shall impair any such right, power or remedy or operate as a waiver thereof or of any other right, power or remedy then or thereafter existing. The Bank may resort to any security given by this Mortgage or to any other security now existing or hereafter given to secure the payment of the Indebtedness secured hereby or any in lebtedness secured by a pledge of the Mortgage, in whole or in part, and it such portions and in such order as may seem best to the Bank in its sole discretion, and any such action shall not in anywise be considered as a waiver of any of the rights, benefits, or liens evidenced by this instrument. The maximum amount for which this Mortgage shall be deemed to secure the obligations of the Mortgager as herein stipulated to reimburse the Bank for all costs, expenses and fees of whatever kind or nature incurred by Mortgagee hereunder is hereby fixed at twenty percent (20%) of the original principal amount of the Mortgage Note.
- 5.8 Waiver by the Bank. Any and all covenants in this instrument may from time to time by instrument in writing signed by the Bank be waived to such extent and in such manner as the Bank may desire, but no such waiver shall ever affect or impair the Bank's rights or liens hereunder, except to the extent specifically stated in such written instrument.
- 5.9 Successors and Assigns. This instrument is binding upon the Mortgagor, the Mortgagor's successors and assigns, and shall inure to the benefit of the Bank, its successors and assigns, and the provisions hereof shall likewise be covenants running with the land.
- 5.10 Article and Section Headings. The article and sec ion headings in this instrument are inserted for convenience of reference and shall not be considered a part of this instrument or used in its interpretation.
- 5.11 Construction. This instrument is in all respects to be construed under the laws of the State of Louisiana as a special mortgage, hypothecation, pledge and assignment and confession of judgment by the Mortgagor in favor of and for the benefit of the Bank, to secure the payment of all Indebtedness and the performance of all obligations of the Mortgagor.
- 5.12 Waiver. The parties hereto waive the production of any mortgage, conveyance, and tax certificates and agree to hold the undersigned Notary harmless in the pr
- 5.13 Notices. Any notice, request, demand or other instrument which may be required or permitted to be given or served upon the Mortgagor shall be sufficiently given when mailed by

first class mail, addressed to the Mortgagor at the address shown below its signature at the end of this instrument or to such different address as the Mortgagor shall have designated by written notice received by the Bank.

gras & &

Now, Personally Intervenes Jerome W. Gales, an individual residing in New York, New York, acting on behalf of any future holder or holders of the Mortgage Note and hereby accepts this Mortgage.

THUS DONE AND PASSED, in multiple originals, before me, the undersigned Notary Public, in and for the County of New York, State of New York, in the presence of the undersigned competent witnesses, who have hereto signed their names with said appearers and me, said Notary Public, after due reading of the whole, on the seventeenth day of February, 1983.

MESA PETROLEUM Co.

By John !

John F. Boros, Vice President

[Corporate Seal]

ATTEST:

Michael J. Moore, Assistant Secretary

Address: One Mesa Square Amarillo, Texas 79189

Jerome W. Gates, Interveno

Norman D. Morrow

WITNESSES TO ALL SIC NATURES:

Hayante St

In and for the State of New York, Commission Expiring

MARGARITA DEUS
MARGARITA DEUS
Notary Public, Stare of New York
0.31—4605970
Qualified in New York County
Commission Expires March 30, 1983

The instrument was prepared by:

R. S. Pieringer, Attorney Liddell, Sapp, Zivley, Brown & LaBoon 3400 Texas Commerce Tower Houston, Texas 77002

EXHIBIT A-1

0

Act of Collateral Mortgage, Collateral Chattel Mertgage, Pledge and Assignment of Production

CERTIFIED RESOLUTIONS OF MESA PETROLEUM CO.

I hereby certify that attached hereto is a multiple original of certified resolutions of MESA PETROLEUM Co. (the "Company"), certified by the Assistant Secretary of the Company, and attached to an Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production (Louisiana), dated February 17, 1983 and executed by the Company, passed before me this seventeenth day of February, 1983.

Notary Public In and for the State of New York, Commis-

sion Expiring

MARGARITA DEUS:
Notary Public, State of New York
No. 31—4605970
Qualified in New York County
Commission Expires March 30, 1983

EXHIBIT A-2

Act of Collateral Mortgage, Collateral Chattel Mortgage, Piedge and Assignment of Production

Certificate of Mesa Petroleum Co.

- I, the undersigned, Assistant Secretary of Mesa Petroleum Co., a Delaware corporation (the "Company"), Do Hereby Certify that:
- 1. Attached hereto as Exhibit A is a true, correct and complete copy of resolutions duly adopted at a meeting of the Board of Directors of the Company, convened and held on the 15th day of February, 1983, which resolutions have not been revoked, modified, amended or rescinded and are still in full force and effect, and the Amended and Restated Credit Agreement, the Notes, the Pledge Agreement and the Mortgages are in substantially the forms of those documents submitted to and approved by the Board of Directors of the Company.
- 2. The person named in Exhibit accepted hereto has been duly elected, has duly qualified as and at all times since April 28, 1982 (to and including the date hereof) has been an officer of the Company holding the office set forth therein opposite his name, and the signature set forth therein opposite his names is his genuine signature.

WITNESS my hand and the seal of the Company this seventeenth day of February, 1983.

Michael J. Moore, Assistant Secretary

[Corporate Seal]

EXHIBIT A to Certificate of Mesa Petroleum Co.

Resolutions of the Board of Directors of Mesa Petroleum Co.

WHEREAS, there has been presented to this meeting: (i) a form of amended and restated credit agreement (the "Credit Agreement"), among this Corporation, the banks parties thereto (collectively, the "Banks"), the managers parties thereto (in such capacities, collectively, the "Managers"), Texas Commerce Bank National Association ("TCB"), as administrative agent for the Banks (in such capacity, the "Administrative Agent"), Citibank, N.A., as collateral agent for the Banks (in such capacity, the "Collateral Agent"), and TCB and Citibank, N.A., as agents for the Banks (in such capacities, collectively, the "Agents" and individually, an "Agent"), providing for the making by the Banks of Credit A Loans (as defined in the Credit Agreement) to this Corporation in an aggregate principal amount not to exceed the Total Credit A Commitment (as defined in the Credit Agreement and not to exceed \$525,000,000). providing for the making by the Banks of Credit B Loans (as defined in the Credit Agreement) to this Corporation in an aggregate principal amount not to exceed the Total Credit B Commitment (as defined in the Credit Agreement and not to exceed \$325,000,000), and providing for the making by the Banks of Credit C Loans (as defined in the Credit Agreement) to this Corporation in an aggregate principal amount not to exceed the Total Credit C Commitment (as defined in the Credit Agreement and not to exceed \$325,009,000); (ii) a form of pledge agreement (the "Pledge Agreement") between this Corporation and the Collateral Agent, pursuant to which this Corporation will pledge to the Collateral Agent, for the ratable and proportionate benefit of the Banks and in accordance with the terms of the Credit Agreement, all of the certificates, instruments and documents representing or evidencing Shares (as defined in the Credit Agreement) to be pledged pursuant to the provisions of the Credit Agreement, for the purpose of securing repayment of the Credit C Loans (as defined in the Credit Agreement), the proceeds of which are used to purchase Shares (as defined in the Credit Agreement); and (iii) forms of mortgage and collateral mortgage documents (including a Mortgage, Deed of Trust, Assignment of Production and Security Agreement, an Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production, a Collateral Mortgage Note, and an Act of Pledge) (collectively, the "Security Documents") from this Corporation to the Collateral Agent and trustees for the Collateral Agent, pursuant to which this Corporation will, in accordance with the terms of the Credit Agreement, mortgage to the Collateral Agent, for the ratable and proportionate benefit of the Banks, certain oil and gas properties, for the purpose of securing repayment of the Credit A Loans and Credit B Loans;

Now, THEREFORE, BE IT RESOLVED, that the President or any Vice President of this Corporation, and each of them, be and he hereby is authorized to execute, in the name and on behalf of this Corporation, and deliver, a credit agreement among this Corporation, the Banks, the Administrative Agent, the Collateral Agent, the Agents and the Managers, substantially in the form of the Credit Agreement presented to this meeting, except for such changes, additions and deletions as to any or all of the terms and provisions thereof as the officer executing the Credit Agreement on behalf of this Corporation shall deem proper, such execution by such officer of the Credit Agreement to be conclusive evidence that such officer deems all of the terms and provisions thereof to be proper;

FURTHER RESOLVED, that the President or any Vice President of this Corporation, and each of them, be and he hereby is authorized to borrow from time to time on behalf of this Corporation the amounts permitted or provided to be borrowed by this Corporation under the Credit Agreement executed by this Corporation pursuant to these resolutions, and to execute and deliver on behalf of this Corporation Credit A Notes, Credit B Notes and Credit C Notes (all

as defined in the Credit Agreement) payable to the order of each of the Banks, substantially in the forms provided for in exhibits to the Credit Agreement, evidencing such borrowings;

FURTHER RESOLVED, that the President or any Vice President of this Corporation, and each of them, be and he hereby is authorized to execute, in the name and on behalf of this Corporation, and deliver, a pledge agreement between this Corporation and the Collateral Agent substantially in the form of the Pledge Agreement presented to this meeting, and a financing statement substantially in the form presented to this meeting, except for such changes, additions and deletions as to any or all of the terms and provisions thereof as the officer executing the Pledge Agreement or such financing statement on behalf of this Corporation shall deem proper, such execution by such officer of each of the Pledge Agreement and such financing statement to be conclusive evidence that such officer deems all of the terms and provisions thereof to be proper;

FURTHER RESOLVED, that the President or any Vice President of this Corporation, and each of them, be and he hereby is authorized to execute, in the name and on behalf of this Corporation, and deliver, one or more mortgages, deeds of trust, assignments of production, security agreements, financing statements and collateral mortgage documents from this Corporation to the Collateral Agent and trustees for the Collateral Agent, substantially in the forms of the Security Documents presented to this meeting, except for such changes, additions and deletions as to any or all of the terms and provisions thereof as the officer executing the Security Documents on behalf of this Corporation shall deem proper, such execution by such officer of the Security Documents to be conclusive evidence that such officer deems all of the terms and previsions thereof to be proper;

FURTHER RESOLVED, that any and all acts, transactions, or agreements undertaken prior to the date of these resolutions by any of the officers or representatives of this Corporation, in the name and on behalf of this Corporation, in connection with any of the foregoing matters, are hereby ratified, confirmed, and adopted by this Corporation; and

FURTHER PESOLVED, that each and every officer of this Corporation be and he hereby is authorized in the name and on behalf of this Corporation from time to time to take such actions and to execute and deliver such certificates, instruments, notices and documents as may be required or as such officer may deem necessary, advisable or proper in order to carry out and perform the obligations of this Corporation under the Credit Agreement executed by this Corporation pursuant to these resolutions, or under any other instrument or document executed pursuant to or in connection with the Credit Agreement; all such actions to be performed in such manner, and all such certificates, instruments, notices and documents to be executed and delivered in such form, as the officer performing or executing the same shall approve, the performance or execution thereof by such officer to be conclusive evidence of the approval thereof by such officer and by this Board of Directors.

EXHIBIT B to Certificate of Mesa Petroleum Co.

Name of Officer

John F. Boros

Office Vice President

Signature

EXHIBIT A-3

Act of Collateral Mortgage, Collateral Chattel Mortgage, Piedge and Assignment of Production

COLLATERAL MORTGAGE NOTE

\$3,060,000,000

February 17, 1983

FOR VALUE RECEIVEL, on demand, the undersigned, MESA PETROLFUM (, a Delaware corporation, promises to pay to the order of Bearer at the office of Citibrok N.A. at 599 Park Avenue, New York, New York, the sum of THREE BILLION 10-1 (\$3,000,000,000) with interest at the rate of eighteen percent (18%) per annum from the late until paid.

In the event this note should be placed in the hands of the atternity, after its maturity, to institute legal proceedings to recover the amount hereof, or any place ereof, in principal or interest to protect the interests of the holder hereof, or in the event the same should be placed in the hands of an attorney for collection, compromise or other action, the makers bind to resolves to pay the fees of the attorney who may be employed for that purpose, which fees are hereby fixed at ten percent (10%) of the amount due or such for, or claimed or sought to be to tested, preserved or enforced.

The parties hereto, whether as original signers or L guaranty or endorsement or thriwise, hereby waive present one for payment, demand, notice of non-payment, procest, and all pleas of division and discussion, and agree that the time of payment increof may be extended from time to time, one or more times, without notice of such extension or extensions and without previous consent, hereby binding themselves in solido, uncontritionally and as original promissors, for the payment hereof a principal, interest, costs and attorneys' fees.

This note shall be construed in accordance the and governed by the laws of the State of Louisiana

| MESA | PETROLEU | м С |), | | | |
|------|----------|------|----|-------|------|----------|
| 0 | | | | | | |
| Ву | | Lahn | 17 | Rorus | Vice | Presider |

NE VARIETUR

For identification with an Act of Collateral Mortgage, Collateral Chattel Mortgage, Pledge and Assignment of Production, executed by MESA PETROLEUM Co., passed before me this seventeenth day of February, 1983.

| | | | | | | | | Nota | ry Publ | ıc |
|------|----|-------|-----|-----|-------|----|-----|-------|---------|----|
| | In | and | for | the | State | of | New | York, | Coinmis | 3- |
| SIUT | E: | cpiri | ng | | | | | | | |

EXAIDIT B

10

Act of Collatoral Mortgage, Collateral Chatter Mortgage, Pledge and Assignment of Production

PROPERTY DESCRIPTIONS

This Exhibit B contains the Jescr point of the "oil ras leases" as defined in the Act of Collateral Mortgage, Collateral Control Mortgage, Pleign and Assignment of Production to which this Exhibit B is attached.

The oil and gas leases are expressly limited inscar and cally insofar as they cover land and der it intervals in which Mortgagor owns or holds an interest and do not include land and der it intervals in which Mortgagor owns no interest.

As used in this Exhibit B. to a term "working interest specifies the leasehold or operating interest (expressed as a decimal 'raction) of Mortgagor in an oil and gas lease, and the term "net interest in production. The fies the interest (expressed as a decimal fraction) of Mortgagor in the production of Hydro arbons from an oil and gas lease after deducting all applicable royalties, overriding revalues and other burdens on production.

1. Oil and Gas Lease OCS-1, 2117, effective February 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering Block 333 Fugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A.

Working Interest .13333 Net Interest in Production .11111

Oi! and Gas Lease effective July 19, 1971, granted by the State of Louisiana, as Lessor. to Mesa revoleum Co., et al., as Lessee, covering entire Tract No. 11960 located in St. Martin and St. Mary Parishes, Louisiana, described as follows: the lands now or formerly constituting the beds and bottoms of all water bodies of every nature and description and all islands and other 'ands formed by accretion or reliction, except tax lands, owned by and not under mineral lease from the State of Louisiana on April 29, 1971 situated in St. Martin and St. Mary Parishes, Louisiana, within the following described boundaries: Peginning at a point having Lambert Plane Coordinates of X = 1,966,700.00 and Y = 446,600.00, being the Northwest corner of the tract herein described; thence East 16,000,00 feet; thence South 13,600,00 feet to a point havin, Lambert Plane Coordinates of X = 1,982,700.00 and Y = 433,000.00; thence West 16,000.00 feet; thence North 13,600.00 feet to a point of beginning, estimated to contain approximately 4.620 acres, all more full shown outlined in red on a plat on file in the State Land Office. All bearings are based on Louisiana Lambert Plane Coordinate System (South Zone), recorded in Book 652, Folio 115, Entry No. 1'4449 of the Conveyance Records of St. Martin Parish, Louisiana and recorded in Book 16-2, Entry No. 147,217 of the Conveyance Records of St. Mary Parish, Louisiana.

Working Interest .18750
Net Interest in Production .13281

3. Oil and Gas Lease OCS-G 3286, effective September 1, 1975, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 613, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 18.

Working Interest .18400 Net Interest in Production .15333

4. Oil and Gas Lease OCS-G 2591, effective May 1, 1974, granged by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 146, South Marsh Island Area, South Addition, Official Leasing Map, Louisiana Map No. 3C.

> Working Interest Net Interest in Production

.16000 .13333

5. Oil and Gas Lease OCS-G 2810 . . covive December 1, 1974, granted by the United States of America, as Lessor, to Mesa P. releum Co., et al., as Lessee, covering all of Block 327, Eugene Island Area, South Addition, CCS Official Leasing App, Louisiana Map No. 4A.

> Working Interest Net Interest in Production

.11500

.09583

6. Oil and Gas Lease OCS-G 2912, effective December 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 329, Eugene Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 4A.

Working Interest

.22500

Net Interest in Production

.18749

7. Oil and Gas Lease OCS-G 3141, effective July 1, 1975, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 397, Verbalion Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 3B.

> Working Interest Net 1 cerest in Production

b" - "

.49999

This property is subject to the Advance Payment Agreement dated December 16, 1975, as amended February 19, 1376 and May 10, 1977, between Mess Petroleum Co., as Seller, and Michigan Wisconsin Tipe Line Company, as Euyer.

8. Oil and Gas, er se OCS-G 2185, effective October 1, 1972, granted by the United States of America as Lessor, as Mess Petroleum Co., et al., as Lessee, covering Block 78, that portion which is more than 3 geogra, 'inal miles seaward from the line described in paragraph 1 of Supplemental Decree of U.S. Supreme Court entered 1: 13-65 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288) South Pass Area, South and East Addition, as shown on Official Leasing Map, Louisiana Map No. 9A.

Working Inte est

.25000

20833

Net Interest in Production

This property is subject to the Exploration and Development Loan Agreement dated ecember 19, 1973, as a nende i December 12, 1974, February 28, 1975 and June 14, 1977, be eveen Mesa Petroleum Co. and Natural Gas Pipeline Company of America.

9. Oil ar Gas Lease OCS-G 3299, effective September 1, 1975, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al. as Lessee, covering all of Block 263, East Cameron A sa, South Addition, as shown on OCS Official Leasing Map, I cuisias Map No. 2A

> Working Interest Net Interest in Production

.10600

.0.3333

This property is subject to the Mortgare, Deed of Trust, Assignment of Poduction, Security Agreement and Financing Statement dated January 21, 1977, between Mesa retrologr Co. as Mortgagor, and McMoRan Experation Co. and to the Act of Mortgage, Ple Ige and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Juner, and McMoRan Exploration Co.

10. Oil and Gas Lease OCS-G 3388 effective January 1, 1977, granted by the United Stones of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 336, East Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 2A.

Working Interest

.05000

Net Interest in Production

.04167

This property is subject to the Mortgage, Treed of Trust, Assignment of Production, Security Agreement at 1 Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortg yor, and McMoRan Exploration Co. and to the Act of Mortgage, Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Owner, and McMoRan Exploration Co.

11. Oil and Gas Lease OCS-G 3390, effective January 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 25, Vermilion Area, as shown on OCS Official Leasing Map, Louisiana Map No. 3.

Working Interest

.04737

Net Interes in Production

.03947

This property is subject to the Mortgage, Deed of Trust, Assignment of Production, Security Agreement and Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortgagor, and McMoRan Exploration Co. and to the Act of Mortgage, Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., us Owner, and McMoRan Exploration Co.

12. Oil and Gas Lease OCS-G 3400, effective January 1, 1977, granted by the United States of America, at Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 310, Vermilion Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map, No. 35.

Working Interest

.05000

.04157

Net Interest in Production

This property is subject to the Mortgage, Deed of Trust, Assignment of Production, Security Agreement and Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortgagor, and McMoRan Exploration Co. and to the Act of Mortgage, Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Owner and McMoRan Exploration Co.

10 Oil and Gas Lease CS-G 3414, effective January 1, 1977, granted by the United States of America, as Lessoc, to Mona Petroleum Co., et al., as Lessee, covering the N/2 of Block 34, West Delta Area, as the wn on OCS Official Leasing Map, Louisiana Map No. 8.

Working Interess.

.05000

Ne allerest in 1 reduction

.04167

This property is subject to the Mortgage, Deed of Trust, Assignment of Production, Security Agreement and Financing Statement dated January 21, 1977, between Mesa Petroleum Co., as Mortgagor, and McMoRan Exploration Co. and to the Act of Mortgage, Pledge and Assignment of Production dated January 21, 1977, between Mesa Petroleum Co., as Owner, and McMoRan Exploration Co.

14. Oil and Gas Lease OCS-G 3589, effective August 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum. Co., et al., as Lessee, covering all of Block 18, South Pelto 1122, as shown on OCS Official Leasing Map, Louisiana Map No. C

.26178

.21815

Working Interest Net Interest in Production

15. Oil and Gas Lease OCS-G 2271, effective February 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering Block 348, Vermilion Area, South Addition, on OCS Cificial Leasing Map, Louisiana Map No. 3B.

Working Interest .75000
Net Interest in Production .62499

16. Oil and Gas Lease No. 6310 effective May 13, 1974, granted by the State of Louisiana, as Lessor, to Mesa Petroleum Co., et al. as Lessee, covering those portions of Blocks 57 and 58, South Pass Area, Playmenines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974, described as follows:

Beginning at the Northwest corner of Block 57, South Pass Area, having Lambert Plane Coordinates of X = 2,615,928.00 and Y = 82,150.00; thence East 9,685.54 feet along the North line of said Block 57 to a point 1 aving Lambert Plane Coordinates of X = 2,625,613.54 and Y = 82,150.00; thence So thwesterly in a straight line to a point having Lambert Plane Coordinates of X = 2.624,340.00 and Y = 80,576.00; thence Southwesterly along an arc which has a radius of 18,240.60 feet and a vertex having Lambert Piane Coordinates of X = 2,610,160.00 and Y = 92,050.00 to a point having Lambert Plane Coordinates of X = 2.621,555.00 and Y = 77,806.00: thence Southwester! in a straight line to a point having Lambert Plane Coordinates of X = 2,621,180.00 and Y = 77,506.00; thence Southwesterly along an arc which has a radius of 18,240.60 feet and a vertex having Lambert Plane Coordinates of X = 2,609,785.00 and Y = 91,750.00 to a o at having Lambert Plane Coordinates of X = 2,617,996.00 and Y = 75,462.00; thence Southwes only in a straight line to a point having Lambert Plane Coordinates of X = 2,617,391.00 and Y = 75,157.00; thence Westerly: long an arc which has a radius of 18,240.60 feet and a vertex having Lambert Plane Coordinates of X = 2,609,180.00 and Y = 91,445.00 to a point having Lambert Plane Coordinates of X = 2,610,702.97 and Y = 73,268.09; thence North 8,881.91 feet to a point of intersection with the North line of Block 58, South Pass Area; thence East 5,225.03 (set along the North line of said Block 58 to its Northeast corner, the same being the point of beginning, estimated to contain approximately 2,026.40 acres, all more fully shown outlined in red on a plat on file in the State Land Office. All bear ngs are based on Louisiana Lambert Plane Coordinate System (South Zone). Reference is made to a survey line described in a Fourth (4th) Supplemental Decree, entered on the 16th day of October, 1372, in that matter style. United States of America v. State of Louisianc, et al., No. 9 Original, on the docket of the Submeme Court of the United States, it is expressly understood that although a segment of that line is incorporated in the description of the property herein leased, that the State of Louisiana is leasing on that portion of the above described block or blocks lying more than one (1) foot landwar. : that line, all as provided for in said Fourth (4th) Supplemental Decree. Note: Portions of this tract are located within the restrictive area as set out by the U.S. Coast Guard and the Corps of Engineers, U.S. Army

Working Interest .33333
Net Interest in Production .24666

This property is subject to *1 e Act of Vortgage and Security Agreement to Secure Future Advances dated November 22, 1. bet ween 1 Mesa Petroleum Co., as Mortgagor, and Columbia Gas Transmission Corporation, as Mortgagee.

17. Oil and Gas Lease OCS-G 3416, effective January 1, 1977, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering that portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975, (United States v. Louisiana), 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 366, December 20, 1971), as showr a OC5 Official Leasing Map. Louisiana Map No. 9.

Working Interest

.33333

Net Interest in Production

.27777

This property is subject to the Act of Mortgage and Security Agreement to Secure Future Advances dated November 22, 1978, between Mesa Petroleum Co., as Mortgagor, and Columbia Gas Transmission Corporation, as Mortgagee.

18. Oil and Gas Lease OCS-G 2619, effective May 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 115, Ship Shoal Area, Official Leasing Map, Louisiana Map No. 5.

Working Interest

.20000

Net Interest in Production

.16667

19. Oil and Gas Lease OCS-G 2887, effective December 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 173, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C.

Working Interest

Net Interest in Production

.039417

20. Oil and Gas Lease OCS-G 2888, effective December 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block 174, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C.

Working Interest

.047300

Net Interest in Production

.039417

21. Oil and Gas Lease OCS-C 2743, effective July 1 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-349, High Island Area, East Addition, South Extension, OCS Official Leasing Map, Texas Map No. 7C.

> Working Interest Net Interest in Production

.20000 .16667

 Oil and Gas Lease OCS-G 2421, effective August 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-330, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C.

> Working Interest Net Interest in Production

.18400 .15.332

23. Oil and Gas Lease OCS-G 2398, effective August 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-273, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas May No. 7C.

Working Interest

.16875

Net Interest in Production

.14062

This property is subject to the Advance Payment Agreement dated October 3, 1974, as amended July 19, 1976, July 23, 1976 and May 10, 1977, between Mesa Petroleum Co., as Seller, and Michigan Wisconsin Pipe Line Company, as Buyer.

24. Oil and Gas Lease OCS-G 2403, effective August 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-279, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C.

Working Interest .15000
Net Interest in Production .12499

This property is subject to the Advance Payment Agreement dated October 3, 1974, as amended July 19, 1976, July 23, 1976 and May 10, 1977, between Mesa Petroieum Co., as Seller, and Michigan Wisconsin Pipe Line Company, as Buyer.

25. Oil and Gas Lease OCS-G 2410, effective August 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-313, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C.

Working Interest .30670
Net Interest in Production .25558

This property is subject to the Advance Payment Agreement dated October 3, 1974, as amended July 19, 1976, July 23, 1976 and May 10, 1977, between Mesa Petroleum Co., as Seller, and Michigan Wisconsin Pipe Line Company, as Buyer.

26. Oil and Gas Lease OCS-G 2412, effective August 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-317, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C.

Working Interest .12270
Net Interest in Production .10225

This property is subject to the Advance Payment Agreement dated October 3, 1974, as amended July 19, 1976, July 23, 1976 and May 10, 1977, between Mesa Petroleum Co., as Seller, and Michigan Wisconsin Pipe Line Company, as Buyer.

27. Oil and Gas Lease OCS-G 2739, effective July 1, 1974, granted by the United States of America, as Les to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-339, High Island Area, E. Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C.

Working Interest .25000 Net Interest in Production .20833

This property is subject to the Advance Payment Agreement dated October 3, 1974, as amended July 19, 1976, July 23, 1976 and May 10, 1977, between Mesa Petroleum Co., as Seller, and Michigan Wisconsin Pipe Line Company, as Buyer.

28. Oil and Gas Lease OCS-G 2426, effective August 1, 1973, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-340, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C.

Working Interest .25000 Net Interest in Production .20833

This property is subject to the Advance Payment Agreement dated October 3, 1974, as amended July 19, 1976, July 23, 1976 and May 10, 1977, between Mesa Petroleum Co., as Seller, and Michigan Wisconsin Pipe Line Company, as Buyer.

29. Oil and Gas Lease OCS-G 2786, effective September 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering all of Block A-315, High Island Area, East Addition, South Extension, as shown on OCS Official Leasing Map, Texas Map No. 7C.

> Working Interest Net Interest in Production

1.00000 .83333

This property is subject to the Advance Payment Agreement dated October 3, 1974, as amended July 19, 1976, July 23, 1976 and May 10, 1977, between Mesa Petroleum Co., as Seller, and Michigan Wisconsin Pipe Line Company, as Buyer, to the Purchase Agreement Development Production Payment dated June 6, 1980, as amended December 12, 1980 and November 19, 1981, between Mesa Petrcleum Co., as Seller, and 110 North Wacker Drive Foundation, Inc., as Buyer, to the Conveyance of Development Production Payment dated June 6, 1980, as amended December 12, 1980 and November 19, 1981, between Mesa Petroleum Co., as WI Owner, and 110 North Wacker Drive Foundation, Inc., as PP Owner, to the Purchase Agreement Development Production Payment dated November 19, 1981, between Mesa Petrolcum Co., as Seller, and 110 North Wacker Drive Foundation, Inc., as Buyer, and to the Conveyance of Development Production Payment dated November 19, 1981, between Mesa Petroleum Co., as WI Owner, and 110 North Wacker Drive Foundation, Inc., as PP Owner.

30. Oil and Gas Lease OCS-G 2607, effective May 1, 1974, granted by the United States of America, as Lessor, to Mesa Petroleum Co., et al., as Lessee, covering Block 312, Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A.

Working Interest

.13333

Net Interest in Production

.11111

LAW OFFICES

LEMLE, KELLEHER, KOHLMEYER & MATTHEWS

EIGHTEETHTH FLOOR

LOUIS C. LEMIE
HANDS D. FILLTHEE
HANDS D. FILLTHEE
HANDS S. SOUCHIN
CHARGES BOUGHISH D. JT
WILLIAM RITING HE
CHARGES E CHICKNOTH
C. MODILLY FILLTH
DAVID AS FELLIN TI
FORM S. LICENSTRUM
JOHN S. LICENSTRUM
JOHN S. LICENSTRUM
JOHN S. CHUNCHELL
DAVID C. CAM DELL
DAVID C. CAM DELL
JULIAN B. DECAL
DAVID C. CAM DELL
JULIAN B. DECAL
DAVID C. CAM DELL
JULIAN B. D. G. CD
BICCIAM S. J. C. NICA

JAMES IN CROSSET
FREDERINGS IN PRINCIPLE
WILLIAM FOR THE STORM JR.
WILLIAM JR.

FIRST NATIONAL BANK OF COMMERCE BUILDING

NEW ORLEAMS 70112

TELEPHONE (04) 500-1241

CABLE ADE (155: LEMMOR TELE) 500272

November 4, 1977

MACK E. CATHEM
OF CHURCE

DON M MICHARD
GEORGE FRAZIER. LY
MAIN S. STEIN
MICHART W. STMMUS.S
TOWN M SARTH, JR
MICHART B. FOSTEN
MICHARL A M. GEONE
TERRITAL J. ELSTELL
ALLAN A. MARI, JR

CHARLES F. THENSTED HALLS T. HUDBANDLIT STEPRY MANCUS WISE PAL C. VELTH STEWART I. PECK DAVID S. RELLY DCUGLAS F. MATTHEWS S. HODDER WELATEN, JR. JAMES H. PROWNLYE

Department of the Interior Bureau of Land Management Outer Continental Shelf Office 500 Camp Street New Orleans, Louisians 70130

Attention: Mr. John L, Fankin, Offshore Manager

Re: Leases Nos. OCS-G2839, 2224, 2225, 3284, 2228, 2850, 2550, 2859, 2860, 2882, 2883, 2587, 2864, 2885, 2900, 3156, 2007, 2317, 2177, 2125, 3193, 3195, 2947, 2193, 3202, 3203, 3206, 3417, 3409, 3416, 3500, Offshore Louisiana

Leases Nos. OCS-G2358, 2366, 2372, 2373, 3118, 2377, 2378, 2583, 2701, 2704, 2779, 2705, 2706, 2384, 2713, 2388, 2369, 2390, 2718, 2719, 2397, 2398, 2399, 2403, 3314, 2409, 2411, 2737, 2413, 2414, 2416, 2418, 2422, 2739, 2426, 2429, 2745, 2746, 2766 Offshore Texas

Gentlemen:

Enclosed are an original executed copy and 70 xerox copies of each of the following documents:

- Amendment to Production Payment Agreement dated as of November 1, 1977 between Pennzoil Company and Pennzoil Louisiana and Texas Offshore, Inc.
- Amendment to Instrument of Conveyance of Production Payment dates as of November 1, 1977 between Pennzoil Louisiana and Texas Offshore, Inc. and Pennzoil Company.

LEMLE, KELLEHER, KOHLMEYER & MATTHEWS

Department of the Inserior Bureau of Land Management Page 2 November 4, 1977

> Confirmation of Assignment dated as of November 1, 19/7 among Pennzoil Louisiana and Texas Offshore, Inc., Pennzoil Company and The First National Bank of Chicago, as Agent for certain banks.

We request that you file the original executed copy of each of these documents, along with this letter, in your file applicable to Lease No. OCS-G2839. Please also place an additional copy of this letter and a xerox copy of these documents in your file applicable to each of the remaining captioned leases for cross-referencing purposes.

We understand that the payment of a fee in connection with this filing is not required by applicable regulations, and that no approval is necessary. Please return one enclosed copy of this letter to evidence such filing and the accomplishment of our request.

Yours very truly,

June, The Bridge

James M. Petersen

JMP/dca

Enclosures

Filed and accomplished as requested.

NUV 8, 1977.

John L. Rankin, Manager

PENNZOIL LOUISIANA AND TEXAS OFFSTIORE, INC.

CONFIRMATION OF ASSIGNMENT

THIS CONFIRMATION OF ASSIGNMENT dated as of November 1, 1977 (the "Confirmation"), between Fennzod Louisiana and Texas Offsbore, Inc. (the "Company"), and The First National Bank of Chicago, as Agent for the Banks mentioned below (the "Agent"),

WITNESSITH THAT:

WHEREAS, the Company entered into a Baul. Credit Agreement (the "Credit Agreement"), dated as of March 1 1974, with The First National Baul, of Chicago, the other bauls referred to in Section 2 of the Credit Agreement and the Agent providing for local (the "Loans") to the Company by the Bauls which are parties to the Credit Agreement to an expregate principal amount not in excess of \$100,000,000, to be evidenced by the promasory notes of the Company (the "Notes").

Winter it, the Company executed, acknowledged and delivered to Pennzoil Company ("Pennzoil") a conveyance of prediction payment dated as of March 1, 1972, substantially in the form of Appendix 1 (the "Conveyance") to that certain Production Payment Agreement, dated us of March 1, 1974, between Pennzoil and the Company (the "Production Payment Agreement"), all as provided in Section 2.5 of the Production Payment Agreement;

Winnexs, in order to secure the full and complete performance by the Company of its obligations under the Credit Agreement, the Company agreed therein to assign its rights and options in, to and under the Production Payment Agreement to cause Pennzoil credit (a) to pay the Company, as the purchase price for the production payment conveyed by the Company to Pennzoil as aforeisid, an amount equal to the principal amount of such production payment, which principal amount shall equal the aggregate principal amount of the loans, plus accuract interest thereon and commitment fee applicable thereto, if any, outstanding under the Credit Agreement on the date, if any, upon which such payment is made, or (by to execut, a landwedge and deliver to the Company, at its expense, such instrument or instruments as may be necessary or appropriate to release, reliaquish and extinguish as fully discharged, such production payment;

Wheneas, in order to induce the Banks which are parties to the Credit Agreement to make loans pursuant to the Credit Agreement and to secure the performance by the Company of its obligations under the Credit Agreement, including its obligation to pay the Notes (the "Credit Agreement Obligations"), the Company executed, acknowledged and delivered to the Agent an Assignment dated as of March 1, 1974 (the "Assignment"), a conformed copy of which is attached hereto as Appendix I (in which Pennzoil joined for the limited purposes set forth therein), by which the Company assigned, transferred and set over to the Agent for the proportionate benefit of the Banks which are parties to the Credit Agreement as their interest may appear all the right, title and interest of the Company in, to and under the Production Payment Agreement, save and except, however, those rights of the Company relating to the sale of gas produced from the Leaachold Interests provided for in Article III of the Production Payment Agreement (such right, title and interest of the Company in the Production Payment Agreement, save and except such rights relating to the sale of gas, being herein and in the Assignment referred to as the "Production Payment Rights"), together with the right, title and interest of the Company in, to and under any amendment of the Production Payment Rights or additions thereto;

Whereas, the Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of May J. 1976 to reflect an extension of the maturity of the credit available to the Company under the Credit Agreement, an increase in the amount of that credit to \$260,000,000 and the addition of Manufacturers Hanover Trust Company, Security Pacific National Bank, Wells Fargo Bank Nat and Association, The Northern Trust Company, European-American Bank & Trust Company and The First National Bank of Boston as parties to the Credit Agreement, and for other purposes;

Whenever, in connection with the aforestid amendment and restatement of the Credit Agreement, the Production Payment Agreement and the Conveyance; the Assignment was confirmed by the Confirmation of Assignment dated as of May 1, 1976 between the Company and the Agent (in which and effective to as of the Assignment and to provide that all references in the Assignment as confirmed thereby to the Credit Agreement, the Production Payment Agreement or the Consequence would be decided to the Banks would be defined to refer to such in trainents as annualled and revisited as of May 1, 1976 and that all references amended and restated.

Wind East, the Credit Agreement the Production Payment Agreement and the Conveyance have now been further amend a and restated as of November 1, 1977 to reflect an extension of the maturity that could be \$270.000.000 and the addition of Morgaer Guaranty Trust Company of New York and First International Earth or Mouston, N.A. as parties to the Credit Agreement, and for other purposes;

Witterss, persuant to the acquest of the Company and the Banks which are parties to the Credit Agreement as amended and reseated as at November 1, 1977 as aforesaid (the "Banks"), Pennzoil proposes to join berein for the limited purposes set forth below:

Now, Theoreton: the Company and counsoil hereby again confirm the continuing validity and effectiveness of the Assignment, which here been heretofore made by the Company to enforce, seeme and provide an additional means of payment of the Credic Agreement Obligations. It is further agreed and uncl. stood that all inferences in the Assignment as confirmed hereby and in Conveyance shall be deemed to refer to such instruments as amended and restated as of November 1, 1977, and that all references in the Assignment as confirmed hereby and in the following parameters as so amended and restated as of November ment as so amended and restated. The Credit Agreement Obligations the payment of which is the Credit Agreement as associated to refer to the Bail's which are parties to the Credit Agreement by the Assignment as confirmed hereby shell include the obligations the payment of which is the Credit Agreement as assented and restated as of November 1, 1977.

The Assignment of Confirmed hereby, (i) shall remain in full force and effect until payment in all indebtedness and obligations of the Company to each of the Banks under and pursuant to Agreement and the Notes executed and which may be executed pursuant thereto, at which I Pennzoli, as their interests may appear, the rights assigned to the Banks hereunder as interest under the Assignment is no longer necessary to protect the Banks, the continuation of expense by reason of any claim that the payment neight constitute a preference or otherwise be successors and assigns, (ii) shall be binding upon the Company and Pennzoil, their respective Banks, their respective successors, transferees and assigns, and (iv) shall be deemed to have been made under and shall be governed by the laws of the State of Texas.

In WITNESS WHEREOF, the parties hereto have caused this Confirmation to be executed by their respective officers thereunto duly authorized as of the day first above written.

DENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

Title: Assistant Vice President

[SEAL]

By Wolf william

Title: Vic. President

Pennezor Company

Title: Group Vice President

The First National Bank of Chicago, as Agent

By Rechard S. Cellinge

THE STATE OF TEXAS COUNTY OF HARRIES SS:

BE IT REMEMBERED that I, the tradersigned, a Notary Public duly qualified, sworn and acting in and for the County and State - escad, hereby certify that, on this 1st day of November, 1977;

(Louisiana)

Before me appeared W. . Wilkinson, to me personally known, who, being by me duly sworn, did say that he is Vice Versident of Pennzon, Leuisiana and Texa. Oursider, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the exporate seal of said corporation and that the seal affixed to the foregoing instrument is the exporation by authority of its Board of Directors and that W. B. Williamson acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared Haroun E. Source we be personally known, who, being by needuly sworn, did say that he is a Group Vice President of P

(lexas)

Before me on this day per-tindly appeared W. B. Wirelisson, known to me to be the person whose name is subscribed to the foregone in frequent and known to me to be a Vice President of Pennzon. Louisiana and Texas Crissona. Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the parabase and consideration for rein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared Paronn E. See seed, become to be the person whose name is subscribed to the forecoing instance to ack known to not to be a Group Vice President of Pennzon. Company, a Delaware corporation, and acknowledged to a constant executed see instrument for the purposes and considerable in the primar pressed, and as the act and deed of said corporation.

IN WITNESS WHEREOF, I have become set my hand and official notarial seal, in the City of Houston, Councy of Harris, State of Texas, the lated cy of Nevermber, 1977.

Notary Public in and for Harris County, Tenas

My Commission Expires Sec. 1.30 1.778

ROSELLE ASHWORTH

Rotor V. M. in continue Harns County, Feros

The County of the South County of South

The County of the County of South County

The County of the County of the County of South County

The County of the County of

[NOTARIAL SEAL]

THE STATE OF ILLINOIS)
COUNTY OF COOK (ss

By IT REMEMBER that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this way of November, 1977;

(Louisiana)

(Texas)

Before me appeared Richard S. Palmus, to me personally known, who, being by me duly sworn, did say that he is an Assistant Vice President of Tur. First National Bank or Cincago, a national banking association, and that the scal affixed to the foregoing instrument is the experate scal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that he group S. Palmus acknowledged the instrument to be the free act and deed of the national banking association.

Before me on the day personally appeared Brewan S. Palmer, known to me to be the person whose name is substituted to the bisa going instrument and known to me to be an Assistant Vice President of Tim First National Basis on Cine was, a national banking association, and acknowledged to me that he executed said but ment for the purpose, and consideration therein expressed and in the capacity therein stated, and as the assumed deed of said national banking association.

In Wirnass Williams, I have here to set my hand and official notarial scal, in the City of Chicago, County of Co.k. State of Illings, this extends of November, 1977.

Clare A Martz

Clare Q

Notary Public in and for Cook County, Illinois
My Commission Expires

My Commission Expires April 9, 1979

[NOTAMAL SEAL]

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC. ASSIGNMENT

THIS ASSIGNMENT dated as of March I, 1974 (the "Assignment") from Pennzoil Louisiana and Texas Offshare I is (the 'Cor., ...') to The First National Bank of Chicago, as Agent for the Broke mentioned below (the 'Agent').

WITHINST IN THAT

Wheneve, the Company Las covered into a Bank Cardit Agreement (the "Credit Agreement"), duted as of March 1, 1974, with the First National Bank of Chicago, the other Banks referred to in Section 2 of the Credit Agreement (the "conks") and the Agent providing for bank (the "Loans") by the Banks to the Company is an aggregate principal amount not in excess of \$100,000,000, to be evidenced by the promissory notes of the Company (the "Notes");

WILLIEAS, the Company has executed, acknowledged and delivered to Pennzoil Company ("Pennzoil") a conveyance of production payment, dated as of March 1, 1974, substantially in the form of Appendix 1 to that certain Production Payment Agreement, dated as of March 1, 1974, between Pennzoil 1 the Company (the "Production Payment Agreement"), all as provided in Section 2.3 of the Production Payment Agreement;

Withheas, in order to secure the full and complete performance by the Company of its obligations under the Credit Agreement, the Company has regreat therein to assign its rights and options in, to and under the Production Payment Agreement to cause Pennzoil either (a) to pay to the Company, as the purchase price for the production payment conveyed by the Company to Pennzoil as aforesaid, an amount equal to the principal amount of such production payment, which principal amount shall equal the aggregate principal amount of the loans, plus accrued interest thereon and commitment fee applicable thereto, if any, outstanding under the Credit Agreement on the date, if any, upon which such payment is made, or (b) to execute, acknowledge and deliver to the Company, at its expense, such instrument or instruments as may be necessary or appropriate to release, relinquish and extinguish as fully discharged, such production payment; and

Whereas, pursuant to the request of the Company and the Banks, Pennzoil proposes to join herein for the limited purposes set forth below;

Now, Therefore, in order to induce the Banks to make loans pursuant to the Credit Agreement and to secure the performance by the Company of its obligations under the Credit Agreement, including its obligation to pay the Notes (the "Credit Agreement Obligations), the Company hereby assigns, transfers and sets over to the Agent for the proportionate benefit of the Banks as their interests may appear all of the right, title and interest of the Company in, to and under the Production Payment Agreement, save and except, however, those rights of the Company relating to the sale of gas produced from the Leasele of Interests provided for in Article III of the Production Payment Agreement, save and except such rights relating to the sale of gas, being hereinafter referred to as the "Production Payment Rights"), together with the right, title and interest of the Company in, to and under any amendment of the Production Payment Rights or additions thereto.

- This Assignment is made by the Company to enforce, secure and provide an additional means of payment of the Credit Agreement Obligations.
- 2. The Production Payment Rights may be exercised by the Agent for and on behalf of the Banks upon the occurrence of an Event of Default as described in Section 11 of the Credit Agreement and at any time during the continuance of such Event of Default. To exercise the right of the

Company to call upon Pennzoil to pay the purchase price of such production payment as provided in Section 2.4 of the Production Fayment Agreement (the "Furchase Price Right"), the Agent shall give Pennzoil notice to such effect which shall

- (a) specify the purchase price of the production payment (determined as provided in Section 2.1 of the Production Payment Agreement) to be paid by Pennzoil pursuant to such request, and
- (b) specify the date upon which the purchase price of such preduction payment is to be paid.

Concurrently with the giving of such notice to Penazoil, the Agyait shall deliver or mail an executed counterpart of such notice to the Company. Giving of such notice by the Ageas to Penazoil shall constitute an effective notice of election by the Company to exercise the Purchase Price Right as provided for in Section 2.5 of the Production Physiciant Agreement. The full amount of the purchase price of such production payment shall be paid by Fernazoil, without sololl, counterclaim or recomponent what over by reason of any claim of Penazoil against the Company in otherwise, directly to the agent on the date specifical in such notice and shall be applied by the Agent to the payment of the Notes in the manner described in Section 6 of the Credit Aga ement.

- 3. Notwithstanding this Assembly the Company and Perezoil shall have the right, without the consent of the Agent or the Earlis, to are ad the Production Payment Agreement and the conveyance of production payment executed pursuant thereto for the sole and limited purposes of releasing or adding to, one or more of the Leaschold Interests or subjecting the same to gas sale,, unit, operating to other agreements providing for the operation of development of the Leasenoid Interests or the deposition of production therefrom; provided, however, that any such amendment shall not otherwise modity in any respect the rights and obligations of the Company or Pennzoil underst the Production 1 s.c. in Agreement or reduce the purchase price of said production payment determined as provided in Section 2.1 of the Production Tajment Agreement.
- 4. No failure or delay on the part of the Agent or the Banks in exercising any power or right hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such aght or power preclude any other single or partial exercise thereof or the exercise of any other right or power hereunder.
- 5. At the represt of the Company and in order to induce the Banks to make the Loans to the Company, Pennzoil agrees
 - (a) with the Company that each certificate as to value of Proven Hydrocarbon Reserves delivered by the Company to the Agent persuant to Section δ(c) of the Credit Agreement shall be signed by Penuzoil to acknowledge its approval thereof and shall be deemed to have been delivered to Penuzoil pursuant to Section 2.4 of the Production Payment Agreement, and
 - (b) with the Agent and the Banks that the breach or failure of the Company to perform any of its obligations under this Assignment or the Production Payment Agreement shall not reduce, impair or adversely affect in any respect the enforceability by the Agent and the Banks of (i) the Production Payment Rights assigned to them hereunder and (ii) the agreement of Pennzoil to pay directly to the Agent upon the date specified in the notice given to Pennzoil by the Agent pursuant to Paragraph 2 above an amount equal to the purchase price of the production payment specified in such notice after any required adjustment as provided in Section 2.5 of the Production Payment Agreement,

it being the purpose and intent of Pennzoil herein to confirm to the Agent and the Banks the continued effectiveness, validity and enforceability of the Production Payment Rights assigned to them hereunder throughout the term of this Assignment without regard to the performance or breach by the Co. pany of its obligations under this Assignment or the 1 seluction Payment Agreement.

6. Pennzoil hereby represents and warrants that

- (a) It is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. It is duly qualified to do business in those states in which it owns a material amount of property or transacts a material amount of business.
- (b) It has corporate power and authority to make and carry out the Production Payment Agreement and this Assignment, and all such action has been duly authorized by all necessary corporate proceedings on its part.
- (c) Since D comber 31, 1973 there has been no material adverse change in the financial condition of Pennzull or of Pennzull and its consolidated subsidiaries.
- (d) The Production Payment Agreement, the conveyance of production payment executed pursuant therebo and this Assignment have been duly and validly executed and delivered by Pennzul a 'triume valio and legally binding agreements of Pennzul coforceable in accordance with their cross, except as enforcement thereaf is limited by bandruptey, resolvency or other laws of general applies ican relating to affecting the enforcement of creditors' rights.
- (e) While this Assignment centimer unreleased Pennzoil shall, if notified of the exercise by the Agent of the right to have the purchase price prior by it to the Agent, be unconditionally obligated to pay such purel we price directly to the Agent in accordance with the terms hereof.
- 7. All notices and other communications provided for herein shall be in writing and shall be delined or mailed (or in the case of telegraphic communication, if by telegram, delivered to the telegraph company and if by the telex, graphic scanning or other telegraphic communications equipment of the sending party, delivered by such equipment) addressed, if to the Company or Fennzuli, as appropriate, at 1600 Southwest Tower, Houston Texas 77002, and if to the Agent, to it at One First National Plaza, Chreago, Illmois 69370. All notices and other communications given to any party hereto in accordance with the processors of this Assignment shall be deemed to have been given when sens by registered or certified mail, if by mail, or when delivered to the telegraph company, charges prepaid of by telegram, or when receipt is acknowledged, if by any telegraphic communications equipment of the sender, in each case addressed to such party as provided in this Paragraph 7 or in accordance with the latest unrevoked direction from such party.
- 8. This Agreement (i) shall remain in full force and effect until payment in full of all indebtedness and obligations of the Company to each of the Banks under and pursuant to the Credit Agreement and the Notes, at which time the Banks will, upon request and at the expense of the Company, release and reassign to the Company and Pennzoil, as their interests may appear, the rights assigned to the Banks hereunder as soon as there has been sufficient clapse of time that, in the opinion of the Banks, the continuation of the security interest under the Assignment is no longer necessary to protect the Banks from loss or expense by reason of any claim that the payment might constitute a preference or otherwise be subject to being voided, (ii) shall be binding upon the Company and Pennzoil, their respective successors and assigns, (iii) shall inure to the benefit of and be enforceable by the Agent and the Banks, their respective successors, transferces and assigns, and (iv) shall be construed in accordance with the laws of the State of Texas.

IN WITNESS WHEREOF, the parties hereto have caused this Assignment to be executed by their respective officers thereunto duly authorized, and in the presence of the undersigned witnesses, as of the day first above written.

| Witnesses: | | PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC. | | |
|------------|-----------------------|---|--|--|
| s/ | MARY FRANCES PORESTER | By s/ R. G. B Vice Pre- | | |
| s/ | Georgia Spaly | ATTEST: | | |
| | | s/ Salaw II Assistant S | | |

PENNZOIL COMPANY

Witnesses:

s/ MARY FRANCES FORESTER

By s/ W. E. GIESON
s/ Georgia Sprvly Vice tresident

ATTEST:

5/ SALLY HAZEN

[CORPORATE SEAL]

Assistant Secretary

[CORPORATE SEAL]

THE STATE OF TEXAS
COUNTY OF HARRIES

SS:

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 21st day of March, 1974:

(Louisiana)

Before me appeared R. C. Bryan, to me personally known, who, being by me duly sworn, did say that he is Vice President of Pennzon Louisiana and Texas Offshore, Inc., a Delaware corporation, and that the scar affixed to the foregoing instrument is the corporate scal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that R. G. Bryan acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared W. E. Girson, to me personally known, who, being by me duly sworn, did say that he is Vice President of P.—szon. Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W. E. Girson acknowledged the instrument to be the free set and deed of the corporation.

(Texas)

Before me on this day personally appeared R. G. Bayars, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President. I Pennson. Louisiana and Penas Orisia v. Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and cheed of said caperation.

(Texas)

Before me on this day personally appeared W. E. Guson, known to me to be the person whose name is subscribed to the foregoing instrument and known to not to be a Vice President of Pennzou. Company, a Delaware comportion, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WHERE OF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 21st day of March, 1974.

8/

Dorothy Stearman

Notary Public in and for Harris County, Texas My Commission Expires June 1, 1975

AMENDMENT

TO

INSTRUMENT OF CONVEYANCE OF PRODUCTION LAYMENT

THIS AMENDMENT TO INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Amendment") dated as of November 1, 1977, between Penszoil Louisiana and Texas Offshore, Inc., a Delaware corporation (the "Grantor"), having its principal office and post office address at Pennzoil Place, P.O. Box 2937, Houston, Texas, and Penszoil. Company, a Delaware corporation (the "Granter"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas.

WITNESSETH:

Winese as by means of that certain Instrument of Convey, we of Production Payment dated as of March 1, 1974 between the Granter and Grantee (such Instrument of Conveyance being hereinafter called the "Conveyance"), the Grantor conveyed to the Grantee, and the Grantee purchased from the Grantor, at the time the first loan was made to the Grantor under that certain Bank Credit Agreement dated as of March 1, 1974, as there after amended and restated, among the Grantor, The First National Bank of Chicago, and the other banks named therein (such Bank Credit Agreement, as amended and restated as of May 1, 1976, being hereinafter called the "Bank Credit Agreement"), a production payment dischargeable from the proceeds of production from or applicable to the interests of the Granter (hereinafter called the 'Leasehold Incrests'), in those certain oil and gas leases covering lands located offshore Louisiana and Texas in the Gulf of Muxico, which interests, leases and lands are described in Exhibit "A" to the Conveyance and in Exhibit "A" to Appendix I to the Pr. duction Payment Agreement hereinafter referred to, in an amount equal to the aggregate principal amount of the loans, plus accord interest and commitment fee, outstanding under the Bank Credit Agreement on the date, if any, upon which the purchase price therefor is paid by the Grantee to the Grantor as provided in that certain Production Payment Agreement dated as of March 1, 1979, as thereafter amended and restated as of May I. 1973, between the Grantee and the Grantor (such Production Payment Agreement, as so amended and restated, being hereinafter called the "Production Payment Agreement"), all as more particularly previded in the Production Payment Agreement;

Wheneas, in order to induce the banks referred to in Section 2 of the Bank Credit Agreement (hereinafter called the "Banks") to make loans pursuant to the Bank Credit Agreemen, and to secure the performance by the Grantor of its obligations under the Bank Credit Agreement, the Grantor, by that certain assignment dated as of March 1, 1974 and confirmed by the Confirmation of Assignment dated as of May 1, 1976 (hereinafter called the "Assignment") from the Grantor to The First National Bank of Chicago, as agent for the Banks (hereinafter called the "Agent"), assigned, transferred and set over to the Agent for the proportionate benefit of the Banks as their interests may appear all the right, title and interest of the Grantor in, to and under the Production Payment Agreement, subject to the exceptions set forth in the Assignment, and which Assignment was also executed by the Grantee for the purpose of its making the representations and special agreements set out therein;

Whereas, the Bank Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of May 1, 1976 to reflect an extension of the maturity of the credit available to the Company under the Bank Credit Agreement, an increase in the amount of that credit to \$200,000,000, the addition of certain banks as parties to the Bank Credit Agreement, the addition to the Leasehold Interests of the interests of the Company in certain oil and gas leases, and for other purposes;

Wheneas, the Bank Credit Agreement is now proposed to be amended to reflect a further extension of the maturity of the credit available to the Company thereunder, an increase in the amount of that credit to \$250,000,000 and the addition of Morgan Guaranty Trust Company of New York and First International Bank in Houston, N.A. as parties to the Bank Credit Agreement, and for other purposes; and

WHEREAS, pursuant to the request of the Grantor and the Grantee, the Agent proposes to join herein for the limited purposes set forth below;

Now, Therefore, in order to add to the Leasehold Interests the interests of the Grantor in certain oil and gas leases covering lands located offshore in the Gulf of Mexico adjacent to the United States of America and to delete from the Leasehold Interests certain other leases, and in consideration of the premises and the mutual covenants and agreements herein contained and intending to be legally bound, the parties hereto ago e as follows:

- 1. The Conveyance is hereby unrended in the following respects:
- (a) The first paragraph of the Conveyance is amended by changing the date "May 1, 1976" therein to "November 1, 1977".
- (b) Subparagraph (f) of Paragraph 1 of the Conveyance is amended by changing the date "May 1, 1976" therein to "November 1, 1977".
- (c) Subparagraph (a) of Paragraph 4 of the Conveyance is unended by changing the date "May 1, 1976" therein to "November 1, 1977".
- (d) Subparagraph (b) of Paragraph 4 of the Conveyance is amended by changing the figure "112%" therein to "105%".
- (e) The "In Witness Wheador" paragraph of the Conveyance is amended by adding after the date "March 1, 1974," the phrase "as heretofore annoted and restated,"
- (f) The signature; and acknowledgments set forth in the Conveyance are amended to read in their entirety as set forth in Annex I hereto as the signatures and acknowledgments thereto.
- (g) Exhibit "A" to the Conveyance is amended to read in its entirety as set forth in Annex J hereto as Exhibit "A" thereto.
- 2. Except as amended hereby, the provisions of the Conveyance shall remain in full force and effect. The Conveyance, as amended hereby, shall be restated to read in its entirety as set forth in Annex I hereto, it being agreed and understood that all references in the Conveyance as amended hereby and so restated to the Bank Credit Agreement shall be deemed to be references to the Bank Credit Agreement as amended and restated as of November 1, 1977.
- 3. The Agent, for and on behalf and in the name of the Banks, hereby consents to the foregoing amendment to and restatement of the Conveyance; provided, however, that notwithstanding any provision hereof to the contrary, this Amendment shall not be construed or operate, except as expressly set forth above, to modify in any respect the rights and obligations of the Company or Pennzoil under the Production Payment Agreement or the Assignment.

In WITNESS WHEREOF, the parties hereto have respective officers thereunto duly authorized as

WITNESSES:

this Amendment to be executed by their and year first above written.

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

[SEAL]

. W10200

PENNZOIL COMPANY

[SEAL]

By / fail Comm Ving Partition

THE FIRST NATIONAL BANK OF CHICAGO, as Agent

[SEAL]

Assistant Vice President

BE IT REMEMBERS that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 1st day of November, 1977:

(Louisiana)

Before me appeared W. B. Wilkermon, to me personally known, who, being by me dely sworn, did say the end of Vice Provident of Pennzon. Louisiana and Texas Offshour. Inc., a Delaware corporation, and that the end affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that W. B. Wilkermon acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared Horand E. Soaron, to me personally known, who, being by me duly sworn, did say that he is a Group View President of Presizent Commany, a Delaware corporation, and that the seal affixed to the foregoing instrument is the composate seal of said corporation and that the instrument was signed and sealed in bullable of the corporation by authority of its Board of Directors and that Harond E. Sogron acknowledged the instrument to be the free act and died of the corporation.

(Texas)

Before be on this day personally appeared W. B. Whikerson, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Pennzon. Leurs and Ann Tenas Operation, the a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared Haron. E. Souton, known to me to be the person whose name is subscribed to the toregoing instrument and known to me to be a Group Vice President of Person. Common, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

In WITNESS WHEREOF, I have here also set my hand and official notarial scal, in the City of Houston, County of Harris, State of Tesas, Phil. Let day of November, 1977.

Notary Public in and for Harris County, Texas My Commission Expires, Jane 35, 1978

ROSELLE / SHWORTH
Notary Politic in and for Harris County Texas

Besile (Lawersh.

My Commercian Evoltes June 13, 137 [Notarial Seal]

THE STATE OF ILLINOIS COUNTY OF COOK

(Louisiana)

(Texas)

BE IT REMEMBERS that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this had of November, 1977:

Before me appeared Bichard S. Palama, to me personally known, who, being by me duly sworn, did say that he is an Assistant Vice President of The First National, Bank of Chicago, a national banking association, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that Richard S. Palamar acknowledged the instrument to be the free act and deed of the rational banking association.

Before me on this day personally appeared Bichard S. Palatia, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be an Assistant Vice President of Tim Phor National, Bank or Cincaco, a national banking association, and acknowledged to me that he executed solid instrument for the purposes and consideration thereis, express in and in the capacity therein stated, and as the act and deed of said national banking association.

In Wilness Williams, I have hereinto set my hand and official notarial scal, in the City of Chicago, County of Cook, State of Illiams, this see they of November, 1977.

Cla . A Mortz

Notary Public in and for Cook County, Illinois

4y Countission Expires

Clare a mong

My Commission Profess A; 11 1979

[NOTARIAL SEAL]

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

PENNZOIL COMPANY

CONVEYANCE OF PRODUCTION PAYMENT

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Conveyance") dated as of March 1, 1977, as an order and restated as of November 1, 1977, between Presson, Louisiana and Trays Oreston, Isc., a Delaware corporation (the "Grantor"), having its principal office and port office address at Pennzoil Place, P.O. Box 2967, Houston, Texas, and Presszon, Conteany, at Delaware corporation (the "Granter"), having its principal office and port office address at Pennzoil Place, P.O. Box 2967, Houston, Texas.

WITNESSETH:

Wite mas, the Grantor desires to sell and the Crentee desires to purchase the Production Payment hereinafter described.

Now, The wrow, it is served by and between the Grant is and the Grentee as follows:

- Conveyance with the meanings given be
 - γ γ "Accounting Month" shall mean any monthly period communing with and including the 21st day of any calendar month and ending on and including the 20th day of the next succeeds t calendar month.
 - (b) "Effective Date" shall mean 7 o'clock, A.M., on the date of this Conveyance, according to the time then in effect at the location of each Subject Interest.
 - (c) "Exhibit a" shall mean Exhibit A attached to this Conveyance and hereby made a part hereof for all purposes.
 - (d) "Hydrocarbons" shall mean all oil, gas and other hydrocarbons or any combination of one or more of such substances appropriate to the context in which such term is used.
 - (e) "Lease" shall mean any oil and gas lease or sub-lease, described in Exhibit A.
 - (f) "Proceeds Commencement Date" shall mean 7 o'clock, A.M., according to the time then in effect at the location of each Subject Interest, on the date upon which the purchase price is paid by the Grantee to the Granter as provided in the Production Payment Agreement dated as of March 1, 1974, as amended and restated as of November 1, 1977, between the Granter and the Grantee.
 - (g) Subject Hydrocarbons" shall mean all Hydrocarbons in and under, and which may be produced and save I from, and which shall accure and be attributable to, the Subject Interests from and after the Proceeds Commencement Date and throughout the period specified in this Conveyance for the term of the Production Payment.
 - (h) "Subject Interests" shall mean the undivided percentage interests of the Grantor in the Leases, and, in addition thereto, each and every other kind and character of right, title, claim or interest which the Grantor now has in, to or under the Leases, which interests or Leases are

either specifically or generally described in Exhibit A, all as the same shall be enlarged by the discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and all renewals and extensions of any of the same, but there shall not be included within the Subject Interests any rights which the Grantor now has or may hereafter obtain with respect to production attributable to the interests of nonconsenting parties under any operating, agreement, unit operating agreement, contract for development or similar instrument.

2. Conveyance of the Pre-loction Payment. The Granton for valuable consideration, the receipt adequacy and sufficiency of which are benefity acknowledged for these presents does hereby grant, bargain, sell, convey, assign, transfer and deliver into the Grantee, as a production payment, an undivided eighty-live percent (55%) interest in the precend from the sale of the Subject Hydrocarbons; subject, however, to the previsions of Paragraph 3 of this Conveyance.

The aforesaid eighty-five percent (85%) interest in the proceeds from the sale of the Subject Hydrocarbons, as granted, bargains U sold, conveyed, assigned, transferred and delivered into the Grantee by the terms of this Conveyance, shall be the "Production Payment."

To Have assemble the Production Payment, together with all and sengular the rights, privileges and appurtenences thereunto in any way belonging, unto the Grance, its successors and assigns, forever.

- Certain Provisions Coverning Subject Hydrocarbons. All the provision of dus Conveyance shall be subject to the following:
 - (a) For the discharge of the Production Payment the Grantee shall look exclusively to the proceeds of Subject Hydrocarbons, and the Grantee shall not be liable for such discharge.
 - (b) The Production Payment shall not be dischargeable from the process of any products resulting from any manufacturing, processing or reliming operation, except to the extent of that portion of such proceeds which represents the fair mail of value at the wellhood of the Subject Hydrocarbons used in such manufacturing, processing or refining operation.
 - (c) The Production Fayment shall not be discharge able out of any points which the Grantor shall receive for any Leave or assignment of any of the Subject Interests or out of any payments made to the Grantor in connection with the drilling or deferring of drilling of any well on any of the Subject Interests or in connection with any adjusticant of any well and leasehold equipment upon unitization of any of the Subject Interests.
 - (d) There shall not be included in the Subject Hydrocarbons any Hydrocarbons unavoidably 'est in the production thereof or produced and saved from any of the Subject Interests and used by the Grantor in conformity with good field practices for drilling and production operations (including gas injection, secondary recovery, pressure maintenance, repressuring or cycling operations) conducted for the purpose of producing Subject Hydrocarbons from the Subject Interests or from any unit to which the Subject Interests are committed, but only so long as such Hydrocarbons are so used.
 - (c) So long as and to the extent that the same may be required by applicable laws and regulations, in the case of any Lease from the United States of America included in the Subject Interests, from which the average production of oil per well per day averaged on the monthly basis is 15 barrels or less, the obligation to pay and the right of the Grantee to receive the proceeds of oil produced from such Lease shall be suspended until said average production of oil per well per day exceeds said minimum amount, and such suspension shall apply separately to any zone or portion of such Lease segregated for computing government royalty.
- 4. Amount and Term of the Production Payment. The Production Payment shall continue and remain in full force and effect until such time as the Grantee shall have received and realized.

out of the proceeds from the sale of the Subject Hydrocarbons, free and clear of all development, operating, mining, producing, treating, processing, handling, storing, marketing, transporting and other rost and expenses of every kind whatsoever, the full aggregate sum of the following amounts:

- (a) The aggregate principal amount of loans, plus accrued interest and commitment fee, outstanding on the Proceeds Commencement Date under the Bank Credit Agreement, dated as of M = h 1, 1974, as thereafter amended and restated as of November 1, 1977, among the Grantor, the First National Bank of Chicago and the other banks named thereis (hereinafter called the "Primary Sum"); plus
- (b) An amount computed at a rate per annum (on the basis of a 365-day year) which shall be equal to 105% of the emporate base rate on 90 day commercial loans to its largest and most credit-weathy commercial horrowers in affect at The First National Bank of Chicago from time to time plus \$250, such rate to change and matically and from time to time effective as of the date of each change in such corporate base rate, from and including the Proceeds Commencement Date on the unliquidated balance from time to time of the Primary Sum (the first such computation to be made on the 21st day of the month next following the Proceeds Commencement Date for the period from and including the Proceeds Commencement Date to and including the 20th day of the month next following the Proceeds Commencement Date on the amount of such unliquidated balance on the Proceeds Commencement Date, and subsequent computations to be made monthly on the first day of each Accounting Month for the preceding Accounting Month on the amount of such unliquidated balance on the first day of each Accounting Month for the preceding Accounting Month); plus
- (c) An amount qual to the agency te of all amounts which are paid by the Granter to any state or political subdivision thereof on account of ad valorem, transfer, mortgage, gross production, gross receipts, income, profits, severance, occupation, sales, use, transhise and other taxes and assessments of any kind whatsoever, including penaltics and interest, imposed or assessed with respect to or measured by or charged against or attributable to the Production Payment, the Subject Hydrocarbons or the proceeds thereof, together with an amount equal to interest on the unliquidated amount thereof at the rate of 7% per amount from the date of payment of each such amount by the Grantee, provided, however, that the Granter shall have the right at its expense to contest any such taxes or assessments in good faith on behalf of the Grantee.

IT BEING THE INTENTION OF THE GRANTOR AND THE GRANTEE IN ENTITIES INTO THIS CONVEYANCE that the Grantee shall be entitled to receive and realize out of the Subject Hydrocarbons the full aggregate sum of the amounts described in subparagraphs (a) and (b) of this Paragraph 4, free and clear of all costs and expenses with respect to the Subject Interests and the Subject Hydrocarbons (which expenses and costs shall be borne by the Grantor) and over and above all taxes, expenses and costs of the character described, specified or referred to in subparagraph (c) of this Paragraph 4

- 5. Termination of the Production Payment, When the full aggregate sum of the amounts specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance (as increased as expressly provided by the terms of this Conveyance) has been received by the Grantee, the Production Payment shall forthwith terminate and all interest therein shall immediately revert to and become vested in the Grantor to the same extent and with the same force and effect as if this Conveyance had not been made. The Grantee agrees that, upon such termination, it shall execute or cause to be executed, upon the request and at the expense of the Grantor, such instruments as may be necessary or appropriate to evidence the termination of the Production Payment.
- 6. Application of the Proceeds of Subject Hydrocarbons. For all purposes of this Conveyance, the proceeds of Subject Hydrocarbons actually received by the Grantee prior to the close of business on 20th day of the month next following the Proceeds Commencement Date shall be deemed to have been received and applied immediately after the opening of business on the 21st day of such month, and thereafter, such proceeds actually received by the Grantee after the opening of business on the first day of each Accounting Month and prior to the close of business on the last day of such Account-

ing Month shall be deemed to have been received and applied inmedito opening of business on the first day of the next succeeding Accounting Month, pro-, that, if any date of application specified above shall be a Saturday Sunday or legal halo oler the law of the jurisdiction in which such proceeds are actually received. proceeds shall be deemed to have been received and applied on the last burn secong such Saturday, Sunday or legal holiday in such jurisdiction, but the amount of such applied persuant to subparagraph (a) of this Paragraph 6 shall neverthease be the one or mice subsequagraph (b) of Paragraph 4 of this Conveyance accurred up to (but her made hag) such Saturday, Sanday or legal holiday. Such proceeds shall be deemed to have been agend on the first day of each Accounting Month as follows:

- (a) First, to the amount referred to in subparact (x,y) of the energy is 4 of this Conveyance accrued up to (but not including) the d. of such application, and
- (b) Second, to the amount referred to be ubprograph (c) of said Paragraph 4, to the extent then ascertain of and
 - (c) Third, to the reduction of the unliquidated balance of the Primary State,

provided, however, that, in the event such proceeds so applied on the first day of any Accounting Month shall be insufficient to cover the full amount specified in subpraggraph (a) of the Paragraph 6, the unliquid ted balance of the Primary Sum shall forthwith be increased (to the extent permitted by law) by an amount equal to the amount of such debricacy.

Within 10 days after the date of each such application by the Grantee of any proceeds of Subject Hydrocarbons, the Grantee will furnish or cause to be furnished to the Granter a written statement showing the application of such proceeds in accordance with the provisions of this Prograph 6.

- 7. Marketing the Subject Hydrocarbons. The Grantor, at its expense, shall market, or cause to be marketed, the Subject Hydrocarbons for the best price reasonably available when marketed, and it shall not be necessary for the Grantee to join in any Salex Contracts (as hereinafter defined) or any amendment thereof. In the excut the Grantor shall take any of the Subject Hydrocarbons for its own use, the Grantor shall pay to the Grantee, on or before the last day of the Accounting Month during which such taking occurs, a price equal to the fair market value at the wellhead of the Subject Hydrocarbons so taken.
- 8. Sales Contracts. The Grantor will duly perform all obligations binding on it under all contracts at Lagreements for the sale of Subject Hydrocarbons at any portion thereof, whether presently existing at hereafter caterial into (all such contracts and agreements being the "Sales Contracts"), in accordance with the terms hereof and will take all appropriate measures to enforce the performance under each of the Sales Contracts of the obligations of the purchaser thereinder. All Subject Hydrocarbons sold by the Grantor, whether pursuant to the terms of the Sales Contracts or otherwise, shall be delivered by the Grantor to the purchases thereof into the pipelines to which the wells producing such Subject Hydrocarbons, may be connected or to such other point of purchase as is reasonably required in the marketing of such Subject Hydrocarbons.
- 9. Withholding and Restitution of Proceeds of Subject Hydrocarbons. All obligations of the Grantor hereunder shall be subject to the applicable provisions of the Natural Gas Act and all applicable rules and regulations of the Federal Power Commission. Rates permitted under the Natural Gas Act and said rules and regulations to be paid for gas included in Subject Hydrocarbons shall be controlling if varying from prices established in Sales Contracts or if different from the fair market price at the wellhead. The Grantor shall be entitled to use its reasonable discretion in making filings with the Federal Power Commission affecting Subject Hydrocarbons. If any proceeds of Subject Hydrocarbons shall be withhold for any reason whatsoever, including, without limitation, withholding by the Grantor pending final approval of rate increases filed with the Federal Power Commission, the Grantee shall not be deemed to have received or realized such proceeds until, and only to the extent that, such proceeds shall actually have been received by the Grantee. If, at any time before

or after receipt of the full aggregate sum specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance, the Grantee shall be compelled, for any reason whatsoever, to make any payment or restitution, or such payment or restitution shall be made for the account of the Grantee, on account of any proceeds of Subject Hydrocarbons, which a proceeds have already been applied pursuant to the provisions of Paragraph 6 of this Conveyance, the unliquidated bilance of the Primary Sum shall forthwith be increased by the amount of the proceeds so paid over by or for the account of the Grantee plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties; provided however, that if at the time of such payment or restitution the entire Primary Sum shall have been liquidated, the Primary Sum shall be deemed not to have been so liquidated and the undiquidated balance of the Primary Sum shall be thereupout deemed to be in the amount of the proceeds so paid over plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties.

- 10. Protection to Purchasers of Production. No person purchasing or taking or processing any Subject Hydrocarbons shall be required to take notice of, or keep informed as to, the discharge termination of the Production Payment until the actual receipt by such person of written not, advising such person of such discharge or termination.
- 11. Payment of Taxes. The Granton will tender and pay, or cause to be paid, punctually, before they shall be one delinquent (or as to any thereof which are being contested in good faith, promptly after the final determination of such contest), all ad valorem taxes (or taxes imposed in lieu thereof) and all transfer, mortgage, gross production, mining, severence, occupation, excise, sales, use, framelise income and other taxes and assessments of every kind and character whatsoever (other than federal income taxes of the Grantee) imposed or assessed with respect to or measured by or charge I against or attributable to the Subject Interests, the Production Payment or the Subject Hydrocarbons, or the proceeds thereof, or against the Grantee by reason of its acquisition or ownership of the Production Payment or against any assignee of the Production Payment by reason of his or its interest as assignee, together with any interest or penalty payadde in connection therewith.
- 12. Covenants of the Grantov. Until such time as the Production Payment has terminated as provided in Paragraph 5 of this Conveyance, the Grantor will, at its expense and regardless of who may be the operator or operators of the Subject Interests, cause:
 - (a) The Subject Interests to be maintained and developed and continuously operated for the production of Hydrocarboas in a good and workmanlike manner and in accordance with sound field practices and all applie able federal, state and local laws, rules and regulations (except those being contested in good faith), provided, however, if the Grantor in good faith and in conformity with sound field practices elects not to participate in any operation which is to be conducted under the terms of an operating agreement, unit operating agreement, contract for development or similar instrument entered into by the Grantor or its predecessors in title pertaining to the development and operation of any Subject Interest (or any portion thereof) and which allows any consenting party or parties to conduct nonconsent operations, then, during the period or periods in which the consenting party or parties are entitled to receive Subject Hydrocarbons attributable to such Subject Interest (or portion thereof) by the terms of such instrument pending the recomponent or receipt of the sums electronic thereof or periods or periods) proceeds from Subject Hydrocarbons attributable to such Subject Interest (or portion thereof) to the extent affected thereby.
 - (b) All rentals, royalties, and all liabilities of any kind or nature, including, without limitation, all liabilities for labor, material, supplies and equipment incurred in, or arising from, the administration, operation or development of, or the gathering, producing, treating, processing, storing, marketing or transporting of, the Subject Hydrocarbons, to be paid punctually when due, or, as to any thereof which are being contested in good faith, promptly after the final determina, in of such contest.

- (c) All machinery, equipment and facilities of any kind now or hereafter located on the Subject Interest, and necessary or useful in the operation thereof for the production of Hydrocarbons therefrom to be provided and to be kept in good and effective operating condition and all repairs, necessals, replacements, additions and improvements thereof or thereto needed for such purpose to be duly made or provided.
- (d) All necessary and proper steps to be taken diligently to protect and defend the Subject Interests, the Subject Hydrocarbons and the proceeds thereof against any adverse claim or demand, including, without limitation, the employment or use of counsel for the prosecution or defense of litigation and the contest, settlement, release or discharge of any such claim or demand.
- (e) Written notice to be given to the Grantee of every adverse claim or demand made by any person (including any government or governmental agency) affecting the Subject Interests, the Subject Hydrocarbons or the proceeds thereof in any manner whatsoever, or of any proceeding instituted with respect thereto.
- (f) The interest of the Grantor in each of the Subject Interests to be kept free and clear of liers, charges and encumbrances of every character, other than those which are of a character customarily found in connection with comparable drilling, and producing operations and which do not materially adversely affect the operation of the Subject Interests and other than those becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Convegance.
- (g) Subject to the provisions of Paragraph 19 of this Conveyance, all Leases included in or relating to the interests described or referred to in Exhibit A and under which the Grantor is lessee, sub-lessee, or assignce to be maintained in full force and effect without amendment, modification or waiver of any provision of any such Lease which might materially adversely affect the rights of the Grantee under the Conveyance.

Anything contained in this Paragraph 12 to the contrary notwithstanding, the Grantor shall not be obligated to perform undertallings performable only by the operator and which are beyond the control of the Grantor, with respect to a dividual Subject Interests of which it is not the operator. In such case, here experience, the Grantor will use its best efforts to seeme the performance of any such undertakings.

- 13. Access to the Subject Interests. The Granton shall period any one or more representatives designated by the Grantee to make any inspection of the Subject Interests at any reasonable time as the Grantee may reasonably request. In addition the Grantor shall furnish to the Grantee such detailed information as the Grantee may reasonably request in writing concerning the Subject Interests, the operation thereof and the production and sale of the Subject Hydrocarbons.
- 14. Pooling and Unitization. Anything herein to the conteary notwithstanding, the Grantor may, without the consent of the Grantee, pool, combine, communitize or unitize any Subject Interest (or any portion the reof) from time to time with other leasehold, mineral or other interests to form units when, in the judgment of the Grantor, it is necessary or advisable to do so in order to facilitate the orderly development or operation (including repressuring, pressure maintenance, cycling, water flood and secondary recovery operations) of any Subject Interest (or any portion thereof). For the with, after the formation of any unit, the Grantor shall furnish to the Grantee a true copy of the pooling agreement, unit designation or any other instrument creating such unit, together with, upon request of the Grantee, any operating agreement relating thereto. The interest in any such unit accruing or attributable to any Subject Interest (or any portion thereof) included therein shall be subject to the Production Payment in the same manner and with the same effect as though such unit were in existence on the Effective Date. The sign stare or joinder of the Grantor—hall of be necessary to the instruments evidencing the pooling and unitization herein authorized.

- 15. Default by the Grantor and the Remedies of the Grantee. Should the Grantor default in the performance or observance of any covenant or agreement provided in this Conveyance to be performed or observed by the Grantor, and should such default continue unremedied for 30 days (or for such longer period as may be necessary to complete performance deligently commenced within 30 days) from the delivery to the Grantor of written notice of such default, the Grantee may, in addition to pursuing any or all other remedies available at law or in equity:
 - (a) Effect performance or observance of such covenant or agreement on behalf of and at the expense of the Grantor, in which event the Grantee may expend funds reasonably necessary to effect such purposes and shall be entitled to reimbursement therefor from the Grantor, together with interest on such amounts at the rate of 7% per amount from the date of such expenditure to the date of reimbursement.
 - (b) Upon written notice to the Grantor, succeed to all rights of the Grantor with respect to the possession, operation and development of the Subject Interest, and may use in connection therewith all of the appurtenant property, both real and personal, situated upon or used or held for future use in connection therewith, or in connection with the production, storing, transporting or sale of the Subject Hydrocarbons, and the Grantoe shall have the right, on behalf of and for the account of the Grantor, to sell the Subject Hydrocarbons and the Grantor shall, upon demand, reinburse the Grantee for all amounts so expended by the Grantee, together with interest on such amounts at the rate of 7% per amount from the date of such expenditure to the date of reimbursement.
 - (e) Apply to a court of equity for the specific performance or observance of such covenant or in aid of the execution of any power briefin granted and for the appointment of a receiver of the Subject Interests and the Subject Hydrogarbons.
- 16. Exercise and Termination of the Remedies of the Grantce. All rights and remedies granted by the provisions of Paragraph 15 of this Conveyance and otherwise belonging to the Grantce shall be cumulative and no exercise of any right or remedy shall preclude the exercise of any other right or remedy. Any right or remedy may be exercised on one or more occasions without exhaustion thereof. All rights and remedies shall terminate:
 - (a) When the Production Payment is discharged and all amounts then due and payable to the Grantee pursuant to the terms of Paragraph 15 of this Conveyance (including amounts payable for interest) shall have been paid in full, or
 - (b) At any carbon time when all defaults of the Grantor shall have been remedied and all such amounts shall have been duly paid in full, without projudice, here ever, to the exercise of any rights and remedies conferred upon the Grantee upon any subseque at failure of the Grantor to perform or observe any of the covenants or agreements herein provided to be performed or observed by the Grantor.
- 17. Reports to the Grantee. While the Production Payment remains in force and effect, the Grantor will, at its expense, furnish to the Grantee within 45 days after the encl of each calendar month, a report showing for such calendar month (i) the quantity of Subject Hydrocarbons produced and sold or taken by the Grantor for its use and (ii) the gross proceeds received from the sale, or attributable to such taking, of such Subject Hydrocarbons.
- 18. Warranty of Title. The Grantor covenants and warrants that at the time of the execution and delivery of this Coaveyance, the Grantor has good and marketable title to the Subject Interests, subject only to such exceptions, charges, liens or encumbrances, if any, as are set forth or referred to in Exhibit A; that the Grantor has the power and right to create and confer upon the Grantee the

rights of the Grantee granted by the terms of this Conveyance; and that there are in suits or proceedings pending or, to the knowledge of the Grantor, threatened, against or affecting the Eulopet Interests before any court or by or before any governmental commission, bureau or any regulatory authority which if decided adversely to the interests of the Grantor shall materially affect the rights of the Grantee pursuant to this Conveyance. The Grantor bureby binds itself, its successors and assigns, to warrant and defend forever the rights under this Conveyance of the Grantee, its successors and assigns, against the lawful claims and demands of every person who shall claim the same or any part thereof, and the Grantor will indemnify and hold the Grantee harmless against and from all loss suffered by the Grantee resolver, from any failure of or defect in the Grantor's title to the Subject Interests and the production of Hydrocarbons as warranted as aforesaid. The covenants and warranties of this Leaguaph 18 shall curvive the discharge of the Preduction Payment.

19. Transfer of the Subject Interests. Except for the conveyance of production payments becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Conveyance, prior to such termination the Greeter shall not, without the written consent of the Granter, make or permit any surrender, abandomoes scale, conveyance, assignment, lease or sublease, in whole or in part of any of the Sul act Interest, provided, however, that the Counter shall have the right to surrende, abandon or consiste terminals (either voluntarily or involuntarily) any of the Subject Interests (m and portion of) when there is no well located thereon which is capable of (i) no existing well may be recompleted thereon as a well capable of producing Hydrocarbons in paying quantities, (i) no additional well could be drilled and completed thereon which would be capable of preducing Hydrocarbons, over and above the cost of drilling and completing same, in paying quantities, and (iii) no pressure maintenance or secondary recovery operations may be instituted with respect to any well thereon which would cause such well to be capable of producing Hydrocarbans, over and above the cost of instituting and completing such pressure maintenance or secondary recovery operation, in paying quantitie. For the pure sets of this Paragraph 19, the words "in paying quantities" as applied at any time to the prod and equality of any well shall mean that, in the light of conditions, existing at the time of determination and which reasonably appear to be not transcribe, such well is producing or is or will be capable of producing Subject Hydrocarbans where aggregate value exceeds or will exceed the Granter's share (as the owner of the Subject Interest on which such well is located) of the direct cost of operating such well.

In each case where the presiding paragraph of this Paragraph 19 permits the Grantor to surrender, abandon or otherwise is usingly any of the Subject Interests (or any portion thereof), then (a) any of the Subject Interests (or any portion thereof) which are subject Interest, abandoned or terminated shall upon surrendering abandoninent or terminating any of the Subject Interest, and (b) the Grantor, in lieu of surrendering abandoning or terminating any of the Subject Interests (or any portion thereof), may self-assign, sublease, farmout or convey the same have and clear of the Production Payment without the consent of the Grantoe, and the Grantoe shall not have the right to any proceeds from such sale, assignment, sublease, farmout or conveyance, provided, however, that any interest in any such Subject Interest (or any portion there it) which may be excepted from such sale, assignment, sublease, farmout or conveyance and retained by the Grantor shall not be released thereby from the Production Payment.

- 20. Exception to Payments. The acceptance by the Grantee of any payment hereunder shall not preclude the Grantee from taking exception to the correctness of the amount thereof; provided, however, that any such exception must be specified by the Grantee in a notice given to the Granter within 12 months after the date such payment is mad.
- 21. Transfer of the Production Payment. Notling contained in this Conveyance shall be construed to limit or restrict many way the right of the Grantee to sell, convey, assign or mortgage the Production Payment, in whole or in part.

- 22. Rights of Mortgagee or Trustee. If the Grantee shall at any time execute a mortgage or deed of trust covering all or part of the Production Payment, the mortgagee or trustee therein named or the holder of any obligation secured thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to exercise all of the rights, remedies, powers and privileges conferred upon the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained hereunder by the Grantor, but the provisions of this Paragraph 22 shall in no way be deemed or construed to impose upon the Grantor any obligation or liability undertaken by the Grantee under such mortgage or deed of trust or under the obligation secured thereby.
- 23. Notices. Any notice request demand, report or other instrument which may be required or permitted to be given or furnished to or served upon either party hereto or other person succeeding to any interest of either party hereto shall be deemed sufficiently given or furnished and served if in writing and delivered to such party or person or to an officer of such party or person or deposited in the United State, mail in a scaled envelope, registered, with postage prepaid, addressed if to the Grantor to Penezoil Place, P.O. Box 2967, Houston, Texas, and if to the Grantee to Pennzoil Place, P.O. Box 2967, Houston, Texas, as the party or person to be addressed shall have designated by written notice to the party or person giving such notice or furnishing such report or making such request or demand.
- 24. Further Assurances. So long as permitted by applicable law so to do, the Grantor will execute and deliver all such other and additional instruments, notices, releases, acquittances and other documents, and will documents and will documents and will documents and things, as may be necessary more fully to accure to the Grantee all the rights and into rests herein and hereby granted or intended so to be.
- 25. Binding Effect. All the covenants and agreements of the Grantor contained in this Conveyance shall be deemed to be covenants running with the Earl and shall be binding upon all successors in interest to, and all assigns of, the Grantor and shall inure to the benefit of an successors in interest to, and all assigns of, the Grantor. The provisions of this Conveyance shall inure to the benefit of and he binding upon the respective successors and assigns of the Grantor and the Grantoe from and after acceptance by the Grantoe. All references herein to either the Grantor or the Grantoe shall include their respective successors and assigns.
- 26. Release or Failure of Title. No release by the Grantee of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect of reducing the amount of the proceeds from the sale of production of Hydrocarhous out of which the Production Payment is dischargeable or reducing the amount of the Production Payment or creating any offset or other prejudice to the Production Payment, and the Production Payment shall come in full force and effect as to the remainder of the Subject Interest until the full amount of the Production Payment has been received by the Grantee as herein provided.
- 27. Final Termination. Notwith touching the other provisions hereof, the Production Payment shall, in any event, terminary 1 day prior to the expiration of 21 years after the death of the survivor of all the descendants of recordore Ecosevelt, late President of the United States, who are living on the date hereof.
- 28. Headings. Paragraph heading used in this Conveyance are for convenience only and shall not affect the construction of this Conveyance.

In Witness Witness, this Restated Instrument of Conveyance — induction Payment amends and restates the Instrument of Conveyance of Production Payment oated as of March 1, 1974, as heretofore amended and restated, and the parties have cause? This Restated Instrument of Conveyance of Production Payment to be duly executed in the presence — in undersigned witnesses on the dates of their respective acknowledgments annexed hereto, in several counterparts, each of which is an original and all of which are identical. Each of such counterparts shall for all purposes be deemed

to be an original, and all such counterparts together shall constitute but one and the same Instrument. The date of this Instrument shall be deemed for all purposes to be the date first above written.

| Witnesses: | PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC. |
|------------------|---|
| Sue J. Prejean | By WBW Western |
| Henre 111.1 fell | ATTEST: |
| | Sally Hay - Secretary |
| | [CORPORATE STAL] |
| Witnesses: | |
| | PENNZOH, COMPANY ///? |
| Jenel Miles | By fold Millerouth Vice President |
| Janel 111/ Self | ATTEST: |
| V | Sally Hay Secretary |

[CORPORATE SEAL]

THE STATE OF TEXAS
COUNTY OF HARRIS

BE IT REMEMBERS that I, the undersigned, a Notary Public July qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on it - 1st day of November, 1977:

in and for the County and State aforesaid, hereby certify that, on the 1st day of November, 1977:

Before me appeared W. B. Willeson, to me personally known, who, being by me duly sworn, did say that he is Vice President of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware

did say that he is Vice President of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corporation, and that the seal affixed to the tolegoing instrument is the corporate scal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W. B. Williamson acknowledged the instrument to be the free act and deed of the corporation.

Before me appeared HAROLD E. SORIOR, to me personally known, who, being by me duly sworn, did say that he is a Group Vice President of PENNZOU. COMPANY, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that HAROLD E. SORTON acknowledged the instrument to be the free act and deed of the corporation.

Before me on this day personally appeared W. B. Wilkerson, known to the to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Pennzon. Louisiana and Texas Orishoae, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed; and as the act and deed of said corporation.

Before me on this day personally appeared Harona E. Souron, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of Pennzon. Company, a Delaware comporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

In Witness Windstor, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 1st day of November, 1977.

Nota: y Public in and for Hagris County, Texas My Commission Expires Jane 30,1918

POSELLE ASH TOPTH
Notary Put 1 c in and for Harry County, Texas
My Commes on Expires June 50, 1973
Bonded by Alexander Lovett, Lawyers Surely Corp.

Rose Church

Louisiana)

(Texas)

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT

| BLM or State Lease No. | C | Description Federal leases are as shown on Official ter Continental Shelf Leasing Maps) | Grantor's Undivided Percentage Interest |
|---------------------------|---------|---|--|
| OCS-G2839 | 12-1-74 | All of Block 352, West Cameron Area, West Addition, OCS Official Leasing Map, Louisiana Map No. 1A | 25.00% |
| OCS-G2224 | 2-1-73 | Block 532 West Cameron Area, South Addition Gfficial Leasing Map, Louisiana Map No. 1B | 26.67°; |
| OCS-G2225 | 2-1-73 | Block 533 West Camerov Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 26.67% |
| OCS-G3284 | 8-1-75 | All of Block 563, West Cameron Area South Addition, as shown on OCS Official Leasing Map, Louisima Map No. 1B | 25.00% |
| OCS-G2228 | 1-1-73 | Block 500 West Cameron Area, South Addition, Official Leasing Map, Louisians Map No. 1B | 24.252561; |
| OCS-G2850 | 12-1-74 | All of Block 609, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 6.6666777 |
| OCS-G2559 | 5-1-74 | All of Block 617 West Cameron Area — South Addition, Official Leasing Map, Louisiana Map No. 1B | 6.6660777 |
| OCS-G2859 | 12-1-74 | All of Block 236, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana M: No. 2A | 16.6666757 |
| OCS-G2860 | 12-1-74 | All of Block 23., Past Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.66667% |
| OCS-G2882 | 12-1-74 | All of Brock 125, Sourn Marsh Island Area, South Addition, CCS Official Leasing Map, Louisiana Map No. 3C | 6.66657% |
| OCS-G2883 | 12-1-74 | All of Block 127, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2587 | 5-1-74 | All of Block 125 South Marsh Island Area. South Addition, Official Leading Map, Louisiana Map No. 3C | 6.66667 |
| OCS-G2884 | 12-1-74 | All of Block 129, South Mar Area, South Addition, OCS Office. Sasing Map Louisian't Map No. 3C | 6.6666757 |
| OCS-G2885 | 12-1-74 | All of Block 141, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2900 | 12-1-74 | All of Block 261, Eugene Island Area, OCS Official Leasing Map, Louisiana Map No. 4 | 16.66667% |
| OCS-G3156 | 7-1-75 | All of Block 262, Eugene Island Area, as shown on OCS Official Leasing M v., Louisiana Map No, 4 | 16.66667% |

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| BLM or State Leave No. | Effective Date | Description | Grantor's Undivided Percentage In crest |
|------------------------|-------------------|---|--|
| OCS-G2607 | 5-1-74 | All of block 312 Eugene Island Area, South Addition, Official Leasing Map Louisiana Map No. 44 | 20.66667% |
| OCS-G2317 | 2-1-73 | Block 333 Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A | 26.67% |
| OCS-G2177 | 11-1-72 | All of Block 49 South Pass Area as shown on official leasing map, Louisiana Map No. 9 | 10,00% |
| State Lease No. 6310 | 5-13-74 | Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and bring more particularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Louisiana | .00% |
| OCS-G2185 | 10-1-72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supleme Court entered December 13, 1965 in the U.S. v. Louisiana No. 9 Original (382 U.S. 258), South Pass Area, South and East Addition, as shown en official leaving map, Louisiana Map No. 9A | 15.00% |
| OCS-G3193 | 7-1-75 | S½ of Block 59, Main Pass Area, that portion located in Zone 2 as that zone was defined in the agreement between the United States and the State of Louisiana, October 12, 1956 and landward ci the Third St pplemental Decree Line (404 U.S. 3SS (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 25.00% |
| OCS-G3195 | 7-1-75 | Block 72, Main Pass Area, that portion lecated more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 21.00% |
| OCS-G2947 | 12-1-74 | All of Block 73, Main Pass Area, OCS Official Leasing Map, Louisiana Map No. 10. That ortion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1977)) | 21.00% |
| OCS-G2193 | .0-1-72 | All of Block 140 Main Pass Area as shown on official leasing map, Louisiana Map No. 10 | 10.00% |
| OCS-G3202 | 7-1 75 | All of Block 947. Ewing Bank Area, as shown on OCS Official Protraction Diagram NII-15-12 | 21.00% |
| | | | |

| BLM or State Lease No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
|---------------------------|-------------------|--|--|
| OC5-G3203 | 7-1-75 | All of Block 903, Ewing Bank Area, as shown on OCS Official Protraction Diagram NII-15-12 | 21.00% |
| OCS-G3206 | 7-1-75 | All of Block N663F63, Mobile South No. 2, as shown on C = Official Leasing Map NH-16-10 (Also k) — a as Block 63, Mississippi Canyon Area, as s—on on OCS Official Protraction Diagram 25-1 16-10) | 10.00% |
| OCS-G2358 | 8-1-73 | All of Block A-442, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2366 | 8-1-73 | All of Block A-474, High Island Area, South Addition, Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2372 | 8-1-73 | All of Block A-489, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2373 | 8-1-73 | All of Block A-496, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 16.00% |
| OCS-G3118 | 4-1-75 | All of Block A-499, II as Island Area, South Addition as shown on OCS Official Leasing Map, Texas Map No. 7B | 16.40% |
| OCS-G2377 | 8-1-73 | All of Block A-519. High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 16.43835% |
| OCS-G2378 | 8-1-73 | All of Block A-520, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2383 | 8-1-73 | All of Block A-535, High Island Area, South Addition. Official Leasung Map, Texas Map No. 7B | 13.333337 |
| OCS-G2701 | 7-1-74 | All of Block A-540, High Island Area – South Addition, OCS Official Leasing Inp. Texas Map No. 7B | 15.00% |
| OCS-G2704 | 7-1-74 | All of Block A-515, High Island Are: - South Addition, OCS Official Leasing Map, Texas Map No. 7B | 10.00% |
| OCS-G2779 | 10-1-74 | All of Block A-546, High Island Area, Earth Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2705 | 7-1-74 | All of Block A-547, High Island - South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2706 | 7-1-74 | All of Block A-548, High Island Area – South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2384 | 8-1-73 | All of Block A-555, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 14.50% |
| | | -14- | - 1.00/0 |

-14-

| | | | Grantor's Undivided |
|---------------------------|-----------------|--|------------------------|
| BLM or State Lease No. | Date Date | Description | Percentage Interest |
| OCS-G2713 | 7-1-74 | All of Block A-562, High Island Area - South Addition, OCS Official Leasing Map, Texas Map No. 7B | 12.715% |
| OCS-G2388 | 8-1-73 | All of Block A-563, High Island Area, South Addition: Official Leasing M. p. Texas 1 ap No. 7B | 12.00% |
| OCS-G2389 | 8-1-73 | All of Plock A-564, High Island Area, South Adda — a. Otheral Leasing Map, Texas May — a 7B | 12.00% |
| OCS-G2390 | 8-1-73 | All of Block 10, High Unid Area, South Advana, Official Leasing Map, Texis Map No. 7B | 20.00% |
| OCS-G2718 | 7-1-74 | All of Block A-575 High Island Area – South Addition, OCS Official Leasing | |
| OCS-G2719 | 7-1-74 | Map, Texas Map No. 7B All of Block A 582 He h Island Area — South Addition, OCs official Leasing | 20.00% |
| OCS-G2397 | 8-1-73 | Map, Texas Map No. 7B All of Block A-272, High island Area, East Addition, South Extension. Official | 14.118% |
| OCS-G2398 | 8-1-73 | Leasing Map, Texas Map No. 7C All of Block A-273, 11), h Island A.ca. East Addition, Sout , Extension, Official | 15.33% |
| OCS-G2399 | S-1-7 33 | Leasing Map, Texts Map No. 7C All of Block A 274, High Island Area, East Addition, South Extension, Official | 12.00% |
| OCS-G2403 | 8-1-73 | Leasing Map, Texas Map No. 7C All of Block A-279, High Island Area. | 13.8461% |
| OCS-G3314 | 4-1-76 | East Addition, South Extension, Official Leasing Map, Texas Map No. 7C All of Block A-288, High Island Area, | 17.00% |
| | | East Addition, South Extension, as shown on GCS Official Leasing Map, Texas Map No. 7C | 17.90% |
| OCS-G2409 | 8-1-73 | All of Biock A-312, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 12.00% |
| OCS-G2411 | 8-1-73 | All of Block A-314, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 22.85714% |
| OCS-G2737 | 7-1 74 | Ali of Block A-316 Hg b Island Area — East Addition — South Extension, | |
| OCS-G2413 | 8-1-73 | OCS Official Leasing Map, Texas Map No. 7C All of Block A-522, High Island Area, East Addition, South Extension. | 14.00% |
| OCS-G2414 | 8-1-73 | Official Leasing Map, Texas Map No. 7C All of Block A-323, High Island Area, East Addition, Sonta Extension. | 14.00% |
| | | Official Leasing Map, Texas Map No. 7C | 14.00% |

EXHIBIT A TO

CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| | CONTEINNEL | T TRODUCTION TRIMING (Communica) | |
|---------------------------|-------------------|--|---|
| BLM or State Lense No. | Effective Date | Description | Grantor's Undivided Percentagee Interest |
| OCS-G2416 | 8-1-73 | All of Block A-325, High Island Area. East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 20.86957% |
| OCS-G2418 | 8-1-73 | All of Block A-327, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G2122 | 8-1-73 | All of Block A 332, High Island Area, East Addition, South Extension, Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G2739 | 7-3-74 | All of Block A 339 High Island Area East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 34.00% |
| OCS-G2420 | 8-1-73 | All of Block A-340, High Island Area, East Addition, South Extension, Official Leasing Map, Texas Map No. 7C | 34.00% |
| OCS-G2429 | 8-1-73 | All of Block A-351, High Island Area, East Addition, South Extension. Official L. asing Map, Texas Map No. 7C | 15.25% |
| OCS-G2745 | 7-1-74 | All of Block A-355, High Island Area - East Addition - South Extension, OCS Official Leasing Map, Texas Map No. 7C | 11.33% |
| OCS-G2746 | 7-1-74 | All of Block A-356, High Island Area — Fest Addition — South Extension, —"S Official Leasing Map, Texas Map No. 7C | 15.00% |
| OCS-G2766 | 7-1-74 | Aft Block A-403 High Island Area – East Addition – South Extension, | 10.00% |
| OCS-C3417 | 1-1-77 | OCS Official Leasing Map, Texas Map No. 76. That portion of Main Pass Blocks 72 and 74 located more than there geographical miles seawar I from the line 6, scribed in the supplemental decrees of the U.S. Supreme Court, June 16, 1975. (U.S. vs. Louis and, 492 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971), as shown on OCS Official Leasing Map | |
| OCS-G3409 | 1-1-77 | Louisiana Map No. 10 All of Block 351, Eugene Island Area, South Addition, as shown on OCS Official Leasing | 25.00% |
| OCS-G3410 | 1-1 77 | Map, Louisiana Map No. 4A That pertion of South Pass Block 57 for ated more than three is occaphical miles save and from the line described in the supplemental decretof the U.S. Supreme Court, June 16, 1975; (U.S. vs. Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 355, December 20, 1971), as shown on OCS Official Leasing Map, Louisiana, Map Map Map 1997. | 20.000 |
| OCS-G350 | 8-1-77 | ana Map No. 9 All of Block 253; West Cameron Area, as she will on OCS Official Leasing Map, Louisiana Map No. 1 —16— | 30.00% |
| | | - 10 | |

This Conveyance is subject to the terms and provisions of the above-described oil and gas leases and accordingly, the Subject Hydrocarbons and to proceeds therefrom shall be calculated after deducting the proportionate share of the royalties reserved in each such lease attributable to Grantov's said undivided percentage interest in such lease.

Lease Nos. OCS-G2839, OCS-G2225, OCS-G2859, OCS-G2559, OCS-G2607, OCS-G2317, OCS-G3193, OCS-G3195, OCS-G2917 and OCS-G3417 retend to above are each subject to an Operating Agreement between Mobil Oil Corporation, as operator, and the Ciantor, et al., as non-operators.

Lease No. OCS-G2224 referred to above is subject to an Operating Agreement between Chevron Dil Company as operator, and the Grantor, et al., as non-operators.

Lease No. OCS-G3284, OCS-G2228, OCS-G2859, OCS-G2860, OCS-G2882, OCS-G2883, OCS-G2887, OCS-G2881, OCS-G2883, OCS-G2031, OCS-G2881, OCS-G2883, O

Lease Nos. OCS-G2177, OCS G2193 and OCS G3206 referred to above are each subject to an Operating Agreement between Gulf Gil Corporation, as operator, and the Grantor, et al. as non-operators.

Lease Nos. OCS-G2358, OCS-G2403 and OCS-G3314 referred to above are each subject to an Operating Agreement between Tarathon Oil Company, as operator, and the Grantor, et al., as non-overal to

Les Nos. OCS-G2373, OCS-G2377, OCS-G2378, OCS-G2411, OCS-G2737, OCS-G2413, OCS-G2414, OCS-G241b OCS-G2745 and OCS-G2746 referred to above are each subject to an Operating Agreement between Cities Service Oil Company, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G²353, OCS-G, 409 OCS-G2418 and OCS-G2422 reterred to above are each subject to an Operating Agreement between Sie Oil Company (Delaware), as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2390 and OCS-G2715 referred to above are each subject to an Operating Agreement between Exxon Corporation, as operator, and the C antor, et al., as non-operators.

Lease Nos. OCS-G2397 and OCS-G2398 referred to above are each subject to an Operating Agreement between Annuoil Development, Inc., as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2701, OCS-G2779. OCS-G2705 and OCS-2706 are each subject to an Operating Agreement between Texaco Inc., as operator, and the Grantor, et al., as non-operators.

The Grantor's interests in Lease Nos. OCS-G2224. OCS-G2225, OCS-G2228 and OCS-G2185 referred to above are subject to Advance Payment Agreements dated as of June 13, 1973, as amended, between Grantor and United Gas Pipe Line Company.

The Grantor's interests in Lease Nos. OCS-G2224, OCS-G2225, OCS-G2550, OCS-G2559, OCS-G2582, OCS-G2587, OCS-C607, OCS-G2317 and OCS-G2193 are subject to liquids purchase agreements between Grantor and Atlas Processing Company.

The Grantor's interests in Lease Nos. OCS-G2224, OCS-G2225 and OCS-G2193 are subject to gas purchase contracts between Grantor and United Gas Pipe Line Combany.

The Grantor's interests in Lease Nos. OCS-G3284, OCS-G2850, OCS-G2559, OCS-G2882, OCS-G2587, OCS-G2883, OCS-G2. are subject to gas purchase contracts between Grantor and Sea Robin Pipe Line Company.

The Grantor's interest in Lease No. OCS G2701 is subject to a Farmout Agreement dated February 25, 1977, as amended, between Pennzoil Offshore Gas Operators, Inc. (now named Pogo Producita Geopany), et al., as owner or farmor, and Mobil Oil Corporation, et al., as farmee.

AMENDMENT TO PRODUCTION PAYMENT AGREEMENT

THIS AMENDMENT TO PRODUCTION PAYMENT AGREEMENT (hereinafter called this "Amendment"), dated as of November 1, 1977 between Pennzon. Company, a Delaware corporation (hereinafter called "Pennzoil"), and Pennzoil. Louisiana and Texas Offshore, Inc., a Delaware corporation (hereinafter called the "Company"),

WITNESSETH:

WHEREAS, by means of that certain Instrument of Conveyance of Production Payment dated as of March 1, 1574 between the Company and Pennzoil (such Instrument of Conveyance being hereinafter called the "Conveyance"), the Company conveyed to Prunzoil, and Prunzoil purchased from the Company, at the time the first loan was made to the Company under that certain Bank Credit Agreement dated as of March 1, 1974, as thereafter amended and restated as of May 1, 1976, among the Company, The First National Bank of Chicago, and the other banks named therein (such Bank Credit Agreement, as amended and restated, being hereinafter called the "1 - k Credit Agreement"), a production payment dischargeable from the proceeds of production from or applicable to the interests of the Company thereinafter called the "Leasehold Interests"), in those certain oil and gas leases covering lands located offshore a misiana and Texas in the Gulf of Mexico, which interests, leases and lands are described in Exhibit "A" to the Conveyance and in Uxbibit "A" to Appendix 1 to the Production Payment Agreement hereinafter referred to in an amount con, if to the aggregate principal amount of the loans, plus accrued interest and committee at fee, outstanding under the Bank Credit Agreement on the date, if any, upon which the purchase price therefor is paid by Pemzoil to the Company as provided in that certain Production Payment Agreement dated as of March 1, 1974, between Pennzoil and the Company (such Production Payment Agreement being hereinafter called the "Production Payment Agreement'), sol as more particularly provided in the Production Payment Agreement,

Windless, in order to induce the banks referred to in Section 2 of the Bank Credit Agreement (hereinafter called the "Banks") to make bans pursuant to the Bank Credit Agreement and to secure the performance by the Company of its obligations under the Bank Credit Agreement, the Company, by that certain assignment dated as of March 1 1974 (hereinafter called the "Assignment") from the Company to The First National Bank of Chicago, as agent for the Banks (hereinafter called the "Agent"), assigned, transferred and set over to the Agent for the proportionate benefit of the Banks as their interests may appear all the right, title and interest of the Company in, to and under the Production Payment Agreement, subject to the exceptions set forth in the Assignment, and which Assignment was also executed by Pennzoil for the purpose of its making the representations and special agreements set out therein;

Wheneas, the Bank Credit Agreement, the Production Payment Agreement and the Conveyance were amended and restated as of May 1, 1976 to reflect an extension of the maturity of the credit available to the Company under the Bank Credit Agreement, an increase in the amount of that credit to \$200,000,000, the addition of certain banks as parties to the Bank Credit Agreement, the addition to the Leasehold Interests of the interests of the Company in certain oil and gas leases, and for other purposes;

Wheneas, the Bank Credit Agreement is now proposed to be amended to reflect a further extension of the maturity of the credit available to the Company thereunder, an increase in the amount of that credit to \$250,000,000 and the addition of Morgan Chananty Tru t Company of New York and First International Bank in Houston, N.A. as parties to the Bank Credit Agreement, and for other purposes; and

WHEREAS, pursuant to the request of the Company and Pennzoil, the Agent proposes to join herein for the limited purpos forth below;

Now, Therefore, in order to reflect the Amendment to the Bank Credit Agreement proposed as aforesaid, to add to the Leasehold Interests the interests of the Company in certain oil and gas leases covering lands located offshore in the Gulf of Mexico adjacent to the United States of America, and to delete from the Leasehold Interests certain other leases, and in consideration of the premises and of the mutual covenants and agreements herein contained and intending to be legally bound, the parties hereto agree as follows:

- 1. The Production Payment Agreement is hereby amended in the following respects:
- (a) The introductory paragraph of the Production Payment Agreement is amended by changing the date "May 1, 1976" therein to "November 1 1277".
- (b) Section 2.1 of the Production Payment is amended by changing the date "May 1, 1976" therein to read "November 1, 1977".
- (c) Section 2.2 of the Production Payment Agreement is amended by changing the figure "11256" therein to read "10556".
 - (d) Section 2.3 of the Production Payment Agreement is amended to read in its entirety as follows:

"Pursuant to the agreement of the Company and Pennzoil as hereinbefore provided, a production payment was sold and conveyed by the Company to Pennzoil by Instrument of Conveyance of Production Payment dated as of March 1, 1974 between the Company and Pennzoil, and thereafter amended and restated as of May 1, 1976. The terms of such production payment shall be further amended by an Amendment to Instrument of Conveyance of Production Payment to be dated as of November 1, 1977, whereby the aforesaid Instrument of Conveyance of Production Payment shall be amended and restated so as to be in substantially the form of Appendix I hereto."

- (e) Section 2.4 of the Production Payment Agreement is amended by adding the clause, "amended as aforesaid," after the phrase "Following the conveyance of such production payment", by changing the date "January 1, 1985" therein to read "July 1, 1986", and by changing the date "May 1, 1976" therein to read "November 1, 1977".
- (f) Section 2.5 of the Production I syment Agreement is amended by changing the date "December 31, 1984" in paragraph (b) thereof to read "June 30, 1986", and by changing the date "May 1, 1976" in paragraph (c) thereof to read "November 1, 1977".
- (g) The "IN WITNESS WHERFOF" paragraph of the Production Payment Agreement is amended to read in its entirety as follows:

"IN Witness Wheneof, this Restated Production Payment Agreement amends and restates the Production Payment Agreement dated as of March 1, 1974, as heretofore amended and restated, and has been executed in multiple counterparts as of November 1, 1977."

- (h) The signatures and acknowledgements set forth in the Production Payment Agreement are amended to read in their entirety as set forth in Annex I hereto as the signatures and acknowledgements thereto.
- (i) The first paragraph of Appendix 1 to the Production Γayment Agreement is amended by changing the date "May 1, 1976" therein to "November 1, 1977".
- (j) Subparagraph (f) of Paragraph I of Appendix I to the Production Payment Agreement is amended by changing the date "May I, 1976" therein to "November I, 1977".
- (k) Subparagraph (a) of Paragraph 4 of Appendix 1 to the Production Payment Agreement is amended by changing the date "May 1, 1976" therein to "November 1, 1977".

- (1) Subparagraph (b) of Paragraph 4 of Appendix 1 to the Production Payment Agreement is amended by changing the figure "112%" therein to "105%".
- (m) The "In Witness Wittness Wittness Wittness in amended by adding after the date "March 1, 1974," therein the phrase "as heretofore amended and restated,".
- (n) The signatures and acknowledgments set forth in Appendix 1 to the Production Payment Agreement are amended to read in their entirety as set forth in Annex 1 hereto as the signatures and acknowledgments thereto.
- (o) Exhibit "A" to Appendix 1 to the Production Payment Agreement is amended to read in its entirety as set forth in Annex 1 hereto as Exhibit "A" to Appendix 1 thereto.
- 2. Except as amended hereby, the provisions of the Production Payment Agreement shall remain in full force and effect. The Production Payment Agreement, as amended hereby, shall be restat to read in its entirity as set forth in Annex I hereto, it being agreed and understood that all reference in the Production Payment Agreement as amended hereby and so restated to the Bank Credit Agreement shall be deemed to be references to the Bank Credit. Agreement as amended and restate of November 1, 1977.
- 3. The Agent, for and on behalf and in the name of the Banks, hereby consents to the forego amendment to and restatement of the Production Paymen. Agreement, provided, however, that notwithstanding any provision hereof to the contrary, this Anaendment shall not be construed or operate, except as expressly set forth, abuse, to modify in any respect the rights and obligations of the Company or Penazoil under the Production Payment Agreement or the Assignment.

IN WITNESS WITEREOF, the parties hereto have caused this Amendment to be executed by their respective officers thereinto duly authorized as of the day and year first above written.

WITNESSES:

| Sue J. Prejean | Pennzoil Louisiana and Texas Offshore, Inc. [Seal] |
|-----------------|--|
| Sent M. Seil | By Wellesson Vice President |
| Sug Prejean | PENNZOR, COMPANY [SEAL] |
| Tr. E. lanville | Group Vice President THE FIRST NATIONAL BANK OF CHICAGO, as Agent |
| 4 | Relation [Stall |

THE STATE OF TEXAS
COUNTY OF HARRIS
SS:

Be It REMEMBELLED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 1st day of November, 1977:

(Louisiana)

Before me appeared W. B. Wilkerson, to me personally known, who, being by me duly sworn, did say that he is Vice President of Pennzoll Louisiana and Texas Offshole, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W. B. Wilkerson acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared. HAROLD E. SORICH, to me personally known, who, being by me duly sworn, did say that he is a Green-Vice President of Pennzon. Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate scal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that Harold E. Sorion acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared W. B. Wharmson, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Pennzoll Logisiana and Texas Offsi out, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared Haron, E. Sorron, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of Pennzon. Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WHERE OF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 1st day of November, 1977.

[NOTABIAL SEAL]

Notary Public in and for Harris County, Texas My Commission Expires March 21, 1979

Let's STEPCER
Let by Labble resident Hubbs County, Texas
My Count, Calon Expires:

Linda Spencer

THE STATE OF ILLINOIS SS:

Louisiana)

Texas)

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this word of November, 1977:

Before me appeared Richard S. Palmer, to me personally known, who, being by me duly sworn, did say that he is an Assistant Vice President of The Fiber National Bank of Chicago, a national banking association, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that Richard S. Palmer acknowledged the instrument to be the free act and deed of the national banking association.

Before me on this day personally appeared RICHARD S. PALMER, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be an Assistant Vice President of The First National, Bank of Chicago, a national banking association, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed and in the capacity therein stated, and as the act and deed of said national banking association.

IN WITNESS WHEREOF, I have hereunto set my hand and official notarial scal, in the City of Chicago, County of Cook, State of Illinois, the residue of November, 1977.

Clare A. Martz

Notary Public in and for Cook County, Illine's

My Commission Expires

My Commission Expires April 9, 1979

[NOTARIAL SEAL]

PRODUCTION PAYMENT AGREEMENT

THIS AGREEMENT, dated as of March 1, 1974, as amended and restated as of November 1, 1977, between Pennzoii. Control a Delaware corporation (hereinafter called "Pennzoii"), and Pennzoii. Louisiana and Texas Offshole, Inc., a Delaware corporation (hereinafter called the "Company").

WITNESSETH:

ARTICLE, I

Recitals

- 1.1 The Company preposes berein to agree to make available to Pennzoil, or 1. mizoil's designee, for purchase on the terms and conditions provided herein all gas produced from or applicable to the interests of the Company (hereinafter called the "Leasehold Interests") in certain oil and gas leases covering lands located offshore in the Gull of Mexico adjacent to the United States of America, which interests, leases and Lands are described in Exhibit "A" to Appendix 1, attached hereto and hereby made a part hereof.
- 1.2 Pennzoil proposes to purchase from the Company, at the time the first loan is made to the Company under that certain Bank Credit Agreement, dated as of the date hereof, among the Company. The First National Bank of Chicago and the other banks named therein (hereinafter called the "Bank Credit Agreement"), a production payment dischargeable from the proceeds of production from or applicable to the Leasehold Interests in an amount equal to the aggregate principal amount of loans, plus accrued interest and commitment fee, outstanding under the Bank Credit Agreement on the date, it any, upon which the purchase price therefor is paid by Pennzoil to the Company as provided herein (hereinafter called the "Purchase I rice Payment Date"), all as more particularly provided herein.

Now, Therefore, in consideration of t' mutual promises become set forth, it is agreed as follows:

ARTICLE 11

Sale and Purchase of coduction Payment

- 2.1 The Company hereby agains to sell and convey to Pennzoil, and Pennzoil hereby agrees to purchase from the Company, a production payment in an amount equal to the aggregate princip? I mount of the loans, plus accrued interest and commitment fee, outstanding under the Bank Creent Agreement, as amended and restated as of November 1, 1977, on the Perchase Price Payment Daje, if any, which shall be specified by the Company as provided in Section 2.5. The purchase price of such production payment shall be an amount equal to the principal amount thereof.
 - 2.2 The conveyance of production payment
 - (a) shall entitle the holder thereof to an amount equal to interest (computed from the Purchase Price Layment Date) on the unpaid balance of the principal amount of the production payment from time to time remaining outstanding at a rate per amount (based on a year of 365 days) which shall be equal to 165% of the corporate base rate on 90 day commercial loans to its largest and most credit-worthy commercial borrowers in effect at The First National Bank of Chicago from time to time plus \$250, such rate to change automatically and from time to time effective as of the date of each change in such corporate base rate.
 - (b) shall provide that the production payment shall be dischargeable from eighty-five percent (85%) of the proceeds of production from or applicable to the Lease, old Interests,

- (c) shall be effective and as of the day on which such production paymen' shall rive been sold and conveyed to Pennzoil as provided in Section 2.3, and
- (d) shall grant to the holder thereof the rights provided for in, and shall be substantially in the form of, the form of conveyance of production payment attached hereto as Appendix 1.
- 2.3 Pursuant to the agreement of the Company and Pennzoil as hereinbefore provide 1 a production payment was sold and conveyed by the Company to Pennzoil by Instrument of Conveyed ce of Production Payment dated as of March 1, 1974 between the Company and Pennzoil, and there it, amended and restated as of May 1, 1976. The terms of such production payment shall be lartly er amended by an Amendment to Instrument of Conveyance of Production Peyment to be dated as of November 1, 1977, whereby the aforesaid Instrument of Conveyance of Production Payment shall be amended and restated so as to be substantially in the form of Appendix 1 hereto.
- 2.4 Following the conveyance of such production payment, amended as aforesaid, by the Company to Pennzoil as provided in Section 2.3, the Company or its assignce shall have the right at any time prior to July 1, 1986, upon notice given to Pennzoil as provided in Section 2.5, to call upon Pennzoil either
 - (a) to pay to the Company or its assignce the purchase price of asch production payment, or
 - (b) to execute, acknowledge and deliver to the Company, at the Company's expense, such instrument or instruments as may be necessary or appropriate to release, relinquish and extinguish as fully discharged such production payment;

provided, however, that the right referred to in subsection (a) above may not be exercised by the Company unless all certificates of the value of Proven Hydrocarbon Reserves which are required to be delivered to The First National Bank of Chicago under the Bank Credit Agreement, as amended and restated as of November 1, 1977, shall have been so delivered as required and an executed counterpart of each such certificate shall have been delivered to Pennzoil.

- 2.5 Notice of the election by the Company or its assignee to exercise either of the rights referred to in Section 2.4 shall specify the following
 - (a) which of such rights the Cor my or its assignce shall have elected to exercise;
 - (b) a date for closing the transactions contemplated by such right which (i) shall be a business day not less than 30 days after the date upon which such notice is received by Pennzoil unless Pennzoil shall agree in writing to an earlier closing date, and (ii) shall not be later than June 30, 1986, and
 - (c) if the Conseany or its assignce shall have elected to exercise its right to receive from Pennzoil the pure an price of the production payment, the aggregate principal amount of loans, plus accrued interest and commitment fire, outstanding under the Bank Credit Agreement, as amended and restated as of November 1, 1977, as of the Purchase Price Payment Date, such amount to be subject to adjustment as required on and as of such date.

The election by the Company or its assignee, in the manner set forth above, of one of the rights referred to in Section 2.4 shall constitute a waiver of the other such right.

- 2.6 The closing of the transactions referred to in Section 2.5 shall be held at the general offices of Pennzoil located at Pennzoil Place, Houston, Texas, or at such other location in Houston, Texas as Pennzoil shall specify in writing to the Company or its assignee. The closing date for such transactions shall be the date specified in the notice to Pennzoil, and on such closing date
 - (a) if the Company or its assignce shall have elected to receive payment of the purchase price of the production payment, Pennzoil shall pay to the Company or its assignce such purchase price in immediately available funds; or

(b) if the Company or its assignee shall have elected to have the Company receive a reconveyance of the production payment. Pennzoil shall execute, have acknowledged and deliver to the Company such instrument or instruments as may be necessary or appropriate to effect such reconveyance to the Company.

ARTICLE III

Pennzoil's Right to Purchase Gas

- 3.1 The Company shall make available to Pennzoil for purchase by Pennzoil, or such other person, firm or corporation as Pennzoil may designate, all gas produced from or applicable to the Leasehold Interests in accordance with the provisions set forth below.
- 3.2 When in Pennzoil's judgment gas reserves of sufficient quantity and deliverability to justify the construction of necessary facilities enstonairly utilized by pur bases of such gas in performance under gas purchase contracts have been developed on lands covered by any one or more of the Leasehold Interests, Pennzoil's shall have the right to purchase such gas. Pennzoil's right to purchase gas hereunder may be excreised whenever and as often as in Pennzoil's judgment gas reserves of sufficient quantity and deliverability are developed on lands covered by one or more of the Leasehold Interests
- 3.3 The piece 'hereinafter called the "Contract Price") to be paid by Pennzoil for gas purchased from the Company hereinder shall be fixed by the contract between the parties hereto for the sale and purchase thereof. Such contract shall contain specific prices competitive with the contract prices then being included in a salar contracts 6, the purchase of gas from the same area but such prices shall not be less than the applicable area ceiling rate set from time to time by the Federal Power Commission.
- 3.4 The procedures to be employed by Pennzoil and the Company on each exercise by Pennzoil of such right to perchase gas are as follows:
 - (a) Pennzoil shall give the Company notice of its election to purchase gas, which notice shall also specify the purchaser if it be a person, firm or corporation other than Pennzoil, and shall transmit with such notice a form of gas purchase contract providing for the sale and purchase of such gas. After each such election to purchase, on or before the expiration of thirty (30) days following the date upon which the pertinent notice is received by the Company, the Company shall execute the gas purchase contract transmitted therewith and return such contract to Pennzoil, unless the Company shall give Pennzoil notice that the Company is of the opic in that the Contract Price provided for therein does not neet the requirements set forth in Section 3.3, in which event the parties shall promptly thereafter undertake in good faith negotiations to determine by agreement such Contract Price. If such Contract Price has not been determined within sixty (60) days after receipt of the initia' office and form of gas purchase contract by the Company, either the Company or Pennzoil shall have the right to cause such Contract Price to be determined by arbitration as provided in Section 3.7.
- 3.5 When (i) in the Company's judgment gas reserves of a sufficient quantity and deliverability to justify the construction of necessary facilities customarily utilized by purchases of such gas in performance under gas purchase contracts have been developed on lands covered by any one or more of the Leasehold Interests, and (ii) Pennzoil shall not theretofore have given the Company notice of its election to purchase such gas, then and in such event the Company shall have the right to sell such gas in accordance with and subject to the provisions of Section 3.6.
- 3.6 The procedures to be employed by the Company and Pennzeil on each exercise by the Company of its rights under Section 3.5 are as follows:
 - (a) Before entering into any contract for the sale of such gas, the Company shall give Pennzoil notice of its desire to sell such gas and shall transmit with each such notice a form of gas purchase contract providing for the sale and purchase of such gas.

- (b) Pennzoil shall have a perio of thirty (30) days from and after its receipt of such notice from the Company within which to elect to purchase such gas. If Pennyoil elects to purchase such gas, it shall give the Company notice to such effect and shall execute such gas purchase contact and return the same to the Company unless Pennzoil shall be of the opinion that the Contract Price provided for therein does I. I meet the requirements set forth in Section 3.3, is which event the parties shall promptly the caster undertake in good faith negotiations to determine by agreement such Contract Price. If Pennzoil and the Company are unable to reach agreement on such Contract Price within sixty (60) days after receipt by Pennzoil of the aforesaid notice, either Pennzoil or the Company shall have the right to conse such Contract Price to be determined by arbitration as provided in Section 3.7.
- (e) In the event Pennzoil does at a cleet to purchase such gas on or before the expiration of thirty (30) days after receipt of such notice from the Company, the Company shall have the right for a period of sixty (60) days from and after the expiration of thirty (30) days next following the receipt by Pennzoil of such original notice from the Company within which to execute with a third party a contract for the sale of such gas in the form submitted to Pennzoil with the original region from the Company. If such contract is not executed within such sixty (60) day period, such gas shall not the cafter he sold he the Company to a third party, without first making the same available to Penazoil in accordance with the provisions hereof.
- 3.7 If, pursuant to the provisions hereof, either party elects to cause the Contract Price to be provided in a gas pu chase contract to be determined by arbitration, such party shall give the other party notice of such election, which notice shall also name one arbitrator. The party receiving such notice shall, within ten (10) days thereafter, by notice to the other, name the second arbitrator. The two (2) arbitrators so appointed shall name the third, or upon their failing to do so within ten (10) days, the third arbitrator may, upon request of either the Company or Pennzoil, be appointed by the person who at the time is the chief judge of the United States District Court for the Southern District of Texas. The arbitrators so appointed shall promptly hear and determine (after giving the parties due notice of heaving and a reasonable opportunity to be heard) the question submitted, and shall render their decision within sixty (60) days after completion of the hearing. The decision of the arbitraters, or a majority thereof, in which in writing shall be final and binding upon Pennzoil and the Company. The expenses of arbitration, including reasonable compensation to the arbitrators, shall be borne equally by Pennzoil and the Company, except that each party shall bear the compensation and expenses of its counsel, witnesses and employees.

ARTICLE IV

- 4.1 All notices and other communications provided for herein shall be in writing and shall be delivered to the addressee thereof at Pennzoil Place, P. O. Boy 2967, Houston, Texas 77001, or to such other address as the addressee shall have designated by written notice to the party giving such notice or making such communication. The date upon which any such notice or other communication is
- 4.2 The terms and provisions of Article III of this Agreement shall constitute covenants running with the lands and leases affected hereby and shall be binding upon and shall inure to the benefit of the parties hereto, their respective successors and assigns.

JN WITNESS WHEREOF, this Restated Production Payment Agreement amends and restates the Production Payment Agreement dated as of March 1, 1974, as heretofore amended and restated, and has been executed in multiple counterparts as of November 1, 1977.

| Witnesses: Sur J. Preje aus Janet M. Liele | Pennzoil Leuisiana and Texas Offshore, Inc By Wice President ATTEST: |
|--|--|
| | Solly, Hayen |
| | [Corforate Seal] |
| Witnesses: Sung-Preje an Janet M. Lear | By Group Vice President ATTEST: |
| | [Corporate-Seal] |

THE STATE OF TEXAS SS:

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 1st day of November, 1977:

(Louisiana)

Before me appeared W. B. Wilkerson, to me personally known, who, being by me duly sworn, did say that he is Vice President of Pennzon, Louisiana and Texar Offshore, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W. B. Wilkerson acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared HAROLD E. Souron, to me personally known, who, being by me duly sworn, did say that he is a Group Vice President of PENNZON. COMPANY, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that HAROLD E. Souron acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared W. B. WILKERSON, known to me to be the person whose name is subscribed to the foregoing austrament and known to me to be a Vice President of Pennzon. Louisiana and Texas Offshour. Inc. a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared Paronn E. Sonron, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of Pennzone Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expresses, and as the act and deed of said corporation.

IN WITNESS WITHEOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 1st day of November, 1977.

[NOTABIAL SEAL]

Notary Public in and for Harris County, Texas My Commission Expires March 31, 1977

Lunda Jones

Not by Public in ... for hairs County, Texas My Contribution Legitest

APPENDIX I

TO

PRODUCTION PAYMENT AGREEMENT BETWEEN

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC. AND

PENNZOIL COMPANY

CONVEYANCE OF PRODUCTION PAYMENT

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Conveyance") dated as of March 1, 1974, as amended and restated as of November 1, 1977, between Pennson Louisiana and Texas Offshora. Inc., a Delaware corporation (the "Grantor"), having its principal office and post office address at Pennson Place 4 O. Box 2967, Houston, Texas, and Pennson Company, a Delaware corporation (the "Grantee"), having its principal office and post office address at Pennsol Place, P.O. Box 2967, Houston, Texas.

WITNESSETH

Winamas, the Grantor desires to sell and the Grantee desires to purchase the Production Payment hereinafter described,

Now, Thenerone, it is agreed by and between the Grantor and the Grantee as follows:

- Certain Definitions. The following terms shall be used in this Conveyance with the meanings given below:
 - (a) "Accounting Month" shall me in any monthly period commencing with and including the 21st day of any calendar month and ending on and including the 20th day of the next succeeding calendar month.
 - (b) "Effective Date" shall mean 7 o'cleek, A.M., on the date of this Conveyance, according to the time then in effect at the location of each Subject Interest
 - (c) "Exhibit A" shall mean Exhibit A attached to this Conveyance and hereby made a part hereof for all purposes.
 - (d) "Hydrocarbons" shall mean all oil, gas and other hydrocarbons or any combination of one or more of such substances appropriate to the context in which such term is used.
 - (e) "Lease" shall mean any oil and gas lease or sub-lease, described in Exhibit A.
 - (f) "Proceeds Commencement Date" shall mean 7 o'clock, A.M., according to the time then in effect at the location of each Subject Interest, on the date upon which the purchase price is paid by the Grantee to the Grantor as provided in the Production Payment Agreement dated as of March 1, 1974, as amended and restated as of November 1, 1977, between the Grantor and the Grantee.
 - (g) "Subject Hydrocarbons" shall mean all Hydrocarbons in and under, and which may be preduced and saved from, and which shall accrue and be attributable to, the Subject Interests from and after the Proceeds Commencement Date and throughout the period specified in this Conveyance for the term of the Production Payment.
 - (h) "Subject Interests" shall mean the undivided percentage interests of the Grantor in the Leases, and in addition thereto, each and every other kind and character of right, title, claim or interest which the Grantor now has in, to or under the Leases, which interests or Leases are

either specifically or generally described in Exhibit A, all as the same shall be enlarged by the discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and all renewals and extensions of any of the same. but there shall not be included within the Subject Interests any rights which the Grantor now has or may hereafter obtain with respect to production attributable to the microsts of noncoasenting parties under any operating agreement, unit operating agreement, contract for development or similar instrument.

2. Conveyance of the Production Payment. The Canator, for valuable consideration, the receipt adequacy and sufficiency of which are hereby acknowledged, by these presents does hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, . . a production payment, an undivided eighty five percent (85%) interest in the proceeds from the sale of the Sebject Hydro-

The aforesaid eighty-five percent $(S_{2}(i))$ interest in the proceeds from the sale of the Subject Hydrocarbons, as granted, bargained, sold, conveyed, assigned, transferred and delivered unto the Grantee by the terms of this Conveyance, shall be the "Production Payment."

- To HAVE AND TO HOLD the Production Payment, together with all and singular the rights, privileges and appurtenances thereunto in any way belonging, unto the Grantee, its successors and assigns,
- 3. Certain Provisions Governing Subject Hydrocarbons. All the provisions of this Conveyance shall be subject to the following:
 - (a) For the discharge of the Production Payment the Grantee shall look exclusively to the proceeds of Subject Hydrocarbons, and the Granto, shall not be liable for such discharge.
 - (b) The Production Payment shall not be dischargeable from the proceeds of any products to the requiring processing or refining operation, except to the extent of that pertion of such proceeds which represents the lan marker value at the wellhead of the Subject
- (c) The Production Payment shall not be dischargeable out of any homes which the Granton shall receive for any Lease or assignment of any of the Subject Interests or out of any payments made to the Grantor in connection with the drilling or deterring of drilling of any well on any of the Subject Interests or in consection with any adjustment of any well and leasehold equipment upon unitization of any of the Subi et Interests.
- (d) There shall not be included in the Subject Hydrocarbons any Hydrocarbons unavoidably lost in the production thereof or produced and saved from any of the Subject Interests and used by the Granter in conformity with good field practices for drilling and production operations by the triangle in comorning was good need provided for until good provided including gas injection, recordary recovery, pressure maintenance, repressuring or cycling operations) conducted for the purpose of producing Subject Hydrocarbons from the Subject operations) conducted for the purpose of producing ourget from any unit to which the Subject Interests we committed, but only so long as such Hydrocarbons are so used.
- (e) So long as and to the extent that the same may be required by applicable laws and regulations, in the case of any Lease from the United States of America included in the Subject Interests, free which the average production of oil per well per day averaged on the monthly basis is 15 bessels or less, the obligation to pay and the right of the Crantee to receive the proceeds of oil produced from such Lease shall be suspended until said average prod. tion of oil per well per day exceeds said minimum amount, and sue's suspension shall apply separately to any zone or Fortion of such Lease segregated for computing government royalty.
- 4. Amount and Term of the Production Payment. The Production Payment shall continue and remain in full force and effect until such time as the Grantee shall have received and realized,

out of the proceeds from the sale of the Subject Hydrocarbons, free and clear of all development, operating, mining, producing, treating, processing, bandling, storing, marketing, transporting and other costs and expenses of every kind whatsoever, the full aggregate sum of the following amounts:

- (a) The aggregate principal amount of loans, plus accrued interest and commitment fee, outstanding on the Proceeds Commencement Date under the Bank Credit Agreement, dated as of March 1, 1971, as thereafter amended and restate 1 as of November 1, 1977, among the Grantor, The First National Bank of Chicago and the other banks named therein (hereinafter called the "Primary Sum"); plus
- (b) An amount computed at a rate per annum (on the basis of a 365-day year) which shall be equal to 105% of the corporate base rate on 90-day commercial loans to its largest and most credit-worthy commercial borrowers in effect at The First National Bank of Chicago from time to time plus 12%, such rate to change automatically and from time to time effective as of the date of cach change in such corporate base rate, from and including in Proceeds Commencement Date on the unliquid sted balance from time to time of the Primary Sum (the first such computation to be made on the 21st day of the month next following the Proceeds Commencement Date for the period from and including the Proceeds Commencement Date to and including the 20th day of the month next following the Proceeds Commencement Date to and including the 20th day of the month next following the Proceeds Commencement Date on the amount of such unliquidated balance on the Proceeds Commencement Date, and subsequent computations to be made monthly on the first day of each Accounting Month for the proceding Accounting Month on the amount of such unliquidated balance on the first day of such proceding Accounting Month); plus
- (c) An amount equal to the aggregate of all amounts which are pair! by the Grantee to any state or political subdivision thereof on account of ad valorem, transfer, mortgage, gross production, gross receipts, income, profits, severance, occupation, sales use, franchise and other taxes and assessments or any kind whatsoever, including penalties and interest, imposed or assessed with espect to or measured by or charged against or attributable to the Production Payment, the Subject Hydrocobous or the proceeds thereof, together with an amount equal to interest on the unliquidated amount thereof at the rate of 7% per amount from the date of payment of each such amount by the Grantee, provided, however, that the Granter shall have the right at its expense to outest any such taxes or assessments in good fair! on behalf of the Grantee.

IT BEING THE INTENTION OF THE GRANTOR AND THE GRANTEE IN ENTERING INTO THIS CONVEYANCE that the Grantee shall be entitled to receive and realize out of the Subject Hydrocarbons the full aggregate sum of the amounts described in subparagraphs (a) and (b) of this Paragraph 4, free and clear of all costs and expenses with respect to the Subject Interests and the Subject Hydrocarbons (which expenses and costs shall be borne by the Grantor) and over at 1 above all taxes, expenses and costs of the character described, specified or returned to in subparagraph (c) of this Paragraph 4

- 5. Termination of the Production Payment. When the full aggregate sum of the amounts specified in subparagraphs (a), (b) and (c) of Paragraph i of this Conveyance (as increased as expressly provided by the terms of this Conveyance) has been received by the Grantee, the Production Payment shall forthwith terminate and all interest therein shall immediately revert to and become vested in the Grantee to the same extent and with the same force and effect as if this Conveyance had not been made. The Grantee agrees that, upon such termination, it shall execute or cause to be executed, upon the request and at the expense of the Grantee, such instruments as may be necessary or appropri to to evidence the termination of the Production Payment.
- 6. Application of the Proceeds of Subject Eydroca-bons. For all psaposes of this Conveyance, the proceeds of Subject Hydrocarlosas as tually received by the Chantee prior to the close of business on 20th day of the month next following the Proceeds: Commencement Date shall be deemed to have been received and applied immediately after the opening of business on the 21st day of such month, and thereafter, such proceeds actually received by the Grantee after the opening of business on the first day of each Accounting Month and prior to the close of business on the last day of such Account-

ing Month shall be deemed to have been received and applied immediately after the opening of business on the first day of the next succeeding Accounting Month, provided, however, that, if any date of application specified above shall be a Saturday, Sunday or legal holiday under the law of the jurisdiction in which such proceeds are actually received by the Grantee, such proceeds shall be deemed to have been received and applied on the last business day next preceding such Saturday, Sunday or legal holiday in such jurisdiction, but the amount of such proceeds to be applied pursuant to subparagraph (a) of this Peragraph 6 shall nevertheless be the amount under subparagraph (b) of Paragraph 4 of this Con cyance accorded up to (but not including) such Saturday, Sunday or legal holiday. Such proceeds shall be deemed to have been applied on the first day of each Accounting Month as follows:

- (a) First, to the amount referred to in subparagraph (b) of Paragraph 4 of this Conveyance accrued up to (but not including) the date of such application, and
- (b) Second, to the amount referred to in subparagraph (c) of said Paragraph 4, to the extent then ascertained, and
 - (c) Third, to the reduction of the unliquidated balance of the Primary Sun;

provided, however, that, in the event such proceeds so applied on the first day of any A counting Month shall be insufficient to cover the full amount specified a subparagraph (a) of this Paragraph 6, the undiquidated balance of the Primary Suna shall forthwith be increased (to the extent permitted by law) by an amount equal to the amount of such deficiency.

Within 10 days after the date of each such application by the Grantee of any proceeds of Subject Hydrocarbons, the Grantee will furnish or cause to be furnished to the Granter a written statement showing the application of such proceeds in accordance with the provisions of this Paragraph 6.

- 7. Marketing the Subject Hydrocarbons. The Grantor, at its expense, shall market, or cause to be marketed, the Subject Hydrocarbons for the best price reasonably available when marketed, and it shall not be necessary for the Grantee to join in any Sales Contracts (as hereinafter defined) or any amendment thereof. In the event the Grantor shall take any of the Sebject Hydrocarbons for its own use, the Grantor shall pay to the Granter, on or before the last day of the Accounting Month during which such taking occurs, a price equal to the fair market value at the wellhead of "Subject Hydrocarbons so taken.
- 8. Sales Contracts. The Grantor will duly perform all obligations binding on it under all contracts and agreements for the sale of Subject II discurbens or any portion thereof, whether presently existing or hereafter entered into (all such contracts and agreements being the "Sales Contracts"), in accordance with the terms bereof and will take all appropriate measures to enforce the performance under each of the Sales Contracts of the obligations of the purchaser thereunder. All Subject Hydrocarbons sold by the Grantor, whether purchasers to the terms of the Sales Contracts or otherwise, shall be delivered by the Grantor to the purchasers thereof into the pipelines to which the wells producing such Subject Hydrocarbons may be connected or to such other point of purchase as is reasonably required in the marketing of such Subject Hydrocarbons.
- 9. Withholding and Restitution of Proceeds of Subject Hydrocarbons. All obligations of the Grantor hereinder shall be subject to the applicable provisions of the Natural Gas Act and all applicable rules and regulations of the Federal Power Commission. Rates permitted under the Natural Gas Act and said roles and regulations to be paid for gas included in Subject Hydrocarbons shall be controlling if varying from prices established in Sales Contracts or if different from the fair market price at the wellhead. The Grantor shall be entitled to use its reasonable discretion in making filings with the Federal Power Commission affecting Subject Hydrocarbons. If any proceeds of Subject Hydrocarbons shall be eithhold for any reason who sever including, without limitation, withholding by the Grantor pending final approval of rate measures filed with the Federal Power Commission the Grantee shall not be deemed to have received or realized such proceeds until, and only to the extent that, such proceeds shall actually have been received by the Grantee. If, at any time before

or after receipt of the full aggregate som specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance, the Grantee shall be compelled, for any reason whatsoever, to make any payment or restitution, or such payment or restitution shall be made for the account of the Grantee, on account of any proceeds of Subject Hydrocorbons, which proceeds have already been applied pursuant to the provisions of Paragraph 6 of this Conveyance, the unliquidated balance of the Primary Sum shall forthwith be increased by the amount of the proceeds so paid over by or for the account of the Grantee plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penaltics; provided, however, that if at the time of such payment or restitution the entire Primary Sum shall have been liquidated, the Primary Sum shall be deemed not to have been so liquid, ted and the unliquidated balance of the Primary Sum shall be thereupon deemed to be in the amount of the proceeds so paid over plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties.

- 10. Protection to Purchasers of Production. No person purchasing or taking or processing any Subject Hydrorarbous shall be required to take notice of, or keep informed as to, the discharge or termination of the Production Payment until the actual receipt by such person of written notice advising such person of such discharge or termination.
- 11. Payment of Taxes. The Grantor will tender and pay, or cause to be paid, punctually, before they shall become delinquent (or as to any thereof which are being contested in good faith, promptly after the final determination of such contest), all ad valorem taxes (or taxes imposed in lieu thereof) and a transfer, mortgage, gross production, mining, severence, occupation, excise, sales, use, franchise, income and other taxes and assessments of every Lind and character whatsoever (other than federal income taxes of the Grantee) imposed or assessed with respect to or measured by or charged against or attributable to the Subject interests, the Production Payment or the Subject Hydrocarbons, or the proceeds thereof, or against the Grantee by reason of its acquisition or ownership of the Production Payment or against any assignee of the Production Payment by reason of his or its interest as assignee, together with any interest or penalty payable in connection therewith.
- 12. Covenants of the Grantor. Until such time as the Production Payment has terminated as provided in Paragraph 5 of this Conveyance, the Grantor will, at its expense and regardless of who may be the operators of the Subject Interest, cause:
 - (a) The Subject Interests to be maintained and developed and continuously operated for the production of Hydrocar sons in a good and workmanlike manner and in accordance with sound field practices and all applicable federal, state and local laws, rules and regulations (except those being contested in good faith); provided, however, if the Grantor in good faith and in conformity with sound field practices elects not to participate in any operation which is to be conducted under the terms of an operating agreement, unit operating agreement, contract for development or similar instrument entered into by the Grantor or its predecessors in title pertaining to the development and operation of any Subject Interest (or any portion thereof) and which allows any consenting party or parties to conduct nonconsent operations, then, during the period or periods in which the consenting party or parties are entitled to receive Subject Hydrocarbons attributable to the Subject Interest (or portion thereof) by the terms of such instrument pending the recomponent or receipt of the sums determined thereunder, the Grantee shall not be entitled to receive or be deemed to have received (but only during such period or periods) proceeds from Subject Hydrocarbons attributable to such Subject Interest (or portion thereof) to the extent affected thereby.
 - (b) All rentals, royalties, and all liabilities of any kind or nature, including, without limitation, all liabilities for labor, material, supplies and equipment incurred in, or arising from, the administration, operation or development of, or the gathering producing, treating, processing, storing, marketing or transporting of, the Subject Hydrocarbons, to be paid punctually when due, or, as to any thereof which are being contested in good faith, promptly after the final determination of such contest.

- (c) All machinery, equipment and facilities of any kind now or hereafter located on the Subject Interests and necessary or useful in the operation thereof for the production of Hydrocarbons therefrom to be provided and to be kept in good and effective operating condition and all repairs, renewals, replacements, additions and improvements thereof or thereto needed for such purpose to be duly made or provided.
- (d) All necessary and proper steps to be taken diligently to protect and defend the Subject Interests, the Subject Hydrocarbons and the proceeds thereof against any adverse claim or demand, including, without limitation, the employment or use of counsel for the prosecution or defense of litigation and the contest, settlement, release or discharge of any such claim or demand.
- (e) Written notice to be given to the Grantee of every adverse claim or demand made by any person (including any government or governmental agency) affecting the Subject Interests, the Subject Hydrocarbous or the proceeds thereof in any manner whatsoever, or of any proceeding instituted with respect thereto.
- (f) The interest of the Grantor in each of the Subject Interests to be kept free and clear of liens, charges and encumbrances of every character, other than those which are of a character customarily found in councetion with comparable drilling and producing operations and which do not materially adversely affect the operation of the Subject Interests and other than those becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Conveyance.
- (g) Subject to the previsions of Paragraph 19 of this Conveyance, all Leases included in or relating to the interest, described or referred to in Exhibit A and under which the Grantor is lessee, sub-lessee, or assignee to be maintained in full force and effect without amendment, modification or waiver of any provision of any such I ase which might materially adversely affect the rights of the Grantee under this Conveyance.

Anything contained in this Paragraph 12 to the contrary notwithstanding, the Grantor shall not be obligated to perform undertakings performable only by the operator and which are beyond the control of the Grantor, with respect to individual Subject Interests of which it is not the operator. In such case, however, the Grantor will use its best efforts to secure the performance of any such undertakings.

- 13. Access to the Subject Interests. The Grantor shall permit any one or more representatives designated by the Grantee to make any inspection of the Subject Interests at any reasonable time as the Grantee may reasonably request. In addition the Grantor shall furnish to the Grantee such detailed information as the Grantee may reasonably request in waving concerning the Subject Interests, the operation thereof and the production and sa'e of the Subject Hydrocarbons.
- 14. Posting and Unitization. Anything herein to the construct notwithstanding, the Granter may, without the consent of the Grantee, pool, combine, communitize or unitize any Subject Interest (or any portion thereof) from time to time with other leasehold, mineral or other interests to form units when, in the judgment of the Granter, it is necessary or advisable to do so in order to facilitate the orderly development or operation (including repressuring, pressure maintenance, cycling, water flood and secondary recovery operations) of any Subject Interest (or any portion thereof). Forthwith, after the formation of any unit, the Granter shall furnish to the Grantee a true copy of the pooling agreement, unit designation or any other instrument creating such unit, together with, upon request of the Grantee, any operating agreement relating thereto. The interest in any such unit accruing or attributable to any Subject Interest (or any portion thereof) included therein shall be subject to the Production Payment in the same manner and with the same effect as though such unit were in existence on the Effective Date. The signature or joinder of the Grantee shall not be necessary to the instruments evidencing the pooling and unitization herein authorized.

- 15. Default by the Grantor and the Remedies of the Grantee. Should the Grantor default in the performance or observance of any covenant or agreement provided in this Conveyance to be performed or observed by the Grantor, and should such default continue unremedied for 30 days (or for such longer period as may be necessary to complete performance diligently commenced within 30 days) from the delivery to the Grantor of written notice of such default, the Grantee may, in addition to pursuing any or all other remedies available at law or in equity:
 - (a) Effect performance or observance of such covenant or agreement on behalf of and at the expense of the Grantor, in which event the Grantee may expend funds reasonably necessary to effect such purposes and shall be entitled to reimbursement therefor from the Grantor, together with interest on such amounts at the rate of 7% per amount from the date of such expenditure to the date of reimbursement.
 - (b) Upon written notice to the Gamtor, succeed to all rights of the Grantor with respect to the possession, operation and development of the Subject Interests, and may use in connection therewith all of the appurtenant property, both real and personal, situated upon or used or held for future use in connection therewith, or its connection with the production, storing, transporting or sale of the Subject Hydrocarbons, and the Grantee shall have the right, on behalf of and for the account of the Grantor, to sell the Subject Hydrocarbons and the Grantor shall, upon demand, reimburse the Grantee for all amounts so expended by the Grantee, together with interest on such amounts at the rate of 7% per amount from the date of such expenditure to the date of reimbursement.
 - (c) Apply to a court of equity for the specific performance or observance of such covenant or in aid of the execution of any power herein granted and for the appointment of a receiver of the Subject Interests and the Subject Hydrocarbons.
- 16. Exercise and Termination of the Remedies of the Grantee. All rights and remedies granted by the provisions of Paragraph 15 of this Conveyance and otherwise belonging to the Grantee shall be cumulative and no exercise of any right or remedy shall produce the exercise of any other right or remedy. Any right or remedy may be exercised on one or more occasions with at exhaustion thereof. All rights and remedies shall terminate:
 - (a) When the Production Payment is discharged and all amounts then due and payable to the Grantee pursuant to the terms of Paragraph 15 of this Conveyance (including amounts payable for interest) shall have been paid in full; or
 - (b) At any earlier time when all defaults of the Grantor shall have been remedied and all such amounts shall have been duly paid in full, without prejodice, however, to the exercise of any rights and remedies conferred upon the Grantee upon any subsequent failure of the Grantor to perform or observe any of the covenants or agazements herein provided to be performed or observed by the Grantor.
- 17. Reports to the Grantee. While the Production Payment remains in force and effect, the Grantor will, at its expense, furnish to the Grantee within 45 days after the end of each calendar month, a report showing for such calendar month (i) the quantity of Subject Hydrocarbons produced and sold or taken by the Grantor for its use and (ii) the gross proceeds received from the sale, or attributable to such taking, of such Subject Hydrocarbons.
- 18. Warranty of Title. The Grantor covenants and warrants that at the time of the execution and delivery of this Conveyance, the Grantor has good and marketable title to the Subject Interests, subject only to such exceptions, charges, liens or encumbrances, if any, as are set forth or referred to in Exhibit A; that the Grantor has the power and right to create and confer upon the Contect the

rights of the Grant or granted by the terms of this Conveyance and that there are no suits or proceedings pending or, to the knowledge of the Gran or, threatens a paint or affecting the Subject Interests before any court or by or before any governmental containistic, bureau or any regulatory authority which if decided adversely to the interests of the Grantor shad materially affect the rights of the Grantee pursuant to this Conveyance. The Grantor here by Lack its successors and assigns, to warrant and defend forever the rights under this Conveyance of the Grantee, its successors and assigns, against the lawful claims and demands of a representation who shall claim the same or any part thereof, and the Grantor dill indometry and held are Grantee harmless against and from all loss suffered by the Grantee resulting from any Laure of or defect in the Constor's title to the Subject Interest, and the production of Hydrocarbon is warranted as aforesaid. The coverants and warrantees of this Paragraph 18 shall durvice the discurse of the Production Payment.

19. Transfer of the Subject Interests. Except for the conveyance of production a cuts becoming effective only after termination of the Production Pryment as provided in Par. 122 C this Conveyance, prior to such tenderation the Grantor shall not, without the written was consist the Grantee, make or permit, any surve der, abandonment, sale, co aveyance, goognament, a se or se lease, in who e or at part, of go, of 1. Subject Interests; provided, however, that the Grantor shan no of the right t surrender, i.b. n ... c herwise terminate (either voluntarily or involuntarily) any of the Subject Interests (or n portion thereof) when there is no well located thereon which is capable of producing Hydroca book in paying quantities and when the Granter has reasonably determined that (i) no existing w. h., b. recompleted thereon as a well capable of producing Hydrocarbons in paying quantities, . .) no additional well could be drilled and completed thereon which would be capable of products. "Androcurbous, over and above the cost of drilling and comparing same, in paying quantities, and (iii) no pressure maintenance or secondary recovery operators may be instituted with respect to any well thereon which would cause such well to be capable of producing Hydrocarbons, over and above the cost of instituting and completing such pressure maintenance or secondary recovery execution, in paying quantities. For the purposa's of this Paragraph 19 the words "in paving quantities" as applied at any time to the product - capability of any well shall mean that, in the light of conditions existing at the time of determinent and which reasonably appear to be not temporary, such well is producing or is or will be an income of producing Subject Hydrocarbons whose aggregate value exceeds or will exceed the Grant as share (as the owner of the Sapje ! Interest on which such well is located, of the direct cost of operating such well.

In each case where the preceding paragraph of this Paragraph 19 permits the Grantor to zurrender, abandon or otherwise terminate any of the Subject Interests (or any portion thereof), then (a) any of the Subject Interests (or any portion thereof) which are so surrendered, abandoned to terminated shall upon surrender, abandonment or termination cease to be a Subject Interest, and (b) the Grantor, in lieu of surrendering, abandonment or terminating any of the Subject Interests (or any portion thereof), may sell, assign, sublease, farmout or convex the same free and clear of the Production payment without the consent of the Grantee, and the Grantee shall not have the right to any proceeds from such sale, assignment, sublease, farmout or conveyance; provided, however, that are interest in any such Subject Interest (or any portion thereof) which may be excepted from such sale, assignment, sublease, farmout or conveyance; all retained by the Grantor shall not be released thereby from the Production Payment.

- 20. Exception to Payments. The a sprance by the Grantee of any payment hereunder shall not proclude the strange from taking exception to the correctness of the amount thereof; provided, however, that any such exception must be specified by the Grantee in a notice given to the Grantee within 12 months after the date such payment is made.
- 21 Transfer of the Production Payment. Nothing continued in this Conveyance shall be construed to linat or restrict it any way the right of the Grante to sell, convey, assign or mortgage the Production Payment, in sel ple or in part.

- 2. Rights of Mortgagee or Trustee. If the Grantee shall at any time execute a moregage or deed of trust covering all or part of the Production Payment, the mortgage or trustee thereis named or the holder of any obligation occurred thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to exercise all of the rights, remedies, powers and privileges confered up in the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained becaused by the Grantor, but the provisions of this Paragraph 22 shall in no way be deemed or construed to impose upon the Grantor any obligation or liability undersafen by the Grantee under such mortgage or deed of trust or under the obligation occurred then
- 22. Notices. Any notice, request, demand, report or other instrument which reads be required or permitted to be a very or furnished to or served upon either peatly hereto or other person succeeding to any interest of citer, perty hereto shall be deemed sufficiently given or furnished and cerved if in writing and delivered to such the type or person or to an officer of such party of person or deposited in the United States small read eithed envelope, regiment by with postage prepare adds so if to the Grantor to Pennzoll Trace Mad. Box 2967, Houston, Texas, and if to the Grantee to the azoil Place, P.O. Box 2067, Houston, resus, or to such other address as the party or person to be addressed shall have designated by written notice to the party of person to be addressed shall have designated by written notice to the party of person to be addressed shall report or making such request or demand.
- 24. Further issurances, so long as permitted by 10%, the law so to do, the Grantor will execute and deliver all such other and additional instruments, of these, releases, requitament and other documents, and will do all such after acts and things, as may be accessary more fully to assure to the Granton all the rights and interests herein and hereby granted or intended so to be.
- 25 In ding Effect. All the covenants and agreements of the Grantor contained in this Conveyant small be defined to be covenants running with the hard and small be binding upon all sace are in interest to, and all assigns of the Grantor and shall inture to the benefit of all successors in interest to, and all assigns of the Grantoe. The provisions of this Conveyance shall inture to the benefit of and be binding upon the respective successors and assigns of the Granton and the Grantee from and after acceptance by the Grantoe. All references herein to either the Grantor or the Grantee shall include their respective successors and assigns.
- 26. Release or Failure c. Title. No release by the Grantes of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect of reducing the amount of the procees, from the sale of production of Hydrocasbons out of which the Production Payment is dischargeable or reducing the amount of the Production Payment or creating any offset, from the Production Payment or creating any offset, from the Production Payment shall continue in full force and effect as to the production of the Subject Interest until the full amount of the Production Payment has been received by the Grantee as herein provided.
- 27. Final Termination. Notwithstanding the other provisions here a, the resolution Payment shall, in any event, terminate 1 day prior to the expiration of 21 years after the death of the survivor of all the descendants of Theodese Roosevell, late President of the United States, who are living on the date hereof.
- 28. Headings, Paragraph headings used in this Conveyance are for convenience only and shall not alread the construction of this Conveyance.

IN WITS 88 WHER 94, this Restated Instrument of Conveyance of Production Payment amends and restates the Instrument of Conveyance of Production Payment dated as of March 1, 1974, as been accorded and restated and the parties have caused this Restated Instrument of Conveyance of Production Payment to be duty executed in the presence of the understand witnesses on the dates of their respective acknowledgments annexed hereto, in several counterparts, each of which is an original and all of which are identical. Each of such counterparts shall for all purposes be deemed

to be an original, and all ruch counterparts together shall constitute but one and the same Instrument. The date of this Instrument shall be deemed for all purposes to be the date first above written.

| **** STORE OF THE THE PURE STORE OF THE STOR | 1 minute at the man and more at their |
|--|---|
| Witnesses: | PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC. |
| 31 1 0 0 10 0000 1 100000 | By Vice President |
| nd na na na natao - na (e managambana) na 1440 - 1990 an | ATTEST: |
| | Secretary |
| | [Corporate 51 at 7 |
| Witnesses: | |
| | PENNZOIL COMPANY |
| | By Group Vice President |
| THE CHARLEST THE STREET, AND S | ATTEST: |
| | Secreta ·· |
| | [Conporate Seal] |
| | ** E |
| | * 43 |
| | 2° x |

THE STATE OF THY
COUNTY OF 1:
\$55:

BE IT READ DEPEND that I, the undersigned, a Notery Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, and this 1st day of November, 1977;

(Lauisiana)

Before me appeared W. B. Wilkelson, to me personally known, who, being by me duly sworn, did say that he is Vice President of Lenszon. Louisiana and Texas Originous, Inc., a Delaware corporation, and that the seal affixed to the foregoing metament is the conservate seal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that W. B. Wilkelson acknowledged the instrument to be the free act and deed of the corporation.

(Lauisiana)

Before me appeared Hyboth E. Sorron, to a copersonally known, who, being by me duly swore, did say that he is a Group Vice President of Pennzon. Company, a Deligage corporation, real that the seal affixed to the foregoing instrument is the corporate seal of said corporation and abat the instrument was signed and sealed in behalf of the corporation by anthonity of its Board of Directors and that Haron E. Souron acknowledged the instrument to be the free act and deed of the corporation.

(l'exas)

Before me on this day personally appeared W. B. Wakerson, known to me to be the person whose name is subscribed to the foregoing in rument and known to me to be a Vice President of Pennzon. Louisiana and Texas Orisione, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared Haronn E. Sonron, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Group Vice President of Pension. Company, a Delayare corporation, and acknowledge I to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WILLTOF, I have heleculto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Proceedings of November, 1977.

Notary Public in and for Harris County, Texas My Commission Expires

EXHIBIT A TO CON' EVANCE OF PRODUCTION PAYMENT

| BLM or State Lease No. | 250 | Description Federal leases are as shown on Official ter Continental Shelf Leasing Maps) | Graptor's Undivided Percentage Interest |
|---------------------------|---------------|--|--|
| OCS-G2539 | 12 1-7- | All of Block 352, West Cameron Area, West Addicion, OCS Cafficial Leasing Map, Louisiana Map No. 14 | 25.00% |
| OCS-G2224 | 2-1-73 | Block 532 Wes, Cameron Area, South Addition, Official Learning Map, Louisiana Map No. 1B | 26.67% |
| OCS-G2225 | 2-1-73 | Block 553 West Cancron Area, South Addition, Official Leasing Map, Louisians Map No. UB | 26.67% |
| OCS-G3284 | 8-1-75 | All of Block 563, 'Vest Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 1B | 25.00% |
| OCS-G2228 | 1-1-73 | Block 590 West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 21.2* 256% |
| OCS-G2870 | 12-1-74 | All of Block Congress Area, South Addition, OCS Official Leasing Map, Loui Jana Map No. 1B | 6.66667% |
| OCS-G2559 | 5-1-74 | All of 10% k 617 West Cameron Area — South Addition, Official Leasing Map, Louisrana * p No 1P | 6.6666797 |
| OCS-G285 | 12-1-74 | . If of Block 236 East Cameron Area, , with Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.66667% |
| UCS G2S60 | 12-1-74 | All of block 237, East Cameron Area. South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.6666798 |
| OCS-G2889 | 12-1 74 | All of Block J25, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 6.60367% |
| OCS-G2583 | 12-1-74 | All of Block 127, South Mars's Island Area, South Addition, GCS Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS G2587 | 5-1 74 | All of Block 128 South Jarsl Island Area, South Addition, Official Leasing Map, Louisiana Map No. (C | 6.66607% |
| OCS-G2884 | 12-1-74 | All of Elock 129, South Tarsh Island Area, South Addition, OCS Official Lesing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2885 | 12-1-74 | All of Block 141, South Marsh asland Area, South Addition, OCS inficial Leaving Map, Louisiana Map No. SC | 6.66667% |
| UCS-G2900 | 12-1-74 | All of Black 261, Euge te Island A OCS Official Leaving Map, 1 ct Map 185, 4 | 16.66667% |
| OCS-G3158 | 7-1-75 | All of Block 262. Europe added Vea, as shown on CCS volad Learning Map, Louisiana Map No. 4 | 16.66667% |

| BLM or State Lease No. | Effective Date | Description | Grautor's Undivided Percentage Interest |
|---------------------------|-------------------|---|--|
| OCS-G2607 | 5-1-74 | All of Block 312 Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A | 26.66667% |
| OCS-G2317 | 2-1-73 | Block 333 Eugene Island Area, South Addition, Official Leasing Map, Louisiana Map No. 4A | 26.67% |
| OCS-G2177 | 11-1-72 | All of Block 49 South Pass Area as shown on official leasing map, Louisiana Map No. 9 | 10.00% |
| State Lease No. 6310 | 5-13-74 | Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and being more particularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Louisiana | 20.00% |
| OCS-G2185 | 10-1-72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court entered December 13, 1935 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288), South Pass Area, South and East Addition, as shown on official leaving map, Louisiana Map No. 9A | 15.00% |
| OCS-G3193 | 7 -1-75 | S½ of Block 59, Main Pass Area, that portion located in Zone 2 as that zone was defined in the agreement between the United States and the State of Louisiana, October 12, 1956 and landward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 25.00% |
| OCS-G3195 | 7-1-75 | Block 72, Main Pass Area, that portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 21.00% |
| OCS-G2947 | 12-1-74 | All of Block 73, Main Pass Area, OCS Official Leasing Map. Louisiana Map No. 10. That portion located more than one foot seaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1977)) | 21.00% |
| CS-G2193 | 10-1-72 | All of Block 140 Main Pass Area as shown on official leasing map, Louisiana Map No. 10 | 10.00% |
| O °S G3202 | 7-1-75 | All of Block 947, Ewing Bank Area, as shown on OCS Official Protraction Diagram NH-15-12 | 21.00% |
| | | | |

| | COMULIATION OF | THOSection Thanks (community) | |
|---------------------------|-------------------|---|--|
| BLM or State Lesse No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
| OCS-G3203 | 7-1-75 | All of Block 903, Ewing Bank Area, as shown on OCS Official Protraction Diagram NH-15-12 | 21.00% |
| OCS-G3206 | 7-1-75 | All of Plock N663E63, Mobile South No. 2, as shown on OCS Official Leasing Map, NII-16-10 (Also known as Block 63, Mississippi Canyon Area, as shown on OCS Official Protraction Diagram NII-16-10) | 10.00% |
| OCS-G2358 | 8-1-73 | All of Block A-442, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2366 | 8-1-73 | All of Block A-474, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2372 | 8-1-73 | All of Block A-489, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2373 | 8-1-73 | All of Block A-496, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 16.00% |
| OCS-G3118 | 4-1-75 | All of Block A-499, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 1B | 16.40% |
| OCS-G2377 | 8-1-73 | All of Block A-519, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 16.43835% |
| OCS-G2378 | 8-1-73 | All of Block A-520, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2383 | 8-1-73 | All of Block A-535, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 13.33333% |
| OCS-G2701 | 7-1-74 | All of Block A-540, High Island Area - South Addition, OCS Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2704 | 7-1-74 | All of Block A-545, High Island Area – South Addition, OCS Official Leasing Map, Texas Map No. 7B | 10.00% |
| OCS-G2779 | 10-1-74 | All of Block A-546, High Island Area, South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2705 | 7-1 | All of Block A-547, High Island - South Addition, OCS Official Leasing Map, Texas Map | 14.00% |
| OCS-G2708 | 7-1 7 . | No. 7B A't of Block A-54S, High Island Area – South Addition, OCS Official Leasing Map, Texas Map No. 7B | 14.00% |
| OCS-G2394 | 8-1-73 | All of Block A-555, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 14.50% |
| | | - 14 - | |

EXHIBIT A
TO
CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| BLM or State Lease No. | Effective Date | Description | Grantor's Undivided Percentage Interest |
|---------------------------|-------------------|--|--|
| OCS-G2713 | 7-1-74 | Ali of Block A-562, High Isla d Area – South Addition, OCS Official Leaving Map, Texas Map No. 7B | 12.715% |
| OCS-G2388 | 8-1-73 | All of Piock A-563, High Island Area, "outh Addition, Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2389 | 6-1-73 | All or Block A-564, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2399 | 8-1-73 | All of Block A-570, High Island Area. South Addition. Onicial Leasing Map, | |
| OCS-G2718 | 7-1-7-1 | Texas Map No. 78 All of Block A-573 High Island Area — South Addition, OCS Official Leasing | 20.00% |
| OCS-G2719 | 7-1-74 | Map, Texas Map No. 7B All of Block A-552 High Island Area — South Addition, OCS Official Leasing | 20.00% |
| OUS G2397 | 8-1-73 | Map, Texas Map No. 7B All of Block A-272, High Island Ar. a, East Addition, Scoth Extension. Official | 14.118% |
| OCS-G2308 | 8-1-73 | Leaving Map, Texas Map No. 7C All of Block A-275, Urah Island Ar. a, | 15 .33% |
| OCS-G2399 | 8-1-73 | East Addition, South Extension, Odicial Leasing Map, Texas Map No. 7C All of Block A-274, High Island Area, | 12.00% |
| OCS-G2403 | 8-1-73 | East Addition, South Extension, Official Leasing Map, Tevas Map No. 7C All of Block A-279, High Island Area, | 13.8461% |
| OCS-C3314 | 4-1-76 | East Addition, South Extension, Official Leasing Map, Texas, Map, No. 7C All of Block A-278, High Island Atea, | 17.00% |
| 003-03-114 | 4-1-70 | East Addition, South Extension, as shown on OCS Official Leasing Map, | 17 0000 |
| OCS-G2409 | 8-1-73 | All of Block A-312, High Island Area, East Addition, South Extension, | 17.00% |
| OCS-G2411 | 8-1-73 | Official Leasing Map, Tevas Map No. 7C All of Block A-314, High Island Area, East Addition, South Extension. | 12.00% |
| OCS-G2737 | 7-1-74 | Official Leasing Map, Texas Map No. 7C All of Block A-316 High Island Area = East Addition — South Extension, | 22.85714% |
| OCS-G2413 | 8-1-73 | OCS Official Leaving Map, Texas Map No. 7C All of Block A-329, High Island Area, East Addition, South Extension. | 14.00% |
| OCS-G2414 | 8-1-73 | Olicial Leasing Map, Texas Map No. 7C All of Block A-323, High Island Area, | 14.00% |
| | | East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 14.00% |

| BLM or State Lease No. | Effective Date | Description | Grantor's Undivided Percentagee Interest |
|---------------------------|-------------------|--|---|
| OCS-G2416 | 8-1-73 | All of Block A-325, Value Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 20.86357% |
| OCS-G2418 | 8-1-73 | All of Block A-327, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G2422 | 8-1-73 | All of Block A-352, High Island Area, East Addition, South Extension. Official Leasing Map. Texas Map No. 7C | 10.00% |
| OCS-G2739 | 7-1-74 | All of Block A-339 High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 34.00% |
| OCS-G2426 | 8-1-73 | All of Block A-340, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 34.90% |
| OCS-G2429 | 8-1-73 | All of Block A-351, High Island Area, East Addition, South Extension. Official Leasing Map, Texas Map No. 7C | 15.25% |
| OCS-G2745 | 7-1-74 | All of Block A-355, High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 11.33% |
| OCS-G2746 | 7-1-74 | All of Block A-356, High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 15.00% |
| OCS-G2766 | 7-1-74 | All of Block A-403 High Island Area — East Addition — South Extension, OCS Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G3417 | 1-J-77 | That portion of Main Pess Blocks 72 and 74 located more than three geographical miles seaward from the fine described in the supplemental decree of the U.S. Supreme Court, June 16, 1975, (U.S. vs Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971), as shown on OCS Official Leasing Map, | |
| OCS-G3109 | 1-1-77 | Louisiana Map No. 10 All of Block 351, Eugene Island Area, South Addition, as shown on OCS Official Leasing | 25.00% |
| OCS-G3416 | 1-1-77 | Map, Louisiana Map No. 4A That portion of South Pass Block 57 located more than three geographical miles seaward from the line described in the supplemental decree of the U.S. Supreme Court, June 16, 1975; (U.S. vs. Louisiana, 422 U.S. 13) to one foot seaward of the Third Supplemental Decree Line (404 U.S. 388, December 20, 1971), as shown on OCS Official Leasing Map, Louisiana, 422 U.S. 1971, as shown on OCS Official Leasing Map, Louisiana, 422 U.S. 1971, as shown on OCS Official Leasing Map, Louisiana, 422 U.S. 1971, as shown on OCS Official Leasing Map, Louisiana, 422 U.S. 1971, as shown on OCS Official Leasing Map, Louisiana, 422 U.S. 1971, as shown on OCS Official Leasing Map, Louisiana, 423 U.S. 1971, as shown on OCS Official Leasing Map, Louisianana, 423 U.S. 1971, as shown on OCS Official Leasing Map, Louisian | 25.00% |
| OCS-G3500 | 8-1-77 | ana Map No. 9 All of Block 253, West Cameron Area, as shown on OCS Official Leasing Map, Louisiana Map | 20,00% |
| | | No. 1 | 30.00% |

This Conveyance is subject to the terms and provisions of the above-described oil and gas leases and accordingly, the Subject Hydrocarbons and the proceeds therefrom shall be calculated after deducting the proportionate share of the royalties reserved in each such lease attributable to Grantor's said undivided percentage interest in such lease.

Lease Nos. OCS-G2838, OCS-G2225, OCS-G2850, OCS-G2559, OCS-G2607, OCS-G2317, OCS-G3193, OCS-G3195, OCS-G2917 and OCS-G3417 referred to above are each subject to an Operating Agreement between Mobil Oil Corporation, as operator, and the Grantor, et al. as non-operators.

Lease No. OCS-G2224 referred to above is subject to an Operating Agreement between Chevron Oil Company, as operator, and the Granton, et al., as non-operators.

Lease Nos. OCS-G3254, OCS-G2228, OCS-G2859, OCS-G2860, OCS-G2882, OCS-G2883, OCS-G2887, OCS-G2884, OCS-G2883, OCS-G3203, OCS-G2884, OCS-G3284, OCS-G3203, OCS-G3286, OCS-G3272, OCS-G3118, OCS-G3104, OCS-G2884, OCS-G2713, OCS-G2388, OCS-G2389, OCS-G2719, OCS-G2399, OCS-G3179, OCS-G3179,

Lease Nos. OCS-G2177, OCS-G2193 and OCS-G3206 referred to above are each subject to an Cherating Agreement between Gulf Oil Corporation, as operator, and the Grantor, et al., as non-cherators.

Lease Nos. GCS-G2355, GCS-G2403 and GCS-G3314 referred to above are each subject to an Operating Agreement between Marathon Oil Company, as operator, and the Granter, et al., as non-operators.

Lease Nos. OCS-G2373. OCS-G2377, OCS-G2378; OCS-G2411, OCS-G2737, OCS-G2413, OCS-G2414, OCS-G2416, OCS-G2745 and OCS-G2746 referred to above are each subject to an Operating Agreement between Cities Service Oil Company, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-C2363, OCS-G2469 OCS-G2418 and OCS-G2422 referred to above tite. A subject to an Operating Agreement between Sun Oil Company (Delaware), as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2350 and OCS-G2715 referred to above are each subject to an Operating Agreement between Exxon Corporation, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2397 and OCS-G2398 referred to above are each subject to an Operating Agreement between Aminoil Development, Inc., as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2704, OCS-G2779, OCS-G2705 and OCS-2706 are each subject to an Operating Agreement between Texaco Inc., as operator, and the Grantor, et al., as non-operators.

The Grantor's interests in Lease Nos. OCS-G2224. OCS-G2225, OCS-G2228 and OCS-G2185 referred to above are subject to Advance Payment Agreements dated as of June 13, 1973, as amended, between Grantor and United Gas Pipe Line Company.

The Grantor's interests in Lease Nos. OCS-G2221 OCS-G2225, OCS-G2550, OCS-G2559, OCS-G2582, OCS-G2557, OCS-G2607, OCS-G2317 and OCS-G2193 are subject to liquids purchase agreements between Grantor and Atlas Processing Company.

The Grantor's interests in Lease No. OCS-G2224, OCS-G2225 and OCS-G2193 are subject to gas purchase contracts between Grantor and United Gas Pipe Line Company.

The Grantor's interests in Lease Nos. OCS-G3284, OCS-G2550, OCS-G2559, OCS-G2582, OCS-G2587, OCS-G2583, OCS-G2000, OCS-G3150, OCS-G2607 and OCS-G2317 are subject to gas purchase contracts between Granton and Sea Robin Pipe Line Company.

The Grantor's interest in Lease No. OCS-G2701 is subject to a Farmout Agreement dated February 25, 1977, as amended, between Pennzoil Offshore Gas Operators, Inc. (now named Pogo Producing Company), et al., as owner or farmor, and Mobil Oil Corporation, et al., as farmee.

12.

.

-

2

LEMLE, KELLEHER, KOHLMEYER & MATTHEWS

EIGHTEENTH FLOOR

LOUIS G. LEMET
HARRY B. RELEBHIN
HARRY C. SOUCH N. C
CHARLES ROME MET M. JAN.
LOUGHOUS B. MATTHING
H. MATTHIN MELNEY, JAR.
LOUNG C. C. MATTHING
LOUIS C. MATTHING

800

FUACH ORTEN, UR.
FRETERING F. MYNAUS
MILLIAM F. CONT.
MIL

FIRST NATIONAL BANK OF COMMERCE BUILDING

NEW ORLEANS 70112

TELEPHONE (504) 586-1241 CABLE ADDRESS: LEMMOR TELEX 564272

May 12, 1976

MACK E. BARHAM OF COUNSEL

ALAN M. GOODMAN
JAMES M. PETERSTN
EDWAND F. ROMANE, J
BTANLET A. MILLAN
GLOROF FRAZIFR, 12
FREVOR G. ROTAM
BTUCE M. HONACH
J. DONICE ALVERSON
MARK S. STEIN
RICHARD W. STEMMORS
JOHN M. SARTIN, JR.

Departmen of the Interior Bureau of Land Management Outer Continental Shelf Office Suite 3200, The Plaza Tower 1001 Howard Avenue New Orleans, Louisiana 70113

Attention: Mr. John L. Rankin, Offshore Manager

Re: Leases Nos. OCS G-2839, 2224, 2225, 3284, 2228, 2850, 2559, 2859, 2860, 2882, 2883, 2537, 2884, 2885, 2900, 3156, 2607, 2317, 2177, 2185, 3193, 3195, 2947, 2191, 2193, 2195, 3202, 3203, 3206

Offshore Louisiana
Leases Nos. OCS G-2351, 2358, 2366, 2372, 2373, 3118, 2377, 2378, 2383, 2701, 2704, 2779, 2705, 27(6, 2384, 2713, 2388, 2389, 2390, 2718, 2719, 2397, 2398, 2399, 2403, 3314, 2409, 2411, 2737, 2413, 2414, 2416, 2418, 2422, 2739, 2426, 2429, 2745, 2746,

ore Texas

Ger

are an original executed copy and 69 Xerox copies of each of the following documents:

- Amendment to Production Payment Agreement dated as of May 1, 1976 between Pennzoil Company and Pennzoil Louisiana and Texas Offshore, Inc.
- Amendment to Instrument of Conveyance of Production Payment dated as of May 1, 1976 between Pennzoil Louisiana and Texas Offshore, Inc. and Pennzoil Company.

RECEIVED

D MONT.

Department of the Interior Bureau of Land Management May 12, 1976 Page 2.

 Confirmation of Assignment dated as of May 1, 1976 among Pennzoil Louisiana and Texas Offshore, Inc., Pennzoil Company and The First National Bank of Chicago, as Agent for certain banks.

We request that you file the original executed copy of each of these documents, along with this letter, in your file applicable to Lease No. OCS G-2839. Please also place an additional copy of this letter and a Xerox copy of these documents in your file applicable to each of the remaining captioned leases for cross-referencing purposes.

We understand that the payment of a fee in connection with this filing is not required by applicable regulations, and that no approval is necessary. Please return one enclosed copy of this letter to evidence such filing and the accomplishment of our request.

Yours very truly,

LEMLE, KELLEHER, KOHLMEYER & MATTHEWS

Murphy Moss

MM/cd Enclosures

Filed and accomplished as requested.

1976

John L. Rankin, Manager

AMENDMENT TO

PRODUCTION PAYMENT AGREEMENT

THIS AMENDMENT TO PRODUCT¹⁶ a "AYMENT AGREEMENT (hereinafte, ea ⁷ed ⁴las "Amendment"), dated as of May 1, 1976 p — een Pennzon. Contrant, a Delaware corporate inafter called "Pennzon"), and Pennzon ¹ outsiana and Texas Orasionu. Inc. a Delaware corporation (hereinafter called the "Company")

WITNESSEIN:

Whereas, by means of that certain Instrument of Conveyance of Production Payment a ated as of March 1, 1974 between the Company and Pennzoil (such fortuneent of Conveyance being hereinafter called the "Conveyance"), the Company conveyed to Ferrizoil, and Penrizoil p reheated from under that certain Bank Credit the Company, at the time the first loan was made to the Com-Agreement dated as of March 1, 1974, as thereafter amended and states as of May 1, 1976, among the Company, The First National Bank of Chicago, and the other mand mained therein (such Bank Credit Agreement, as amended and restated be up hereinafter called the "Bank Credit Agreement"), a production payment lischargeable from the proceeds of production from e, applicable to the interests of the Company (herein 'ter called the 'Lease' 11 Interests'), in those certain oil and gas leases covering lands located offshore Louisiana and Texas in the Gulf of Mexico, which interests, leases and lands are described in Exhibit "A" to the Conveyance and in Exhibit "A" to Appendix 1 to the Produc tion Payment Agreement hereinafter referred to mean amount equal to the aggregate principal amount of the loans, plus accrued interest and commutment fee, outstanding under the Baok Credit Agreement on the date, if any, upon which the purchase price therefor is paid by Pennzall to the Company as provided in that ce,tai: Production Layment Agreement d. t. C. a. of March 1, 1974, between Pennzoil and the Company (such Production Physical Agreement being Inscinal - extent the "Production Payment Agreement"), all as more particularly provided in the Production Payment Agreement;

WHEREAS, in order to induce the banks referred to in Section 2 of the Bank Credit Agreement (hereinafter called the "Banks") to in the leaves pursuant to the Bank Credit Agreement and to secure the performance by the Company of its obline are under the Bank Credit Agreement, the Company, by that certain assignment dated a set March 1, 1974 (hereinafter called the "Assignment") from the Company to The First National Pass of Chicago, as expent for the Banks (hereinafter called the "Agent"), assigned true-car dead set over to the Agent for the proportionate benefit of the Banks as their interests may a sexual the fact the and interest of the Company and to and under the Production Payment Agreement such as their exceptions of the the Assignment and which Assignment was also executed by the most for the purpose of its moding the representations and special agreements set out the reset and

WHEREAS, passward to the respect of the Company and Pennzor! the Arent proposes to join herein for the limited purposes set forth below;

Now, Transcours in solver is add to the "early-bloom streets the interests of the Company in certain oil and as because ang lands but they assembly being the as in the Gulf of Mexico adjacent to the cheed States of America, and in counds to med the possess and of the mutual covernants and agreements berein contained and interesting to be legally bound, the parties bereto agree as follows:

- 1. The Production Payment Agreement is hereby accorded a the following respects:
- (a) Section 2.1 of the Production Payment Agreement is concided by adding there in the clause, as amended and restated as of May 1, 1966, after the word "Agreement".
- (b) Section 2.3 of the Fredhiction Parison it Agreement is anomaled by adding therein the clause.", as annoaled and restated as of Moy 1, 1976, after the word "Agreement".

- (c) Section 2.4 of the Production Payment Agreement is an acided by changing the date "January 1, 1983" therein to read "January 1, 1985", and by Adrig therein the clause ", as amended and resta of as of May 1, 1976," after the word "Agreement".
- (d) S som 2.5 of the Production Payment Agreement is amended by changing the date "December '1, 1982" in preceipth (b) thereof to read "December 31, 1984", and by adding therein the clause "as inscaled and estated as of May 1, 1976," after the word "Agreement.".
- (e) Section 2.6 of the Production Payment Agreement is amended by changing the address "900 Southwest Tower" therein to "Pennzeil Place"
- (f) Section 4.1 of the Production Payment Agreement is amended by changing the address "900 Southwest Tower, Houston, Texas 7700," to "Penozoil Place, P. O. Box 2967, Houston, Texas 77001".
- (g) The "IN Wirkiss Wire Form" paragraph of the Presbaction Payment Agreement is amended to read in its entirety as follows:

"In Whiten's Winners, this Restated Production Payment Agreement amends and restates the Production Payment Agreement dated as of March 1, 1974 and has been executed in multiple counterparts as of May 1, 1976,"

- (h) The signatures and acknowledgments set forth in the Production Payment Agreement are amended to read in their entirety as set forth in Armex 1 hereto as the signatures and acknowledgments thereto.
- (i) The first paragraph of App. adix 1 to the Production Payment Agreement is amended to read in its entirety as follows:

"THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (dis "Conveyance") dated as of March 1, 1974, as amended and restated as of May 1, 1976, between Planzon Louisiana and Texas Official Inc., a Dilloware corporation (the "Granton"), having its principal office and pot office address at Pennzon Planc, P.O. Box 2967, Houston, Texas, and Prinzon Conveya, a 1901 evane corporation (the "Grantec"), having its principal office and post office address at Pennzon Plance, P.O. Box 2967, Houston, Texas,".

- (j) Subparagraph (f) of Paragraph 1 of Appendix 1 to the Production Payment Agreement is amended by adding therein the charse ", as anneaded and restated as of May 1, 1976," after the number "1974".
- (k) Sub-paragraph (a) of Paragraph 4 of Appendix 1 to the Production Payment Agreement is amended by adding therein the chars, "a is thereafter amended and restated as of May 1, 1976," after the number "1974".
- (1) Paragraph 23 of Appendix 1 to the Production Favorant Agreement is amended by changing the address "900 Southwest Tower" to "Prompod Place, P. O. Box 2367" at each occurrence therein.
- (iii) "The "Is Wrixess Winester" paragraph of Appendix 1 to the Prometion Payment Agreement is amended to a admits entirely a follow:

"Is Winness Windows, this Restated Instrument of Consequence of Production Payment amends and restates the Instrument of Consequence of Production Instrument of Conveyance of March I, 1974, and the parties have care of this Restated Instrument of Conveyance of Production Payment to be duly assented in the presence of the underigned witnesses on the dates of their respective a Production and all of whole are made. I as heat on a constraint small for all purpose, by a larger to be an original and all uses constraints to rether shall constitute but one and the same Instrument. The date of this Instrument shall be decreated for all purpose to be the date first above written."

- (n) The signatures and acknowledgments set forth in Appendix 1 to the Production Payment Agreement are amended to read in their entirety as set forth in Annex I hereto as the signatures and acknowledgments thereto.
- (o) Exhibit "A" to Appendix 1 to the Production Payment Agreement is amended to read in its entirety as set forth in Annex 1 hereto as Exhibit "A" to Appendix 1 thereto.
- 2. Except as annualed hereby, the provides softhe Production Payment Agreement shall remain in full force and effect. The "anduction Paym at Agreement, as amended hereby, shall be restated to read in its estately as set forth in Amery I I seto, it being agreed and understood that all references in the Production Payment Agreement as amended hereby and so restated to the Bank Caedit Agreement shall be deemed to be references to the Bank Credit Agreement as amended and restated as of May 1, 1976.
- 3. The Agent, for and on behalf and in the name of the Banks, hereby consents to the foregoing amendment to and restatement of the Production Payment Agreement; provided, however, that not withstanding any provision hereof to the contrary, this Amendment shall not be construed or operate, except as expressly set forth above, to modify in any respect the rights and obligations of the Company or Pennzoil under the Production Payment Agreement or the Assignment

IN WITNESS WILLREOF, the parties hereto have caused this Amendment to be executed by their respective officers there—sto-duly authorized as of the day and year first above written.

| their respective officers there—to duly authorized | as of the day and year first above written. |
|---|--|
| Witnesses: Blitis C. Cittle Le oneder (Clapethe) | PINNOR LOUISIAN, AND TEXAS OF SHORE, INC. By Control of the Contr |
| Blacin & Stitle Benedic (Clampter) | PENNZOH COMPANY (SEAL) No Company (SEAL) |
| Carmoth Discharge | ISEAL |

THE STAIL OF TEXAS COUNTY OF HARRIS (88)

By Ir Restrainable that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 6 ds day of May, 1976.

(Louisiana)

Before me appeared White exist, it is to me personally known, who being by me duly sworn, did say that he is Vice President of Priszon Louisiana and Trias Orismone, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that to the extremely ple acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared W.B. Callines of to me personally known, who, being by me duly sworn, Cid say that he is Vice President of Presson Company, a Delaware corporation, and that the scal affixed to the foregoing instrument is the corporate scal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that W.C. Callee we acknowledge 1 be instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared W. P. Erwald, J. Tr., known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Preszon Loristes and Texas Oristion. Is a ** Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared while the largest known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Planzon. Company a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consider tion therein expressed, and as the act and deed of said corporation.

In Witness Windows, I have been onto set my hand and official notarial seal, in the City of Hourton, County of Harris, State of Texas, this $\mathcal{L}^{t,h}$ day of May, 1976.

Lette of Haye Hacter Notary Public in and for Harris County, Texas

My Commission Expires James 1, 1977

THE STATE OF LILINOIS COUNTY OF COOK SS:

Bi. It Reservance that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, or this 1 day of May, 1976:

(Louisiana)

Before me appeared $\{\cdot,\cdot\}$ and $\{\cdot,\cdot\}$ by $\{\cdot,\cdot\}$, to me personally known, who, being by me duly sworn, did say that he is $\{\cdot\}_{I\in\mathcal{I}}^{n}\}_{I\in\mathcal{I}}^{n}$ of Tin. First National Bank or Cincaco, a national banking association, and that the veal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Bo... In cettors and that $U(L)_{I\in\mathcal{I}}$ and L be a Vacknowledged the instrument to be the free act and deed of $\{\cdot,\cdot\}$ national banking association.

('rexas)

' fore me on this day personally appeared + \(\cdot \cdot \dagger L \cdot \cdo

 Weines Witherot, I have becomes set my hand and official notarial scal, in the City of Chicago, County of Cook, State of Illinois, this Z_i day of May, 1976.

Notary Public in and for Cook County, Illinois

My Commission Expires

II., Comil In Explies April 9, 1979

PRODUCTION PAYMENT AGREEMENT

THIS AGREEMENT, disked as of March I, 1974, as amended and restated as of May 1, 1976, between Penzzon Contenny, a Delaware corporation (becominater called "Pennzoil"), and Penzzon. Lowerman and Tryes Orisinous Inc., a Delaware corporation (becomparer called the "Company"),

WITNESSETH:

ARTICLE I

Recituls

- 1.1 The Company purposes herein to agree to make available to Pennzoil, or Pennzoil's designee, for purchase on the terms and conditions provided herein all gas produced from or applicable to the interests of the Company (hereins on collect the "Leuschold Interests") in certain oil and gas leaves covering lands located offshore in the Carlott Mexico adjacent to the United States of America, which interests, leaves and lands are described in Exambit "A" to Appendix 1, attached hereto and hereby made a part hereof.
- 1.2 Pennzeil proposes to purchase from the Company, at the time the first loan is made to the Company under that certain Bank Credit agreement, dated as of the date hereof, among the Company. The First National Bank of Chicae, and the other banks named therein ("encinafter called the "Bank Credit Agreement", a production payment dechargeable from the proceeds of production from or applicable to the Leasehold Interest in an amount equal to the aggregate principal amount of loans, plus accured interest and commitment fee, outstanding under the Bank Credit Agreement on the date, if any, upon which the purchase price therefor is paid by Pennzoil to the Company as provided herein (hereinafter called the "Purchase Price Payment Date"), all as more particularly provided herein.

Now, Theratour, in consideration of the materal promises herein set forth, it is agreed as follows:

ARTICLE II

Sale and Purchase of Production Payment

2.1 The Company a reby agrees to sell and convey to Pennzoil, and Pennzoil hereby agrees to purchase from the Company, a production payment in an amount equal to the aggregate principal amount of the loans, plus accured intenset and commitment fee, outstanding under the Bank Credit Agreement as amended and restated as of May 1, 1976, on the Purchase Price Payment Date, if any, which shall be specified by the Company as provided in Section 2.5. The purchase price of such production payment shall be an amount equal to the principal amount thereof.

2.2 The conveyance of production payment

- (a) shall entitle the holder thereof to an amount equal to interest (computed from the Purchase Price Payment Date) on the unpaid balance of the principal amount of the production payment from time to time remaining outstanding at a rate per annum (based on a year of 365 days) which shall be equal to 11% of the corporate base rate on 90 day commercial loans to its largest and most credit worthy commercial betrowers in effect at The First National Bank of Chicago from time to time place 1,5%, such rate to change automatically and from time to time effective as of the date of each change justice corporate base rate.
- (b) shall provide that the production payment shall be dischargeable from eighty-five percent (85%) of the proceeds of production from or applicable to the Leasehold Interests,
- (c) shall be effective from and as of the day on which such production payment shall have been sold and conveyed to Penazoil as provided in Section 2.3, and

- (d) shall grant to the hobber thereof the rights provided for in, and shall be substantially in the form of, the form of conveyance of production payment attached hereto as Appendix 1.
- 2.3 Such postuction payment shall be sold and conveyed by the Company to Pennzoil on the data that the first loan is a add to the Company under the Baul. Credit Agreement, as amended and rest, to does of May 1, 1956. Our sould date, the Company shall execute, have acknowledged and deliver to Econzoil a conveyance of production payment in sub-radially the form of Appendix 1, with the blands presently appropriate these in appropriately completed.
- 2.4 Following the consequence of such production payment by the Company to Pennzoil as provided in Section 2.3, the Corogany or its assigner shall have the right at any time prior to Journal 1 1985 upon notice given to Pennzoil as provided in Section 2.5, to call upon Pennzoil either
 - (a) to pay to the Company or its assigned the purchase price of such production payment, or
 - (b) to execute, acknowledge and deliver to the Company, at the Company's expense, such instrument or instruments as may be necessary or appropriate to release; relinquish and extinguish as fully discharged such production payment,

provided, however, that the right induced in an subsection (a) above may not be exercised by the Company and as all certificates of the value of Proxin Hydrocardae. Resultes which are required to be collevered as The First National Bank of the ago much the Bank viscolit Agreement, as amended and restated as at May 1. 1975 of all himse been so delivered as required and an executed counterpart of each such a trivial is shall incorporated by Proposition.

- 2.5 Notice of the election by the Company as issuess, mee to exercise either of the rights referred to in Sec. on 2.4 shall specify the rollowing.
 - (a) which of such rights the Conq by or its assignce shall have elected to exercise;
 - (b) a date for the ing the transactions contemplated by such right which (i) shall be a business day not less than 30 days after the date upon which such notice is received by Pennzoil unless Pennzoil shall agree in writing to an earlier closing date; and (ii) shall not be later than December 31, 1954, and
 - (c) if the Company of its assigned shall have elected to exercited a field to receive from Pennzoul the purchase price of the production payment, the agent gate principal amount of loans, plus accound noticest and commutation for ourstanding under the Bank Could Agreement, as amended and to tasted accord May 1, 1976, as of the Purchase Price Payment Date; such a number to be subject to adjustment as required enough as of such date.

The electron by the Company of its assignar, in the normal set faith above, of one of the rights reterred from Section 2.45 bill on soften a ways not the other such right.

- 2.6 The choice of the transactions referred to in Section 2.5 shall be held at the general offices of Pennizori located at Penni oil Plan. Houston Texas, or at such other location in Houston, Texas as Pennizoril shall grady in writing to the Company or its assignmen. The closing date for such transactions shall be the date specified in the notice to Pennizoril and ensuch closing late.
 - (a) if the Company or its assigner shall have elected to receive payment of the purchase price of the production payment. Permixed shall pay to the Company or its assigner such purchase price in named fels available tonds, or
 - (b) if the Company of its a some shall have charted to have the Company receive a reconservation of the production person of Teneral shall a vecite, have acknowledged and deliver to the Company such instruments or instruments as may be necessary or appropriate to effect such reconservance to the Company.

ARTICLE III

Pennzoil's Right to Purchase Gas

- 3.1 The Company shall make available to Pennzoil for purchase by Pennzoil, or such other person, firm or corporation as Pennzoil may designate, all gas produced from or applicable to the Leasehold Interests in accordance with the provisions set forth below.
- 3.2 When in Pennzoil's judgment gas reserves of sufficient quantity and deliverability to justify the construction of necessary facilities customarily utilized by purchasers of such gas in performance took case purchase contracts have been developed on lands covered by any one or more of the Leasehold Interests, Pennzoil's shall have the right to purchase such gas. Pennzoil's right to purchase gas hereunder may be exercised whenever and as often as in Pennzoil's judgment gas reserves of sufficient quantity and deliverability are developed on lands covered by one or more of the Leasehold Interests.
- 3.3 The price (hereinafter called the "Contract Price") to be paid by Pennzoil for gas purchased from the Company hereunder shall be fixed by the contract between the parties hereto for the sale and purchase thereof. Such contract shall contain specific prices competitive with the contract prices then being included in similar contracts for the purchase of gas from the same area but such prices shall not be less than the applicable area ceiling rate set from time to time by the Federal Power Commission.
- 3.4 The procedures to be employed by Pennzoil and the Company on each exercise by Pennzoil of such right to purchase gas are as follows:
 - (a) Pennzoil shall give the Company notice of its election to purchase gas, which notice shall also specify the purchaser if it be a person, firm or corporation other than Pennzoil, and shall transmit with such notice a form of gas purchase contract providing for the sale and purchase of such gas. After each such election to purchase, on or before the expitation of thirty (30) days following the date upon which the pertinent notice is received by the Company, the Company shall execute the gas purchase contract transmitted therewith and return such contract to Permzoil, unless the Company shall give Pennzoil notice that the Company is of the opinion that the Contract Price provided for therein does not meet the requirements set forth in Section 3.3, in which event the parties shall promptly thereafter undertake in good faith negotiations to determine by agreement such Contract Price. If such Contract Price has not been determined within sixty (60) days after receipt of the initial notice and form of gas purchase contract by the Company, either the Company or Pennzoil shall have the right to cause such Contract Price to be determined by arbitration as provided in Section 3.7.
- 3.5 When (i) in the Company's judgment gas reserves of a sufficient quantity and deliverability to justify the construction of necessary facilities customarily utilized by purchasers of such gas in performance under gas purchase contracts have been developed on lands covered by any one or more of the Leasehold Interests, and (ii) Pennzoil shall not the retofore have given the Company notice of its election to purchase such gas, then and in such event the Company shall have the right to sell such gas in accordance with and subject to the provisions of Section 3.6.
- 3.6 The procedures to be employed by the Company and Pennzoil on each exercise by the Company of its rights under Section 3.5 are as follows:
 - (a) Before entering into any contract for the sale of such gas, the Company shall give Pennzoil notice of its desire to sell such gas and shall transmit with each such notice a form of gas purchase contract providing for the sale and purchase of such gas.
 - (b) Pernzoil shall have a period of thirty (30) days from and after its receipt of such notice from the Company within which to elect to purchase such gas. If Peanzoil elects to purchase such gas, it shall give the Company notice to such effect and shall execute such gas

purchase contract and return the same to the Company unless Pennzoil shall be of the opinion that the Contract Price provided for therein does not meet the requirements set forth in Section 3.3, in which event the parties shall promptly thereafter undertake in good faith negotiations to determine by a present such Contract Price. If Pennzoil and the Company are unable to reach agreement on such Contract Price with a sixty core days after receipt by Pennzoil of the aforesaid notice, either Pennzoil or the Company shall have the right to cause such Contract Price to be determined by arbitration, as provided in Section 3.7.

- (c) In the event Francoil does not clear to purchase such the company shall have the right for a period of sixty (60) days from and after the expiration of thirty (30) days next following the receipt by Pennzuil of such enigned netter from the Company within which to execute with a third party a contract for the sale of such gas in the form submitted to Pennzuil with the original notice from the Company. If such contract is not executed within such sixty (60) day period, such gas shall not the reafter be sald by the Company to a third party, without first making the same available to Pennzuil in accordance with the provisions hereof.
- 3.7 If, pursuant to the provisions hereof, either party elects to cause the Contract Price to be provided in a gas purchase contract to be determined by arbitration, such party shall give the other party notice of such election, which notice shall also mane one arbitrator. The party receiving such notice shall, within ten (10) days it creater, by notice to the other, mane the second arbitrator. The two (2) arbitrators so appointed shall make the third, or upon their failing to do so within ten (10) days, the third arbitrator may, upon request of either the Company or Pennzoil, be appointed by the person who at the time is the chird judge of the United States District Court for the Southern District of Texas. The arbitrators so appointed shall promptly hear and determine (after giving the parties due notice of lessing and a reasonable opportunity to be heard) the question submitted, and theil resolve their decision within sixty (60) days after completion of the learning. The decision of the arbitrators, or a majority thereof, made in writing shall be final and binding upon Pernsoil and the Company. The expenses of arbitrators, including reasonable compensation to the arbitrators, shall be borne equally by Pennzoil and the Company. Such cache party shall bear the compensation and expenses of its counsel, witnesses and employ on

ARTICLE IV

- 4.1 All notices and other communications provided for herein shall be in writing and shall be delivered to the addressee thereof at Permanil 19 as P. O. Bax 2967. Houston, Texas 7.000, or to such other address as the address as shall have does mated by written notice to the party giving such notice or making such communication. The date upon which has such notice or other communication is delivered shall be derived to be the date the same is received.
- 4.2 The terms and provision and Article III at this America and shall constitute coverants running with the lands and leaves affected beauty and shall in binding upon and shall inure to the benefit of the parties beauty their respective are covers and assemb.

IN WILKISS WHEREOF, this Restated Production Payment Production Payment Agreement dated as of March 1, 1974 and 1 parts as of May 1, 1976.

amends and restates the ecuted in multiple counter-

Bycsin C. Seittle Edonaly Wappoler

ATTEST:

Sall, Kaya-

[CORPORATE SEAL]

Agrica to Settle Exemple Chappler

ATTEST:

Sall, Hoje -

[CORPORATE SEAL]

THE STATE OF TEXAS COUNTY OF HARRING \$ 881

Br. It Restembero that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aloresaid, hereby certify that, on this 2.25 day of May, 1976:

(Louisiana)

tion, and that the scal added to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that the black of the acknowledged the instrument to be the free act and deed of the cor;

(Lou. na)

Before me appeared W.E.P. P. Serry, to me personally known, who, being by me duly sworn, did say that he is Vice President of Lunnzon Commany, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of raid corporation and that the instrument was signed and scaled in what of the euporation by authority of its Board of Directors and that acknowledged the instrument to be the free act and deed of the corporation. Will William some

(Texas)

Before me on this day personally appeared W. H. Ecwalet, Ite, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of PENNZOIL LOUISIANA AND TEXAS OLISHOM, INC., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared hit williams, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of PENNZOIL Company, a Delaware corporation, and relative hed ad to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

In Witten Wilmanor, 11 a become set re- hand and official notarial scal, in the City of Houston, County of Plant's State of Tex that the Care on May, 1976.

Relicion hays Hoster Notary Public in and for Harris County, Texas

My Commission Expires June 1, 1977

APPENDIX 1

TO

PRODUCTION PAYMENT AGREEMENT BETWEEN

PENNZOH, LOUISIANA AND TEXAS OFFSHORE, INC. AND

PENNZOIL COMPANY

CONVEYANCE OF PRODUCTION PAYMENT

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Conveyance") dated as if March 1 1974, as at a had neglected as of March 1 1976, between Penazon Louissana and Divisional Research Penazon (the "Graphor"), having its principal office and port office address at Penazon Phase, 120 Bus 2967, Heaving Texas, and Penazon Contexas, a Delivery a penaton of the "Grantic" a having its principal office and post office address at Penazon Place, 120 Bus 2967, discission Texas.

WITNESSEL

Williams the Granton doings for all analytic Grantee in sign to procluse the Production Payment hereinafter described,

Now, Turns out it is seried by smalls tween the contor and the contor as follows:

- Certain Definitions. The following terms of all a sed in this Conveyance with the meanings given below:
 - (a) "As spire. At the double possible possible possible and in lading the 21st stay of an include ment and anding on and include the 20th day of the next succeeding call it is small.
 - (b) "Etherica Date" shall mean 7 o'clock, A.M., our the date of the Convernee, according to the transition in effect at the heating of each Subject Interact.
 - (c) "Exhibit A" shall mean Exhibit A attached to the Conveyance and hereby made a part here d for all purposes.
 - (d) "Hy have a brown" of all true in all off, gas and other hydrograph insite any combination of one or more of such and frame appropriate to the countest to which such learn is used.
 - (e) "Lesse" de 2 et al., godt and gas lease of sub-lease, described in Exhibit A.
 - (f) "For each Court may must Date" shall mean 7 old As, A.M., ascording to the time then made of add a superal real stational transfer and the distance of which they have price is paid by the Courts are the Court as a provided in the Production Production American dated as of March 1 1 to be a model and in the Law of May 1 1 Point 1 forces the Courter and the Grantes.
 - (g) "Subject 14, " mad se" shift is an all Hydrocarbone in and under, and which may be produced and a statement and which belief accurated by attendant blatter, the Subject Interests from and after the consent C announce at Date and theory best the period specified in this Conveyance to, the terms of the constant of the constant of the constant.
 - (h) Subject totals of the translation of the method of the contents of the Granton in the Leases, and in addition thereby. As and every other band and character of right, title, claim or interest which the Grant is now to income the Library which interests or Leases are either appropriate ally or generally all labed in Lybrit A, all as the same shall be enlarged by the

discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and all renewals and extensions of any of the same, but there shall not be included within the Subject Interests any rights which the Grantor now has or may be realter obtain with a great to production attributable to the interests of nonconsenting parties under any operating agreement, unit operating agreement, contract for development or similar instrument.

2. Conveyance of the Production Payment. The Capator, for valuable consideration, the receipt, added by and sufficiency of a libbrate heads a flowed died by these presents does hereby mant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, as a production payment, an undivided eighty-lave procent (SET) interest in the proceeds from the sale of the Subject Hydrocarbers, so 'et, however, to the provisions of Far, graph 3 of this Conveyance.

The afore siderighty five percent (2.5%) interest in the proceeds from the sale of the Subject Bydrocarbere, as a santal, bargained, sold, conveyed, assigned, transferred and delivered unto the Grantee by the term of this Conveyance, shall be the "Production Payment."

To II we ware allocothe Production Payment, together with all and singular the rights, privileges and appurtenances thereinto in any way belonging, unto the Grant or, its successors and assigns, forever.

- 3. Certain Process Governing Subject Hydrocarbons. All the provisions of this Conveyance shall be subject to to following
 - (a) For the discharge of the Production Payment the Grantee shall look exclusively to the proceeds of Sunjan. Hydrocardons, and the Grantor shall not be hable for such discharge.
 - (b) The Production Payment shall not be discharged. from the proceeds of any products resulting from any material inding, proceeding or refining operation except to the extent of that portion of only proceeds which represents it. fair market value at the wellhead of the Subject Hydrocarbons used in such manufacturing, processing or refining operation.
 - (c) The Freduction Payment shall not be discharged be out of any bonus which the Grantor shall receive for any Leave or assignment of any of the Subject Interests or out of any payments made to the Grantor in connection with the drifting or deferring of drifting of any well on any of the Subject Interests or in connection with any adjustment of any well and leasehold equipment upon unitiration of any of the Subject Interests.
 - (d) There shall not be include I in the Subject Hydrocarbons any Hydrocarbons unavoidably lost in the prediction theoret or produced and saved from any of the Subject Interests and used by the Granton in condomnity with good field penaltion for during and production operations (including gas injection, recording record, present to subject recording or cycling operations conducted for the purpose of producing Subject Interests are consisted, but only so long as such Hydrocarb are remard.
 - (c) So long as and to the cets it that the same any be required by appleable laws and regulations, in the cere of may be a form the United States of America included in the Subject between the tree when himself, a population of oil per well per day averaged on the monthly be cere in the cets of the other states per and the night of the Granter to receive the precedent oil per day exceeds with the meaning and main and average production of oil per well per day exceeds with measures mount, and such suspension shall apply separately to any cone or participated so takes a vegetable to computing procument royalty.
- 4. Amount and Term of the Production Payment. The Froduction Payment shall one and renomin to full to court discount until so to take use the Greater shall have receive and a little to the court until so to take use the Greater shall have receive and a little to the court until so to take use the Greater shall have receive and a little to the court until so to take use the Greater shall have received and a little to the court until so to take use the Greater shall have received and the court until so to take use the court until so take use the co

out of the properly from the sale of the Subject Hydrocarbons, free and clear of all development, operating, mining produced to along processing, building, storing, resketing, trasporting and other costs and expenses of every kind what moves the full aggregate sum of the following amounts:

- (a) The apprecial pencipal ances to beans, plus accorded interest and commitment fee, outstanding on the Proceeds Come accorded Pate used to the East. Credit Agreement, dated as of March 1 1975, withoutstanding as a bed code is stated as of March 1 1976; among the Grantor, The Pust National Book of Change as 1 the other books manned then in thereinafter called the "Primary Sum", plus
- (b) An anomal conjusted at a rate per annum (on the baris of a 565-day year) which shall be expected to H = of the corporate been to as the per to its largest and most credit wouthers come all barrows is in a fact of Ha I in a National Bank of Classica from time to time phase by a such rate to change a domait such a transition of the action of the date of each one in such corporate less with forementally and from those the Press ets Commencement Date on the rate of a disputational plane from the most interest the Press ets Commencement Date on the 21st 4 of the right has a fill being a fall brain to be made on the 21st 4 of the right has a transition to be made in bading the Press of Commencement Date for the period from and in bading the Press of Commencement Date to and including the 20th day of the most has a following the Press of Commencement Date to and including the 20th day of the most has a constitution of the most increase of the pressure of the pressure of the most monthly on the first day of each Accounting Most before the pressuring Most of the amount of such undequalified but now on the first day of each Accounting Most before the pressuring Most of plus.
- (e) An arount equal to the agreement of all amounts which are paid by the Grantee to any state of political sub-loss as the correct of advanteem, teacher, mortgage, gross production, gross preceipts, factors, profiles, severages, occupation scales, use, francise and other taxes and assessments of any 150 by his according penaltics and interest, imposed or asserted with respect to or measured by or charm discussion attributable to the Production Payment, the Subject 11 droughter or the proceed discussion of the indicate it is amount about the interest on the unliquided amount discussion, it for each of 7% per annum from the date of payment of each such as the first Canton, provided, boxester, that the Granton shall have the right at its counter to comest any such taxes or asserspents in good faith on behalf of the Grantee.
- It Brise the Indianos of the Generol and the Granter in Element into This Conveyance that the Grantee shell be entitled to receive and redice out of the Subject Hydrocubous the full aggregate many of the about a described in ordinary applies (a) and (b) of this Panagraph 4, free and clear of all cost and copy accounts to the following part between and the Subject Hydrocubous (which expenses and notes." It is bound by the Granten) and over and above all taxes, expenses and costs of the changles 6 of Subjects of an artists. It is in subject, as he (c) of this Panagraph 4.
- 5. Termination of the Production Payment where the full agency of sum of the amounts specified an original plane (1), i.e. (1) and its of Lagragah 4 of this Conveyance (as increased as expressly passed that the result is Conveyance at a bound and by the Greeter, the Production Payment shadily to passed in an analysis of the same expression of the immediately contributed was done to the same expression of the immediately contributed was done to the same expression of the immediately contributed was done to the account of the conveyance had not become as a linear expression of the upper such termination at shall execute or cause to be executed agree the contributed at the conveyance of the Grandon, such instruments as may be necessary or agreement to explore the Lagragae of the Production (agree).
- 6. Application of the Proceeds of Subject Discheralisms. For all purposes of this Conveyance, the process I and Subject 111. It is a second for the Conveyance of the Grantes point to the close of business on 20th clay at the small result is a 1-th sum. The is a self-transmentement Date shall be deemed to have been received it and applications of delay at a flavor process of business continued to day of such mouth, and therefore as her sees a network over the delay of such mouth, and therefore as her sees a network over the delay of such as a flavor on the first day of such Account-

is a Month shall be deemed to have been received and applied immediately after the signing of looks a on the first day of the next succeeding Accounting Month; provided, however, that, if any date of application specified above shall be a Saturday, Sunday or legal holiday under the law of the resolution in which such proceeds are actually received by the Grantee, such proceeds shall be deemed to have been reserved and applied on the last business cay next preceding such Saturday. Sunday on the all holiday in such just diction, but the amount of such proceeds to be applied present to such appear (a) of this Tasagraph 6 shall revertheless be the account under subjurgaph (b) of Pasagraph fool this Convey one account up to (but not including) such Saturday, Sunday on legal heliday. Such proceeds shall be deemed to have been applied on the first day of each Accounting Month as follows:

- (a) First, to the amount referred to in substance the property of this Conveyance accound up to (but not included) the date of such property or, and
- (b) Second, to the amount referred to in subparagraph (c) of said Paragraph 4, to the extert them assertanced and
 - (c) Third, to the reduction of the unliquidated balance of the Primary Sum;

provided, however, that, in the event such presents so applied on the first day of any Accounting Month shall be in afficient to exact the full amount specified in subparagraph (a) of this Paragraph 6, the unliquidated balance of the Primary Sum shall forther these increased (to the extent permitted by law) by an amount equal to the amount of such deficiency.

Within 10 days after the date of each such application by the Grantee of any proceeds of Subject Hydrocarbons, the Grantee will turnish or cause to be furnished to the Grantee a written statement showing the application of such proceeds in accordance with the provisions of this Pangraph 6.

7. Marketing the califest 11 decomposes. The Granton at its expense, shall market, or cause to be marketed, the Subject Hydromathons for the best price resonably available when marketed, and it shall not be necessary for the Granton to join many Sales Centracts (as larginafter defaced) or any amendment thereof. In the except the Granton shall take any of the Subject Hydrocarbons for its own use, the Granton shall pay to the Cenaton, on or belong the last day of the Accounting Morah during which such taking oroms, a price a rual to the fair market value at the wellhead of the Subject Hydrocarbons sort dam.

So Sales Centracts. The Crantor will delt perform all obligation, building on it under all contracts and agreements are the sale of Subject Harris or larger any postern thread, whether presently existing or here. It is not excell into gall such contracts and agreements being the "Soles Contracts"), meanworldow with the norms har of and will take all appropriate mounts to enforce the performance under each of the Soles of the obligations of the purchase thereinder. All Subject Hydrogarbors sold by the Congress, who then present to the topos of the Sales Contracts of otherwise, shall be delivered by the Granton in the purchase therein into the pipelines to which the will production such Soles. Hydrogarbors may be consisted or to such other point of purchase as is tractionally required in the total cline of such buildy of Hydrogarbors.

9. Withhold be and Restitution of Proceed of Subject II data chans. All objections of the Granton handed by the subject to the spylatic be provinced of the Natural Cos Act and all applicable tubes and regarding set the Festival Power Commission. Rates a quantited under the Natural Cas Act and said who said with the Festival Power Commission. Rates a quantited under the Natural Cas Act and said who said with the problem of the said with the contribution of the said of the said of the contribution of the region in making filings with the Festival Power Commission. Subject Hydrosubsection per backback and say would be rate to seen, including without limitation, withholding by the Grantor per backback and say would be rate to seen, including with the Festival Power Commission.

', to the the Crantee shall not be deemed to have received or realized such proceeded before extent that, such proceeds shall actually have been received by the Garater of or after receipt of the full aggregate sum specified in subjectar aging (i) (i) (i) and is : aramaph 4 of this Conveyance, the Grantee Stall Learning Ped, it any reserve CHY DOV-2 the Crantez on ment or restitution or such payment or restitution stall is a section account of my proceeds of Subject Hydrocarbons, where we say the exa mand me must to the provenue of Engangh tool this Conveyance, the will be a distallation of the Primary Sum shall forthwith be increased by the around of the policids at point of the recount of the Grantes plus the amount of any payments node by or many a minst the Grantee in the nature the in or each payment or of interest damages and penalties, provided, however, sarestitution the cuture Principly Sum shall have been liquidated. Criminy Sum al. if he deemed not to have been a lipped and and the unbiquidated I once of the Primary Sam, shall be thereupon deemed to be in the assume of the proceeds so poid or a plus the amount of any payment a made by or napseed against the Collection the nature of interest, damages and penalties

- 10. Protection to Purcha cas of Production. No person purchasing or taking or processing any Subject Hydrogadeers, shall be required to take notice of, or keep informed as to, the discharge or termination of the Production Payment until the actual receipt uch person of written notice advicing such person of each discharge or termination.
- II. Payment of Taxes. The G. eter will torder and pay. come to be paid, punctually, before they shall become defin junct (et as to bey thereof whicher a being contested in good faith, proughly after the final determination of such confect), all ad valuern taxes (or taxes imposed in ficultin colly and all transfer, months are given posturizer, mining, severance, occupation, excise, sales, use, franchise, income and other tax sound a research of every kind and character whatso ver (other than federal income text of the Galacter) imposed to a several with respect to or measured by or character in attractable in the Subject Interests, the Production Payment or the Subject Hydrocarbons or the process de the reaf, or as and the Grantee by reason of its acquisition or ownership of the Production Payment by reason of his or its interest as assistance to either with any assistance of the Production Payment by reason of his or its interest as assistance to either with any instance of the Production Payment by reason of his
- 12. Coverants of the Counter. Until sects that as the Production Paymer t has terminated as provided in Paramarks 5 of this Counce, the Grantor will, at its expense and regardless of who may be the equation or operators of the Subject but out, cause:
 - (a) The Sody of this peaks to be maintained and developed and continuously operated for the peak, which is it highest as a series and seather, b't manner and in accordance with sound fieldy. The condition of the peaks and seather, b't manner and in accordance with sound fieldy. The conditions of the leaves and heaves and heaves and seather and in accordance with and in continuously with sound is by a mass about and to participate in may operation which is to be conditionally with sound in a peaks the conditional unit of all peaks are an expensive and conditional which is to be conditionally as an all one peaks the conditional unit of a superficient field of the conditional and the peaks and the peaks and the peaks are also as a superficient factor of the field of the conditional and the peaks are the conditional and the peaks are the conditional and the
 - (b) We as be now to a scale of the soft and bid or nature including without factors of a 1 to the soft and the soft and the proposed in or arising from the polarisation of a soft and the soft and the processing treating processing

storage maketing or transporting of the Subject II chocarbons, to be paid punctually when due, or, as to any thereof which are being contact in good feith, promptly after the final determinations is such context.

- (c) All machinery, represent and Lade a of any 1 ad now or hereafter located on the Subject Indicates and more many meter full and temperate and more production of Hydrocanbans therefore to be a relief and to be kept a result, all the time operating condition and all repairs, removed, repeats and here a radio provides a threat or thereof or therefore needed for such property also doly in a case parallel.
- (d) All reservoir and proper it prote by the mobile entity to proceed and defend the Subject Interests the Subject II. Proposition and the proceeds the code ground any adverse claim of demand, including actionary functions, the emphasization are of council for the prosecution or defense of higheston well the contest, settless at, release or decharge of any such claim or depend.
- (c) Written notice to be given to the Grantes of every adverse claim or demand made by any person (in but, arony conserved to give more that agency) affecting the Subject Interests, the Subject Hydrocalor match proceeding instituted with a part thereto.
- (f) The intensity of the Granton in each of the Subject Index als to be kept free and clear of line, charge and clear of the control of every the analytic than these which are of a charge estimately to sell in compactable thilling and providing operations and which do not restrictly adversely affect the my stanted the Subject Intensity and other than those becoming effect or analytic terms and of the Landon Conveyor.
- (g) Subject to the pressions of Farmer's Limit the Conveyance, all leaves included in or relating to the feature back on L. Farmel to in Palithic A and under which the Gundou is Leaves on lesses, or resolver to be monthing by full force and effect without amendment, machinest or or waver of may provide not of any such Leaves which might materially adversely affect the right-scale the Convey which the Convey tree.

Anything a place A in the Phys. (1) 12 to the clother cate an potential ading, it is Creator shall not be obtained to perfect a confertable and a sum A. (2) by the sum at a and which are beyond the control of the Camera, a offer or good to a finished budged Entered and which it is not the operator. In such case, hear we goe a reason as a finished budged to finish to see a certification of any such undertakents.

- 13. As a set the Subsect of a set. The United A. It permit any and in more representatives of shorts for a Court state of a set of the a first of the Subsect of the Grant state of the
- If for a small Court was a New to the real time of many and its any Sulfield limits of a superior to the Court of the court of the superior to the court of the c

shall be subject to the Production Payment in the same manner and with the same effect as though such unit were in existence on the Effective Date. The signature or joinder of the Grantee shall not be necessary to the instruments exploring the pooling and unitization herein authorized.

- 15. Default by the Grantor and the Remedies of the Grantee. Should the Grantor default in the performance or observance of any covernant or agreement provided in this Conveyance to be performed or observed by the Grantor, and should such default continue unremedied for 30 days (or for such longer pariod as may be necessary to complete performance diligently commenced within 30 days) from the delivery to the Grantor of written notice of such default, the Grantee may, in addition to pursuase any or all other remedies available at law or in equity:
 - (a) Effect performance or observance of such covenant or agreement on behalf of and at the expense of the Grantor, in which event the Grantee may expend funds reasonably necessary to effect such purposes and shall be entitled to reimbursement therefor from the Grantor, together with interest on such anomals at the rate of 7% per annum from the date of such expenditure to the date of reimburs, next.
 - (b) Upon written notice to the Granter, succeed to all rights of the Granter with respect to the possession, operation and development of the Subject Interests, and may use in connection therewith all of the appartenant property, both real and personal, situated upon or used or held for future use in connection their with, or in connection with the production, storing transporting or sale of the Subject Hydrocarbons, and the Grantee shall have the right, on behalf of and for the account of the Granter, to sell the Subject Hydrocarbons and the Granter shall, upon demand, reindures the Grantee for all amounts so expended by the Grantee, together with interest on such accounts at the rate of 7% per annum from the date of such expenditure to the date of reinhausement.
 - (c) Apply to a court of equity for the specific performance or observance of such covenant or in aid of the extension of any power herein granted and for the appointment of a receiver of the Subject Interests and the Subject Heisenberg.
- 16. Exercise and Tennination of the Repaches of the Grantee. All rights and remarks granted by the provisions of Paragraph 15 of the Conveyance and otherwise belonging to the Grantee shall be cumulative and no creatise of any right or remark shall preclude the exercise of any other right or teniedy. Any right or remarks on one or more occasions without exhaustion thereof. All rights and remarks the interaction.
 - (a) When the I section Payment is discharged and all amounts then due and payable to the Grantee parts. To the term of Payment 15 of this Conveyance (including amounts payable for interact) of the Payment full payment for the Conveyance (including amounts payable for interact) of the Payment full paymen
 - (b) At any earlier time who will definite of the Granton dealthave been remedied and all such amounts shall have been remedied and all such amounts shall have been club, p. directall, without projudies however, to the exercise of any tiplus and made a condition of the Grant of the performed or perform or observe any of the sevenants or agreements bearing provided to be performed or observed by the Grant of a
- 17. Reports to the theretee. While the Leader-teer Payment remains in force and effect, the Grantor will, at its expose, finally, to the Grantor within 45 days after the end of each calendar month, an particle of a conficulty to the sale for the quantity of Subject Hydrocarbons produced and sold or taken by the Granton for the end of the gross proceeds received from the sale, or attributable from hit days of such pulses (11) for given
- 18. Warranty of Title. The Grandon conceants and wa rants that at the time of the execution and delivery of this Commission. the Grandon has good and marketable title to the Sul-ject Interests,

subject only to such exceptions, charges, lieux or encumbrances, if any, as are set forth or referred to in Exhibit A, that the Grantee has the power and right to create and confer upon the Grantee the rights of the Grantee granted by the terms of this Conveyance; and that there are no suits or proceedings pending or, to the knowledge of the Granter threatened, against or affecting the Subject Interests before any count or by or before any governmental commission, bureau or any regulatory authority a high if decided adversely to the interests of the Granter shall materially affect the rights of the Grantee pursuant to this Conveyance. The Granter hereby birds itself, its successors and assigns, against the lawful claime and demands of every person who shall claim the same or any part thereof, and the Grantee made incoming and hold the Grantee harmless against and from all loss suffered by the Grantee made incoming and hold the Grantee harmless against and from all loss suffered by the Grantee made incoming thom any tailout of or delect in the Granter's title to the Subject Interests and the products and thom, about pursue the discharge of the Production Payment.

19. Transfer of the Subject Interests, Except for the conveyance of production payments becoming effective only . Iter termination of the Preduction Payment as provided in Paragraph 5 of this Conveyance, prior to such a maintain the Granton shall not, without the written consent of the Grantee, make or permit any source nebra, of and nament, sale, conveyance, assignment, lease or sublease, in whole or in part, of any of the Subject Interests, provided, however, that the Grantor shall have the right to smeender, absorber or otherwise Lemmate (either voluntarily or involuntarily) any of the Subject later is (in any portion time of) who a there is no well located three on which is capable of producing lively-carbons in paying quantities and when the Grantor has reasonably determined that (i) no existing well may be recompleted the even as a well capable of producing Hydro arbons in paying quantitie, (ii) no ed-himen well well weld be drilled and completed thereon which capable of producing IIA2 - there one and above the cost of drilling and completing same, in paying quantities, and that his pressure that termine or theoretay recovery operations may be Instituted with a specific may well the control be would cause such well to be capable of producing Hydrocarbons over and above the institution and completing such pressure maintenance or secondary recovery operation, in paying operation. For the purposes of this Paragraph 19, the woods "in paying quartities" as applied at any time to the producing capability of any well shall mean that, in the light of combining exist in at the time of thermination and which reasonably appear to be not temperary, such well is policing or is or will be capable of producing Subject Hydrocarbon, who e are a some a boreast been will and I the Granton's chare (as the owner of the Subject indicast on which we have have have the of the dead too followering such well.

In cas', the where the procedure personnels of the Real sphr 19 yumits the Grantor to surrender, absorber of the safe, a first series are personnels and the Sulpert Index to (or any partion threeff), then (a) and of the safe, a first series are personnels and the safe are constructed, abundanced or termined of shelling a series and a safe are at order a surface tradecast (or any portion then first shelling are a first series and the safe are safe and the safe are all the safe are at a safe and the safe are at a safe are at a safe and the safe are at a safe are at a safe and the safe are at a safe are at a safe and the safe are at a safe are at a safe are at a safe at a safe are at a safe at a safe are at a safe at a safe are at a saf

20. Useration to Payment: The same as let the Count of except and being der shall not provided the Grands have a 1 and 1 so a second the regards of the recount thereof, provided. Lower set, that its authors place must be a result by the payment in in a set or given to the Grandon scaled 4.2 a sub-of a payment a payment a made.

- 21. Transfer of the Production Payment. Nothing contained in this Conve ance shall be construed to limit or restrict in any way the right of the Grantee to sell, convey, assign or mortgage the Production Payment, in whole or in part.
- 22. Rights of Mortgages or Trustee. If the Grantee shall at any time execute a managage or deed of trust covering all or part of the Production Payment, the mortgages or trustee therein named or the holder of any oldication second thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to carryise all of the rights, remadics, powers and privileges conferred upon the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained hereunder by the Grantor, but the provisions of this Paragraph 22 shall in no way be deemed or construed to impose upon the Granter any oblig, tion or liability undertaken by the Grantee under such mortgage or deed of trust or under the obligation secured thereby.
- 23 Notices. Any notice, request, demand, report or other instrument which may be required or permitted to be given or furnish d to or served upon either party hereto or other person succeeding to any interest of other party hereto shall be deemed sufficiently given or furnished and served if in writing and delivered to such party, or person or to an officer of such party or person or deposited in the United States mail in a scaled envelope, registered, with portage prepaid, addressed if to the Granter to Pennzoil Place, P. O. Box 2967, Houston, Texas, and if to the Grantee to Pennzoil Place, P. O. Box 2967, Houston, Texas, or to such other address as the party or person to be addressed shall have designated by written needs to the party or person giving such notice or furnishing such report or making such request or so and
- 24. Further Assurances. So long as permitted by applicable law so to do, the Grantor will execute and deliver all such other and additional instruments, notices, releases, acquittances and other documents, and will do all such other acts and things, as may be necessary more fully to assure to the Grantee all the rights and interests herein and hereby presented or intended so to be.
- 25. Binding Effect. All the exceptoral and agreements of the Granter contained in this Conveyance shall be deemed to be covenants maining with the land and shall be binding upon all successors in interest to, and all assigns of, the Granter and shall innie to the benefit of all successors in interest to, and all assigns of, the Grantee. The provisions of this Conveyance shall innie to the benefit of and be binding upon the respective source-ors and assigns of the Granter and the Grantee from and after accept one by the Grantee. All references herein to either the Granter or the Grantee shall include their respective successors and assigns.
- 26. Release or Failure of Title. No indexe by the Grantes of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect of a long the amend of the more cells from the sale of production of Hydrocarbons out of which the Production Payment or creation any offset or offset projudes to the Production Payment or creation any offset or offset projudes to the Production Payment, and the Production Payment shall entire in full these and sheet a to the remaind a of the Subject Interests until the full amound of the Production Payment shall entire in Payment by the Grantes as herein provided.
- 27. Final Templastica. Note if istanche : we other provisions here for the Production Payment shall, in any execution as a 1 day point to the expectation of A years after the death of the survivor of all the descentises of Therefore is a well, I is Tresident of the United States, who are living on the date bearing.
- 28. Deadings Paragraph headings used in this Conveyance are for convenience only and shall not affect the construction of this Conveyance

IN WILLIES Will note, this Rest ded Instrument of Councyance of Production Payment amends and restates the Instrument of Councyano of Production Payment dated as of March 1, 1874, and the parties have considering the taked Instrument of Councyano of Production Payment to be duly executed in the presence of the reader igned witnesses on the dates of their respective acknowledge.

ments annexed hereto, in several counterpaces, each of which is an original and all of which are identical. Each of such counterpacts small for all purposes be deemed to be an original, and all such counterpacts together shall constitute but one and the same Instrument. The date of this Instrument shall be deemed for all purposes to be the date first above written.

| Witnesses: | | | PENNZOH LOUISIANA AND TENAS OFFSHORE, INC. | | |
|------------|--|--|--|-----------------|-------------------|
| 1 2 | | | □ ② | Бу | |
| | | | | ATTEST: | |
| | | | | | SS2 NR SR |
| | | | | | [Cor-orate Seal.] |
| Witnesses. | | | | | |
| | | | | Pennzon Comeany | |
| | | | | Ву | |
| | | | | ATTEST: | |
| | | | | | 0.505 |
| | | | | | [Component Seal] |

TID: STATE OF TEXAS COUNTY OF HABBIS SS:

Bi. In Restr. and that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the C. and State aforesaid, hereby certify that, on this day of May, 1976:

(Louisiana)

Before me appeared to me person lly Luowo, who, being by me duly swom, did say that he is Vie - President of Franzon Lorisiana and Trans Orisione, Isc., a Delaware corporation, and that the scal allised to the foregoing instrument is the corporate scal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared to me personally known, who, being by me duly sworn, did say that he is Vice President of Preszon. Cospens, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and scaled in behalf of the corporation by authority of its Board of Directors and that acknowledged the instrument to be the free set and deed of the corporation.

(Texas)

Before no on this day personally appeared — , known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of PERSZON LOUSIANS AND TEXAS OFFSHOM, INC., a Delaware corporation, and acknowledged to me that he executed and in trumout for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared ... known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Prinzon. Convent, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITERS WITERS, I have hereunto set my hand and official notarial seal, in the City of Houston, County of II rris, State of Texas, this day of May, 1976.

Notary Public in and for Harris County, Texas My Cour is sion Expires

EXHIBIT A

TO

CONVEYANCE OF PRODUCTION PAYMENT

| | (.(), 11.1.1 | NOT OF THODOGOTON LAISING | 1000 USD - 4001 |
|------------------------------------|-------------------|--|--|
| BLM or State Leave No. | Effective Dete | Description | Crantor's Undivided Percentage Interest |
| P | | edetal icases are as shown on Official | 53.12.57.30 |
| | | ter Continental Sheli Leasing Maps) | |
| OCS G2839 | 12-1-74 | All of Elvel, 352, West Colleges Steal | |
| (A) (1// | 12.1.1.1 | West Addition, Of 'S Other d Leaving Map. | |
| | | Long tima Mage No. 1A | 25.00% |
| OCS (2:22) | 2-1-73 | Block 512 West Carreton Area, South Addition, | |
| AND THE WATER AND EAST OF THE SAME | | Official Leavis, May Louisiana Map No. 1B | 26.67% |
| OCS GEET | 2.1.73 | Die Frankest Carre een Xina, Smith Addition, | |
| (20): (1020) | 6.1 | Official Leaver: Map Locasian, Map No. 1B | |
| OCS-G3281 | 8-1 75 | All of Block 59 is West Conneron As a South | |
| | | Addition as shown on OCS Official Leasing Map Unorsing Map No. 1B | 25.00% |
| OCS COSS | 1 1-73 | Rivel 1999 West Cas a ren Area, South Addition, | |
| | A 160 1050 | Official Learning Map. Louisians Map. No. 118 | 23.20% |
| OCS-CC VA | 12-174 | All of the A. Cert Ver et Common Area. | 98-88-87-2004 To |
| | | South Addition Of S. Official Learning Map, | |
| | | Lactional a Map 5 x 4 B | 6.6666777 |
| OCS-G. 559 | 5-1-71 | All 1 1 517 Med a aperon Area | |
| | | Son Addition to be all Leasher Map | 02111 |
| OCS-02559 | 12.1.71 | Louisant Mayer 5 | 63(1) |
| (11,11,11,11,11) | 1 . 1 . 1 | All of Block 2 to 1 of Common Area, South Add trees of S.O. of Leading Map, | |
| | | Lorronna M. p. No. 2 V | 16 6066711 |
| OCS-02569 | 12-1-74 | All of Block "AT I at Conserve Areas | |
| | | South Additions to saying all Language Map. | |
| | | Low Signer M. p. Xao. 2 V | 16.6680.5 |
| OCS C2 32 | 12 1-71 | All of block Info Small March Plant Are a | |
| | | South Addition Of Settle South Map, | 0.666677 |
| OCS G.S. 3 | 12-1 74 | All of I had EN So of Mr. All North | O. Original . |
| 570-1.808915538-05- 05 - 40 | 3530 F 14 55 | South A history Co. S. Committee as in Map. | |
| | | I = 6 = 40.2 M qu Na 16 | 6 (477 77) |
| OCS GEG | 5171 | All of the Late See as the Late of Mr. | |
| | | Soft filler of the officer. Man | 2000 0 |
| OCS-CPS1 | 1.5 1 5 5 | Les con M | 0-2. |
| (A.5-(, .5) | 1. 1 | AE of birth 1. 1. 28 with M white I Wood, South School School and was Map. | |
| | | Let we take the M | G. Charter |
| OCS-CESSS | 12 1 74 | Alter Devil 111 Secret, March 111 March | 10000 001 1010 |
| | | Smile wild as the street little in Map, | |
| 93 PW - 98 | | 1.6 3.6 31 3 3 4 0 | 6.666674 |
| OCS CERT | 12 ! 74 | | Table 1912 Commence |
| OCS CHA. | 7.1 % | ABer David Comment of the Comment of | 16.66667* |
| | 4 1 4 4 | and the state of t | |
| | | Large and A. C. and I | Iti taddi. 7. |
| ()() = () + () = | 5-1-74 | All products the second of the second of the | |
| | | Addition, top of the control of | |
| anno no anno nome | SE 17 M/2 | Map X of Y | 142 C |
| OC 25 (17, 817) | 2 1-73 | Photo House Handle to the Not town | .66. |
| | | G 55.0 I may Map I at such M p S or 4A 11 | 2014.1V3 |
| | | 1.8 | |

EXHIBIT A

TO

CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| CON | (1,1,1,1,1) L. OI | TROMPORTAL (Communica) | |
|--------------------------|--------------------|--|--|
| BLM of State Lease No | La'es tive Date | Description | Grantor's Undivided Fercentage Interest |
| OCS GEITT | 11 1-72 | All of Block 1948 off, Phys. Area as shown on official brasiling trap. Louishing M. p. No. 9 | 10.00% |
| State Lease No. 6310 | 5 111 | Tract 1301S. Lating those particus of Blocks 57 and 58 South Plass Area. Physicannes Parish, Longuez behavior to the State of Louisvan; and not made a control large on Match 4, 1974 and has to be purificularly described in Exhibit A. Association to add Jan C.O.B. Notation, Physical Review of Physicaniaes Parish Louisvania. | 20,00% |
| OCS G2185 | In 1 . 2 | Block 7%, that portion which is more than 3 geographical radius scaward from the line distributed in parastraph Lof the Supplemental Discretion tills U.S. Supplement Court intered Discretion 13 PC 7 at 15 U.S. v. Louisina No. 9 Original visit U.S. v. S. S. S. th. Pass Aten, South and East Addition, as shown on official because map, Louisiana Majy No. 9A | 15.00% |
| OC\$-0.31-13 | 7-1-75. | SU2 of Phys 1 (8) Mant. Pass Area, the position located by Z or Pus that zone was defined in the areas of the control of the 2 (8) desired the State of Landau and O tobas 12 (16) and landward of the Third Supplemental Decrea- Ling (40) U.S. avec (16) and a 20 (1971) A OCS Official Learning Mep. Louis at Map | |
| OCS Care | 7 1 77 | No. 10 Die 4, 72 Mais, P. S. Area, that person located more than a selection award of the Third Supplyment of Device 100 (49) U.S. 388 (December 2) Programme Vis. 5 Office J.Lensing | 25.00% |
| OCS COLT | 12 1 74 | Map, Lorentzia Map, No. 10. All of Park L. Mar, Pro-Asia, Of S. Official Lorentzia Mag, Lowesto Map, No. 10. That portion Land Lorentzia on food servend of the Thord Sept to state in over those (404) L. S. S. (112) and C. O. Larentzia. | 21.00% |
| OCS C. 10 | 10.17. | What do know Mark Pas Are as driver or offs. The rate of the North Passage of the control Map No. 10 | ereniu u |
| Cic S (.2) | In 1.7.3 | $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | d |
| Other Comme | 11 1 72 | At a time the state of the stat | 7.50 |
| O6 5 () 2 · 2 | 7.1 7.5 | on OCS Co. ** 1 C. are Major XII Fe to | 21.(%)*; |
| OC'S * | 7.175 | Mi of Glock Section 122 New Original and Jouxne on Ot S Charlet Leaster, "Lap NM Le12 | 21.00 |
| OCS C State | 7-1-75 | What files k No. 3 t 3 24 2 b. Seeds No. 2. The second of the second Map. | |
| OCS GOA | 4 1 7.3 | All 11 Section Advantage Anna Scotth Addition | 10.00% |
| | | tion to a little loop Map, Let a Map No. 6A | 02 (R) |

EXHIBIT A TO CONVIA ANCE: OF PRODUCTION PAYMENT (Continued)

| | COZZAT / ZCT, OE | PRODUCTION PAYMENT (Continued) | 5291 (J. 48) |
|---------------------------|------------------|--|------------------------|
| | | | Crantor's Undivided |
| BLM or State Tease No. | 1) 10 % | Descriptions | Percentage Interest |
| 008 (338 | 8 1-75 | Albert Phys. A 142 H. 5; Island Area, South Address Oner I basing Map, Texas Map No. 7B | 15.00 à |
| OCS (-2'Ya) | 8 1 7.1 | Ad of Elect A 174 Hi b Island As a South Addition Official Leave. Map. Texas Map. No. 76 | 12.00% |
| OCS-C2.372 | 8 1%,0 | All of block A-980 High Island Area, South Arthropic Offs ald resp. Map. Texas Neap N. 549 | 15.00% |
| OCS-G2373 | 8 1-7.4 | All of Block A fix High I land Area, South Addition, Officed Leasing Map, Texts Map | 12.00% |
| OCS-03115 | 4-1-75 | No 7B All et blad A-Per, Helle Pland Area, South Addition as slavar on OCS Oficial Leasing | |
| OCS G2377 | 8 1 73 | M. p. Tr. as M. p. No. 748 All of Phys. 2, 12, 12, 15 Island Area, South Addelman Chicago Le Sara, Map. Texas Map. | 16.40% |
| OCS C2778 | 8 1-73 | No. 7B A" of 6b d. 2 520, theh bland Area, South | 12.00° i |
| 068-02355 | 8-1 7-8 | Add in All and I having Map, Texas Map No. 76 All at blood A S.V. Buch Island Area, South | $12.00^{\circ}e^{-}$ |
| Octors the second | \$200.04 24 U. | Addition Office! I for the Map, Texas Map. | 12,000 |
| OCS-C2701 | 7-1-7-1 | All of 1963 As 20 Hubbashad Areas South Addition Of Sourceal Leasing Map, Texas Map 've 545 | 15.00% |
| OCS-G2704 | 7171 | All of decl. A life, H. k. Lland. An accessorable Addition, Carlo Official Leading Map. Trees. Moreover 40. | 10 00% |
| OCS (12779) | 10 1 ° | All of the " A " - He had telled Areas South Addragon to a second Lesson Maps Texas | 1400 |
| OCS C.2005 | 7.1 . 1 | Map No. 43: Ali : 1 v 2 v v ir ir v v ir v v ir v v v v v v v v v | TASS) |
| ous ceau | 7.1 % | No off Alice to "A fide it is Digit Associated Advisor for the property of the Texas | 14 (6) |
| ors Grant | 8 1 5 V | Ali, 5 2 18 7B Ali 8 22 | 14.00 |
| OUS CAPE | 7 4 11 11 | A to the second of the second | 14.50 |
| CALL AND EX | W 14 (III)6 | of the first of the second sec | 127150 |
| O(5-(.) 6 6 | 8171 | The Latin Annual South Made and The Latin Annual South Made and The Latin Annual South Made and The Latin Annual Made and The Research Made and The Latin Annual Made and The | 12 00% |
| mes or the | 8 1 5 3 | All of W = 1 at the left and Area See 1 at 1 at 1 at 1 at 2 are | |
| CH. S. F. T. | S I | All and South Administration | 12.0051 |
| | | и - М.р. и и м.р. и - М.р. з - Э | 20.(n) |

EXHIBIT A TO CONVEYANCE OF FRODUCTION PAYMENT (Continued) Grantor's

| BI M or State Leave No. | I ffective Date | Description | Grantor's Undivided Percentage Interest |
|----------------------------|--------------------|--|--|
| OCS G2718 | 7 1 74 | All of Block A-575 High Island Area - South Addition, O(5,O), and Leasing | |
| OCS-G2719 | 7-1-74 | Map Texas Map No. 716 All of 4.5 a k A 582 Birds 4.5 and Area — S. 24. Addition Of 5 Official Lausin | 20.00% |
| OCS-G2397 | 8-1-73 | N. p. Gesas Map. So. 745 Allor Blod J. A.77 Heel, P. Ind. Area. | 14.118% |
| | 2 2 8 8 | East Addition South Lybrosion, Official Leasing Map, Texas Map No. 7C | 15.33% |
| OCS-G2398 | 8-1-73 | Alboi Block A 27 i High Island Area. East Addition, South Extension, Official | 10.00.7 |
| OCS-C2399 | 8-1-73 | Leasing Map, Texas Map No. 7C All of Ele 1 A 274, Hegh Island Area, East Act to a South Extension Official | 12.00% |
| OCS G2403 | 8-1-73 | Leasin, Map. Tevas, Map. No. 7C. All of Block A 279 (1) 11 and Area. | 13.8461¢ i |
| | No. of Control | East Addition South Falsacian Official Leasing Map. Texas '1 p No. 7G | 17.00% |
| OCS G3314 | 4-1-76 | All of Block A 288, He heldend Area, East Addition, South Extension, as shown on OCS Off and Leasing Map, | |
| OC8-G24(//) | 8-1-73 | Texas Map No. 7C All of Block A-312, High Lot of Area, | 17.00% |
| 67.00.613.111 | 100 N 2001 | East Addation, South Extension, Official Leasing M. p. Texas Map No. 7C | 12.00% |
| OCS-G2411 | 8-1-73 | All of Black A-314, H., he Island Area East Addition, South Ustension Official Leasing May, Texas Map No. 7C | 0.) 957147 |
| OCS-G2737 | 7-1-71 | All of block A 316 Heal, Island Area = East Addition Scatta Extension. | 22.85714% |
| OCS-G2413 | 8-1-73 | OCS Official Learning Map. Texas Map No. 7C All of Block A-322, 10: helshard Area, | 14.00% |
| OC5-G2411 | 8-1-7-1 | East Addition. So the Latension. Official Ceasing Map. Fexas Map No. 7C Ali of Elock A 323. Health Land Area. | 14.00% |
| VV.1 (12.1) | 21-1-1-1 | Get Addition, Society, Johnston Official Lawring Way, Texts Indian | 14.50 |
| OCS G2416 | 8173 | All of Phys. LA 322, 110 half and Area, East Addition South Latension | |
| OCS G2118 | 5.1-7;0 | Official Haussey, Mayo Trans, Map No. 7C All of Block A 4.7, 11, 11, 11, 12, and Area, No. 1, 21, 12, and Area, | 20.8605 |
| OCS (1242) | 8173 | East Addition for C. List usion. Official Leastin; Map Jonas Map No. 7C. All of ithe UA 3x; B. A. Lind Area. | 10.00% |
| | | East Addition bounted a traveur. Official League, Mape Testy, Map No. 7C | 10.00% |
| OCB G27.39 | 7.1.74 | All of Res I. A. 2011. A shad Area — East Addition — S. 2011. Later arm. O(2) O(2) | D. CARNO |
| OCS G2426 | 8-1-73 | OCS Objected Jeneiner, Map, Texas, Map No. 7C. All of Blood Acido, The let Lland Area, East velofitons, sensibility tension. | 34,00% |
| | | Official Learning Map, Texas Map No. 7C | 34.00% |

EMHBIT A TO CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| Id Mor State Terse No. | 1.ttective Date | Description | Cranter's Undivided Percentage Interest |
|---------------------------|--------------------|--|--|
| OKS (22129) | 8-1-73 | All of Black A-351, High Island Area, East Addition, South Extension | |
| OCS G2715 | 7-1 71 | Official Leasing Map. Texas. Map No. 7C. All of Tobal. A 355. High-Jel and Area East Addition. South Extension. | 15.25% |
| OCS C= % | 7-1-74 | OCS Official Leasury Map, Texas Map No. 7C All of Block A 356-46 b Fland Area East Addition — South 3 stension, | 11.33% |
| OCS G2760 | 7-1-71 | OCS Official Teasine p. Texas Map No. 7C All of Block As Jos He and Area strension East Addition. Some strension | 15.00% |
| | | OCS Official Feasing, Map, Texas Map No. 7C | 10.00% |

This Conveyance is subject to the terms and provisions of the above described oil and gas leases and accordingly, the Subject Hydrocubous and the proceeds therefrom shell be calculated after deducting the proportionate share of the royalties reserved in each such lease attributable to Grantor's said undivided percentage concrest in such lease.

Lease No. OCS-G2839, OCS-G2725, OCS-G2830, OCS-G2759, OCS-G2607, OCS-G2317, OCS-G3193, OCS-G3195 and OCS-G2947 referred to also clare each subject to an Operating Agreement between Medal Oil Corporation, as operator and the Granton et also as occuparators.

Leave No. OCS C5504 referred to all we is subject to an Open best Agreement between Chevron Oil Company, as operator, and the Grantor, et al. as non-operator.

Lease Nos. OCS-G3284, OCS-G3288, OCS-G3288, OCS-G2889, GCS-G2889, OCS-G3888, OCS-G3881, GCS-G3267, OCS-G3284, OCS-G3284,

Lease Nos. OCS C2177, OCS C2191, DES C2193, OCS C2195. —LOCS C2206 referred to above are used subject to an Operating Associated between Call Col Corporation, as executed, and the Grandon et al., as non-noise stars.

Lease No. OCS-G235 OCS G 103 and OCS-G3314 referred to above or each subject to an Operating Agreement between Marghan G company as operator and the Graytor, et al. ox non-operators.

Large Nov. OCS C237. CCS C237. OCS C237. OCS C2311. OCS C2337. OCS C2313. OCS C2314. OCS C2314. OCS C2314. OCS C2316. OCS C2315. and OCS C. 136 rebared to the coarse caches best for an Operating Agreement between Cities. Service Oct. Company good rador, and the Careties of all is non-operators.

Lause No. OCS CCPS, OCS CC. — OCS CCAS and CCCCCCC contend to above an each subject to an Operating Agreement between Sun Oil Company (Delivere) as operator and the Granton et al. as non-operators.

Leave Nov. Of S.G. 2760 and Of S.G. 212 web real to all α are each self-self-self-self on Operating Agreement between Exam Corporation as operator is 4.0 by Graphic et al., as sept-specifies.

Lease No. OCS C2377 and OCS C. Sexus modern arose are unit subject to an Operating Agreement between Lease In Oil Developes at Lease Superators, and the Grantin, chall as non-operators.

The Grador's interests in Labor No. 100 to 0.2004, but 8-0.2005, but 8-0.2005, and 100's 0.2185 teleprod for all the present subject to Advance President Automorphis date lab of June 13, 1973, as amorphis to two or Grador and Units at Co. Type Line Company.

AMENDMENT

TO

INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT

THIS AMENDMENT TO INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Amendment") dated as of May 1, 1976, between Pennzoil Louisiana and Texas Offsitore, Inc., a Delaware corporation (the "Grantor"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas, and Pennzoil Company, a Delaware corporation (the "Grantee"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas,

WITNESSETH:

Whereas, by means of that certain Instrument of Conveyance of Production Payment dated as of March 1, 1974 between the Grantor and Grantee (such Instrument of Conveyance being hereinafter called the "Conveyance"), the Grantor conveyed to the Grantee, and the Grantee purchased from the Grantor, at the time the first loan was made to the Grantor under that certain Bank Credit Agreement dated as of March 1, 1974, as thereafter amended and restated, among the Grantor, The First National Bank of Chicago, and the other banks named therein (such Bank Credit Agreement, as amended and restated as of May 1, 1976, being hercinafter called the "Bank Credit Agreement"), a production payment dischargeable from the proceeds of production from or applicable to the interests of the Grantor (hereinafter called the "Leasehold Interests"), in those certain oil and gas leases covering lands located offshore Locassiana and Texas in the Gulf of Mexico, which interests, leases and lands are described in Exhibit "A" to the Conveyance and in Exhibit "A" to Appendix 1 to the Production Payment Agreement hereinafter referred to, in an amount equal to the aggregate principal amount of the loans, plus accrued interest and commitment fee, outstanding under the Bank Credit Agreement on the date, if any, upon which the purchase price therefor is paid by the Grantee to the Grantor as provided in that certain Production Payment Agreement dated as of March 1, 1974, as thereafter amended and restated as of May 1, 1976, between the Grantee and the Grantor (such Production Payment Agreement, as so amended and restated. It ag have inafter called the "Production Payment Agreement"), all as more particularly provided in the coduction Payment Agreement;

Whereas, in order to induce the banks referred to in Section 2 of the Bank Credit Agreement (hereinafter called the "Banks") to make loans pursuant to the Bank Credit Agreement and to secure the performance by the Grantor of its obligations mader the Bank Credit Agreement, the Grantor, by that certain assignment dated as of March 1, 1974 and confirmed by the Confirmation of Assignment dated as of May 1, 1976 (hereinafter called the "Assignment") from the Grantor to The First National Bank of Chicago, as agent for the Banks (hereinafter called the "Agent"), assigned, transferred and set over to the Agent for the proportionate brucht of the Loaks as their interests may appear all the right, title and inferest of the Grantor to be and under the Troduction Payment Agreement, subject to the everptions set forth in the Assignment and whath Assignment was also executed by the Grantee for the purpose of its making the representations and special agreements set out therein; and

Whenever, pursuant to the request of the Cranton and the Grantee, the Agent proposes to join herein for the limited purposes set forth below;

Now, Timuarona, in order to a 'd to the Leasehold Interests the interests of the Grantia in certain oil and passlesses covering Linds located offshore Lean time and Texas in the Gulf of Mexico adjacent to the United States of America, and consideration of the premises are the mutual covenants and agreements herein contained and intending to be legally bound, the parties hereto agree as follows:

- 1. The Conveyance is hereby a nended in the following acspects:
- (a) The first pare staph of the Conveyance is amended to read in its entirety as follows:

"THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Conveyance") dated as of March 1, 1974, as amended and restricted as of May 1, 1976, between Pennzou. Louisiana and Texas Offshow, Inc., a Delaware corporation (the "Crantor"), having its principal office and post office address at Pennzoul Place, P.O. Box 2967. Houston, Texas, and Pennzou. Company, a Delaware corporation (the "Grantee"), aving its principal office and post office address at Pennzou Place, P.O. Bex 2967, Plouston, Texas,".

- (b) Subparagraph (f) of Paragraph 1 of the Conveyance is amended by adding therein the clause", as amended and restated as of May 1, 1976," after the number "1974".
- (c) Subparagraph (a) of Paragraph 4 of the Conveyance is arounded by adding therein the clause", as thereafter amended and restated as of May 1, 1976," after the number "1974".
- (d) Paragraph 23 of the Conveyance is emended by changing the address "900 Southwest Tower" to "Pennzoii Place, P.O. Box 2967" at each occurrence therein.
- (e) The "In Witness Willemeor" paragraph of the Conveyance is amended to read in its entirety as follows:

"In WITNESS WHEREOF, this Restated Instrument of Conveyance of Production Payment amends and restates the Instrument of Conveyance of Production Payment dated as of March I, 1974, and the parties have caused this Restated Instrument of Conveyance of Production Payment to be duly executed in the presence of the undersigned witnesses on the dates of their respective acknowledgments annexed hereto, in several counterparts, each of which is an original and all of which are invertical. Each of such counterparts shall for all purposes be deemed to be an original, and all is a production together shall constitute but one and the same Instrument. The date of this Taxament shall be deemed for all purposes to be the date first above written."

- (f) The signatures and acknowledgments set forth in the Conveyance are amended to read in their entirety as set forth in Annex 1 hereto as the signatures and acknowledgments thereto.
- (g) Exhibit "A" to the Conveyance is anreaded to read in its cutirety as set forth in Annex I hereto as Exhibit "A" thereto.
- 2. Except as amended hereby, the provisions of the Conveyance shall remain in full force and effect. The Conveyance, as amended hereby, shall be restated to re. I in its entirety as set forth in Annex I hereto, it being agreed and under tood that all references in the Conveyance as amended hereby and so restated to the Bank Credit Agreement shall be deemed to be references to the Bank Credit Agreement as amended and restated as of May 1, 1976.
- 3. The Agent for and on behalf and in the rame of the Banks, he chy consents to the foregoing amendment to and restatement of the Corsa cancer, provided, however that notwithstanding any provision hereof to the contrary this Amendment shall not be construed in operate, except as expressly set forth above, to modify in any respect the rights and obligations of the Company or Pennzoil under the Production Payment Agreement or the Assignment.

1

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective officers thereunto duly authorized as of the day and year first above written.

Benefic Carmenseta Ronaldon

Pennzoil Louisiana and Texas Offshore, Inc.

By WM Brund J.

Seal

Pennzoil Company

Pennzoil Company

By WBULLister

Vice President

The First National Bank of Chicago, as Agent

Pany Car Resident

Pennzoil Company

[Seal]

THE STATE OF TEXAS COUNTY OF HARRIS

BE IT REMEMBERS: that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this of the day of May, 1976.

(Louisiana)

Before me appeared W. M. Brumla 1, Jr., to me personally known, who, being by me duly sworn, did say that he is Vice President of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W. M. Brumley, Jr. getting ledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared W.B. Wilkerson, to me personally known, who, being by me duly sworn, did say that he is Vice President of Pennzon Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W. & When we acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared W. H. Bumley, Jr. , known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me or, this day personally appeared W. B. W. Mercon, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of PENNZOIL Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WITNESS WITNESS, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 6th day of May, 1976.

Richion Haye Haiter. Notary Public in and for Harris County, Texas

My Commission Expires Jene 1, 1977

THE STATE OF ILLINOIS COUNTY OF COCK SS:

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 10 day of May, 1976:

(Louisiana)

Before me appeared Petred L. Lines, to me personally known, who, being by me duly sworn, did say that he is Vice-President of The First National Bank of Chicago, a national banking association, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that Lie Lie L. Evans acknowledged the instrument to be the free act and deed of the national banking association.

(Texas)

Before me on this day personally appeared Laberd L. Files , known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be an Vice-President of The First National Bank of Chicago, a national banking association, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed and in the capacity therein stated, and as the act and deed of said national banking association.

In WITNESS WHEREOF, I have beccunto set my hand and off stal notarial seal, in the City of Chicago, County of Cook, State of Illinois, this /C day of May, 1976.

Notary Public in and for Cook County, Ilisis

My Commission Expires

My Commission Expires April 9, 1979

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

PENNZOIL COMPANY

CONVEYANCE OF PRODUCTION PAYMENT

THIS INSTRUMENT OF CONVEYANCE OF PRODUCTION PAYMENT (this "Conveyance") dated as of March 1, 1974, as amended and restated as of May 1, 1976, between Pennzon. Louisiana and Texas Offshow, Inc., a Delaware corporation (the "Grantor"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas, and Pennzoil Company, a Delaware corporation (the "Grantee"), having its principal office and post office address at Pennzoil Place, P.O. Box 2967, Houston, Texas,

WITNESSETH:

Whereas, the Grantor desires to sell and the Grantee desires to parenase the Production Payment hereinafter described,

Now, Therefore, it is agreed by and between the Grantor and the Grantee as follows:

- Certain Definitions. The following terms shall be used in this Conveyance with the meanings given below:
 - (a) "Accounting Month" shall mean any monthly period commencing with and including the 21st day of any calendar month and ending on and including the 20th day of the next succeeding calendar month.
 - (b) "Effective Date" shall mean 7 o'clock, A.M., on the date of this Conveyance, according to the time then in effect at the location of each Subject Interest.
 - (e) "Exhibit A" shall mean Exhibit A attached to this Conveyance and hereby made a part hereof for all purposes.
 - (d) "Hydrocarbons" shall mean all oil, gas and other hydrocarbons or any combination of one or more of such substances appropriate to the context in which such term is used.
 - (e) "I ease" shall mean any oil and gas lease or sub-lease, described in Exhibit A.
 - (f) "Proceeds Commencement Date" shall mean 7 o'clock, A.M., according to the time then in effect at the location of each Subject Interest, on the date upon which the purchase price is paid by the Grantee to the Grantor as provided in the Production Pay and Agreement dated as of March 1, 1974, as amended and restated as of May 1, 1976, between the Grantor and the Grantee.
 - (g) 'Subject Hydrocarbons' shall mean all Hydrocarbons in and under, and which may be produced and saved from, and which shall accure and be attributable to, the Subject Interests from and after the Proceeds Commencement Date and throughout the period specified in this Conveyance to: the term of the Production Payment.
 - (h) "Subject Interests" shall mean the undivided percentage interests of the Grantor in the Leases, and, in addition thereto, each and every other kind and cheacter of right, title, claim or interest which the Grantor now has in, to or under the Leases, which interests or Leases are either specifically or generally described in Exhibit A, all as the same shall be enlarged by the

discharge of any payments out of production or by the removal of any charges or encumbrances to which any of the same are subject and any and all renewals and extensions of any of the same, but there shall not be included within the Subject Interests any rights which the Grantor now has or may hereafter obtain with respect to production attributable to the interests of nonconsenting parties under any operating agreement, unit operating agreement, contract for development or similar instrument.

2. Conveyance of the Production Payment. The Grantor, for valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, by these presents does hereby grant, bargain, sell, convey, assign, transfer and deliver unto the Grantee, as a production payment, an undivided eighty-five percent (85%) interest in the proceeds from the sale of the Subject Hydrocarbons; subject, however, to the provisions of Paragraph 3 of this Conveyance.

The aforesaid eighty-five percent (85%) interest in the proceeds from the sale of the Subject Hydrocarbons, as granted, bargained, sold, conveyed, assigned, transferred and delivered unto the Grantee by the terms of this Conveyance, shall be the "Production Payment."

To HAVE AND TO HOLD the Production Payment, together with all and singular the rights, privileges and appurtenances thereunto in any way belonging, unto the Grantee, its successors and assigns, forever.

- 3. Certain Provisions Governing Subject Hydrocarbons. All the provisions of this Conveyance shall be subject to the following:
 - (a) For the discharge of the Production Payment the Grantee shall look exclusively to the proceeds of Subject Hydrocarbons, and the Grantor shall not be liable for such discharge.
 - (b) The Production Payment shall not be dischargeable from the proceeds of any products resulting from any manufacturing, processing or refining operation, except to the extent of that portion of such proceeds which represents the fair market value at the well ead of the Subject Hydrocarbons used in such manufacturing, processing or refining operation.
 - (c) The Production Payment shall not be dischargeable out of any bonus which the Grantor shall receive for any Lease or assignment of any of the Subject Interests or out of any payments made to the Grantor in connection with the drilling or deferring of drilling of any well on any of the Subject Interests or in connection with any adjustment of any well and leasehold equipment upon unitization of any of the Subject Interests.
 - (d) There shall not be included in the Subject Hydrocarbons any Hydrocarbons unavoidably lost in the production thereof or produced and saved from any of the Subject Interests and used by the Grantor in conformity with good field practices for drilling and production operations (including ges injection secondary recovery, pressure maintenance, repressuring or cycling operations) conducted for the purpose of producing Subject Hydrocarbons from the Subject Interests or from any unit to which the Subject Interests are committed, but only so long as such Hydrocarbons are so used.
 - (c) So long as and to the extent that the same may be required by applicable laws and regulations, in the case of any Lease from the United States of America included in the Subject Interests, from which the average production of oil per well per day averaged on the monthly basis is 15 barrels or less, the obligation to pay and the right of the Grantee to receive the proceeds of oil produced from such Lease shall be suspended until said average production of oil per well per day exceeds said minimum amount, and such suspension shall apply separately to any zone or portion of such Lease segregated for computing government royalty.
- 4. Amount and Term of the Production Payment, The Production Payment shall continue and remain in full force and effect until such time as the Grantee shall have received and realized,

out of the proceed. • The sale of the Subject Hydrocarbons, free and clear of all development, operating, mining, process, and expenses of a ylar distribution of the following amounts:

- (a) The aggregate principal amount of leans, plus accrued interest and commitment fee, outstanding on the Proceeds Commencement Date under the Bank Credit Agreement, dated as of March 1, 1974, as thereafter ancerd in and restated as of May 1, 1976, among the Grantor, The First exitional Bank of Chicago and the other banks named therein (barcinafter called the "Primary Sum"); plus
- (b) An amount computed at a rate per annum (on the basis of a 365-day year) which shall be equal to 112% of the corporate base rate on 90-day commercial loans to its largest and most credit-worthy commercial borrowers in effect at The First National Bank of Chicago from time to time plus ½%, such rate to change automatically and from time to time effective as of the date of each change in such corporate base rate, from and including the Proceeds Commencement Date on the unliquidated balance from time to time of the Primary Sum (the first such computation to be made on the 21st day of the mouth next following the Proceeds Commencement Date for the period from and including the Proceeds Commencement Date to and including the 20th day of the month next following the Proceeds Commencement Date on the amount of such unliquidated balance on the Proceeds Commencement Date, and subsequent computations to be made monthly on the first day of each Accounting Month for the preceding Accounting Month on the amount of such unliquidated balance on the first day of such preceding Accounting Month); plus
- (c) An amount equal to the aggregate of all amounts which are paid by the Grantee to any state or political subdivision thereof on account of ad valorem, transfer, mortgage, gross production, gross receipts, income, profits, severance, occupation, sales, use, franchise and other taxes and assessments of any kind whatsoever, including penalties and interest, imposed or assessed with respect to or measured by or charged against or attributable to the Production Payment, the Subject Hydrocarbons or the proceeds thereof, together with an amount equal to interest on the unliquidated amount thereof at the rate of 7% per annum from the date of payment of each such amount by the Grantee; provided, however, that the Granter shall have the right at its expense to contest any such taxes or assessments in good f. "I on belight of the Grantee.

IT BEING THE INTENTION OF THE GRANTOR AND THE GRANTEE IN ENTERING INTO THIS CONVEY-ANCE that the Grantee shall be entitled to receive and realize out of the Subject Hydrocarbons the full aggregate sum of the amounts described in subparagraphs (a) and (b) of this Paragraph 4, free and clear of all costs and expenses with respect to the S is set Interests and the Subject Hydrocarbons (v. toch expenses and costs shall be borne by the Granton) and over and above all taxes, expenses and costs of the character described, specified or referred to in subparagraph (c) of this Paragraph 4.

- 5. Termination of the Production Payment. When the full aggregate sum of the amounts specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance (as increased as expressly provided by the terms of this Conveyance) has been received by the Grantee, the Production Payment shall forthwith terminate and all interest therein shall immediately revert to and become vested in the Granter to the same extent and with the same force and effect as if this Conveyance had not been made. The Grantee agrees that, upon such termination, it shall execute or cause to be executed, upon the request and at the expense of the Granter, such instruments as may be necessary or appropriate to evidence the termination of the Production Payment.
- 6. Application of the Proceeds of Subject Hydrocarbons. For all purposes of this Conveyance, the proceeds of Subject Hydrocarbons actually received by the Grantee prior to the close of business on 20th day of the month next following the Proceeds Communication Date shall be deemed to have been received and applied immediately after the opening of business on the 21st day of such month, and thereafter, each proceeds actually received by the Grantee after the opening of business on the first day of each Accounting Month and prior to the close of business on the last day of such Account-

ing Month shall be deemed to have been received and applied immediately after the operator of business on the first day of the next succeeding Accounting Month; provided, however, any date of application specified above shall be a Saturday, Sunday or legal holiday under the jurisdiction in which such proceeds are actually received by the Grantee, such proceeds shall be deemed to have been received and applied on the last business day next proceeds to be applied pursuant to subparagraph (a) of this Paragraph 6 shall nevertheless be the amount under subparagraph (b) of Paragraph 4 of this Conveyance accrued up to (but not including) such haturday, Sunday or legal holiday. Such proceeds shall be deemed to have been applied on the first day of each Accounting Month as follows:

- (a) First, to the amount referred to in subparagraph (b) of Paragraph 4 of this Conveyance accrued up to (but not including) the date of such application, and
- (b) S. and, to the amount referred to in subparagraph (c) of said Paragraph 4, to the extent then ascertained, and
 - (c) Third, to the reduction of the unliquidated balance of the Primary Sum;

provided, however, that, in the event such proceeds to applied on the first day of any Accounting Month shall be insufficient to cover the full amount specified in subparagram (a) of this Paragraph 6, the unliquidated balance of the Primary Sum shall forthwith be increased to the extent permitted by law) by an amount equal to the amount of such deficiency.

Within 10 days after the date of each such application by the Grantee of any proceeds of Subject Hydrocarbons, the Grantee will furnish or cause to be furnished to the Granter a written statement showing the application of such proceeds in accordance with the provisions of this Paragraph 6.

- 7. Marketing the Subject Hydrocarbons. The Grantor, at its expense, shall market, or cause to be marketed, the Subject Hydrocarbons for the best price reasonably available when marketed, and it shall not be necessary for the Grantee to join in any Sales Contracts (as hereinafter defined) or any amendment thereof. In the event the Grantor shall take any of the Subject Hydrocarbons for its own use, the Grantor shall pay to the Grantee, on or before the last day of the Accounting Month during which such taking occurs, a price equal to the fair market value at the wellhead of the Subject Hydrocarbons to taken.
- 8. Sales Contracts. The Grantor will duly perform all obligations binding on it under all contracts and agreements for the sale of the tributed Hydrocarbons or any portion thereof, whether presently existing or hereafter entered into (dissuch contracts and agreements being the "Sales Contracts"), in accordance with the terms becoff and will take all appropriate measures to enforce the performance under each of the Sales Contracts of the additions of the purchase thereunder. All Subject Hydrocarbons sold by the Granton, whether pursuant to the terms of the Sales Contracts or otherwise, shall be delivered by the Granton to the purchases thereof into the pipelines to which the wells producing such hulp at Hydrocarbons may be connected or to such other point of purchase as is reasonably required in the marketing of such Subject Hydrocarbons.
- 9. Withholding and Restitution of Proceeds of Subject Hydrocarbous, All obligations of the Granton hereunder shall be subject to the applicable rules and regulations of the Tederal Fower Commission. Butes permitted under the Natural Gas Act and sain substance of the Tederal Fower Commission. Butes permitted under the Natural Gas Act and sain substance of the Tederal Fower Commission in buded in Subject Hydrocarbon shall be controlling if verying from praces a tablished in Subject Contracts or it different from the fair market price at the wellhand. The Gaustic shall be ented at to use its reasonable discretion in making filings with the Federal Fower Commission affects. Subject Hydrocarbons, If any proceeds of Subject Hydrocarbons shall be withheld to may in a say video en, including, without limitation, withholding by the Grantor pending final approval of rate increases filed with the Federal Power Commission,

the Grantee shall not be deemed to have received or realized such proceeds until, and only to the extent that, such proceeds shall actually have been received by the Granice. If, at any time before or after receipt of the full agas egate sum specified in subparagraphs (a), (b) and (c) of Paragraph 4 of this Conveyance, the Grantee shall be compelled, for easy cason whatsoever, to make any paya or uns Conveyance, the Grantee snan the compened, for any cason wratsoever, to make any payment or restitution, or such payment or to fitution shall be made for the account of the Grantee, on nem or resumnon, or such payment of resonation state to made for the account of the organice, of any proceeds of Subject Hydrocarbons, which proceeds have already been applied pursuant to the provisions of Paragraph 6 of the Conveyance, the unliquidated balance of the Primary Sum shall forthwith be increased by the amount of the proceeds so paid over by or for the account of the Grantee plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties provided, however, that if at the time of such payment or restitution the entire Primary Sum shall have been liquidated, the Primary Sum shall be deemed not to have been so liquidated and the unliquidated balance of the timary Sum shall be thereupon deemed to be in the amount of the proceeds so paid over plus the amount of any payments made by or imposed against the Grantee in the nature of interest, damages and penalties.

- 10. Protection to Furchasers of Production. No person purchasing or taking or processing any Subject Hydrocarbons shall be required to take notice of, or keep informed as to, the discharge or description of the Production Payment until the actual receipt by such person of written notice
- 11. Payment of Taxes. The Greator will tender and pay, or cause to be paid, punctually, before they shall become clinquent (or a to any thereof which are being contested in good faith, advising such person of such discharge or termination. to the final determination of such contest), all ad valorem taxes (or taxes imposed in prompty to the man determinance of such contest), an an value of taxes unposed in licu thereof) and all transfer, mortgage, gross production, mining, severance, occupation, excise, sales, use, franchise, income and other taxes and assessments of every kind and character whatsoever (other than federal success taxes of the Grantee) imposed or assessed with respect to or measured by or charged against or attributable to the Subject Interests, the Production Payment or the Subject Hydrocarbons, or the proceeds thereof, or against the Grantee by reason of its acquisition or ownership of the Production Payment or against any assignce of the Production Payment by reason of his or its interest as assignce, together with any interest or penalty payable in connection therewith.
 - 12. Covenants of the Grantor. Until such time as the Production Payment has termir ited as provided in Paragraph 5 of this Conveyance, the Grantor will, at its expense and regardless of who
 - (a) The Subject Interests to be maintained and developed and continuously operated for may be the operator or operators of the Subject Interests, cause: the production of Hydrocarbons in a good and workmanlike manner and in accordance with one prometion of tryonocarouns in a room and workmanner manner and in accordance with sound field practices and all applicable federal, state and local laws, rules and regulations (except some non practices and an applicante residue, some and recarring, times and regulations (except those being contested in good faith): provided, however, if the Granter in good faith and in more being concess or in gover ratings provided, more year, it the crantor in good ratin and in conformity with sound held practices elects not to participate in any operation which is to be conducted under the terms of an operation agreement, unit operating agreement, contract for development of shalls to a more cut, for linto by the Grantot or its predecessors in title perfaming to the development and operation of any Subject Interest (or any portion thereof) and which allows any consenting party or parties to conduct nonconsent operations, then, during the period or period, in which the consenting party or parties are enlitted to receive Subject Hydrocarbons to provide in which the constitutes party or parties due construction terms of such instrument pending attributable to such imbged lulerest (or parties due red) by the terms of such instrument pending the recoupment or receipt of the sums determined the remider, the Grantee shall not be entitled to receive us be deemed to have ten arred (but only during such period or periods) proceeds from Subject Hydrocarbons, actributable to such Subject Interest (or portion thereof) to the extent affected thereby.
 - (b) All rentals royalt as and I halalities of any kind or nature, including, without In An actual towards are a parameter of any kind of mature, meaning, we not huntation, all habilities for labor mate hal, supplies and equipment incurred in, or arising from, mananan, an natures a too face, analy not supplied and repopulated measures in, or arising producing, treating, processing, the administration, operation or development of, or the gathering, producing, treating, processing,

storing, marketing or transporting of, the Subject Hydrocarbons, to be paid punctually when due, or, as to any thereof which are being contested in good faith, promptly after the final determination of such contest.

- (c) All machinery, equipment and facilities of any kind now or hereafter located on the Subject Interests and necessary or useful in the operation thereof for the production of dydrocarbons therefrom to be provided and to be kept in good and effective operating condition and all repairs, renewals, replacements, additions and improvements thereof or thereto needed for such purpose to be duly made or provided.
- (d) All necessary and proper steps to be taken diligently to protect and defend the Subject Interests, the Subject Hydrocarbons and the proceeds thereof against any adverse claim or demand, including, without limitation, the employment or use of counsel for the prosecution or defense of litigation and the contest, settlement, release or discharge of any such claim or demand.
- (e) Written notice to be given to the Grantee of every adverse claim or demand made by any person (including any government or governmental agency) affecting the Subject Interests, the Subject Hydrocarbons or the proceeds thereof in any manner whatsoever, or of any proceeding instituted with respect thereto.
- (f) The interest of the Grancor in each of the Subject Interests to be kept free and clear of liens, charges and encumbrances of every character, other than those which are of a character customarily found connection with comparable drilling and producing operations and which do not materially cersely affect the operation of the Subject Interests and other than those becoming effective only after termination of the Production Payment as provided in Paragraph 5 of this Conveyance.
- (g) Subject to the provisions of Paragraph 19 of this Conveyance, all Leases included in or relating to the interests described or referred to in Exhibit A and under which the Grantor is lessee, sub-lessee, or assignee to be maintained in full force and effect without amendment, modification or waiver of any provision of any such Lease which might materially adversely affect the rights of the Grantee under this Conveyance.

Anything contained in this Paragraph 12 to the contrary notwithstanding, the Grantor shall not be obligated to perform undertakings performable only by the operator and which are beyond the control of the Grantor, with respect to individual Subject Interests of which it is not the operator. In such case, however, the Grantor will use its best efforts to secure the performance of any such undertakings.

- 13. Access to the Subject Interests. The Grantor shall permit any one or more representatives designated by the Grantee to make any inspection of the Subject Interests at any reasonable time as the Grantee may reasonably request. In addition the Grantor shall furnish to the Grantee such detailed information as the Grantee may reasonably request in writing concerning the Subject Interests, the operation thereof and the production and sale of the Subject Hydrocarbons.
- 14. Pooling and Unitization. Anything herein to the contrary notwithstarding, the Grantor may, without the consent of the Grantee, pool, combine, communitize or unitize any Subject Interest (or any portion thereof) from time to time with other leavehold, mineral or other interests to form units when, in the judgment of the Grantor, it is necessary or advisable to do so in order to facilitate the orderly development or operation (including repressured pressured pressured parameters (excling, water flood and secondary recovery operations) of any Subject Interest (or any portion thereof). Forthwith, after the formation of any unit, the Grantor shall furnish to the Grantee a true copy of the pooling agreement, unit designation of any other instrument creating such unit, together with, upon represt of the Grantee, any operating greement relating thereto. The interest in any such unit according or attributable to any Subject Interest (or any portion thereof) included therein

shall be subject to the Production Payment in the same manner and with the same effect as though such unit were in existence on the Effective Date. The signature or joinder of the Grantee shall not be necessary to the instruments evidencing the pooling and unitization herein authorized.

- 15. Default by the Grantor and the Remedies of the Grentee. Should the Grantor default in the performance or observance of any covenant or agreement provided in this Conveyance to be performed or observed by the Grantor, and should such default continue unremedied for 30 days (or for such longer period as may be necessary to complete performance diligently commenced within 30 days) from the delivery to the Grantor of written notice of such default, the Grantee may, in addition to pursuing any or all other remedies available at law or in equity:
 - (a) Effect performance or observance of such overnant or agreement on behalf of and at the expense of the Grantor, in which event the Grantee may expend funds reasonably necessary to effect such purposes and shall be entitled to reimbursement therefor from the Grantor, tegether with interest on such amounts at the rate of 7% per annum from the date of such expenditure to the date of reimbursement.
 - (b) Upon written notice to the Grantor, succeed to all rights of the Grantor with respect to the possession, operation and development of the Subject Interests, and may use in connection therewith all of the appurtenant property, both real and personal, situated upon or used or held for future use in connection therewith, or in connection with the production, storing, transporting or sale of the Subject Hydrocarbons, and the Grantee shall have the right, on behalf of and for the account of the Grantor, to sell the Subject Hydrocarbons and the Grantor shall, upon demand, reimburse the Grantee for all amounts so expended by the Grantee, together with interest on such amounts at the rate of 7% per annum from the date of such expenditure to the date of reimbursement.
 - (c) Apply to a court of equity for the specific performance or observance of such covenant or in aid of the execution of any power herein granted and for the appointment of a receiver of the Subject Interests and the Subject Hydrocarbons.
- 16. Exercise and Termination of the Remedies of the Grantee. All rights and resides granted by the provisions of Paragraph 15 of this Conveyance and otherwise belonging to the scantee shall be cumulative and no exercise of any right or remedy shall preclude the exercise of any other right or remedy. Any right or remedy may be exercised on one or more occasions without exhaustion thereof. All rights and remedies shall terminate
 - (a) When the Production Payment is discharged and all amounts then due and payable to the Grantee pursuant to the terms of Paragraph 15 of this Conveyance (including amounts payable for interest) shall have been paid in tuli, or
 - (b) At any earlier time when all defaults of the Grantor shall have been remedied and all such amounts shall have been duly paid in full, without prejudice, however, to the exercise of any rights and remedies conferred upon the Grantor upon any subsequent failure of the Grantor to perform or observe any of the covenants or agreements herein provided to be performed or observed by the Grantor.
- 17. Reports to the Grantee. While the Production Payment remains in force and effect, the Granter will, at its expense, furnish to the Grantee within 45 days after the end of each calendar month, a report showing for such each adar month (i) the quantity of Subject Hydrocarbons produced and sold or taken by the Granton for its use and (ii) the gross proceeds a ceived from the sale, or attributable to such taking, of such Subject Hydrocarbons.
- 18. Warranty of Title. The Grantor covenants and warrants that at the time of the execution and delivery of this Conveyance, the Grantor has good and marketable title to the Subject Interests,

subject only to such exceptions, charges, liens or encumbrances, if any, as are set forth or referred to in Exhibit A; that the Grantor has the power and right to create and confer upon the Grantee the rights of the Grantee granted by the terms of this Conveyance, and that there are no suits or proceedings pending or, to the knowledge of the Grantor, threatened, against or affecting the Subject Interests before any court or by or before any goveranemal commission, bureau or any regulatory authority which if decided adversely to the interests of the Grantor shall materially affect the rights of the Grantee pursuant to this Conveyance. The Grantor hereby binds itself, its successors and assigns, to warrant and defend forever the rights under this Conveyance of the Grantee, its successors and assigns, against the lawful claims and demands of every person who shall claim the same or any loss suffered by the Grantor will indemnify and hold the Grantee harmless against and from all loss suffered by the Grantee resulting from any failure of or defect in the Grantor's title to the Subject Interests and the production of Hydrocarbons as warranted as aforesaid. The covenants and warranties of this Paragraph 18 shall survive the discharge of the Production Payment.

19. Transfer of the Subject Interests. Except for the conveyance of production payments becoming effective only after termination of the Production Payment - provided in Paragraph 5 of this Conveyance, prior to such termination the Grantor shall not, with at the written consent of the Grantee, make or permit any surrender, abandonment, sale, conveyance, assignment, lease or sublease, in whole or in part, of any of the Subject Interests; provided, however, that the Grantor shall have the right to surrender, abandon or otherwise terminate (either voluntarily or involuntarily) any of the Subject Interests (or any portion thereof) when there is no well located thereon which is capable of producing Hydrocarbons in paving quantities and when the Grantor has reasonably determined that (i) no existing well may be recompleted thereon as a well capable of preducing Hydrocarbons in paying quantities, (ii) no additional well could be dilled and completed thereon which would be capable of producing Hydrocarbons, over and above the cost of drilling and completing same, in paying quantities, and (iii) no receive maintenance cosecondary recovery operations may be instituted with respect to any well Percon which would cause such well to be capable of producing Hydrocarbons, over and above the cost of instituting and completing such pressure maintenance or secondary recovery operation, in paying quantities. For the purposes of this Paragraph 19, the words "in paying quantities" as applied at any time to the producing capability of any well shall mean that, in the light of conditions existing at the time of determination and which reasonably appear to be not temporary, such well is producing or is or will be capable of producing Subject Hydrovarbons whose aggregate value exceeds or will exceed the Grantor's share (as the owner of the Subject Interest on which such well is located) of the direct cost of operating such well.

In each case where the preceding paragraph of this Paragraph 19 permits the Grantor to surrender, abandon or otherwise terminate any of the Subject Interests (or any portion thereof), then (a) any of the Subject Interests (or any portion thereof) which are so surrendered, abandoned or terminated shall upon surrender, abandoning or termination cease to be a Subject Interest, and (b) the Grantor, in lieu of surrendering abandoning or terminating any of the Subject Interests (or any portion thereof), may sell, assign, sublease, farmout or convey the same free and clear of the Production Payment without the consect of the Granton, and the Grantee shall not have the right to any proceeds from such sale, assignment, sublease, farmout or conveyance; provided, however, that any interest in any such Subject Interest (or any portion thereof) which may be excepted from such sale, assignment, sublease, farmout or conveyance induction by the Grantor shall not be released thereby from the Production Payment.

20. Exception to Payments. The acceptance by the Grantee of any payment hereunder shall not preclade the Grantee from taking a sception to the correctness of the amount thereof; provided, however, that any such exception must be specified by the Grantee in a notice given to the Grantor within 12 months after the date such payment is made.

- 21. Transfer of the Production Payment. Nothing contained in this Conveyance shall be construed to limit or restrict in any way the right of the Grantee to sell, convey, assign or mortgage the Production Payment, in whole or in part.
- 22. Rights of Modgagee or Trustee. If the Grantee shall at any time execute a mortgage or deed of trust covering all or part of the Production Payment, the mortgagee or trustee therein named or the holder of any obligation secured thereby shall be entitled, to the extent such mortgage or deed of trust so provides, to exercise all of the rights, remedies, powers and privileges conferred upon the Grantee by the terms of this Conveyance and to give or withhold all consents required to be obtained hereunder by the Grantor, but the provisions of this Paragraph 22 shall in no way be deemed or construed to impose upon the Grantor any obligation or liability undertaken by the Grantee under such mortgage or deed of trust or under the obligation secured thereby.
- 23. Notices. Any notice, request, demand, report or other instrument which may be required or permitted to be given or furnished to or served upon either party hereto or other person succeeding to any interest of either party hereto shall be deemed sufficiently given or furnished and served if in writing and delivered to such party or person or to an officer of such party or person or deposited in the United States mail in a scaled envelope, registered, with postage prepaid, addressed if to the Grantor to Pennzoil Place, P. O. box 2967, Houston, Texas, and if to the Grantee to Pennzoil Place, P. O. Box 2967, Houston, Texas, or to such other address as the party or person to be addressed shall have designated by written notice to the party or person giving such notice or furnishing such report or making such request or demand.
- 24. Further Assurances. So long as permitted by applicable law so to do, the Grantor will execute and deliver all such other and additional instruments, notices, releases, acquittances and other documents, and will do all such other acts and things, as may be necessary more fully to assure to the Grantee all the rights and interests herein and hereby granted or intended so to be.
- 25. Binding Effect. All the covenants and agreements of the Grantor contained in this Conveyance shall be deemed to be covenants running with the land and shall be binding upon all successors in interest to, and all assigns of, the Grantor and shall inure to the benefit of all successors in interest to, and all assigns of, the Grantee. The provisions of this Conveyance shall inure to the benefit of and be binding upon the respective successors and assigns of the Grantor and the Grantee from and after acceptance by the Grantee. All references herein to either the Grantor or the Grantee shall include their respective successors and assigns.
- 26. Release or Failure of Title. No release by the Grantee of any part of the Subject Interests from the Production Payment, and no loss or failure of title to any part of the Subject Interests, shall have the effect of reducing the amount of the proceeds from the sale of production of Hydrocarbons out of which the Production Payment is dischargeable or reducing the amount of the Production Payment or creating any offset or other projudice to the Production Payment, and the Production Payment shall continue in full force and effect as to the remainder of the Subject Interests until the full amount of the Production Payment has been received by the Grantee as herein provided.
- 27. Final Termination. Notwithstanding the other provisions hereof, the Production Payment shall, in any event, branched 1 day prior to the expiration of 21 years after the death of the survivor of all the descendants of Theorem Roosevelt, late President of the United States, who are living on the date hereof.
- 28. Headings, Paragraph headings used in this Conveyance are for convenience only and shall not affect the construction of this Conveyance.

In Wirrass Wherever, this Restated Instrument of Conveyance of Production Payment amouds and restates the Instrument of Conveyance of Production Payment dated as of March 1, 1974, and the parties have caused this Restated Instrument of Conveyance of Production Payment to be duly executed in the presence of the undersigned witnesses on the dates of their respective acknowledge-

ments annexed hereto, in several counterparts, each of which is an original and all of which are identical. Each of such counterparts shall for all purposes be deemed to be an original, and all such concerparts together shall constitute but one and the same Instrument. The date of this Instrument shall be deemed for all purposes to be the date first above written.

| Witnesses: | PENNZOIL LOUISIANA AND TEXAS OFFSHORE, IMC. |
|--|---|
| Beneily Wapple | By WMY Tunky J. VICE PRESIDENT ATTEST: Sally Hay. ASSISTANT SECRETARY |
| · | [Corporate Seal] |
| Witnesses: Blevin & Stille Beverly Wagpler | PENNZOIL COMPANY By WBWWisson Vice President |
| | Sally Hay |

[CORPORATE SEAL]

THE STATE OF TEXAS
COUNTY OF HARRIS

BE IT REMEMBERED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certicy that, on this Cothe day of May, 1976:

(Louisiana)

Before me appeared W.M. Drumley, Ic., to me personally known, who, being by me duly sworn, did say that he is Vice President of Pannzen. Louisiana and Texas Offshore, Inc., a Deliware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W.M. Brownley, Jr. acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared W.E. Wilkerson, to me personally known, who, being by me duly sworn, did say that he is Vice President of Pennzoil Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W.B. Wilker and acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared W.F. Brandey, Tr. , known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Pennzon. Louisiana and Texas Offshore, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared W. Ew. Weekee , known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Pennzon. Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

IN WILNESS WHENEOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Herris, State of Texas, this & A.A. day of May, 1976.

Notary Public in a od for Harris County, Texas
My Commission Expires James, 1977

EXHIBIT A

TO

CONVEYANCE OF PRODUCTION PAYMENT

Grantor's

| BLM or State | Effective | Description | Undivided Percentage Interest |
|--------------|-----------------|---|--|
| Lease No. | Date / All E- I | 90 Htt | interest |
| | | eral leases are as shown on Official Continental Shelf Leasing Maps) | |
| OCS-G2839 | 12-1-74 | All of Block 352, West Cameron Arca, West Addition, OCS Official Leasing Map, Louisiana Map No. 1A | 25.00% |
| OCS-G2224 | 2-1-73 | Block 5th West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 26.67% |
| OCS-G2225 | 2-1-73 | Block 533 West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 26.67% |
| OCS-G3284 | 8-1-75 | All of Block 563, West Cameron Area, South Addition, as shown on OCS Official Leasing Map, Louisiana Map No. 1B | 25.90% |
| OCS-G2228 | 1-1-73 | Plock 590 West Cameron Area, South Addition, Official Leasing Map, Louisiana Map No. 1B | 23.20% |
| OCS-G2850 | 12-1-74 | All of Block 609, West Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 1B | 6.66667% |
| OCS-G2559 | 5-1-74 | All Block 617 West Cameron Area — South Addition, Official Leasing Map, Louisiana Map No. 1B | 63/3 % |
| OCS-G2859 | 12-1-74 | All of Block 236, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.66667% |
| OCS-G2860 | 12-1-74 | All of Block 237, East Cameron Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 2A | 16.66667% |
| OCS-G2882 | 12-1-74 | All or Plock 125, South Marsh Island Area, South Addition, OCS Official Leasing Map, Louisiana Map No. 3C | 6.66667% |
| OCS-G2983 | 12-1-74 | All of Block 127, South Marsh Island Area, South Addition, OC's Official Leasing Map, Louisiana Map No. 3C | 6.03667% |
| OCS-G2587 | 5-1-74 | All of Block 128 South Marsh Island Area, South Addition, Official Leasing Map, Louisiana Map No. 3C | 6%% |
| OCS-G2884 | 12-1-74 | All of Block 129, South Marsh Island Area, South Addition, OCS Official Leasing Map, | 100 00000000000000000000000000000000000 |
| OCS-G2885 | 12-1-74 | Louisiana Map No. 3C All of Block 141, South Marsh Island Area, South Addition, OCS Official Leasing Map, | 6.66667% |
| OCS-G2900 | 12-1-74 | Louisiana Map No. 3C All of Block 261, Eugene Island Area, OCS Official Leasing Map Louisiana Map No. 4 | 6.6666757 16.6666757 |
| OCS-G3156 | 7-1-75 | All of 1 - k 262. Furgine Island Area, as she in on OCS Official Leasing Map, Louistana Map No. 4 | 16.66667% |
| OCS-G2607 | 5-1-74 | All of Block 312 U gene Island Area, South Addition, Official Leasing Map, Louisiana | on Continue to Continue the Continue to Co |
| OCS-G2317 | 2-1-73 | Map No. 4A Block 333 Engene Island Area, South Addition, Official Leaving Map. Louisiana Map No. 4A —12— | 2634% 26.67% |

EXHIBIT A TO

CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| CONT | EIMICE OF | THOSection TAINEMI (Continued) | |
|---------------------------|-------------------|---|--|
| BLM or State Lanso No. | Effective Date | Description | Grantor's Undivided Percentess Interest |
| | 11.1.770 | | - |
| OCS-G2177 | 11-1-72 | All of Block 49 South Pass Area as shown on official leasing map, Louisiana Map No. 9 | 10.00% |
| State Lease No. 6310 | 5-33-74 | Tract 13018, being those portions of Blocks 57 and 58, South Pass Area, Plaquemines Parish, Louisiana, belonging to the State of Louisiana and not under mineral lease on March 4, 1974 and being more particularly described in Exhibit A to Assignment recorded in C.O.B. No. 406, Folio 465, Records of Plaquemines Parish, Louisiana | 20,72% |
| OCS-G2185 | 10-1-72 | Block 78, that portion which is more than 3 geographical miles seaward from the line described in paragraph 1 of the Supplemental Decree of the U.S. Supreme Court entered December 13, 1965 in the U.S. v. Louisiana No. 9 Original (382 U.S. 288), South Pass Area, South and East Addition, as shown on official leasing map, Louisiana Map No. 9A | 15.00% |
| OCS-G3193 | 7-1-75 | S½ of Block 59, Main Pass Area, that portion located in Zone 2 as that zone was defined in the agreement between the United States and the State of Louisiana, October 12, 1956 and landward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Official Leasing Map, Louisiana Map No. 10 | 25.00% |
| OCS-G3195 | 7-1-75 | Block 72, Main Pass Area, that portion located more than one foot scaward of the Third Supplemental Decree Line (404 U.S. 388 (December 20, 1971)), OCS Odicial Leasing Map, Louisiana Map No. 10 | 21.00% |
| OCS-G2947 | 12-1-74 | All of Block 73, Main Pass Arca, OCS Official Leasing Map, Louisiana Map No. 10. That portion located more than one foot seaward of the Third Supplemental Decree Line (404) | |
| OCS-G2191 | 10-1-72 | U.S. 388 (December 20, 1971)) All of Block 138 Main Pass Area as shown on official leasing map, Louisiana Map No. 10 | 21.00% |
| OCS-G2193 | 10-1-72 | All of Block 140 Main Pass Area as shown on official leasing map, Louisiana Map No. 10 | 10.00% |
| OCS-G2195 | 11-1-72 | All of Block 146 Main Pass Area as shown on official leasing map. Lanisiana Map No. 40 | 7.50% |
| OC3-G3202 | 7-1-75 | All of Block NG15E1-17, New Orleans, as shown on OCS Official Leasing Map, NH-15-12 | 21.00% |
| OCS-G3203 | 7-1-75 | All of Block NOTIE147, New Orleans, as shown on OCS Official Leasing Map, NII-15-12 | 21.00% |
| DCS-G3206 | 7-1-75 | All of Block No. 3E63, Mobile South No. 2, as shown on OCS Official Leasing Map NH-16-16 | 10.00% |
| CCS-G2351 | 8-1-73 | All of Block A-209, Galveston Area, South Addition, Official Leasing Map, Texas Map No. 6A | 20.00% |
| | | | |

EXHIBIT A TO CCNVEYANCE OF PRODUCTION PAYMENT (Continued)

Grantor's Unslivided

| BLM or State Lease No. | Effective Date | Description | Undivided Percentage Interest |
|---------------------------|-------------------|--|-------------------------------------|
| OCS-G2358 | 8-1-73 | All of Block A-442, High Island Area, South Addition, Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2366 | 8-1 73 | All of Block A-474, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.60% |
| OCS-G2372 | 8-1-73 | All of Block A-489, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-G2373 | 8-1-73 | All of Block A-496, High Lland Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G3118 | 4-1-75 | All of Block A-499, High Island Area, South Addition, as shown on OCS Official Leasing Map, Texas Map No. 7B | 16.40% |
| OCS-G2377 | 8-1-73 | All of Block A-519, High Island Area, South Addition. Official Leasing Map, Texas Map | 12.00% |
| OCS-G2378 | 8-1-73 | All of Block A-520, High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2383 | 8-1-73 | All of Block A. S. High Island Area, South Addition. Official Leasing Map, Texas Map No. 7B | 12.00% |
| OCS-G2701 | 7-1-74 | All of Block A-540, High Island Arca - South Addition, OCS Official Leasing Map, Texas | |
| OCS-G2734 | 7-1-74 | Map No. 7B All of Block A-545, High Island Area – South Addition, OCS Official Leasing Map, Texas Map No. 7B | 15.00% |
| OCS-C2779 | 10-1-74 | All of Block A-546, High Island Area, Scuth Addition, OCS Official Leasing Map, Texas | 14.00% |
| OCS-G2705 | 7-1-74 | Map No. 7B All of Block A-547, High Island – South Addi- tion, OCS Official Leasing Map, Texas Map No. 7P | S. Harris Christian |
| OCS-G2706 | 7-1-74 | All of Block A-548, High Island Area - South Addition, OCS Official Leasing Map, Texas | 14.00% |
| OCS-G2381 | 8-1-73 | Map No. 7B All of Block A-555, High Island Area, South Addition, Official Leasing Map, Texas Map | 14.00% |
| OCS-G2713 | 7-1-74 | No. 7B All of Flock A-562, High Island Area — South Addition, OCS Official Leasing Map, Texas | 14.50% |
| OCS-G2388 | 8-1-73 | Map No. 7B All of Block A-563, High Island Area, South Addition, Official Leasing Map, Texas Map | 12.715% |
| OCS G2389 | 8-1-73 | No. 7B All of Block A-564, Theh Island Area, South Addition. Official Leasing Map, | 12.00% |
| OCS-C2390 | 8-1-73 | Texas Map No 7B All of Blo UA 570 High Island Area, South Addition. Official Leasing Map, | 12.00% |
| | | Texas Niap No. 7B 14 | 20.00% |

EXHIBIT A TO CONVEYANCE OF PRODUCTION PAYMENT (Continued)

| | CONVEYANCE OF | PRODUCTION PAYMENT (Continued) | Grantor's |
|---------------------------|--|---|---|
| BLM or State Lease No. | Effective Date | Description | Undivided Percentage Interest |
| OCS-G2718 | 7-1-74 | All of Block A-575 High Island Area - | |
| | | South Addition, OCS Official Leasing | |
| | | Map, Texas Map No. 7B | 20.00% |
| OCS-G2719 | 7-1-74 | All of Block A-592 High Island Area - | 120 |
| | | South Addition, OCS Official Leasing | |
| | NA PARTITION AND | Map, Texas Map No. 7B | 14.118% |
| OCS-G2397 | 8-1-73 | All of Block A-272, High Island Area, | |
| | | East Addition, South Extension, Official | 15,000 |
| OCS-G2398 | 0.1.70 | Leasing Map, Texas Map No. 7C | 15.33% |
| OC3-G2398 | 8-1-73 | All of Block A-273, High Island Area, | |
| | | East Addition, South Extension, Official Leasing Map, Texas Map No. 7C | 12.00% |
| OCS-G2399 | 8-1-73 | All of Block A-274, High Islan I Area, | 2200 /10 |
| 000 02000 | 0 2 - 10 | East Addition, South Extension, Official | |
| | | Leasing Map, Texas Map No. 7C | 13.8461% |
| OCS-G2403 | 8-1-73 | All of Block A-279, High Island Area, | |
| | | East Addition South Extension. Official | |
| | | Leasing Map, Texas Map No. 7C | 17.00% |
| OCS-G3314 | 4-1-76 | All of Block A 288, High Island Area, | |
| | | East Addition, South Extension, as | |
| | | shown on OCS Official Leasing Map, | 117 0000 |
| OCS-G2409 | 0 1 70 | Texas Map No. 7C | 17.00% |
| OC3-G2405 | 8-1-73 | All of Block A-312, High Island Area, East Addition, South Extension. | |
| | | Official Leasing Map, Texas Map No. 7C | 12.60% |
| OCS-G2411 | 8-1-73 | All of Block A-314, High Island Area. | 12.00/0 |
| 000 02111 | 0.1.0 | East Addition, South Extension | |
| | | Official Leasing Map, Texas Map No. 7C | 22.85714% |
| OCS-G2737 | 7-1-74 | All of Block A-316 High Island Area → | • |
| | | East Addition - South Extension, | |
| | | OCS Official Leasing Map, Texas Map No 7C | 14.00% |
| OCS-G2413 | 8-1-73 | All of Block A-322, High Island Area, | |
| | | East Addition, South Extension, | 1.4.00.04 |
| OCS-G2414 | 8 1 72 | Official Leasing Map. Texas Map No. 7C | 14.00% |
| 003-62414 | 8-1-73 | All of Block A-323, Heth Island Area, East Addition, South Extension. | |
| | | Official Leasing Map, Texas Map No. 7C | 14.00% |
| OCS-G2416 | 8-1-73 | All of Block A-325, High Island Area. | 14.00/0 |
| OGD CEII | | East Addition South Extension. | |
| | | Official Leasing Map, Texas Map No. 7C | 20.86957% |
| OCS G2418 | 8-1-73 | All of Block A 327 High Island Area, | 100-0-100-0-100 (100-0-1 -0-1-) |
| | | East Addition, South Extension. | |
| | | Official Leasing Map, Texas Map No. 7C | 10.00% |
| OCS-G2/122 | 8-1-73 | All of Block A 332, High Island Area, | |
| | | East Addition, South Extension, | tooper |
| 000 00700 | 7174 | Official Leasing Map Treas Map No. 7C | 10.00% |
| OCS-G2739 | 7-1-74 | All of block A-339 High Island Area = East Addition - South Extension. | |
| | | OCS Official Leasing Map, Texas Map No 7C | 34.00% |
| OCS-G2426 | 8-1-73 | All of block A-340, High Island Area, | Carried / |
| | C0000000000000000000000000000000000000 | East Addition, South Extension. | |
| | | Official Leasing Map, Texas Map No. 7C | 34.00% |
| | | | |

- 15 --

EXHIBIT 4

| | CONVEYANG | LXIIIBIT A | |
|---------------------|------------------------|--|--------------------------------------|
| JA: or State | LIMNCE | OF PRODUCTION DE | |
| OCS-6,2429 | Effective Date | EXHIBIT A TO OF PRODUCTION PAYMENT (Continued | 1) |
| OCS-G2745 | 8-1-73 | | Giantor's Undivided Percentage |
| 003-132745 | 7.1.74 | All of Block A-351, Property of Area, Official Leasing Map, Texas Map No. 7C East Addition, Sci. Stand Area, All of Block A 355, The Stand Area Control of Block A 351, The Stand Area Control of Block A 351, The Stand Area Control of Block A 355, The Stand Area Control of Block A 3 | Interest |
| OCS-G2746 | 7-1-74 | OR S (1) 12 | 15.2.0 |
| OCS-G2766 | 7-1-74 | East Addition - South Bland Area - | 11 33% |
| This Conveya | nnce is subject to the | Ear Addition Shigh Island / ms Map No. 7C | 15.00% |
| dedireting the prop | the Subject Hy roca | therms and provisions of the state Map No 7C | 10.00cz |

This Conveyance is subjected the terms and provisions of the above-described oil and gas leases and accordingly, the Schieet Hydrocarbons and the proceeds therefrom shall be calculated after deducting the proportionale state of the royalties reserved in each such heave attributable to Grante is said undivided percentage int - st in such lease.

Lease Nos. OCS-G2539, G. G2225, UCS-G2550, OCS-G2559, OCS-G2607, OCS-G2317, OCS-G Lease Nos. OCS-G2830, G. G2225, OCS-G2850, OCS-G2559, OCS-G2607, OCS-G2317, O

Lease No. OCS-G2234 referred to above is subject to an Operating Agreement between Chevron Oil Company, as operator, and the Grantor, et al., as non-operator; Lease Nos. OCS-G254, OCS-G2528, OCS-G2559, OCS C2509, OCS-G2586

2587, OCS-G2884, OCS-G2885, OCS-G2900, OCS-G3156, OCS-G2185, OCS-G32 2557, OCS-C2566 OCS-G2372, OCS-G2900, OCS-G3150, OCS-G3150, OCS-G32

C2351, OCS-C2566 OCS-G2372, OCS-G311e, OCS-G2701 OCS-G2704, OCS-G2705, OCS OCS-G2706, OCS-G2364, OCS-G2713, OCS-G2715, OCS-G2701, OCS-G2704, OCS-G2704, OCS-G2705, OCS-G2705, OCS-G2705, OCS-G2705, OCS-G2719, OCS-G2705, OCS-G2426, OCS-G2584, OCS-G2715, OCS-G2588, OCS-G2715, ¹C.S.G2SS3, OCS-G to an operating agreement between the Grantee, as operator, and the Grantor, et al., as non-operators.

Lease Nos. OCS-G2177, OCS-G2191, OCS-G2193, OCS-G2154 and OCS-G3206 referred to above Lease Nos. OCS-G2177, OCS-G2171, OCS-G2171, OCS-G2173, OCS-G2173, and OCS-G2173 and OCS-G2174 and OC

Lease Nos. OCS-G2355, OC3-G2403 and OCS 3341 referred to above are each subject to an Operating A. a mean between Marathon Oil Company, as person, and the Granton et al., as

Lease Nov. OCS-G2373, OCS-G2377, OCS-G2378, OCS-G2411, OCS-G2737, OCS-G2413, OCS-C241-4, OCS C2416, OCS C2745 and OCS C. The referred is above are such subject to an Operating

Agreement between Gities Service Oil Company upon step and it the Grantor, et al., as non-operators, Lease Nos. OCS G2383, OCS G2301 OCS + 21 2 and OCS U2323 televied to a we are each the et to an Operating Agreement between Sun and Company (Delaware), as operator, and the Gran, w. et al as non-oper rs.

Lause Nov. OCS-C. Sal and OCS C. Vills referred to above are each subject to an Operating Agreement between Exon Corporation, as operator, and the Granton, Clab., Conor operators.

Lease Nov. OCS G.397 and OCS-G2395 referred to above are each subject to an Operating Agree-

Lease Nos. OCS Gestrand Cr.S-Cestrementer to above an each impect to an Operating Agreement between Burnah (2.1 Development, Live, assopication, and the Giovanne of also as non-operators. The Granton is terest, in Least Nov. Oc. S. Commission Const. Vi. 100, vi. at. as non-operators.

Res. Commission of Const. Commission The Granton in terests in Lease Are a restriction of the Granton in terests and the state of the above are subject to Advance Payment Views of the dated as of June 13, 1973, as an ended. between Granter and United Cas Pipe 1 ne Company.

DECEIVERD

PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

CONTIRM TION OF ASSIGNMENT AND DEZZARI 786

T IIS CONFIRMATION OF ASSIGNMENT dated as of May 1, 1976 (the "Confirmation"). between Pennzoil Louisiana and Texas Offshore, Inc. (the "Company"), and Theekird An off Sank of Chicago, as Agent for the Banks mentioned below (fire "Agent"),

WILNESSETH THAT.

"VILLELAS, the Company has entered into a Bank Credit Agreement (the "Credit Agreement"), dated as of March I, 1974, with The First National Bank or Chicago, the other Bank's referred to in Section 2 of the Credit Agreement and the Agent providing for loss (the "Loans") to the Company by the Banks which are parties to the Credit Agreement in an avegate principal a excess of \$100,005,700, to be evidenced by the promissory notes of the Company (the "Not, s"),

Witting s, the Company has executed, acknowledged and delivered to Pennzoli Company ("Pennzoil") a convey reas of production payment dated as a March I, 1974, also entially at the form of Appendix I (the "Co..., valice") to that certain Production Payment Agreeme a little I as of March 1, 1974, between Pen and the Company (the "Production Payment Agreen ut"), all as provided in Section 5.3 at 11 Production Payment Agreement.

WHEREAS, in order to sectore the full and complete performance by the Company of its or ligations under the Credit Agreement, the Company has agreed the following its rights and options in, to and under the Production Payment Agreement to cause Pennzeel John (a) to pay the Company, as the purchase price for the production payment conceved by the Company to Penrzoil as aforesaid, an amount equal to the principal amount of such production (a ment, which i incipal amount shall equal the aggregate principa" amount of the loans, plus accrue I interest ther on and commitment fe applicable thereto, if any, or standing under the Credit Agreement on the date, if any, upon wl 'ch - A payment 's made, or (b) to execute, acknowledge and deliver to the Company, at its expense, so a instrument or instruments as may be necessary or appropriate to release, relinquich and oxin a start fully discharged such production payment;

We all As, in order to induce the Banks which are partie, to the Credit Agreement to make loans pursuant to the Credit Agreement and to secure the performance by the Company of its obligations under the Credit Agreement, inc'oding its obligation to pay the Notes (the "Credit Agreement Obligations"), the Company has executed, acknowledged and delivered to the Agent an Assignment dated as of March 1, 1974 (the "Assignment"), a conformed copy of which is attached hereto as Armex I (in which Pennzoll joined for the limited purposes set forth therein), by which the Company assigned, transferred and set over to the Agent for the proportionate benefit of the Banks which are parties to the Credit Agreement as their interests may appear all the right, title and interest of the Company in to and under the Production Payment Agreement, save and except, however, those rights of the Con, no relating the sale of g.s produced from the Leasehold Interests provided for in Article III of the Production Payment Agreement (such right title and interest of the Company in the Production Payment Agreement, save and except such rights relating to the sale of gas, being herein and in the Assignment referred to as the "Production Payment Rights"), together with the right, title and interest of the Company in, to and under any amendm at at the Pro to tion Payment Rights or additions thereto,

Whereas, the Credit Agreement, the Fraduction Payment Agreement and the Conveyance have been amended and restated as of May 1, 1976 to reflect an extension of the ma crity of the credit available to the Company under the Credit Agreement, an increase in the amount of that credit to \$200,090,700 and the addition of Manufacturers Hanover Trust Company, Secur & Pacific National Bank, Wells Fargo Bink Sational Association, The Northern Trust Compan. European American Back & Trust Company and The First N Social Bank of Boston as parties to the Credit Agreement and fer other purposes, and

". In mays, pursuant to the request of the Company and the Banks which are parties to the Credit Agreement as amended and restated as aforesaid (the "Bands"), Pennzes, proposes to join betein for the limited purposes set forth below,

Now, Time cour, the Company and Pennzoil hereby confirm the continuing validity and effect reness in accordance with its terms of the Assignment, which has been heretolore made by the Company to enforce, secure and provide an additional mesos of payment of the Credit Agreemest Obligacions.

The Assignment, as confirmed hereby, (i) shall remain in full force and effect until payment in full of all relichtedness and obligations of the Company to each of the Ban's under and pursuant to the Gredit Agreement and the Notes executed and which it is be executed purse or thereto, at which time the Banks will upon request and at the expense of the Company release and reassign to the Company and Pemizol as their interests may appear the rights assire of to the banks bereamder as soon as there has been sun, seen a lapse of time that, in the opinion and Banks, the esanimuation of the security is cost under the Assignment is no longer necessary to potent the Books from loss or expense by rea on of any claim that the payment might constitute a preference or otherwise be subject to being socied, the shall be building upon the Company and Penazoil, their respective successors and assigns, (iii) hall invice to the benefit of and be enforceable by the Agent and the Bank their respective successors, transferees and assigns, and (iv) shall be deemed to have been made . The and shell be governed by the laws of the State of Texas. It is agreed and understood that a confirmed bereby to the Credit Agreemen!, the Production Cayment Agreement or the Conveyance sha!! be deemed to refer to such instruments as amended and restated as hereir above specified, and that all references to the Banks shall be decined to refer to the Banks which are parties to the Credit Agreement as so amended and restated.

In Witness William, "The parties acreto have caused this Confirmation to be executed by their respective officers thereinto duly authorized as of the day first above written.

PENNZOH LOUISIANA AND TEXAS OFFSHORE, INC.

[SEAL]

by What Stan Chy J.

Practi Company

[SEAL]

B, Wellesson

THE PHET NAME ALL BANK OF CHICAGO, AS Agent

[SEAL]

By Ruland L. Fran

The State of Tex's County of Harris (SS:

Bi Ir Riso to that I, the rederigned, a Notary Public duly qualified sworn and acting in and for the county and State aforesaid, berelay certify that, on the field day of May, 1976:

.Lani iana)

Before are appeared "Physiciality Is to me personally known, who, being by me duly sworn, did say that he is Vice President on Prinzen Louisiana 280. This Offshore, Isc., a Delaware corporation, and that the scal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and scaled in inhall of the corporation by authority of its bond of Directors and that a 227 in Major acknowledged the instrumen, to be the free act and deed of the corporation.

(Louisi ma)

Before me appeared were WAVES OF, to me personally known, who, being by me duly sworn, did say that he is Vice President of PLENZON, CENTENNY, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate of all corporation and that the instrument was signed and sealed in behalf of the corporation by an active of its Board of Directors and that w.B. Walkerson acknowledged the instrument to be the free acc and deed of the corporation.

(Texa.)

Before me on this day personally appeared. The Variable, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of PENZON LOCKIANA AND TEXAS OFFSIONE, INC., a Delawar corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Teans)

Before me on this day personally appears. W.B.JUARCE TO known to me to be the person whose to be a vice President of Pennzon. Contenty, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

In WHEES WHEREOF, I have hereunto set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this $b^{(E)}$ day of M_{eff} , 1976.

Killian Lage Hester. Notary Public in and for Parris County, Texas

My Commission Expires June 1, 1977

The State of Trinois | ss: County of County | ss:

Bi Ti Br (18tic) that I the undersigned, a Notary Public duly qualified, sworn and acting in and for the County (6d State aforesaid hereby critify that, on this Z* day of May, 1976;

(Louisiana)

Before me appears to the trace of the trace of the trace of the mean authority of its Board of Directors and that T, the trace of th

(Texas)

Between on the day personally appeared of the roll Livers whose themselves the person whose many is subscribed to the foregoing instrument and known to me to be an Virtham land and acknowledged as we that he except docaid instrument for the purposes and consideration therein expressed and in the capacity there is stated, and as the act and deed of said national banking association.

In Will ESS Williamor, I have hereunto set my hand and official notarial seal, in the City of Chicago, Co. 65 of Cook, State 64 Bilmois, this 25 day of Stay, 1976.

Of Carte (1) 1611 (5)
Notary Public in and for Cook County, Illinoise

My Commission Expires

My Commission Expires April 9, 1923

[CONFORMED COPY]

PENNZOIL LOUISIANA AND TEXAS OF ISHORE, INC.

ASSIGNMENT

THIS ASSIGNMENT dated as of March 1974 (the "Assignment"), from Pennzoil Louisiana and Texas Olfshore, Inc. (the "Company") to the First National Bank of Chicago, as Agent for the Banks mentioned below (the "Agent"),

WITNESSETH THAT:

Whereas, the Company has entered into a Bank Credit Agreement (the "Credit Agreement"), dated as of March I. '974, with The First National Bank of Chicago, the other Banks referred to in Section 2 of the Creat Agreement (the "Banks") and the Agent providing for loans (the "Loans") by the Banks to the Company in an aggregate principal amount not in excess of \$100,000,000, to be evidenced by the promissory notes of the Company (the "Notes");

Whenexs, the Company has executed, acknowledged and delivered to Pennzoil Company ("Pennzoil") a conveyance of production payment, dated as of March 1, 1974, substantially in the form of Appendix 1 to that certain Production Payment Agreement, dated as of March 1, 1974, between Pennzoil and the Company (the "Production Payment Agreement"), all as provided in Section 2.3 of the Production Payment Agreement;

WHEREAS, in order to secure the full and complete performance by the Company of its obligations under the Credit Agreement, the Company has agreed therein to assign its rights and options in, to and under the Production Payment Agreement to cause Pennzoil either (a) to pay to the Company, as the purchase price for the production payment conveyed by the Company to Pennzoil as afe esaid, an amount equal to the principal amount of such production payment, which principal amount shall equal the aggregate principal amount of the loans, plus accrued interest thereon and commitment fee applicable thereto, if any, outstanding under the Credit Agreement on the date, if any, upon which such payment is made, or (b) to execute, acknowledge and deliver to the Company, at its expense, such instrument or instruments as may be necessary or appropriate to release, relinquish and extinguish as fully discharged, such production payment; and

Whereas, pursuant to the request of the Company and the Banks, Pennzoil proposes to join herein for the limited purposes set forth below;

Now, Therefore, in order to induce the Banks to make loans pursuant to the Credit Agreement and to secure the performance by the Company of its obligations under the Credit Agreement, including its obligation to pay the Notes (the "Credit Agreement Obligations"), the Company hereby assigns, transfers and sets over to the Agent for the proportion to benefit of the Banks as their interests may appear all of the right, title and interest of the Company in to and under the Production Payment Agreement, save and except, however, those rights of the Company relating to the sale of gas produced from the Leasehold Interests provided for in Article III of the Production Payment Agreement (such right, title and interest of the Company in the Production Payment Agreement, save and except such rights relating to the sale of gas, being hereinafter referred to as the "Production Payment Rights"), together with the right, title and interest of the Company in, to and under any amendment of the Production Payment Rights or additions thereto.

- This Assignment is made by the Company to enforce, secure and provide an additional rate. I symment of the Credit Agreement Obligations.
- 2. The Production Payment Rights may be exercised by the Agent for and on hehalf of the 3 siks a son the occurrence of an Exent of Default, as described in Section 11 of the Credit Agreement as no covering the continuance of such Event of Default. To exercise the right of the

Company to call upon Penuzoil to pay the purchase price of such production payment as provided in Section 2.4 of the Production Payment Agreement (the "Purchase Price Right"), the Agent shall give Penuzoil notice to such effect which shall

- (a) specify the purchase price of the production payment (determined as provided in Section 2.1 of the Production Payment Agreement) to be paid by Pennzoil pursuant to such request, and
- (b) specify the date upon which the purchase price of such production payment is to be paid.

Concurrently with the giving of such notice to Pennzoil, the Agent shall deliver or mail an executed counterpart of such notice to the Company. Giving of such notice by the Agent to Pennzoil shall constitute an effective notice of election by the Company to exercise the Purchase Price Right as provided for in Section 2.5 of the Production Payment Agreement. The full amount of the purchase price of such production payment shall be paid by Pennzoil, without setoff, counterclaim or recompment whatsoever by reason of any claim of Fennzoil against the Company or otherwise, directly to the Agent on the date specified in such notice and shall be applied by the Agent to the payment of the Notes in the manner described in Section 6 of the Credit Agreement

- 3. Notwithstanding this Assignment, the Company and Pennzoil shall have the right, without the consent of the Agent or the Banks, to amend the Production Payment Agreement and the conveyance of production payment eve uted pursuant thereto for the sole and hinted purposes of releasing, or raining to, one or more of the Leaschold Interests or subjecting the same to gas sales, unit, operating or other agreements providing for the operation or development of the Leaschold Interests or the disposition of production therefrom, provided, however, that any such amendment shall not otherwise modify in any respect the rights and obligations of the Company or Pennzoil under the Production Payment Agreement or reduce the purchase price of said production payment determined as provided in Section 2.1 of the Production Payment Agreement.
- 4. No failure or delay on the part of the Agent or the Banks in exercising any power or right hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power preclude any other single or partial exercise thereof or the exercise of any other right or power hereunder.
- At the re-mest of the Company and in order to induce the Banks to make the Loans to the Company, Pennzoil agrees
 - (a) with the Company that each certificate as to value of Proven Hydrocarbon Reserves delivered by the Company to the Agent pursuant to Section S(e) of the Credit Agreement shall be signed by Fennzoil to acknowledge its approval thereof and shall be deemed to have been delivered to Pennzoil pursuant to Section 2.4 of the Production Payment Agreement, and
 - (b) with the Agent and the Banks that the breach or failure of the Company to perform any of its obligations under this Assignment of the Production Payment Agreement shall not reduce, impair or adversely affect in any respect the endorcability by the Agent and the Banks of (i) the Production Eavisor of Right are good to them becomed and (ii) the agreement of Pennzoil to pay directly to the Agent upon the date specified in the notice given to Pennzoil by the Agent pursuant to Panagasph 2 above an amount equal to the purchase purce of the production payment specified in such respectable or required adjustment as provided in Section 2.5 of the Production Payment Agreement,

it being the purpose and intent of Pennzoul berein to confirm to the Agent and the Banks the continued effectiveness validity and enforceability of the Production Payment Rights assigned to them hereunder throughout the term of the Assignment without regard to the performance or breach by the Company of its obligation, under this Assignment or the Production Payment Agreement.

- 6. Pennzoil hereby represents and warrants that
- (a) It is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. It is duly qualified to do business in those states in which it owns a material amount of property or transacts a material amount of business.
- (b) It has corporate power and authority to make and carry out the Production Payment Agreement and this Assignment, and all such action has been duly authorized by all necessary corporate producing on its part.
- (c) Since December 31, 1973 there has been no praterial adverse change in the financial condition of Pennzoil and its consolidated subsidiaries.
- (d) The Production Payment Agreement, the conveyance of production payment executed pursuant thereto and this Assignment have been de'v and validly executed and delivered by Pennaoil and constitute valid and legally binding agreements of Pennaoil enforceable in accordance with their terms, except as enforcement thereof is limited by bankruptcy, insolvency or other laws of general application relating to or affecting the enforcement of creditors' rights.
- (c) While this Assignment continues unreleased Pennzoil shall, if notified of the exercise by the Agent of the right to have the purchase price paid to the Agent, be unconditionally obligated to pay such purchase price directly to the Agent in accordance with the terms hereof.
- 7. All notices and other communications provided for herein shall be in writing and shall be delivered or mailed (or in the case of telegraphic communication, if by telegram, delivered to the telegraph company and if by the telex, graphic scanning or other telegraphic communications equipment of the sending party, delivered by such equipment) addressed, if to the Company or Pennzoil, as appropriate, at 900 Southwest Tower, Houston, Texas 77002, and if to the Agent, to it at One First National Plaza, Chicago, Illimost 60670. All notices and other communications given to any party hereto in accordance with the provisions of this Assignment shall be deemed to have been given when sent by registered or certified mail, if by mail, or when delivered to the telegraph company, charges prepaid, if by telegram, or when receipt is acknowledged, if by any telegraphic communications equipment of the sender, in each case addressed to such party as provided in this Paragraph 7 or in accordance with the latest unrevoked direction from such party.
- 8. This Agreement (i) shall remain in full force and effect until payment in full of all indebtedness and obligations of the Company to each of the Banks under and pursuant to the Credit Agreement and the Notes, at which time the Banks will, upon request and at the expense of the Company, release and reassign to the Company and Pennzoil, as their interests may appear, the rights assigned to the Banks hereunder as soon as there has been sufficient chapse of time that, in the opinion of the Banks, the continuation of the security interest under the Assignment is no longer necessary to protect the Banks from loss or expense by reason of any claim that the payment might constitute a preference or otherwise be subject to being voided, (ii) shall be binding upon the Company and Pennzoil, their respective successors and assigns, (iii) shall inuic to the benefit of and be enforceable by the Agent and the Banks, their respective successors, transferces and assigns, and (iv) shall be construed in accordance with the Lays of the State of Texas.

In WITNESS WHEREOF, the parties hereto have caused this Assignment to be executed by their respective officers thereunto duly authorized, and in the presence of the undersigned witnesses, as of the day first above written.

Witnesses: PENNZOIL LOUISIANA AND TEXAS OFFSHORE, INC.

R. C. BRYAN By s/ MARY FRANCES FORESTER 5/

Vice President

GEORGIA SPIVEY ATTEST:

> 5/ SALLY HAZEN Assistant Secretary

> > [CORPORATE SEAL]

Witnesses:

s/

PENNZOIL COMPANY MARY FRANCES FORESTER s/

W. E. GIPSON By s/ Vice President 5/ GEORGIA SPIVEY

ATTEST:

5/ SALLY HAZEN Assistant Secretary

[CORPORATE SEAL]

THE STATE OF TEXAS COUNTY OF HARUS

BE IT REMINIBURED that I, the undersigned, a Notary Public duly qualified, sworn and acting in and for the County and State aforesaid, hereby certify that, on this 21st day of March, 1974:

(Louisiana)

Before me appeared R. G. Bryxy, to me personally known, who, being by me duly sworn, did say that he is Vice President of Prinzon, Louisiana and Trans Officion, Inc., a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that R. G. Bryxy acknowledged the instrument to be the free act and deed of the corporation.

(Louisiana)

Before me appeared W. E. Gresos, to me personally known, who, being by me duly sworn, did say that he is Vice President of Fennzon Company, a Delaware corporation, and that the seal affixed to the foregoing instrument is the corporate seal of said corporation and that the instrument was signed and sealed in behalf of the corporation by authority of its Board of Directors and that W. E. Greson acknowledged the instrument to be the free act and deed of the corporation.

(Texas)

Before me on this day personally appeared R. G. Bryan, known to me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice President of Pennzou. Louisiana and Trias Offshour, Inc., a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

(Texas)

Before me on this day personally appeared W. E. Gipson, known (2) me to be the person whose name is subscribed to the foregoing instrument and known to me to be a Vice. President of Pennzon. Company, a Delaware corporation, and acknowledged to me that he executed said instrument for the purposes and consideration therein expressed, and as the act and deed of said corporation.

In Witness Witten or, I have become set my hand and official notarial seal, in the City of Houston, County of Harris, State of Texas, this 21st day of March, 1974.

5/

DOROTHY STEARMAN Dorothy Stearman

Notary Public in and for Harris County, Texas My Commission Expires June 1, 1975

BALLAY & BRAUD

A PROFESSIONAL LAW CORPORATION

513 BELLE CHASSE HIGHWAY NORTH

Belle Chasse, Louisiana 70037 · 2699

CHARLES J. BALLAY STEPHEN C. BRAUD TELEPHONE (504) 394-9841

July 18, 1985

United States Dept. of the Interior Minerals Management Service Gulf of Mexico OCS Region Imperia! Office Bldg. P. O. Box 7944 Metairie, LA 70010

Attn: LE-3-1 Ms. Boehm

RECEIVED

JUL 2.2 1985

i. ern entitied"

47Vi-2

Re: Lease Numbers OCS-G 2185 OCS-G 2360

Gentlemen:

Enclosed you will find an Act of Collateral Chattel Mortgage dated July 15, 1985. In order that third persons may be placed on notice as to the execution and efficacy of the Mortgage, please file one copy of this letter, with the attached duplicate original of the Mortgage in the files in your office relating to each of the above-captioned leases.

A check in the amount of \$50.00 is enclosed for this service.

Very truly yours,

BALLAY & BRAUD

Charles J. Ballay

CJB/jwb

Enclosures

Filing Accomplished As Requested Minerals Management Service

La Nelle Boehm

July 22, 1985

80164B.24

Date:

COLLATERAL CHATTEL MORTGAGE

BY: HuB ENERGY SERVICES, INC.

UNITED STATES OF AMERICA STATE OF LOUISIANA

PARISH OF PLAQUEMINES

BE IT KNOWN, that on this _ 15 day of July, in the year of Our Lord, one thousand nine hundred and eighty-five;

for the Parish of Plaquemines, State of Louisiana, and in the

PERSONALLY CAME AND APPEARED:

Hub ENERGY SERVICES, INC., a Louisiana corporation, domiciled and doing business in the Parish of Plaquemines, State of Louisiana, represented herein by its undersigned officer, as per the attached certified resolution.

hereinafter sometimes referred to as the mortgagor, who declared to me, Notary, that, desiring to secure funds from any person, firm or corporation willing to loan same, and for such purpose, said mortgagor does by these presents declare and acknowledge a debt in the sum of FIVE HUNDRED TWENTY FIVE THOUSAND AND NO/100 (\$525,000.00) DOLLARS, and to evidence such indebtedness, has executed, under date of these presents, one (1) certain promissory note for the sum of FIVE HUNDRED TWENTY FIVE THOUSAND AND NO/100 (\$525,000.00) DOLLARS, made payride to " of bearer, ine on demand at Delta Bank & Trust Company, s to bear interest at the rate of fifteen (15%) per which said :ntil paid, payable on demand and 25% attorney's fees, cent per which saide, ... aving been paraphed "Ne Varietur" by me, Notary, for identification with t...s act was delivered to the said mortgagor, who acknowledged receipt thereof. Said mortgagor further declared that said note would be negotiated for the purpose of raising funds as heretofore stated, and said mortgagor does acknowledge, by these presents, to be indebted to any future holder or holders of said note in the full amount thereof, together with interest, attorney's fees, insurance premiums, taxes and court costs, should they accrue.

In case the said note should be placed in the hands of an attorney-at-law to institute legal proceedings to recover the amount thereof or any part hereof,

M. O. B. 41 FOLIO 4/95

DY. CLERK OF COURT

in principal or interest, or to protect the interest of the holder or holders
thereof, or in case the same should be placed in the hands of an attorney for
collection, compromise or other action, the mortg for hereby agrees to pay the
fee of the attorney who may be employed for that purpose, which fee is hereby
fixed at 25% on the amount due or sued for or claimed or sought to be protected,
preserved or enforced.

Now, in order to secure the payment of the said indebtedness evidenced by the said note, together with all interest, attorney's fees, premiums of insurance, taxes and court costs, the said mortgagor does by these presents, specially mortgage, affect and hypothecate unto and in favor of any future holder of said note, whether the same be held as an original obligation or in pledge, the following described property:

A. One 350 HP Worthington Model OF6MU-3, balanced opposed compressor, serial no. A15311 with 7½ x 5½ x 5° cylinders, direct connected to a Waukesha Model 3711 engine, serial no 48985, located on a cilfield type skid, including inlet and intermediate scrubbers, intermediate and after cooling, pneumatic stainless steel panel and overhead galvanized cooler, serial no. 813815.

Located at South Pass Block 78, Platform "A", OCS-G-2185, Louisiana, U.S.A.

B. One Caterpillar Engine, serial no. 72B01062, Model 379 NA, direct connected to an Ariel Compressor, serial no. F2802, Model JGW-2 with two (2) 3 7/8" cylinders fabricated on an oilfield type skid, with cooler, serial no. 82145.

Located at High Island Block A-447, OCS-G-2360, Texas, U.S.A.

C. One Waukesha Engine, serial no. 326911, Model F-1197, direct connected to an Ariel Compressor, serial no. F1253, Model JG-2, with two (2) 3 5/8" cylinders, scrubbers, instrumentation and related piping fabricated on a concrete filled skid, with cooler, serial no. 783828.

Located at State Lease #4977, Plaquemines Parish, Louisiana.

Whenever any of the above described properties are not located at the locations designated above, it shall be located at the principal office of mortgagor, at 1202 Belle Chasse Hwy. South, Belle Chasse, Plaquemines Parish, Louisiana.

Mortgagor binds and obligates itself not to remove any of the property from its present said location without the prior written consent of the then holder or holders of the hereinabove described note, having first been obtained.

Together with all the improvements, appurtenances and attachments, thereunto belonging or in anywise appertaining, now or hereafter forming part of and attached to or connected with some property or used in connection therewith.

Said property to remain . ally mortgaged, affected and hypothecated unto and in favor of any future holder or holders of said note until the full and final payment thereof, said mortgagor being hereby obligated not to sell, alienate, deteriorate or otherwise encumber the said property to the prejudice of this act, and not to permit or suffer the same to be sold, alienated, deteriorated or encumbered.

And the said mortgagor does, by these presents, confess judgment "c" the amount as said note in principal, interest attorney's fees and all ther costs and charges, and does consent, agr - and stipulate that, in the event the said note or any interest thereon is not punctually paid at maturity, or upon the mortgagor's death, suspension, failure, liquidation or insolvency, or upon application for said mortgagors adjudication in bankruptcy, or upon the application by or against said mortgagor for a respite, assignment, composition, extension, reorganization or receivership, or in the event of the mortgagor's failure to comply with any obligation by said mortgagor herein undertake... anything herein contained to the contrary notwithstanding, it shall be lawful for and the said mortgagor does hereby authorize the then holder or holders of the said note without making a demand or putting the said mortgagor in default being hereby expressly waived, to cause and singularly the property herein mortgaged and hereinabove described, to be seized and sold under executory process or any other legal process at the option of the holder or holders of said note. said mortgagor especialy waives the benefit of any and all laws or parts of laws relative to the appraisement of the property seized and sold under executory process, or any other legal process, and consents that said property be sold without appraisement to the highest bidder for cash or on such terms as the plaintiff in such proceedings may direct.

Possession of the said note at any time by the said mortgagor herein shall not in any manner extinguish the said note or the present mortgage securing payment thereof, but the said mortgagor shall have the right to issue and reissue the said note from time to time and as often as convenience may require without in any manner extinguishing or affecting the obligation of said note or the security of this mortgage.

And the said mortgagor further agrees that, in the event any proceedings are undertaken under this mortgage by way of executory process or otherwise, any and all declarations of the facts made by authentic act before a Notary Public

and in the presence of two witnesses, by a person declaring that such facts lightly within his knowledge, shall constitute authentic evidence of such facts for that purpose of executory process. Said mortgagor hereby promises and agrees to keep the above described property constantly insured against the risk or loss by fire, wind, storm and tornado, in solvent insurance companies up to the full insurable value and to transfer and deliver the said policies of insurance to the holder or holders of said note, and hereby authorizes the holder or holders of said in te, in the event of a default with regard to said insurance, to cause said insurance to be effected at the cost of the mortgagor, at the then current rates, and the reimbursements of all sums paid for the purpose shall be secured by the privilege and mortgage aforesaid. However, this clause is not to be taken as making it obligatory upon the holder or holders of said note to cause said insurance to be effected upon the default of said mortgagor, or as imposing any liability upon the holder or holders of said note for failure to do so.

The mortgagor declares that there are no taxes due and eligible on the property herein mortgaged.

The maximum amount for which this act of mortgage shall be deemed to secure the obligations of the mortgagor, as herein stipulated, to reimburse any holder or holders of said note, the amounts paid for premiums of insurance or taxes is hereby fixed at 20% of the face value of said note.

The mortgagor hereby deciares that the property herein mortgaged stands registered in the name of the mortgagor, and that it has not been alienated by the said mortgagor, and according to the Chattel Mortgage certificate attached hereto, there are no mortgages or encumbrances against the said property.

The mortgagor further binds itself to pay all taxes and assessments that may be levied against the mortgaged property, and his failure to so do shall give the mortgagee the right to pay such taxes and assessments and the right to reimbursement of such sums from the mortgagee, which sums shall be secured by the mortgages aforesaid; to keep the hereinabove described property in constant repair and to exhibit same to the mortgagee at any time the mortgagee may desire; that in the event of foreclosure, mortgagee shall have the right to cause the hereinabove described property to be put in merchantable condition, and that the expenses so incurred shall operate as a further lien and privilege on said property, and that the costs thereof shall become a part of this mortgage and secured thereby; to notify mortgagee at once in the event that the

RESOLUTION

BE IT RESOLVED, that V. A. Bradshaw, Secretary-Treasurer of HuB Energy Services, Inc., be and is hereby authorized to borrow on behalf of this corporation the sum of FIVE HUNDRED TWENTY-FIVE THOUSAND AND NO/100 (\$525,000.00) DOLLARS at such interest rates, and upon such terms and conditions as he deems necessary and proper in his sole discretion, and to secure said loan with a Collateral Chattel Mortgage on the property described as follows:

A. One 350 HP Worthington Model OF6MU-3, balanced opposed compressor, serial no. A15311 with 7½" x 5½" x 5" cylinders, direct connected to a Waukesha Model 3711 engine, serial no 48985, located on a oilfield type skid, including inlet and intermediate scrubbers, in emediate and after cooling, pneumatic stainless steel panel and overhead galvanized cooler, serial no. 813315.

Located at South Pass Block 78, Platform "A", OCS-G-2185, Louisiana, U.S.A.

B. One Caterpillar Engine, serial no. 72B01062, Model 379 NA, direct connected to an Ariel Compressor, serial no. 2802, Model JGW-2 with two (2) 3 7/8" cylinders fabricated on an cilfield type skid, with cooler, serial no. 82145.

Located at High Island Block A-447, OCS-G-2360, Texas, U.S.A.

C. One Waukesha Engine, serial no. 326911, Model F-1197, direct connected to an Ariel Compressor, serial no. F1253, Model JG-2, with two (2) 3 5/8" cylinders, scrubbers, instrumentation and related piping fabricated on a concrete filled skid, with cooler, serial no. 783828.

Located at State Lease #4977, Plaquemines Parish, Louisiana.

BE IT FURTHER RESOLVED, that the said V. A. Bradshaw, Secretary/Treasurer, be and he is hereby authorized to execute any and all mortgages, promissory notes, instruments, assignments and pledges and/or documents necessary and proper to carry out the purposes and intents of this resolution, and that all of his aforesaid actions in connection therewith are hereby ratified and approved by the Board of Directors of this corporation.

This is to certify that the above and foregoing is a true and correct copy of a resolution unanimously adopted by the Board of Directors of HuB Energy Services, Inc., at a special meeting held by this corporation on the 15 day of July, 1985, and that said resolution is still in effect and has not been rescinded or amended as of this 15 day of July, 1985, at 10 o'clock A.m.

M. A. Bu Man

ATTESTY

WILLE HESS, PRESIDENT

within described property should be seized by a person the mortgagee.

And now to these presents personally came and intervened Darrell D. Roy, who on behalf of any future holder or holders of said note accepts this act of mortgage.

THUS DONE AND PASSED, in triplicate original, in Belle Chasse, Louisiana, on the day, month and year first above written, in the presence of the undersigned competent witnesses who hereunto sign their name with the said appearers and me, Notary, after a reading of the whole.

WITNESSES:

HUB ENERGY SERVICES, INC.

Malened Natherne Delta Bank & TRUST COM

DELTA BANK & TRUST COMPANY